U.S. DEPARTMENT OF TRANSPORTATION FEDERAL AVIATION ADMINISTRATION



Aviation Rulemaking Committee Charter

Effective Date: 10/5/2012

SUBJ: 14 CFR 21 / Safety Management Systems Aviation Rulemaking Committee

1. PURPOSE. This Charter creates the Aviation Rulemaking Committee (ARC) for Part 21 / Safety Management Systems (SMS) according to the Administrator's authority under Title 49 of the United States Code (49 U.S.C.) 106(p)(5). This charter also outlines the committee's organization, responsibilities, and tasks.

2. BACKGROUND.

On May 22, 2012, the Aircraft Certification Process Review and Reform ARC submitted a report to the FAA recommending that we undertake a review to update part 21 certification procedures to reflect a systems safety approach to product certification processes and oversight of design organizations. Design organizations must have full responsibility and accountability through the establishment of regulatory requirements for minimum qualification, performance, and management systems.

Consistent with FAA Order VS 8000.367, and the International Civil Aviation Organization (ICAO) Annex 8, the Aircraft Certification Service (AIR) has been actively developing and implementing an internal and external SMS. The initial focus was primarily on developing an internal set of processes, tools, and methodologies that facilitate the transition into the future state. AIR began that effort in 2005 and has made progress in defining key processes and tools. Later, with support from industry participants, the activities expanded to include development of standards for design and manufacturing organizations. Through implementation of pilot SMS projects with certain companies, the FAA is collecting information that will help define the scope of the SMS for Design Approval Holders (DAHs), validate certain best practices, and expand the knowledge base within the workforce and industry with respect to the essential elements of a robust SMS for manufacturers.

SMS requires a proactive approach to discovering and addressing hazards before they exhibit safety consequences. SMS also includes processes that seek to identify potential organizational breakdowns and necessary process improvements which allow management to address a safety issue before a noncompliant or unsafe condition results. SMS is not a substitute for compliance with FAA regulations or FAA oversight activities.

3. OBJECTIVES AND TASKS OF THE ARC. AIR wants to evaluate certain improvements to the effectiveness and efficiency of existing "certification procedures for products and parts," along with incorporating SMS in the design and manufacturing environment. This includes considering the effects of certain changes to the existing regulations, such as applicant qualifications, hazard (or safety) reporting, compliance assurance, and continued operation safety assurance systems for

Distribution:

all DAHs. The intent is to facilitate shifting towards a systems approach for DAHs that is similar to that used for production approval holder requirements, which involves a clear understanding of roles, responsibilities, and privileges. As part of this evaluation, we want to determine the best way industry and the FAA can effectively fulfill their respective compliance and safety responsibilities.

The ARC will provide a forum for the U.S. aviation community to discuss and provide recommendations to the FAA. The committee is expected to provide general information and guidance regarding proposed changes to part 21 and the AVS SMS program as it relates to design and manufacturing certificate and approval holders.

- a. The ARC will provide the FAA recommendations, which may include proposals for rulemaking, suggested processes, policies and guidance, and any further action it determines the agency should contemplate for part 21 to align with the SMS requirements documented in proposed 14 CFR part 5, which is the central component of the NPRM entitled Safety Management Systems for Part 121 Certificate Holders [Docket No. FAA-2009-0671; Notice No. 10-15].
- b. The ARC, serving in an advisory capacity, is expected to present and discuss whatever input, guidance, and recommendations its members consider critical to the FAA's ultimate disposition, development, and implementation of proposed regulatory requirements and related guidance and policy as necessary to the future direction for part 21 to include applicant pre-qualifications, approval holder recognition, and SMS considerations.
- c. The ARC will also consider proposed revisions to clarify and update engineering/designoriented regulatory requirements to part 21. In support of design certification and continued airworthiness, the evaluation should include improvements in the areas of:
 - 1. Application process
 - 2. Applicant qualifications
 - 3. Standardized certification criteria
 - 4. Identifying design approval holder responsibilities and privileges
 - 5. Clarifying continued airworthiness requirements
 - 6. Clarifying design approvals needing Instructions for Continued Airworthiness
 - 7. Clarifying TSO design approval processes
 - 8. Process definition for determining eligibility of U.S. surplus military aircraft in the restricted category

This proposal additionally corrects regulatory language, implements editorial changes for clarification, and standardizes regulatory language to reflect the global aviation environment. While this information will be shared with the ARC, responses to "clean-up" proposals are not required as part of the deliverables.

d. Proposed part 5 and International Civil Aviation Organization (ICAO) Annex 8 and Annex 19 (draft) serve as the foundation for the ARC's consideration regarding how the FAA will address its responsibilities for developing and implementing SMS requirements and the management and oversight of its regulated product/service providers. The ARC must respect the framework outlined in proposed part 5 and the ICAO Annexes when it provides the FAA recommendations with respect to application of SMS. However, the FAA will consider proposed changes to part 5 as deemed necessary from a design and manufacturing perspective.

Recommendation Report. The ARC shall make recommendations and submit a report addressing the following:

- a. Improvements, which may include proposals for rulemaking, processes, policies and guidance for 14 CFR part 21 that reflect a systems approach for safety. This will promote an effective and efficient certification process, which includes considering the effects of certain changes to the existing regulations, such as:
 - 1. Minimum qualifications and organizational requirements for design approval applicants and holders including responsibilities and privileges
 - 2. SMS for design approval holders
 - 3. Compliance assurance
 - 4. Continued operational safety assurance
 - 5. Hazard reporting
- b. Cost and benefit and other impact information in support of developing the required Regulatory Evaluation(s) and Regulatory Flexibility economic analysis for applying any proposed changes to 14 CFR part 21 FAA certificate and approval holders. Cost and benefit analysis should include information obtained through the AIR SMS pilot project and should identify the specific areas of impact and present this information in quantitative terms to the extent possible.
- c. Part 21 design and production approval holder organizations to which the proposed SMS requirements should apply, taking into consideration cost and benefit information as well as public comments to the part 5 NPRM and the SMS-ARC *Design and Manufacturing Working Group Report High-Level Recommendations for SMS Requirements* dated March 12, 2010.
- d. Changes to the FAA oversight methodology based on any recommendations for changes to part 21 that takes into account existing FAA processes and oversight and delegation programs for design and manufacturing related certificates and approvals and authorizations.
- e. Definitions and processes to be included in advisory, policy, and procedures material for addressing safety risk management responsibilities within a design and/or manufacturing organization. These definitions and processes should include:
 - 1. An operational definition of a "hazard" throughout the life cycle of a product in safety risk management.
 - 2. Definition of the term "organization" with respect to design and production approval holders to identify the limits of applicability of proposed SMS requirements, in

consideration of the broad range of organizational structures and activities within modern design and/or manufacturing organizations.

- 3. Hazard identification procedures.
- 4. Processes for the determination of acceptable safety risk.
- 5. Procedures to be included in advisory, policy, and procedures material for addressing safety assurance responsibilities within a design and/or manufacturing organization, including specific recommendations regarding "employee reporting systems".

The Director of Aircraft Certification Service (AIR-1) may propose additional tasks as necessary in support of a potential part 21 rulemaking action. The ARC may also request that AIR-1 add other tasks deemed relevant to the success of this initiative.

4. ARC PROCEDURES

- a. The ARC advises and provides written recommendations to AIR-1 and acts solely in an advisory capacity. Once the ARC recommendations are delivered to AIR-1, it is within his/her discretion to determine when and how the report of the ARC is released to the public.
- b. The ARC may propose additional tasks as necessary to AIR-1 for approval.
- c. The ARC will submit a report detailing recommendations within 18 months from the effective date of this charter. The chair of the ARC sends the recommendation report to both AIR-1 and the Director of the Office of Rulemaking.
- d. The ARC may reconvene following the submission of its recommendations for the purposes of providing advice and assistance to the FAA, at the discretion of AIR-1, provided the charter is still in effect.
- 5. ARC ORGANIZATION, MEMBERSHIP, AND ADMINISTRATION. The FAA will establish a committee of members of the aviation community. Members will be selected based on their familiarity with 14 CFR part 21, Safety Management Systems analysis, and regulatory compliance. Membership will be balanced in viewpoints, interests, and knowledge of the committee's objectives and scope. ARC membership is limited to promote discussion. Active participation and commitment by members will be essential for achieving the ARC's objectives. Attendance is essential for continued membership on the committee. When necessary, the ARC may set up specialized work groups that include at least one ARC member and invited subject matter experts from industry and government.

This ARC will consist of members from U.S. and foreign industry including representatives from designers and manufacturers holding part 21 certificates and approvals and other private sector aviation industry associations and advocacy groups. Invited foreign authorities and International Civil Aviation Organization (ICAO) representatives provide a valuable perspective from the global aviation community. These representatives are encouraged to fully participate in committee discussions; however, their participation does not include voting privileges on committee issues. The FAA's participation and support for the ARC will come from all affected lines-of-business.

- a. The ARC sponsor is AIR-1 who:
 - 1. Appoints members or organizations to the ARC, at the Director's sole discretion;

- 2. Selects the industry chair(s) from the ARC membership;
- 3. Selects the FAA's designated federal official for the ARC;
- 4. Receives all ARC recommendations and reports; and
- 5. Provides administrative support for the ARC through the Safety Management Design and Analysis Branch (AIR-150).
- b. Once appointed, the industry chair(s) will:
 - 1. Coordinate required committee and subcommittee (if any) meetings in order to meet the ARC's objectives and timelines;
 - 2. Provide notification to all ARC members of the time and place for each meeting;
 - 3. Ensure meeting agendas are established and provided to the committee members in a timely manner;
 - 4. Keep meeting minutes;
 - 5. Perform other responsibilities as required to ensure the ARC's objectives are met; and
 - 6. Provide status updates in writing to AIR-1 at 6 months and 12 months from the effective date of this charter.
- 6. COST AND COMPENSATION. The estimated operating cost (including *pro rata* share of salaries of FAA employees) to the Federal Government for this ARC is approximately \$400,000 annually. All travel costs for government employees will be the responsibility of the government employee's organization. Non-government representatives serve without government compensation and bear all costs related to their participation on the committee.
- 7. PUBLIC PARTICIPATION. ARC meetings are not open to the public. Persons or organizations outside of the ARC who wish to attend a meeting must get approval in advance of the meeting from a committee co-chairperson or designated federal official.
- 8. AVAILABILITY OF RECORDS. Consistent with the Freedom of Information Act, Title 5, U.S.C., section 522, records, reports, agendas, working papers, and other documents that are made available to or prepared for or by the committee will be available for public inspection and copying at the FAA's Office of the Director, Aircraft Certification Service (AIR-1), 800 Independence Avenue SW, Washington, DC 20591. Fees will be charged for information furnished to the public according to the fee schedule published in Title 49 of the Code of Federal Regulations, part 7.

You can find this charter on the FAA Web Site at: http://www.faa.gov/about/committees/rulemaking/.

- **9. DISTRIBUTION.** This order is distributed to director-level management in the Office of the Associate Administrator for Aviation Safety, the Office of Aviation Policy and Plans, the Office of Rulemaking, and the director- and division-level management in the Aircraft Certification Service.
- **10. EFFECTIVE DATE AND DURATION.** This committee is effective upon issuance of this charter. The committee shall remain in existence for 2 years, unless sooner terminated or

extended by the Administrator.

The effective date of this charter is October 5, 2012.

٠

Michael P. Huerry Acting Administrator



US Department of Transportation Federal Aviation

Administration

January 14, 2015

Mr. Walter Desrosier Part 21 SMS ARC Co-Chair 1400 K Street NW, suite 801 Washington, DC 20005

Dear Mr. Desrosier:

On behalf of the Aircraft Certification Service I would like to thank you for serving as the industry chairperson of the Part 21/Safety Management Systems (SMS) Aviation Rulemaking Committee (ARC).

The Federal Aviation Administration (FAA) acknowledges receipt of the ARC report dated October 5, 2014 and would like to extend our gratitude to all those who participated on the committee. The recommendations made by the ARC will be very helpful in assisting the FAA to address certification process improvements recommended by the Aircraft Certification Process Review and Reform (ACPRR) ARC.

We appreciate the thoughtful consideration of how to shift towards a systems approach for product certification including revisiting the recommendations submitted by the Certified Design Organization (DO) ARC in May 2008. We agree with the recommendation to use a "building block" approach for implementing DO. The FAA looks forward to working with industry on the near term concepts of including SMS for design and manufacturing organizations and recognition of voluntary systems development for compliance. We concur with the ARC recommendation to consider future rulemaking to establish regulatory requirements for issuance and oversight of certificated DOs after successful implementation of the necessary building blocks.

Again, thank you for your time, effort, and leadership you devoted to this committee.

Sincerely,

alounde al Bale

Dorenda D. Baker Director, Aircraft Certification Service Federal Aviation Administration



US Department of Transportation Federal Aviation

Administration

January 14, 2015

Mr. Walter Desrosier Part 21 SMS ARC Co-Chair 1400 K Street NW, suite 801 Washington, DC 20005

Dear Mr. Desrosier:

On behalf of the Aircraft Certification Service I would like to thank you for serving as the industry chairperson of the Part 21/Safety Management Systems (SMS) Aviation Rulemaking Committee (ARC).

The Federal Aviation Administration (FAA) acknowledges receipt of the ARC report dated October 5, 2014 and would like to extend our gratitude to all those who participated on the committee. The recommendations made by the ARC will be very helpful in assisting the FAA to address certification process improvements recommended by the Aircraft Certification Process Review and Reform (ACPRR) ARC.

We appreciate the thoughtful consideration of how to shift towards a systems approach for product certification including revisiting the recommendations submitted by the Certified Design Organization (DO) ARC in May 2008. We agree with the recommendation to use a "building block" approach for implementing DO. The FAA looks forward to working with industry on the near term concepts of including SMS for design and manufacturing organizations and recognition of voluntary systems development for compliance. We concur with the ARC recommendation to consider future rulemaking to establish regulatory requirements for issuance and oversight of certificated DOs after successful implementation of the necessary building blocks.

Again, thank you for your time, effort, and leadership you devoted to this committee.

Sincerely,

alounde al Bale

Dorenda D. Baker Director, Aircraft Certification Service Federal Aviation Administration

A Report from the Part 21/Safety Management Systems (SMS) Aviation Rulemaking Committee to the Federal Aviation Administration

Recommendations on Certification Procedures for Products and Parts

October 5, 2014

Prepared for:

Dorenda D. Baker, Director Aircraft Certification Service Federal Aviation Administration Washington, DC

TABLE OF CONTENTS

Lett	er Fro	m the ARC Co-Chairsv				
Exe	cutive	Summary vii				
1.0	1.0 Background					
	1.1	Introduction1				
	1.2	ARC Scope and Structure 1				
	1.3	Rulemaking Process				
	1.4	Previous Reviews and Recommendations				
2.0	Cu	rrent State				
	2.1	FAA and Industry Share the Responsibility for Current Safety Success				
	2.2	Delegation				
	2.3	Current Oversight Process				
	2.4	Demand for FAA Resources				
	2.5	Economic Impact of Aviation Industry				
3.0	Fu	ture State				
	3.1	Case for Change				
	3.2	Vision for the Future				
	3.3	Performance-Based Oversight (PBO) Recommendations 17				
4.0	Hig	gh-Level Recommendations & Implementation 19				
	4.1	Systems Approach to Certification				
	4.2	Safety Management Systems Requirements				
	4.3	Evolution of Oversight				
	4.4	Part 21 Miscellaneous Cleanup and TSO Modernization				
5.0	SM	SMS Recommendations and Requirements				
	5.1	SMS Implementation				
	5.2	Other Party SMS Accreditation				
	5.3	Additional SMS Recommendations				
6.0	De	sign Organization Model and Framework				
	6.1	DO and Organization Control Systems (OCS)				
	6.2	Compliance Assurance System				
	6.3	Safety Management System				
	6.4	DO Project Applications and Activity Reporting				

	6.5	DO Relationships and Supplier Control	47			
	6.6	Business Structure Variation	51			
	6.7	Scope, Limitations, and Privileges of DO Certificates	51			
	6.8	DO Approval of Data	60			
7.0	Otl	her Processes for Small Businesses	67			
	7.1	Accredited Organization	67			
	7.2	Agent Design Organization	68			
	7.3	Modified Current Model—Enhanced Showing	69			
8.0	Mi	nimum Design and Production Approval Applicant/Holder Requirements	70			
	8.1 Overs	Minimum Design Approval Applicant/Holder Requirements in Support of SMS a ight Objectives	und 70			
	8.2	Changes to § 21.3 Reporting to Reflect SMS	75			
9.0	Par	rt 21 Cleanup & TSO Modernization	84			
	9.1	Part 21 Cleanup	84			
	9.2	Part 21 Recommendations for TSO Program	87			
10.0) Bu	ilding Block Approach to Design Organization	91			
	10.1	Evolution of FAA Oversight: Centralized and Performance-Based	92			
	10.2	Accountability Framework/Applicant Showing Compliance Assurance	. 100			
	10.3	Systems Engineering	. 104			
	10.4	Optimum Use of ODA	. 106			
	10.5	Bilateral Considerations for Products and Articles Certification	. 107			
11.0) Ce	rtification Cost/Benefit Overview	. 110			
12.0	D Dis	ssenting Opinions	. 113			
App	pendix	A—Part 21/SMS ARC Membership	A-1			
App	pendix	B—Glossary	.B–1			
App	pendix	C—Acronyms	.C–1			
App	pendix	D—Part 21/SMS ARC Charter	D-1			
App	pendix	E—Key Considerations Supporting ARC Recommendations	.E–1			
App	pendix	F—Design Organization Working Group Report	. F–1			
Appendix G—Safety Management Systems Working Group ReportG-1						
App	Appendix H—Oversight Working Group Report					
App	Appendix I—TSO Subteam Working Group ReportI-1					
App	pendix	J—Cost-Benefit Analysis Working Group Report	J–1			

Appendix K—SM ICG Findings on SMS Equivalence	K–1
Appendix L—EASA DO Approval Certificate	L–1
Appendix M—Responsibilities of Exporters	M–1
Appendix N—Applicant Declaration and Holder Obligations	N–1
Appendix O—Accountability Framework	0–1
Appendix P—Use of Applicant Showings on ODA Projects	P–1

LETTER FROM THE ARC CO-CHAIRS

October 5, 2014

Ms. Dorenda Baker Director Aircraft Certification Service Federal Aviation Administration 800 Independence Avenue, SW. Washington, DC 20591

Dear Ms. Baker,

On behalf of the Part 21/Safety Management Systems (SMS) Aviation Rulemaking Committee (ARC), we are pleased to submit the enclosed report, which provides recommendations to improve the overall effectiveness and efficiency of the certification procedures in Title 14, Code of Federal Regulations (14 CFR) part 21, Certification Procedures for Products and Parts, by updating regulations and policies to reflect a systems safety approach to product certification and FAA oversight of design organizations (DO).

This report directly addresses the Aircraft Certification Process Review and Reform (ACPRR) recommendation to Congress to review and update part 21and provides detailed considerations of minimum qualification and organizational requirements for design approval applicants and holders including responsibilities and privileges. The ARC also considered other key ACPRR recommendations on the importance of ensuring a comprehensive means for implementing certification process improvements and change management planning to prepare both industry and the FAA workforce for their new and evolving roles and responsibilities in a systems safety approach to certification and oversight.

The ARC's goal was to determine the best way the FAA and industry can effectively fulfill their respective compliance and safety responsibilities while improving the efficiency and robustness of the certification process.

In this report, the ARC provides four high-level recommendations:

- Phased implementation of a systems approach to certification.
 - o Promote accountability framework and enhanced applicant showings.
 - o Establish minimum requirements for design approval applicant/holder.
 - Establish requirements for voluntary certificated Design Organizations (DO).
- Application of SMS requirements to design and production approval holders.
- Evolution of FAA oversight toward performance based systems safety (SMS) approaches.
- Part 21 miscellaneous cleanup and TSO modernization.

An overview of these high-level recommendations and discussion of the ARC's proposed phased implementation through a building block approach is provided in section 4 of this report. The remaining sections of the report provide detailed discussion on the issues and concepts considered by the ARC and the basis for its findings and high-level recommendations. It also provides "key considerations," which are supporting detail level recommendations necessary for practical implementation, summarized in appendix E.

On behalf of the Part 21/SMS ARC, it has been a privilege to be selected to undertake this important initiative. We are confident the ARC recommendations, when implemented, will result in a safer, more effective, and more efficient certification process for both the FAA and industry. Furthermore, the changes recommended have the potential to facilitate increased international cooperation and efficiencies, strengthening the FAA's global aviation safety leadership and supporting the competitiveness of U.S. products.

The ARC members express our appreciation for this opportunity and offer our support for future activities to consider these recommendations and develop additional clarification, guidance, and policies necessary for implementation.

Sincerely,

Mike Reinert, Co-Chair Manager, System Performance and Development Branch, AIR–150 Aircraft Certification Service Federal Aviation Administration

water Desroster

Walter Desrosier, Co-Chair Vice President, Engineering & Maintenance General Aviation Manufacturers Association (GAMA)

EXECUTIVE SUMMARY

This executive summary and summary of recommendations will provide a high-level overview of the Part 21/Safety Management Systems (SMS) Aviation Rulemaking Committee (ARC) activities and recommendations. This report provides the background for the ARC, then describes the current state for aircraft and component certification activities, followed by the ARC's vision for the future direction of the certification process.

The Part 21/SMS ARC's objectives and tasks were to evaluate certain improvements to the effectiveness and efficiency of existing "certification procedures for products and parts," along with incorporating SMS in the design and manufacturing environment. This included considering the effects of certain changes to the existing regulations, such as applicant qualifications, hazard (or safety) reporting, compliance assurance, and continued operational safety assurance systems for all design approval holders (DAH). The intent was to facilitate shifting toward a systems approach for product design approval/certification similar to that used for production approvals, which involves a clear understanding of roles, responsibilities, and privileges. The ARC's goal was to determine the best way the Federal Aviation Administration (FAA) and industry can effectively fulfill their respective compliance and safety responsibilities while improving the efficiency and robustness of the certification process.

The Aircraft Certification Process and Review Reform (ACPRR) ARC report, dated August 13, 2012, responded to section 312, Aircraft certification process review and reform, of the FAA Modernization and Reform Act of 2012 (Public Law 112-95), and provided recommendations to reform, streamline, and reengineer the certification process to meet future challenges. Previous assessments of the certification process found industry development of new aviation products and technologies is expected to continue growing at a pace that exceeds the FAA's ability to support. Of particular interest is the ever-increasing demand for FAA resources in areas other than certification such as continued operational safety of the existing fleet of aircraft, resulting in fewer available resources for aircraft certification in the future. This highlights the need for change to the certification process. The ACPRR's key recommendation is shifting the FAA certification process from a detailed product approach toward a systems safety approach. This report builds on the recommendations of the ACPRR report to Congress in support of an ongoing evolutionary shift to a systems approach to certification and provides high-level recommendations and detailed key considerations for a phased implementation using a building block approach. This enables a natural progression of the maturity and robustness of the current certification process, including industry compliance and FAA oversight processes.

In this report, the ARC provides four high-level recommendations on the following:

- 1. Phased implementation of a systems approach to certification.
 - a. Promote accountability framework and enhanced applicant showings.
 - b. Establish minimum requirements for design approval applicant/holder.
 - c. Establish requirements for voluntary certificated design organizations (DO).
- 2. Application of SMS requirements to design and production approval holders.

- 3. Evolution of FAA oversight toward performance-based systems safety (SMS) approaches.
- 4. Part 21 miscellaneous cleanup and technical standard order (TSO) modernization.

Each recommendation is discussed in detail within the report, which provides significant supporting information and "key considerations," which are supporting detail level recommendations necessary for practical implementation. Appendix E provides a summary overview of the key considerations that support the high-level recommendations and where they are discussed in the report.

1.0 BACKGROUND

1.1 INTRODUCTION

This report documents the results of the Part 21/Safety Management Systems (SMS) Aviation Rulemaking Committee (ARC), chartered by the Federal Aviation Administration (FAA) on October 15, 2012, to update the certification procedures in Title 14, Code of Federal Regulations (14 CFR) part 21, Certification Procedures for Products and Parts, to reflect a systems safety approach to product certification processes and oversight of design organizations (DO). The charter tasked the ARC to evaluate and recommend requirements for design approval applicants, application qualifications, hazards (or safety) reporting, compliance assurance, and continued operational safety (COS) systems for all design approval holders (DAH). The intent was (1) to provide guidance to the FAA on shifting toward a systems safety approach for DAHs similar to the approach used for production approval holder (PAH) requirements, which involves a clear understanding of roles, responsibilities, and privileges, and (2) to determine how the FAA and industry can most effectively fulfill their respective safety and compliance responsibilities.

Appendix A to this report contains a list of ARC members and support staff. Appendix B contains a glossary of terms, and appendix C is a list of acronyms used in the report. A copy of the charter is found in appendix D.

1.2 ARC SCOPE AND STRUCTURE

1.2.1 SCOPE

The scope of the ARC's tasks was to review and provide recommendations to update part 21 certification procedures to reflect a systems safety approach to product certification processes and oversight of DO. The goal was to identify the regulatory requirements for implementing DOs that would be fully accountable for determining compliance with the applicable product regulations using a systems approach that integrated the safety management requirements of the proposed 14 CFR part 5, Safety Management Systems. The part 21 requirements would establish minimum qualifications, performance, and management systems to ensure the DO could accomplish this activity with minimal FAA involvement and oversight. (Refer to section 1.3 below for an overview of the rulemaking process Federal agencies use to create regulations.)

1.2.2 ARC STRUCTURE

To accomplish its goals, the ARC formed four working groups to identify applicable regulatory requirements: the Design Organization Working Group (and Technical Standard Order Subteam), the Safety Management Systems Working Group, the Oversight Working Group, and the Cost-Benefit Analysis Working Group. The ARC created a charter for each working group to identify objectives, and each group produced a summary report containing the results of its deliberations and a set of final recommendations. The working group and subteam reports are included in their original form as appendixes F, G, H, I, and J to this report.

1.2.3 ORGANIZATION OF THIS REPORT

The body of this report consists of an executive summary and 11 numbered sections. Section 1 provides an introduction to the ARC's taskings and organization. Section 2 summarizes the current state of the certification processes, and section 3 introduces the ARC's proposed future state. A high-level summary of each of the four major recommendations is found in section 4. Key elements of the future state are explored in section 5 (regarding SMS), section 6 (regarding the DO model), and section 7 (regarding alternatives to DO for small businesses). The next two sections detail means by which the FAA and industry can achieve the desired future state: section 8 explains the ARC's recommended changes to part 21, and section 9 sets out the building block approach to transitioning from a delegated process to a system safety process for certification. Section 10 summarizes cost information related to the current and proposed certification processes. Dissenting or minority opinions from ARC members not included in the applicable sections of the report can be found in section 11. The report also includes 16 appendixes.

1.3 RULEMAKING PROCESS

The following section is an overview of the rulemaking process. The end of this section includes references to provide a more in-depth study of the rulemaking process.

Federal regulations are created through a process known as rulemaking. Regulations are issued by Federal agencies, boards, or commissions, and explain how the agency intends to carry out a law. By law, Federal agencies must consult the public when creating, modifying, or deleting rules in the CFR. Once an agency decides a regulation should be added, changed, or deleted, it typically publishes a proposed rule in the Federal Register to ask the public for comments.

After the agency considers public feedback and makes changes where appropriate, it then publishes a final rule in the Federal Register with a specific date for when the rule will become effective and enforceable. When the agency issues a final rule for comment, it must describe and respond to the public comments it received.

Refer to section 8 of this report for the ARC's recommendations on regulatory changes to part 21. Further information about the rulemaking process can be found at the following online sources:

- Electronic Code of Federal Regulations (eCFR): http://www.ecfr.gov/
- A Guide to the Rulemaking Process, Prepared by the Office of the Federal Register: http://www.federalregister.gov/uploads/2011/01/the_rulemaking_process.pdf
- The Rulemaking Process (U.S. Department of Transportation): http://www.dot.gov/regulations/rulemaking-process

1.4 PREVIOUS REVIEWS AND RECOMMENDATIONS

The FAA Modernization and Reform Act of 2012 (Public Law 112–95) required the FAA Administrator, in consultation with the aviation industry, to conduct an assessment of the aircraft certification and approval process and to make recommendations to streamline and reengineer to improve overall effectiveness and efficiency. The final report submitted to Congress included a review of previous reports conducted by an FAA/industry ARC that recommended changes to the certification process.

These reports were conducted by independent expert bodies such as the U.S. Aerospace Commission and National Research Council, as well as oversight agencies such as the U.S. Government Accountability Office (GAO) and the U.S Department of Transportation Office of the Inspector General (OIG). Some of these reports focused on specific FAA programs such as organization designation authorization (ODA) and risk-based resource targeting (RBRT) tools, and provided recommendations to improve implementation and effectiveness of FAA program management and safety oversight. Other reports assessed the rapidly changing aviation environment (increased growth in industry activity combined with the accelerated development of new technology and products) and provided recommendations for reengineering the FAA certification processes to meet future challenges and continue to improve safety levels.

Certification Process Reform Streamlining and Reengineering

The Aircraft Certification Process and Review Reform (ACPRR) ARC report (often referred to as the "312 report") dated August 13, 2012, was a response to section 312, Aircraft certification process review and reform, of the FAA Modernization and Reform Act, and summarized key recommendations for certification process improvements from previous reports. The ARC identified multiple instances where recommendations from previous reports converged and overlapped. Several independent assessments of the certification process found that industry development of new aviation products and technologies is expected to continue growing at a pace that exceeds the FAA's ability to support. All of the reports provided recommendations to reform, streamline, and reengineer the certification process to meet future challenges. The common theme among these recommendations is shifting the FAA certification process from a detailed product approach toward a systems safety approach. This Part 21/SMS ARC report builds on the ACPRR report to Congress in support of an ongoing evolutionary shift to a systems approach to certification, and provides high-level recommendations and detailed key considerations for a phased implementation using a building block approach.

Certified Design Organization ARC

The FAA Certified Design Organization (CDO) ARC submitted a final report to the FAA in May 2008 with recommendations for rulemaking, policies, and guidance to serve as the foundation for the establishment of FAA certification and oversight of design organizations (DO). The Part 21/SMS ARC notes that some of the information in sections 2 and 3 of this report was taken from the CDO ARC report because it remains relevant in describing some of the conditions driving the desire for an approved/certified DO.

2.0 CURRENT STATE

2.1 FAA AND INDUSTRY SHARE THE RESPONSIBILITY FOR CURRENT SAFETY SUCCESS

The flying public currently enjoys an unprecedented level of safety as a direct result of the FAA and industry's collaborative certification, maintenance, and operational approaches. By all measures, aircraft and aircraft operations have never been safer. Manufacturers, operators, repair stations, and the FAA have improved design, modernized operations, improved infrastructure, and delivered on a mandate for safe air travel.

This collective work by government and industry has resulted in a 5-fold reduction in air carrier accidents, reducing the rate of accidents that result in passenger fatalities to about one in every 14 million commercial flights. This has been accomplished, in part, with advances in technology, the use of structured data and analysis, and improved processes for the design certification, production, maintenance, and operation of aviation products. Notably, many of the actions to achieve this record safety level were developed and implemented as a result of voluntary actions by industry.

A fundamental aspect of the current certification process (the "detailed product approach") is that FAA resources (FAA engineers, inspectors, designated engineering representatives (DER), or ODA unit members (UM)) must be involved in an applicant's critical path activities to make discrete compliance findings with all U.S. aviation regulations applicable to the project. To determine opportunities to minimize critical path exposure, the ARC reviewed the requirement for FAA/DER/Designated Airworthiness Representative (DAR), or ODA UMs to find compliance and the delegation process for potential improvements. This section identifies some of the issues associated with the current certification process that need attention if the FAA and industry were to shift from a delegation process to a DO process.

2.2 DELEGATION

The current process for issuing an aircraft, engine, or propeller design certification or approving a change to type design is authorized under Title 49, United States Code (49 U.S.C.) § 44702(d), Issuance of Certificates. The FAA has increasingly recognized industry's expertise and resources in creating its system of individual and organizational delegations. Civil aviation expansion has far outpaced the FAA's growth in resources, and to leverage its resources, the FAA often relies on designees or delegated organizations to make discrete findings of compliance with regulatory requirements on its behalf.

The following outlines the history of the FAA's delegation system:

- 1940s: DER, designated manufacturing inspection representative (DMIR), designated pilot examiner (DPE) individual designees.
- 1950s: Design organization approval (DOA) organizational delegations for small airplanes, propellers, and engines.
- 1958: Federal Aviation Act of 1958 (Public Law 85–726) reaffirms delegation.

- 1960s: Designation alternation station (DAS) organizational delegation for repair stations.
- 1970s: Special Federal Aviation Regulation (SFAR) 36, Development of Major Repair Data, authorizations for operators.
- 1980s: DAR individual designees.
- 1990s: Organizational Designated Airworthiness Representative (ODAR) organizational delegations.
- 2006: ODA organizational delegations for all products and organizations; replaces DOA, DAS, ODAR, and SFAR 36.

The FAA has the authority to decide which aspects of a project require direct involvement, which aspects it should delegate, and to whom those aspects should be delegated. The FAA has tailored its delegation programs to ensure it retains responsibility for direct oversight of safety-critical areas.

For decades the aviation statutes and regulations have contained the provisions for delegation to both individuals and companies, but have also recognized that the FAA may rely on industry resources and expertise for compliance activities without delegation. Industry has assumed an enhanced role in making compliance showings that the FAA relies upon without making discrete findings in areas (such as primary category aircraft certification) that permit reduced FAA involvement before the issuance of a type certificate (TC) and technical standard order authorization (TSOA) design approvals.

The current U.S. certification processes require applicants to show compliance with all applicable airworthiness requirements, then relies on FAA resources, company DERs, and ODA UMs to review and verify the applicant showing to find compliance. For every applicant action, there is a corresponding need for an FAA reaction. This "show" and "find" process is repeated for nearly every discrete individual airworthiness requirement. Consequently, this certification process can be time consuming for larger and more complex projects.

Moving to a more effective and efficient certification process through a systems approach to safety, leveraging both the capability of applicant (with past experience) and the use of risk management techniques within the FAA, would benefit the overall timeliness of certification.

A systems approach that allows the FAA to accept the company statement of compliance through enhanced showings will allow FAA resources to focus on more safety-related issues would save the FAA and industry time and money.

This is an issue that both industry and the FAA must address. Industry members must ensure they have processes in place and personnel with the experience necessary to determine compliance with minimal FAA involvement. Industry members must also develop their processes to allow less reliance on their DERs and ODA UMs to both show and find compliance, particularly for low-risk requirements that can be approved by a company statement of compliance in accordance with § 21.20, Compliance with applicable requirements. This will require that the company fully understand and accept the FAA accountability framework. The

FAA must review its decisions for being involved in a project if the applicant has the capability to perform the compliance determination. Unless valid reasons exist to withhold complete delegation, a project should be delegated. If delegations are withheld, the FAA and applicant must develop a procedure to agree on how the applicant can perform the determination without FAA involvement.

As the U.S. aviation industry expands (in terms of both increased production and introduction of new products and technologies), more effective and efficient approaches to certification and related findings of compliance is needed so industry can more effectively and efficiently plan and execute the certification of new products with improved safety and utility.

2.3 CURRENT OVERSIGHT PROCESS

The FAA and applicant have a dual responsibility leading to the issuance of a certificate. Industry has the responsibility of full compliance and the FAA has the responsibility to make a finding of compliance before issuing any certificate. The current oversight process can be broken out into two different classifications: design certification oversight and production certification oversight. This section gives a high-level overview of the oversight process for both design and production.

Today's system for design certification is based on a "showing" (that is, a showing of compliance) by the applicant and a "finding" by the FAA. For every action by the applicant, there is an equal action by the authority on a project-by-project basis. An Aircraft Certification Office (ACO), Manufacturing Inspection District Office (MIDO), Flight Standards District Office (FSDO) and/or Aircraft Evaluation Group (AEG) office conducts design certification oversight through several methods.

With the introduction of ODA, the FAA instituted organizational management teams (OMT) for every ODA. FAA Order 8100.15, Organization Delegation Authorization Procedures, captures the OMT operations and requirements. An OMT's fundamental purpose is to oversee the ODA holder, which includes the ODA unit. In certain ODA applications, the FAA created "customized" OMTs for companies with large and/or complex ODAs. In general, these OMTs use dedicated resources to oversee the volume, complexity, and integration of ODA operations.

This structure is designed to combine the resources of current FAA offices (such as ACOs, MIDOs, and FSDOs). The centralized nature of these OMT structures enables increased commonality for ODA oversight. The dedicated resources reduce priority ODA support from competing with other FAA priorities.

In addition to design certification, the FAA is required to inspect aircraft during manufacture. Certificate management is the FAA's method for meeting this requirement, and auditing is its key component. The purpose of an audit is to verify a PAH has established and continues to follow approved procedures in the production of products, articles, and parts that conform to their approved type design and are in an airworthy condition for safe operation.

Certificate management of a production approval consists of four processes:

- **Principal inspector (PI) audit:** An ongoing progressive review of the PAH's quality system over a given period of time.
- **Supplier audit (SA):** An audit of the PAH's supplier control system, generally conducted at the supplier's facility.
- **Quality system audit (QSA):** A periodic, complete review of a PAH's approved quality system by FAA personnel.
- **Product audit:** Using a product or article produced by the PAH to prove the PAH is following its approved procedures/processes and their effectiveness. Product audits are performed during PI audits, QSAs, and SAs.

Certificate management of a PAH is a planned activity based on risk. The PI responsible for the PAH will enter required information concerning the PAH into the RBRT tool. The RBRT assessment tool is used to assign risk to a PAH according to the likelihood that it will produce nonconforming products, articles, or parts, and consequential results associated with introducing those products, articles, or parts into the system. RBRT assessments and associated procedures provide a consistent and justifiable basis for effective deployment of FAA resources when performing certificate management. Each directorate must annually assess PAHs using RBRT assessments. The RBRT assessment of each applicable facility is based on organizational and technical indicators that demonstrate a facility's potential for producing nonconforming products, articles, or parts. The RBRT assessment results in assigning a facility one of the following risk levels:

- 1. **High:** Having a facility with the greatest potential to produce nonconforming products, articles, or parts.
- 2. **Medium (Medium Low and Medium High):** Having a facility with moderate potential to produce nonconforming products, articles, or parts.
- 3. Low: Having a facility with low potential to produce nonconforming products, articles, or parts.

Audit data resulting from PAH certificate management activities is stored in the Certificate Management Information System (CMIS). CMIS data is used to detect shifts in performance and statistically significant trends within the manufacturing industry, by directorate, production approval type, or other categories as supported by the data available within CMIS. CMIS data may also be used to study various aspects of QSA performance as needed.

A key difference between design and production oversight is that design oversight is still more of a product-based oversight. Production oversight has moved to recognizing an organization and performing oversight based on risk.

2.4 DEMAND FOR FAA RESOURCES

The FAA must support several planned or implemented activities with the potential to further reduce FAA support for certification activity, including—

- Increased globalization. Companies around the world are constantly looking for opportunities to conduct innovative aircraft design and production. This changes the way the FAA is involved in projects and adds to the need for resources that can address those needs. This can potentially reduce the FAA's aircraft certification budget and personnel available for other certification activity.
- **Safety management systems.** Safety management is an FAA-supported mandate from the International Civil Aviation Organization (ICAO). Safety management may not affect design significantly but could affect the continued airworthiness process and require additional personnel and resources that could affect the budget available to support certification.
- Next Generation Air Transportation System (NextGen). The NextGen project, although not directly involved in certification, affects it by requiring certification of new equipment that must be installed to operate in the system. The implementation of this system will lead to a significant demand for certification services.
- Unmanned Aircraft Systems (UAS). UAS technology is on the cusp of explosive growth, requiring the FAA to allocate resources to support safe and successful integration of UAS into U.S. airspace.
- Environmental considerations (such as noise, emissions/carbon dioxide, halon, lead, and chemicals). There is a continuing demand from the U.S. Environmental Protection Agency (EPA) and public, to further restrict noise, emissions, chemicals, and other materials from the atmosphere. This requires additional FAA resources and could potentially affect the FAA's budget for certification activity.
- Military and public use aircraft. There are increasingly requirements to support military aircraft such as the McDonnell Douglas KC–10 and public use aircraft for the Department of the Interior, U.S. Forest Service, and others.

2.5 ECONOMIC IMPACT OF AVIATION INDUSTRY

The civil air transport industry plays a crucial role in fostering trade and making any location worldwide easily and quickly accessible. U.S. industry and consumers depend on the vital services of air transportation, which continue to maintain and stimulate the U.S. economy. Even with the extreme fluctuations in the economy and government sequestration, the aviation industry continues to readjust itself and regain stability.

- In 2009, air carriers operating in U.S. airspace transported 793 million passengers over 1,039.3 billion revenue passenger miles (RPM).
- More than 53 billion revenue ton-miles (RTM) of scheduled freight passed through U.S. airports in 2009.

- The U.S. civil aviation manufacturing industry continues to be the top U.S. net exporter. According to 2009 data from the U.S. International Trade Commission (USITC), the U.S. civil aviation manufacturing industry supported a positive trade balance of over \$75 billion.
- The 2011 FAA Aerospace Forecast expects average annual growth rates of 2.7 percent per year through 2034 for U.S. air carriers.

Despite these achievements, the U.S. aerospace industry may be facing some of its greatest challenges in decades. While weathering numerous hardships during 2013, the industry produced relatively flat results compared with 2012. An overall slight decrease in sales is forecasted, reaching \$220.1 billion for 2013—down from \$222 billion in 2012—with only civil aircraft sales showing growth. Figure 1 below, from the Aerospace Industries Association (AIA) Annual Report, depicts annual aerospace industry sales from 1998 through 2014.



Figure 1. Aerospace Industry Sales

Data has proven that civil aviation is on the rise, and even through difficult transitions, continues to increase. However, the FAA's ability to support this continuous growth at its current capacity is a concern. According to the Bureau of Transportation Statistics (BTS), employment projections show an increase in aviation transportation employees by 0.7 percent annually and, conversely, a decline of 1.6 percent in government employment through the year 2022. Additionally, the FAA, and the Office of Aviation Safety (AVS) in particular, has maintained a relatively flat increase in hiring because of a slow and steady hiring rate as well as attrition. Figure 2 illustrates this gap.



Figure 2. Employment Projections

If the anticipated industry growth continues by 0.7 percent per year through 2022 and the FAA employment rate remains level, the rate at which the FAA is able to support critical activities becomes increasingly more difficult. This significant gap will translate to lost industry revenue, stifling of innovation and entrepreneurs, a possible decline in air travel, and possible future safety-impacting occurrences. Civil aviation is moving faster than the government can keep up with, and if this trend continues, the direct and indirect impacts could be substantial. The longer a company is waiting for the FAA to proceed with certification activities, the longer product time to market will be.

Mitigation strategies have been implemented for certification activity to assist with the increasing trend of industry growth. However, systems such as sequencing and designee programs are no longer mitigating the existing gap appropriately, as evidenced by sequencing wait times and FAA employees' shifting areas of focus. The United States has reached a time where significant change is needed. This change can be made through an adjustment of the FAA's certification system, a topic considered in the next section of this report.

References

- Aerospace Industries Association (AIA), "2013 Year End Review and Forecast," 2013.
- Deloitte, "The Aerospace and Defense Industry in the U.S.: A Financial and Economic Impact Study," March 2012.
- FAA, "FAA Aerospace Forecast, Fiscal Years 2014-2034," June 2014. http://www.faa.gov/about/office_org/headquarters_offices/apl/aviation_forecasts/aerospa ce_forecasts/2014-2034/media/2014_FAA_Aerospace_Forecast.pdf

- FAA, "The economic impact of civil aviation on the U.S. economy," August 2011.
- U.S. Census, "Annual Survey of Manufactures: Geographic Area Statistics," 2007-2011.
- U.S. Department of Transportation, Bureau of Transportation Statistics (BTS), T-100 Segment, System revenue ton-miles.
- U.S. Department of Transportation, BTS, "New Quarterly Statistics Detail Industries' Economic Performance," April 2014. (http://bea.gov/newsreleases/industry/gdpindustry/gdpindnewsrelease.htm)

3.0 FUTURE STATE

This section examines the future challenges in U.S. aviation and the ways in which the FAA and industry can effectively face these challenges. Sections 5, 6, and 7 of this report detail three key elements of the ARC's desired future state of a systems approach to certification and oversight: SMS, DO, and minimum organizational requirements for design approval applicants and holders.

3.1 CASE FOR CHANGE

3.1.1 SAFETY THROUGH FUTURE COLLECTIVE ACTIONS OF FAA AND INDUSTRY

The continued growth of aviation will lead to an aviation system of the future with more general aviation aircraft and significantly more commercial aircraft flying more operations. New and different aircraft designs will also emerge at an increasing rate. Cost pressures have increasingly led to new business models for producing and selling aircraft and for the use of aftermarket parts in service and maintenance operations.

The FAA's current efforts to (1) rewrite part 23, Airworthiness Standards: Normal, Utility, Acrobatic, and Commuter Category Airplanes; (2) develop standard means of compliance to the regulations; and (3) find ways to implement new safety improvements in older airplanes represent a new approach to collective actions between the FAA and industry that could lead to additional collaboration. The European Aviation Safety Agency (EASA), Transport Canada Civil Aviation (TCCA), National Civil Aviation Agency of Brazil (ANAC), Civil Aviation Administration of China (CAAC), and the Civil Aviation Authority of New Zealand are also involved in this activity, representing a major change in the approach to collaborative effort among the FAA, industry, and major certification authorities.

3.1.2 THE CHALLENGE TO MEET PUBLIC DEMANDS FOR INCREASED SAFETY

The FAA and industry's joint challenge for the future is to continue the unprecedented safety improvements of the last decade. The level of safety the public has come to expect will be challenged by the FAA's ability to respond to the effects of new technology, new regulations, acute global competition, and global engineering and manufacturing.

With the FAA Aircraft Certification Service's (AIR) budget nearly flat, the increasing amount of time AIR spends on continued airworthiness and other high priority activities leads to less time available to support certification activity. At the same time, industry is attempting to increase certification activity and needs more support or an alternative approach to certification. The lack of sufficient FAA and designee resources and clearly defined means of compliance proposed by industry and accepted by the FAA exacerbates industry's inability to efficiently certify and market products that could help further improve safety.

Because history has shown the FAA's growth rate to be less than the growth rate of industry, the FAA must continue to seek solutions to improve safety while optimizing the use of its resources. Significant changes must be made to the certification process if the U.S. aviation industry is to continue to be a world leader in the production of aircraft products and maintain or improve safety. New aircraft sales and service support contribute significantly to the balance of trade. The loss of these economic factors because of the failure to improve the certification process would greatly harm the U.S. economy.

3.1.3 **OPPORTUNITIES FOR ADDITIONAL SAFETY IMPROVEMENTS**

Several studies, reports, and initiatives have been developed and implemented to "streamline" the certification process, shifting from reliance on FAA resources (FAA engineers, DERs, DMIRs, DARs, or ODA UMs) to an increased use of robust, predictable industry processes. The ACPRR ARC report (refer to section 1.4 above) summarized a need to "shift the FAA certification process from a detailed product approach toward a systems safety approach." This systems safety approach would use processes developed by the company or jointly by the FAA and industry as a means of determining compliance.

These documented industry systems and procedures to determine compliance with appropriate verification/assurance enable the FAA to propose this new shift in the certification process. This shift would also allow the FAA and industry to satisfy ICAO's SMS implementation mandate.

3.1.4 BILATERAL AGREEMENT SMS CONSIDERATIONS FOR PRODUCTS AND ARTICLES CERTIFICATION

The desired future state for SMS is that importing (validating) authorities will rely on the exporting (domestic) authority's overall system for aeronautical product design, certification, and production, including SMS implementation, in accordance with ICAO principles and existing bilateral agreements. Specifically, the importing (validating) authority would not—

- Evaluate or re-approve an organization's SMS to be able to accept/validate that organization's aeronautical products.
- Impose its domestic SMS requirements in addition to the exporting (domestic) authority's SMS requirements, which would result in the organization needing to adapt its SMS to multiple authority requirements and hold multiple SMS approvals.
- Impose its domestic SMS requirements on an organization even if the exporting (domestic) authority has not yet formally accepted/approved an organization's SMS at the time of product acceptance by the importing (validating) authority.

The Safety Management International Collaboration Group (SM ICG) (refer to appendix K to this report) is working with ICAO to embed these principles for contracting states' SMS development and safety programs. As a result, the FAA should continue to pursue the above principles when developing modified bilateral and validation agreements.

3.1.5 WHY ODA IS NOT ENOUGH—SHIFTING TO A SYSTEMS APPROACH TO CERTIFICATION

The FAA adopted the ODA program expanding organizational delegation as a means to provide more effective certification services to the aviation industry. As stated in the final rule preamble, the level of industry certification project activity and needs continue to grow at a rate exceeding that of FAA resources, and expansion of the available authorized functions will reduce the time and cost for these certification activities. This is a shift toward a systems approach to certification and oversight through authorization of industry organizations with the appropriate capabilities and processes to make compliance findings on behalf of the FAA so it may better focus its resources on oversight and safety areas.

The basis for a systems approach to certification is increased responsibility and capability of applicants for compliance activities, which the FAA can recognize and rely upon to support a shift from discrete product-level show/find activities toward systems safety oversight. The maximum opportunity to achieve the potential benefits and efficiency of a systems approach to certification and oversight requires the applicant/DAH and FAA engineers to understand and fully embrace the concept of accountability framework (refer to section 4.1.2 of this report).

However, ODA programs are delegation, which means the FAA remains responsible for the discrete compliance activities. ODA certification programs depend on the traditional "show/find" certification process whereby the applicant is responsible to "show compliance" and the FAA through its own resources or its designees (such as ODA unit member) must "find compliance" for each individual discrete requirement. Although the FAA's oversight of the ODA organization facilitates increased delegation, the roles and responsibilities of industry and the FAA are mixed because delegation is reliance on industry resources to act on behalf of the FAA. Likewise, industry compliance activities rely on the FAA/ODA review and verification of its showing. In addition, the use of delegation is completely discretionary for the FAA, which means that on any given project or area of a project the FAA can decide to be directly involved at any level and even retain any discrete findings that it chooses.

FAA delegation to industry individuals and organizations is also challenged by significant public and political perceptions and misunderstandings. Following an aviation safety issue or incident, there are typically many concerns raised and challenges that question the level of delegation the FAA exercised in certification of the product. The FAA's discretion to fully utilize delegation and have a high level of direct involvement in certification projects can swing like a pendulum, which poses significant challenges in establishing an efficient certification process that is effective and predictable.

Therefore, simply optimizing ODA is not enough—an update to part 21 certification procedures to approve/certify design organizations that can make compliance determinations that the FAA can rely upon is necessary to achieve the full potential of a systems safety approach for effective and efficient certification processes. FAA approval or certification of DOs provides significant opportunity for improvements in safety, safety culture, and the overall effectiveness and efficiency of aircraft certification processes through an evolutionary shift to a full systems safety approach to certification and FAA oversight.

3.2 VISION FOR THE FUTURE

The ARC envisions a future in which the FAA continues its leadership in advancing efficient certification and effective safety risk management (SRM). Ideally, the certification workload would be managed by a risk-based system using approved processes that places responsibility and accountability on industry in designing and manufacturing products in compliance with all applicable requirements with minimal direct FAA involvement. This new certification system would be more robust than the delegation system currently in place, and would be based on accountability and risk using a systems approach to determine and oversee compliance. This system would significantly help the United States maintain its outstanding safety rating and protect competitiveness in an increasingly global aircraft manufacturing market. Without significant changes in the certification process, the U.S. industry will not be able to grow at the rate needed to keep up with the demands for new products while remaining competitive in the global market and meeting new mandates created by NextGen and other required programs.

Listed below are five primary attributes of the future certification system that would provide significant benefits to the FAA and industry.

- 1. Compliance assurance system for industry compliance determination;
- 2. A systems approach to FAA oversight;
- 3. Greater industry control of project schedules;
- 4. Better leveraging of FAA and industry resources, enabling the growth that the market demands; and
- 5. Industry compliance determination best practices.

Each of these attributes contains significant change, but also significant benefits that would contribute to a balanced certification system and allow for increased project capacity and continued safety enhancements. These five attributes are briefly described below.

3.2.1 COMPLIANCE ASSURANCE SYSTEM FOR INDUSTRY SHOWINGS

A compliance assurance system (CAS) in which the FAA can confidently accept industry compliance determinations without an additional finding of compliance will depend heavily on defined processes that are continually reviewed and updated to ensure all required activities are conducted in a consistent and acceptable manner. In contrast, the current process for finding compliance depends on a one-for-one show/find process. This process is frequently inconsistent and has a large number of variables because it lacks clearly defined and accepted means of compliance and personnel on both sides of the show/find equation have different levels of experience and interpret requirements differently. If the system has clearly defined processes that the applicant consistently follows, the FAA will be able to evaluate the health of the organization and develop trust in the system. This process also places more accountability on applicant management to ensure the processes are properly maintained and followed. The current process can be likened to a production system that attempts to inspect quality into a product rather than build it in.

3.2.2 SYSTEMS APPROACH TO FAA OVERSIGHT

Taking a systems approach to industry compliance allows the FAA to also take a systems approach to oversight. In the desired future state, the current oversight process would transition from the traditional one-for-one show/find process to oversight of predetermined company processes. The FAA would perform oversight by reviewing an organization's self-assessment activities and auditing the processes. The FAA would not normally audit the results, unless a process was found to be deficient; if so, the FAA could review the results, or any other part of the process, to determine whether the results were unsatisfactory and whether any safety issues should be corrected. This approach reduces the amount of resources the FAA must allocate toward oversight and will enable it to allocate resources in an area of greater need.

3.2.3 GREATER INDUSTRY CONTROL OF PROJECT SCHEDULES

Currently, the industry is subject to the FAA's availability to initiate certification projects. With the transition to a compliance assurance procedure (CAP), the industry will have full control of each project schedule and will move the FAA off the project critical path. This benefit will be seen through reduced product development time and cost, as well as potentially increased revenues by quicker product time to market and allocation of resources to areas other than FAA detailed project-level involvement.

3.2.4 BETTER LEVERAGING OF FAA AND INDUSTRY RESOURCES

A systems approach allows for better leveraging of FAA and industry resources. Industry would be able to use resources that would normally be allocated to FAA detailed level of project involvement and product audits in areas of greater need such as project development and research and development (R&D). This enables a company to reduce its product time to market, have more control of project schedule, and eliminate sequencing or FAA resource and response wait times with a higher level confidence than for a product being developed in the current environment.

The FAA will experience similar benefits. By implementing a systems approach, the FAA can reduce the amount of resources allocated to support the find compliance process and product audits, and focus instead on the health of the organization. Additionally, with the increased demand of civil aviation, the FAA will be able to provide support at its current capacity with less concern of being able to keep up with industry demand.

3.2.5 INDUSTRY COMPLIANCE DETERMINATION BEST PRACTICES

Compliance by process is a main contributor to the overall performance-based system. This process depends on using FAA-accepted industry best practices for determining compliance. These best practices can be any of the following:

- 1. Industry standards such as SAE International (SAE), RTCA, Inc. (RTCA), or ASTM International (ASTM);
- 2. Company-developed and maintained compliance determination processes;
- 3. FAA advisory circulars (AC), policy, etc.;

- 4. A combination of the above when accepted by the FAA; or
- 5. Other FAA-accepted compliance determination processes.

These industry best practices are frequently referred to as a "compliance library." Most companies can review existing compliance reports and select compliance determinations that are considered acceptable and easily create a standard for that regulation. These practices will take time to establish, but will result in a system that when properly constructed allows repeated, consistent compliance determination that is accepted by the FAA. Because the library defines how compliance will be determined for a particular regulation, many determinations can be accomplished by engineers with less experience, freeing the senior engineers to do more critical work. Having an FAA-accepted compliance library will increase the FAA's confidence in the system and eliminate the need for find compliance inspections conducted through delegation.

3.3 PERFORMANCE-BASED OVERSIGHT (PBO) RECOMMENDATIONS

The recommended future oversight model is fundamentally composed of assessment and surveillance. The ARC envisions in the future the FAA will require minimum organization capabilities to justify application. Those capabilities and expanded capabilities will be determined by an FAA oversight assessment methodology. Similarly, the FAA will define capability requirements to become a DO. The same FAA assessment approach will be used to determine whether a company meets the DO qualifications. As that DO company develops additional capabilities that are further assessed by the FAA, it may gain additional privileges. In between the application, DO certification, and expanded privileges, the future FAA oversight will be surveillance of the company's performance.

The future oversight model concept includes oversight for design/manufacturing organizations, product/article, and post-certification using COS processes. The three areas introduce levels of capability (organization), levels of project involvement (product/article), and levels of surveillance (post-certification). The model envisions a single FAA oversight presence for the three key areas:

- Organizational: Transition from traditional show/find compliance to an organizational PBO model.
- Products and Articles: Transition from the FAA's traditional role of direct project involvement to a performance-based project oversight model. FAA participation will be limited to LOPI.
- Post-Certification Using COS Processes: Transition from traditional reactionary model to a systemic (process-based) surveillance model.

The future FAA oversight system will be based on two principles: PBO and compliance-based oversight (CBO). Each oversight principle has advantages and disadvantages, and collectively they balance the safety performance goals. PBO has greater advantages than CBO. PBO focuses FAA resources on areas of higher risk in the aviation system and moves the FAA from a total dependence on compliance findings, audits, and inspections to a more effective approach of monitoring safety and compliance performance data from the aviation industry. CBO is at point

of manufacture/integration rather than storefront, while PBO can be virtual. Increases in compliance, conformity, and safety performance will adjust the traditional CBO activity and frequency.

Performance data has target goals that are mutually agreed to by the company and the FAA (vitals show the health of the company's compliance, conformity, and safety). A company maintains its system to move in the direction of acceptable/better performance. Developing performance indicators would enable selection of the indicators best suited to establish the safety and compliance health of a company. The types of indicators used could be added, deleted, or adjusted based on acceptable performance trends. The FAA could also identify minimum types of risk-based performance data the FAA should monitor.

The ARC recommends a dedicated FAA and industry effort to develop guidance for determining performance indicators that are mutually acceptable before implementing the new oversight model. Consideration for the effort would include review of an existing documentation such as the AVS Strategy and Framework to Manage Safety Performance in AVS document, dated June 30, 2013 (refer to appendix D to the Oversight Working Group Report, included as appendix H to this report). In addition, the development of the performance level indicators should consider the EASA approach, which intends to assign a performance level to a company based on performance parameters from its surveillance of the company's organization as well as its involvement in projects and activities. This is captured in the EASA Embodiment of LOI and SMS Requirements into Part 21 document (refer to appendix E to the Oversight Working Group Report).

4.0 HIGH-LEVEL RECOMMENDATIONS & IMPLEMENTATION

The Part 21/SMS ARC's objectives and tasks were to evaluate certain improvements to the effectiveness and efficiency of existing "certification procedures for products and parts," along with incorporating SMS in the design and manufacturing environment. This included considering the effects of certain changes to the existing regulations, such as applicant qualifications, hazard (or safety) reporting, compliance assurance, and COS assurance systems for all DAHs. The intent was to facilitate shifting toward a systems approach for product design approval/certification similar to that used for production approvals, which involves a clear understanding of roles, responsibilities, and privileges. The ARC's goal was to determine the best way the FAA and industry can effectively fulfill their respective compliance and safety responsibilities while improving the efficiency and robustness of the certification process.

In this report, the ARC provides four high-level recommendations on the following:

- 1. Phased implementation of a systems approach to certification.
 - a. Promote accountability framework and enhanced applicant showings.
 - b. Establish minimum requirements for design approval applicant/holder.
 - c. Establish requirements for voluntary certificated DOs.
- 2. Application of SMS requirements to design and production approval holders.
- 3. Evolution of FAA oversight toward performance-based systems safety (SMS) approaches.
- 4. Part 21 miscellaneous cleanup and technical standard order (TSO) modernization.

Appendix E identifies all of the key considerations that support each of these four high-level recommendations and where they may be found in the report.

4.1 SYSTEMS APPROACH TO CERTIFICATION

The ACPRR ARC conducted an assessment of the aircraft certification and approval process, and the FAA's final report to Congress makes recommendations to streamline and reengineer to improve overall effectiveness and efficiency. A key recommendation is shifting the FAA certification process from a detailed product approach toward a systems safety approach.

Effective implementation of the shift from the current certification process and ODA procedures and the FAA's internal SMS for risk-based decisions on oversight and direct level of involvement in certification activity to a systems approach to certification leverages the FAA's limited resources and improves the efficiency of certification process. However, a reengineering update to part 21 certification procedures is necessary to achieve the full potential of a systems safety approach for effective and efficient certification processes. The ARC finds that following the systems approach to the certification method outlined in this report is a natural progression of the maturity and capability of current industry and FAA processes.

FAA approval or certification of DOs provides significant opportunity for improvements in safety, safety culture, and the overall effectiveness and efficiency of aircraft certification processes through an evolutionary shift to a systems safety approach. This is consistent with how the FAA provides safety oversight of aircraft production, air carrier operations, and repair stations. In addition, this approach to aircraft certification is used in Europe, Canada, and Brazil. The ARC strongly supports initiatives to continue shifting the aircraft certification and FAA oversight process toward a systems approach including the establishment of new regulatory requirements for design approval applicants/holders, SMS, and approved or certified DOs.

In support of an ongoing evolutionary shift to a systems approach to certification and the broad range of project activities, the ARC has structured its recommendations into a phased implementation using a building block approach.

4.1.1 PHASED IMPLEMENTATION USING A BUILDING BLOCK APPROACH

Table 1 below provides a high-level view of the three phases the ARC anticipates in implementing the identified report recommendations. The three phases consist of "Today," "Transitional," and "Transformational," which stretch across a 7+ year timeframe to reach the goal of having certified DOs. There are individual actions and recommendations from different sections of this report within each phase for both industry and the FAA. This table should not be viewed as including all of the final, definitive requirements but as a high-level proposal to achieve the desired goal of a systems approach to certification and oversight through maturity of industry organizational and compliance capabilities to an approved DO (that is, the building block approach).

	Today Near Term (0-3 Years)	Transitional Intermediate Term (3-7 Years)		Transformational Long Term (7+ Years)
•	Optimize ODA and organizational oversight. Systems approach to	 Develop and promulgate new Part 21 regulations: Minimum organizational 	•	Develop and promulgate new Part 21 regulations for voluntary DO—a
	 Enhance policy and understanding of accountability framework. Promote enhanced showings (showing only, no discrete findings). 	 requirements for all design approval applicants and holders. SMS requirements for certain organizations. Compliance Assurance Procedures (CAP). Compliance libraries. Continue to mature oversight methods as the FAA transitions to a more systems safety-based oversight (SMS). 	•	certificated organization with compliance verification process. Feedback loop to ensure processes are continuously monitored and updated. Reduction in designees.

Table 1	Three_Phase	Implementation	of Systems A	nnroach to	Cartification
Table 1.	Inree-Fnase	Implementation	oj systems A	прионение	Certification
The ARC has recognized that neither industry nor the FAA is ready to go directly to a mandatory DO implementation or to define a specific future date where we think robustness of industry compliance processes and FAA oversight will be ready. The phased implementation is a methodical, although slower, transition that allows both industry and the FAA to mature into a robust systems-based approach to safety. Upon completion of the phased building block approach, an applicant would be expected to have sufficient processes in place to meet the requirements of an approved DO. This allows the FAA and industry to properly shift their cultures and business structures without extreme cost while gradually obtaining benefits. With successful implementation of these building blocks, the ARC supports a future rulemaking to consider mandatory implementation of DO.

4.1.2 SYSTEMS APPROACH: TODAY—NEAR TERM (0-3 YEARS)

The near-term activities focus primarily on improving the implementation of current certification processes and developing policies and guidance to promote a better understanding of systems approaches to certification and oversight and voluntary implementation/use in certain areas. A major focus in the near term needs to be a better understanding by both industry and FAA personnel of the accountability framework on which much of the remainder of the activities will depend.

Optimize ODA and Organizational Oversight

The FAA should continue to manage ODA initiatives as a priority, including the monitoring of field performance to ensure consistent implementation and intended benefits for both industry and the FAA. ODA streamlining and certification efficiency is characterized by "full utilization," which can be defined as ODA management of any function that is not inherently governmental.

Optimizing ODA will begin with the ODA holder developing defined processes it can use to satisfy the ODA and certification requirements. As these processes are developed and validated in accordance with an FAA-approved procedures manual, the ODA holder will be granted additional approvals to perform the applicable tasks with minimal direct involvement and will rely more on the processes to ensure the ODA holder is performing the tasks in a satisfactory manner.

These processes must cover all aspects of an ODA holder's responsibilities. This would typically include all administrative requirements as well as maintaining the appropriate design documents, training, and personnel records; developing a compliance library and COS procedures, etc.

There should be a continuous effort on the part of both the ODA holder and the FAA to review all ODA limitations and identify areas where improvements can be made to reduce FAA direct involvement.

Maximizing the full capacity of ODA will further reduce the FAA's certification workload. This will ensure alignment with the objectives to improve efficiency, reduce cost, and redirect FAA resources to enable development of new products and technologies and establish a strong foundation for building blocks toward DO.

Systems Approach to Certification—Voluntary

The ARC finds that FAA certification of DOs is a natural progression of the maturity and capability of organizational delegation that provides a significant opportunity for improvements in safety, safety culture, and the overall effectiveness and efficiency of aircraft certification processes.

One of the biggest challenges related to the systems approach to certification is the maturity and robustness of processes and the cultural shift that must occur with both industry and the FAA. Accomplishing this cultural shift will require education, patience, diligence, and perseverance on both sides. Using a phased approach is the most realistic method of achieving a transition to a systems approach to certification and oversight. The basis for a systems approach to certification is increased responsibility and capability of applicants for compliance activities, which the FAA can recognize and rely upon to support a shift from discrete product-level show/find activities toward systems safety oversight. Several examples of voluntary approaches between industry organizations and the FAA reflect a systems approach to certification on specific tasks. This is based on the concept of the accountability framework and applicant enhanced showings.

Accountability Framework

To optimize a systems approach, it is critical that the applicant/DAH, along with the ODA holder and FAA ACO, understand and fully embrace the concept of the accountability framework. The underlying premise of the accountability framework is that applicants and approval holders have full responsibility (legal and regulatory) for compliance with all applicable requirements (refer to 14 CFR). The FAA Administrator has the authority under the statutes to exercise discretion in promulgating airworthiness regulations and standards in the interest of safety, and defining by regulations and directives how the FAA will oversee compliance by those it regulates. Amendment 92 to §§ 21.20 and 21.97, effective April 16, 2011, provides the regulatory basis for the company statement of compliance, and AC 21–51, Applicant's Showing of Compliance and Certifying Statement of Compliance, provides clear guidance on the intent of the accountability framework.

The practical implementation of the accountability framework is for the FAA to exercise its discretion on the level of involvement necessary to make a finding that the applicant has shown compliance with all the applicable requirements before issuing a design approval. In the current show/find process, the FAA is involved in nearly 100 percent of all discrete compliance activities but this is not required. The FAA may rely on an applicant's showing and not make a discrete finding based on demonstrated capability or accepted processes. Increased reliance and acceptance of applicant showings and the ability for the FAA to focus its resources primarily in risk-based areas will require both additional training on the technical aspects of robust processes and a cultural shift in responsibilities.

Enhanced Showings

This phased approach begins moving toward a more uniform and consistent approach to those activities the applicant must perform. Among those processes would be FAA and industry agreed-upon standards for performing such tasks as burn testing, an FAA-approved compliance library, and a process for receiving and processing field reports to ensure the continued airworthiness of products. It is anticipated that the compliance library could be an individual library for each applicant or a consensus standard developed by industry and the FAA that is available for anyone to use. The benefit of a consensus-based means of compliance is that it takes advantage of input from numerous sources, including other CAAs, to ensure all aspects of an issue are covered, and it can reduce the number of different means of compliance the FAA would have to approve. This would significantly reduce the need for direct FAA involvement for most projects and would allow the applicant more flexibility to meet their schedules. Once the standard procedures are developed and an applicant documents compliance to those procedures, the FAA involvement would depend on the LOPI deemed necessary by the FAA based on the risk. It should be noted that there will be overlap from one area to the next in some of the objectives.

Table 2 below depicts the requirements for a low, medium, or high risk project to determine the FAA LOPI privilege. The objective is to decrease the LOPI as the capability of the applicant increases.

Project Risk Level	Requirements to Earn LOPI Privilege	Begin Implementing Today
High	The safety assurance feedback loop has been achieved when the ability to ensure COS tracking and compliance verification has been accomplished through a systems approach to safety.	Creates a feedback loop (or safety assurance function) that allows high-risk projects to be reduced to medium or low risk.
Medium	Approved Compliance Assurance Procedures (CAP).	
	Example: CAP for flammability.	
Low	Documented and agreed-upon means of compliance.	
	Self-audit program.	

Table 2. Near-Term Systems Approach to Certification

Recommendation 1a—Systems Approach to Certification – Voluntary Initiatives: The ARC recommends that the FAA issue policy and guidance to promote the understanding of the accountability framework as a basis for a systems approach to certification and facilitate voluntary approaches to implementation through FAA recognition and acceptance of applicant enhanced showings. (Refer to section 10.2 of this report.)

4.1.3 Systems Approach: Transitional—Intermediate Term (3-7 Years)

The intermediate "Transitional" phase will focus on new part 21 requirements to clarify applicant responsibilities and continue maturing applicant compliance processes. This includes new requirements for applicants to show a minimum level of understanding of certification processes to make application to the FAA as well as the application of SMS requirements into the certification and COS processes for certain organizations. Throughout this phase, applicant CAPs and FAA oversight will continue to mature in robustness and capability.

New Minimum Organizational Requirements for All Design Approval Applicants and Holders

One of the effects of shifting from a detailed product approach to a systems approach to certification is the increased responsibility on the applicant to know and understand the regulations, certification process, and applicable airworthiness requirements before starting a project. The current part 21 requirements do not require the applicant to have any knowledge or capability, so often much of the work falls on the FAA to educate applicants on certification processes and airworthiness requirements and to ensure the applicant carries out their responsibility to show compliance. For applicants that do not have experience or understanding of the certification requirements and processes, this introduces a significant number of unknowns that result in projects that can require significant back-and-forth with the FAA and can take a substantial amount of time beyond the intended schedule and budget. There is also a significant impact on the FAA due to increased workload and a high level of involvement in detailed project activities that often result in incomplete certification projects. This has a much broader impact across the industry as inefficient use of FAA resources detracts from other safety activities and certification projects.

The ARC proposes changes to part 21 to more clearly define the minimum regulatory requirements for all design approval applicants and holders to ensure they understand the responsibilities of an applicant and to explain to the FAA how they intend to carry them out. The ARC's recommendation for minimum applicant/holder requirements is discussed in detail in section 9 of this report.

Recommendation 1b—Systems Approach to Certification – Minimum Applicant/Holder Requirements: The ARC recommends establishing minimum requirements for design approval applicant/holder qualification and responsibilities to ensure they fully understand the type certification process and how they intend to carry them out.

New SMS Requirements

The ARC recommends applying most of the requirements of the proposed part 5 to certain design and production approval holder organizations and incorporating these elements into certification and COS processes. The ARC's SMS recommendation is discussed in detail in section 4.2 below.

CAPs and Compliance Libraries

The evolution to a systems approach requires increased responsibility and capability of applicant processes and compliance activities consistent with the accountability framework. To support efficient certification processes and the ability for increased FAA recognition of applicant enhanced showings, the ARC recommends the development of policy/guidance to formally establish compliance libraries and a CAP. Any applicant can establish a compliance library that reflects those methods acceptable to the FAA and with which the applicant has demonstrated experience. This allows for efficient acceptance of the same methods of compliance for similar projects in the future. Organizations will implement SMS and CAPs that provide robust verification and monitoring of safety and compliance activities within their safety assurance functions. In conjunction with appropriate oversight, this can recognized and relied on by the FAA as enhanced showings without need for discrete compliance findings.

Continue to Mature Oversight Methods in the Transition to a More Systems Safety-Based Oversight

The ARC recommends that the FAA continue to mature performance-based oversight methods throughout this phase and that the FAA define reporting requirements. This will provide an opportunity to test and evaluate feedback processes that provide a clear picture of FAA and industry activity and will assist in identifying areas that may need assistance, or different reporting methods or data. The ARC's oversight recommendation is discussed in detail in section 4.3 below.

4.1.4 Systems Approach: Transformational—Long Term (7+ Years)

The transformational period will be critical in terms of completing the transition to approved DOs. It will depend highly on how successful the previous phases were in meeting their objectives.

New DO Requirements

Recommendation 1c—Systems Approach to Certification – DO

Requirements: The ARC recommends establishing regulatory requirements for the issuance and oversight of voluntary certificated DOs, including the necessary compliance assurance, safety management, and controls to make all compliance determinations through applicant showing and verification processes. Through FAA DO certificate management oversight and direct project involvement in defined risk-based areas, the FAA may rely on the DO compliance determinations to make its finding for the issuance of a design approval. This report builds on the recommendations submitted to the FAA by the CDO ARC in May 2008. The ARC recommends a building block approach to implementing DO, which includes establishing a clear accountability framework, transitioning the FAA's oversight of design activities to a centralized systematic model, optimizing use of ODA, and implementing new organizational and SMS requirements. With successful implementation of these building blocks and voluntary DO, the ARC supports future rulemaking to consider mandatory DO.

Feedback Loop

The ARC recommends that feedback loops be created that will provide information on the activities throughout the transition to ensure the transition is moving in the direction that supports reaching DO status during the third phase. Following the transition to DO, there will still need to be feedback loops to ensure all of the processes are functioning as intended and to assist in spotting issues that one or more DOs may be having that may need to be addressed either individually or as a group. However, the feedback loops used during the transition may not continue into the mature DO stage or may be modified to provide different types of information or at different frequencies.

Reduction in Designees

The ARC recognizes the need for designees and ODA UMs will be reduced when the transition to DO occurs. However, it is expected that the designees will continue to function in much the same role as they do today for much of the certification work. Their technical expertise will still be needed to help maintain the compliance libraries and determine compliance on which the company statement of compliance can rely.

4.2 SAFETY MANAGEMENT SYSTEMS REQUIREMENTS

The ARC's vision for the future of the FAA certification process depends on a systems-based approach by both industry and the FAA. The goal of this approach is to gain efficiencies and maintain effectiveness while also increasing the safety of the aviation system as a whole. Thus the ARC envisions SMS to be an integral part of an organization's processes for type certification compliance and post-TC COS processes.

An SMS includes an organization-wide safety policy, formal methods for identifying hazards, controlling and continually assessing risk, and promoting a safety culture. When systematically applied, an SMS provides a set of decisionmaking tools that certificate holders can use to improve safety. SMS requires a proactive approach to discovering and addressing hazards before they exhibit safety consequences. SMS also includes processes to assess potential organizational and process changes to ensure a compliant conformed product or article and the proper functioning of the COS processes, which allow management to address a safety issue before a noncompliant or unsafe condition results. SMS is not a substitute for compliance with FAA regulations or FAA oversight activities.

The ARC's SMS Working Group analyzed current requirements and planned future requirements in the proposed part 5, and developed a concept of operations (CONOPS) describing the practical application of part 5 for design and manufacturing (D&M) organizations. Consistent with ICAO guidance, the ARC recommendations allow a phased approach to implementation, with a manageable series of steps to follow including allocation of resources, effectively managing the workload associated with implementation, and providing for a realistic implementation of an effective SMS (avoiding a "cosmetic compliance"). This phased approach is described in more detail in sections 5.1.5 and 5.1.6 of this report.

Recommendation 2—SMS Requirements: The ARC recommends establishing regulatory requirements for implementing SMS consistent with the proposed part 5 for design and production approval organizations. This new requirement should apply to organizations that design or manufacture type-certificated products (under a TC or production certificate) and those that design or manufacture articles (under a TSO or parts manufacturer approval (PMA)) or make changes to products (under a supplemental type certificate (STC)) that could directly prevent continued safe flight and landing if they fail.

For additional details on the ARC's SMS evaluation and recommendations, refer to section 5 of this report.

4.3 EVOLUTION OF OVERSIGHT

Establishment of a systems approach to certification requires a different approach to FAA oversight of an applicant or DAH. Using FAA-approved standards for determining compliance will require the FAA to develop and implement processes for ensuring the applicant is following their approved processes rather than looking at discrete findings of compliance except as necessary to support the procedures-based oversight. In turn, industry must develop robust processes that demonstrate compliance and can be used as part of an FAA-approved compliance library. The oversight process will also ensure the appropriate SMS requirements are included in the applicant's processes. Additionally, an applicant's quality management system (QMS) should ensure a continuous self-evaluation and improvement process to address those issues.

The development of this new approach to oversight will be best managed using a centralized FAA oversight system to provide consistent and progressive assessment and surveillance processes leading to performance-based standards.

Establishing a central FAA oversight organization will-

- Achieve standard surveillance practices.
- Centralize policy responsibility ensuring consistency in interpretations.
- Allow the ACO to focus on safety-critical functions.
- Provide "third-party objectivity," as the office does not work programs with the DO.
- Provide a single source/repository for the oversight data, which will drive the risk-based modeling controls.
- Manage skill development practices for the surveillance staff.
- Allow for a highly trained staff in system surveillance.
- Provide a single source for oversight of corrective actions.

Recommendation 3—Evolution of FAA Oversight Toward Performance-Based Systems Safety (SMS) Approaches: The ARC recommends development of a performance-based single surveillance oversight approach that aligns with proposed changes to design and production organizational requirements and a systems approach to certification. The three key oversight areas are: (1) Organizational—transition from traditional show/find compliance to organizational PBO model; (2) Product and Articles-transition from the FAA's traditional role of direct project involvement to a LOPI approach focused on performing governmental functions and enhanced showing capabilities; (3) Post-Certification (COS)-transition from a traditional reactionary approach to a systemic (process-based) surveillance model. Establishing a central FAA oversight organization will achieve standard surveillance practices, create centralized policy, be a single source/repository for the oversight data that will drive the risk-based modeling controls, and allow for a highly trained staff in system surveillance, skill management, and a single source for corrective actions. As companies continue to evolve to a systems safety-based certification and organizational oversight, a centralized FAA oversight system will provide consistent and progressive assessment and surveillance processes leading to the performance-based standard.

The detailed recommendations for this process are found in section 3.3, section 9, and appendix H of this report.

The ARC developed proposed practices for FAA oversight that correlate with recommended D&M organizational changes. This enables a shift to performance-based oversight where the FAA can effectively allocate resources based on D&M system risk management performance, and moves the FAA from a total dependence on discrete compliance findings, audits, and inspections.

Figure 3 illustrates the stepping stone to oversight as envisioned by the ARC and is designed to align with the table 1 above.

		DO Rulemaking Transformational	
	Mandatory Implementation and Rulemaking TC/PC/STC/PMA Organizational and SMS Based on Risk Transitional		
Systems Approach to Certification—Voluntary Today (Ongoing Effort)			

Figure 3. Evolution of Oversight

4.4 PART 21 MISCELLANEOUS CLEANUP AND TSO MODERNIZATION

The Part 21 Cleanup section was developed based on the broad nature of this ARC. There were many items to be addressed within part 21 that may have been deemed insignificant, or that could not be grouped with one of the other three primary recommendations. However, the ARC felt these items were significant enough to address and should not go unaccounted for. The part 21 cleanup items can be grouped into three subject areas:

- 1. Recommendations to simply clean up issues that would eliminate confusion and inability to comply in the current regulation, if addressed.
- 2. Response to recognition of part 21 related recommendations from the Part 23 Reorganization ARC to reduce costs related to certification.
- 3. Changes to § 21.3 reporting, which also aligns with SMS and part 183, Representatives of the Administrator, § 183.63.

The details of these recommendations can be found in section 9 of this report.

The ARC's TSO Subteam recommends a number of changes to modernize the TSO process that require changes to part 21. However, some of these recommendations can be addressed by policy changes and do not require part 21 rule changes. These proposals are discussed in detail in sections 9.1.3 and 9.2 of this report. The ARC believes those TSO modernization changes that do not require rule changes could be accomplished in the first of the three phases.

Recommendation 4—Part 21 Cleanup and TSO Modernization: The ARC recommends FAA consideration of several proposed changes and updates to various part 21 regulations, which primarily reflect clarifications to eliminate confusion, modernization to reflect current practices, and updates to align with other recommendations in this report for a systems approach to certification.

5.0 SMS RECOMMENDATIONS AND REQUIREMENTS

5.1 SMS IMPLEMENTATION

5.1.1 SMS Recommendations

The ARC's vision for the future of the FAA certification process depends on a systems-based approach by both industry and the FAA. The goal of this approach is to gain efficiencies and maintain effectiveness while also increasing the safety of the aviation system as a whole.

An SMS includes an organization-wide safety policy, formal methods for identifying hazards, controlling and continually assessing risk, and promoting a safety culture. When systematically applied, an SMS provides a set of decisionmaking tools that certificate holders can use to improve safety. SMS requires a proactive approach to discovering and addressing hazards before they exhibit safety consequences. SMS also includes processes that seek to assess organizational changes to allow management to address a noncompliant condition before an unsafe condition results. SMS is not a substitute for compliance with FAA regulations or FAA oversight activities.

The ARC's SMS Working Group performed a gap assessment of the current part 21 requirements to the proposed part 5 requirements (refer to appendix G to this report). The working group determined part 21 only partially addresses the requirements of the proposed part 5, primarily through meeting the airworthiness standards and quality system requirement. Therefore, gaps in safety management exist that must be closed by the application of SMS.

In addition, ICAO Annex 19, Safety Management, mandates the implementation of an SMS for organizations responsible for design and/or manufacture of aircraft. The United States must adopt a system meeting the ICAO standards as a member state.

In consideration of the above, the ARC recommends SMS be applied to D&M organizations. Refer to the National Air Traffic Controllers Association's (NATCA) dissenting opinion of section 5.1.1 in section 12 of this report.

The ARC has determined the SMS requirements in the proposed part 5, with consideration given to D&M sector comments for recommended changes contained in the docket, are appropriate for D&M organizations. In addition, the ARC evaluated the proposed § 5.27, Coordination of emergency response planning, and determined it is not necessary for D&M organizations. Therefore, the ARC recommends the FAA modify part 21 to make part 5, excluding § 5.27, the SMS requirements for organizations meeting the SMS applicability threshold.

The ARC recommends the proposed part 5 be referred to by the appropriate part 21 regulations because it was determined to be applicable (with only minor deviation), minimizes the regulatory revisions required to implement, ensures harmonization with FAA and ICAO SMS requirements, and allows for efficient oversight by the FAA.

5.1.2 SMS APPLICABILITY

ICAO Annex 19 only requires SMS for organizations that design or produce products (aircraft, engines, and propellers). That requirement does not include DAHs for STCs, PMAs, or TSOs. Several ways to apply SMS requirements to D&M organizations were considered:

- Organizations that design or manufacture products (that is, aircraft, engines, or propellers);
- All D&M organizations that hold a design approval (that is, a TC, STC, PMA, or TSO) or hold a production certificate to manufacture products or articles; and
- Certain D&M organizations based on safety risk.

The ARC considered the scope of applicability for SMS and determined it should be based on safety risk. To that end, applying SMS to only the product-level D&M organizations does not adequately address safety risk as it does not apply to many aspects of COS, and an SMS requirement for all D&M organizations would be too broad, including organizations with minimal impact to increased safety risk. A risk-based approach provides a fair solution among small and large businesses and among various business structures.

Based on the SMS Working Group's considerations and recommendation, the ARC makes the following recommendation for SMS applicability to D&M organizations.

The ARC recommends the FAA and industry develop guidance for an SMS applicability threshold requiring an SMS for organizations that—

- Design or manufacture products (that is, aircraft, engines, or propellers);
- Design or manufacture articles (TSO, PMA) whose failure could directly prevent continued safe flight and landing; or
- <u>Make design changes to a product through an STC, failure of which could directly</u> prevent continued safe flight and landing.

Note that this recommendation is not intended to discourage voluntary implementation of SMS for organizations producing articles with criticality falling below the SMS applicability threshold.

Recommendation 2—SMS Requirements: The ARC recommends establishing regulatory requirements for implementing SMS consistent with the proposed part 5 for design and production approval organizations. This new requirement should apply to organizations that design or manufacture type-certificated products (under a TC or production certificate) and those that design or manufacture articles (under a TSO or PMA) or make changes to products (under an STC) that could directly prevent continued safe flight and landing if they fail.

5.1.3 SMS APPLICATION TO D&M ORGANIZATIONS WITHOUT A CERTIFICATE

Other regulatory authorities have applied SMS to the aviation industry by linking SMS with an "operating certificate" (such as a maintenance organization approval, operator certificate, or production approval) that allows the organization to conduct specific activities/operations. The SMS is a condition of initial certificate issuance and ongoing certificate maintenance. A discrete SMS approval is not issued; instead, the "operating certificate" is issued when the applicant/holder demonstrates the organization meets all certificate requirements, including those for SMS. In this case, deficiencies in the SMS can lead to the regulator refusing initial certificate issuance, or taking certificate action such as fines or limited certificate privileges.

Application of an SMS to the U.S. aeronautical product design industry is challenging, as no "operating certificate" exists similar to other aviation industry sectors. The main organizational approval (ODA) is a "delegation" from the FAA, and is not an "operating" certificate. The main certificates issued by the FAA are specific product design approvals (for example, TCs, STCs, TSOs, or PMAs) issued to a holder. The design approval confers no "operating" privileges to the holder, other than the ability to assign a production organization to produce the design under an FAA certificate or approval.

To apply the SMS requirements to non-certificated organizations, the SMS requirement should be applied through part 21 to design approval applicants and DAHs as follows:

- Part 21 TCs and STCs. If the design to be approved/held meets the SMS applicability threshold, the applicant/holder should have an SMS per part 5.
- Part 21, subpart K—PMA. If the article meets the SMS applicability threshold, the applicant/holder should have an SMS per part 5.
- Part 21, subpart O—TSO. If the article meets the SMS applicability threshold, the applicant/holder should have an SMS per part 5.

Regarding production organizations, it would be consistent to have any organization producing a part/product that requires an SMS from a design approval perspective (as outlined above) to also have an SMS as follows:

• Part 21 Production (PC for TCs and STCs). If the product to be produced meets the SMS applicability threshold, the applicant/holder should have an SMS per part 5.

5.1.4 SMS ACCEPTANCE/OVERSIGHT

The FAA should develop an approval document (letter or certificate) to indicate FAA acceptance of the organization's SMS and any associated limitations and conditions. In addition, the FAA should establish methods to address deficiencies in the organization's SMS, because FAA certificate action will not be effective for an organization without an FAA certificate. The traditional means of FAA compliance and enforcement procedures can be updated to clarify how to apply to non-certificated organizations to ensure compliance with SMS implementation requirements.

5.1.5 SMS WITH DIFFERENT DESIGN AND PRODUCTION ORGANIZATIONS/DIFFERENT COUNTRIES

Given the evolving nature of the D&M industry, DAH organizations with different production organizations, and even production organizations in different countries (and therefore under different State of Design/State of Manufacture authorities), are not uncommon. From an SMS perspective, this means the potential for two SMS systems in different companies and potentially under different national authorities. In these cases, there should be appropriate coordination between the two organizations' SMS systems and their regulators (if in different countries). The ARC recommends the FAA work with other authorities and SMS teams to review requirements and establish guidance as required.

5.1.6 SMS IMPLEMENTATION

SMS implementation strategy for D&M organizations should maintain alignment with the ICAO Safety Management Manual (SMM). The document allows a phased approach to implementation, with a manageable series of steps to follow, including allocation of resources, effectively managing the workload associated with implementation, and providing for a realistic implementation of an effective SMS (avoiding a "cosmetic compliance").¹

There are four levels to SMS implementation:²

- Level 1: Planning and Organization. This consists of orientation, defined safety policy, gap analysis (preliminary and detailed), an SMS training plan, and an implementation plan.
- Level 2: Reactive Processes, Basic Risk Management. The organization develops and implements a basic SRM process. The organization is engaged for further SMS development.
- Level 3: Proactive Processes, Looking Ahead—A Fully Functioning SMS. This consists of applying the SRM to the initial design of systems, processes, organizations, and products.
- Level 4: Continuous Improvement, Continued Assurance. This level is achieved when all SMS processes have been implemented and their performance has been verified.

5.1.7 EFFECTIVE SMS REQUIREMENT DATES FOR D&M ORGANIZATIONS

The timeline for initial SMS implementation for D&M organizations meeting the SMS working group's applicability threshold would be similar to that for the current proposed § 5.1, Applicability. Modification of the appropriate sections of part 21 would point to the SMS requirement of the proposed part 5. The ARC recommends § 5.1 refer to the period of time following the *approval* of an implementation plan, rather than the *effective date* of the final rule, as the deadline for an accepted SMS. This would allow sufficient time to work with the FAA on revising the submitted implementation plan and to develop and enact proper processes.

¹ FAA: SMS Implementation Guide, Rev 3

² D&M SMS Pilot Project Guide, Rev C

For organizations that have already implemented an SMS, evidence of a suitable SMS could be submitted to the FAA within the implementation plan submittal deadline. The suitable evidence would consist of details such as a detailed gap analysis, previous voluntary implementation plan, and records from all levels of the active SMS that demonstrate meeting the requirements. If deficiencies are found, a streamlined implementation plan would be submitted to address those deficiencies within the mandated timeline to have an accepted SMS. Otherwise, if found acceptable, the SMS would be accepted immediately.

5.1.7.1 Existing DAHs/PAHs Meeting the SMS Applicability Threshold

Existing D&M organizations holding a design or production approval should submit an implementation plan within 6 months of the rule's effective date, and have a full SMS within 3 years of an approved implementation plan.

For new applications by an existing D&M holding a design or production approval, the requirement to have an SMS should still be dictated by the above requirements.

5.1.7.2 New Applicants for DAHs/PAHs Meeting the SMS Applicability Threshold³

New applicants for a design or production approval would be held to the same timelines as current D&M organizations. However, the implementation plan should be submitted at the time of application and must be approved by the FAA before issuance of a design approval, production approval, or PMA. The full SMS would be required within 3 years of the implementation plan approval. A requirement to have a fully functioning SMS on initial design approval, production approval, or PMA issuance may be unrealistic or cost-prohibitive due to resources required within a short timeframe. Because the timeline required to have a fully implemented SMS is the same as that required for current D&M organizations at the final rule issuance, allowing new D&M organizations to enter the market with the same grace period afforded to the existing D&M organizations retains fair treatment of new organizations. This process also requires the applicant to show full understanding of the COS and safety requirements (that is, an understanding of certain aspects and responsibilities of part 21 requirements) before a design approval, production approval, or PMA can be issued.

³ That is, currently not holding a design or production approval.

5.1.7.3 Proposed Regulatory Text for SMS Implementation

The following provides the ARC's recommended regulatory text for the implementation of SMS upon design and production approval holders.

The changes to § 21.135, Organization, are intended to satisfy the requirements for an SMS process for production certificate holders.

§ 21.135 Organization

Each applicant for or holder of a production certificate must-

(a) Provide the FAA with a document describing how its organization will ensure compliance with the provisions of this subpart. At a minimum, the document must describe assigned responsibilities and delegated authority, and the functional relationship of those responsible for quality to management and other organizational components.

(b) Within 3 years after the FAA approval of the implementation plan for design and manufacturing, each applicant for or holder of production certificate for a product or change to a product whose failure could directly prevent continued safe flight and landing must have a safety management system that meets the requirements of part 5 of this chapter (except § 5.27) and is acceptable to the Administrator. The applicant for or holder of a production certificate required to have a safety management system must—

(1) Submit an implementation plan to the FAA Administrator for review no later than [date 6 months after the effective date of the final rule for existing production certificate holders, or with the application for a new production certificate].

(2) The implementation plan may include any of the certificate holder's existing programs, policies, or procedures that it intends to use to meet the requirements of part 5 of this chapter.

The changes to § 21.305, Organization, are intended to satisfy the requirements for an SMS process for PMA certificate holders.

§ 21.305 Organization

(a) Each applicant for or holder of a PMA must provide the FAA with a document describing how its organization will ensure compliance with the provisions of this subpart. At a minimum, the document must describe assigned responsibilities and delegated authority, and the functional relationship of those responsible for quality to management and other organizational components.

(b) Within 3 years after the FAA approval of the implementation plan for design and manufacturing, each applicant for or holder of a PMA whose failure could directly prevent continued safe flight and landing must have a safety management system that meets the requirements of part 5 of this chapter (except § 5.27) and is acceptable to the Administrator. The applicant for or holder of a PMA required to have a safety management system must—

(1) Submit an implementation plan to the FAA Administrator for review no later than [date 6 months after the effective date of the final rule for existing PMA certificate holders, or with the application for a new PMA].

(2) The implementation plan may include any of the certificate holder's existing programs, policies, or procedures that it intends to use to meet the requirements of part 5 of this chapter.

The changes to § 21.605, Organization, are intended to satisfy the requirements for an SMS process for TSO certificate holders.

§ 21.605 Organization

(a) Each applicant for or holder of a TSO must provide the FAA with a document describing how its organization will ensure compliance with the provisions of this subpart. At a minimum, the document must describe assigned responsibilities and delegated authority, and the functional relationship of those responsible for quality to management and other organizational components.

(b) Within 3 years after the FAA approval of the implementation plan for design and manufacturing, each applicant for or holder of a TSO whose failure could directly prevent continued safe flight and landing must have a safety management system that meets the requirements of part 5 of this chapter (except § 5.27) and is acceptable to the Administrator. The applicant for or holder of a TSO required to have a safety management system must—

(1) Submit an implementation plan to the FAA Administrator for review no later than [date 6 months after the effective date of the final rule for existing TSO certificate holders, or with the application for a new TSO certificate holder].

(2) The implementation plan may include any of the certificate holder's existing programs, policies, or procedures that it intends to use to meet the requirements of part 5 of this chapter.

The following text would be included as part of any new regulation for minimum design approval applicant and holder regulatory requirement to satisfy the requirements for an SMS process for design approval holders.

(e) Within 3 years after the FAA approval of the implementation plan for design and manufacturing, each applicant for, or holder of, a type certificate or for a product or change to a product whose failure could directly prevent continued safe flight and landing must have a safety management system that meets the requirements of part 5 of this chapter (except § 5.27) and is acceptable to the Administrator. The applicant for or holder of a type certificate required to have a safety management system must—

(1) Submit an implementation plan to the FAA Administrator for review no later than [date 6 months after the effective date of the final rule for existing design approval holders, or with the application for a new design approval].

(2) The implementation plan may include any of the certificate holder's existing programs, policies, or procedures that it intends to use to meet the requirements of part 5 of this chapter.

5.1.7.4 New Applicants for DAH/PAHs Meeting the SMS Applicability Threshold <u>After</u> 3 Years of Rule Effective Date

Once 3 years has passed since the rule's effective date, a full approved/accepted SMS would be a requirement for the applicant at the time of application. The 3-year phase-in period for the SMS requirement would allow sufficient time for new applicants to prepare before proceeding with an application.

5.1.7.5 Voluntary Implementation for Current and New D&M Organizations

Voluntary submittals of implementation plans or evidence of an existing active SMS can be made at any time. The FAA resources would be prioritized to those D&M organizations requiring an SMS based on the applicability threshold; therefore, those D&M organizations with the highest risk would be being evaluated first. Subsequent prioritizations could be based on FAA RBRT criteria if resource restraints prevent immediate FAA review.

5.2 OTHER PARTY SMS ACCREDITATION

5.2.1 SUMMARY

The international scope of ICAO Annex 19 suggests SMS requirements can be uniquely defined on the country/state level. This creates a concern within industry because of the potential of having to comply with a complex array of SMS requirements. The ARC believes developing an "other" party recognition or accreditation could solve this problem. The proposal presented here is to adopt a model similar to the current aviation standards associated with QMS (via the International Aerospace Quality Group (IAQG)) to apply to SMS for D&M organizations.

5.2.2 DEFINITIONS

AS9100/AS9110. International standards that include ISO 9001, Quality Management System, requirements and specifies additional requirements for a QMS for the aerospace industry.

International Organization for Standardization (ISO) 9001. A quality systems model for quality assurance in design, development, production, installation, and servicing.

5.2.3 ORGANIZATIONS

IAQG—International Aerospace Quality Group. "The IAQG Council sets the policy, purpose, and objectives of the organization and drives initiatives to meet the goals and objectives of the organization. There are 26 Council Voting Members; 10 from the Americas, 10 from Europe, and 6 from Asia-Pacific."⁴

AAQG—**Americas Aerospace Quality Group.** "The AAQG is a cooperative organization within the aerospace industry in the Americas (including North, Central, and South Americas). Its processes are established in a set of agreed, documented, operational procedures."⁵

5.2.4 "OTHER PARTY"-IAQG DEFINITION

"The Other Party Scheme is based on-

- The use of identical or equivalent international, sector, and national standards based on the 9104/1/2/3 trilogy of standards;
- An industry oversight system at international, sector and national levels to ensure that scheme's requirements are fulfilled; and
- Auditors authenticated against identical requirements."⁶

5.2.5 "OTHER PARTY" MANAGEMENT TEAM STANDARDS

9104-1 Requirements for Aerospace QMS are-

- Certification/Registrations Programs, and
- Globally harmonized standard defining the certification/accreditation process.

9104-2 Requirements for Oversight of Aerospace QMS Registration/Certification Programs

• Globally harmonized standard defining the surveillance and oversight processes.

9104-3 Requirements for Aerospace Auditor Competency and Training Courses

• Globally harmonized standard defining the auditor qualification and auditor training processes.

5.2.6 "OTHER PARTY" SMS

Requirements for SMS:

• Based on ICAO Annex 19.

Oversight of SMS:

• Based on SMS ICG Evaluation Tool.

⁴ http://www.sae.org/iaqg/organization/council.htm.

⁵ http://www.sae.org/aaqg/.

⁶ http://www.sae.org/iaqg/organization/opmt.htm.

Auditor Competency and Training Courses:

• Analogous to IAQG.

The proposed plan is to work initially with the AAQG toward a new set of standards (AS91xx-1/2/3) based on the above to leverage existing local and international groups.

The primary advantage of this approach is the ability to begin implementing this activity immediately. It also serves as one of the bricks in the "building blocks" approach detailed in section 10 of this report. This timely approach also provides a means for D&M companies that are voluntarily implementing SMS aligned with ICAO Annex 19 intent to get validated without waiting on proposed regulations (part 21/part 5) that face cost-benefit challenges.

5.3 Additional SMS Recommendations

The SMS Working Group, comprised of industry and FAA subject matter experts, developed SMS regulatory material and a basis for preamble, policy, and guidance material as provided in this report. However, the working group determined additional work is necessary to produce detailed guidance material for the practical implementation of SMS requirements to part 21 design and production approval holder organizations and processes. The ARC tasked the working group to develop a framework for additional guidance, which will be provided as an addendum to the SMS Working Group Report (included as appendix G to this report). The ARC recommends that this guidance material be fully developed by the subject matter experts that comprised the SMS Working Group before the issuance of an SMS notice of proposed rulemaking (NPRM), so it is available for concurrent review and comment.

Specific guidance that requires further development includes-

- How safety objectives are established;
- Evaluating the performance of an organization's SMS;
- Development of an "operational" definition of a hazard throughout the life cycle of a product;
- Acceptable criteria for the effectiveness of safety risk controls at a system level;
- A process the D&M organization implements to meet § 5.55(b), Safety risk assessment and control, and the extent to which the FAA is engaged; and
- Flexibility in applying SMS to the D&M organization.

Notwithstanding the gap assessment (refer to section 4.1 of appendix G to this report) that identifies regulatory gaps between part 21 and part 5, the D&M community has established business practices that accomplish, to varying degrees, the intent of SRM and safety assurance as part of its design, certification, production, and continued airworthiness efforts (although not using the SMS terminology). However, much of what has been written about SRM and safety assurance has been from an operational perspective (for example, air carrier flight operations or airport air traffic operations), and limited documentation/guidance exists for how these SMS elements relate or are equivalent to what D&M organizations do as part of their compliance to the current airworthiness regulations and design and certification procedures.

To ensure the intent of SRM and safety assurance is applied to D&M organizations in the most effective and efficient manner, the ARC recommends the interpretations and explanations provided herein be included in any preamble, policy, or guidance material. Also, depending on the future state of delegation and approved or certificated organizations, regulation, policy and guidance should allow an organization to use the same processes and procedures to satisfy the intent of the equivalent regulations. This approach would minimize the economic burden to industry while maximizing the enhancement to aviation safety.

The ARC has developed an SMS CONOPS describing the intent of the part 5 SMS framework (safety policy, SRM, safety assurance, and safety promotion) for D&M organizations as it applies to each life cycle phase (design and certification, production and airworthiness certification, and continued airworthiness) of a product or article. <u>The ARC recommends the CONOPS form the basis for the development of preamble, policy, and guidance material for D&M organizations. The ARC also recommends that, as described in the CONOPS, existing processes and procedures should be considered as meeting the intent of part 5.</u>

The following SMS recommendation was not directly tied to the ARC charter or its taskings, but supports the § 21.3, Reporting of failures, malfunctions, and defects, requirements and proposed changes discussed in section 8.2 of this report.

The ARC recommends the FAA develop an approach to make fleet data already provided to the FAA (hours, flights, reported failures, malfunctions, and defects and service difficulty reports) readily available to D&M organizations, in support of executing SRM (§ 5.71, Safety performance monitoring and measurement).

6.0 DESIGN ORGANIZATION MODEL AND FRAMEWORK

A DO must have systems in place that ensure the FAA may rely on any and all statements of compliance when it issues a certificate. That activity must occur under established minimum requirements of a DO.

An integrated and systematic approach to compliance and safety therefore encompasses several elements, including an organization, CAS, and SMS. Successful execution of these elements should enable the continued growth of a compliance and safety culture within the DO. Although the DO should satisfy each of the elements of the systems discussed below, the elements may be arranged or grouped differently or encompassed within organizational systems with different names to accomplish the intended purpose.

The DO must be able to establish and show that the organization—

- Possesses the required competence to determine that the certificate holder's designs meet all applicable airworthiness standards within the scope of the DO certificate.
- Has procedures for assuring compliance to the airworthiness standards.
- Maintains the essential process controls to deliver repeatable and sustainable compliance.

Recommendation 1c—Systems Approach to Certification – DO Requirements: The ARC recommends establishing regulatory requirements for the issuance and oversight of voluntary certificated DOs, including the necessary compliance assurance, safety management, and controls to make all compliance determinations through applicant showing and verification processes. Through FAA certificate management oversight and direct project involvement in defined risk-based areas, the FAA may rely on the DO compliance determinations to make its finding for the issuance of a design approval. This report builds on the recommendations submitted to the FAA by the CDO ARC in May 2008. The ARC recommends a building block approach to implementing DO, which includes establishing a clear accountability framework, transitioning the FAA's oversight of design activities to a centralized systematic model, optimizing use of ODA, and implementing new organizational and SMS requirements. With successful implementation of these building blocks and voluntary DO, the ARC supports future rulemaking to consider mandatory DO.

The DO Working Group Report contains a number of detailed recommendations proposed when the ARC envisioned the DO to be mandatory. Because these recommendations only apply to a mandatory DO and the ARC is now recommending the DO be voluntary, these working group recommendations have not all been brought forward into this final ARC report. However, the intent of the building block approach is to eventually get to a DO during the "Transformational" phase, at which time the detailed information provided in the working group report becomes relevant and should be reviewed. Additionally, the ARC recommends that proposed regulations, preamble language, and guidance material should be discussed as a follow-on activity to mature the information provided in this report.

6.1 DO AND ORGANIZATION CONTROL SYSTEMS (OCS)

The ARC does not recommend that a particular organizational structure be required; however, certain functional roles, aspects, and elements must be defined in the FAA-approved DO procedures manual. Each holder of a DO certificate must have a procedures manual that defines the procedures and processes to be used (that is, OCS) that meets the requirements specifically required by the DO regulation to be in the procedures manual.

The holder of a DO certificate must follow the procedures in the manual. The manual must be in the English language and retrievable in a form acceptable to the FAA.

The DO procedures manual contains the DO's procedures for meeting its regulatory requirements. The manual must address all relevant DO requirements.

The DO procedures manual processes and procedures must be sufficient for the FAA to determine that regulatory compliance is properly addressed. The manual is intended to be a top-level document that will guide the development of lower-level processes and work instructions that the DO can develop and change as it finds necessary (that is, without the need for FAA approval) to meet the top-level requirements and objectives. If the FAA determines the procedures manual lacks the detail necessary to ensure regulatory compliance, it will request a change to the manual. The DO is obligated to respond to the FAA's request within an agreed-upon timeframe. The ARC recommends further discussion on the necessary level of detail for inclusion in the procedures manual and the appropriate reasons/rationale for FAA requests for changes to the procedures manual.

The procedures manual must be consistent with all issued FAA regulations and guidance related to the proper function of a DO.

The DO procedures manual may be in any format proposed by the DO and acceptable to the Administrator. There is no expectation that each DO procedures manual would be formatted the same.

The following are the organizational and OCS requirements:

1. Identified DO Executive. The DO Executive is accountable for all activities covered within the scope of the DO certificate. This executive must be identified by name and position within the company. The DO Executive may also act as the primary point of contact (POC) for the DO. The DO procedures manual must contain an explanation of the reporting relationships between the DO Executive and senior company management, as well as the organizational relationships within the DO.

Identified DO point(s) of contact. The DO POC is the person(s) within the DO with whom the FAA will communicate. A formal list of POCs must be maintained by the DO. The DO POC must be familiar with the DO processes and the applicable FAA regulations consistent with the scope of the DO certificate. The DO POC must also have unencumbered, but not necessarily

direct, access to the DO Executive. Additionally, defined procedures for communication between the DO and the FAA, including agreement on expectations and expediency, will be stated.

Each DO certificate holder will have qualified staff, as appropriate to DO privileges and obligations. The DO is responsible for ensuring the staff in technical departments has the experience, training, and authority to be able to discharge their allocated responsibilities, and that these, together with the accommodation, facilities, and equipment, are adequate to enable the staff to achieve the airworthiness objectives for the product. Refer to the NATCA dissenting opinion in section 12 of this report.

A process for verification of personnel qualifications. The scope of personnel is intended to focus on those people who hold accountability for maintaining the organizational oversight of the DO and those people with prime accountability for the design management system (DMS), compliance certification system (CCS), and SMS. Personnel qualification includes essential competencies, experience, and training.

A process for verification of systemic performance of duties. This includes procedures for planning, conducting, and documenting internal audits to ensure compliance with the approved procedures manual, DMS, CCS and SMS. The procedures must include reporting of internal audit results to the manager responsible for implementing corrective and preventative actions.

A process for retaining records that are required to be produced. This includes the identification of records that must be retained, the method and means of storage and retrieval, control and access privilege, and retention period. These records typically comprise DO procedures manual approvals (including changes to the manual), design approval records, design change approvals, internal audit records, and CCS records.

A process for identifying which certification projects require an application for establishing details of the project list, and how often this information should be provided to the FAA. (Refer to section 6.4 of this report)

A process for notifying the FAA if circumstances prevent the DO from meeting DO obligations.

The process, timetable, and authority for obtaining and agreeing on changes to the DO procedures manual, DMS, CAS, and SMS.

6.2 COMPLIANCE ASSURANCE SYSTEM

The DO applicant must demonstrate that it has established and is able to maintain a regulatory CAS for—

- The control and management of the design approval(s),
- Design changes of products and articles covered by the scope of the certificate, and
- Any production activities associated with those design approvals.

The CAS should result in assurance that the compliance determinations are correct and consistent with what would result from an independent skilled review of compliance.

The DO applicant's regulatory CAS must contain a means to provide assurance that the design and design changes of the applicant's products and articles comply with the applicable airworthiness standards. This is presented in this report as being accomplished through a DMS and CCS.

After the issuance of the DO certificate, any changes to the CAS materially affecting compliance with the certification basis or airworthiness standards must be submitted to the FAA for approval before implementation. The DO must identify to the FAA how the proposed changes to the CAS will result in continued compliance after implementation.

Assurance of compliance with the airworthiness standards, including the performance of suppliers, is of critical importance to the success of DO. No FAA designees are used by the DO. Therefore, the quality of the DO's processes for determinations of compliance and process adherence, and the robustness of the CAS are the basis for enabling the FAA to rely on the DO's statement of compliance when making its finding.

The DO CAS is composed of two elements:

- 1. **Design Management System.** The DMS is the system by which a DO creates and maintains product design data. DMS requirements are—
 - A process for creating and maintaining configuration and control of design data. Establishing a structured and controlled system for the development of design data, the control of changes to the data, and the assurance that the descriptive design data is current and approved is fundamental to this process.
 - A process for retaining, retrieving, protecting, and maintaining design descriptive and substantiating data. This process should also include any agreements between the DO and FAA regarding availability of data, access to the data, and any electronic system requirements required to view the data.
 - A process for engineering supplier control that defines how design activities performed externally to the DO are controlled, validated, and assured. This also includes inter-DO relationships (refer to section 6.5.4 of this report).
 - A process for creating eligible data that can be used for showing compliance when applied to a particular certification activity.
 - A process for receiving and processing safety data related to approved designs.

Note: "Safety" data is meant to pertain to information relevant to identification of product safety hazards, risk analysis, and mitigation, as appropriate.

• A process for applying relevant "safety" data to the DMS for purposes of continuous improvement.

Compliance Certification System. The CCS is the system by which a DO ensures product compliance (showing of and verification of, as appropriate) to the applicable airworthiness standards. The CCS requirements are—

- A process for identifying and/or establishing regulatory certification basis. This process includes an assessment of the product's intended usage and determination of the applicable airworthiness standards. The establishment of a product's certification basis may be a pre-decisional FAA approval based on the agreed processes defined within the DO procedures manual.
- A process for identifying regulatory changes that affect the design system. The DO must have an active means of monitoring regulatory changes that affect the product(s) for which the organization is responsible and evaluating the need for implementing changes or taking other actions. This includes any activities resulting from the issuance of an airworthiness directive (AD) against a DAH's product. A defined system that shows compliance to the applicable airworthiness standards under § 21.20(a), and how verification of compliance will be conducted including any process for assuring independence and objectivity. The "show" responsibility is identical to that already prescribed under § 21.20(a). However, in the DO model, the "show" and "verification" functions are intended to be accomplished within the DO.
- A process for using memorandums of cooperation (MOC) that are acceptable to the FAA, including, as appropriate, a process for verification and validation of analytical tools and a process for managing testing and inspection required to support each showing of compliance. This would include conformity inspections used to support showings of compliance. It is expected that this process will delineate between showings within well-established MOCs and procedural distinctions when addressing a MOC that is new or novel.
- A process for declaring to the Administrator that a design is in compliance with the applicable airworthiness standards at the time the design approval is requested. This documentation fulfills the requirement under § 21.20(b).
- A process for reporting other approvals such as design changes, changes to Instructions for Continued Airworthiness (ICA), and approval of repair data.
- A process for receiving, processing, and implementing corrective action regarding "compliance" data related to approved designs. If during the course of regular business or as the result of a finding during any internal or externally conducted audit a noncompliance to the approved procedures or airworthiness standard is identified, the mechanism for evaluating the risk of the noncompliance and identification of any corrective actions must be defined.

DO Show and Verify Compliance Certification Functions. The CCS processes must include a showing of compliance by the DO to each applicable airworthiness standard. The following principles provide guidance in this regard:

- Show and verify functions are part of the CCS.
- The show function is a demonstration of compliance to an airworthiness standard (that is, 14 CFR parts 23, 25, 27, 29, 33, and 35) and is composed of substantiating data,

statements, and/or other acceptable methods of demonstration (for example, acceptable MOC).

- The verify function is an independent check, or equivalent, of the show function.
- The show and verify functions are distinct functions or activities, where appropriate. The degree of independence between the individuals performing the show and verify functions or activities should be established in the DO procedures manual.

A DO determination of compliance is a showing with an independent verification function unless the DO procedures manual identifies a specific area where a verification is not required or where the procedures manual identifies a process that allows the DO to assess risk to make decisions on whether a verification step is required.

• The showing of compliance requirement remains identical to that which exists in the current regulatory system. There is no intent to establish a changed requirement for showing of compliance for a DO.

Tools Used for Performing Compliance Activities. Where the system is dependent on the use of a tool for performing some of the processes and methods, means must be provided to ensure—

- The tool performs its required function,
- The tool and its output are being controlled under a configuration management program,
- The tool is periodically verified for its applicability regarding the processes and methods for which it is intended to apply, and
- A record is kept of the use of the tool to accomplish the compliance activity.

6.3 SAFETY MANAGEMENT SYSTEM

The SMS requirements for a DO are to be provided in a separate report by the ARC's SMS Working Group.

The ARC anticipates the SMS to be the system for actively monitoring product safety, identifying and managing risks to product safety, and promoting a strong safety culture throughout the organization, which would likely consist of—

- Safety policy and objectives including management commitment, responsibilities, accountabilities, key safety personnel, and coordination of emergency response planning;
- Safety risk management including hazard identification, and risk assessment and mitigation;
- Safety assurance through performance monitoring, measurement, change management, and continuous improvement; and
- Safety promotion through training, education, and communication.

6.4 DO PROJECT APPLICATIONS AND ACTIVITY REPORTING

The ARC believes the DO certificate holder may complete many projects without notifying the FAA, because the DO will be making all determinations of compliance in accordance with its FAA-approved procedures manual. There are other projects the FAA must be immediately aware of, as they require the FAA to validate the existing type certification basis or establish a new one.

Part 21 already specifies when an application must be made to the FAA. This includes applications for TC, STC, and PMA. The existing required applications could be used to discriminate between those projects that required FAA notification and those that did not. If the project would require an application under part 21, the FAA must be notified when the project is initiated. Such projects would be any new design approval, amended TCs requiring a new model designation, new STCs, and any project that would be expected to have a revised type certification basis under § 21.101, Designation of applicable regulations. Any change that does not rise to this level will be handled by the DO under its approved procedures manual.

The details of the project list, how often it should be provided to the FAA, and how the FAA should be notified of projects requiring an application are some of the matters that should be discussed with the FAA and included in the procedures manual.

6.5 DO RELATIONSHIPS AND SUPPLIER CONTROL

6.5.1 GENERAL REQUIREMENTS

A design approval applicant or DAH has sole responsibility for proper control of all its suppliers, whether they are suppliers of engineering services, manufacturing of pre-production parts, special process, or any other part of its compliance responsibilities. Should there be any deficiency or noncompliance on the part of a supplier, even if it is a supplier of compliance determinations, products, parts, or appliances, the FAA holds the applicant or holder of the design certificate responsible to correct the deficiency. This longstanding principle remains for the DAH even if the DAH is also the holder of a DO certificate.

The DO must qualify its engineering suppliers, provide oversight, and define the process by which suppliers function within the DO system. This means a DO may authorize suppliers to make determinations of compliance only after the DO has evaluated the supplier's system and determined that the supplier is qualified to act in that capacity for the DO. This does not mean suppliers must adopt the DO's system of processes, but it does mean the DO must determine the system of processes to be used by the suppliers are acceptable and meet the DO's requirements.

To fulfill its supplier management responsibilities, a DO must have as part of its system a process by which it will determine the appropriate level of oversight required for its suppliers. The DO should consider such things as—

- The criticality of the design,
- Its experience with the supplier,
- The supplier's standing as an FAA-approved DO or as a holder of other FAA design approvals or delegations,

- Additional compliance determinations to be made during integration testing of the design, and
- Any other appropriate factors.

The ARC recommends DO certificate holders be able to cooperate with other companies to pool supplier oversight responsibilities, in a manner similar to what is currently done by manufacturing facilities and airlines under the Coordinating Agency for Supplier Evaluation (C.A.S.E.), http://www.caseinc.org/. As an example, several companies buying avionics components from a single supplier may cooperate in the surveillance of that supplier by allowing one of the companies to conduct the audit and the other companies to use the results as if they conducted the audit themselves. To gain the credit for such pooling of audit requirements, it is essential that the supplier processes be consistent across all companies, or that the company auditing the supplier assess all the requirements of those companies wishing to share the benefits of the single-party audit.

In selecting suppliers, the DO must consider that, for both engineering and production suppliers, there must be means for the FAA to gain access to the facility for the purposes of DO oversight.

6.5.2 OVERSIGHT OF FOREIGN SUPPLIERS

Oversight of foreign suppliers is required whether or not a bilateral agreement exists between the FAA and the country in which a foreign supplier is located. A DO may authorize foreign suppliers to make determinations of compliance only after the DO has evaluated the suppliers' systems and determined they are qualified to act in that capacity under the DO processes.

To fulfill its supplier management responsibilities, a DO may also propose, as part of its quality system, a foreign supplier oversight process for its design services supplier, just as companies currently do for parts suppliers under AC 21–1B, Production Certificates. This process could be based on using an approved organization as a supplier to its DO, for example, a contract with an EASA DOA holder in good standing. In doing so, it must be recognized that the foreign civil aviation authority (CAA) may not be performing any oversight of that activity if it does not lead to an approval under the CAA. If representatives of an EASA DOA, for example, are authorized by the DO to make compliance determinations, those determinations must be acceptable and meet the DO's requirements.

The DO remains fully responsible for all compliance determinations made by the foreign suppler holding an organizational approval from its cognizant CAA, just as it is for all other compliance determinations. However, in its supplier oversight function, the DO may take credit for the surveillance of the supplier by its CAA. That credit would result in a reduced need for oversight by the DO, and the foreign supplier oversight process should be defined within the supplier surveillance portion of the DO procedures manual. The DO, for example, could review periodic reports from audits performed by the supplier CAA or establish some other means of tracking supplier performance. The DO should consider the criticality of the design, experience with the supplier, and other factors in determining the degree of oversight necessary, as it does with all supplier oversight. The DO oversight methodology applied to foreign suppliers having capabilities recognized by their respective CAA would be evaluated as part of the FAA's

oversight function. Additionally, the DO supplier oversight process could include a qualified third-party organization (such as Bureau Veritas).

6.5.3 THE NEED FOR SPECIALTY SERVICE PROVIDERS

For the design and production companies in the aviation system, the FAA has been moving away from a system of approvals based on the use of individual designees, and toward organizational delegations based on demonstrated and approved processes within a company. This is especially true for those seeking or holding original design approval certificates. The DO concept further propagates this trend.

At the same time, many companies in the aviation community are becoming very specialized in their ability to perform unique technical services. In certain highly technical areas, the number of qualified organizations that can perform specialized services has been reduced to a critical few. The designers and producers of approved products and articles rely on these specialty services to supplement their capabilities. The 2008 CDO ARC Report recognized a need to create a new process wherein these specialty companies are recognized for their capability, and that capability can be used to supplement DO (and ODA) compliance activities, as well as those of other applicants. The 2008 CDO ARC chose to call these companies "Specialty Service Providers" (SSP).

The ARC recommends the FAA give priority to developing a means for recognizing an accreditation system for these SSPs. This concept could encompass technical specialties ranging from the more complicated (such as flammability, dynamic seats, icing, electromagnetic interference, and high-intensity radiated fields) to the more routine specialties (such as materials testing, nondestructive inspection processes, and environmental testing of components). These are only a few examples of the scope of activities that could be included under this concept.

The ARC recommends that industry develop this concept with the direct participation of the FAA because of the variety of issues that must be addressed and the need to create industry consensus standards.

- The SSP could perform compliance with industry consensus standards acceptable to the FAA and DO as evidence of compliance with specific airworthiness standards, resulting in a determination of compliance.
- The possibility of third-party approval and oversight of SSPs should be considered.
- The data developed by SSPs must be directly useable by all applicants without further verification of the data's integrity.
- The FAA should pursue international acceptance of the SSP system.
- Although SSPs may have individuals who are designees of the Administrator, such individuals do not exercise delegation in the course of a DO project as a SSP.
- Persons using SSPs must do so under their supplier control system. The amount of supplier oversight conducted by the user of these services can take into consideration the accreditation of these SSPs.

- The continued compliance responsibility of applicants who use SSPs is not reduced by the fact that the SSPs are recognized by the FAA for their expertise. The TC, PMA, or part approval holders still have the total responsibility for initial and continued compliance of the design approvals they hold, and the resolution of all COS issues.
- The ARC recognizes that there are different methods that may be used to implement the SSP concept, and that the concept is not necessarily tied to implementation of the DO. Regardless of the method of implementation, the ARC recommends the FAA give priority to developing a means for recognizing an accreditation system for SSPs (for example, Nadcap or similar) and the system be made available to the aviation community.

6.5.4 INTER-DO RELATIONSHIPS AND RESPONSIBILITIES AS DESIGN PARTNERS

Inter-DO relationships will be controlled by a supplier interface document (refer to section 6.8 of this report). This interface document will allow the supplier DO to use its own procedures and processes within its scope of authority as a DO. The applicant DO will identify how supplied data (and potentially determinations of compliance) will be assessed for integration and applicability to the product.

6.5.5 AGENT DESIGN ORGANIZATION (ADO)

The ARC discussed the creation and recognition of a DO that has been contracted to act on behalf of a design approval applicant. The ARC refers to such a DO as an ADO. An ADO may or may not hold a design approval and must meet the minimum DO requirements specified within this report. The ADO provides an avenue for an interested design approval applicant to use the professional DO services that the applicant either does not have or does not have the financial ability to create. The use of an ADO promotes consistency in design activity (safety enhancement), an accountable DO (responsibility), and may reduce the necessary aviation system assessment and oversight resources. The ARC believes the ADO role fills a critical need in the DO model where a full DO would be more than was needed and would have an adverse impact on an applicant's ability to create needed designs or design changes, particularly regarding small business applications.

The following text from EASA 21.A.2 describes a similar application of the ADO concept:

"Undertaking by another person than the applicant for, or holder of, a certificate. The actions and obligations required to be undertaken by the holder of, or applicant for, a certificate for a product, part, or appliance under this section may be undertaken on its behalf by any other natural or legal person, provided the holder of, or applicant for, that certificate can show that it has made an agreement with the other person such as to ensure the holder's obligations are and will be properly discharged.

Refer to section 7.2 of this report for more information about ADOs.

6.6 BUSINESS STRUCTURE VARIATION

The ARC recognized that additional business structures may exist where a DO is part of a business structure in which companies are not in a traditional supplier relationship. An example of such a business structure is a consortium, where each company is an equal business partner and is not considered a supplier to the other. In some cases one business partner is U.S.-based while the other is foreign-based. Also the consortium company may or may not be U.S.-based. As such, neither company has complete oversight, in the supplier sense, of the other company. Each business partner wants recognition of their respective DO in a certification project. Such is the situation, in any business structure, where one company does not have full supplier oversight responsibilities over another company.

Industry currently has such business structures with U.S.-based companies and foreign-based companies. As the aviation business continues to grow globally, industry members foresee continued growth of these business structures to mitigate associated business risks and leverage various strengths of different companies.

The ARC discussed the notion that when the consortium company is foreign-based, a U.S.-based DO should be able to manage the integration responsibilities for the FAA portion of a certification project, including the project statement of compliance to the applicable airworthiness standards, knowing that the U.S.-based DO may not possess or have access to all project descriptive and/or substantiating data. This is a result of the need to protect proprietary methods and information within each respective company. Completion of all required determinations of compliance must be documented to support the project statement of compliance.

The ARC also discussed that a consortium may consist of two U.S.-based DOs. Industry recognizes there is one applicant and certificate holder for each TC; that has historically been the consortium company. The FAA has worked with consortium companies to execute shared-responsibility agreements between the members, which are acceptable to the FAA. That has allowed each consortium member to autonomously execute its role independent of the other consortium members, including findings of compliance under FAA delegation, thus protecting its proprietary data. A similar FAA-accepted working agreement could describe how those autonomously performed design and compliance determination processes under DOs are integrated into a single type design, TC, and continued airworthiness process.

The ARC believes further discussion is needed between the FAA and industry to develop how this would be accomplished. <u>Therefore, the ARC recommends further development of how such business structures will be accommodated under the proposed DO framework.</u> Input from industry, especially those who currently have consortium programs, should be further considered.

6.7 SCOPE, LIMITATIONS, AND PRIVILEGES OF DO CERTIFICATES

There are many variations in design and production organizations and their products throughout the aviation system. They range from organizations dealing with a full line of products, like transport airplanes, high-tech general aviation aircraft, helicopters, and high-bypass engines, to PMA holders with a more narrow focus. In some cases, FAA compliance approvals for the activities of these organizations are made either directly by FAA resources, or by using individual or organizational delegation approvals from the FAA. Repair stations may have ODA authorizations or their own company designees, or may contract with consultant designees to perform design approval functions. This describes only a few of the organizations that make a business of engaging in design and production certification activities.

In addition to the FAA's ability to implement SMS, the safety benefits of a more complete corporate focus on compliance and safety can further permeate the industry if these organizations are required to obtain DO certificates. For this reason, the criteria for obtaining and holding a DO certificate must be such that they can be tailored to the size and functions of the specific DO certificate holder.

A total "culture of compliance" must exist within each DO company, but how that culture is established will likely differ for each DO certificate holder. The key is to define criteria against which all potential DO certificate holders will be measured, but recognize that there will be variables in how the criteria are met based on different types and sizes of companies and associated regulatory obligations.

6.7.1 FAA LIMITATIONS ON THE DO CERTIFICATE

A DO certificate may cover type certification activities, supplemental type certification activities, and PMA issuance activities, as well as production approval activities. For a particular DO, the FAA may limit the scope of activities that might be accomplished by that DO.

Type Certificates. For type certification activities, it would be rare that a certificate holder would be able to perform all the responsibilities necessary to demonstrate compliance for all products that are eligible to receive a TC. For this reason, the FAA may restrict a DO certificate to only products covered by a specific part of the airworthiness/design requirements, such as 14 CFR part 23 (small airplanes), part 25 (large airplanes), part 27 (small rotorcraft), part 29 (large rotorcraft), part 31 (balloons), part 33 (engines), or part 35 (propellers).

The FAA may further limit the scope of DO certificate activities within a given regulatory part. For instance, a manufacturer might only have the experience necessary to properly comply with DO requirements for small transport airplanes under part 25 airplanes, but not large transport airplanes; or for reciprocating engines under part 33, but not large turbofan engines. The FAA may use other parameters it determines to be necessary to further limit the scope of a DO certificate. The intent is to allow the widest scope of certificate for which the applicant has been able to demonstrate its capability to comply with the relevant design and airworthiness requirements.

Supplemental Type Certificates. In the case of STCs, the scope would also likely be defined in more narrow terms. For example, the scope might be limited by the products that a particular air carrier operates, or by technical discipline and subpart (part 23 structures, for instance), or by the complexity of the product (large turbofan engines, for instance), or by other generic parameters the FAA determines to be appropriate.

PMAs. In the case of PMAs, the scope would likely be tailored to each certificate holder.

Determining the Appropriate Scope of a DO. When determining an appropriate scope for a DO certificate, the FAA must ensure the certificate holder has, and will continue to maintain, the capability to meet all the requirements of the subpart within the scope of its certificate.

As part of this determination, the FAA may consider providing multiple DO certificates in unusual situations for applicants with substantially decentralized organizations, or who have a wide range of products or capabilities. When evaluating whether a single certificate or multiple certificates is most appropriate, the FAA would consider the organizational structure of the applicant, interactions of remote or co-located design and production facilities, and the use of common processes and procedures.

The scope of any DO certificate will be clearly defined so that all persons, including other CAAs, will understand the scope of authority for FAA-approved data granted under that certificate.

Transfer of a DO Certificate. A DO certificate holder cannot transfer the DO certificate.

6.7.2 Scope of DO Certificate Privileges Must Be "Functionally Complete"

The DO certificate privileges must be functionally complete, which means they must cover all activities that would have to be undertaken to fully complete a design approval project within the scope of the DO's authority. The DO holder's competence and capabilities must also be functionally complete to certify compliance with the applicable airworthiness safety standards within the scope of its authority. Those include—

- All certification activities leading to the issuance of an original or amended design approval, including design, airworthiness, manufacturing, and maintenance and operations activities as they relate to a design approval. This includes engineering inspection, analysis, and tests; flight tests; ICA; and aircraft flight manuals.
- All determinations of compliance, including those that involve a subjective evaluation.
- Continued airworthiness activities, including changes to those approved designs for product improvements or safety enhancements, such as those contained in service bulletins, or repair data.
- Manufacturing and airworthiness activities, such as the pre-production manufacturing of parts, components, and subassemblies; and conformity of test articles and products, and their airworthiness certification for flight test.
- The development and testing of designs and processes for possible inclusion in future approved designs (that is, "eligible data").
- Any other activities leading to the development of data necessary for the FAA to determine compliance with the requirements issued by those countries from which validation is sought and with which the FAA has a bilateral airworthiness agreement covering that compliance activity.

6.7.3 FORM OF A DO CERTIFICATE

The ARC reviewed and discussed examples of EASA DOA certificates and terms of approval in assessing application to DOs. <u>The ARC recommends a certificate structure similar in nature</u> to the EASA DOA certificate and terms of approval for an FAA DO certificate. This would provide consistency between FAA and CAA certificates.

Refer to appendix L to this report for an example of an EASA DO certificate.

6.7.4 TRANSFER OF DESIGN APPROVALS UNDER DO

A TC/STC issued after the effective date of regulation (for example, under the DO framework) is transferable provided the following requirements are met:

1. The transferor of the TC/STC makes a statement, in writing, to the FAA that they are providing all descriptive data and providing, or making available, all substantiating data to the transferee. This statement must describe the conditions under which the substantiating data is made available to the transferee if the data will not be provided (that is, contractual agreement).

A TC/STC issued before the effective date of regulation is transferable provided the descriptive data and substantiating data the transferor has, or has access to, is provided to the transferee. The ARC recognized that existing TC/STC holders may or may not have the descriptive data and/or substantiating data and imposing a regulatory restriction on the transfer would negatively impact the asset value.

The ARC recommends non-DO design approval transfer requirements be provided by a separate follow-on activity to the ARC.

Note: Because of the unique combined design and production nature of a PMA, PMA is not eligible for transfer.

6.7.5 THE GENERATION OF "ELIGIBLE DATA"

Under a DO certificate, compliance is an intended by-product of an FAA-approved DO system properly functioning under its procedures manual, which includes a formal internal audit and oversight process. For the compliance determination for a particular part or component to be complete, it is essential that a certification basis be established for the product on which it is to be installed or for the article itself. Another essential element is that a type or article design be fully defined so that the interaction of products, parts, and components may be assessed. This interaction may establish additional certification needs.

It is common for the normal engineering and production system of a company to develop products, parts, components, and processes for future use in certification programs. In the case of a DO certificate holder, if that development is accomplished under the approved DO system, that development could be eligible for inclusion in subsequent designs, except for the establishment of a product final certification basis and complete product or article definition. It would be inappropriate to consider such development activity as meeting the standards for complete compliance determination because those two elements would be missing. It is appropriate, however, to give credit for any compliance activities accomplished under a DO. The ARC refers to this as "eligible data."

"Eligible" data is data developed under the processes of an approved DO system, given a specified, but not necessarily final, certification basis and product type or article design.

To use "eligible" data, the DO holder must assess the data's compliance against the final type certification basis of the product or article and final type or article design, respectively, in which it is to be used. It would not be necessary to repeat the compliance activities, provided those activities were appropriate for the final product or article and its certification basis.

The creation of "eligible" data is a concept that is intended for use internal to the DO. No approval or compliance determination can be conferred on the data if the data is provided for use outside the DO.

6.7.6 VOLUNTARY DISCLOSURE PRIVILEGES

The FAA has several active voluntary disclosure programs for air carriers, PAHs, and organizations that have an ODA, among others. These programs are designed to encourage the reporting of product and process deficiencies so they can be corrected before unsafe conditions occur. The programs also apply to discovered deviations from FAA-approved procedures manuals and inadvertent regulatory violations. If the deficiencies or noncompliance activities reported were not intentional or criminal in nature, the FAA will refrain from using the disclosures as the basis for any civil penalty, as long as the certificate holder takes swift action to correct the deficiencies discovered.

As stated on the FAA's Voluntary Safety Programs Branch Web site-

"... the FAA believes that aviation safety is well served by providing incentives for certificate holders to correct their own instances of noncompliance and to invest more resources in efforts to preclude their recurrence. The FAA's policy of forgoing civil penalty actions when a certificate holder meets the requirements of this program, is designed to encourage compliance with the FAA's regulations, foster safe operating practices, and promote the development of internal evaluation programs."⁷

Although the DO is a new type of certificate, the ARC concludes that the information presented above remains equally applicable for a DO, and the FAA voluntary disclosure policy should be extended to DO certificate holders. Activities under a production approval are already covered by FAA voluntary disclosure policy.

The FAA voluntary disclosure reporting program is presented in AC 00–58B, Voluntary Disclosure Reporting Program. Under section 1, Purpose, there is an important exception that must be recognized. The AC states, in part—

"The procedures and practices outlined in this AC <u>cannot</u> be applied to those persons who are required to report failures, malfunctions, and defects under 14 CFR part 21, § 21.3, and who do not make those reports in the timeframe required by the regulation."

⁷ http://www.faa.gov/about/office_org/headquarters_offices/avs/offices/afs/afs200/branches/afs280/descriptions/

This exception to the program is in recognition of a determination made by the FAA that, because there is a regulatory requirement to report under § 21.3, the voluntary disclosure of a failure to report cannot relieve the certificate holder from any enforcement that might be based on that failure to report. This exception still appears to be appropriate for a DO certificate holder under the same defined.

6.7.7 MANUFACTURING AND PRODUCTION FUNCTIONS UNDER DO

There are two types of manufacturing and production functions that should be addressed regarding a DO and a production organization:

- Those pre-production manufacturing functions associated with obtaining a design approval, and
- Those associated with a production approval (that is, post-design approval production).

The intention is to allow a DO to use its existing FAA-approved production system during pre-production manufacturing functions associated with obtaining a design approval. If the organization, within the scope of its design authority, chooses to use its approved production quality system, it must use the DO procedures manual processes for any of the following:

- Conformity inspection,
- Determining conformity of parts and test articles,
- Determining conformity of test setup, and
- Determining conformity of installations.

For DOs performing pre-production manufacturing as part of their scope of activities, in addition to the above, the DO procedures manual must also contain procedures for—

- Controlling documents and data associated with pre-production manufacturing;
- Ensuring each supplier furnished product, part, or appliance conforms to its design;
- Controlling manufacturing processes to ensure conformity to its design;
- Conducting inspections and tests;
- Ensuring calibration and control of all inspection, measuring, and test equipment;
- Documenting the inspection and test status of products, parts, and appliances supplied or manufactured to the design;
- Ensuring discarded articles are rendered unusable;
- Implementing corrective and preventive actions to eliminate the causes of an actual or potential nonconformity to the design or noncompliance with the approved DO procedures manual;
- Preventing damage and deterioration of each product, part, and appliance during handling, storage, preservation, packaging, and delivery;
- Identifying, storing, protecting, retrieving, and retaining quality records; and
- Planning, conducting, and documenting internal audits to ensure compliance with the approved DO procedures manual.

For post-design approval production, the production approval requirements remain the same.

The DO processes would also support the FAA's issuance of special airworthiness certificates in the experimental category for the purpose of R&D or show compliance.

Although the ARC intended to further discuss the manufacturing and production functions to include the combined design and production organizations (that is, one certificate comprising design and production), it was limited to the pre-production concepts noted above because of schedule constraints. Further discussion would be required to recommend additional privileges that may be available for a combined Design Production Organization (DPO).

6.7.8 FLIGHT STANDARDS FUNCTIONS

Section 21.17(a)(1) requires an applicant for a TC to show that its product meets "the applicable requirements of this subchapter that are in effect on the date of application for that certificate." Part 21 resides in 14 CFR chapter 1, subchapter C, Aircraft. This subchapter covers parts 21 through 59, which includes the type certification airworthiness standards found in parts 23 through 35. The operating rules applicable to these same type certificated aircraft are found in subchapters F and G, which include parts 91 through 135.

Although a TC may legally be awarded without the product complying with appropriate operating requirements, the practice has been to provide an initial operational evaluation of aircraft during the type certification program. That operational evaluation is carried out by the Flight Standards AEG that has the responsibility for the particular product being type certificated. The AEG performs or coordinates the following activities associated with the type certification of products, which are discussed in FAA Order 8900.1, Flight Standards Information Management System (FSIMS):

- ICA—Review and find acceptable the maintenance aspects of the ICA which are required under § 21.50, Instructions for continued airworthiness and manufacturer's maintenance manuals having airworthiness limitations sections, and §XX.1529, Instructions for Continued Airworthiness, in the respective aircraft certification standards.
- Flight Operations Evaluation Board (FOEB)—Develop and revise the master minimum equipment list (MMEL).
- Flight Standardization Board (FSB)—Determine the requirements for pilot type ratings, develop minimum training recommendations, and ensure initial flight crewmember competency.
- Maintenance Review Board (MRB)—Establish the minimum maintenance and inspection requirements for transport category aircraft, engines, propellers, and auxiliary power units. Participate in industry steering committee meetings to review the Maintenance Steering Group (MSG)–3 analyses.
- Participate in Type Certification Board and Flight Manual Review Board activities.

During type certification, all determinations of compliance to the airworthiness requirements in parts 23 through 35 are made by the DO certificate holder, with appropriate FAA oversight. Because compliance with the ICA requirement in §§ XX.1529, 31.82, 33.4, and 35.4 are to be determined by the DO certificate holder, the ARC recommends the maintenance aspects of those requirements also be determined by the DO. The DO procedures manual would need to contain appropriate procedures that ensure the maintenance aspects of the ICA are properly addressed, and consistent with § 21.50 and FAA Flight Standards' regulatory guidance.

The ARC believes the formulation and execution of the FOEB, FSB, and MRB should continue as Flight Standards AEG functions, with support from the DO certificate holder. All determinations of compliance to airworthiness standards associated with those boards would be made by the DO certificate holder consistent with its procedures manual. Some additional responsibilities associated with the operation of those boards might be assigned to a DO certificate holder, under Flight Standards policy, after experience is gained. This would necessitate a revision to the DO procedures manual.

AEG participation in Type Certification Board and Flight Manual Review Board activities would continue to the degree that AIR participates in those functions. For new TCs and amended TCs requiring a model change there would be a review by the Type Certification Board, but it is expected that most major changes would be conducted under DO procedures and would not require board review. This is because the type boards are identified in an FAA Order and the DO certificate holder is free to propose its own procedures instead of those identified in existing FAA Orders.

For a DO, a Flight Manual Review Board would not have the sole responsibility for determining compliance with the flight manual, which would reside with the DO certificate holder. Any operational regulations and associated flight standards guidance regarding flight manuals would be complied with through processes and procedures defined in the DO procedures manual.

6.7.9 Noise, Fuel Venting, and Exhaust Emissions

Although Congress has granted the FAA full statutory authority over the airworthiness certification of civil aviation products in the United States, the EPA guides FAA requirements regarding noise, fuel venting, and exhaust emissions (14 CFR part 34, Fuel Venting and Exhaust Emission Requirements for Turbine Engine Powered Airplanes, and part 36, Noise Standards: Aircraft Type and Airworthiness Certification).

Under the current system, FAA Order 1050.1E, Environmental Impacts: Policies and Procedures, sets policies and procedures and assigns responsibilities for ensuring the FAA complies with environmental procedures as required by the National Environmental Policy Act under the direction of the Council on Environmental Quality. The order contains examples of actions that normally require an environmental assessment, including noise and emission requirements.

In addition, the Noise Control Act of 1972 requires the FAA to make findings, notwithstanding any delegation to companies, other private persons, CAAs, or any procedures for type certificating foreign-manufactured aircraft. The FAA's Office of Environment and Energy (AEE) delegates the authority to make these types of findings to the appropriate

FAA certification directorate, depending on the type of aircraft involved. That directorate may not re-delegate the authority and the FAA must base its finding on actual examination of each type design. Individual delegations have been granted by the FAA but they are only for recommending approval, not finding compliance.

Although the ARC recognizes the distinction between the airworthiness requirements of 14 CFR and the noise, fuel venting, and emissions requirements, it believes a DO could be found to have the necessary capabilities and expertise to make compliance determinations regarding the environmental requirements contained in parts 34 and 36. Specific noise, fuel venting, and emissions processes would be developed within the DO compliance, safety, and quality systems to ensure proper compliance determinations. This is in keeping with the principle of a DO making 100 percent of the compliance determinations.

The ARC recommends the FAA propose to the EPA that the process-based approach to compliance, as established by DO program principles, is far more robust than the normal delegation process and is sufficient to ensure compliance with the environmental aspects of parts 34 and 36. The ARC believes this to be consistent with the recommendations set forth by the FAA in response to section 312 of the FAA Modernization and Reform Act of 2012, which recommend "expanding delegation capability to include support for all certification airworthiness standards when appropriate, particularly low-risk or routine activities such [as] those related to noise and emissions tests and ICA." Refer to NATCA's dissenting opinion to section 10 of this report.

6.7.10 ESTABLISHING A CERTIFICATION BASIS

An applicant for a design approval may propose a certification basis to the FAA, and the FAA can establish the certification basis. Regarding design changes under a DO, the FAA could establish a certification basis within the limited boundaries set forth in the DO approved procedures manual. This is considered an FAA pre-decisional approval process provided to the DO. Projects that fall within the approved boundaries would be performed under the domain of the DO without additional FAA input. Industry has identified such pre-decisional approval as critical to the success of DOs.

6.7.11 CHANGED PRODUCT RULE

The derivative type certification requirements, which apply to TCs and STCs, specify the need for the FAA to make a determination of the appropriateness of the original type certification basis. This is sometimes referred to as the "changed product rule requirements." As a general rule, those projects would be subject to FAA LOPI, and it is expected that a DO would notify the FAA when it undertakes such projects.

As the FAA gains more confidence in specific DO certificate holders, it may be willing to rely on specific DO approved processes to assist the FAA in making its determinations under the changed product rule.

6.8 DO APPROVAL OF DATA

The DO privileges associated with approval of data discussed below are made in consideration of a 14 CFR part 21 accountability framework policy initiative, which the FAA has indicated it is currently pursuing for FAA-managed certification projects. The FAA has advised that it will formalize the policy in FAA Order 8110.4D, Type Certification, and future changes to FAA Order 8110.37, Designated Engineering Representative (DER) Handbook, and FAA Order 8100.15. The new policy will clarify that discrete substantiating or descriptive data generated in support of part 21 certification projects do not need to be discretely "approved." Although the ARC accepts the FAA's intent to return to the basic foundations of part 21, the ARC believes this specific change alters an accepted practice that has been established over several decades within FAA policy. However, if the FAA continues to pursue this new policy, the ARC strongly believes that whether a project is managed by the FAA with designees or by a DO, the status of any data, regardless of the part 21 process that produced it, should be the same. Should the FAA not pursue this policy change, the FAA must ensure the DO regulation includes the privilege of making discrete compliance determinations that result in data that is (or is equivalent to) FAA-approved, and that the data is internationally recognized within the scope of bilateral agreements. Refer to section 12 of this report for a dissenting opinion on approval of data.

Data Supporting a Design Approval Under Part 21

Current practice has facilitated FAA approval of discrete substantiating data as well as discrete descriptive data that ultimately makes up the type design. However, part 21 contains no requirement for approval of this discrete data separate from issuing the TC (refer to § 21.41, Type certificate, where the TC is defined to include the type design, any operating limitations, the certificate data sheet, the certification basis, and any other conditions or limitations). The substantiating data is the documentation related to the applicant's showing, while the descriptive data defines the type design that should be determined "compliant" to the regulations and ultimately approved when the FAA issues a design approval. The ARC agrees with the FAA that making this clear would eliminate the perceived value and pedigree many currently attach to data and the assumption that the reason data has value is because the FAA has labeled it "FAA-approved."

Getting back to the basics of part 21 is relevant to all discrete data showings and will become even more relevant as the FAA begins to implement risk-based decisionmaking and when it chooses not to be involved in certain aspects of a certification project. In such cases, requiring the FAA to provide a status for discrete data "showings" would not be logical when it has chosen not to be involved in making a discrete finding of compliance. Doing so wrongly implies that all data should be statused by the FAA, which implies some level of FAA involvement, even when the FAA has already determined not to be involved in a given aspect of a certification program.

The ARC acknowledges that regardless of how any data was used in the past, it remains the applicant's responsibility to show compliance even when such data is used to support subsequent part 21 projects. The ARC understands the FAA intends to clarify this in future type certification policy to further reinforce the accountability framework concepts related to design certification. This should include revisions to FAA Forms 8110–3 and 8110–9 to distinguish between when the form is being used for a part 21 project and when it is supporting

a maintenance or other operational requirement under part 43, Maintenance, Preventative Maintenance, Rebuilding and Alternation, associated with an existing design approval.

Although not the case for part 43 maintenance actions, and unlike current practice, the data produced for the purposes of showing compliance under part 21 needs no label or pedigree designating it as "approved" by the FAA. Instead, the descriptive and substantiating data will only be designated as "found to have shown compliance to the airworthiness standards by the FAA" or "determined compliant to the airworthiness standards by a DO." In either case, only when the FAA issues its design approval will it make the single finding required by part 21, and in so doing approve the descriptive data that defines the complete type design.

Data Supporting Major Design Changes Under Part 21

Part 21 currently contains provisions for FAA approval of both major and minor design changes. In promulgating the DO regulation, once a type design is held by a DO, the DO must have the privilege to make all necessary determinations of compliance associated with any change and also approve the descriptive data by incorporating the change into its type design. Although existing statutory law (49 U.S.C. § 44702) allows only the FAA to issue a certificate, there is no restriction preventing the FAA from allowing a DO to make changes to an existing TC as a privilege of its DO certificate (that is, DO is not a delegation). In exercising this privilege, the DO will approve the descriptive data associated with any design change once it incorporates the change in accordance with its approved procedures. This privilege should be given regarding any type of design approval held by a DO, including PMAs (which are not addressed in 49 U.S.C.). This privilege should also be given to a DO that is contracted to act as an agent of the applicant to manage its type design. In a supplier role, a DO may provide determinations of compliance that support a design change; however, as a supplier a DO cannot approve design changes because it is not the custodian of the type design.

In promulgating this privilege, the existing part 21 regulations regarding major changes should be changed. For major changes, the current regulations require that 1) an application be made, 2) the regulations under § 21.101 be considered, and 3) the "person" obtain either an STC or an amendment to the TC. There is no alternative process that allows this to happen without the FAA. Today, only the FAA or its designee can address the certification basis or issue an amended TC or STC. Therefore, the approval of the major change is currently an FAA activity. In the case of § 21.101, the function is inherently governmental and will require the FAA to be involved. However, the ARC foresees that the FAA could facilitate its involvement through a "pre-decisional" process that is part of the DO procedures manual. The DO could follow such a process to ensure the FAA is properly engaged on certification basis decisions that are outside what is allowed by the pre-decisional process. In this way the DO is not held up and forced to wait on the FAA for the majority of its project activity.

Data Supporting Part 43 Maintenance or Any Operational Requirement

Although DO determinations of compliance associated with seeking a design approval or supporting a minor or major change do not need to result in discretely approved data, a DO must have the authority to make determinations of compliance that result in the creation of approved data when required to support part 43 maintenance or any operational requirement associated with an existing type design. In such cases, the data may or may not be part of the

approved type design. For example, when a DAH issues a service bulletin or service letter, the data is conveyed as a change to the type design. However, a holder (or even the FAA) can provide data to support a unique repair or alteration for an individual owner/operator, an action which does not result in a change to the type design.

The ARC contends that the "approved data" required by part 65, Certification: Airmen Other Than Flight Crewmembers, for use in part 43 is the approved descriptive data and any other technical data required to perform the maintenance. (This includes any drawings, material specifications, process specifications, procedures, and other data describing an approved repair or alteration). The ARC contends that part 43 does not require direct FAA approval of any substantiating data used to show compliance. Thus, the DO's authority to approve data in support of part 43 maintenance applies only to the descriptive data (that is, not the substantiating data) associated with the design approvals it holds. This means that, at its discretion and without any action by the FAA, the DO may create "approved data" to support repairs and alterations by third-party owner/operators for design approvals held by the DO (or regarding any other design approval for which the DO is authorized to provide data for maintenance under its scope of authority).

The ARC sees three possible regulatory options for the FAA to consider in addressing DO "approved data" to support major changes and to support maintenance:

Option 1: The DO has authority to create "FAA-approved data." This option assumes the DO will be authorized to create FAA-approved data in a manner that does not include delegation. In the FAA's current system, all type design data approved by the FAA for use in the global aviation system has been referred to as "FAA-approved." This option continues with that approach.

For decades § 21.95, Approval of minor changes in type design, has allowed minor changes to a type design to be "approved under a method acceptable to the Administrator before submitting to the Administrator any substantiating or descriptive data." Additionally, in 14 CFR part 1, the term "approved" is defined as "approved by the Administrator, unless used with reference to another person." Because § 21.95 makes no reference to another person, the regulation allows for the creation of FAA-approved data without the data being submitted to the FAA or reviewed by the FAA. The FAA-approved data is created when the TC holder executes the "method acceptable to the Administrator."

Although § 21.95 applies only to minor changes to a TC, the ARC believes this existing approach can be applied to major type design changes determined to be compliant by a DO (that is, DO creation of FAA-approved data before any substantiating or descriptive data is submitted to the FAA). The DO regulatory requirements, along with the processes and procedures contained in an FAA-approved DO procedures manual, must be sufficiently thorough for the FAA to approve the data resulting from it before being submitted to the FAA. As with minor type design changes, the FAA may review any "compliance determinations" and supporting data after the DO determines it is compliant.

Under this concept, a DO is not approving data on behalf of the FAA, because a DO is not a delegation. Once the FAA-approved DO process for making a compliance determination has been properly executed, the descriptive data are FAA-approved.

Option 2: The DO has authority to create "DO-approved data" equivalent to "FAA-approved data." This option would facilitate recognition of DO-approved data. It would require a change to 14 CFR part 1, where the term "approved" is defined as "approved by the Administrator, unless used with reference to another person." The regulation should be changed to include other entities entrusted by the Administrator to approve. Under this option, the ARC would recommend the definition be revised to "approved by the Administrator or under the authority of a certificate granted by the administrator, unless used with reference to another person." In granting this privilege to each DO, it is essential that the FAA affirm to its international airworthiness partners that such DO data is equivalent to being "FAA-approved."

Option 3: Create a different term for "DO-approved data." This would require a change to parts 65, 121, 135, and 145, instead of part 43 to make it clear that it can be used.

Although it may seem easy to simply permit certificate holders to issue "approved" data, this function would actually be far more difficult to reconcile with current regulatory practice than it appears at first glance. The FAA Chief Counsel's Office has already met with the ARC to explain that the word "approved" is currently defined to encompass inherently governmental tasks, and the office has expressed reservations at permitting a certificate holder to issue approved data. The word "approved" is also a difficult word to redefine in this situation because it is used in a variety of different contexts in the FAA regulations to reflect things that are approved by the Administrator, and this use imposes certain constraints on the ability to make changes in the use of the term.

Options 1 and 2 are trying to affect a very specific use of the term "approved" (a subset of the ways that it is used): the use of the term in the context of data on which maintenance providers may rely in the case of major repairs and major alterations. Rather than trying to craft language that does not adversely affect the other uses of the term "approved," it might be preferable to adopt a different adjective to describe the data that is appropriate for use to support major repairs and major alterations. To effect this change, one could define a new term ("purple data" and "eligible data" have been used as placeholders, but any adjective not already in use in the regulations could be acceptable). The data included under the definition of this adjective would include both FAA-approved data and DO-approved data (it should be defined in § 1.1 of the regulations to have global impact on all FAA parts of the regulations). The requirements for approved data currently found in parts 65, 121, 135, and 145 could then be updated.

Some positive aspects of this proposed change would include (1) accomplishing the goal of permitting DOs to issue data on which the maintenance community could rely in a manner that is nearly identical to current practice, except with limited FAA involvement (limited to FAA-chosen LOPI), (2) being consistent with existing statutory authority, (3) limiting the possibility of unintended consequences (impact on other uses in the regulations of the term "approved"), and (4) avoiding potential delegation to the public of inherently governmental functions. The negative aspect of this proposal would be the potential need to update international executive agreements (like the maintenance-acceptance provisions of the bilateral

agreement with Canada) to reflect the new terminology; however, this impact would likely also apply to expanded use of the term "approved," because the inherent definition of a term on which the bilateral agreements rely would be changing, which may cause trading partners to want to revisit the affected bilateral agreements.

Although any of the above three options would be acceptable to industry, the FAA indicated that option 3 is likely the most viable option for supporting part 43 maintenance without the use of delegation (an action achieved in the past through recognition of SFAR 36 organizations). The FAA expressed concern regarding whether it was statutorily possible to grant DOs the privilege of approving major changes to the design approvals under part 21, as industry prefers. Current FAA thinking is that some form of delegation would be required. FAA members of the ARC's DO Working Group acknowledge that, to work within the DO construct, any delegation would essentially be performing an administrative function in making its statutorily required finding. In this role, the delegation would perform the following administrative actions before certificate issuance or approving a type design change:

- Verify the FAA's planned project LOPI is complete without open issues.
- Verify the statement of compliance was issued by a DO representative authorized to make the statement.
- Ensure there is no knowledge of any noncompliance or unsafe condition conveyed by FAA or DO personnel involved in the project. (Note: This is intended to be a yes-or-no answer without conducting research.)

In summary, three key privileges have been identified that a DO must be granted regarding how it documents its compliance decisions and dispositions data. In all cases, the data would require no further "showing" on the applicability and acceptance of its intended use.

- 1. New Part 21 Design Approvals. Make determinations of compliance related to the design approvals the DO is seeking to hold under part 21, and similarly make them as either a supplier to another DO or as an agent contracted to manage a certification project for an applicant seeking a design approval, when authorized through a formal interface agreement.
- 2. Part 21 Type Design Changes. Make determinations of compliance related to both major and minor changes to the design approvals the DO holds under part 21, or for those it has been contracted as an agent for another holder evidenced by a formal interface agreement, and to also approve those type design changes. (Note: In a supplier role, a DO does not approve design changes because it is not the custodian of the type design.)
- 3. Maintenance and Operations. Create and distribute "approved data" to support part 43 maintenance or any operational requirement associated with a design approval the DO holds, and similarly regarding third-party design approvals (that is, not held by the DO), but where the data approval is executed within the DO's FAA-authorized scope of authority.

6.8.1 SERVICE BULLETINS AND A STANDARDIZED FORM FOR DO TRANSMITTAL OF APPROVED DATA

When issued, service bulletins constitute a change in type design by the holder and convey the necessary "approved data" to implement the change by owner/operators under part 43. <u>The ARC</u> recommends DO-issued service bulletins or other types of service data should be a means for DOs to provide "approved data" for general use, and a new or revised form is needed for domestic and international recognition of "approved data" created under the DO concept. The new or revised form should contain the same basic information as FAA Forms 8110–3 and 8100–9 regarding compliance data and the approval's purpose.

The form should be titled to recognize the source of the data approval and should be traceable to the originator. In addition, there should be no provisions for recommending approval of data. Thus, "approved" is the only statement that can be made about the data. The form should also address the date the determination of compliance was made and the date the form was signed.

The ARC believes allowing FAA designees and DOs to use the same form is the preferred option. This would help reinforce the equivalency of the data when executed by a DO. The FAA should also consider whether electronic formats would be acceptable for transmitting this type of information to owner/operators.

6.8.2 DO USE OF PREVIOUSLY APPROVED DATA

The concept of previously approved "substantiating" or "descriptive" data implies the data has an established pedigree with an FAA approval that makes the data more valuable than data without such a pedigree. However, no inference can be made regarding the applicability of such data to another design approval project unless a determination of compliance has been made by the DO. This does not mean the DO should regenerated, recalculated, or retested the data, but no relief regarding the part 21 showing is implied regarding the new project based on how it was used in the past. The DO will still have to assess the data to determine its applicability to the new project and to make a determination of compliance to the certification basis of the new certification project.

In addition, when a DO incorporates a type certificated or TSO component into its type design, it is only required to show the product's type design, including the installation of that component, is compliant. There is no requirement for the installer to "look behind" the design or compliance status (for example, TSOA status) of the component itself. A similar approach can be taken with STCs and PMA parts when they are installed by owner/operators. However, if a DO desires to make an STC or PMA part directly a part of the type design on which it is being installed, the DO should obtain, or have access to, both the substantiating and descriptive data associated with the STC to make its determination of compliance. Drawing this distinction in no way prevents a TC holder from installing an STC in its production line, but in such cases the STC is not a part of the DO's type design, but rather a change to that design, installed at the time it is manufactured, but held by another entity.

6.8.3 Use of the FAA's Delegation System

One of the basic principles developed for DOs is that the FAA makes no discrete findings of compliance. DOs are issued a DO certificate because they have a demonstrated engineering capability and commitment to compliance. This enables the FAA, using its discretionary authority, not to direct its resources to making numerous discrete compliance findings. Instead, it can rely on the DO's statement of compliance in making its overall compliance finding when issuing a TC or other design approval.

Because the FAA is making no discrete compliance findings, there is no basis for allowing the use of engineering designees, either within the DO itself or at its partners/suppliers. Designees are authorized only to perform tasks the FAA itself would otherwise perform. Because the FAA is not making any discrete findings of compliance under the DO concept, there is nothing to delegate. Thus, the advantage to industry of being able to make all determinations of compliance is that the DO is not dependent on the existing delegation system.

This does not mean DOs cannot use individuals and companies that also hold FAA delegations, but those designees would be acting solely as a design supplier resource to the DO and any compliance determinations made by such suppliers must be conducted under a system determined acceptable by the DO. DOs are not acting as representatives of the Administrator.

6.8.4 DO RECOGNITION AND USE OF DESIGN SUPPLIERS WITH FAA CREDENTIALS (INCLUDING OTHER DOS)

A DO may take the status of an FAA designee or another DO into consideration when determining the appropriate method and level of supplier oversight it should perform. That oversight must be defined within the DO supplier procedures and must include both the qualification of that supplier and periodic oversight. In conducting its oversight of the supplier, the "project DO" may include as one of its considerations the fact that the supplier is a designee of the FAA, but it must recognize that the FAA will not be conducting oversight on any non-delegation activity. If the supplier is another DO, the FAA will perform oversight of the "supplier DO," but this does not relieve the project DO of performing oversight.

Regardless of a particular supplier's FAA credentials, the project DO must, under its CAS and SMS processes, assess and find acceptable the compliance and safety risk associated with its degree of reliance on this type of supplier. The project DO must also be satisfied that these organizations or individuals are performing as expected, and must be aware of any FAA corrective action related to their performance. The project DO could achieve this awareness by contractually requiring the designee or supplier DO to provide records of any FAA corrective action, such as designee counseling letters or audit records.

A formal supplier interface agreement must exist between a project DO and every supplier DO providing it with determinations of compliance. The interface document should address the scope of what the supplier DO may accomplish for the project DO. The document may authorize the supplier DO to follow its existing DO procedures when making determinations of compliance associated with the project DO's certification plan. The presence of a supplier interface document does not relieve the project DO from showing compliance responsibility as an applicant.

7.0 OTHER PROCESSES FOR SMALL BUSINESSES

The goal of recommending change to part 21 is to streamline certification so the FAA and industry are able to maintain or improve the current level of safety while keeping up with industry growth and ensuring small business are not adversely affected. To encourage small business and innovation, the ARC identified three options to obtain similar responsibilities and certification process efficiencies for any applicant that chooses not to pursue a DO with privileges or, in the long term, falls below the DO applicability threshold:

- Accredited organization (AO),
- Agent design organization (ADO), and
- Modified current model.

The ARC proposed these alternatives in response to the concern that moving most organizations to a DO could introduce costs that would be detrimental to many small businesses. These options would pose less of a burden to small businesses, and each is designed to reduce the FAA's daily involvement in low-risk activities while still maintaining or improving aviation safety. Because of the ARC's inability to focus primarily on those companies falling below the DO applicability threshold, these options have been explored as a preliminary effort. Further research and definition are required for each option following the conclusion of this ARC.

7.1 ACCREDITED ORGANIZATION

An AO would be authorized to determine compliance to specific regulations for the articles or the work the organization performs. The accreditation would be performed by an FAA-approved organization that provided an accreditation process acceptable to the FAA.

The AO approach would seek to achieve many of the same objectives as the DO/SMS approach, but with a goal of making such achievements cost effective for smaller business while maintaining or improving safety. This would be an optional program, with a goal of voluntary adoption by industry.

To parallel or harmonize with the international mandate for SMS principles, the AO would implement COS principles or SMS principles. COS principles incorporate many but not all principles of SMS, and many companies below the DO applicability threshold are already familiar with, or have already implemented, COS.

Those companies seeking to export products to SMS-requiring ICAO countries may elect to implement appropriate additional SMS requirements; however, adoption of SMS requirements would not be mandatory for those implementing an AO.

Adoption of certain established COS/SMS principles would be required for accreditation to a voluntary industry standard. The AO approach would be a voluntary industry accreditation program similar to AC 00–56, Voluntary Industry Distributor Accreditation Program, for distributors. Under such an approach, one or more industry representatives (for example, SAE) would develop industry accreditation standards articulating the necessary requirements for an AO system, including implementation of COS/SMS principles. Those "below-the-threshold"

companies successfully accredited to such standards would be identified in an FAA tool indicating compliance. The objective of such an accreditation program is to encourage voluntary participation by industry, and for customers to make inclusion in such a program a prerequisite to doing business. As accreditation to the standard becomes an accepted norm, more companies will seek to implement COS/SMS principles under the AO model.

The AO approach may also provide companies that meet the required accreditation standard with the opportunity to use an approved "compliance library." The compliance library could be one the AO develops on its own and has FAA approval, or one developed as part of a consensus standard approved by the FAA. If a consensus standard were to be used, the AO would use only those standards included on a list approved for the AO by the FAA. The compliance library would enable a "below-the-threshold" company to take advantage of the reliability indicated by accreditation to self-start projects fitting in its compliance library. This would permit the AO to take advantage of the benefits of avoiding the FAA sequencing/project prioritization queue to more quickly initiate projects, with the end result of bringing end-products to market more quickly. Projects not within an AO's compliance library would still be subject to FAA sequencing/project prioritization. The compliance library could be expanded to demonstrate more competencies that would permit the AO to avoid the FAA queue for additional projects.

The AO approach will also provide for a reduced LOPI. The extent to which the LOPI is reduced or increased will depend on the proposed project's complexity. Complex projects will involve more a significant LOPI from the FAA in terms of systems oversight and findings, though not rising to the level of the one-for-one show/find process of the current model. Projects becoming less complex will allow for correspondingly reduced LOPI (and in some cases almost zero LOPI), reflecting their level of complexity and effect on safety, and preserving FAA resources.

7.2 AGENT DESIGN ORGANIZATION

An ADO is a person or organization that could perform certification and COS activities on behalf of an applicant through a contractual arrangement. This would be similar to the current consultant ODA.

ADO is a concept that needs further discussion and understanding by both the FAA and industry. Persons or organizations qualifying as ADOs would have responsibilities similar to those of the AO approach.

ADOs would provide cost-effective options for small companies, STC applicants, and small PMA applicants that do not have resources to implement an AO approach, or that have limited needs for FAA resources because of limited applications.

An ADO may provide services to comply with the § 21.3 requirements and COS monitoring on behalf of holder of a TC (including amended or supplemental TCs), a PMA, or a TSOA, or the licensee of a TC. This enables small companies to comply with the § 21.3 and COS requirements without the maintaining the required resources on their own. This concept needs further exploration and the ADO approach would not be applicable until DOs become available.

7.3 MODIFIED CURRENT MODEL—ENHANCED SHOWING

This process would be open to all applicants and would probably be most used by small applicants with limited resources. This process would require projects to enter the FAA's resource prioritization process and depend on the availability of FAA engineers or DERs to assist the applicant, but could be subject to potentially significant delays because of FAA engineer or DER availability.

The modified current model approach reflects the use of designees by small businesses, including PMA companies and STC applicants. Although the process has not been changed, the model is "modified" in the sense that the FAA anticipates a reduction in the number of DERs available to provide services to these small applicants. With a transition of organizations toward ODA and future DO concept, DERs will be reduced by approximately two-thirds of current numbers by attrition, non-renewal of privileges, and limitations of new DER privileges. RBRT and/or project prioritization will still apply, causing certain applicants to be substantially delayed in the FAA queue based on the perceived value and safety considerations of their application, while giving preference to those applicants whose projects are deemed to have a greater impact on safety.

Criteria for applicant-only showing will be developed by means of a standard (for example, ISO or SAE) for low-risk projects. Specific criteria for applicant-only showing would include the article being considered low risk, meeting criteria of the compliance library, and the ability to issue a § 21.20 statement. If the standard is adopted, applicants will gain privileges. A compliance library would be developed and accessible in a repeatable manner, allowing the applicant to initiate low-risk projects independent of the FAA. However, for the applicant to take advantage of the privileges, all criteria must be met. If an applicant deviates from the defined criteria, there will be stipulations potentially involving the FAA or a DER. Additionally, the company would have a system in place to meet the specific criteria and have a process with proper oversight and/or checks in the system. The system would have reporting requirements back to the authority regarding self-disclosures and a corrective action program regarding noncompliances. This system would then provide for additional privileges under applicant-only showing program.

The modified current model will encourage those companies that can afford to implement a DO or an AO to do so, to avoid being subject to sequencing with the FAA. Smaller businesses unable to establish an AO or DO will still have access to the FAA but may be subject to delays in FAA compliance findings in areas of FAA involvement such as flight testing, software, human factors, noise, test witnessing, and inherently governmental functions, like exemptions. The FAA must balance the certification needs of applicants producing future products for the National Airspace System with the need to maintain an ever-growing COS responsibility within that airspace. Most FAA resources are currently focused on lower risk projects because of the sheer number of such projects initiated each year. Transitioning small business and lower risk activities to this type of system has the potential to affect the product time to market for the individual companies, allowing the FAA to focus resources in areas of greater risk. The FAA has been successful in the use of abbreviations of this model such as memorandums of understandings between the FAA and applicant.

8.0 MINIMUM DESIGN AND PRODUCTION APPROVAL APPLICANT/HOLDER REQUIREMENTS

8.1 MINIMUM DESIGN APPROVAL APPLICANT/HOLDER REQUIREMENTS IN SUPPORT OF SMS AND OVERSIGHT OBJECTIVES

8.1.1 BACKGROUND

The objective of the ARC is to improve the overall effectiveness and efficiency of part 21 certification procedures by updating regulations and policies to reflect a systems safety approach to product certification and FAA oversight of DOs. The ARC considered minimum qualification and organizational requirements for all design approval applicants and holders as a systems approach requires the ability to recognize and establish FAA oversight for the responsibilities and privileges of design approval applicant/holder organizations.

The concept of establishing new minimum design approval applicant/holder requirements is consistent with the ARC's recommendation to establish new SMS requirements for organizations that design or manufacture type-certificated products (under a TC or PC) and those that design or manufacture articles (under a TSO or PMA) or make changes to products (under an STC) that could directly prevent continued safe flight and landing if they fail. SMS is a structured approach to safety that relies on qualified personnel and documented processes and actions requiring some type of minimum organization. Current regulations for design approval eligibility (§ 21.17) allow any person to make application for design approval and do not provide for any minimum qualification.

In addition, minimum qualifications for applicants is necessary to ensure they fully understand the type certification process and their roles and responsibilities so as to ensure effective and efficient certification programs and ongoing continued airworthiness, which allow the FAA's limited resources to focus on safety and value-added activities.

The challenge the ARC faces is to develop a minimum qualification for an applicant without discouraging innovation or disenfranchising small businesses, while at the same time improving the certification efficiency and quality assurance.

To this end, the ARC reviewed the typical small business activities with TSO application and production as well as PMA application and production. From this review, the ARC determined a two-step process was necessary so initial applicants are not discouraged, or overly burdened by new regulatory requirements, from bringing new products into aviation while at the same time ensuring they understand the safety standards applicable to their intended products.

As a result, the ARC developed a progressive approach beginning with an "applicants" minimum understanding of the regulations and compliance process followed by a producer's requirement for a quality system appropriate to the products being produced. The minimum applicant standards are intended to be demonstrated either by the applicant's personal knowledge and/or experience or by contracting with a consultant with the requisite knowledge and experience.

Once the initial application and demonstration of certification is completed and production is ready to begin, the applicant should upgrade their system to include a quality system similar to the quality system requirements of the TSOA and PMA appropriate to their intended product.

Recommendation 1b—Systems Approach to Certification – Minimum Applicant/Holder Requirements: The ARC recommends establishing minimum requirements for design approval applicant/holder qualification and responsibilities to ensure they fully understand the type certification process and how they intend to carry them out.

8.1.2 RECOMMENDED REGULATORY TEXT FOR MINIMUM REQUIREMENTS

A subteam of ARC members that represent small businesses and TSO, PMA, and STC applicants and holders developed a proposed new part 21 requirement establishing minimum standards for all design approval applicants and holders, referred to as "§ 21.nnn." These proposed regulatory requirements for all design approval applicants are similar to requirements currently in place for TSO and PMA applicants/holders:

<u>§ 21.305 [605] Organization.</u>

Each applicant for or holder of a PMA[TSO authorization] must provide the FAA with a document describing how its [the applicant's] organization will ensure compliance with the provisions of this subpart. At a minimum, the document must describe assigned responsibilities and delegated authority, and the functional relationship of those responsible for quality to management and other organizational components.

<u>§ 21.316 [616] Responsibility of holder.</u>

Each holder of a PMA [TSO authorization] must—

(a) Amend the document required by § 21.305 [605] as necessary to reflect changes in the organization and provide these amendments to the FAA. [...]

§ 21.nnn Minimum standards for design approval applicants and holders.

(a) No person may apply for a design approval unless that person submits to the FAA—

(1) The following organizational information:

(i) The identity of an accountable manager.⁸

(ii) The identity of a primary point of contact during the approval

process.

⁸ For the purposes of this requirement, the accountable manager means the person designated by an applicant or design approval holder who is responsible for and has the authority over all design approval operations that are conducted under part 21, including ensuring that design approval holder personnel follow the regulations and serving as the primary contact with the FAA. (Refer to the glossary in appendix C of this document.)

(*iii*) The reporting relationship between the point of contact and the accountable manager, if they are different.

(*iv*) How the accountable manager fits within the company's reporting and budget structure.

(2) A document describing how the applicant intends to show compliance with all applicable requirements for the issuance of the design approval.

(b) No person may hold a design approval unless that person submits to the <u>FAA</u>

(1) The following organizational information:

(i) The name and address of the design approval holder;

(ii) The identity of an accountable manager;

(iii) The identity of a primary point of contact;

(iv) The reporting relationship between the point of contact and the accountable manager, if they are different.

(2) A document describing how the design approval holder intends to comply with all applicable requirements including —

(i) Process for controlling descriptive and substantiating data and subsequent changes to that descriptive and substantiating data to ensure that only current, correct, and approved data is used in production;

(ii) Process for complying with the reporting requirements of this

<u>Part.</u>

(*iii*) Process to make available the instructions of continued airworthiness and changes thereto pursuant to § 21.50.

(c) A design approval holder must submit to the FAA each change to any process required by this section.⁹

The proposed § 21.nnn is intended to assist those new applicants who do not have prior experience certifying aircraft or other products better understand what is required before they submit an application for a design approval. This can help them potentially reduce their costs and frustration with the complex nature of the certification process. Most established companies will already have processes in place to address all of these requirements.

8.1.3 CORRESPONDING CHANGES TO RELATED REGULATIONS

The FAA provided the ARC with initial proposals for "part 21 cleanup," which includes different approaches and concepts to establish minimum requirements for design approval applicants/holders. Although considered, the ARC decided to propose new § 21.nnn regulatory text to provide a complete understanding of the objectives and intent. However, the ARC

⁹ The ARC notes that during the pre-application consultation phase, it would expect that the potential applicant might not yet meet the applicant requirements, but the potential applicant would be able to identify point of contact and would have some idea of how it plans to demonstrate compliance.

recognizes there are several existing part 21 regulatory requirements affected that will need to be changed to either incorporate these concepts or ensure consistency with new requirements. The following table reflects proposed changes in the original FAA cleanup document and is provided for reference as the ARC's proposed § 21.nnn may also require consideration of corresponding changes to these related regulatory paragraphs.

Existing Regulation	FAA Cleanup Proposed Change	FAA Explanation
§ 21.13 Eligibility. Any interested person may apply for a type certificate.	 § 21.13 Eligibility. a) Prior to application, applicants for type certificate must propose an acceptable system/process for establishing compliance. b) Any interested person may apply for a type certificate. The FAA may decline the application if the submittal is incomplete or the project has no U.S. interest. 	Require applicants for TCs to propose acceptable system/process for establishing compliance prior to application.
 §21.15 Application for a type certificate (a) An application for a type certificate is made on a form and in a manner prescribed by the FAA and is submitted to the appropriate aircraft certification office. (b) An application for an aircraft type certificate must be accompanied by a three- view drawing of that aircraft and available preliminary basic data. (c) An application for an aircraft engine type certificate must be accompanied by a description of the engine design features, the engine operating characteristics, and the proposed engine operating limitations. 	 \$21.15 Application for a type certificate (a)*** (b) An applicant for an aircraft type certificate must submit with the application: 1) a detailed description of the design features 2) a detailed engineering graphic representation of the aircraft with multiple views 3) any available preliminary basic data. 4) proposed certification basis and other data (e.g., preliminary means of compliance). 	Need to require more information initially from the applicant, ensuring early understanding of certification process and schedules before application.
 § 21.21 Issue of type certificate: normal, utility, acrobatic, commuter, and transport category aircraft; manned free balloons; special classes of aircraft; aircraft engines; propellers. An applicant is entitled to a type certificate for an aircraft in the normal, utility, acrobatic, 	§ 21.21 Issue of type certificate: normal, utility, acrobatic, commuter, and transport category aircraft; manned free balloons; special classes of aircraft; aircraft engines; propellers.	To require a standard high level of certitude by applicants (i.e. the applicant cannot just be a leasing company, front company, etc.). Align with EASA (e.g.,

Table 3. Recommended Changes to Part 21 Regulatory Requirements

Existing Regulation	FAA Cleanup Proposed Change	FAA Explanation
commuter, or transport category, or for a manned free balloon, special class of aircraft, or an aircraft engine or propeller, if—	*** ***	procedures manual, milestones, staff and their authority, applicable
(a) The product qualifies under §21.27; or	(a) the applicant has	requirements for a project
(b) The applicant submits the type design, test reports, and computations necessary to show that the product to be certificated meets the applicable airworthiness, aircraft noise, fuel venting, and exhaust emission requirements of this subchapter and any special conditions prescribed by the FAA, and the FAA finds—	established and maintained an engineering system and processes to establish compliance with airworthiness standards and part 21 requirements.	
(1) Upon examination of the type design, and after completing all tests and inspections, that the type design and the product meet the applicable noise, fuel venting, and emissions requirements of this subchapter, and further finds that they meet the applicable airworthiness requirements of this subchapter or that any airworthiness provisions not complied with are compensated for by factors that provide an equivalent level of safety; and	(b) The applicant qualifies under 21.43. (text shown is manufacturing link)	
(2) For an aircraft, that no feature or characteristic makes it unsafe for the category in which certification is requested.		
§ 21.47 Transferability.	§ 21.47 Transferability.	Need to ensure continuing
(a) A holder of a type certificate may transfer it or make it available to other persons by licensing agreements.	***	airworthiness and design holders reporting responsibilities will be continued. Better aligns
(b) For a type certificate transfer in which the State of Design will remain the same, each transferor must, before such a transfer, notify in writing the appropriate aircraft certification office. This notification must include the applicable type certificate number, the name and address of the transferee, and the anticipated date of the transfer.	(e) FAA will agree/approve transfer and reissue of Type Certificate/Supplemental Type Certificate (TC/STC) after review of new potential holder's qualifications and capabilities as a certificate applicant and holder.	with EASA regulations that require transfer of both TC and substantiation data. This would provide enforceability on the selling or licensing party.
(c) For a type certificate transfer in which the State of Design is changing, a type certificate may only be transferred to or from a person subject to the authority of another State of Design if the United States has an agreement with that State of Design for the acceptance of	(f) The transferor of a type or supplemental type certificate must provide to the transferee all type design and substantiation data related to the certificate.	
Each transferor must notify the appropriate aircraft certification office before such a transfer in a form and manner acceptable to the	supplemental type certificate must provide to the licensee all of the type design and	

Existing Regulation	FAA Cleanup Proposed Change	FAA Explanation
FAA. This notification must include the applicable type certificate number; the name, address, and country of residence of the transferee; and the anticipated date of the transfer.	substantiation data necessary for the licensee to fulfill its obligations under this subchapter.	
(d) Before executing or terminating a licensing agreement that makes a type certificate available to another person, the type certificate holder must notify in writing the appropriate aircraft certification office. This notification must include the type certificate number addressed by the licensing agreement, the name and address of the licensee, the extent of authority granted the licensee, and the anticipated date of the agreement.		
§21.XX Design Approval Holder Responsibilities	Add new section – Specifies or references Design Approval holder responsibilities.	To reference responsibilities of a design approval holder in central location. Similar to current 14 CFR 21.146 for Production Approval holders.
		Require holder to submit address changes to maintain currency and for notification purposes.

8.2 CHANGES TO § 21.3 REPORTING TO REFLECT SMS

8.2.1 BACKGROUND

Section 21.3 was originally proposed in March 1969 and codified later in 1970. The list of reportable items in § 21.3(c), as promulgated in 1970, was very similar to Civil Aviation Regulation (CAR) 40.508, Mechanical Reliability Reports, which required air carriers to submit mechanical reliability reports. CAR 40.508 was last changed in March 1962. Section 21.3(c) has remained unchanged for close to 45 years. Airplane design, the aviation transportation system, and understanding of how accidents occur have changed dramatically in those years. The proposed changes to § 21.3, namely the addition of a new §§ 21.3(d) and 21.3(e), aim to improve reporting of failures, malfunctions, and defects so it correlates to the current aviation system. Because these new regulations are process-based they will enable reporting requirements to be adapted to future reporting needs.

The purpose of the proposal as expressed in the original proposal for § 21.3 was to provide the FAA with the earliest possible notification of failures, malfunctions, or defects so the FAA may take appropriate mandatory action, such as issuing an AD. AC 21–9B, Manufacturers Reporting of Failures, Malfunctions, or Defects, further implies that compliance with § 21.3 will provide

the earliest possible notification to the FAA of a potentially unsafe condition and will ensure appropriate corrective action by the manufacturer.

An update of § 21.3 is also needed to align reporting requirements and processes with existing COS systems in place at many D&M organizations. These changes will also facilitate the FAA in meeting its responsibility to ensure D&M organizations are managing risk acceptably. Lastly, the proposed changes to § 21.3 will increase harmonization with other regulatory authorities' reporting requirements.

One of the ARC's objectives was to develop these changes while minimizing impact on many smaller D&M organizations that already meet the existing reporting requirements. The articles from these organizations typically introduce relatively little risk in the system. The ARC therefore did not want to negatively impact these smaller organizations by requiring changes that were not likely to improve risk management.

8.2.2 RECOMMENDED REGULATORY CHANGES

The proposed solution provides options for D&M organizations to declare the means by which to report. This is evidenced by enabling either reporting in accordance with the existing 21.3(c) or reporting under a process-based methodology in the new § 21.3(d).

The current part 21 does not provide a complete set of COS requirements. The risk management of a product or article following a design approval is a necessary function to assure the public's expectation of safety. Most, if not all, D&M organizations have implemented COS programs. The TC/PC holders and larger STC, PMA, and TSOA holders typically operate sophisticated COS systems. The new § 21.3(e) establishes regulatory expectations to implement COS systems for all D&M organizations and provides a clearer path for FAA oversight of these systems.

Section 21.3(e) contains minimum COS requirements. Section 21.3(e)(1) attempts to take advantage of existing processes many affected parties may have that comply with § 21.137, Quality system, and that use the existing list of occurrences in § 21.3(c) to report to the FAA. Section 21.3(e)(2) aligns with § 21.3(d) and is consistent with many existing COS agreements between ACOs and D&M organizations.

The change in § 21.3(g)(1) to increase the time to report to 72 hours was originally proposed in the FAA's initial submittal to the ARC. Increasing reporting time from 24 hours to 72 hours aligns with EASA. However, EASA has other stipulations that are not represented in the proposed change.

The ARC recommends implementing the changes to §§ 21.3 and 21.4 identified in the following table.

Section	Recommended Change	Existing 21.3 regulation, Amendment 21–92, Wording
21.3 Title	Failures, malfunctions and defects	Reporting of failures, malfunctions, and defects
21.3 (a)	 (a) The holder of a type certificate (including amended or supplemental type certificates), a PMA, or a TSO authorization, or the licensee of a type certificate must report any failure, malfunction, or defect in any product or article manufactured by it, that has <u>resulted in:</u> (1) Any occurrence listed in paragraph (c) or (2) Any occurrences identified by the holder as <u>determined by their safety analysis process in accordance with paragraph (d) of this section.</u> 	(a)The holder of a type certificate (including amended or supplemental type certificates), a PMA, or a TSO authorization, or the licensee of a type certificate must report any failure, malfunction, or defect in any product or article manufactured by it that it determines has resulted in any of the occurrences listed in paragraph (c) of this section.
21.3(b)	 (b) The holder of a type certificate (including amended or supplemental type certificates), a PMA, or a TSO authorization, or the licensee of a type certificate must report, <i>in accordance with paragraph</i> (c) or (d) of this section, any defect in any product or article manufactured by it, that could result in: (1) Any occurrence listed in paragraph (c) or (2) Any occurrences identified by the holder as determined by their safety analysis process in accordance with paragraph (d) of this section. 	(b) The holder of a type certificate (including amended or supplemental type certificates), a PMA, or a TSO authorization, or the licensee of a type certificate must report any defect in any product or article manufactured by it that has left its quality system and that it determines could result in any of the occurrences listed in paragraph (c) of this section.
21.3(c)	No Change	No Change
21.3(d)	(d) Occurrences, as determined by a safety analysis process developed by the holder of a type certificate (including amended or supplemental type certificates), a PMA, or a TSO authorization, or the licensee of a type certificate, and approved by the Administrator, must be reported as provided in paragraphs (a) and (b) of this section of which it is aware that has resulted in or may result in a finding of an unsafe condition by the Administrator.	New requirement. Resulted in renumbering of existing requirements so § 21.3(d) becomes § 21.3(f).

Table 4. Recommended Changes to §§ 21.3 and 21.4

Section	Recommended Change	Existing 21.3 regulation, Amendment 21–92, Wording
21.3(e)	(e) The holder of a type certificate (including amended or supplemental type certificates), a PMA, or a TSO authorization, or the licensee of a type certificate must:	New requirement. Resulted in renumbering of existing requirements so § 21.3(e) becomes § 21.3(g).
	(1) Identify, analyze, and initiate appropriate corrective action for any of the occurrences reported under paragraph (c) of this section, or otherwise;	
	(2) Monitor reported service problems, analyze, and initiate appropriate corrective action for any of the occurrences reported under paragraph (d) of this section.	
21.3(f)	The requirements of paragraph (a) of this section do not apply to—	§ 21.3(f) is same as current § 21.3(d)
	 Failures, malfunctions, or defects that the holder of a type certificate (including amended or supplemental type certificates), PMA, TSO authorization, or the licensee of a type certificate determines— 	
	(i) Were caused by improper maintenance or use;	
	(ii) Were reported to the FAA by another person under this chapter; or	
	(iii) Were reported under the accident reporting provisions of 49 CFR part 830 of the regulations of the National Transportation Safety Board.	
	(2) Failures, malfunctions, or defects in products or articles—	
	 (i) Manufactured by a foreign manufacturer under a U.S. type certificate issued under § 21.29 or under an approval issued under § 21.621; or 	
	(ii) Exported to the United States under § 21.502.	

Section	Recommended Change	Existing 21.3 regulation, Amendment 21–92, Wording
21.3(g)(1)	 Each report required by this section— (1) Must be made to the Aircraft Certification Office in the region in which the person required to make the report is located within <u>72</u> hours after it has determined <u>a reportable occurrence has</u> <u>occurred.</u> However, a report that is due on a Saturday or a Sunday may be delivered on the following Monday and one that is due on a holiday may be delivered on the next workday; 	 § 21.3(g) is same as current (e) Formerly § 21.3(e)(1) (e)(1) Must be made to the Aircraft Certification Office in the region in which the person required to make the report is located within 24 hours after it has determined that the failure, malfunction, or defect required to be reported has occurred. However, a report that is due on a Saturday or a Sunday may be delivered on the following Monday and one that is due on a holiday may be delivered on the next workday:
21.3(g)(2) and (3)	 (2) Must be transmitted in a manner and form acceptable to the FAA and by the most expeditious method available; and (3) Must include as much of the following information as is available and applicable: (i) The applicable product and article identification information required by part 45 of this chapter; (ii) Identification of the system involved; and (iii) Nature of the failure, malfunction, or defect. 	§ 21.3(g)(2) and (3) are same as current (e)(2) and (3)
21.3(h)	If an accident investigation or service difficulty report shows that a product or article manufactured under this part is unsafe because of a manufacturing or design data defect, the holder of the production approval for that product or article must, on request of the FAA, report to the FAA the results of its investigation and any action taken or proposed by the holder of that production approval to correct that defect. If action is required to correct the defect in an existing product or article, the holder of that production approval must send the data necessary for issuing an appropriate airworthiness directive to the appropriate aircraft certification office.	§ 21.3(h) is same as current (f)
21.4	§ 21.4(a)(5) The type certificate holder must identify the sources and content of data that it will use for its system. The data must be adequate to evaluate the specific cause of any in-service problem reportable under this section or § 21.3 that could affect the safety of ETOPS.	Refer to section 8.2.4 on Relationship to § 21.4, ETOPS Reporting Requirements

8.2.3 ALIGNMENT WITH OTHER EXISTING AND POTENTIAL REGULATIONS

The recommended changes to §§ 21.3(d) and 21.3(e) align with organizations that—

- 1. Hold an ODA under part 183;
- 2. Will be required to comply with the proposed SMS Requirements in part 5; and
- 3. Will implement an approved DO concept and associated reporting requirements.

ODAs that meet the requirements of § 183.63, Continuing requirements: Products, parts or appliances, should be able to show that the related processes will comply with the new \$ 21.3(d) and 21.3(e). Further, processes developed in accordance with part 5 will facilitate compliance to \$ 21.3(e)(2). Duplication of requirements will be avoided while allowing future changes in safety management.

The ARC recommends developing or revising the following guidance materials to support the recommended rule changes and facilitate FAA oversight:

- 1. <u>Criteria for consistent understanding of the language in § 21.3(d)</u> "has resulted in or may result in a finding of an unsafe condition by the Administrator".
- 2. <u>Acceptable compliance demonstration and verification regarding the safety analysis</u> referenced in § 21.3(d).
- 3. <u>Changes to AC 21–9B</u>, <u>Manufacturers Reporting of Failure, Malfunctions, or Defects,</u> to include the recommended guidance for §§ 21.3(d) and 21.3(e).

8.2.4 RELATIONSHIP TO § 21.4, ETOPS REPORTING REQUIREMENTS

The ARC also considered the potential impact to § 21.4, ETOPS Reporting Requirements. Section 21.4(a)(5) will need to be amended to include applicability of the new § 21.3(d) by deleting only reference to § 21.3(c) as follows:

(5) The type certificate holder must identify the sources and content of data that it will use for its system. The data must be adequate to evaluate the specific cause of any in-service problem reportable under this section or § 21.3 that could affect the safety of ETOPs.

8.2.5 SHORT-TERM RISK MANAGEMENT

8.2.5.1 Background

FAA Order VS8000.367A, Aviation Safety (AVS) Safety Management System Requirements, requires the FAA to define a process for acceptable safety risk in the short term while long-term safety risk control/mitigation plans are developed and implemented. In some situations there can be a short-term design solution that adequately mitigates the risk in the short term but is not fully certifiable because it does not adequately address a known hazard. From a safety risk management perspective, it may be desirable to enhance the safety of the in-service fleet by implementing the short-term solution. For economic reasons, it may also be desirable to continue delivering products with the short-term design solution until the long-term solution can be developed.

Current part 21 regulations, policy, and guidance require any changes in type deign to be fully compliant to all applicable regulations at the installation or product level, regardless of whether the change improves the safety or product capability. There are ongoing situations where time-limited exemptions are required to allow the incorporation of a product safety enhancement for an interim period until a fully compliant product-level change can be identified. There are also situations in which a product improvement that maintains or enhances the product safety cannot be incorporated without an exemption because the installation or product level cannot be shown to fully comply. The nature of the exemption process does not lend itself to expeditiously addressing in-service fleet safety concerns.

Occasionally after a TC has been issued, the FAA discovers the product does not comply with one or more of the applicable airworthiness provisions. If the noncompliant product is considered airworthy, manufacture and operation of that product type is allowed to continue. However, certificating a major change for that product, even if the change clearly improves the level of safety, currently requires the applicant to bring the product back to the level of safety intended by its basis of certification or be granted an exemption. This may place an inequitable burden on major design changes, which is clearly a deterrent to discretionary product improvements as well as production incorporation of improvements mandated by an AD. A short-term solution to a known hazard may be available faster than the process for obtaining an exemption and therefore may prolong the incorporation of a safety enhancement into the fleet.

SMS is intended to move to a more risk-based approach and aligns the responsibility of safety management with the product or service provider while ensuring compliance to the airworthiness regulations. To achieve this objective, the regulations for changes to type design should reflect and allow for incorporation of short term product and safety enhancements when the risk is shown to be acceptable, even though the installation or product may not fully comply, provided there is a plan for a long-term design change that does comply with the airworthiness regulations. Accordingly, the changes to part 21 compliance regulations should be tied to part 5 SMS applicability to products. Additionally, there should be criteria for when an exemption is necessary.

8.2.5.2 Recommended Regulatory Changes

The ARC recommends that it continue to develop regulatory proposals aligned with SMS philosophies to enhance safety. The objective would be to complete the task by October 2014 by providing an addendum to this report.

The regulatory applicability would be limited to non-substantial, non-significant § 21.101 changes to type design and hence only be major changes per § 21.93, Classification of changes in type design. The TC holder would rely on their SMS SRM processes to define the short and long-term acceptable risk while presenting a plan for a long-term design change that complies with the airworthiness regulations.

Criteria is needed for when the part 21 process should be used instead of an exemption,¹⁰ defining the long-term compliance plan, process for applicant and regulatory authority agreement on short-term mitigation and safety assurance of the short-term risk mitigation or product enhancements. The proposal should enhance part 21 so parts 5, 11, and 21 function more systemically to maintain or enhance safety of the product while ensuring compliance with the airworthiness regulations.

The ARC recommends changes to part 21 to address situations described above, examples of which are provided below:

- Short-term solutions:
 - There is an unacceptable risk in the fleet, and a short-term solution that adequately mitigates the risk is available but not fully certifiable because it does not fully address a known hazard. For economic reasons it may also be desirable to continue delivering products with the short-term design solution until the long-term solution can be developed.
 - There is a known noncompliance in production for which the holder has defined a recovery plan. An interim fix is available that mitigates some of the risk, but is not fully compliant.
- Product enhancements that maintain or enhance product safety:
 - Some types of changes are often considered where because of the constraints of having to comply with the latest regulatory requirements, whether because of safety issues or revisions to regulations, it becomes economically impractical to implement the change even though those changes would provide an incremental improvement in the performance, functionality, reliability, or safety of the product. These fall under the following categories:
 - Product improvements for—
 - Added or improved functionality,
 - Reliability enhancements,
 - Improved performance,
 - Maintainability enhancements, and
 - Producibility enhancements;
 - Part obsolescence; and
 - Alternate or substitute parts.

The ARC recommends developing proposed regulation changes and guidance or process proposals to address safety risk that is acceptable in the short term while long-term safety risk control/mitigation plans are developed and implemented.

¹⁰ In the context of product improvements that do not have a plan for a long-term fully compliant product change, NATCA's position is an exemption should be required.

Recommendation 4—Part 21 Cleanup and TSO Modernization: The ARC recommends FAA consideration of several proposed changes and updates to various part 21 regulations, which primarily reflect clarifications to eliminate confusion, modernization to reflect current practices, and updates to align with other recommendations in this report for a systems approach to certification.

9.0 PART 21 CLEANUP & TSO MODERNIZATION

9.1 PART 21 CLEANUP

The following proposed changes are cleanup issues that would eliminate confusion and inability to comply in the current regulation, if addressed. They do not affect product safety.

9.1.1 SECTION 21.8 AND 21.9 ISSUE PAPER

As recommended by the 14 CFR Part 23 Reorganization ARC's 2013 report, the part 21/SMS ARC recommends—

- 1. Clarifying the applicability of § 21.8, Approval of articles, and § 21.9, Replacement and modification articles; and
- 2. Adding provisions to § 21.9 for the productions of parts allowed under the authority of § 21.8.

Over the past decade there have been discussions regarding approved parts and unapproved parts. The level of activity seems to parallel the FAA field applications regarding the Suspected Unapproved Parts program. There numerous opinions regarding the applicability of §§ 21.8 and 21.9.

It is important to highlight that in the 1995 Suspected 'Unapproved Parts' Program Plan, which was prepared and submitted to the Administrator by the FAA Suspected 'Unapproved Parts' Task Force, the task force clarified that an "approved part" is not synonymous with "a part that has received a formal FAA approval."

The terms "approved parts" and "unapproved parts" as used in this report are not legal definitions, but a reflection of the need to have a broad term that identifies parts that should, or should not, be installed on an aircraft. In this report, parts that should be used on an aircraft (that is, "approved parts") are described as parts "acceptable for installation" or "eligible for installation."

9.1.1.1 Applicability of § 21.8

Section 21.8 begins with the scope of the section by stating "If an article is required to be approved under this chapter ..." In clarifying the scope of § 21.8, which chapter is this regulation addressing?

§ 21.8 Approval of articles.

If an article is required to be approved under this chapter, it may be approved—

- (a) Under a PMA;
- (b) Under a TSO;
- (c) In conjunction with type certification procedures for a product; or
- (d) In any other manner approved by the FAA.

Part 21 is contained within chapter I, Federal Aviation Administration, Department of Transportation of 14 CFR. Therefore when § 21.8 refers to an article required to be approved under "this chapter" it is referring to chapter I, the Federal Aviation Regulations. Although this is readily apparent for required equipment within subchapter C, Aircraft, as these required equipment are approved via the type certification process, the applicability is not as apparent for non-required/optional equipment nor certain applicability's within part 91, General Operating and Flight Rules. Part 91, carries specific criterion for operational equipment which are often absent from specific "approval" requirements.

The following are examples of regulatory language that cause the confusion:

- Section 91.205(b)(11) serves as an example where the article must be approved: "For small civil airplanes certificated after March 11, 1996, in accordance with part 23 of this chapter, an approved aviation red or aviation white anti-collision light system."
- Section 91.225(a)(1) serves as an example where the article is approved by specific • TSO: "After January 1, 2020, and unless otherwise authorized by ATC, no person may operate an aircraft in Class A airspace unless the aircraft has equipment installed that— Meets the requirements in TSO-C166b ..."
- Section 91.215(a) serves as an example where the article must meet the performance of • the TSO standards but not necessarily be approved to the TSO: "For operations not conducted under part 121 or 135 of this chapter, ATC transponder equipment installed must meet the performance and environmental requirements of any class of TSO-C74b (Mode A) or any class of TSO-C74c (Mode A with altitude reporting capability) as appropriate, or the appropriate class of TSO-C112 (Mode S)."
- Section 91.205(d)(2) serves as an example of where the operating rules are silent on the subject of approval: "Two-way radio communication and navigation equipment suitable for the route to be flown."

9.1.1.2 Applicability of § 21.9

Under the final rule for the Production and Airworthiness Approvals, Part Marking, and Miscellaneous Amendments (74 FR 53368, October 16, 2009) the FAA clarified "The provisions of § 21.9 apply to the *producer* of any part that may be used as a replacement or modification article, not just parts that were produced specifically as replacement or modification articles."

When answering questions submitted by the U.S. Small Business Administration, the Administrator reiterated that "Section 21.9 governs the production, not the sale, of articles and does not prohibit distributors from selling articles,"¹¹ and "The requirements of this rule apply to products or articles as they are manufactured."¹²

Therefore, it is clear that the applicability of § 21.9 is for the production of replacement and modification parts and does not apply to the sale, distribution or installation of parts.

¹¹ 74 FR 53368, 53373. ¹² 74 FR 53368, 53374.

9.1.1.3 Recommendations

The ARC recommends AC 20–62, Eligibility, Quality, and Identification of Aeronautical Replacement Parts, paragraph 6, Discussion, be amended to include an explanation of the scope of §§ 21.8 and 21.9.

Specifically, the ARC recommends adding the following paragraphs to AC 20–62:

6. DISCUSSION. [...]

d. Applicability of Section 21.8. Section 21.8 requires that if an article is required to be approved under the Federal Aviation Regulations, that the part may be approved under a PMA, TSO, in conjunction with type certification procedures for a product; or in any other manner approved by the FAA. If the regulations are mute about explicit approval requirements such as for non-essential, non-required equipment, section 21.8 does not apply and the installer will default to section 43.13 for in-service installations. Section 21.8 applies to—

(1) All required equipment within subchapter C,

(2) Anywhere in part 91 where the equipment requirements explicitly require approval, and

(3) Any required equipment in part 121, 129, and part 135.

e. Applicability of Section 21.9. Section 21.9 applies to the production of replacement and modification parts and is does not apply to the sale, distribution or installation of parts.

Proposed Revision to § 21.9

A recent development originating with the ARC highlighted an apparent unintended limitation of § 21.9. Although § 21.8 allows for parts to be approved "In any other manner approved by the FAA." (§ 21.8(d)) there are no provisions within § 21.9 to produce those "otherwise approved" parts. As such, it is the recommendation of the ARC to include similar language to § 21.8(d) within § 21.9.

The ARC recommends amending § 21.9(a) to add paragraph (a)(7), which reads: "Produced in any other manner approved by the FAA."

9.1.2 SECTION 21.335, RESPONSIBILITY OF EXPORTERS

Section 21.335(b) states that each exporter of aircraft parts must "[p]reserve and package products and articles as necessary to protect them against corrosion and damage during transit or storage and state the duration of effectiveness of such preservation and packaging." The requirement to state the duration of a packaging's effectiveness arises from ATA Spec 300, Specification for Packaging of Airlines Supplies. However, Spec 300 does not describe or contemplate a duration requirement for a single-use packaging. The duration and effectiveness of single-use packaging is not a metric that is typically available from packaging manufacturers, thus making it impossible to satisfy § 21.335(b) under current industry standards and practices. Furthermore, the FAA has produced no guidance material to explain to industry or FAA personnel the appropriate methods for determining and stating the duration and effectiveness of packaging, and it is unclear whether the provision offers any safety benefits, as it has never been

treated as an enforcement priority. The ARC therefore recommends the FAA remove the phrase "and state the duration of effectiveness_of such preservation and packaging" from § 21.335(b). A copy of the Aviation Suppliers Association (ASA) white paper providing additional details on this proposal is located in appendix M to this report.

The ARC recommends the removal of § 21.335(b) from 14 CFR.

9.1.3 TSO CLEANUP RECOMMENDATIONS NOT TIED TO ARC OBJECTIVES

The TSO Sub team identified four items it would like to see changed that were not tied to the ARC charter or taskings.

For these four items the ARC recommends-

- <u>Maintaining the privilege for TSO holders to make minor or insignificant (sub-minor)</u> changes to articles without further approval;
- <u>Clarifying the TSO application data, manufacturer data and furnished data requirements.</u> (Refer to the TSO Sub team Report, included as appendix H to this report);
- <u>Developing expanded guidance to promote the uniform definition and treatment of integrated non-TSO functions by applicants, installation developers, and the FAA.</u> (Refer to the TSO Subteam Report, included as appendix H to this report); and
- An applicant should submit to the Administrator a signed undertaking to carry out the responsibilities as a DAH before issuance of a design approval.

9.2 PART 21 RECOMMENDATIONS FOR TSO PROGRAM

Regarding the TSO program value and challenges identified in appendix C to the TSO Subteam Report included as appendix H to this report, the TSO Subteam reviewed the list of issues and problem statements from the three sources (FAA, ARC members, and FAA-industry workshop), to distill the key issues that, if appropriately addressed, would lead to material improvements in the effectiveness and efficiency of the TSO program. These key issues are—

- Acceptance/approval of integrated non-TSO functions,
- Management of post-TSOA design discrepancies,
- Definition of substantiation data and data submittal items,
- Minimizing re-review of TSOA substantiation data at installation approval, and
- Elimination of unnecessary TSO deviation requests.

The TSO Subteam continued to develop six rulemaking recommendations and proposals for related policy and guidance, for consideration by the ARC in its final report to the FAA. Although the recommendation in section 9.2.1 below is written with reference to a "certified TSO organization," these recommendations are generally intended to be implementable by either a certified TSO organization or an expanded FAA (ODA) delegation system, incorporating TSO design functions.

9.2.1 TSO RECOMMENDATIONS

The TSO system has been described as a "self-certification" process, but currently requires FAA involvement to verify the manufacturer's compliance statement and issue an authorization to apply TSO marking for each article. The FAA could better manage its resources and streamline the TSO process by allowing the traditional FAA review portion to be performed by certified TSO organizations or appropriately delegated organizations, up to and including issuance of the TSOA letter. The FAA already has a rule basis to qualify TSO organizations under § 21.605, but part 21 subpart O, Technical Standard Order Approvals, currently requires an application to be submitted to the ACO (under § 21.603, Application) and the FAA to issue the authorization (under § 21.611, Issuance).

<u>The ARC recommends allowing TSO organizations to issue their own TSOAs, relative to</u> scalable privileges for particular types of TSO standards. (Alternate approaches via a certified <u>TSO organization or expansion of TOS ODA functions.)</u>

Under current rules, the holder of a TSOA is expected to maintain the performance of the TSO article relative to the TSO standard. However, to support airworthiness or contractual requirements, TSO manufacturers are typically required by their customers to maintain performance of the article relative to requirements that are not part of the TSO standard. The TSO system could be modified to allow manufacturers to better align their requirements under the TSO system with those requirements they are electing to meet as part of airworthiness or contractual obligations. EASA's system supports the declaration of this type of manufacturer-defined performance by requiring the submission of a Declaration of Design and Performance (DDP), and expects TSO manufacturers to control the article's performance relative to that DDP. Refer to appendix M to this report.

The ARC recommends clarifying the types of data that can be approved under a TSOA (that is, type design of the article and declared performance of the article including non-TSO functions and incomplete TSO), and expectations for acceptance of approved TSO data for installation. (Require and approve DDP via revision to § 21.601(b)(2) and proposed new § 21.601(b)(2) and § 21.603(b)(3).)

Non-TSO functions have become increasingly prevalent in avionics systems mainly because of added processing capabilities in newer integrated circuits as well as larger and cheaper memory capabilities. However, mechanical systems TSOs, such as seats, can also contain provisions to host non-TSO functions (for example, embedded passenger entertainment devices). These added non-TSO functions are mainly market or original equipment manufacturer driven to optimize installation capabilities while minimizing certification costs. Added non-TSO functions have historically been handled through FAA policy, because part 21 subpart O does not specifically address or codify their embedded existence. This resulted in a problem with some ACOs who feel the current part 21 subpart O does not allow for a more detailed evaluation of non-TSO functions are inseparable from the hosting TSO article design at the time of manufacture.

The ARC's proposed change would provide several benefits. For the TSO applicant, an initial review of the manufacturer-declared performance can be made, and credit for software, hardware, and environmental testing in support of installation can be acknowledged in the TSOA

letter. For the FAA, more efficient use of engineering resources because the review of the added functions can be more thoroughly accomplished with the TSO manufacturer as opposed to attempting to accomplish this review during each subsequent installation approval. Both the TSO article as well as the integrated non-TSO functions, must be appropriate to support the intended installation.

The ARC recommends proposed new §§ 21.603(a)(3) and 21.619(d), Design changes, for subsequent design changes. (Additional guidance including a "decision table" to assist in differentiating between TSO supporting features and integrated non-TSO functions.)

Current regulations require the use of model numbers to maintain configuration control of TSOAs. From a practical perspective, however, model numbers are frequently used as marketing identifiers; they do not consistently provide a meaningful reference for configuration control of TSO article designs.

The ARC recommends rule revision to remove the term "model number" from TSO rules and replace it with a requirement for a "unique identifier." (Revision to §§ 21.603(b) and 21.619 for subsequent design changes.)

Currently § 21.603(a)(1) implies that application for TSO is made after all design and development is completed. Designing and developing a TSO article is a lengthy process especially for complex articles. During this time, TSOs could be revised or new TSOs could be introduced. Current policy, as defined in FAA Order 8150.1C, Technical Standard Order Program, paragraph 6–1(b), allows an applicant 6 months from release of a newer revision to apply with the previous revision. Although 6 months seems to be a reasonable timeframe, many complex system developments take significantly longer and the 6 month grace period is not sufficient. The two current options to address this situation are—

- 1. Request a petition for exemption (per 14 CFR part 11, General Rulemaking Procedures), which an applicant is required to submit at least 120 days before the exemption is needed. After submission, approval can take many months because of the requirement of publication in the Federal Register and a public comment period.
- 2. Comply with the newer revision TSO or add a newly released TSO.

If a new revision of a TSO or newly introduced TSO provides no benefit or does not impact flight safety, this additional work to submit a petition for exemption or complying with the latest TSO during an in-process development project could be a large burden on an applicant, potentially driving redesign and/or retesting and preventing expeditious introduction of safety enhancing products to market. Because TSOA is a self-certification based on a statement of conformance, the responsibility of reviewing and ensuring any new or revised TSO(s) does not affect the certification basis or design is the burden of the TSO applicant regardless of when application is made.

If the applicant is allowed to declare the effective TSO revision levels at the beginning of a project, FAA/applicant communication on complex certification issues could be improved. Currently, when an application is submitted, there could be several iterations because of certification basis disagreements causing potential applicant design rework and weeks to months

of delay in issuance of a TSOA letter. Based on the regulation change, adoption of the new or recently revised TSO(s) would be voluntary unless the change is because of a safety-of-flight issue or required for other reasons, such as interoperability, and deemed mandatory by the FAA. The benefits of this change can be tracked through improved turnaround time of TSOA letter issuance from the FAA, on final submittal of a statement of conformance certifying that the applicant has met the requirements of the subpart and that the article meets the applicable TSO(s).

The ARC recommends changing part 21 to establish the effective TSO revision level at the beginning of the project, not at the end. (Revision to § 21.603(a).)

Before the changes to part 21 and part 45, Identification and Registration Marking, implemented by amendment 21–92, dated April 16, 2011, and amendment 45–26, dated April 16, 2011, respectively, the requirement that an article meet applicable TSO performance standards to be so marked was defined in § 21.603 as shown below:

"With the part 21 and 45 amendment changes implemented in April 16, 2011, the previous § 21.603(a) requirements were re-codified into § 45.10(b) as shown below, which still requires the article to meet applicable performance standards before marking can be applied."

Currently, when a TSO holder identifies a design deficiency, the holder must stop shipment (that is, stop marking per § 45.10(b), Marking) and report the deficiency to the ACO. If the ACO determines the deficiency does not result in an unsafe condition (that is, does not require a § 39.19 action), then to resume shipping of articles the TSO holder must either correct the deficiency immediately (which may not be practical) or request a deviation under § 21.618, Approval for deviation. However, the intent of the § 21.618 provision was to provide a means for the TSO "applicant" to propose a true "equivalent level of safety" (ELOS) to a TSO performance requirement, and was not meant to forgive a design deficiency or oversight of the TSO holder.

The ARC's proposal provides for a risk-based approach to handle TSO design deficiencies that do not rise to the level of an "unsafe" condition. It may also be in the public's interest in the case where the stop shipment could result in a major economic burden to the end-user of the article.

The ARC recommends a process for the TSO holder to continue marking TSO articles following a determination of "a design discrepancy that does not result in an unsafe condition." (Revision of § 45.10(b) and proposed new§ 21.616(i), Responsibility of the holder.)

10.0 BUILDING BLOCK APPROACH TO DESIGN ORGANIZATION

<u>The ARC was concerned with the anticipated challenges in implementing a DO system</u> in the current state, so the ARC chose to recommend a voluntary DO approach, combined with a "building block" transition plan, to better position the FAA and industry for a future <u>DO environment.</u> This building block approach would allow the FAA and industry to immediately begin taking steps to improve processes and make the changes necessary for DO implementation. These steps will enable a much more manageable transition to DO in the future.

The building block approach requires the FAA and industry to make changes, including-

- Establishing accountability framework/applicant showing as the foundation for a CAS,
- Transitioning to a centralized, systemic oversight model, and
- Optimizing use of the existing ODA system.

One of the key aspects of the U.S. aviation industry is its continuous innovation and ability to foster entrepreneurs to develop the next generation of safety improvements. Therefore, it was important to the ARC's DO Working Group to establish a "lower overhead" path that would allow for "low-risk" innovation but leverage the strengths of established organizations for "higher risk" designs. By making DO voluntary, each organization would be making the decision to become a DO based on the cost/benefit (business case) for the organization. This approach promotes actions by both industry and the FAA to ensure the benefits of the privileges of a DO outweigh the costs and burden of implementation and maintenance. In addition, all organizations that meet the minimum requirements are eligible to apply to become a DO (refer to section 6 of this report); DOs do not require an FAA determination of need.

Therefore, <u>the ARC recommends that promulgation to establish DO requirements would</u> make it a voluntary choice for organizations to obtain, which means there is no need to define applicability thresholds for any design approval applicants or considerations for <u>small business and cost/benefit</u>.

The following sections provide additional details on how the FAA and industry can achieve these steps.

10.1 EVOLUTION OF FAA OVERSIGHT: CENTRALIZED AND PERFORMANCE-BASED

Recommendation 3—Evolution of FAA Oversight Toward Performance-Based Systems Safety (SMS) Approaches: The ARC recommends development of a performance-based single surveillance oversight approach that aligns with proposed changes to design and production organizational requirements and a systems approach to certification. The three key oversight areas are: Organizational - Transition from traditional show/find compliance to organizational PBO model; Product and Articles - Transition from the FAA's traditional role of direct project involvement to a LOPI approach focused on performing governmental functions and enhanced showing capabilities; Post Certification (COS) - Transition from a traditional reactionary approach to a systemic (process based) surveillance model. Establishing a central FAA oversight organization will achieve standard surveillance practices, create centralized policy, be a single source/repository for the oversight data that will drive the risk-based modeling controls, and allow for a highly trained staff in system surveillance, skill management, and a single source for corrective actions. As companies continue to evolve to a systems safety-based certification and organizational oversight, a centralized FAA oversight system will provide consistent and progressive assessment and surveillance processes leading to the performance-based standard.

Establishing a single FAA oversight model is a fundamental first step in the building block approach to DO. As companies evolve from ODA processes to CAS and SMS systems, a centralized FAA oversight system will provide consistent and progressive assessment and surveillance processes leading to the performance-based standard. The ARC recommends FAA oversight teams report to a centralized organization, which will—

- Achieve standard surveillance practices.
- Centralize policy responsibility ensuring consistency in interpretations.
- Allow the ACO to focus on safety-critical functions.
- Provide "third-party objectivity," as the office does not work programs with the DO.
- Provide a single source/repository for the oversight data, which will drive the risk-based modeling controls.
- Manage skill development practices for surveillance staff.
- Allow for a highly trained staff in system surveillance.
- Provide a single source for oversight of corrective actions.

The FAA oversight team will be a team of individuals that have a corresponding role with the organization. For example, if a company holds design, manufacturing, and/or repair certificates, the oversight team will consist of engineers (ASEs), manufacturing and flight standards inspectors (ASIs), and AEG members to parallel the DO's capabilities. The surveillance activities are system-level, not program-specific. Therefore, policy will not contain specific
criteria by which the FAA will inspect and evaluate the organization. The oversight team will base its surveillance on the procedures contained within the organization's operating manual, adherence to the procedures/processes, and evaluating the process outcomes for quality.

To address these concerns, the ARC recommends the FAA create a single oversight presence to address three key oversight areas:

- <u>Organizational: Transition from traditional show/find compliance to organizational</u> <u>PBO model.</u>
- <u>Product and Articles: Transition from the FAA's traditional role of direct project</u> involvement to a LOPI approach focused on performing governmental functions.
- <u>Post-Certification (COS):</u> Transition from traditional reactionary approach to a systemic (process-based) surveillance model.

10.1.1 OVERSIGHT OF THE ORGANIZATION

For organizational oversight, capability is shown or determined by an organization's experience or demonstration of capability based on the existence of adequate processes used to conduct projects with FAA involvement before DO. Additional third-party suppliers can be used to achieve capability not already existing within the DO. FAA oversight consists of organization processes such as company-required processes (SMS, CAS, and QMS) as well as Maintenance, Repair, and Alteration (MRA) and manufacturing processes. The FAA will focus on the process evaluation as opposed to individual product or article compliance.

10.1.2 OVERSIGHT OF PRODUCT/ARTICLE CERTIFICATION

The design/manufacturing oversight model for product or article certification includes the introduction of FAA LOPI. If a company has previously demonstrated its engineering and design capabilities (applicant showing) and is not seeking to expand that capability, the LOPI should be "0" outside of any required governmental functions, and the company will be expected to have the autonomy to complete all certification efforts.

Once these efforts are complete, the company makes a declaration of compliance, which does not require an FAA assessment before it issues the certification/approval to the company for that product or article. In cases where special conditions, ELOS, issue papers, alternate methods of compliance, or any other governmental functions are required, there will be a LOPI by the FAA. A LOPI is required but should minimize impact to the company's critical path activities. In terms of oversight, surveillance of the company is expected to be the primary function performed by the FAA, regardless of the LOPI. Company activities that required LOPI for a specific product or article can be used as demonstration of capability for future efforts.

10.1.3 OVERSIGHT OF POST-CERTIFICATION (COS)

The FAA and certificate holders will transition from the traditional reactionary model to a systemic (process-based) surveillance model. The level of surveillance will depend on processes for hazard identification, criticality of products, and risk-based safety criteria determinations. The process will include outputs for corrective actions to the FAA (such as service bulletins,

ADs, and special airworthiness information bulletins). Parameters must be set to ensure these corrective actions do not impose added costs to operators.

Hazard identification will include monitoring and trending safety analysis data with multilevel inputs by incorporating a data-driven, risk-based approach for safety assurance and SRM. Current governmental and holder processes (such as Monitor Safety/Analyze Data (MSAD) and processes at Boeing and Bombardier) should be evaluated for effectiveness and readiness. For more information on hazard identification, refer to the SMS Working Group Report in appendix F to this report.

A company with a functional SMS will be responsible for proposing to the FAA the corrective actions required to address safety issues. For example, a company could request the FAA release an AD based on the company's risk assessment.

10.1.4 OVERSIGHT ASSESSMENT METHODOLOGIES

10.1.4.1 Initial Assessment

The initial assessment¹³ of a company should be an evaluation of its ability to satisfy the requirements for an applicant showing/DO. The initial assessment establishes that the organization meets the basic requirements for the authority, and has the necessary process capabilities in place. The FAA will assess the organization's effectiveness in ensuring compliance, product safety, and SRM using performance measures that will determine the frequency and depth of ongoing surveillance activities.

The initial assessment is intended to establish whether an organization has the required process coverage and level of process maturity. Several models are used across the industry to assess process maturity. In April 2012, the SM ICG published the Safety Management System Evaluation Tool as an objective method "to indicate the expected standard of an organization's SMS in terms of compliance with the SMS regulation and its performance to effectively manage safety risk."¹⁴ This tool provides an internationally harmonized standard for assessing process maturity, and can be expanded to include additional regulatory requirements for a DO.

The ARC's Oversight Working Group expanded the ICG's SMS assessment tool to include requirements for a CAS and QMS. These components—SMS, CAS, and QMS—form the basic components of a CDO. In advance of final requirements for CAS and QMS, the working group used the draft proposed regulations from the 2008 CDO ARC Report. The working group presented this "prototype" tool in appendix C to its report (included as appendix G to this report), as an example of what an assessment tool could look like. Before endorsing this as a tool for broader use, the FAA and industry should revise the model to incorporate lessons learned from recent applications of the ICG's SMS assessment tool.

In the 2008 CDO ARC Report, the CDO ARC recommended that "an organization applying for a CDO certificate or an expansion of its existing certificate undertake a self-assessment. This self-assessment should be a formal undertaking with records generated of the findings and observations of the evaluators." The self-assessment should apply the same evaluation tool as

¹³ Referred to as "appraisal" in section IV. J of the 2008 CDO ARC Report.

¹⁴ http://www.skybrary.aero/bookshelf/books/1774.pdf

the FAA assessment; however, the FAA may consider the extent and thoroughness of the self-assessment in developing its plan for assessment. Following satisfactory completion of the self-assessment, the candidate organization may submit its application for CDO to the central FAA DO office.

The application process (preferably electronic/online), will be defined by FAA policy and should consist of a completed application form, a written request for assessment, identification of requested capabilities, and an enclosure of self-assessment.

10.1.4.2 Surveillance

As part of its fundamental oversight responsibilities, the FAA will conduct surveillance on the organizations to evaluate performance through inspection. This surveillance will use a systematic approach focused on validating the processes/procedures by means of inspection, and a verification of the organizations' capability to follow its procedures through an evaluation of the products and/or approvals that result from the system. Surveillance could include, but is not limited to—

- Reviewing the company process/procedures within the operating manual,
- Reviewing the company self-assessment/self-surveillance data and corrective actions,
- Inspecting the company for compliance with its process/procedures within the operating manual,
- Reviewing the work performed and evaluating performance for quality assurance,
- Ensuring required training has been completed,
- Providing constructive feedback, and
- Taking corrective action, as necessary.

Surveillance does not include program-specific involvement; it is based at the system and process levels. The FAA will still have the responsibilities contained in governmental functions. However, the ARC has termed these actions as LOPI and will be the tasks required of the FAA office in charge of the respective program and not within the FAA oversight office's job duties.

Risk-Based Decisionmaking: The interval and depth of surveillance activities should be based on the safety risk of the product or article, capability of the organization, past performance of the organization, complexity of the programs, and authority/privileges granted to the organization. An organization's self-surveillance activity, including its performance in addressing voluntary disclosures and notifications of noncompliance, is an indication of a closed-loop system that will ensure continual improvement of the organization and address lessons learned. In addition, an organization's performance, as indicated by the following factors, will determine the level and frequency of surveillance:

- Capability,
- Past performance, including
 - o ADs/safety findings on approved products,

- o Quality escapes,
- Noncompliances, and
- Self-surveillance findings;
- Complexity, including—
 - New and novel technology concerning current processes,
 - LOPI from ACO (If the program has a high level of involvement from the ACO, is it a good candidate for oversight?),
 - o The ability to manage and maintain control over large programs, and
 - o Process robustness; and
- Multiple authorizations/privileges/certificates.

10.1.4.3 Expansion Assessment

When a company requests an expansion to the scope of its authority that involves substantially new processes that have not been previously demonstrated, the company must demonstrate it is capable of operating under its proposed processes before receiving an amended certificate with the expanded capabilities. As a tool in the performance of the self-assessment and the subsequent FAA evaluation, the DO applicant should have traceability between its processes and the regulatory requirements they are intended to address.

Self-assessment: The application for expanded scope would include a self-assessment to show a company's readiness to function with the new processes, and exercise its capabilities to make determinations under the changed scope. The self-assessment should be a formal undertaking with records generated of the findings and observations of the evaluators using the same process/performance measures used for the initial assessment. This assessment may be abbreviated based on the difference between regulatory requirements regarding the additional capabilities requested. The FAA may use this assessment in focusing its assessment activities.

FAA assessment: Following company application for expanded scope, the FAA will perform an assessment based on the criteria used for the initial assessment to determine that the company has shown it is fully capable of operating within the changed scope.

In the case of an expansion in scope for a company in good standing, the FAA may rely on the self-assessment in issuing the expansion of the applicant showing/DO certificate.

- If the expansion in scope is minor, the applicant showing/DO self-assessment may be sufficient to allow the FAA to expand the certificate scope with no further demonstration.
- The use of the applicant's self-assessment to adjust the scope of FAA activities is solely at the FAA's discretion and should follow the safety management principles of targeting safety-critical efforts.
- The FAA is under no obligation to complete its assessment within a minimum time limit or number of projects.

The ARC recommends a centralized, systemic (process-based) oversight approach for initial and ongoing assessments. The assessment methodology will cover a standardized approach to quality, design, and safety. In support of this recommendation, the ARC's Oversight Working Group has provided a capability-based assessment tool, PROs/CONs analysis of oversight management options, and supporting rationale for the recommendation. FAA oversight teams would report to a centralized FAA organization. Establishing a central FAA oversight organization will achieve standard surveillance practices, create centralized policy, be a single source/repository for the oversight data that will drive the risk-based modeling controls, and allow for a highly trained staff in system surveillance, skill management, and a single source for corrective action.

10.1.5 FAA COLLABORATIVE OVERSIGHT

10.1.5.1 Define and Teach Systems Engineering

Systems engineering is a cultural shift the FAA must understand and undergo. This change in culture will only help the FAA and industry when moving toward systemic self-surveillance and collaborative oversight. Understanding systems engineering means that in some cases FAA employees may move from a subject matter expert role to a systems engineering role. The FAA must be able to maintain its current level of oversight to assist in maintaining the current U.S. safety rating. As stated in section 2.5, if the anticipated industry growth continues by 0.7 percent per year through 2022 and the FAA employment remains level, the rate at which the FAA is able to support critical activities becomes increasingly more difficult.

A Systems Engineering Management Plan (SEMP) is a technical planning document used in a variety of industries to develop, oversee, and maintain high risk and geographically dispersed programs and processes. This type of document can be used to assist the FAA in understanding how to audit a set of processes. This technique includes understanding that company processes are not a universal solution and that there could be more than one correct way of performing a function, documenting a process, and meeting regulatory compliance. Auditing a process is confirming the process is performing well, not auditing a specific product.

A SEMP is generally used by the organization responsible for generating and managing technical programs. The SEMP should be coordinated with the project plan for integration of the technical planning and modifications related to the allocated resources including cost, schedule, personnel, facilities, and deliverables required. The SEMP is also used to evaluate the team's technical approach, make technical risk assessments, and measure progress. It may identify key milestones where the FAA has assigned oversight. This is the area of the SEMP where the FAA and industry would use the plan jointly, similarly to the way interagency agreements are used between government agencies. The SEMP can identify specific processes used in the design environment.

In a system engineering environment, the process will be the FAA's area of oversight, compared to its current history of product oversight. As long as the process meets the intent of the requirements, it may be defined at the discretion of the company. Processes will vary from company to company. The SEMP assists the FAA in ensuring the organization is following its process and the process is working well, in turn assuring the FAA that products are being designed according to the required regulations. Figure 4 below provides an illustration of the systems engineering concept.



Figure 4. Systems Engineering Approach to Process Management and Oversight

This illustration describes how an effective assessment will validate that the organization has procedures and supporting processes in place to satisfy the requirements, and verify that the processes are followed by personnel who meet the qualifications defined in those processes. The validation step can be described as "did you build the right product?," and the verification step can be thought of as "did you build the product right?"

It is important to ensure both steps are effective. For example, an applicant should rely on the expectation that when people in the organization follow approved processes, compliant products will result. This requires effective validation of the processes during their development and initial assessment. Effective validation ensures all applicable requirements are allocated to the procedures (no requirements are missed), requirements captured are valid (based in regulatory requirements), and supporting processes define clear, unambiguous steps to be performed. Refer to NATCA's dissenting opinion in section 12 of this report.

Furthermore, the industry should expect a CDO to meet its obligations, and ensure qualified personnel follow its approved processes. This includes the individual performing a self-assessment of their own work, the organization performing self-assessments of its own activities, and the FAA performing ongoing surveillance activity of the DO.

Moving the FAA workforce to this kind of systems-based thinking will require training, practice, and a cultural shift. However, this is an attainable goal with an agile workforce that is willing to learn. The concepts can be promoted through webinars, orientation sessions, and pilot projects. Implementing and recognizing systems engineering concepts at an earlier stage will assist the workforce transition. This will provide for a phased-in approach for the FAA and industry.

10.1.5.2 Central Oversight

Table 5–1 of the Oversight Working Group Report (refer to appendix G to this report) provides a detailed analysis of options for oversight management. Although this analysis was developed in the context of a DO oversight model, all of the principles are equally applicable in an ODA environment. Refer to section 3.3 of this report for more information about the ARC's recommended future oversight model.

The ARC recommends the oversight implementation include three major transition steps:

- 1. <u>Proof of Concept—Pre-Implementation: Ensure through proof-of-concept plans that the requirements proposed by the ARC are practical, effective, and efficient. Determine if the transition from "mature ODA" to DO has benefits to the FAA and industry.</u>
- 2. <u>FAA Transition Plan Transition Principle:</u> The FAA should not release a final rule before it has demonstrated the necessary cultural shift to perform system oversight. To achieve a cultural shift, policy and organizational changes may be required.
- 3. <u>Industry Transition Plan: The organization must establish the systems required of an approved organization while still working as a non-certificated applicant or a delegated organization. Applicants working toward becoming a DO demonstrate compliance to those requirements on an "as ready" basis.</u>

10.2 ACCOUNTABILITY FRAMEWORK/APPLICANT SHOWING COMPLIANCE ASSURANCE

10.2.1 INDUSTRY

A CAS that includes the following aspects will be a key feature of an approved organization that can determine compliance without direct FAA involvement.

10.2.1.1 Engineering Accountability for Compliance

Experience has shown that when compliance responsibility is embedded within the organizations that develop the design, resultant designs are less likely to encounter certification problems and associated delays. Engineering organizations that develop certification plans, take ownership for the certification requirements, and are accountable for certification issues are more effective in ensuring compliance.

The accountability framework begins with Congressional statutes and is applied through FAA regulations that establish clear roles and responsibilities for both the FAA and industry. This framework is largely derived from part 21 and 49 U.S.C., and addresses the roles and responsibilities of applicants, certificate holders, and the FAA. This framework includes each stakeholder's role in the certification process and continued airworthiness, as well as FAA's role in developing standards, policy, and guidance, and its enforcement responsibility.

The DO concept uses an accountability framework as a foundation that clearly distinguishes the roles and responsibilities of the FAA and industry. Currently, applicants lacking certification experience, as well as the use of numerous FAA designees by many companies, sometimes results in a blurred distinction between the showing of compliance by industry and the finding of compliance by the FAA. The FAA and industry must be able to strengthen and understand each of their roles in the aviation system for the DO vision to work. Figure 5 below summarizes the accountability framework.

	Issuing regulations.
The FAA promotes aviation safety by:	• Specifying the certification basis consistent with issued regulations.
	• Providing guidance regarding acceptable means of compliance.
	Overseeing compliance.
	• Taking enforcement actions as necessary.
	• Issuing certificates and approvals.
	• Mandating corrective action as necessary.
Applicants for a design approval have a	• Use means of compliance acceptable to the FAA.
regulatory obligation to:	• Show their designs are compliant.
Applicants for a production approval have a	• Establish a fabrication inspection system or a quality control system.
regulatory obligation to:	• Demonstrate they can produce products that meet the approved design.
DAHs have an ongoing regulatory	Maintain compliant designs with no unsafe feature.
obligation to:	• Report all known failures, malfunctions, and defects for their products.

Figure 5. Accountability Framework

10.2.1.2 DAH Manual

A required DAH manual will identify all processes the DAH will used to perform certification work and meet the requirements for a DAH. This manual may ultimately serve as the manual for the DO but does not need to have requirements unique to a DO included until needed to support a DO application. The transition period should be used to ensure the DAH manual requirements are sufficient to satisfy DO requirements for certification and compliance assurance or any other processes applicable to the DAH.

The ARC's DO Working Group provided guidance on what the DAH manual should contain. A summary can be found in section IV(C)(1) of the DO Working Group Report (refer to appendix E to this report). The 14 CFR Part 23 Reorganization ARC also provided recommendations on manual content based on AS 9100 requirements that could satisfy the DAH manual requirements. A manual template using this process is available from the part 23 activity.

Only top-level requirements should be in the DAH manual. However, these requirements should identify lower-level processes and work instructions in separate documents that satisfy top-level requirements. The FAA may review the DAH manual and any referenced documents as needed to audit the DAH.

The ODA manual focuses on requirements for the holder to act like the FAA in finding compliance. The DAH manual must focus on CAS processes for determining compliance with regulations and processes for self-surveillance.

Many companies planning to become DOs already have many of the processes necessary to satisfy the DO manual requirements in a condition that is acceptable or, with relatively minor changes, could be acceptable. Using existing documented processes "as is" or with minor revisions will minimize the amount of effort required to satisfy the DO manual requirements.

In addition, the DO Working Group recommended establishing an OCS, DMS, and CCS. These are described in more detail in the working group's report, included as appendix E to this report.

The ARC recommends further discussion on the necessary level of detail to include in the procedures manual and the appropriate reasons/rationale for FAA requests for changes to the procedures manual.

10.2.1.3 Applicant Showing Process and FAA LOPI

The FAA has generally not promulgated regulations governing the substantiation or "showing" of compliance, but has focused instead on regulating the "finding." Under the accountability framework, the applicant is responsible for ensuring the substantiation data is correct (showing compliance), and the FAA decides what inspections and tests it will witness or conduct and what data it will review. This establishes the FAA LOPI. FAA involvement is not required for the applicant to show compliance and fulfill their regulatory obligations. Using this discretionary function authorized by statute, the FAA may choose to rely on an applicant showing and statement of compliance instead of making a discrete finding of compliance. For more on the accountability framework and applicant showing, refer to appendixes N and O to this report, respectively.

The accountability framework requires applicants to document the determination of compliance in a way that would support an understanding of how compliance was demonstrated in a subsequent audit or investigation. Moreover, it requires the identification of auditable processes that ensure the substantiation data will be consistently developed in an acceptable manner. The accountability framework requires applicants and DAHs to take responsibility for ensuring compliance. An organization authorized to perform applicant showing may use ODA UMs or DERs to validate compliance determinations, but when doing so, it must be clear that they are acting on behalf of the applicant or holder, not the Administrator.





The current state is one of delegation and very limited applicant showing application. Oversight is compliance-based, meaning every "show" has a "find" from the FAA or a delegate (ODA or DER). Expanding the applicant showing system would result in an increase in required FAA resources on a qualitative basis. This is driven by the requirement that the FAA would need to review each "show" process to ensure the determination could be left to the applicant for a CBO system. The solution to this is a movement to a PBO system. Applicants rely on their SMS umbrella to drive assurance of their show processes. FAA resources are then used for surveillance of the applicants' SMS instead of individual applicant showing processes.

10.2.1.4 Compliance Library

An essential part of an applicant showing system is a collection of FAA-approved best practice methods of compliance applicable to the design. Industry has made various efforts to develop and document these methods with limited success. In some cases, the methods were developed to be so specific that they had only limited applicability. In other cases, applicants relied on a pattern or history of substantiation without definitive documentation of the methods.

A compliance library would include methods of compliance that lead an applicant from the regulation to an FAA-accepted compliance determination using a documented process. The methods could make use of ACs, FAA policy, or ASTM, SAE, RTCA, or other FAA-recognized consensus standards to determine compliance following a prescribed process. To maximize the opportunity for applicant showing and minimize the FAA LOPI, the applicant should have, or have access to, a compliance library containing an FAA-approved means of compliance to all of the regulations for which the applicant would normally be expected to determine compliance. ASTM means of compliance consensus standards being developed with the FAA and other CAAs for use with part 23 aircraft would be an acceptable compliance library as applicable.

A possible approach to creating the initial version of the compliance library has the applicant identifying all of the regulations they wish to have included in their library, then identifying one of their recent compliance reports that successfully showed compliance and is FAA-approved. This could potentially be used as an interim means of compliance until the applicant has the time to put the means of compliance process in an on-purpose format for the library. To ensure this is not continued indefinitely, a limit could be placed on the number of times its use would be allowed, or a conversion date could be set.

10.2.2 FAA

10.2.2.1 Comprehensive Plan to Accept Applicant Showing Only

Recommendation 1a—Systems Approach to Certification – Voluntary Initiatives: The ARC recommends that FAA issue policy and guidance to promote the understanding of the accountability framework as a basis for a systems approach to certification and facilitate voluntary approaches to implementation through FAA recognition and acceptance of applicant enhanced showings.

One of the limitations on implementing applicant showing has been lack of instruction to FAA personnel in applying and overseeing the concept. The FAA should publish an order providing instruction to personnel on the appropriate (risk-based) uses of applicant showing, the expectations for accountability framework, and any special oversight considerations unique to applicant showing processes. Because applicant showing is applied in low-risk areas, the oversight should be minimal.

10.2.3 FAA AND INDUSTRY (METRICS)

The FAA and industry should work together to define metrics to measure the effectiveness of the accountability framework and applicant showing in focusing FAA and industry resources on safety-critical areas. The metrics should be able to show where systemic issues may exist across multiple DOs and where there may be isolated issues. It should also show where the company statement of compliance is working well and does not require attention. Use of these metrics would help identify when an organization is ready for DO, and after achieving DO certification, how well it is performing.

The ARC recommends the FAA consider the DO model's impact on the existing accountability framework, particularly regarding how a design approval applicant, DAH, and DO are related.

10.3 SYSTEMS ENGINEERING

Systems engineering can be defined as an interdisciplinary, collaborative approach that derives, evolves, and verifies a life-cycle balanced system solution that satisfies customer expectations and meets public acceptability.¹⁵

¹⁵ IEEE P1220, Standard for Application and Management of the Systems Engineering Process, [Final Draft], September 26, 1994).

The FAA and industry are beginning to embrace the concept of systems engineering. The SEMP concept (refer to section 10.1.5.1 of this report) can be applied to technical planning documents used in a variety of industries to develop, oversee, and maintain high-risk and geographically dispersed programs and processes. The SEMP is designed to be an integrated planning document for the conduct and management of a specific effort, project, program, or set of processes.

Systems engineering is the field that enables systemic self-surveillance and collaborative oversight to occur. These are two key components of the systems engineering field. These two components are key pieces of aircraft certification that will allow the industry to continue on its current growth path and the FAA to perform oversight at a manageable level.

10.3.1 INDUSTRY

10.3.1.1 Systemic Self-Surveillance

Internal systemic self-surveillance can play a vital role in obtaining and maintaining a healthy compliance system. Quality management is a significant player in the systemic self-surveillance piece of systems engineering.

Systemic self-surveillance can be achieved through quality process management, performing internal audits, regular program and process reviews, continuous risk management, enhanced internal communication, and employee reporting programs. Employee reporting programs are noteworthy in that the employee must be open to report and identify risks and issues without fear of retribution.

Errors, variability, omissions, and other process problems cost time, program resources, and lives. It is the employee's responsibility to know how the quality of the process affects their projects and related processes to achieve an optimal level of process quality by encouraging best practices. A self-correcting feedback loop should be built into the systems engineering process starting from the conceptual phase throughout the life cycle of the system. The system's life cycle remains a continuous feedback loop.

These important areas are not the only components of a systemic self-surveillance system, but contribute to systems-managed processes and assist the FAA in understanding the overall health of the organization. Under a DO, establishing a QMS working in conjunction with the CAS will provide systemic self-surveillance and corrective action for the CAS processes.

10.3.1.2 Commitment to Continuous Improvement

A strong commitment to continuous improvement is an important indicator that an organization has a healthy system. An organization where employees may report risks or issues without fear of retribution is more likely to meet requirements and process and system goals. Continuous improvement can be illustrated in many ways; however, employee participation is a primary component. A company's trust in its employees, and employee trust in management, will build strong relationships internally, enhance communication, and assist in identifying potentially costly risks at an earlier stage by following defined processes and openness to report problems as soon as they are realized. Additionally, training and staying up on new technology provides an open forum for questions and answers. This assists with internal communication and openness to report risk identified in the process.

10.3.2 FAA AND INDUSTRY (METRICS)

The FAA and industry should work together to define metrics to measure the effectiveness of the efforts to shift to a more systemic approach to oversight.

10.4 OPTIMUM USE OF ODA

10.4.1 INDUSTRY

10.4.1.1 ODA Optimization

To reach a point at which the FAA and industry can effectively implement a DO model, it is important to continue optimizing the current ODA delegation system. As discussed in section 1.4 of this report, numerous studies have found a need to shift FAA certification processes from a detailed product approach to a systems safety approach. This shift is highlighted in the ACPRR ARC report as a key recommendation. The FAA and industry should continue to implement the recommendations of the ACPRR ARC and Consistency of Regulatory Interpretation ("section 313") ARC activities and work to improve the efficiency and effectiveness of current certification processes.

10.4.2 FAA

10.4.2.1 Full ODA Minus Limitations

FAA Order 8100.15 states, "The OMT may impose any limitations on an ODA holder's authority, as warranted by the ODA holder's staffing and experience, that the OMT determines appropriate. The OMT must limit the authority based on the qualifications and capabilities of the ODA UMs. The OMT may, for example, retain authority for the approval of test plans, requiring them to be submitted for approval by the ACO." However, the FAA often retains compliance findings in areas that do not have documented limitations.

Under a more systemic approach, any time a delegation is withheld, the FAA would provide a written reason for withholding delegation. An ODA that has no documented limitations would enjoy full ODA authority, and have delegation to perform all activities with the exception of those defined as inherently governmental.

10.4.2.2 Improve the Issue Paper Process to Reduce Governmental Activity

In revision B to FAA Order 8100.15, the FAA added a requirement that the "PNL response must identify the rationale for all [FAA] specific findings and reasons for any other FAA participation." Preliminary data gathered from program notification letter (PNL) responses indicates a large majority of FAA involvement in ODA activity is associated with items considered inherently governmental. Further analysis of this data indicates much of this activity could be reduced or eliminated with a more strategic approach to the development of MOCs, ELOS, and special condition issue papers. Because many of these issue papers are developed for specific installations, subtle variations in the design often require developing a new issue paper. This results in long lead times, project delays, and additional FAA work statements for issues where the compliance path has already been established. The FAA should provide instructions to the responsible directorates that will increase the usability of issue papers, thereby reducing the volume of inherently governmental activity. Such instruction would include, but not be limited to, multi-use statements, bounding design variables (for example, covering a range of seat pitch and cant angles), and multi-model issue papers.

10.4.3 FAA AND INDUSTRY (METRICS)

The FAA and industry should work together to define metrics to measure the effectiveness of the efforts to optimize the usage of ODA capability.

10.5 BILATERAL CONSIDERATIONS FOR PRODUCTS AND ARTICLES CERTIFICATION

The ARC has acknowledged that its recommendations will impact the bilateral agreements pertaining to airworthiness and environmental certification, which the U.S. Government has developed and agreed with its international partners to ease the export of U.S. products and the import of non-U.S. products.

The foundations of the bilateral agreements on airworthiness and environmental certification rely on a mutual trust between the FAA and its bilateral CAAs' partners that their respective certification systems provide equivalent results. These agreements may be based on systems that use different certification processes and procedures. What matters is not the way the certification of the products or articles is performed but that the bilateral partners mutually accept that their systems produce equivalent results. This mutual trust in each other's systems relies on effective communication between the authorities that will keep each other informed of any changes in their certification systems (such as their statutory responsibilities, organizational structure, airworthiness and environmental standards and procedures, production quality control system oversight, or delegated or contracted functions).

These bilateral agreements provide the ground for reciprocal acceptance of compliance demonstrations, findings, and approvals through a "type validation" process. This process is typically developed into technical implementation procedures, which specify the principles by which the validation authority accepts the compliance demonstrations, findings, and approvals made by the prime certification authority system. These procedures also define the validation items, which are the compliance demonstrations items of particular interest to the validation authority and for which its involvement will normally go beyond the familiarization process. Those validation items must be justified, for example by technical differences in the standards. Except for those justified validation items, the key concept of the validation process is that the validation authority should depend on the compliance determination made by the certification authority system to the maximum extent possible.

Those bilateral agreements are of primary importance for the industry because they streamline the importation and exportation of products and articles between the bilateral partners and reduce the cost of certification while producing an equivalent level of safety. They are also very effective for the CAAs of the importing countries, which can focus their involvement on the limited significant standards differences and otherwise rely on the compliance determinations of the exporting country's certification system for all the other compliance items. The ARC recognizes the significant changes introduced by the DO model and its associated FAA-industry accountability framework will have an impact on the bilateral agreements and their associated technical implementation procedure. Under the terms of these agreements, the FAA will need to communicate to its bilateral partners about those changes in its certification system and to implement the necessary changes in the bilateral documentation.

Considering the importance of the validation process for the industry and authorities, the ARC recommends the following high-level objectives be considered in this task of reviewing the bilateral documentation:

- Validation by the FAA bilateral partners of U.S. products and articles certified using a LOPI process: The certification of products and articles under a LOPI approach will be considered equivalent to the current system and should not lead to an increase of the validation items by the validation authorities.
- Validation by the FAA of non-U.S. products and articles certified under an FAA bilateral partner deemed as equivalent to the FAA certification process: The involvement of the FAA in the validation activities should, in most cases, be limited to issues that are risk-based and meet the requirements for LOPI for the validation items as defined in the bilateral technical implementation procedures. Under this principle, validation activities will no longer include retained findings of compliance.

10.5.1 BILATERAL AGREEMENT DO CONSIDERATIONS FOR PRODUCTS AND ARTICLES CERTIFICATION

One privilege of a CDO is the authority to make determinations of compliance, and present the Administrator with a statement of compliance the FAA may rely on to issue a certificate.

Principle of Reciprocal Acceptance. Current U.S. bilateral airworthiness agreements with other states contain the clause that each party recognizes and accepts the other party's safety oversight and regulatory system. That principle of reciprocal acceptance has allowed one authority to act on behalf of the other in making compliance determinations, and performing other functions as defined within the bilateral agreement.

This principle also includes accepting each other's system of delegation, if applicable. As a matter of policy, the FAA has decided to use its delegation system in performing specific functions or when making compliance determinations to foreign airworthiness requirements when requested under a bilateral agreement.

Bilateral reciprocal acceptance also has enabled "approved" data to be used internationally to facilitate CAA-approval of manuals, repairs, and modifications. This has considerably reduced the compliance finding burden on the FAA and the schedule burden on the industry during type validation programs.

Need for Bilateral Changes Relevant to DOs. Because the ARC's future vision will be a new organizational approval system, the FAA is obligated to notify its bilateral partners under the terms of the existing bilateral agreements. These authorities have the right to evaluate the new FAA system and determine whether it meets the intent of the bilateral agreement and can be accepted. It is envisioned that importing (validating) authorities will rely on the exporting

(domestic) authority's overall system for aeronautical product design, certification, and production. Specifically, the importing (validating) authority would—

- Evaluate the state's system requirements for regulatory compliance, including SMS and system oversight.
- Mutually recognize the two systems as equivalent, or define where there are differences and address them with the bilateral partner.

As it cannot be assumed that a new FAA organizational approval system will be accepted internationally, the FAA must engage with other CAAs through early and regular communication of the FAA's DO concept to help gain international acceptance.

11.0 CERTIFICATION COST/BENEFIT OVERVIEW

Historically, aviation safety rulemaking cost-benefit analysis (CBA) has focused on preventing of accidents and lives lost as the benefit justification to offset the costs of implementing new rules. The ARC's purpose in evaluating the proposed transition to a systems-based approach to new rules was not to create a rule that relies on accident prevention as a justification, but rather a rule that can be justified and supported by enhanced improvements to effectiveness and efficiency for both the FAA and industry. As a result, the ARC's Cost-Benefit Analysis Working Group was tasked to find new ways of capturing the benefits and costs associated with such improvements and identify a supporting methodology.

The working group found that overall change is needed for both the FAA and industry. This conclusion is a combination of statistical data, quantitative and qualitative analysis, and assumptions gathered by the working group. There appears to be a uniform concern that the current certification system is not sustainable given industry growth and the FAA change in resources and budgetary constraints of recent years. The working group could not, within its limited scope, collectively conclude that a DO with the inclusion of SMS is the appropriate change; however, it is increasingly apparent that a major change to the current certification system is needed.

The constraints and advantages of the working group led to the following findings:

• The cost vs. benefit of implementing SMS without a DO must be further researched. Given the late decision by the ARC's Organization Working Group to recommend DO as optional, the CBA Working Group did not have sufficient time to gather data on SMS independent of a DO.

Regarding the data gathered from small organizations pertaining to implementing mandatory DO with SMS, a scaled DO (AO), or the modified current model (refer to section 7 of this report), the CBA Working Group made the following key findings:

- The 2013 MARPA Conference was extremely useful because small business does not always have the resources to participate full time on ARCs. Holding a workshop and distributing a survey during the MARPA conference proved small business has a strong interest in proposed changes to certification. Additionally, there is a common feeling that a change to the current system is needed.
- The resulting average estimated cost increases of adopting mandatory DO with SMS, an AO, or the modified current model were approximately
 - o 15 percent for DO with SMS,
 - 20 percent for an AO, and
 - Modified Current Model was approximately 8.3 percent for the modified current model.
- The data also revealed that the resulting average estimated increase in revenue from immediate project initiation (no sequencing queue delay) was approximately 15 percent. This is a significant benefit that may compensate for cost of a DO with SMS or an AO.

• Respondents estimated the effect on revenues if their company was not required to wait in the sequencing or project prioritization queue to initiate a project. The purpose of the question was to gauge the perceived benefit of privileges associated with both the mandatory DO and AO models. The resulting average estimated time savings as a result of immediate project initiation (no sequencing queue) was approximately 4 months.

Regarding the data gathered from large organizations pertaining to implementing mandatory DO with SMS, the working group made the following key findings:

- The large company survey responses considered a mandatory DO system to be far from cost-beneficial. Two companies provided no cost-benefit estimates, stating estimates were "difficult to estimate at this time" or "too premature to determine." One company's responses were large outliers, so just four useful responses were received. This gave a median response of approximately 2 to 1, costs to benefits.
- Discussion with ARC company representatives suggests possible reasons for this result are that the large company representatives believe—
 - The DO system provides no additional benefits beyond what is provided by the ODA, or that will be provided when ODA is "fully matured;"
 - o The DO proposal is premature because ODA has not fully matured; or
 - The DO system is far from cost beneficial now, but could be cost beneficial after other companies have adopted it and worked out issues with implementation.
- Additionally, informal discussions with company representatives after the survey was completed indicated the companies did not sufficiently focus on their potential benefits in responding to the survey. This finding suggests greater attention must be given to potential benefits in future surveys.

Additional key findings of the working include the following:

- Manufacturers had difficulty articulating baseline certification costs. Part 21 is a process-oriented rule, which made for a very broad and difficult rule to cost out. The majority of participants calculated cost in a more general format and described it mostly in percentages.
- Large and small manufacturers had difficulty articulating benefits. Although participants understood the benefits from a conceptual level, most were unable to express benefits in monetary values. Therefore, a benefits survey is strongly encouraged as a follow-on activity to this ARC and to future rulemaking efforts.

Lastly, in the event a formal rulemaking project takes place, the formal cost-benefit analysis should consider the following recommendations in addition to the traditional process:

• Consider a separate survey to gather benefit data. A key finding was that respondents had significant difficulty articulating benefits. Gathering data from multiple benefit questions can provide adequate data to calculate efficiencies.

- Relying on the industry to calculate this information is not a viable solution. In the case of SMS and changes to part 21, process-oriented requirements are far too broad to calculate in one question.
- When benefits were described in the form of a question, respondents were better able to estimate cost savings.
- Efficiencies can be calculated through a number of equations commonly used in industry and by other agencies. The most simple of these includes the following:

Efficiency = Expense/Revenue

This is only one example of a possible solution to calculating efficiencies in typical business operations.

- Efficiencies must be broken out into tangible items whenever possible. In areas where this is not possible, a qualitative assessment is acceptable to the Office of Management and Budget (OMB), according to OMB Circular A–4, Regulatory Analysis.
- The working group did not examine SMS independent of a DO; however, a common response by the industry was to consider the following:
- Identify industry cost of maintaining and satisfying multiple SMSs *compared with* single SMS accepted internationally. The industry expressed substantial concern with developing and maintaining a SMS for multiple countries. The concern related to cost of development, implementation, maintenance, and accommodation of external audits. This is a specific area of concern and should be further researched.

The CBA Working Group took a proactive approach to identify cost and benefits at a preliminary stage. The methodology applied focused on collecting real data and active interaction with the parties impacted, including small and large business and the FAA. Data gathered by the working group is *supporting data only* and does *not* represent a formal economic analysis. All data gathered by the working group has been shared with the FAA's Office of Aviation Policy and Plans Economic Analysis Division (APO–300) with the intent to serve as supporting data, and may be referenced during the formal economic analysis in the event a rulemaking project takes place. The CBA Working Group Report located in appendix J to this report contains the details of the methodology as well as results from all data gathered.

12.0 DISSENTING OPINIONS

All of the ARC members were offered the opportunity to present a dissenting opinion on the ARC's decisions and recommendations. All dissenting opinions are collected in this section with a reference back to the section they apply to.

Dissenting Opinion—Approved Data

The following dissenting opinion to Approved Data in section 6.8 was submitted by a member of the ARC's DO Working Group from General Electric:

A member of the working group does not agree with the opinion expressed above, related to data considered to be approved by the FAA. A working group member agrees that the type design, as defined in § 21.31, is not only found to comply to applicable airworthiness requirements but is considered to be approved by the FAA, with the issuance of a design approval. Thus, the issue is with the FAA desire to not call the substantiating data "approved data."

While it could be argued from the above discussions that there is nothing changed about the data itself by not calling it "approved," an ARC member fails to see the need for the FAA to change several orders and reeducate the entire international aviation community on why it is not calling substantiating data approved, when in fact the character of the data has not change.

For many decades the FAA has called that "found-to-comply" data "approved data." The words "approved data" have been common place within the U.S. aviation system and have been recognized as having significance by international authorities. Any change to this entrenched concept in U.S. aviation, and the enormous task of reeducating FAA employees, U.S. industry, ICAO, and other airworthiness authorities, should only be undertaken after the FAA has presented a good cause argument for that change. In all of the discussions of this issue, FAA has never defined the problem they are trying to correct or the new message they are trying to convey by no longer allowing substantiating data to be called "approved data."

Dissenting Opinion—Correcting Non-Compliances

This dissenting opinion was expressed by NATCA during an ARC meeting and will not be found elsewhere in the report.

NATCA recommends the DO regulations, either part 5 SMS or part 21, include a requirement to report and provide a corrective action plan for non-compliances to the airworthiness regulations that are discovered by the DO. This would be similar to the current reporting requirement of 14 CFR 183.63(b)(2) and the associated procedures required for ODA in Order 8100.15 for production products.

Section 5.1.1: NATCA does not have a position on the portion of the recommendation that part 5 be revised from the NPRM proposal based on the Design and Manufacturing sector comments. NATCA did not review the NPRM because it was titled "Safety Management Systems for part 121 Certificate Holders" when it was published for comments on November 5, 2010. The title of the rulemaking did not include any indication it would apply to parts of 14 CFR other than part 121 air carrier operations. The final rule is now at OMB for review, so providing comments to the FAA through this ARC, which is outside the NPRM docket, might be considered a violation of DOT ex parte policy. NATCA will therefore review the final rule and consider submitting comments to the final rule docket or to the new rulemaking docket if FAA publishes a new NPRM that would apply SMS requirements in part 5 to part 21.

Note, FAA submitted a summary of an ex part meeting with industry on the NPRM docket number FAA-2009-0671, item FAA-2009--671-0136, titled "Summary of an Safety Management System Aviation Rulemaking Committee Meeting held on November 16, 2010." It includes the following explanation that part 5 will be reopened for comment should a NPRM be issued to apply part 5 SMS requirements to other parts, including part 21:

"The docket item includes the following: "First, the ARC members were concerned that the title of the NPRM would not adequately notify others, such as repair stations, part 135 operators, and aircraft design and manufacturers, who may be impacted by this rulemaking in the future through expansion of the applicability requirements of part 5. The FAA noted that the NPRM states that although the proposed rule would only apply to 14 CFR part 121 certificate holders, the FAA may consider applying the part 5 SMS requirements to parts 135, 145, and 21, as appropriate. In the event that the FAA would extend the general requirements to these populations, the FAA would initiate rulemaking and these populations would have an opportunity to comment on the requirements and the applicability through future NPRMs."

Section 6.1 and section 10.1.5.1: NATCA recommended the qualification requirements for staff show and verify compliance to the airworthiness regulations should be addressed prior to DO rulemaking activity, and that consideration should include certification or licensing.

Section 6.7.9, Environmental Compliance Determinations: NATCA does not agree with the statement in the recommendation that "*the process-based approach to compliance, as established by DO program principles, is far more robust than traditional delegation process.*" However, NATCA understands that FAA is moving toward a DO process. Given a DO process will be utilized in the future, NATCA does not see a reason it should not also be applied to 14 CFR parts 34 and 36.

Section 8.1, Changes to 14 CFR 21.21: NATCA disagrees with the proposal to replace 21.21(a) and (b) with the proposed text. The existing 21.21(a) provides for type certification of surplus aircraft from the U.S. Armed Forces. The existing 21.21(b) provides requirements that an applicant (1) provides data to show the product meets the applicable sections of subchapter C and (2) that "*For an aircraft, that no feature or characteristic makes it unsafe for the category in which certification is requested.*" These existing requirements of 14 CFR 21.21 need to remain and would apply equally to DOs, although an alternate provision to 21.21(b)(1) could be developed that applied more directly to DOs.

Page intentionally left blank.

APPENDIX A—PART 21/SMS ARC MEMBERSHIP

MEMBERS AND ALTERNATES

- Mr. Walter Desrosier, Co-Chair, GAMA
- Mr. Michael Reinert, Co-Chair, FAA AIR-150
- Mr. Rick Baggette, Boeing
- Mr. Glenn Baxter, Bombardier
- Mr. Edmond Boullay, Alternate, ASD/US Crest
- Mr. Rafael Borges, Observer, ANAC
- Mr. John Bouma, Cessna
- Mr. Dave Chapel, GE
- Ms. Maria Clara, Observer, ANAC
- Mr. Michael Collins, NATCA
- Mr. Robert Cook, FAA AIR–200
- Mr. Peter Corbeel, Observer, EASA
- Ms. Doris Costa, Observer, ANAC
- Ms. Jodi Diamant-Boustead, Alternate, AIAC/Pratt Canada
- Mr. Jason Dickstein, MARPA
- Mr. Tomaso DiPaolo, Alternate, NATCA
- Mr. Chris Eick, Honeywell
- Mr. Jimmy Eyl, Banyan Air Service
- Mr. Robert Ferguson, Observer, TCCA
- Ms. Amy Garzaro, FAA AIR-150
- Mr. Scott Geddie, FAA AIR-110
- Mr. Paul Greer, FAA AGC
- Mr. Julian Hall, Observer, EASA

- Mr. David Hempe, FAA AIR-100
- Ms. Katrina Holiday, FAA ARM
- Mr. Charles Huber, FAA SMS/ANM-100
- Mr. Daniel Leach, FAA APO
- Mr. Eric Lesage, Airbus
- Mr. Michael Linegang, FAA AIR–110
- Mr. Mark Lopez, Alternate, A4A
- Mr. Stacey Mason, Observer, TCCA
- Ms. Linda Navarro, Observer, FAA AIR-150
- Mr. George Novak, AIA
- Ms. Jan Novak, Observer, EASA
- Mr. Ric Peri, AEA
- Mr. Dennis Piotrowski, BELAC
- Mr. Tom Rogozinski, Honeywell
- Mr. Dan Shapiro, Sikorsky General
- Mr. Eric Sivel, Observer, EASA
- Mr. Roger Southgate, Rockwell Collins
- Mr. Giles Strickler, FAA ARM
- Mr. Marcus Tittiger, Observer, TCCA
- Mr. Scott VanBuren, Observer, FAA AVP (SMS)
- Mr. Mark Watton, Delta Airlines
- Mr. Bill Whitton, Gulfstream

PROGRAM SUPPORT

Mr. Larry Van Dyke, GAMA

APPENDIX B—GLOSSARY

Term	Definition	Source
Acceptable Risk	The level of risk that individuals or groups are willing to accept given the benefits gained. Each organization will have its own acceptable risk level, which is derived from its legal and regulatory compliance responsibilities, its threat profile, and its business/organizational drivers and impacts.	AVS Order 8000.367A
Accident	An unplanned event or series of events that results in death, injury, or damage to, or loss of, equipment or property.	AVS Order 8000.367A FAA Order 8040.4A
Accountable Manager	(a) Accountable manager means the person designated by an applicant or design approval holder who is responsible for and has the authority over all design approval operations that are conducted under part 21, including ensuring that design approval holder personnel follow the regulations and serving as the primary contact with the FAA.	14 CFR Part 145.3(a) modified for use by DAH
Accountability Framework	An established set of responsibilities and commitments of the FAA and industry	Refer to appendix O to this report.
Aerospace System	U.S. airspace, all manned and unmanned vehicles operating in that airspace, all U.S. aviation operators, airports, airfields, air navigation services, pilots, regulations, policies, procedures, facilities, equipment, and all aviation-related industry.	AVS Order 8000.367A FAA Order 8040.4A
Aircraft Accident	An occurrence associated with the operation of an aircraft that takes place between the time any person boards the aircraft with the intention of flight and all such persons have disembarked, and in which any person suffers death or serious injury, or in which the aircraft receives substantial damage.	49 CFR 830.2
Analysis	The process of identifying a question or issue to be addressed, examining the issue, investigating the results, interpreting the results, and possibly making a recommendation. Analysis typically involves using scientific or mathematical methods for evaluation.	FAA Order 8040.4A
Applicant Show With Capability (ASOC)	Based on the Oversight Working Group's model of DOs getting recognized for demonstrated capabilities.	

Term	Definition	Source
Approved Data	Data approved by FAA employees, its designees, or a DO acting under the authority of its certificate.	
Assessment	Process of measuring or judging the value or level of something.	FAA Order 8040.4A
Certificate Surveillance	FAA actions to monitor the DO certificate holder and to determine the holder's compliance with the provisions of its certificate. Note: In the Oversight section we discuss managing these organizations through surveillance.	
Compliance Assurance System (CAS)	DO holder's system for ensuring that it complies with the applicable regulations.	
Compliance Finding	FAA decision (either directly or through a designee) that compliance has been shown with the applicable regulatory requirements.	
Control	Refer to Safety Risk Control.	AVS Order 8000.367A
Corrective Action	Action to eliminate or mitigate the cause or reduce the effects of a detected nonconformity or other undesirable situation.	AVS Order 8000.367A
	An action required to be taken by the DO to address noncompliances and problems with the organization's procedures or performance.	
	The noncompliances may result from—	
	• Internal audits conducted by the DO,	
	• FAA surveillance,	
	• DO employee observations, and	
	Voluntary disclosures.	
Culture of Compliance	Knowledge, beliefs, attitudes, and behaviors of an organization that are focused on ensuring regulatory compliance with all its activities.	

Term	Definition	Source
Descriptive	From the DO Working Group:	
Data	• Data that defines the type design that needs to be determined compliant to the applicable airworthiness standards. The descriptive data is what is approved by the FAA when a design approval certificate is issued.	
	• The drawings and specifications necessary to define the configuration shown to comply.	
Determination of Compliance	A decision made by the certificate holder that compliance has been shown with the applicable regulatory requirements. [Note: The ARC has referred to "regulatory requirements" rather than just "airworthiness standards" because its recommendation that DOs eventually include determination of compliance with other 14 CFR parts, such as parts 26, 34, and 36.] It may also be a decision made by the certificate holder that data previously approve by the FAA or data determined to comply by another CAA under the provisions of a bilateral airworthiness agreement between the United States and a foreign country or jurisdiction, are valid and applicable to the design of the product, part, or appliance for which it is to be used, including the applicable certification or approval basis.	
DO Executive	The company individual directly responsible for ensuring that the DO meets all of its regulatory responsibilities.	
DO Point(s) of Contact	The individual(s) within the DO responsible for all communications with the FAA.	
Eligible Data	Data developed under an approved DO system, assuming a specified, but not FAA-established, certification basis, and product type design if appropriate.	

Term	Definition	Source
Enforcement	An action taken by the FAA most appropriate to promote safety and compliance with the statutory and regulatory requirements. The program provides a wide range of options for addressing noncompliance:	14 CFR Part 13
	 Educational and remedial training efforts, Administrative action in the form of either a warning notice or letter of correction, 	
	 Certificate suspensions for a fixed period of time, Civil penalties, 	
	 Indefinite certificate suspensions pending compliance or demonstration of qualifications, 	
	Certificate revocations,	
	• Injunctions, and	
	Referrals for criminal prosecution.	
Evaluation	Determining the adequacy and effectiveness of an organization through a review of organizational policies, procedures, and systems.	
FAA Oversight Team	FAA personnel assigned to provide guidance and oversight of the DO in meeting its regulatory requirements.	
Hazard	A condition that could foreseeably cause or contribute to an accident.	AVS Order 8000.367A FAA Order 8040.4A
Inspection	A formal systematic and independent review of organizational policies, procedures, and systems.	
Interoperability	The ability for each SMS to be part of the system or systems through interdependent processes and/or components with shared principles, information, and governance.	AVS Order 8000.367A

Term	Definition	Source
Level of Project Involvement (LOPI)	The interactive process that the DO shares with its assigned Aircraft Certification Office (ACO) for specific engineering/design elements and with the Manufacturing Inspection District Office (MIDO) for specific production elements during certification projects. The criteria/factors influencing the decision of when to be involved will include but is not limited to Governmental functions, such as—	
	 Novel or unusual features which may require issuance of Special Conditions, Significant issues which may require Issue Papers, and Defining Equivalent Level(s) of Safety 	
	(ELOS).	
Likelihood	The estimated probability or frequency, in quantitative or qualitative terms, of a hazard's effect or outcome.	AVS Order 8000.367A FAA Order 8040.4A
Management System	See the definition for System.	
Methods (or) Means of Compliance	 Need a definition from DO Working Group Notes: Method: Process Means: Capability 	
Mitigation	A means to reduce the risk of a hazard. Refer to Safety Risk Control.	AVS Order 8000.367A FAA Order 8040.4A
Nonconformity	Non-fulfillment of a requirement. This includes but is not limited to noncompliance with Federal regulations. It also includes an organization's requirements, policies, and procedures as well as requirements of safety risk controls developed by the organization.	AVS Order 8000.367A

Term	Definition	Source
Novel or Unusual	"The phrase "novel or unusual" as used in 14 CFR 21.16 is a very relative term. As used hereafter in applying a4 CFR 21.16 to justify the issuance of special conditions, "novel or unusual" will be taken with respect to the state of technology envisaged by the applicable airworthiness standards of this subchapter. It must be recognized that in some areas which will vary from time to time the state of the regulations may somewhat lag the state of the art in new design because of the rapidity in which the state of the art is advancing in civil aeronautical design and because of the time required to develop the experience base needed by the Applicants for type certification of a new design have the opportunity to mitigate the impact of not knowing the precise airworthiness standards to be applied for "novel or unusual design features: by consulting with the FAA early in their certification planning when such features are suspected or known by the applicant to exist. It should also be recognized that, because of the intentional objective nature of the airworthiness standards of this subchapter, many new design features which might be thought of as "novel or unusual design features" may already be adequately covered by existing regulations, thus obviating the need to issue special conditions." Preamble material to 14 CFR 21.16.	
Oversight	 A systems approach to review an organization's performance, validate the development of their defined system and verify compliance to the requirements of a certified DO to determine sufficiency. Oversight activities include— Reviewing the work performed, Evaluating performance for quality assurance, Ensuring that required training has been completed, Providing constructive feedback, and Taking corrective action, including enforcement as necessary. 	

Term	Definition	Source
Procedure	A fixed, step-by-step sequence of activities or course of action (with definite start and end points) that must be followed in the same order to correctly perform a task.	
Product/Service Provider	An organization engaged in the delivery of aviation products or services.	AVS Order 8000.367A
Quality Management System	From the DO Working Group—A set of interrelated or interacting quality processes accomplished by the organization through the establishment of policy and objectives, and achieving those objectives.	
Risk	Refer to Safety Risk. The terms risk and safety risk are used synonymously.	AVS Order 8000.367A FAA Order 8040.4A
Safety	The state in which the risk of harm to persons or property damage is acceptable.	AVS Order 8000.367A FAA Order 8040.4A
Safety Assurance	Processes within the SMS that function systematically to ensure the performance and effectiveness of safety risk controls and that the organization meets or exceeds its safety objectives through the collection, analysis, and assessment of information.	AVS Order 8000.367A FAA Order 8040.4A
Safety Culture	The shared values, actions, and behaviors that demonstrate a commitment to safety over competing goals and demands.	AVS Order 8000.367A
	The product of individual and group values, attitudes, competencies, and patterns of behavior that determine the commitment to an organization's safety programs.	
Safety Management	The act of understanding and making decisions and taking actions to lower risk, inherent in all human activity, to acceptable levels.	
Safety Management System (SMS)	The formal, top-down, organization-wide approach to managing safety risk and assuring the effectiveness of safety risk controls. It includes systematic procedures, practices, and policies for the management of safety risk.	AVS Order 8000.367A
Safety Objective	A measurable goal or desirable outcome related to safety.	AVS Order 8000.367A
Safety Performance	Realized or actual safety accomplishment relative to the organization's safety objectives.	AVS Order 8000.367A

Term	Definition	Source
Safety Policy	The organization's documented commitment to safety, which defines its safety objectives and the accountabilities and responsibilities of its employees in regards to safety.	AVS Order 8000.367A
Safety Promotion	A combination of training and communication of safety information to support the implementation and operation of an SMS in an organization.	AVS Order 8000.367A
Safety Requirement	A safety condition or capability that must be met or passed by a system to satisfy a contract, standard, specification, or other formally imposed document or need.	AVS Order 8000.367A
Safety Risk	The composite of predicted severity and likelihood of the potential effect of a hazard. Initial—The predicted severity and likelihood of a hazard's effects or outcomes when it is first identified and assessed; includes the effects of preexisting risk controls in the current environment. Current—The predicted severity and likelihood at the current time. Residual—The remaining predicted severity and likelihood that exists after all selected risk control techniques have been implemented.	AVS Order 8000.367A FAA Order 8040.4A
Safety Risk Control	A means to reduce or eliminate the effects of hazards.	AVS Order 8000.367A FAA Order 8040.4A
Safety Risk Management (SRM)	A process within the SMS composed of describing the system, identifying the hazards, and analyzing, assessing, and controlling risk.	AVS Order 8000.367A FAA Order 8040.4A
Senior Company Management	Those in the company management chain above the DO Executive who are accountable for the actions of the DO.	
Severity	The consequence or impact of a hazard's effect or outcome in terms of degree of loss or harm.	AVS Order 8000.367A FAA Order 8040.4A
Showing	Determination of compliance to the Airworthiness Regulations by the applicant.	
Show/Find	The process by which the applicant "shows" how they complied with a regulation and the FAA "finds" that the applicant has adequately shown compliance to the regulation.	FAA Order 8110.4C

Term	Definition	Source
Statement of Compliance	A statement from the DO to the Administrator certifying that compliance with the applicable regulatory requirements has been determined and the procedures listed in its FAA-approved DO procedures manual have been followed.	
Substantiating Data	Documentation related to a design approval applicant's showing or compliance to the applicable airworthiness standards.	
Substitute risk	Risk unintentionally created as a consequence of safety risk control(s).	AVS Order 8000.367A
Supplier DO	A separate DO entity in its own right provides an article to an applicant/holder DO.	
Surveillance	The combination of evaluation and inspection to accomplish a review of organizational system to determine the adequacy and effectiveness of an organization.	
System	An integrated set of constituent elements that are combined in an operational or support environment to accomplish a defined objective. These elements include people, hardware, software, firmware, information, procedures, facilities, services, and other support facets.	AVS Order 8000.367A FAA Order 8040.4A
System State	An expression of the various conditions, characterized by quantities or qualities, in which a system can exist.	AVS Order 8000.367A
Validation	Validation is the process of proving that the functions, procedures, controls, and safety standards are correct and the right system is being built. i.e. the requirements are unambiguous, correct, complete, and verifiable.	
Verification	The process that ensures that the system requirements have been met by the design solution and the system is ready to be used in the operational environment for which it is intended.	

Page intentionally left blank.
APPENDIX C—ACRONYMS

AAQG	Americas Aerospace Quality Group
AC	Advisory Circular
ACO	Aircraft Certification Office
ACPRR	Aircraft Certification Process Review and Reform
AD	Airworthiness Directive
ADO	Agent Design Organization
AEE	Office of Environment and Energy
AEG	Aircraft Evaluation Group
AFS	Flight Standards Service
AIA	Aerospace Industries Association
AIR	Aircraft Certification Service
ANAC	National Civil Aviation Agency of Brazil
AO	Accredited Organization
APO-300	Office of Aviation Policy and Plans, Economic Analysis Division
ARC	Aviation Rulemaking Committee
ASA	Aviation Suppliers Association
ASE	Oversight Team of Engineers
ASI	Aviation Safety Inspector
ASTM	ASTM International
ATC	Air Traffic Control
AVS	FAA Office of Aviation Safety
BTS	Bureau of Transportation Statistics
CAA	Civil Aviation Authority (of another country)
CAAC	Civil Aviation Administration of China

CAP	Compliance Assurance Procedure
CAR	Civil Aviation Regulation
CAS	Compliance Assurance System
C.A.S.E.	Coordinating Agency for Supplier Evaluation
СВА	Cost Benefit Analysis
СВО	Compliance Based Oversight
CCS	Compliance Certification System
CDO	Certified Design Organization
CFR	Code of Federal Regulations
CMIS	Certificate Management Information System
CONOPS	Concept of Operations
COS	Continued Operational Safety
D&M	Design and Manufacturing
DAH	Design Approval Holder
DAR	Designated Airworthiness Representative
DAS	Designated Alteration Station
DDP	Declaration of Design and Performance
DER	Designated Engineering Representative
DMIR	Designated Manufacturing Inspection Representative
DMS	Design Management System
DO	Design Organization
DOA	Design Organization Approval
DPE	Designated Pilot Examiner
DPO	Design Production Organization
EASA	European Aviation Safety Agency

ELOS	Equivalent Level of Safety	
EPA	Environmental Protection Agency	
ETOPS	Extended Operations	
FAA	Federal Aviation Administration	
FOEB	Flight Operations Evaluations Board	
FSB	Flight Standardization Board	
FSDO	Flight Standards District Office	
GAO	General Accountability Office	
IAQG	International Aerospace Quality Group	
ICA	Instructions for Continued Airworthiness	
ICAO	International Civil Aviation Organization	
IEEE	Institute of Electrical and Electronics Engineers	
ISO	International Standards Organization	
LOPI	Level of Project Involvement	
MIDO	Manufacturing Inspection District Office	
MMEL	Master Minimum Equipment List	
MOC	Memorandum of Cooperation	
MRA	Maintenance, Repair, and Alteration	
MRB	Maintenance Review Board	
MSAD	Monitor Safety/Analyze Data	
MSG	Maintenance Steering Group	
NATCA	National Air Traffic Controllers Association	
NPRM	Notice of Proposed Rulemaking	
OCS	Organization Control System	
ODA	Organization Designation Authorization	

ODAR	Organizational Designated Airworthiness Representative
OIG	Office of Inspector General
OMB	Office of Management and Budget
OMT	Organizational Management Team
РАН	Production Approval Holder
PBO	Performance-Based Oversight
PI	Principal Inspector
PMA	Parts Manufacturer Approval
PNL	Project Notification Letter
РОС	Point of Contact
QMS	Quality Management System
QSA	Quality System Audit
R&D	Research and Development
RBRT	Risk-based Resource Targeting
RPM	Revenue Passenger Miles
RTCA	Radio Technical Commission for Aeronautics
RTM	Revenue Ton Miles
SA	Supplier Audit
SAE	Society of Automotive Engineers
SEMP	Systems Engineering Management Plan
SFAR	Special Federal Aviation Regulation
SM ICG	Safety Management International Collaboration Group
SMM	Safety Management Manual
SMS	Safety Management System
SRM	Safety Risk Management

SSP	Specialty Service Providers	
STC	Supplemental Type Certificate	
TC	Type Certificate	
TCCA	Transport Canada Civil Aviation	
TSO	Technical Standard Order	
TSOA	Technical Standard Order Authorization	
UAS	Unmanned Aircraft Systems	
UM	Unit Member	
U.S.C.	United States Code	
USITC	U.S. International Trade Commission	

Page intentionally left blank.

APPENDIX D—PART 21/SMS ARC CHARTER



U.S. DEPARTMENT OF TRANSPORTATION FEDERAL AVIATION ADMINISTRATION Aviation Rulemaking Committee Charter

Effective Date: 10/5/2012

SUBJ: 14 CFR 21 / Safety Management Systems Aviation Rulemaking Committee

 PURPOSE. This Charter creates the Aviation Rulemaking Committee (ARC) for Part 21 / Safety Management Systems (SMS) according to the Administrator's authority under Title 49 of the United States Code (49 U.S.C.) 106(p)(5). This charter also outlines the committee's organization, responsibilities, and tasks.

2. BACKGROUND.

On May 22, 2012, the Aircraft Certification Process Review and Reform ARC submitted a report to the FAA recommending that we undertake a review to update part 21 certification procedures to reflect a systems safety approach to product certification processes and oversight of design organizations. Design organizations must have full responsibility and accountability through the establishment of regulatory requirements for minimum qualification, performance, and management systems.

Consistent with FAA Order VS 8000.367, and the International Civil Aviation Organization (ICAO) Annex 8, the Aircraft Certification Service (AIR) has been actively developing and implementing an internal and external SMS. The initial focus was primarily on developing an internal set of processes, tools, and methodologies that facilitate the transition into the future state. AIR began that effort in 2005 and has made progress in defining key processes and tools. Later, with support from industry participants, the activities expanded to include development of standards for design and manufacturing organizations. Through implementation of pilot SMS projects with certain companies, the FAA is collecting information that will help define the scope of the SMS for Design Approval Holders (DAHs), validate certain best practices, and expand the knowledge base within the workforce and industry with respect to the essential elements of a robust SMS for manufacturers.

SMS requires a proactive approach to discovering and addressing hazards before they exhibit safety consequences. SMS also includes processes that seek to identify potential organizational breakdowns and necessary process improvements which allow management to address a safety issue before a noncompliant or unsafe condition results. SMS is not a substitute for compliance with FAA regulations or FAA oversight activities.

3. OBJECTIVES AND TASKS OF THE ARC. AIR wants to evaluate certain improvements to the effectiveness and efficiency of existing "certification procedures for products and parts," along with incorporating SMS in the design and manufacturing environment. This includes considering the effects of certain changes to the existing regulations, such as applicant qualifications, hazard (or safety) reporting, compliance assurance, and continued operation safety assurance systems for

Distribution:

Initiated By:

all DAHs. The intent is to facilitate shifting towards a systems approach for DAHs that is similar to that used for production approval holder requirements, which involves a clear understanding of roles, responsibilities, and privileges. As part of this evaluation, we want to determine the best way industry and the FAA can effectively fulfill their respective compliance and safety responsibilities.

The ARC will provide a forum for the U.S. aviation community to discuss and provide recommendations to the FAA. The committee is expected to provide general information and guidance regarding proposed changes to part 21 and the AVS SMS program as it relates to design and manufacturing certificate and approval holders.

- a. The ARC will provide the FAA recommendations, which may include proposals for rulemaking, suggested processes, policies and guidance, and any further action it determines the agency should contemplate for part 21 to align with the SMS requirements documented in proposed 14 CFR part 5, which is the central component of the NPRM entitled *Safety Management Systems for Part 121 Certificate Holders* [Docket No. FAA-2009-0671; Notice No. 10-15].
- b. The ARC, serving in an advisory capacity, is expected to present and discuss whatever input, guidance, and recommendations its members consider critical to the FAA's ultimate disposition, development, and implementation of proposed regulatory requirements and related guidance and policy as necessary to the future direction for part 21 to include applicant pre-qualifications, approval holder recognition, and SMS considerations.
- c. The ARC will also consider proposed revisions to clarify and update engineering/designoriented regulatory requirements to part 21. In support of design certification and continued airworthiness, the evaluation should include improvements in the areas of:
 - 1. Application process
 - 2. Applicant qualifications
 - 3. Standardized certification criteria
 - 4. Identifying design approval holder responsibilities and privileges
 - 5. Clarifying continued airworthiness requirements
 - 6. Clarifying design approvals needing Instructions for Continued Airworthiness
 - 7. Clarifying TSO design approval processes
 - 8. Process definition for determining eligibility of U.S. surplus military aircraft in the restricted category

This proposal additionally corrects regulatory language, implements editorial changes for clarification, and standardizes regulatory language to reflect the global aviation environment. While this information will be shared with the ARC, responses to "clean-up" proposals are not required as part of the deliverables.

d. Proposed part 5 and International Civil Aviation Organization (ICAO) Annex 8 and Annex 19 (draft) serve as the foundation for the ARC's consideration regarding how the FAA will address its responsibilities for developing and implementing SMS

requirements and the management and oversight of its regulated product/service providers. The ARC must respect the framework outlined in proposed part 5 and the ICAO Annexes when it provides the FAA recommendations with respect to application of SMS. However, the FAA will consider proposed changes to part 5 as deemed necessary from a design and manufacturing perspective.

Recommendation Report. The ARC shall make recommendations and submit a report addressing the following:

- a. Improvements, which may include proposals for rulemaking, processes, policies and guidance for 14 CFR part 21 that reflect a systems approach for safety. This will promote an effective and efficient certification process, which includes considering the effects of certain changes to the existing regulations, such as:
 - 1. Minimum qualifications and organizational requirements for design approval applicants and holders including responsibilities and privileges
 - 2. SMS for design approval holders
 - 3. Compliance assurance
 - 4. Continued operational safety assurance
 - 5. Hazard reporting
- b. Cost and benefit and other impact information in support of developing the required Regulatory Evaluation(s) and Regulatory Flexibility economic analysis for applying any proposed changes to 14 CFR part 21 FAA certificate and approval holders. Cost and benefit analysis should include information obtained through the AIR SMS pilot project and should identify the specific areas of impact and present this information in quantitative terms to the extent possible.
- c. Part 21 design and production approval holder organizations to which the proposed SMS requirements should apply, taking into consideration cost and benefit information as well as public comments to the part 5 NPRM and the SMS-ARC *Design and Manufacturing Working Group Report High-Level Recommendations for SMS Requirements* dated Mareh 12, 2010.
- d. Changes to the FAA oversight methodology based on any recommendations for changes to part 21 that takes into account existing FAA processes and oversight and delegation programs for design and manufacturing related certificates and approvals and authorizations.
- e. Definitions and processes to be included in advisory, policy, and procedures material for addressing safety risk management responsibilities within a design and/or manufacturing organization. These definitions and processes should include:
 - 1. An operational definition of a "hazard" throughout the life cycle of a product in safety risk management.
 - 2. Definition of the term "organization" with respect to design and production approval holders to identify the limits of applicability of proposed SMS requirements, in

consideration of the broad range of organizational structures and activities within modern design and/or manufacturing organizations.

- 3. Hazard identification procedures.
- 4. Processes for the determination of acceptable safety risk.
- 5. Procedures to be included in advisory, policy, and procedures material for addressing safety assurance responsibilities within a design and/or manufacturing organization, including specific recommendations regarding "employee reporting systems".

The Director of Aircraft Certification Service (AIR-1) may propose additional tasks as necessary in support of a potential part 21 rulemaking action. The ARC may also request that AIR-1 add other tasks deemed relevant to the success of this initiative.

4. ARC PROCEDURES

- a. The ARC advises and provides written recommendations to AIR-1 and acts solely in an advisory capacity. Once the ARC recommendations are delivered to AIR-1, it is within his/her discretion to determine when and how the report of the ARC is released to the public.
- b. The ARC may propose additional tasks as necessary to AIR-1 for approval.
- c. The ARC will submit a report detailing recommendations within 18 months from the effective date of this charter. The chair of the ARC sends the recommendation report to both AIR-1 and the Director of the Office of Rulemaking.
- d. The ARC may reconvene following the submission of its recommendations for the purposes of providing advice and assistance to the FAA, at the discretion of AIR-1, provided the charter is still in effect.
- 5. ARC ORGANIZATION, MEMBERSHIP, AND ADMINISTRATION. The FAA will establish a committee of members of the aviation community. Members will be selected based on their familiarity with 14 CFR part 21, Safety Management Systems analysis, and regulatory compliance. Membership will be balanced in viewpoints, interests, and knowledge of the committee's objectives and scope. ARC membership is limited to promote discussion. Active participation and commitment by members will be essential for achieving the ARC's objectives. Attendance is essential for continued membership on the committee. When necessary, the ARC may set up specialized work groups that include at least one ARC member and invited subject matter experts from industry and government.

This ARC will consist of members from U.S. and foreign industry including representatives from designers and manufacturers holding part 21 certificates and approvals and other private sector aviation industry associations and advocacy groups. Invited foreign authorities and International Civil Aviation Organization (ICAO) representatives provide a valuable perspective from the global aviation community. These representatives are encouraged to fully participate in committee discussions; however, their participation does not include voting privileges on committee issues. The FAA's participation and support for the ARC will come from all affected lines-of-business.

- a. The ARC sponsor is AIR-1 who:
 - 1. Appoints members or organizations to the ARC, at the Director's sole discretion;

- 2. Selects the industry chair(s) from the ARC membership;
- 3. Selects the FAA's designated federal official for the ARC;
- 4. Receives all ARC recommendations and reports; and
- 5. Provides administrative support for the ARC through the Safety Management Design and Analysis Branch (AIR-150).
- b. Once appointed, the industry chair(s) will:
 - 1. Coordinate required committee and subcommittee (if any) meetings in order to meet the ARC's objectives and timelines;
 - 2. Provide notification to all ARC members of the time and place for each meeting;
 - 3. Ensure meeting agendas are established and provided to the committee members in a timely manner;
 - 4. Keep meeting minutes;
 - 5. Perform other responsibilities as required to ensure the ARC's objectives are met; and
 - 6. Provide status updates in writing to AIR-1 at 6 months and 12 months from the effective date of this charter.
- 6. COST AND COMPENSATION. The estimated operating cost (including *pro rata* share of salaries of FAA employees) to the Federal Government for this ARC is approximately \$400,000 annually. All travel costs for government employees will be the responsibility of the government employee's organization. Non-government representatives serve without government compensation and bear all costs related to their participation on the committee.
- PUBLIC PARTICIPATION. ARC meetings are not open to the public. Persons or organizations outside of the ARC who wish to attend a meeting must get approval in advance of the meeting from a committee co-chairperson or designated federal official.
- 8. AVAILABILITY OF RECORDS. Consistent with the Freedom of Information Act, Title 5, U.S.C., section 522, records, reports, agendas, working papers, and other documents that are made available to or prepared for or by the committee will be available for public inspection and copying at the FAA's Office of the Director, Aircraft Certification Service (AIR-1), 800 Independence Avenue SW, Washington, DC 20591. Fees will be charged for information furnished to the public according to the fee schedule published in Title 49 of the Code of Federal Regulations, part 7.

You can find this charter on the FAA Web Site at: http://www.faa.gov/about/committees/rulemaking/.

- 9. DISTRIBUTION. This order is distributed to director-level management in the Office of the Associate Administrator for Aviation Safety, the Office of Aviation Policy and Plans, the Office of Rulemaking, and the director- and division-level management in the Aircraft Certification Service.
- **10. EFFECTIVE DATE AND DURATION.** This committee is effective upon issuance of this charter. The committee shall remain in existence for 2 years, unless sooner terminated or

extended by the Administrator.

The effective date of this charter is October 5, 2012.

Michael P. Huerra Acting Administrator

APPENDIX E—KEY CONSIDERATIONS SUPPORTING ARC RECOMMENDATIONS

This appendix provides a comprehensive summary of all recommendations from the ARC report that support the four main recommendations and can be used as reference material. Each supporting recommendation has been grouped under the appropriate core recommendation as a key consideration. The purpose of this appendix is to provide any future rulemaking team a comprehensive overview of the key considerations the ARC feels should be taken into consideration for each of these areas if a rulemaking effort takes place.

RECOMMENDATION 1: SYSTEMS APPROACH TO CERTIFICATION—KEY CONSIDERATIONS

The systems approach to certification will be separated into three subsections that must be satisfied to move from the current state to a DO:

- a. Promote accountability framework and enhanced applicant showing.
- b. Establish minimum requirements for design approval applicant/holder.
- c. Establish requirements for voluntary certificated DOs.

1a. PROMOTE ACCOUNTABILITY FRAMEWORK AND ENHANCED APPLICANT SHOWING

Key Considerations:

- 1. DAH Procedures Manual: The ARC recommends further discussion on the necessary level of detail to include in the procedures manual and the appropriate reasons/rationale for FAA requests for changes to the procedures manual. (Refer to section 10.2.1.2.)
- 2. Accountability Framework: The ARC recommends the FAA consider the DO model's impact on the existing accountability framework, particularly regarding how a design approval applicant, DAH, and DO are related. (Refer to section 10.2.3.)

1b. ESTABLISH MINIMUM REQUIREMENTS FOR DESIGN APPROVAL APPLICANT/HOLDER

Key Considerations:

1. Establish minimum standards for design approval applicant/holder qualification and obligations to ensure applicants fully understand the type certification process and their roles and responsibilities.

1c. ESTABLISH REQUIREMENTS FOR VOLUNTARY CERTIFICATED DOS

Key Considerations:

- 1. Proposed Regulations, Preamble Language, and Guidance Material: The ARC recommends that proposed regulations, preamble language, and guidance material should be discussed as a follow-on activity to mature the information provided in this report. (Refer to section 6.)
- 2. Supplier Oversight "Pooling": The ARC recommends DO certificate holders be able to cooperate with other companies to pool supplier oversight responsibilities in a manner similar to what is currently done by manufacturing facilities and airlines under the

Coordinating Agency for Supplier Evaluation (C.A.S.E.), http://www.caseinc.org/. (Refer to section 6.5.1.)

- 3. Specialty Service Providers (SSP): The ARC recommends the FAA give priority to developing a means for recognizing an accreditation system for SSPs (for example, Nadcap or similar). (Refer to section 6.5.3.)
- 4. Establish requirements for the issuance and oversight of certificated DOs that includes the necessary compliance assurance, safety management, and controls to make all compliance determinations through applicant showing and verification processes. Through FAA certificate management oversight and direct project involvement in defined risk-based areas, the FAA may rely on the DO compliance determinations to make its finding for the issuance of a design approval. This report builds on the recommendations submitted to the FAA by the CDO ARC in May 2008.

However, the ARC had significant concerns about attempting to set a specific date when a certified DO would be required. It was felt that this could cause both industry and the FAA to have to push other things aside just to satisfy the DO schedule requirements, affecting industry's ability to deliver its products in a timely manner and the FAA's ability to support that activity. After much deliberation, it was determined that a phased approach to the DO implementation would be more feasible than a single-step process.

The Part 21/SMS ARC recommends a building block approach to implementing DO, which includes establishing a clear accountability framework, particularly regarding how a design approval applicant, DAH, and DO are related; transitioning the FAA's oversight of design activities to a centralized systematic model; optimizing full use of ODA authorization; and implementing new organizational and SMS requirements. (Refer to section 10.)

Summary of the Building Block Approach to a Design Organization: The phased approach is described as a building block approach that would let a company build pieces of the DO requirements on a more flexible schedule. This building block approach would allow the FAA and industry to immediately begin taking steps to improve processes and make the changes necessary for DO implementation. These steps shown in figure E–1 below will enable a much more manageable transition to DO

in the future. System / Organization Future State (DO) LOPI Performance Govern **Based Oversight** (PBO) Product / Article Transition via Building Blocks (Accredited Org) LOPI DD Retained Govern CBO & PBO Key steps: 1. FAA Central Oversight and Mature ODA's 2. SMS demonstrates Performance Based Oversight 3. Expansion of Applicant Show from risk based to "Level of Capability" based determination (ASOC) LOPI Delegation Decision (DD) Retained Govern Compliance Based Oversight (CBO) Current State: ODA / DER

Figure E–1. Building Block Approach to DO

With successful implementation of these building blocks, the ARC supports a future rulemaking to consider mandatory implementation of DO. (Refer to section 10.2.1.3.)

- 5. Business Structure Variation: The ARC recommends further development of how business structure variations will be accommodated under the DO framework. (Refer to section 6.6.)
- 6. Form of the DO Certificate: The ARC recommends a certificate structure similar in nature to the EASA DOA certificate and terms of approval. (Refer to section 6.7.3.)
- 7. DA Transfer: The ARC recommends non-DO design approval transfer requirements be provided by a separate follow-on activity to the ARC. (Refer to section 6.7.4.)
- 8. Maintenance Aspects of ICA: The ARC recommends the DO determine maintenance technical material aspects of ICA requirements. (Refer to section 6.7.8.)
- 9. Environmental Compliance Determinations: The ARC recommends the FAA propose to the EPA that the process-based approach to compliance, as established by DO program principles (which is far more robust than the normal delegation process) is sufficient to ensure compliance with the environmental aspects of 14 CFR parts 34 and 36. (Refer to section 6.7.9.)
- 10. Form for DO Transmittal of Approved Data: DO-issued service bulletins should be a means for DOs to provide "approved data" for general use. When issued, service bulletins constitute a change in type design by the holder and convey the necessary "approved data" to implement the change by owner/operators under part 43. A new or revised form is also needed for domestic and international recognition of "approved data" created under the DO concept. The ARC recommends the form be similar to the FAA Forms 8110–3 and 8100–9 currently used to approve data in the FAA's delegation system. (Refer to section 6.8.1.)

11. DO Is a Choice: The ARC recommends DO applicability thresholds for any design approval applicants or DAHs should be optional. (Refer to section 10.)

RECOMMENDATION 2: SAFETY MANAGEMENT SYSTEM—KEY CONSIDERATIONS

In addition to the Safety Management Systems Requirements section in the report (section 4.2), the following are key considerations that the ARC feels should be taken into consideration if a rulemaking effort takes place.

Key Considerations:

- The ARC evaluated the proposed § 5.27 and determined it is not necessary for D&M organizations. Therefore, the ARC recommends the FAA modify part 21 to make part 5, excluding § 5.27, the SMS requirements for organizations meeting the SMS applicability threshold. The ARC will continue to develop guidance material for D&M implementation of part 5 and appreciates the FAA's willingness to engage industry in this regard. (Refer to section 5.1.1.)
- 2. Establish a requirement for implementing SMS consistent with the proposed part 5 for design and production approval organizations. The ARC recommends this new requirement apply to organizations that design or manufacture type-certificated products (under a TC or production certificate) and those that design or manufacture articles (under a TSO or PMA) or make changes to products (under an STC) that could directly prevent continued safe flight and landing if they fail. (Refer to section 5.1.2.)
- 3. Policy and Guidance Material: The ARC recommends the FAA and industry develop SMS guidance for organizations that: design or manufacture products (that is, aircraft, engines, or propellers); design or manufacture articles (TSO, PMA) whose failure could directly prevent continued safe flight and landing; or make design changes to a product through an STC, failure of which could directly prevent continued safe flight and landing. The ARC developed SMS regulatory material and a basis for preamble, policy, and guidance material, as provided in this report; however, it determined more work was necessary to produce supporting guidance material. The ARC recommends the SMS Working Group continue to develop supporting guidance material with a goal of completing the task by Spring 2015 by providing an addendum to this report through the ARC. (Refer to section 5.3.)
- 4. SMS Concept of Operations (CONOPS): The ARC has developed a CONOPS describing the intent of the part 5 SMS framework (safety policy, SRM, safety assurance, and safety promotion) for D&M organizations as it applies to each life cycle phase (design and certification, production and airworthiness certification, and continued airworthiness) of a product or article. The ARC recommends the CONOPS form the basis for the development of preamble, policy, and guidance material for D&M organizations. The ARC also recommends that, as described in the CONOPS, existing processes and procedures should be considered as meeting the intent of part 5. (Refer to section 5.3.)

- 5. Availability of Data for SRM: The ARC recommends the FAA develop an approach to make fleet data already provided to the FAA (hours, flights, reported failures, malfunctions, and defects and service difficulty reports) readily available to D&M organizations, in support of executing SRM (§ 5.71). (Refer to section 5.3.)
- 6. The ARC recommends implementing the changes to §§ 21.3 and 21.4 identified in the table found in section 8.2.2 of the ARC report. (Refer to section 8.2.2.)
- The ARC recommends developing or revising the following guidance materials to support the recommended rule changes and facilitate FAA oversight: (1) criteria for consistent understanding of the language in § 21.3(d) "has resulted in or may result in a finding of an unsafe condition by the Administrator"; (2) acceptable compliance demonstration and verification regarding the safety analysis referenced in § 21.3(d); (3) changes to AC 21–9B, Manufacturers Reporting of Failure, Malfunctions or Defects, to include the recommended guidance for §§ 21.3(d) and 21.3(e). (Refer to section 8.3.2.)
- 8. The ARC recommends developing proposed regulation changes and guidance or process proposals to address safety risk that is acceptable in the short term while long-term safety risk control/mitigation plans are developed and implemented. (Refer to section 8.2.5.2.)

RECOMMENDATION 3: EVOLUTION OF OVERSIGHT—KEY CONSIDERATIONS

In addition to the Evolution of Oversight section in the report (section 4.3), the following are key considerations that the ARC feels should be taken into consideration if a rulemaking effort takes place.

Key Considerations:

- 1. Performance-Based Oversight: The ARC developed proposed practices for FAA oversight that is correlated with recommended D&M organizational changes. This enables a shift to performance-based oversight where the FAA can effectively allocate resources based on D&M system risk management performance, and moves the FAA from a total dependence on discrete compliance findings, audits, and inspections. The ARC recommends chartering a dedicated effort with the FAA and industry to develop guidance for determining performance indicators that are mutually acceptable before implementation of the new oversight model. (Refer to section 3.3.)
- 2. Single Centralized Oversight Organization: The ARC recommends a single centralized oversight presence and systemic (process-based) approach for initial and ongoing assessments. The three key oversight areas are: (1) Organizational—transition from traditional show/find compliance to organizational PBO model; (2) Product and Articles—transition from the FAA's traditional role of direct project involvement to a LOPI approach focused on performing governmental functions; (3) Post-Certification (COS)—transition from traditional reactionary approach to a systemic (process-based) surveillance model discussed in more detail later in this section.

- 3. The assessment methodology will cover a standardized approach to quality, design, and safety. In support of this recommendation, the ARC's Oversight Working Group has provided a capability-based assessment tool, PROs/CONs analysis of oversight management options, and supporting rationale for the recommendation. FAA oversight teams would report to a centralized FAA organization. Establishing a central FAA oversight organization will achieve standard surveillance practices, create centralized policy, be a single source/repository for the oversight data that will drive the risk-based modeling controls, and allow for a highly trained staff in system surveillance, skill management, and a single source for corrective actions. (Refer to section 10.1.4.3.)
- 4. The recommended three transitional steps to the centralized oversight organization are—

1. Proof of Concept—Pre Implementation: Ensure through proof of concept plans that the requirements proposed by the ARC are practical, effective, and efficient. Determine if the transition from "mature ODA" to DO has benefits to the FAA and industry.

2. FAA Transition Plan Transition Principle: The FAA should not release a final rule before it has demonstrated the necessary cultural shift to perform system oversight. To achieve a cultural shift, policy and organizational changes may be required.

3. Industry Transition Plan: The organization must establish the systems required of an approved organization while still working as a non-certificated applicant or a delegated organization. Applicants working toward becoming a DO demonstrate compliance to those requirements on an "as ready" basis. (Refer to section 10.1.5.2.)

RECOMMENDATION 4: TSO MODERNIZATION AND PART 21 MISCELLANEOUS CLEANUP

In addition to the Part 21 Miscellaneous Cleanup and TSO Modernization section in the report (section 4.4), the following are key considerations that the ARC feels should be taken into consideration if a rulemaking effort takes place.

TSO MODERNIZATION

Key Considerations:

The ARC recommends the following changes to modernize the TSO requirements:

- 1. Allow TSO organizations to issue their own TSOAs, relative to scalable privileges for particular types of TSO standards. (Alternate approaches via a certified TSO organization or expansion of TSO ODA functions.) (Refer to section 9.2.1.)
- Clarify the types of data that can be approved under a TSOA (that is, type design of the article and declared performance of the article including non-TSO functions and incomplete TSO), and expectations for acceptance of approved TSO data for installation. (Require and approve DDP via revision to § 21.601(b)(2) and proposed new § 21.603(a)(3).) (Refer to section 9.2.1.)
- Proposed new §§ 21.603(a)(3) and 21.619(d), Design changes, for subsequent design changes to declare non-TSO functions. (Additional guidance including a "decision table" to assist in differentiating between TSO supporting features and integrated non-TSO functions.) (Refer to section 9.2.1.)

- 4. Rule revision to remove the term "model number" from TSO rules and replace it with a requirement for a "unique identifier." (Revision to §§ 21.603(b) and 21.619 for subsequent design changes.) (Refer to section 9.2.1.)
- 5. The ARC recommends changing part 21 to establish the effective TSO revision level at the beginning of the project, not at the end. (Revision to § 21.603(a).) (Refer to section 9.2.1.)
- 6. The ARC recommends a process for the TSO holder to continue marking TSO articles following a determination of "a design discrepancy that does not result in an unsafe condition." (Revision of § 45.10(b) and proposed new § 21.616(i), Responsibility of the holder.) (Refer to section 9.2.1.)
- 7. The ARC recommends maintaining the privilege for TSO holders to make minor or insignificant (sub-minor) changes to articles without further approval. (Refer to section 9.1.3.)
- 8. The ARC recommends clarifying the TSO application data, manufacturer data and furnished data requirements. (Refer to the TSO Subteam Report, included as appendix H to this report). (Refer to section 9.1.3.)
- 9. The ARC recommends developing expanded guidance to promote the uniform definition and treatment of integrated non-TSO functions by applicants, installation developers, and the FAA. (Refer to the TSO Subteam Report, included as appendix H to this report). (Refer to section 9.1.3.)
- 10. The ARC recommends an applicant should submit to the Administrator a signed undertaking to carry out the responsibilities as a DAH before issuance of a design approval. (Refer to section 9.1.3.)

MISCELLANEOUS PART 21 CLEANUP

Key Considerations:

The ARC recommends—

- 1. AC 20–62, Eligibility, Quality, and Identification of Aeronautical Replacement Parts, paragraph 6, Discussion, be amended to include an explanation of the scope of §§ 21.8 and 21.9. (Refer to section 9.1.1.3.)
- 2. Amending § 21.9(a) to add paragraph (a)(7), which reads: "Produced in any other manner approved by the FAA." (Refer to section 9.1.1.3.)
- 3. The removal of § 21.335(b) from 14 CFR. (Refer to section 9.1.2.)

APPENDIX A—PART 21/SMS ARC MEMBERSHIP

MEMBERS AND ALTERNATES

- Mr. Walter Desrosier, Co-Chair, GAMA
- Mr. Michael Reinert, Co-Chair, FAA AIR-150
- Mr. Rick Baggette, Boeing
- Mr. Glenn Baxter, Bombardier
- Mr. Edmond Boullay, Alternate, ASD/US Crest
- Mr. Rafael Borges, Observer, ANAC
- Mr. John Bouma, Cessna
- Mr. Dave Chapel, GE
- Ms. Maria Clara, Observer, ANAC
- Mr. Michael Collins, NATCA
- Mr. Robert Cook, FAA AIR–200
- Mr. Peter Corbeel, Observer, EASA
- Ms. Doris Costa, Observer, ANAC
- Ms. Jodi Diamant-Boustead, Alternate, AIAC/Pratt Canada
- Mr. Jason Dickstein, MARPA
- Mr. Tomaso DiPaolo, Alternate, NATCA
- Mr. Chris Eick, Honeywell
- Mr. Jimmy Eyl, Banyan Air Service
- Mr. Robert Ferguson, Observer, TCCA
- Ms. Amy Garzaro, FAA AIR-150
- Mr. Scott Geddie, FAA AIR-110
- Mr. Paul Greer, FAA AGC
- Mr. Julian Hall, Observer, EASA

- Mr. David Hempe, FAA AIR-100
- Ms. Katrina Holiday, FAA ARM
- Mr. Charles Huber, FAA SMS/ANM-100
- Mr. Daniel Leach, FAA APO
- Mr. Eric Lesage, Airbus
- Mr. Michael Linegang, FAA AIR–110
- Mr. Mark Lopez, Alternate, A4A
- Mr. Stacey Mason, Observer, TCCA
- Ms. Linda Navarro, Observer, FAA AIR-150
- Mr. George Novak, AIA
- Ms. Jan Novak, Observer, EASA
- Mr. Ric Peri, AEA
- Mr. Dennis Piotrowski, BELAC
- Mr. Tom Rogozinski, Honeywell
- Mr. Dan Shapiro, Sikorsky General
- Mr. Eric Sivel, Observer, EASA
- Mr. Roger Southgate, Rockwell Collins
- Mr. Giles Strickler, FAA ARM
- Mr. Marcus Tittiger, Observer, TCCA
- Mr. Scott VanBuren, Observer, FAA AVP (SMS)
- Mr. Mark Watton, Delta Airlines
- Mr. Bill Whitton, Gulfstream

PROGRAM SUPPORT

Mr. Larry Van Dyke, GAMA

APPENDIX B—GLOSSARY

Term	Definition	Source
Acceptable Risk	The level of risk that individuals or groups are willing to accept given the benefits gained. Each organization will have its own acceptable risk level, which is derived from its legal and regulatory compliance responsibilities, its threat profile, and its business/organizational drivers and impacts.	AVS Order 8000.367A
Accident	An unplanned event or series of events that results in death, injury, or damage to, or loss of, equipment or property.	AVS Order 8000.367A FAA Order 8040.4A
Accountable Manager	(a) Accountable manager means the person designated by an applicant or design approval holder who is responsible for and has the authority over all design approval operations that are conducted under part 21, including ensuring that design approval holder personnel follow the regulations and serving as the primary contact with the FAA.	14 CFR Part 145.3(a) modified for use by DAH
Accountability Framework	An established set of responsibilities and commitments of the FAA and industry	Refer to appendix O to this report.
Aerospace System	U.S. airspace, all manned and unmanned vehicles operating in that airspace, all U.S. aviation operators, airports, airfields, air navigation services, pilots, regulations, policies, procedures, facilities, equipment, and all aviation-related industry.	AVS Order 8000.367A FAA Order 8040.4A
Aircraft Accident	An occurrence associated with the operation of an aircraft that takes place between the time any person boards the aircraft with the intention of flight and all such persons have disembarked, and in which any person suffers death or serious injury, or in which the aircraft receives substantial damage.	49 CFR 830.2
Analysis	The process of identifying a question or issue to be addressed, examining the issue, investigating the results, interpreting the results, and possibly making a recommendation. Analysis typically involves using scientific or mathematical methods for evaluation.	FAA Order 8040.4A
Applicant Show With Capability (ASOC)	Based on the Oversight Working Group's model of DOs getting recognized for demonstrated capabilities.	

Term	Definition	Source
Approved Data	Data approved by FAA employees, its designees, or a DO acting under the authority of its certificate.	
Assessment	Process of measuring or judging the value or level of something.	FAA Order 8040.4A
Certificate Surveillance	FAA actions to monitor the DO certificate holder and to determine the holder's compliance with the provisions of its certificate. Note: In the Oversight section we discuss managing these organizations through surveillance.	
Compliance Assurance System (CAS)	DO holder's system for ensuring that it complies with the applicable regulations.	
Compliance Finding	FAA decision (either directly or through a designee) that compliance has been shown with the applicable regulatory requirements.	
Control	Refer to Safety Risk Control.	AVS Order 8000.367A
Corrective Action	Action to eliminate or mitigate the cause or reduce the effects of a detected nonconformity or other undesirable situation.	AVS Order 8000.367A
	An action required to be taken by the DO to address noncompliances and problems with the organization's procedures or performance.	
	The noncompliances may result from—	
	• Internal audits conducted by the DO,	
	• FAA surveillance,	
	• DO employee observations, and	
	Voluntary disclosures.	
Culture of Compliance	Knowledge, beliefs, attitudes, and behaviors of an organization that are focused on ensuring regulatory compliance with all its activities.	

Term	Definition	Source
Descriptive	From the DO Working Group:	
Data	• Data that defines the type design that needs to be determined compliant to the applicable airworthiness standards. The descriptive data is what is approved by the FAA when a design approval certificate is issued.	
	• The drawings and specifications necessary to define the configuration shown to comply.	
Determination of Compliance	A decision made by the certificate holder that compliance has been shown with the applicable regulatory requirements. [Note: The ARC has referred to "regulatory requirements" rather than just "airworthiness standards" because its recommendation that DOs eventually include determination of compliance with other 14 CFR parts, such as parts 26, 34, and 36.] It may also be a decision made by the certificate holder that data previously approve by the FAA or data determined to comply by another CAA under the provisions of a bilateral airworthiness agreement between the United States and a foreign country or jurisdiction, are valid and applicable to the design of the product, part, or appliance for which it is to be used, including the applicable certification or approval basis.	
DO Executive	The company individual directly responsible for ensuring that the DO meets all of its regulatory responsibilities.	
DO Point(s) of Contact	The individual(s) within the DO responsible for all communications with the FAA.	
Eligible Data	Data developed under an approved DO system, assuming a specified, but not FAA-established, certification basis, and product type design if appropriate.	

Term	Definition	Source
Enforcement	An action taken by the FAA most appropriate to promote safety and compliance with the statutory and regulatory requirements. The program provides a wide range of options for addressing noncompliance:	14 CFR Part 13
	 Educational and remedial training efforts, Administrative action in the form of either a warning notice or letter of correction, 	
	 Certificate suspensions for a fixed period of time, Civil penalties, 	
	• Indefinite certificate suspensions pending compliance or demonstration of qualifications,	
	Certificate revocations,	
	• Injunctions, and	
	Referrals for criminal prosecution.	
Evaluation	Determining the adequacy and effectiveness of an organization through a review of organizational policies, procedures, and systems.	
FAA Oversight Team	FAA personnel assigned to provide guidance and oversight of the DO in meeting its regulatory requirements.	
Hazard	A condition that could foreseeably cause or contribute to an accident.	AVS Order 8000.367A FAA Order 8040.4A
Inspection	A formal systematic and independent review of organizational policies, procedures, and systems.	
Interoperability	The ability for each SMS to be part of the system or systems through interdependent processes and/or components with shared principles, information, and governance.	AVS Order 8000.367A

Term	Definition	Source
Level of Project Involvement (LOPI)	The interactive process that the DO shares with its assigned Aircraft Certification Office (ACO) for specific engineering/design elements and with the Manufacturing Inspection District Office (MIDO) for specific production elements during certification projects. The criteria/factors influencing the decision of when to be involved will include but is not limited to Governmental functions, such as—	
	 Novel or unusual features which may require issuance of Special Conditions, Significant issues which may require Issue Papers, and Defining Equivalent Level(s) of Safety 	
	(ELOS).	
Likelihood	The estimated probability or frequency, in quantitative or qualitative terms, of a hazard's effect or outcome.	AVS Order 8000.367A FAA Order 8040.4A
Management System	See the definition for System.	
Methods (or) Means of Compliance	 Need a definition from DO Working Group Notes: Method: Process Means: Capability 	
Mitigation	A means to reduce the risk of a hazard. Refer to Safety Risk Control.	AVS Order 8000.367A FAA Order 8040.4A
Nonconformity	Non-fulfillment of a requirement. This includes but is not limited to noncompliance with Federal regulations. It also includes an organization's requirements, policies, and procedures as well as requirements of safety risk controls developed by the organization.	AVS Order 8000.367A

Term	Definition	Source
Novel or Unusual	"The phrase "novel or unusual" as used in 14 CFR 21.16 is a very relative term. As used hereafter in applying a4 CFR 21.16 to justify the issuance of special conditions, "novel or unusual" will be taken with respect to the state of technology envisaged by the applicable airworthiness standards of this subchapter. It must be recognized that in some areas which will vary from time to time the state of the regulations may somewhat lag the state of the art in new design because of the rapidity in which the state of the art is advancing in civil aeronautical design and because of the time required to develop the experience base needed by the Applicants for type certification of a new design have the opportunity to mitigate the impact of not knowing the precise airworthiness standards to be applied for "novel or unusual design features: by consulting with the FAA early in their certification planning when such features are suspected or known by the applicant to exist. It should also be recognized that, because of the intentional objective nature of the airworthiness standards of this subchapter, many new design features which might be thought of as "novel or unusual design features" may already be adequately covered by existing regulations, thus obviating the need to issue special conditions." Preamble material to 14 CFR 21.16.	
Oversight	 A systems approach to review an organization's performance, validate the development of their defined system and verify compliance to the requirements of a certified DO to determine sufficiency. Oversight activities include— Reviewing the work performed, Evaluating performance for quality assurance, Ensuring that required training has been completed, Providing constructive feedback, and Taking corrective action, including enforcement as necessary. 	

Term	Definition	Source
Procedure	A fixed, step-by-step sequence of activities or course of action (with definite start and end points) that must be followed in the same order to correctly perform a task.	
Product/Service Provider	An organization engaged in the delivery of aviation products or services.	AVS Order 8000.367A
Quality Management System	From the DO Working Group—A set of interrelated or interacting quality processes accomplished by the organization through the establishment of policy and objectives, and achieving those objectives.	
Risk	Refer to Safety Risk. The terms risk and safety risk are used synonymously.	AVS Order 8000.367A FAA Order 8040.4A
Safety	The state in which the risk of harm to persons or property damage is acceptable.	AVS Order 8000.367A FAA Order 8040.4A
Safety Assurance	Processes within the SMS that function systematically to ensure the performance and effectiveness of safety risk controls and that the organization meets or exceeds its safety objectives through the collection, analysis, and assessment of information.	AVS Order 8000.367A FAA Order 8040.4A
Safety Culture	The shared values, actions, and behaviors that demonstrate a commitment to safety over competing goals and demands.	AVS Order 8000.367A
	The product of individual and group values, attitudes, competencies, and patterns of behavior that determine the commitment to an organization's safety programs.	
Safety Management	The act of understanding and making decisions and taking actions to lower risk, inherent in all human activity, to acceptable levels.	
Safety Management System (SMS)	The formal, top-down, organization-wide approach to managing safety risk and assuring the effectiveness of safety risk controls. It includes systematic procedures, practices, and policies for the management of safety risk.	AVS Order 8000.367A
Safety Objective	A measurable goal or desirable outcome related to safety.	AVS Order 8000.367A
Safety Performance	Realized or actual safety accomplishment relative to the organization's safety objectives.	AVS Order 8000.367A

Term	Definition	Source
Safety Policy	The organization's documented commitment to safety, which defines its safety objectives and the accountabilities and responsibilities of its employees in regards to safety.	AVS Order 8000.367A
Safety Promotion	A combination of training and communication of safety information to support the implementation and operation of an SMS in an organization.	AVS Order 8000.367A
Safety Requirement	A safety condition or capability that must be met or passed by a system to satisfy a contract, standard, specification, or other formally imposed document or need.	AVS Order 8000.367A
Safety Risk	The composite of predicted severity and likelihood of the potential effect of a hazard. Initial—The predicted severity and likelihood of a hazard's effects or outcomes when it is first identified and assessed; includes the effects of preexisting risk controls in the current environment. Current—The predicted severity and likelihood at the current time. Residual—The remaining predicted severity and likelihood that exists after all selected risk control techniques have been implemented.	AVS Order 8000.367A FAA Order 8040.4A
Safety Risk Control	A means to reduce or eliminate the effects of hazards.	AVS Order 8000.367A FAA Order 8040.4A
Safety Risk Management (SRM)	A process within the SMS composed of describing the system, identifying the hazards, and analyzing, assessing, and controlling risk.	AVS Order 8000.367A FAA Order 8040.4A
Senior Company Management	Those in the company management chain above the DO Executive who are accountable for the actions of the DO.	
Severity	The consequence or impact of a hazard's effect or outcome in terms of degree of loss or harm.	AVS Order 8000.367A FAA Order 8040.4A
Showing	Determination of compliance to the Airworthiness Regulations by the applicant.	
Show/Find	The process by which the applicant "shows" how they complied with a regulation and the FAA "finds" that the applicant has adequately shown compliance to the regulation.	FAA Order 8110.4C

Term	Definition	Source
Statement of Compliance	A statement from the DO to the Administrator certifying that compliance with the applicable regulatory requirements has been determined and the procedures listed in its FAA-approved DO procedures manual have been followed.	
Substantiating Data	Documentation related to a design approval applicant's showing or compliance to the applicable airworthiness standards.	
Substitute risk	Risk unintentionally created as a consequence of safety risk control(s).	AVS Order 8000.367A
Supplier DO	A separate DO entity in its own right provides an article to an applicant/holder DO.	
Surveillance	The combination of evaluation and inspection to accomplish a review of organizational system to determine the adequacy and effectiveness of an organization.	
System	An integrated set of constituent elements that are combined in an operational or support environment to accomplish a defined objective. These elements include people, hardware, software, firmware, information, procedures, facilities, services, and other support facets.	AVS Order 8000.367A FAA Order 8040.4A
System State	An expression of the various conditions, characterized by quantities or qualities, in which a system can exist.	AVS Order 8000.367A
Validation	Validation is the process of proving that the functions, procedures, controls, and safety standards are correct and the right system is being built. i.e. the requirements are unambiguous, correct, complete, and verifiable.	
Verification	The process that ensures that the system requirements have been met by the design solution and the system is ready to be used in the operational environment for which it is intended.	

Page intentionally left blank.

APPENDIX C—ACRONYMS

AAQG	Americas Aerospace Quality Group
AC	Advisory Circular
ACO	Aircraft Certification Office
ACPRR	Aircraft Certification Process Review and Reform
AD	Airworthiness Directive
ADO	Agent Design Organization
AEE	Office of Environment and Energy
AEG	Aircraft Evaluation Group
AFS	Flight Standards Service
AIA	Aerospace Industries Association
AIR	Aircraft Certification Service
ANAC	National Civil Aviation Agency of Brazil
AO	Accredited Organization
APO-300	Office of Aviation Policy and Plans, Economic Analysis Division
ARC	Aviation Rulemaking Committee
ASA	Aviation Suppliers Association
ASE	Oversight Team of Engineers
ASI	Aviation Safety Inspector
ASTM	ASTM International
ATC	Air Traffic Control
AVS	FAA Office of Aviation Safety
BTS	Bureau of Transportation Statistics
CAA	Civil Aviation Authority (of another country)
CAAC	Civil Aviation Administration of China

CAP	Compliance Assurance Procedure
CAR	Civil Aviation Regulation
CAS	Compliance Assurance System
C.A.S.E.	Coordinating Agency for Supplier Evaluation
СВА	Cost Benefit Analysis
СВО	Compliance Based Oversight
CCS	Compliance Certification System
CDO	Certified Design Organization
CFR	Code of Federal Regulations
CMIS	Certificate Management Information System
CONOPS	Concept of Operations
COS	Continued Operational Safety
D&M	Design and Manufacturing
DAH	Design Approval Holder
DAR	Designated Airworthiness Representative
DAS	Designated Alteration Station
DDP	Declaration of Design and Performance
DER	Designated Engineering Representative
DMIR	Designated Manufacturing Inspection Representative
DMS	Design Management System
DO	Design Organization
DOA	Design Organization Approval
DPE	Designated Pilot Examiner
DPO	Design Production Organization
EASA	European Aviation Safety Agency

ELOS	Equivalent Level of Safety
EPA	Environmental Protection Agency
ETOPS	Extended Operations
FAA	Federal Aviation Administration
FOEB	Flight Operations Evaluations Board
FSB	Flight Standardization Board
FSDO	Flight Standards District Office
GAO	General Accountability Office
IAQG	International Aerospace Quality Group
ICA	Instructions for Continued Airworthiness
ICAO	International Civil Aviation Organization
IEEE	Institute of Electrical and Electronics Engineers
ISO	International Standards Organization
LOPI	Level of Project Involvement
MIDO	Manufacturing Inspection District Office
MMEL	Master Minimum Equipment List
MOC	Memorandum of Cooperation
MRA	Maintenance, Repair, and Alteration
MRB	Maintenance Review Board
MSAD	Monitor Safety/Analyze Data
MSG	Maintenance Steering Group
NATCA	National Air Traffic Controllers Association
NPRM	Notice of Proposed Rulemaking
OCS	Organization Control System
ODA	Organization Designation Authorization

ODAR	Organizational Designated Airworthiness Representative
OIG	Office of Inspector General
OMB	Office of Management and Budget
OMT	Organizational Management Team
РАН	Production Approval Holder
PBO	Performance-Based Oversight
PI	Principal Inspector
PMA	Parts Manufacturer Approval
PNL	Project Notification Letter
POC	Point of Contact
QMS	Quality Management System
QSA	Quality System Audit
R&D	Research and Development
RBRT	Risk-based Resource Targeting
RPM	Revenue Passenger Miles
RTCA	Radio Technical Commission for Aeronautics
RTM	Revenue Ton Miles
SA	Supplier Audit
SAE	Society of Automotive Engineers
SEMP	Systems Engineering Management Plan
SFAR	Special Federal Aviation Regulation
SM ICG	Safety Management International Collaboration Group
SMM	Safety Management Manual
SMS	Safety Management System
SRM	Safety Risk Management

SSP	Specialty Service Providers
STC	Supplemental Type Certificate
TC	Type Certificate
TCCA	Transport Canada Civil Aviation
TSO	Technical Standard Order
TSOA	Technical Standard Order Authorization
UAS	Unmanned Aircraft Systems
UM	Unit Member
U.S.C.	United States Code
USITC	U.S. International Trade Commission
Page intentionally left blank.

APPENDIX D—PART 21/SMS ARC CHARTER



U.S. DEPARTMENT OF TRANSPORTATION FEDERAL AVIATION ADMINISTRATION Aviation Rulemaking Committee Charter

Effective Date: 10/5/2012

SUBJ: 14 CFR 21 / Safety Management Systems Aviation Rulemaking Committee

 PURPOSE. This Charter creates the Aviation Rulemaking Committee (ARC) for Part 21 / Safety Management Systems (SMS) according to the Administrator's authority under Title 49 of the United States Code (49 U.S.C.) 106(p)(5). This charter also outlines the committee's organization, responsibilities, and tasks.

2. BACKGROUND.

On May 22, 2012, the Aircraft Certification Process Review and Reform ARC submitted a report to the FAA recommending that we undertake a review to update part 21 certification procedures to reflect a systems safety approach to product certification processes and oversight of design organizations. Design organizations must have full responsibility and accountability through the establishment of regulatory requirements for minimum qualification, performance, and management systems.

Consistent with FAA Order VS 8000.367, and the International Civil Aviation Organization (ICAO) Annex 8, the Aircraft Certification Service (AIR) has been actively developing and implementing an internal and external SMS. The initial focus was primarily on developing an internal set of processes, tools, and methodologies that facilitate the transition into the future state. AIR began that effort in 2005 and has made progress in defining key processes and tools. Later, with support from industry participants, the activities expanded to include development of standards for design and manufacturing organizations. Through implementation of pilot SMS projects with certain companies, the FAA is collecting information that will help define the scope of the SMS for Design Approval Holders (DAHs), validate certain best practices, and expand the knowledge base within the workforce and industry with respect to the essential elements of a robust SMS for manufacturers.

SMS requires a proactive approach to discovering and addressing hazards before they exhibit safety consequences. SMS also includes processes that seek to identify potential organizational breakdowns and necessary process improvements which allow management to address a safety issue before a noncompliant or unsafe condition results. SMS is not a substitute for compliance with FAA regulations or FAA oversight activities.

3. OBJECTIVES AND TASKS OF THE ARC. AIR wants to evaluate certain improvements to the effectiveness and efficiency of existing "certification procedures for products and parts," along with incorporating SMS in the design and manufacturing environment. This includes considering the effects of certain changes to the existing regulations, such as applicant qualifications, hazard (or safety) reporting, compliance assurance, and continued operation safety assurance systems for

Distribution:

Initiated By:

all DAHs. The intent is to facilitate shifting towards a systems approach for DAHs that is similar to that used for production approval holder requirements, which involves a clear understanding of roles, responsibilities, and privileges. As part of this evaluation, we want to determine the best way industry and the FAA can effectively fulfill their respective compliance and safety responsibilities.

The ARC will provide a forum for the U.S. aviation community to discuss and provide recommendations to the FAA. The committee is expected to provide general information and guidance regarding proposed changes to part 21 and the AVS SMS program as it relates to design and manufacturing certificate and approval holders.

- a. The ARC will provide the FAA recommendations, which may include proposals for rulemaking, suggested processes, policies and guidance, and any further action it determines the agency should contemplate for part 21 to align with the SMS requirements documented in proposed 14 CFR part 5, which is the central component of the NPRM entitled *Safety Management Systems for Part 121 Certificate Holders* [Docket No. FAA-2009-0671; Notice No. 10-15].
- b. The ARC, serving in an advisory capacity, is expected to present and discuss whatever input, guidance, and recommendations its members consider critical to the FAA's ultimate disposition, development, and implementation of proposed regulatory requirements and related guidance and policy as necessary to the future direction for part 21 to include applicant pre-qualifications, approval holder recognition, and SMS considerations.
- c. The ARC will also consider proposed revisions to clarify and update engineering/designoriented regulatory requirements to part 21. In support of design certification and continued airworthiness, the evaluation should include improvements in the areas of:
 - 1. Application process
 - 2. Applicant qualifications
 - 3. Standardized certification criteria
 - 4. Identifying design approval holder responsibilities and privileges
 - 5. Clarifying continued airworthiness requirements
 - 6. Clarifying design approvals needing Instructions for Continued Airworthiness
 - 7. Clarifying TSO design approval processes
 - 8. Process definition for determining eligibility of U.S. surplus military aircraft in the restricted category

This proposal additionally corrects regulatory language, implements editorial changes for clarification, and standardizes regulatory language to reflect the global aviation environment. While this information will be shared with the ARC, responses to "clean-up" proposals are not required as part of the deliverables.

d. Proposed part 5 and International Civil Aviation Organization (ICAO) Annex 8 and Annex 19 (draft) serve as the foundation for the ARC's consideration regarding how the FAA will address its responsibilities for developing and implementing SMS

requirements and the management and oversight of its regulated product/service providers. The ARC must respect the framework outlined in proposed part 5 and the ICAO Annexes when it provides the FAA recommendations with respect to application of SMS. However, the FAA will consider proposed changes to part 5 as deemed necessary from a design and manufacturing perspective.

Recommendation Report. The ARC shall make recommendations and submit a report addressing the following:

- a. Improvements, which may include proposals for rulemaking, processes, policies and guidance for 14 CFR part 21 that reflect a systems approach for safety. This will promote an effective and efficient certification process, which includes considering the effects of certain changes to the existing regulations, such as:
 - 1. Minimum qualifications and organizational requirements for design approval applicants and holders including responsibilities and privileges
 - 2. SMS for design approval holders
 - 3. Compliance assurance
 - 4. Continued operational safety assurance
 - 5. Hazard reporting
- b. Cost and benefit and other impact information in support of developing the required Regulatory Evaluation(s) and Regulatory Flexibility economic analysis for applying any proposed changes to 14 CFR part 21 FAA certificate and approval holders. Cost and benefit analysis should include information obtained through the AIR SMS pilot project and should identify the specific areas of impact and present this information in quantitative terms to the extent possible.
- c. Part 21 design and production approval holder organizations to which the proposed SMS requirements should apply, taking into consideration cost and benefit information as well as public comments to the part 5 NPRM and the SMS-ARC *Design and Manufacturing Working Group Report High-Level Recommendations for SMS Requirements* dated Mareh 12, 2010.
- d. Changes to the FAA oversight methodology based on any recommendations for changes to part 21 that takes into account existing FAA processes and oversight and delegation programs for design and manufacturing related certificates and approvals and authorizations.
- e. Definitions and processes to be included in advisory, policy, and procedures material for addressing safety risk management responsibilities within a design and/or manufacturing organization. These definitions and processes should include:
 - 1. An operational definition of a "hazard" throughout the life cycle of a product in safety risk management.
 - 2. Definition of the term "organization" with respect to design and production approval holders to identify the limits of applicability of proposed SMS requirements, in

consideration of the broad range of organizational structures and activities within modern design and/or manufacturing organizations.

- 3. Hazard identification procedures.
- 4. Processes for the determination of acceptable safety risk.
- 5. Procedures to be included in advisory, policy, and procedures material for addressing safety assurance responsibilities within a design and/or manufacturing organization, including specific recommendations regarding "employee reporting systems".

The Director of Aircraft Certification Service (AIR-1) may propose additional tasks as necessary in support of a potential part 21 rulemaking action. The ARC may also request that AIR-1 add other tasks deemed relevant to the success of this initiative.

4. ARC PROCEDURES

- a. The ARC advises and provides written recommendations to AIR-1 and acts solely in an advisory capacity. Once the ARC recommendations are delivered to AIR-1, it is within his/her discretion to determine when and how the report of the ARC is released to the public.
- b. The ARC may propose additional tasks as necessary to AIR-1 for approval.
- c. The ARC will submit a report detailing recommendations within 18 months from the effective date of this charter. The chair of the ARC sends the recommendation report to both AIR-1 and the Director of the Office of Rulemaking.
- d. The ARC may reconvene following the submission of its recommendations for the purposes of providing advice and assistance to the FAA, at the discretion of AIR-1, provided the charter is still in effect.
- 5. ARC ORGANIZATION, MEMBERSHIP, AND ADMINISTRATION. The FAA will establish a committee of members of the aviation community. Members will be selected based on their familiarity with 14 CFR part 21, Safety Management Systems analysis, and regulatory compliance. Membership will be balanced in viewpoints, interests, and knowledge of the committee's objectives and scope. ARC membership is limited to promote discussion. Active participation and commitment by members will be essential for achieving the ARC's objectives. Attendance is essential for continued membership on the committee. When necessary, the ARC may set up specialized work groups that include at least one ARC member and invited subject matter experts from industry and government.

This ARC will consist of members from U.S. and foreign industry including representatives from designers and manufacturers holding part 21 certificates and approvals and other private sector aviation industry associations and advocacy groups. Invited foreign authorities and International Civil Aviation Organization (ICAO) representatives provide a valuable perspective from the global aviation community. These representatives are encouraged to fully participate in committee discussions; however, their participation does not include voting privileges on committee issues. The FAA's participation and support for the ARC will come from all affected lines-of-business.

- a. The ARC sponsor is AIR-1 who:
 - 1. Appoints members or organizations to the ARC, at the Director's sole discretion;

- 2. Selects the industry chair(s) from the ARC membership;
- 3. Selects the FAA's designated federal official for the ARC;
- 4. Receives all ARC recommendations and reports; and
- 5. Provides administrative support for the ARC through the Safety Management Design and Analysis Branch (AIR-150).
- b. Once appointed, the industry chair(s) will:
 - 1. Coordinate required committee and subcommittee (if any) meetings in order to meet the ARC's objectives and timelines;
 - 2. Provide notification to all ARC members of the time and place for each meeting;
 - 3. Ensure meeting agendas are established and provided to the committee members in a timely manner;
 - 4. Keep meeting minutes;
 - 5. Perform other responsibilities as required to ensure the ARC's objectives are met; and
 - 6. Provide status updates in writing to AIR-1 at 6 months and 12 months from the effective date of this charter.
- 6. COST AND COMPENSATION. The estimated operating cost (including *pro rata* share of salaries of FAA employees) to the Federal Government for this ARC is approximately \$400,000 annually. All travel costs for government employees will be the responsibility of the government employee's organization. Non-government representatives serve without government compensation and bear all costs related to their participation on the committee.
- PUBLIC PARTICIPATION. ARC meetings are not open to the public. Persons or organizations outside of the ARC who wish to attend a meeting must get approval in advance of the meeting from a committee co-chairperson or designated federal official.
- 8. AVAILABILITY OF RECORDS. Consistent with the Freedom of Information Act, Title 5, U.S.C., section 522, records, reports, agendas, working papers, and other documents that are made available to or prepared for or by the committee will be available for public inspection and copying at the FAA's Office of the Director, Aircraft Certification Service (AIR-1), 800 Independence Avenue SW, Washington, DC 20591. Fees will be charged for information furnished to the public according to the fee schedule published in Title 49 of the Code of Federal Regulations, part 7.

You can find this charter on the FAA Web Site at: http://www.faa.gov/about/committees/rulemaking/.

- 9. DISTRIBUTION. This order is distributed to director-level management in the Office of the Associate Administrator for Aviation Safety, the Office of Aviation Policy and Plans, the Office of Rulemaking, and the director- and division-level management in the Aircraft Certification Service.
- **10. EFFECTIVE DATE AND DURATION.** This committee is effective upon issuance of this charter. The committee shall remain in existence for 2 years, unless sooner terminated or

extended by the Administrator.

The effective date of this charter is October 5, 2012.

Michael P. Huerra Acting Administrator

APPENDIX E—KEY CONSIDERATIONS SUPPORTING ARC RECOMMENDATIONS

This appendix provides a comprehensive summary of all recommendations from the ARC report that support the four main recommendations and can be used as reference material. Each supporting recommendation has been grouped under the appropriate core recommendation as a key consideration. The purpose of this appendix is to provide any future rulemaking team a comprehensive overview of the key considerations the ARC feels should be taken into consideration for each of these areas if a rulemaking effort takes place.

RECOMMENDATION 1: SYSTEMS APPROACH TO CERTIFICATION—KEY CONSIDERATIONS

The systems approach to certification will be separated into three subsections that must be satisfied to move from the current state to a DO:

- a. Promote accountability framework and enhanced applicant showing.
- b. Establish minimum requirements for design approval applicant/holder.
- c. Establish requirements for voluntary certificated DOs.

1a. PROMOTE ACCOUNTABILITY FRAMEWORK AND ENHANCED APPLICANT SHOWING

Key Considerations:

- 1. DAH Procedures Manual: The ARC recommends further discussion on the necessary level of detail to include in the procedures manual and the appropriate reasons/rationale for FAA requests for changes to the procedures manual. (Refer to section 10.2.1.2.)
- 2. Accountability Framework: The ARC recommends the FAA consider the DO model's impact on the existing accountability framework, particularly regarding how a design approval applicant, DAH, and DO are related. (Refer to section 10.2.3.)

1b. ESTABLISH MINIMUM REQUIREMENTS FOR DESIGN APPROVAL APPLICANT/HOLDER

Key Considerations:

1. Establish minimum standards for design approval applicant/holder qualification and obligations to ensure applicants fully understand the type certification process and their roles and responsibilities.

1c. ESTABLISH REQUIREMENTS FOR VOLUNTARY CERTIFICATED DOS

Key Considerations:

- 1. Proposed Regulations, Preamble Language, and Guidance Material: The ARC recommends that proposed regulations, preamble language, and guidance material should be discussed as a follow-on activity to mature the information provided in this report. (Refer to section 6.)
- 2. Supplier Oversight "Pooling": The ARC recommends DO certificate holders be able to cooperate with other companies to pool supplier oversight responsibilities in a manner similar to what is currently done by manufacturing facilities and airlines under the

Coordinating Agency for Supplier Evaluation (C.A.S.E.), http://www.caseinc.org/. (Refer to section 6.5.1.)

- 3. Specialty Service Providers (SSP): The ARC recommends the FAA give priority to developing a means for recognizing an accreditation system for SSPs (for example, Nadcap or similar). (Refer to section 6.5.3.)
- 4. Establish requirements for the issuance and oversight of certificated DOs that includes the necessary compliance assurance, safety management, and controls to make all compliance determinations through applicant showing and verification processes. Through FAA certificate management oversight and direct project involvement in defined risk-based areas, the FAA may rely on the DO compliance determinations to make its finding for the issuance of a design approval. This report builds on the recommendations submitted to the FAA by the CDO ARC in May 2008.

However, the ARC had significant concerns about attempting to set a specific date when a certified DO would be required. It was felt that this could cause both industry and the FAA to have to push other things aside just to satisfy the DO schedule requirements, affecting industry's ability to deliver its products in a timely manner and the FAA's ability to support that activity. After much deliberation, it was determined that a phased approach to the DO implementation would be more feasible than a single-step process.

The Part 21/SMS ARC recommends a building block approach to implementing DO, which includes establishing a clear accountability framework, particularly regarding how a design approval applicant, DAH, and DO are related; transitioning the FAA's oversight of design activities to a centralized systematic model; optimizing full use of ODA authorization; and implementing new organizational and SMS requirements. (Refer to section 10.)

Summary of the Building Block Approach to a Design Organization: The phased approach is described as a building block approach that would let a company build pieces of the DO requirements on a more flexible schedule. This building block approach would allow the FAA and industry to immediately begin taking steps to improve processes and make the changes necessary for DO implementation. These steps shown in figure E–1 below will enable a much more manageable transition to DO

in the future. System / Organization Future State (DO) LOPI Performance Govern **Based Oversight** (PBO) Product / Article Transition via Building Blocks (Accredited Org) LOPI DD Retained Govern CBO & PBO Key steps: 1. FAA Central Oversight and Mature ODA's 2. SMS demonstrates Performance Based Oversight 3. Expansion of Applicant Show from risk based to "Level of Capability" based determination (ASOC) LOPI Delegation Decision (DD) Retained Govern Compliance Based Oversight (CBO) Current State: ODA / DER

Figure E–1. Building Block Approach to DO

With successful implementation of these building blocks, the ARC supports a future rulemaking to consider mandatory implementation of DO. (Refer to section 10.2.1.3.)

- 5. Business Structure Variation: The ARC recommends further development of how business structure variations will be accommodated under the DO framework. (Refer to section 6.6.)
- 6. Form of the DO Certificate: The ARC recommends a certificate structure similar in nature to the EASA DOA certificate and terms of approval. (Refer to section 6.7.3.)
- 7. DA Transfer: The ARC recommends non-DO design approval transfer requirements be provided by a separate follow-on activity to the ARC. (Refer to section 6.7.4.)
- 8. Maintenance Aspects of ICA: The ARC recommends the DO determine maintenance technical material aspects of ICA requirements. (Refer to section 6.7.8.)
- 9. Environmental Compliance Determinations: The ARC recommends the FAA propose to the EPA that the process-based approach to compliance, as established by DO program principles (which is far more robust than the normal delegation process) is sufficient to ensure compliance with the environmental aspects of 14 CFR parts 34 and 36. (Refer to section 6.7.9.)
- 10. Form for DO Transmittal of Approved Data: DO-issued service bulletins should be a means for DOs to provide "approved data" for general use. When issued, service bulletins constitute a change in type design by the holder and convey the necessary "approved data" to implement the change by owner/operators under part 43. A new or revised form is also needed for domestic and international recognition of "approved data" created under the DO concept. The ARC recommends the form be similar to the FAA Forms 8110–3 and 8100–9 currently used to approve data in the FAA's delegation system. (Refer to section 6.8.1.)

11. DO Is a Choice: The ARC recommends DO applicability thresholds for any design approval applicants or DAHs should be optional. (Refer to section 10.)

RECOMMENDATION 2: SAFETY MANAGEMENT SYSTEM—KEY CONSIDERATIONS

In addition to the Safety Management Systems Requirements section in the report (section 4.2), the following are key considerations that the ARC feels should be taken into consideration if a rulemaking effort takes place.

Key Considerations:

- The ARC evaluated the proposed § 5.27 and determined it is not necessary for D&M organizations. Therefore, the ARC recommends the FAA modify part 21 to make part 5, excluding § 5.27, the SMS requirements for organizations meeting the SMS applicability threshold. The ARC will continue to develop guidance material for D&M implementation of part 5 and appreciates the FAA's willingness to engage industry in this regard. (Refer to section 5.1.1.)
- 2. Establish a requirement for implementing SMS consistent with the proposed part 5 for design and production approval organizations. The ARC recommends this new requirement apply to organizations that design or manufacture type-certificated products (under a TC or production certificate) and those that design or manufacture articles (under a TSO or PMA) or make changes to products (under an STC) that could directly prevent continued safe flight and landing if they fail. (Refer to section 5.1.2.)
- 3. Policy and Guidance Material: The ARC recommends the FAA and industry develop SMS guidance for organizations that: design or manufacture products (that is, aircraft, engines, or propellers); design or manufacture articles (TSO, PMA) whose failure could directly prevent continued safe flight and landing; or make design changes to a product through an STC, failure of which could directly prevent continued safe flight and landing. The ARC developed SMS regulatory material and a basis for preamble, policy, and guidance material, as provided in this report; however, it determined more work was necessary to produce supporting guidance material. The ARC recommends the SMS Working Group continue to develop supporting guidance material with a goal of completing the task by Spring 2015 by providing an addendum to this report through the ARC. (Refer to section 5.3.)
- 4. SMS Concept of Operations (CONOPS): The ARC has developed a CONOPS describing the intent of the part 5 SMS framework (safety policy, SRM, safety assurance, and safety promotion) for D&M organizations as it applies to each life cycle phase (design and certification, production and airworthiness certification, and continued airworthiness) of a product or article. The ARC recommends the CONOPS form the basis for the development of preamble, policy, and guidance material for D&M organizations. The ARC also recommends that, as described in the CONOPS, existing processes and procedures should be considered as meeting the intent of part 5. (Refer to section 5.3.)

- 5. Availability of Data for SRM: The ARC recommends the FAA develop an approach to make fleet data already provided to the FAA (hours, flights, reported failures, malfunctions, and defects and service difficulty reports) readily available to D&M organizations, in support of executing SRM (§ 5.71). (Refer to section 5.3.)
- 6. The ARC recommends implementing the changes to §§ 21.3 and 21.4 identified in the table found in section 8.2.2 of the ARC report. (Refer to section 8.2.2.)
- The ARC recommends developing or revising the following guidance materials to support the recommended rule changes and facilitate FAA oversight: (1) criteria for consistent understanding of the language in § 21.3(d) "has resulted in or may result in a finding of an unsafe condition by the Administrator"; (2) acceptable compliance demonstration and verification regarding the safety analysis referenced in § 21.3(d); (3) changes to AC 21–9B, Manufacturers Reporting of Failure, Malfunctions or Defects, to include the recommended guidance for §§ 21.3(d) and 21.3(e). (Refer to section 8.3.2.)
- 8. The ARC recommends developing proposed regulation changes and guidance or process proposals to address safety risk that is acceptable in the short term while long-term safety risk control/mitigation plans are developed and implemented. (Refer to section 8.2.5.2.)

RECOMMENDATION 3: EVOLUTION OF OVERSIGHT—KEY CONSIDERATIONS

In addition to the Evolution of Oversight section in the report (section 4.3), the following are key considerations that the ARC feels should be taken into consideration if a rulemaking effort takes place.

Key Considerations:

- 1. Performance-Based Oversight: The ARC developed proposed practices for FAA oversight that is correlated with recommended D&M organizational changes. This enables a shift to performance-based oversight where the FAA can effectively allocate resources based on D&M system risk management performance, and moves the FAA from a total dependence on discrete compliance findings, audits, and inspections. The ARC recommends chartering a dedicated effort with the FAA and industry to develop guidance for determining performance indicators that are mutually acceptable before implementation of the new oversight model. (Refer to section 3.3.)
- 2. Single Centralized Oversight Organization: The ARC recommends a single centralized oversight presence and systemic (process-based) approach for initial and ongoing assessments. The three key oversight areas are: (1) Organizational—transition from traditional show/find compliance to organizational PBO model; (2) Product and Articles—transition from the FAA's traditional role of direct project involvement to a LOPI approach focused on performing governmental functions; (3) Post-Certification (COS)—transition from traditional reactionary approach to a systemic (process-based) surveillance model discussed in more detail later in this section.

- 3. The assessment methodology will cover a standardized approach to quality, design, and safety. In support of this recommendation, the ARC's Oversight Working Group has provided a capability-based assessment tool, PROs/CONs analysis of oversight management options, and supporting rationale for the recommendation. FAA oversight teams would report to a centralized FAA organization. Establishing a central FAA oversight organization will achieve standard surveillance practices, create centralized policy, be a single source/repository for the oversight data that will drive the risk-based modeling controls, and allow for a highly trained staff in system surveillance, skill management, and a single source for corrective actions. (Refer to section 10.1.4.3.)
- 4. The recommended three transitional steps to the centralized oversight organization are—

1. Proof of Concept—Pre Implementation: Ensure through proof of concept plans that the requirements proposed by the ARC are practical, effective, and efficient. Determine if the transition from "mature ODA" to DO has benefits to the FAA and industry.

2. FAA Transition Plan Transition Principle: The FAA should not release a final rule before it has demonstrated the necessary cultural shift to perform system oversight. To achieve a cultural shift, policy and organizational changes may be required.

3. Industry Transition Plan: The organization must establish the systems required of an approved organization while still working as a non-certificated applicant or a delegated organization. Applicants working toward becoming a DO demonstrate compliance to those requirements on an "as ready" basis. (Refer to section 10.1.5.2.)

RECOMMENDATION 4: TSO MODERNIZATION AND PART 21 MISCELLANEOUS CLEANUP

In addition to the Part 21 Miscellaneous Cleanup and TSO Modernization section in the report (section 4.4), the following are key considerations that the ARC feels should be taken into consideration if a rulemaking effort takes place.

TSO MODERNIZATION

Key Considerations:

The ARC recommends the following changes to modernize the TSO requirements:

- 1. Allow TSO organizations to issue their own TSOAs, relative to scalable privileges for particular types of TSO standards. (Alternate approaches via a certified TSO organization or expansion of TSO ODA functions.) (Refer to section 9.2.1.)
- Clarify the types of data that can be approved under a TSOA (that is, type design of the article and declared performance of the article including non-TSO functions and incomplete TSO), and expectations for acceptance of approved TSO data for installation. (Require and approve DDP via revision to § 21.601(b)(2) and proposed new § 21.603(a)(3).) (Refer to section 9.2.1.)
- Proposed new §§ 21.603(a)(3) and 21.619(d), Design changes, for subsequent design changes to declare non-TSO functions. (Additional guidance including a "decision table" to assist in differentiating between TSO supporting features and integrated non-TSO functions.) (Refer to section 9.2.1.)

- 4. Rule revision to remove the term "model number" from TSO rules and replace it with a requirement for a "unique identifier." (Revision to §§ 21.603(b) and 21.619 for subsequent design changes.) (Refer to section 9.2.1.)
- 5. The ARC recommends changing part 21 to establish the effective TSO revision level at the beginning of the project, not at the end. (Revision to § 21.603(a).) (Refer to section 9.2.1.)
- 6. The ARC recommends a process for the TSO holder to continue marking TSO articles following a determination of "a design discrepancy that does not result in an unsafe condition." (Revision of § 45.10(b) and proposed new § 21.616(i), Responsibility of the holder.) (Refer to section 9.2.1.)
- 7. The ARC recommends maintaining the privilege for TSO holders to make minor or insignificant (sub-minor) changes to articles without further approval. (Refer to section 9.1.3.)
- 8. The ARC recommends clarifying the TSO application data, manufacturer data and furnished data requirements. (Refer to the TSO Subteam Report, included as appendix H to this report). (Refer to section 9.1.3.)
- 9. The ARC recommends developing expanded guidance to promote the uniform definition and treatment of integrated non-TSO functions by applicants, installation developers, and the FAA. (Refer to the TSO Subteam Report, included as appendix H to this report). (Refer to section 9.1.3.)
- 10. The ARC recommends an applicant should submit to the Administrator a signed undertaking to carry out the responsibilities as a DAH before issuance of a design approval. (Refer to section 9.1.3.)

MISCELLANEOUS PART 21 CLEANUP

Key Considerations:

The ARC recommends—

- 1. AC 20–62, Eligibility, Quality, and Identification of Aeronautical Replacement Parts, paragraph 6, Discussion, be amended to include an explanation of the scope of §§ 21.8 and 21.9. (Refer to section 9.1.1.3.)
- 2. Amending § 21.9(a) to add paragraph (a)(7), which reads: "Produced in any other manner approved by the FAA." (Refer to section 9.1.1.3.)
- 3. The removal of § 21.335(b) from 14 CFR. (Refer to section 9.1.2.)

APPENDIX F—DESIGN ORGANIZATION WORKING GROUP

21ARC Working Document - Not for Distribution

Design Organization Concept Report

by

Design Organization Working Group of the Part 21/SMS Aviation Rulemaking Committee

7 February 2014

This page is intentionally blank.

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

ii

TABLE OF CONTENTS

TABLE OF CONTENTS	iiiiii
I. INTRODUCTION	
I.A. SUMMARY OF RECOMMENDATIONS	
H. BACKGROUND	
ILA. CHARTERED DESIGN ORGANIZATION WORKING GROUP	
ILB. DO WG REPORT	
II.C. WHAT IS DO?	
II C.(1) DO Buhances Compliance. Efficiency, and Effectiveness. II C.(2) DO is Not "Self-Certification".	
II.D. THE ROLE OF THE FAA	
III. ACCOUNTABILITY FRAMEWORK	
III.A. ACCOUNTABILITY CONSIDERATIONS	
IV. DO GUIDING PRINCIPLES AND ATTRIBUTES	
IV.A. OVERVIEW	
IV.B. DO MODEL AND FRAMEWORK	
IV B (1) (a) DA Applicant DO and DAH DO IV B (1) (a) DA Applicant DO and DAH DO IV B (2) A DO Certificate is Nor a Delegation IV B (3) Eligibility and Thresholds IV B (4) Certificated Organization IV B (5) Applicability of Port 21 Requirements IV B (6) Transition Provisions	11 16 16 24 24 25
IV.C. DO MINIMUM REQUIREMENTS	
IV C (1) DO and OCS IV C (2) Compliance Assurance System IV C (3) Salery Management System (SMS) IV C (4) DO Project Applications and Activity Reporting	27 30 33 34
IV.D. DO RELATIONSHIPS AND SUPPLIER CONTROL	
 IV D (1) General Requirements IV D (2) Oversight of Foreign Suppliers IV D (3) The Need for Specially Survice Providers IV D (4) Inter-DO Relationships and Responsibilities as Design Partners IV D (5) Agent DO IV D (6) Business Structure Variation 	15 36 17 38 38 38 38
IV.E. SCOPE, LIMITATIONS, AND PRIVILEGES OF DO CERTIFICATES	
(V,E.(1) FAA Limitations on the DO Certificate, (V,E.(2) Scope of DO Certificate Privileges Must Be "Punctionally Complete".	
Dessan Disancestian Working Group	in .

Part 21/SMS Aviation Rulemaking Committee

IV.E.(3) Form of a DO Certificate	43
IV.E.(4) Transfer of Design Approvals under DO	44
IV.E.(4).(a) PMA Specific Design Approvals under DO	44
IV.E.(5) The Generation of "Eligible Data"	44
IV.E.(6) Voluntary Disclosure Privileges	45
IV.E.(7) Manufacturing and Production Functions Under DO	46
IV.E.(8) Flight Standards Functions	47
IV.E.(9) Noise, Fuel Venting, and Exhaust Emissions	49
IV.E.(10) Establishing a Certification Basis	50
IV.E.(11) Changed Product Rule	50
IV. F. DO APPROVAL OF DATA	.51
IV F (1) Service Bulletins and a Standardized form for DO Transmittal of Approved Data	57
IV.F.(2) DO Use of Previously Approved Data	
IV.F.(3) Use of FAA's Delegation System	59
IV.F.(4) DO Recognition and Use of Design Suppliers with FAA Credentials (including other DOs)	59
V. CHANGE MANAGEMENT PLAN	61
Appendix A. DO Working Group Members	.62
Appendix B. DO WG Charter	.63
Appendix C. Glossary of Terms	. 68
Appendix D. List of Acronyms	.71
Appendix E. DO Challenges, Benefits and Concerns	.73
Appendix F. DO Threshold Decision Tree	.77

I. INTRODUCTION

This report provides concepts for a Design Organization (DO) developed by the Design Organization Working Group (WG) of the Part 21 / Safety Management System (SMS) Aviation Balemalong Committee (ARC)

The intent of this report is to define and address the details necessary for the Federal Aviation Administration (FAA) to recognize a DO that meets organizational and system requirements sufficient to ensure that a DO is capable of making compliance statements upon which the FAA may rely. Specifically the DO framework, microtum requirements relationships, limitations, and privileges are addressed by this report.

This conceptual report establishes a framework and captures the inteni of a DO. In many instances, the concepts put forth in this report are either identical or build upon the May 2008 Certified Design Organization (CDO) ARC report. The DO WG considered the CDO) ARC report extensively and in many cases quoted language directly in this report without explicit estation.

I.A. Summary of Recommendations

The WG provides the following recommendations in response to the WG charter and associated taskings.

- Proposed Regulations, Preamble Language, and Guidance Material: The DO WG recommends that proposed regulations, preamble language, and guidance material should be discussed as a follow-on activity to mature the information provided in this report. See II B - DO WG Report
- Accountability Framework: The WG recommends the FAA consider the DO nodel unpact on the existing Accountability Framework, particularly with respect to how an design approval (DA) applicant, design approval holder (DAH) and DO are related. See III A – Accountability Considerations.
- DAB responsibility statement: The WG recommends that a separate, yet to be tasked WG, consider whether an applicant should submit to the Administrator a signed undertaking to carry out the responsibilities as a DAH prior to issuance of a DA. See III.B (1) – DO Model and Framework
- DO is a choice: Industry's recommendation is that there be no manufatory thrasholds for any DA applicants or DAHs. All DOs would be optional. Sec-IV 0 (3) – filigibility and Throsholds.
- Making the Failure Modes Effects (FME) Assessment Transparent: The DO WG proposes to use the current regulations and guidance wherever possible, and

Deopa Organization Working Group But 21/8MS Aviation Rulemaking Committee

м

recommends encouraging the various directorates to work with industry to develop regulations and guidance where it is currently absent or unclear. See IV.B.(3) – Secondary "Risk Based" Threshold

- Areas Without Existing FME Assessment Regulation/Guidance: The DO WG recommends the various directorates work with industry to develop regulations and guidance for any areas where existing FME Assessment is absent, or unclear. See IV.B.(3) Secondary "Risk Based" Threshold
- Other disciplines and Product Level Safety Assessments: The WG recommends that the secondary threshold be limited to <u>Product Level</u> Safety Risk Assessments. Any "other" secondary threshold assessment should be set using the criteria of will the potential failure mode "affect the continued safe flight and landing of the aircraft." See IV.B.(3) Secondary "Risk Based" Threshold
- Major/Minor Determinations: While Major/Minor determinations are out of scope of this WG, the DO WG recommends that the FAA and industry drive towards a consistent, easily understood, safety risk-based determination of Major/Minor change. See IV.B.(3) Other Threshold Models Discussed
- Significant Change: While significant change determinations are out of scope of this WG, the DO WG recommends that the FAA and industry drive towards a consistent, easily understood, safety risk based determination of significant change. See IV.B.(3) Other Threshold Models Discussed Paragraph
- **DO Procedures Manual:** The WG recommends further discussion on the necessary level of detail for inclusion in the procedures manual and the appropriate reasons/rationale for FAA requests for changes to the procedures manual. See IV.C.(1) DO and Organization Control System (OCS)
- SMS as a requirement of DO: The WG recommends further discussion by the ARC on the subject of SMS being directly tied to DOs. See IV.C.(4) Safety Management System
- Supplier Oversight "Pooling": The DO WG recommends that DO certificate holders be able to cooperate with other companies to pool supplier oversight responsibilities, in a manner similar to what is currently done by manufacturing facilities. See IV.D.(1) General Requirements
- Speciality Service Providers (SSP): The DO WG recommends that the FAA give priority to developing a means for recognizing an accreditation system for SSPs. See IV.D.(3) The Need for Speciality Service Providers
- Business Structure Variation: The DO WG recommends further development of how business structures variations will be accommodated under the DO framework. See IV.D.(6) Business Structure Variation
- Form of the DO Certificate: The DO WG recommends a certificate structure similar in nature to the European Aviation Safety Agency (EASA) Design Organization Approval (DOA) Certificate and Terms of Approval. See IV.E.(3) Form of DO Certificate

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

- **DA Transfer:** The WG recommends non-DO DA transfer requirements be provided by a separate WG to the ARC. See IV.E.(4) Transfer of Design Approvals under DO
- Maintenance Aspects of Instructions for Continued Airworthiness (ICA): The WG recommends that the maintenance aspects of ICA requirements also be determined by the DO. See IV.E.(8) Flight Standards Functions
- Environmental Compliance Determinations: The WG recommends that the FAA propose to the Environmental Protection Agency (EPA) that the process-based approach to compliance, as established by DO program principles, is far more robust than the normal delegation process and is sufficient to ensure compliance with the environmental aspects of the 14 Code of Federal Regulations (CFR) Parts 34 and 36. See IV.E.(9) Noise, Fuel Venting and Exhaust Emissions
- Form for DO Transmittal of Approved Data: DO-issued service bulletins should be a means for DOs to provide "approved data" for general use. When issued, service bulletins constitute a change in type design by the holder and convey the necessary "approved data" to implement the change by owner/operators under part 43. A new or revised form is also needed for domestic and international recognition of "approved data" created under the DO concept. The DO WG recommends that the form be similar to the FAA Forms 8110-3 and 8100-9 that are currently used to approve data in the FAA's delegation system. See IV.F.(1) Service Bulletins and a Standardized form for DO Transmittal of Approved Data

II. BACKGROUND

II.A. Chartered Design Organization Working Group

On October 5, 2012, the FAA Administrator chartered an Aviation Rulentaling Committee (ARC) for Part 21/SMS. The ARC was tasked to recommend improvements to the effectiveness and efficiency of existing certification procedures for products and parts along with incorporating SMS in the design and manufacturing environments. The ARC was tasked with making recommendations, including proposals for rulemaking suggested processes, policies, guidance, and any other actions the agency should take in support of its goal.

In addressing its charter, the ARC forther chartered the DO WO on March (7, 2013 to assist it in developing details associated with FAA recognition of DOs in Part 21. The WG was tasked to define and address the details necessary for the FAA to recognize a DO that meets organizational and system requirements sufficient to ensure that a DO is capable of making compliance statements upon which the FAA toay rely. The WG presented and discussed in-depth principles, guidance, and recommendations that the members of the committee considered relevant to the implementation of the DO concept.

The DO WG membership is listed in Appendix A. The WG charter is contained in Appendix B.

ILB. DO WG Report

This report represents the consensus position of the WG and its recommendations to the ARC for the development, scope, and operation of a DO from Industry and FAA perspectives. Dissenting opinions expressed by members against specific section concepts or recommendations are also presented within each section. Specifics relevant to SMS and Oversight requirements were assigned to other ARC WGs.

This report is intended to be a thorough presentation of background material to be considered by the ARC in making its recommendations to the FAA regarding the requirements, privileges, and responsibilities of a FAA recognized DO. Taken in in entirety, if constitutes the WG recommendations detailing the guiding principles, concepts, and attributes necessary to prepare regulatory language for the drafting of a Notice of Proposed Rulemaking (NPRM). The WG recommends that proposed regulations, preamble language, and guidance material should be discussed as a followon activity to mature the information provided in this report.

A glossary of terms and a list of acronyous used in this report are contained in Appendix-C and Appendix D, respectively

> Design Organization Working Group Pan 21/SMS Activition Rolenoking Committee

-44

II.C. What is DO?

A DO is a regulatory-recognized organization that meets organizational and system requirements sufficient to ensure that it is capable of making compliance statements upon which the FAA may rely in support of obtaining Design Approvals (DA) under 14 Code of Federal Regulations (CFR) part 21. As such, the DO maintains and follows processes to manage its certification projects, as well as the continued airworthiness of its products.

A DO will include organizations applying for, supporting the application for, or holding Type Certificates (TC), Supplemental Type Certificates (STC), or Parts Manufacturer Approvals (PMA). Further details regarding eligibility are included in section IV.B.(3) of this report.

II.C.(1) DO Enhances Compliance, Efficiency, and Effectiveness

Moving Beyond Delegation. Under the current FAA delegation system, there are several "persons" working together to find compliance with the FAA requirements, be they individuals or delegated organizations. In a typical program, the FAA personnel

- will make some of the findings,
- will delegate some to designees or delegated organizations, and
- by using the FAA's statutory discretionary authority, may choose not to review some demonstrations by the applicant in less-safety-critical areas or where the FAA has confidence in the applicant's compliance with the regulations.

The current process of obtaining a DA places no requirement on the applicant to establish a system of documented processes and procedures to show compliance. As such, the variety in applicant capabilities makes it highly resource-intensive for the FAA to effectively deal with the certification process.

With DO, the design organization operates in accordance with their FAA-approved processes and compliance assurance system (CAS). If a non-compliance is found by the DO or by the FAA, the DO's CAS is subject to review and change, as required by its FAA-approved procedures manual. The DO is also subject to enforcement action, including civil penalty, for not following its approved procedures and for not adhering to the regulatory requirement to present an accurate statement of compliance to the Administrator for approval.

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

Compliance Assurance System (CAS) Enhances Compliance. Under DO, the establishment and determination of compliant designs will be made through a CAS, with appropriate internal checks and balances to ensure it is functioning properly. The WG envisions that the CAS will be composed of a design management system (DMS) and a compliance certification system (CCS), both of which are defined in greater detail in section IV.C.(3) of this report.

Organizations must have a thorough understanding of the regulatory requirements and what constitutes compliance. They must incorporate design and quality systems so that compliance is designed into the product along the path toward certification. When this occurs, every step along the path of product design and development is a step along the path toward compliance, and is not dependent on the FAA or its designees to make the compliance determination.

This system will be required under DO in order to provide a high degree of regulatory compliance assurance that is shown to be as effective as a skilled independent check. Such a system with appropriate FAA oversight increases assurance that compliance with the requirements has been established by the applicant. The FAA is able to rely on this increased assurance when making its finding for the issuance of the certificate, rather than requiring the FAA's direct involvement in making discrete findings.

Other Systems Also Enhance Compliance. In addition to the CAS, the DO is subject to requirements for a SMS and organization control system (OCS). The systematic approach to the engineering certification process, coupled with SMS and OCS enhances the organization's overall ability to consistently perform the compliance assurance functions, and to identify and correct problems that may arise. The above noted system requirements are addressed within section IV.C of this report.

Enhanced Efficiency and Effectiveness. Industry's product development and certification efficiency is paced, in part, by the FAA under the existing delegation system. Delays like certification plan approval generate additional uncertainty in program schedule and costs. DO provides greater industry control over a product delivery schedule by allowing the FAA to rely on statements of compliance by the DO. This allows the FAA to shift its focus from specific compliance findings to compliance and safety system oversight, providing the opportunity for greater capacity in the system.

Additionally, FAA approval of DO relational or supplier processes allows a DO to utilize supplier or consortium-member contributions to certification projects without data duplication.

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

II.C.(2) DO is Not "Self-Certification"

Under DO, all determinations of compliance within the DO certificate holder authority will be made by the DO organization. This does not mean that DO is selfcertification by industry.

The FAA will retain the right to review, audit, and otherwise oversee the operation of the DO while the DO makes compliance determinations, as well as after the PAA has issued design and airworthiness certificates. These functions are being addressed by the Oversight WG

Approval by the FAA of DO processes, certification bases, Means of Compliance (MOC), and continued oversight during certification activities differentiate DO from only delf-certification process. This contrasts with self-certification, wherein the government would issue standards and the applicant would certify that it has met those standards when it introduces its product into service. For example, the National Highway Transportation Safety Administration (NHTSA) establishes safety standards for motor vehicles and there is no government involvement or review until after products are introduced into service.

II.D. The Role of the FAA

The role of the FAA with respect to DOs will be provided by the Overaight WG of the Part 21/SMS ARC. The Oversight WG will address DO assessment, issuance of DO certificates, ongoing DO oversight, and FAA level of project involvement (LOPI)(e.g., certification blood approved MOC) in DO certification program.

The DO WG expects that the FAA will be able to rely on the DO-in-good-standing 5 statement of compliance to issue the DA, without further showing, verification, or involvement is poted in Section IV F1

Design Organization Working Group Bird 21/SMS Avioneo Rukensking Lemonitee

III. ACCOUNTABILITY FRAMEWORK

The foundation for development of the DO program, as with any design certification unogram developed by the FAA, must be an accountability framework that begins with Congressional statutes and is applied through FAA regulations that establish clear roles and responsibilities for both the FAA and Industry. This framework is largely derived from Title 49 and 14 CFR part 21, and addresses the roles and responsibilities of applicants, certificate holders, and the FAA. This framework includes each stakeholder's role in the certification process and continued airworthiness, as well as FAA's role in developing standards policy and guidance, and its enforcement responsibility.

The foundation of DO is an accountability framework that clearly distinguishes the roles and responsibilities of both Industry and FAA. Applicants lacking certification experience and many companies' use of numerous FAA designees have sometimes resulted in a blurred distinction between the showing of compliance by the Industry and the finding of compliance by the FAA.

The accountability framework principles, based on current regulatory requirements, are summarized below

F 1A planades aviation sufety by:	Imutog regulations
	Specifying the certification basis consistent with issued regulations
	Providing guidance regarding acceptable means of compliance
	Overseeing compliance
	Taking enforcement actions as necessary
	Issuing certificates and approvals after FAA finding of compliance
	Mandating corrective action as necessary
Applicants for a design approval have a regulatory obligation to:	Use means of compliance acceptable to the FAA.
	Show that their designs are compliant
Applicants for a production approval have a regulatory obligation to:	Establish an inspection system or a quality control system to support production efforts
	Demonstrate that they can produce products that meet the approved design
Design Approval Halders have an ungoing regulatory obligation to:	Maintain compliant designs with no unsafe Teature
	Report all known failures, malfunctions, and defects for their products

Design Organization Working, Group Part 21/SMS Availon Rolenaking Committee

III.A. Accountability Considerations

In the event of an issue where the FAA is required to take action, the action would be taken based on the obligations of the involved Design Approval Holder (DAH) and DO Certificate holder, as appropriate. For example:

• If an unsafe condition is identified in an approved product and is determined to be the result of a design deficiency by the applicant's DO or by an Agent DO, the FAA would address the product issue with the DAH as is done today (e.g., Airworthiness Directive (AD) issuance, etc.). In addition, the FAA would address the design deficiency with the DO and its systems.

The WG recommends the FAA consider the DO model impact on the existing Accountability Framework, particularly with respect to how a DA applicant, DAH, and DO are related.

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

IV. DO GUIDING PRINCIPLES AND ATTRIBUTES

This section defines the guiding principles and system attributes that form the foundation of the DO concept. They are the framework for the WG proposal for consideration by the ARC for recommendation to the FAA in developing DO policy. These guiding principles are intervined and must be viewed in their entirety.

The overarching objectives are:

- Define a regulatory framework by which the FAA can recognize a company's system and process capabilities for DA applicant and holder requirements.
- Leverage the capability of DOs allowing more efficient use of FAA oversight resources
- Consistency with the accountability framework.
- Continuous improvement in safety processes and compliance.

IV.A. Overview

The DO principles and attributes are defined in terms of overall objectives, not specific implementation. This is intended to allow flexibility in the creation of systems to address these objectives and to assure that the systems created may be of appropriate scale for the organization involved.

These principles and attributes may be applied to a range of DA applicant and holder scenarios [e.g. a small organization seeking approvals for a limited set of design modifications (defined by STC), an organization with a full line of products, including aircraft, or a small organization that holds a single TC for an aircraft no longer in production but with a significant fleet still in service]. This range of organizational sizes and responsibilities demands that the DO be scaled appropriately to the organization. It is obvious that a "one size fits all" principle is not appropriate for DO. The WG has approached this task by creating a single set of scalable requirements, with the expectation that each requirement would be assessed for effective implementation in the organization.

> Design Organization Working Group Part 21/SMS Available Rolenaking Committee

IV.B. DO Model and Framework

IV.B.(1) DO Model and Framework

The key entities in the proposed DO model are consistent with the Accountability Framework and consist of:

- 14 CFR part 21 DA Applicant required to meet obligations in part 21 to apply for and obtain the DA (certificate)
- 14 CFR part 21 DAH required to meet obligations outlined in part 21 as a DA holder
- DO Applicant
- DO Certificate Holder
- FAA issues DA certificates once a finding of compliance has been made that the product meets the airworthiness standards of the certification basis; certifies DOs and conducts oversight of DOs

The requirements and obligations of DA Applicants and Holders under CFR Part 21 are distinct. DA Applicants seek approval of a design and are required to demonstrate that the design meets the applicable airworthiness standards, whereas DA Holders are obligated to meet "Holder" requirements for reporting, mandatory corrective actions and Continued Operational Safety (COS) activities. This has the potential to lead to different DO requirements and capabilities for different phases of the DO operations.

In considering the implementation of a DO for DA applicants, the WG identified associated potential benefits in the form of a number of desired privileges. These privileges have the advantage of balancing the addition / implementation of DO systems within the organization, as well as providing benefits to the regulator. In the case of DAHs, there are limited associated privileges for an organization and as a result the benefits of a DO from an industry perspective are much more difficult to quantify. The benefits of SMS, as a required element of a DO, have not been sufficiently quantified to be considered by this WG.

•

IV.B.(1).(a) DA Applicant DO and DAH DO

The WG deliberated on whether DO should apply to DA applicants, DAHs, or both. This section provides insight into the deliberations prior to the WG reaching a consensus on the threshold applicability approach to be applied to DO (see Section IV.B.(3)). As such, it represents the considerations for DO under a mandated threshold approach.

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

DO for DA Applicants

The WG considered a risk-based approach to establishing requirements for a DO as a DA applicant. The model for incorporation of the DO with respect to applicants in certification process is proposed as follows:

- An applicant for a DA under 14 CFR part 21 will be required to have a DO if the DA being sought meets specific criteria (thresholds) as outlined in Section IV.B.(3)
- If an applicant does not have a DO at time of application, the applicant may:
 Make application to the FAA for a DO, or
 - Use an "Agent" DO to act on their behalf [see section IV.D.(5)]
- The DA applicant's DO must meet the requirements and have the systems in place outlined in Section IV.C that assure that any and all statements of compliance may be relied upon by the FAA when the FAA makes its ultimate finding of compliance by the act of issuing the DA certificate.

DO for **DAH**

As noted previously, the privileges and obligations of a DAH are not the same as a DA Applicant. In addition, there are a number of types of DAHs, many out of production and with no active DAH to support. For example:

- DAH in production, active or non-active applicant may require DO
- DAH out of production /some fleet, non-active applicant low risk No DO required

In particular, for a DAH, there are no associated identifiable DO privileges and the prirmary "benefit" of a DO is related to having an SMS that can support the Part 21 COS obligations, which would be most applicable for organizations that are DAH for products that are "above the threshold" per Section IV.B.(3) and are in production

In considering DO requirements for DA holders, the group reviewed the relationship between a DO and SMS and the need for specific DAH's to have a DO and / or an SMS. As a result, two possible options were identified with associated pros and cons. An outline of each option follows:

Option 1: DO required to apply SMS to DAH

Assumptions:

- · SMS is needed for some DAH above threshold / meeting certain criteria
- DO is needed to apply SMS to DAH

Model:

"Holder" DO will capture DAH above threshold / meeting certain criteria



- DAH DO Threshold / specific criteria is established based on risk and need to have an SMS
 - Potential additional criteria: in production / number of products in service / product risk level
- Holder DO capabilities / functions / privileges are specifically identified vs Applicant DO capabilities / functions / privileges
- Could have a Holder DO with no (embedded) Applicant DO

Rationale:

- DO provides an "operational" certificate to which SMS can be applied as a condition no other (identified) approach to apply SMS
- Part 21 plus + 21.605-like model is not equivalent to full SMS so may not be adequate as an alternative for DAH that would be "above the threshold" from a risk perspective
- Applicant and Holder capabilities / functions / privileges are not the same and can't assume Applicant DO & SMS can/will apply to Holder aspects.

PROS :

- Having DO makes it easier to make sure SMS aspects are applied to the DAH responsibilities; No need to try to make "21.3 Plus" work for higher risk DAHs
- Makes DAH transfer scenario easier since if DO required as DAH, new DAH should also have DO
- Aligned with other authorities' approach with respect to DO for DAH and SMS (Harmonized) and facilitates bilateral acceptance

CONS:

- Cost / benefit:
 - hard to justify to add DO and SMS to DAH (no quantifiable "benefit" to industry similar to additional Applicant DO privileges and SMS benefits not yet quantified
 - Impacts "value" of DA for transfer / sale
- Threshold:
 - Challenge to establish
 - Not exactly same as applicant needs to be reviewed
 - Adding additional requirements and complexity to the system / reg to implement
- Could have retro-active effect if not properly implemented

Option 2: DO and / or SMS for a DAH not required

Assumptions:

- DO is not needed to apply SMS to DAH
- "Full" SMS not required and minimum SMS elements could be applied through some other mechanism to a DAH

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

• Organizations will voluntarily implement DO and SMS

Model:

- Applicant DO will capture large organizations above threshold
- DO = Design Org and there is no need for a <u>design org</u> as a DAH, except to do design changes, where an (applicant) DO would then be invoked.
- Part 21 plus + 21.605-like will be sufficient to cover DAH COS aspects for all types of DAHs
- A DA applicant / holder that is required to have an Applicant DO (which includes SMS) should apply the SMS to the entire DA / part 21 aspects, including holder obligations

Rationale:

- As a <u>DAH</u>: There are no obvious privileges or demonstrated benefits to industry to have an SMS with / without DO
- SMS cost / benefit to industry not established and difficult to justify required changes and additional processes / elements
- SMS cost/ benefit to FAA would need to be further examined and quantified

PROS:

- Cost / benefit easier
 - No need to explain lack of privileges and positive benefit for DO and / or SMS on DAH
 - Retains "value" of DA from Transfer perspective
- Abandoned / surrendered DAs:
 - Makes possibility to have another organization be DAH, since no DO/SMS applied [Maintains "value" of DA]
- No need to define DAH DO Threshold

CONS:

- If SMS is required, challenge to apply SMS since no DO. No approach yet defined to apply SMS to DAH (which is NOT an operational certificate) without a DO
- Could lead to DAH Applicants (> threshold) to immediately transfer DA to avoid DO / SMS
- Relies on reviseed 21.3 / COS regulations to be sufficient to meet intent of SMS requirements [SMS WG / 21.3+ yet to be tasked WG]
- Would require concept of Agent DO to be in place to allow non-DO orgs to do (above threshold) design changes
- Possible issue with respect to FAA action and non-compliance to (new) Part 21.3+/COS requirements for SMS aspects / elements linked to (product) DA versus an operational certificate (DO). Would penalize product operators, not DAH
- May not be harmonized with other authority approach and could impede bilateral product acceptance and / or validating authority SMS review of DAH

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

WG Consensus:

Based on the cost / benefit challenge and the possibility to apply any required SMS elements to DAHs via enhanced Part 21 requirements or alternate approaches, the group recommends Option 2 above as the model, with the additional considerations as follows:

- No specific DO threshold for DAH, as the Applicant DO will capture large organizations above threshold
- A DA applicant / holder that is required to have an Applicant DO (which includes SMS) will be required to apply the SMS to the entire DA / part 21 aspects, including DAH obligations
- Separate, yet to be tasked, WG recommendations for all DAHs will be sufficient to cover DAH COS aspects for all types of DAs (anticipated to include 21.3+, 21.605-like recommendations)
- If a DAH does not have a DO, the Holder may use an "Agent" DO to act on their behalf [see Section IV.D.(5)] for 14 CFR part 21 Applicant or Holder activities

Dissenting Opinion

One industry member of the WG dissented with the decision to recommend Option 2, with the following comments:

- If an SMS is to be applied to DA entities, the greatest benefit, from a safety enhancement perspective, is for DAHs particularly in the area of COS activities. It seems counter-intuitive to have a DO with an SMS applied to a DA applicant, (based on a successful cost benefit analysis) yet not apply it to certain DAH above the threshold.
- It can not be assumed that all DAH are going to be "compliant" and voluntarily adopt an SMS. Experience of other authorities is, if SMS is deemed necessary, a non-mandatory (or voluntary) approach to SMS is not that effective in capturing all organizations that should have an SMS. Many organizations will pro-actively adopt the SMS and gain the benefits, but many will not.
- There is an assumption that if SMS needs to be mandated, it can be applied to DAHs via another mechanism other than an "operating certificate" such as an DO. However, no other approach has been determined (yet) by other authorities for DA entities. If the SMS is made a condition of being a DAH, and the FAA needs to take some certificate action against the organization due to SMS issues, the certificate action is against the specific design / product and impacts the operators of the product, not just the DAH.
- There is a potential risk of reduced bilateral acceptance of products that have not been designed / certified / produced under an FAA-accepted SMS and the possibility that other authorities will want to review an organization's voluntary

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

SMS if it is not explicitly FAA "endorsed". It is unlikely that the FAA will oblige FAA resources to "endorse" voluntary SMS. It may be possible to have third party "endorsement" but this may also be subject to lack of acceptance by other authorities and additional cost (to pay third party org.)

 Flue overall industry perception of SMS as a burden due to the difficulty in regulator implementation approach and required cultural changes is not unwarranted, however, the bonofits of a having an operational and recognized SMS are not well enough understood and are currently underestimated in terms of incident / accident / liability cost avoidance.

IV.B.(2) A DO Certificate is Not a Delegation

A DO certificate is not a delegation as defined under section 44702(d) of Title 49. While it provides functions similar to those available to designees and delegated organizations, the DO's ability to perform these functions is a privilege of its DO certificate as defined under section 44704(c) of Title 49 and not delegation.

While DO is a significant regulatory change, it is only astep in the maturing relationship between the FAA and Industry related to product certification compliance determinations. The LAA is still responsible to avaluate the capability of the DO and us determinations of compliance to monitor and audit these determinations, and to issue certificates. In addition, the ability of the Administrator to amend, modify, auspend, revolve, or otherwise after the DO certificate as specified in sections 44709 and 44711 of Title 49, is not affected in any way

IV.B.(3) Eligibility and Thresholds

Eligibility

All organizations who meet the minimum requirements (see Section IV C) are eligible to apply to become a DO DO does not require a FAA determination of need. Not does a DO application require previous FAA approval experience

No Thresholds, DO is optional

One of the key aspects of the United States (US) available initiative is the continuous innovation and ability to foster entrepreneurs to develop the next generation of safety improvements. Therefore it was important to the WG to establish a "lower overhead" path that would allow for "low risk" innovation and ver leverage the strengths of established organizations for "logier risk" designs.

Industry's recommendation is that there be no mandating thresholds for any DA applicants of DAHs. All DOs would be optional. Benefits of this approach are that

> Design Organization Working Group Part 21/SMS Available Rollenaking Committee

each DO applicant would be making the decision to become a DO based solely on the cost/benefit (business case) of that organization. This approach would be dependent on the benefits of the privileges outweighing the costs of implementation and maintenance of DO.

The "Mandatory Thresholds" described below could still be included, but changed from DO Required to DO Recommendedor DO expected, as guidance.

Mandatory Thresholds (if required)

In the event that the "No threshold, all DO's are optional" recommendation is not accepted. The WG developed a set of logical thresholds that could be used to minimize overhead where not required, and leveraging the strengths of DOs.

The WG explored various methods of developing thresholds to define which types of DAs would be above vs. below the threshold. The threshold structure was developed in an attempt to capture the highest-safety risk DAs, while allowing the lower-risk DAs to be handled by the current "FAA Managed" process.

The WG desired to develop thresholds that could be completely objective and evaluated without detailed analysis or interpretation. However it quickly became apparent that a more nuanced approach would be required to meet the intent of the thresholds. Various threshold models were evaluated. (See *Other Threshold Models Discussed* below for details on the other models.) Ultimately the WG settled on a Primary/Secondary (Risk Based) Threshold model as a logical approach.

Threshold Decision Tree

The decision tree (see Appendix F) is intended to be an easy-to-follow process to determine whether a project is above or below the threshold. The intent of the decision tree is only to determine if a DO is required to work the project. There are only two possible outcomes of the decision tree:

- 1. DO is NOT required, FAA-Managed or optional DO are acceptable
- 2. DO is required for Project, Develop DO (or have access to DO) w/ DA Application

Note that the projects worked with a DO and with the "FAA Managed" process would be required to meet the same airworthiness standards. DAs obtained through a DO or "FAA Managed" processes would have the same approval status and acceptance.

Existing DOs

The WG expects applicants who are already DOs to work any new project through their DO. If the DO makes application for a project beyond its existing authority,

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

the WG would expect the DO to work with the EAA to expland its existing authority.

Primary Threshold

The primary thresholds were developed to be completely objective. See Appendix *U*, Figure P-1 for the primary threshold decision tree. The primary threshold is designed to be a coarse screen for those projects that would be coally identifiable as "low on "high" risk. This section discusses some of the rationale for the decision points.

New TC applicants

Projects that present a small enough risk/population that they would not require a DO

- 1 Part 31 (Manned Free Balloons)
- 7 Part 21 24 (primary category aircraft).
- 1 Part 21 25 (restricted category aircraft)
- 1 Part 21 27* (surplus aircraft of the Armed Forces).
- 1 Part 33 (engines) & Part 35 (propellers) intended for Part 21 XX Aircraft.

*Rotorcraft Directorate asked to exclude rotorcraft from the Part 21.27 exemption. The reasoning bound this request was due to the high volume of safety issues and ADs being issued on the large number of rotorcraft being pulled from military surplus boneyards and applying for Restricted Category TC using 14 CFR part 21.27. These applicants are using old maintenance manuals that are inadequate and outdated. However, the local FAA offices lack the knowledge of the history with these aircraft and issue the Restricted TC. The Rotorcraft Directorate wants to require applicants for Restricted Category TC using 14 CFR part 21.27 to be or use a DO. The benefit of this would be that the DO would require the applicant to have a better engineering staff than what they currently have, and it would also have an improved COS Program. It is the opinion of this WG that this issue needs to be addressed outside of this WG.

STC. PMA and changes to TC. STC or PMA

Projects that present a small enough risk/population that they would not require a DO

- 1 Part 31 (Manned Free Balloons)
- 2 Part 33 (angina) non-turbina
- 7 Part 35 (propeller) fixed prich
- 4 Part 23 Small Airplane
 - a =9 Passenuors.
 - b Not certificated as a commuter category or predecessor commuter category airplane (Special Federal Aviation Regulation (SFAR) 23 and SFAR 41), AND
 - c. Not pressurized and certificated to fly at altitudes 24,000 ft or above

Design Organization Working Group Part 21/SMS Aviation Rolenaking Committee
Note: The Small Airplane Directorate's primary threshold criteria are consistent with the safety continuum work associated with the concurrent Part 23 ARC

3. One Time Only STC (and changes to a one time only STC) Since, by definition. One Time Only STCs apply to only one product (arcraft, engine or propeller), the risk associate with the STC will be limited to one arcraft.

Note: Any attempt to apply a One Time STC to multiple different products should be discouraged.

- Non-Critical PMA (and changes to Non-Critical PMA)
- Every PMA applicant has to make a safety assessment determine if a part is critical or non-critical (See FAA Order 8110.42). Therefore this is a well underatood and accepted process and can be used as part of the primary threshold.

Secondary "Risk Based" Threshold

For Multiple STC and changes to PC or Multiple STC not captured by the primary thresholds, the threshold decision true directs the applicant evaluate the secondary "Risk Based" threshold. (See Appendix F. Figure F-2.)

Note that in order for a project to be evaluated by this secondary threshold, it must have passed through the primary threshold flowchart without a "DO is required" or "DO is NOT required" determination

As stated above, the WG desired to develop the thresholds that could be completely objective and evaluated without detailed analysis or interpretation. Since the fidelity sequired to get to a good determination of product safety risk was not possible with the purely objective approach, the Failure Modes Effects (FME). Assessment was chosen. (See Other (Threshold Model's Discussed below for details on the other models examined by the WG.)

Most applicants make a formal FME Assessment (or Analysis) at some point in time fairing the development of the design change. In most cases, this FME Assessment (or Analysis) is part of the formal application or change record. For some CFR 14 Parts and disciplines this FME Assessment (or Analysis) currently has a regulation, order and/or guidance material. Others unfortunately do not.

In the interest of making the FME Assessment as transparent and well understood as possible, the WG proposes to use the current regulations and guidance wherever possible, and recommends the various directorates work with industry to develop regulations and guidance where it is currently absent or nuclear.

The secondary risk-based threshold is designed to be a more numced screen for these projects that would be identifiable as "low" or "high" risk. This section discusses some of the rationale for the decision points.

Dengii Organizatian Working Group Part 21/SMS Avanton Rulemaking Committee

Existing FME Assessment regulation/guidance

As stated above, the WG desired to use existing regulation and guidance material where it is available. The following are examples of this material:

- 1. Part 27/29 Structures
 - a. Part XX.571(b)(2) & Part XX.573(c)(2) Principal Structural Element (PSE)
 - b. Part XX.602 Critical Parts
- 2. Part 23/25/27/29 System and Equipment
 - a. Part XX.1309 Equipment, systems, and installations
- 3. Part 33.75 Safety analysis
- 4. Part 35.15 Safety analysis

The part 27/29 structures assessment is relatively simple. If the change affects a PSE or Critical Part, then "DO is required". If not, "DO is NOT required."

For the Part 23/25/27/29 System and Equipment disciplines and Part 33 and Part 35 assessments, the applicant only has to make an <u>assessment</u> as to the failure modes and effects. If the potential failure mode of the change would introduce a new failure mode(s) or otherwise impact the existing failure mode(s) which could result in either Hazardous or Catastrophic classification, then "DO is required." If not (i.e., the potential failure mode is Major or less), then "DO is NOT required."

Note: the regulations and guidance material provide for not only an <u>assessment</u> but also an <u>analysis</u>. The <u>assessment</u> provides the category of potential failures. The <u>analysis</u> provides the probability of occurrence for the potential failures. For the purposes of threshold determination ONLY the <u>assessment</u> is required.

The WG expects "below the threshold" applicants to include their FME assessment with their application/proposal for any FAA managed projects. Since this is currently done with most applications, the WG does not expect this to be an undue burden on "below the threshold" applicants.

Areas without existing FME Assessment regulation/guidance

The WG recommends encouraging the various directorates to work with industry to develop regulations and guidance for any areas where existing FME Assessment is absent, or unclear. There should be a clear delineation between "High risk" and "Low Risk", similar to the Hazardous/Catastrophic vs. Major or less that is currently used for other directorates/disciplines.

Other disciplines and Product Level Safety Assessments

The WG explored adding detailed discipline-level FME assessments and other secondary criteria. The WG recommends that the secondary threshold be limited to <u>Product Level</u> Safety Risk Assessments. Any "other" secondary threshold assessment should be set using the criteria of will the potential failure mode "affect

the continued safe flight and landing of the aircraft." As an example, the WG explored occupant safety assessments. While occupant safety and crash survivability are important and there are specific regulations written around these areas, they do not rise to the level of "continued safe flight and landing."

Any applicant for a STC (or change to a TC) would be expected meet all the regulatory requirements whether they use a DO or an FAA-managed approval process.

Risk of Under/Over Classification

If an applicant makes an error in the application of the secondary "Risk Based" threshold, the proposed system has a robust 'fail safe' mechanism built in.

If the applicant <u>under classifies</u> the FME Assessment and incorrectly determines that "DO is NOT required," then they would submit that assessment as part of their FAA-Managed project. The FAA would then have the opportunity to review the assessment and correct the error. The applicant would then be required to become a DO or to obtain access to a DO to process the application.

If the applicant <u>over classifies</u> the FME Assessment and incorrectly determines that "DO is required," then the applicant would (on their own) become a DO or obtain access to a DO to process the application. A fully-functioning DO would be able to detect the error and correct the FME assessment.

Optional DO

Because of the privileges associated with a DO (see section IV.E), non-DO Applicants who are operating below the thresholds may desire to become a DO. Such applicants would be treated in the same manner as any other DO/DO Applicant. (As discussed above, there is no "need determination" required to become a DO.)

Other Threshold Models Discussed

Major/Minor Change

Major vs. Minor design change would be a convenient delineation for setting a DO threshold. Unfortunately, the current state of the industry and guidance does not allow for an easy, consistent determination of Major/Minor change, nor is the Major/Minor threshold necessarily safety risk-focused. For example, adding a placard in the lavatory of a Part 25 aircraft may be classified as a Major Change if no placard currently exists in that location. The WG noted many more examples where Major/Minor change determinations have been controversial, with disagreements between FAA and industry. For these reasons the Major/Minor change was not selected for the threshold determination.

While Major/Minor determinations are out of scope of this WG, the WG

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

recommends that the FAA and industry drive towards a consistent, easily understood, safety risk-based determination of Major/Minor change.

Significant Change

Similar to Major/Minor, Significant Change would also be a convenient delineation for setting a DO threshold. Unfortunately, the current state of the industry and guidance does not allow for an easy, consistent determination of significant change. The WG noted many examples where significant change determinations have been very controversial, with disagreements between FAA and industry. For this reason the significant change was not selected for the threshold determination.

While significant change determinations are out of scope of this WG, the WG recommends that the FAA and industry drive towards a consistent, easily understood, safety risk based determination of significant change.

<u>Proxy Thresholds</u>: Substitute objective thresholds (such as weight, thrust, altitude, passenger count) for the risk-based secondary threshold. Benefits of this approach are that the determination of above vs. below the threshold would become completely objective and not dependent on analysis or interpretation. Drawbacks of this approach include that some "low risk" changes will be caught by the proxy thresholds, and that some "high risk" changes could be below the threshold.

Technical Standard Order Approval (TSOA)

TSOA Applicants/Holders would not be required to have a DO. The WG identified no appreciable benefits or privileges that a TSOA Applicant/Holder would attain by becoming a DO that they do not currently have as a TSOA Holder/Applicant nor would there be a reduction in FAA resources required. Therefore TSOA Holder/Applicants were determined to be out of scope, since a simple cost-benefit evaluation was perceived as negative.

Dissenting Opinion - DO requires previous FAA approval experience

A member of the WG dissented with the eligibility statement that DO application does not require previous FAA approval experience. The dissenting opinion is that an applicant for DO should be required to demonstrate capability as a mature Organization Delegation Authorization (ODA) before being granted a DO certificate. This approach provides experience in building toward a DO certificate.

Dissenting Opinion – Secondary risk-based mandatory thresholds

A member of the WG objects to the manner in which the mandatory thresholds are established; particularly to the secondary threshold mechanism which uses an overly complicated analysis that is subjective and that makes it difficult for new market entrants to identify their regulatory obligations.

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

The methodology posits a primary threshold, which would be based on easily ascertained objective criteria. But it then continues to posit a secondary threshold, which would be based upon an analysis that is analogous to a failure modes and effects analysis (FMEA).

The audience for this threshold distinction won't be large and well-established companies. Their participation in the WG suggests that these sorts of companies would voluntarily obtain DO Certificates regardless of whether there is a mandate. Rather, the audience for this threshold distinction will be new market entrants seeking their first DA. These are more likely to be small businesses, with limited resources and limited experience. Such companies may not have past experience in performing FMEA, so mandating that they perform an analysis equivalent to an FMEA on their design idea just to determine whether or not they need to perform the design process under the auspices of a DO Certificate seems ill-advised.

Furthermore FMEA is typically performed in the context of an existing design. The engineer is identifying failure modes and effects based upon the design features. But the FMEA posited by this secondary analysis would not be performed by analyzing a design. Instead, it would be performed by analyzing an idea that has not yet been designed. It would need to be performed at such an early stage because it would be a vital part of the analysis that drives the prima facie question of whether the design work needs to be performed in the context of a DO Certificates. So new companies would be caught in the catch-22 of trying to perform FMEA on a part that has not yet been designed in order to compare the FMEA to the secondary threshold metrics, and being unable to perform the design work until first receiving an answer to the prima facie question of whether a DO Certificate was needed.

Finally, there is no need for a complicated mechanism for identifying whether a company needs to hold a DO Certificate. It is possible to establish an objective standard that serves as a proxy for the safety thresholds upon which the WG wants to rely.

Under the discussion of Existing DOs in this Report, the WG has asserted its intent that existing DOs would work all new projects through their existing DOs. Thus, the threshold question of whether you need a DO is an important one that has an effect on all future projects from that company – and companies may be dissuaded from pursuing complex projects if they fear that this sort of complex project would adversely affect the resource requirement for all future projects.

It is recommended to retain the primary threshold only; because the companies that would be asked to perform the risk assessment are the ones that have the least experience in performing such analyses, and the fact that they would need to predict failure modes and effects based on incomplete engineering means that they will not have adequate data upon which to make their FMEA predictions. The primary threshold adequately serves as a proxy for the safety standards that are necessary to establish the DO Certificate threshold.

It is also strongly recommended that all threshold enteria related to this new certificate (like eligibility and necessity) be established using objective standards. For example, it is objectively obvious to an applicant whether their aircraft parts concept is intended for use on a Part 25 aircraft, and thus a criteria that uses Part 25 is likely to be both objective and observationally simple.

IV.B.(4) Certificated Organization

The WG discussed what form of recognition a DO should be provided and determined that a DO should be a certificated organization. The alternative options discussed included approved, authorized, and accepted organizations. The WG agreed that a certificated organization aligned best with the accountability framework model.

Design Organization Certificate

Like any other FAA certificate, a DO certificate comes with privileges and requirements. This is similar to how an air carrier operating certificate is structured. Section 119.7 of i4 CFR indicates the kinds of authorized operations that constitute a part of the operations certificate. Similarly, the types of activities authorized under a DO certificate, including the types of covered products and services, would constitute a part of the DO certificate.

Not all DO certificate holders will have the same scope of authority-

DO Certificate. All persons who meet the minimum requirements are qualified to be issued a DO certificate. DO does not require a FAA determination of next.

The DO certificate holder has the right to retain the certificate unless the FAA takes certificate action in accordance with 40 United States Code (USC) 44700

IV.B.(5) Applicability of Part 21 Requirements

There are several places within 10 CFR, chapter 1, subchapter C, most of thim being in Part 21, where the principles embodied within the DO concept conflict with existing regulations.

The WG saw two possible avenues available to resolve those conflicts

- Change each rule wherein a conflict occurred, or
- Place a single section within the DO requirements that contained the MOC by which the DO would comply with the conflicted rules.

Design Organization Working Group Part 21/SMS Availing Relenaking Committee

In no case would the compliance objective of the conflicted rule be changed. The WG chose the latter approach consistent with the CDO ARC, as it results in all the DO requirements residing in one place within Part 21 Refer to the CDO ARC report proposed NPRM (e.g., paragraph 21 741)

It is important to recognize that the DO certificate holder, like any other applicant or certificate holder, must comply with the other requirements of 14 CFR Part 21, unless otherwise stated or otherwise excluded.

IV.B.(6) Transition Provisions

A WG subteam addressed transition provisions prior to forming the consensus opinion of the threshold applicability for a DO (i.e., no threshold) The following information is presented should a dissenting opinion that DO become a mandatory requirement be further considered.

Should a role for mandatory DO becomes effective there should be a transition period to allow applicants time to meet the new requirements. The WG made the following assumptions about this transition period.

- All parties will act in good faith.
- The effective date of the rule will be a date one to two years after the rule's publication date
- There will be no link of requirement between TC/STC/PMA issuance and DO certificate issuance.
- Issuance of a DO certificate should be within five years of date of DO application.
- There will be no mandated date requiring new applicants to have a DO certificate. New applicants may apply for TC/STC/PMA and a DO certificate concurrently
- A process needs to be in place to address applicationa/projects that started prior to effective date of new rule, but not yet finished. This is only required if the applicant decides they want to apply for a DO certificate otherwise all applications submitted prior to effective date of the rule may continue as FAA-assisted applications.
- For TC/STC/PMA applications. ODA will not exist. This might require a "sunset" ODA regulation

The WG proposd DO certificate application requirements with the following progression

Applicants applying for a DA (TC/STC/PMA) above the threshold after the
effective date of the rule AND (a) do not already fold a DO certificate OR.
(b) are not utilizing an agent DO will be required to

Design Organization Working Group Bin 21/SMS Aviation Rulemaking Committee

- Sobrmi application for a DO certificate (a plan should be submitted with the application, but a proposed procedures manual 15 not required at this point).
- Submit DO procedures manual to the FAA within 2 years of initial application date.
- Within 3 years of DO procedures manual submittal, the FAA should have a means to approve the manual and issue the DO cartificate.

IV.C. DO Minimum Requirements

As previously discussed, a DO must have systems in place that assure any and all statements of compliance may be relied on by the FAA when making its ultimate finding of compliance by the act of issuing a certificate. That activity must occur under established minimum requirements of a DO.

An integrated and systematic approach to compliance and safety therefore encompasses several elements, including an organization, CAS and SMS. Successful execution of these elements should enable the continued growth of a compliance and safety culture within the DO. While each of the elements of the systems discussed below need to be satisfied by the DO, they may be arranged or grouped differently or encompassed within organizational systems with different names to accomplish the intended purpose.

The DO must be able to establish and show that the organization;

- possesses the required competence to determine that the certificate holder's designs meet all applicable airworthiness standards within the scope of the DO certificate,
- has procedures for assuring compliance to the airworthiness standards, and
- maintains the essential process controls to deliver repeatable and sustainable compliance.

IV.C.(1) DO and OCS

The WG does not recommend that a particular organizational structure be required; however, there are certain functional roles, aspects and elements that must be defined in the FAA-approved DO Procedures Manual. Each holder of a DO certificate must have a Procedures Manual that defines the procedures and processes to be used (i.e., OCS) which meet the requirements specifically required by the DO regulation to be in the procedures manual.

The holder of a DO certificate must follow the procedures in the manual. The manual must be in the English language and retrievable in a form acceptable to the FAA.

The DO Procedures Manual contains the DO organization's procedures for meeting its regulatory requirements. The manual must address all relevant DO requirements.

The DO Procedures Manual processes and procedures must be sufficient for the FAA to determine that regulatory compliance is properly addressed. The manual is intended to be a top level document that will guide the development of lower level processes and work instructions that the DO can develop and change as it finds necessary (i.e., without the need for FAA approval) to meet the top level

requirements and objectives. While these lower level process documents will not be FAA approved, they must be cross-referenced to the procedures manual. These lower level processes and procedures are auditable by the FAA. If the DO fails to comply with any procedure contained or referenced in the procedures manual, this non-compliance could result in enforcement action. This means that all lower level processes and work instructions within the DO that are related to compliance must have a means to tie them to the FAA-approved Procedures Manual. Internal company processes and procedures that are not required to show DO regulatory compliance would not be referenced in the procedures manual and would not be auditable by FAA as part of its DO oversight.

If the FAA determines that the procedures manual lacks the detail necessary to ensure regulatory compliance, the FAA will request a change to the manual. The DO is obligated to respond to FAA's request within an agreed upon time frame. The WG recommends further discussion on the necessary level of detail for inclusion in the procedures manual and the appropriate reasons/rationale for FAA requests for changes to the procedures manual.

The procedures manual must be consistent with all issued FAA regulations and guidance related to the proper functioning of a DO. The manual may not be used by the FAA to apply policy that has not been formally implemented through a public process. The certificate holder may not use the manual for relief from any regulatory requirement or to create unique policy for its sole benefit.

The DO Procedures Manual may be in any format proposed by the DO and acceptable to the Administrator. There is no expectation that each DO procedures manual would be formatted the same.

The organizational and OCS requirements are:

- (a) Identified DO Executive. The DO Executive is accountable for all the activities covered within the scope of the DO certificate. This executive must be identified by name and position within the company. The DO Executive may also act as the primary point of contact (POC) for the DO. The DO Procedures Manual must contain an explanation of the reporting relationships between the DO Executive and senior company management, as well as the organizational relationships within the DO.
- (b) Identified DO point(s) of contact. The DO POC is the person(s) within the DO with whom the FAA will communicate. A formal list of POC must be maintained by the DO. The DO POC must have familiarity with the DO processes and the applicable FAA regulations consistent with the scope of the DO certificate. The DO POC must also have unencumbered, but not necessarily direct, access to the DO Executive. Additionally, defined procedures for communication between the DO and the FAA, including agreement on expectations and expediency, shall be stated.



(v) Each DO certificate holder shall have qualified staff as appropriate to the DO privileges and obligations. The DO is responsible for ensuring the staff in technical departments have the experience, training, and authority to be able to discharge their allocated responsibilities, and that these, together with the accommodation. Facilities and equipment, are adequate to enable the shaff to achieve the airworthiness objectives for the product.

Note: The intent is for the DO to preserve autonomy in defining the structure and number of staff in stignment with its business model. The intent is not to require identification by name of staff in the procedures manual, however, it is intended to provide a means to trace determinations of compliance to qualified staff. While the WG dat and discuss details regarding staff qualifications, a WG member suggested DO staff qualification requirements should be addressed prior to rulemaking activity.

- (d) A process for verification of personnel qualifications. The scope of personnel is intended to focus on those people who hold accountantity for maintaining the organizational oversight of the DO and those people with prime accountability for the DMS, CCS and SMS. Personnel qualification includes essential competencies, experience and training.
- (ii) A process for verification of systemic performance of duties. The includes procedures for planning, conducting, and documenting internal audits to ensure compliance with the approved Procedures Manual, DMS CCS and SMS. The procedures must include reporting of internal audit results to the manager responsible for implementing corrective and preventative actions.
- (f) A process for retaining records that are required to be produced. This includes the identification of records that must be retained, the method and means of storage and retrieval, control and access privilege, and retention period. These records typically comprise DO Procedures Manual approvals (including changes to), DA records, approval of design changes internal multi records and CCS records.
- (a) A process for identifying which certification projects require an application, for establishing details of the project list, and for how offenthis information should be provided to the FAA. (Refer to Section IV/C (4))
- (b) A process for notifying the FAA if circumstances prevent the DO from meeting the DO obligations. Notification is not required if the circumstances have been anticipated and mitigated. This process includes agreed expectations pertaining to communication and notification protocols between the organization and the FAA regarding identification and investigation of potential for actual i non-compliance.
- The process, timetable and authority for obtaining and agreeing on changes to the DO Procedures Manual, DMS, CAS, and SMS.

Design Organization Working Group Bin 21/SMS Astanon Rulemaking Committee

IV.C.(2) Compliance Assurance System

The DG applicant must demonstrate that it has established and is able to maimain a regulatory CAS for

- the control and management of the DA(a).
- design changes of products and articles covered by the scope of the certificate, and
- any production activities associated with those DAs.

The CAS should result in assurance that the compliance determinations are correct and consistent with what would result from an independent skilled review of compliance

The DO applicant's regulatory CAS must contain a means to provide assurance that the design and design changes of the applicant's products and articles comply with the applicable airworthiness standards. This is presented in this report as being accomplished through a DMS and a CCS, details of which are contained below.

As appropriate to compliance and safety, the compliance assurance process for the various aspects of the product or article design may include safeguards, risk-based assessments, and/or checking functions of the showings of compliance. As an example of a safeguard, a computer-aided design system could preclude designers from inadvertently selecting materials that had not been qualified by the DO as compliant with the airworthmess standards. In these instances where safeguards have been implemented or where a risk-based assessment has been accomplianed, a checking or verification function may not necessarily be required.

After the issuance of the DO cortificate, any changes to the CAS materially affecting compliance with the cortification basis or airworthiness standards must be submitted to the FAA for approval, prior to implementation. The DO must identify to the FAA how the proposed changes to the CAS will result in continued compliance after implementation.

Assurance of compliance with the airworthiness standards, including the performance of suppliers, is of critical importance to the success of DO. As will be discussed in a later section, there are no FAA designees used by the DO. Therefore, the quality of the DO's processes for determinations of compliance and process adherence, and the robustness of the CAS are the basis for enabling the FAA to rely on the DO's statement of compliance when making its finding.

The DO CAS is comprised of two elements:

 Design Management System. The DMS is the system by which a DO creates and maintains product design data. The DMS requirements are.

> Design Organization Working Group Pan 21/SMS Availon Rolenaking Committee

- (a) A process for creating, and maintaining configuration and control of, design data. The establishment of a structured and controlled system for the development of design data, the control of changes to, and the assurance that the descriptive design data is current and approved is fundamental to this process.
- (b) A process for retaining, retrieving, protecting and maintaining design descriptive and substantiating data. This process should also include any agreements between the DO and the FAA regarding availability of data, access and any electronic system requirements required to view the data.
- (c) A process for engineering supplier control that defines how design activities performed external to the DO are controlled, validated, and assured. This also includes Inter-DO relationships (see Section IV.D.(4)).
- (d) A process for creating eligible data that can be used for showing compliance when applied to a particular certification activity.
- (e) A process for receiving and processing "safety" data related to approved designs.

Note: "Safety" data is meant to pertain to information relevant to identification of product safety hazards, risk analysis and mitigation, as appropriate.

- (f) A process for applying relevant "safety" data to the DMS for purposes of continuous improvement.
- 2. **Compliance Certification System.** The CCS is the system by which a DO assures product compliance (showing of and verification of, as appropriate) to the applicable airworthiness standards. The CCS requirements are:
 - (a) A process for identifying and/or establishing regulatory certification basis. This process includes an assessment of the product's intended usage and determination of the applicable airworthiness standards. The process should also include protocols for coordination with and, if required, agreement from the FAA. For changes to existing DAs, an assessment and determination of the changed product rule shall be performed. The establishment of a product's certification basis may be a pre-decisional FAA approval based on the agreed processes defined within the DO Procedures Manual.
 - (b) A process for identifying regulatory changes that affect the design system. The DO must have an active means of monitoring regulatory changes that affect the product(s) for which the organization is responsible and evaluating the need for implementing changes or taking other actions. This includes any activities resulting from the issuance of an AD against a DAH's product.
 - (c) A defined system that shows compliance to the applicable airworthiness standards under 14 CFR 21.20(a) and how verification of compliance will be conducted including any process for assuring independence and objectivity. The "show" responsibility is identical to that already prescribed under 14

CFR 21.20(a). However, in the DO model, the "show" and "verification functions are intended to be accomplished within the DO

(d) A process for using MOC that are acceptable to the FAA, including, as appropriate, a process for verification and validation of analytical tools and a process for managing testing and inspection required to support each showing of compliance. This would include conformity inspections used to support showings of compliance.

It is expected that this process will define the between showings within wellestablished MOC and procedural distinctions when addressing a MOC that is new or novel.

- (b) A process for declaring to the Administrator that a design is in compliance with the applicable account finess standards at the time that the DA is requested. This documentation fulfills the requirement under 14 CTR 21 20(b).
- (f) A process for reporting other approvals such as design changes, changes to Instructions for Communed Airworthiness (ICA), and approval of repair data.
- (g) A process for receiving, processing, and implementing corrective action with respect to "compliance" data related to approved designs. If during the course of regular business or as the result of a finding during any internal or externally conducted audit a non-compliance to the approved procedures or airworthiness standard is identified, the mechanism for evaluating the risk of the non-compliance and identification of any corrective actions must be defined.

DO show and verify compliance vertification functions. The CCS processes must include a showing of compliance by the DO to each applicable airworthiness submard. The following principles provide guidance in this regard

- Show and verify functions are part of the CCS
- The show function is a demonstration of compliance to an airworthiness standard (i.e., 14 CFR 23, 25, 27, 29, 33, 35) and is comprised of substantiating data, statements, and/or other acceptable methods of demonstration (c.g., acceptable MOC).
- The verify function is an independent check, or equivalent, of the show function
- The show and verify functions are distinct functions or activities where appropriate. The degree of independence between the individuals performing the show and verify functions or activities should be established in the DO Procedures Manual
- A DO determination of compliance is a showing with an independent verification function unless the DO Procedures Manual identifies a specific area where a verification is not required or where the Procedures Manual

identifies a process that allows the DO to assess risk to make decisions on whether a verification step is required. As such, a mathematical expression for a determination of compliance may be considered as the following: determination of compliance = showing of compliance + verification of compliance, as appropriate per Procedures Manual

• The showing of compliance requirement remains identical to that which exists in the current regulatory system. There is no intent to establish a changed requirement for showing of compliance for a DO.

Tools used for performing compliance activities. Where the system is dependent on the use of a tool for performing some of the processes and methods, means must be provided to ensure:

- The tool performs it required function;
- The tool and its output are being controlled under a configuration management program;
- The tool is periodically verified for its applicability with respect to the processes and methods for which it is intended to apply; and
- A record is kept of the use of the tool to accomplish the compliance activity.

IV.C.(3) Safety Management System (SMS)

The SMS requirements for a DO are to be provided in a separate report by the SMS WG of the Part 21/SMS ARC.

The DO WG anticipates the SMS to be the system for actively monitoring product safety, identifying and managing risks to product safety, and promotion of a strong safety culture throughout the organization, which would likely consist of;

- Safety policy and objectives including management commitment, responsibilities, accountabilities, key safety personnel and coordination of emergency response planning.
- Safety risk management including identification of hazards, assessment of risk and mitigation.
- Safety assurance through performance monitoring, measurement, management of change and continuous improvement.
- Safety promotion through training, education and communication

The majority of the industry represented in the WG disagrees that SMS must be directly tied to DO. While there is agreement that specific elements of SMS should be addressed by a DO (i.e., DO minimum requirements), the majority of industry believes that SMS as proposed in Part 5 should be independent from DO

requirements. As the development of SMS for DO was outside the scope of this WG charter, the WG recommends further discussion by the ARC on this subject.

IV.C.(4) DO Project Applications and Activity Reporting

[Note: This section is consistent with the CDO ARC report of May 2008.]

The WG believes there are many projects that the DO certificate holder may complete without having to notify the FAA, since the DO will be making all determinations of compliance in accordance with its FAA-approved Procedures Manual. There are other projects that the FAA must be immediately aware of as they require the FAA to validate the existing type certification basis or establish a new one.

Part 21 already specifies when an application must be made to the FAA. This includes applications for TC, STC, and PMA. The WG concluded that the existing required applications could be used to discriminate between those projects that required notification of the FAA and those that did not. If the project would require an application under Part 21, then the FAA must be notified when the project is initiated. Such projects would be any new DA, amended TCs requiring a new model designation, new STCs, and any project that would be expected to have a revised type certification basis under section 21.101. Any change that does not rise to this level will be handled by the DO under their approved Procedures Manual.

The WG discussed other DO activities that would not require an application. The WG concluded that in all cases the FAA must be provided access to a record of all compliance activities being performed by the DO. This could be a paper record but in most cases it is expected to be electronic. Such a system of records would include activities such as major and minor changes to an existing design as well as repair approvals. The WG envisions a constantly updated database that may be accessed by the FAA as it desires. This database would need to contain the type of information that the FAA currently uses to measure the significance of a project, similar to the data collected through its Certification Project Notification (CPN) process. The database should also address whether or not the type certification basis may need to be revised, and the scope of FAA LOPI. This complete project listing would provide the FAA with information regarding the DO's activities and would help guide FAA oversight of the DO.

The details of the project list, how often it should be provided to the FAA, and how the FAA would be notified of projects requiring an application are some of the matters that should be discussed with the FAA and included in the Procedures Manual.

IV.D. DO Relationships and Supplier Control

IV.D.(1) General Requirements

An applicant for, or hulder of, a DA has sole responsibility for proper control of all its suppliers, be they suppliers of engineering services, manufacturing of preproduction parts, special process, etc., or any other part of its compliance responsibilities. Should there be any deficiency or non-compliance on the part of a supplier, even if it is a supplier of compliance determinations, products, parts or appliances, the FAA holds the applicant or holder of the design cartificate responsible to correct the deficiency. This long-standing principle remains for the DAH even if the DAH is also the holder of a DO certificate.

The DO must qualify its engineering suppliers, provide overloght, and define the process by which suppliers function within the DO system. This means that a DO may authorize suppliers to make determinations of compliance only after the DO has evaluated the supplier is system and determined that the supplier is qualified to not in that capacity for the DO. This does not mean that the suppliers must adopt the DO's system of processes, but it does mean that the DO must determine that the system of processes to be used by the suppliers are acceptable and meet the DO's requirements.

Tu fulfill us supplier management responsibilities, a DO must have as part of its system a process by which it will determine the appropriate level of oversight required for its suppliers. The DO would need to consider such things as

- · the articality of the design,
- its experience with the supplier.
- the supplier's standing as a FAA-approved DO or as a holder of other I/AA DAs or delegations.
- additional compliance determinations to be made during integration resting of the design, and
- any other appropriate factors

The WG recommends that DO certificate holders be able to cooperate with other companies to pool supplier oversight responsibilities. In a manner similar to what is currently done by manufacturing facilities. As an example, several companies hoying aviones components from a single signifier may cooperate in the surveillance of that supplier by allowing one of the companies to conduct the anduand the other companies to use the results as if they conducted the midit themselves. To gain the credit for such pooling of audit requiraments, it is essential that the supplier processes be consistent across all companies or that the company auditing

Denga Organization Working Group Part 21/8MS Availion Rulemaking Committee

A Report from the Part 21/SMS ARC to the FAA

the supplier assess all the requirements of those companies wishing to share the benefits of the single-party audit.

In selecting suppliers, the DO must consider that, for both engineering and production suppliers, there must be means for FAA to gain access to the facility for the purposes of DO oversight.

IV.D.(2) Oversight of Foreign Suppliers

Oversight of foreign suppliers is required whether or not a bilateral agreement exists between the FAA and the country in which a foreign supplier is located. A DO may authorize foreign suppliers to make determinations of compliance only after the DO has evaluated the suppliers' systems and determined that they are qualified to act in that capacity under the DO processes.

To fulfill its supplier management responsibilities, a DO may also propose as part of its quality system a foreign supplier oversight process for its design services supplier, just as companies do today for parts suppliers under Advisory Circolat (AC) 21-1B, "Production Certificates." This process could be based on using an approved organization as a supplier to its DO, for example, a contract with an European Aviation Safety Agency (EASA) Design Organization Approval (DOA) holder in good standing. In doing so, it must be recognized that the foreign civil aviation authority (CAA) may not be performing any oversight of that activity if it does not lead to an approval under the CAA authority. If representatives of an EASA DOA, for example, are authorized by the DO to make compliance determinations, those determinations must be acceptable and meet the DO's requirements.

The DO remains fully responsible for all compliance determinations made by the foreign supplet holding an organizational approval from its cognizant CAA, just as it is for all other compliance determinations. However, in its supplier oversight function, the DO may take credit for the surveillance of the supplier by its CAA. This credit would result in a reduced need for oversight by the DO, and the foreign implier oversight process should be defined within the supplier surveillance portion of the DO Procedures Manual. The DO, for example, could review periodic reports from audits performed by the supplier CAA or establish some other means of that ing supplier performance. The DO would need to consider the criticality of the design, experience with the supplier, and other factors in determining the degree of oversight necessary, as it does with all supplier oversight. The DO oversight methodology applied to foreign supplier shaving capabilities recognized by their respective CAA would be evaluated as part of the FAA's oversight function. Additionally, the DO supplier oversight process could in finde a qualified third party organization (such as Bravat Ventus).

IV.D.(3) The Need for Specialty Service Providers

For the design and production companies in the aviation system, the I/AA has been moving away from a system of approvals that is based on the use of individual designces, and towards organizational delegations that are based upon demonstrated and approved processes within a company. This is especially true for those seeking or holding original DA certificates. The DO concept further propagates this trend

At the same time, many companies in the aviation community are becoming very specialized in their ability to perform unique technical services. In certain highly technical areas, the number of qualified organizations that can perform specialized services has been reduced to a critical few. The designers and producers of approved products and articles rely on these specialty services to supplement their capabilities. The CDO ARC recognized a need to create a new process wherein these specialty companies are recognized for their capability, and that capability can be used to supplement DO (and ODA) compliance activities, as well as those of other applicants. The CDO ARC choise to call these companies "Specialty Service Providers" (SSP)

The WG recommends that the FAA give proving to developing a means for recognizing an accreditation system for these SSPs. This concept could encompass technical specialties ranging from the more complicated [such as flamouability, dynamic seats, icing, electromagnetic interference, and high-intensity radiated fields] to the more routine specialties [such as materials testing, non-destructive inspection processes, and environmental testing of components]. These are but a few examples of the scope of activities that could be included under this concept.

The WG charter does not authorize the committee to develop the details of such a concept or offer specific proposals to the FAA on this subject. However, the WG agrees with the CDO ARC on the following initial principles and details for the FAA's consideration. The WG recommends that this CDO ARC concept be developed by industry with the direct participation of the FAA because of the variety of issues that must be addressed and the need to create industry consensus standards.

- The SSP could perform compliance with Industry consensus standards acceptable to the FAA and DO as evidence of compliance with specific airworthiness standards resulting in a determination of compliance
- The possibility of third party approval and oversight of SSPs should be considered.
- The data developed by SSPs must be datacily usuable by all applicants without further verification of the data's integrity
- The FAA should porsue informational acceptance of the SSP system

Design Organization Working Group Bin 21/8MS Aviation Rulemaking Committee

17.

- While SSPs may have individuals who are designees of the Administrator, such individuals do not exercise delegation in the course of a DO project as a SSP.
- Persons using SSPs must do so under their supplier control system. The amount of supplier oversight conducted by the user of these services can take into consideration the accreditation of these SSPs.
- The continued compliance responsibility of applicants who use SSPs is not reduced by the fact that the SSPs are recognized by the FAA for their expertise. The TC, PMA approval, or part approval holders still have the total responsibility for initial and continued compliance of the DAs they hold, and the resolution of all COS issues.

The WG recognizes that there are different methods that may be used to implement the SSP concept, and that the concept is not of necessity fiel to implementation of DO Regardless of the method of toplementation, the WG recommends that the SSP system be made available to the available community.

IV.D.(4) Inter-DO Relationships and Responsibilities as Design Partners

Inter-DO relationships will be controlled by a supplier interface document (see section IV F). This interface document will allow the supplier DO to initize its own procedures and processes within its scope of authority as a DO. The applicant DO will identify how supplied data (and potentially determinations of compliance) will be assessed for integration and applicability to the product.

IV.D.(5) Agent DO

The WG discussed the creation and recognition of a DO that has been contracted to act on behalf of a DA applicant. The WG refers to such a DO as an Agent DO. An Agent DO may or may not hold a DA and must meet the minimum DO requirements specified within this report. The Agent DO provides an avenue for an interested DA applicant to utilize the professional DO services that the applicant either does not have the financial ability to create. The use of an Agent DO promotes consistency in design activity (safety enhancement), an accountable design organization (responsibility), and may reduce the necessary aviation system assessment and oversight resources. The WG believes the Agent DO rote fills a critical need in the DO model where DO thresholds impact an applicant, ability to create needed designs or design changes, particularly with respect to small business applications.

Design Organization Working, Group Part 21/SMS Availan Rulemaking Committee

The WG provides the following text from EASA 21.A.2 to consider, as a similar application of the Agent DO concept.

21.A.2 Undertaking by another person than the applicant for, or holder of, a certificate

The actions and obligations required to be undertaken by the holder of, or applicant for, a certificate for a product, part or appliance under this Section may be undertaken on its behalf by any other natural or legal person, provided the holder of, or applicant for, that certificate can show that it has made an agreement with the other person such as to ensure that the holder's obligations are and will be properly discharged.

IV.D.(6) Business Structure Variation

The WG recognized that additional business structures may exist where a DO is part of a business structure in which companies are not in a traditional supplier relationship. An example of such a business structure is a consortium, where each company is an equal business partner and is not considered a supplier to the other. In some cases one business partner is US-based while the other is foreign-based. Also the consortium company may or may not be US-based. As such, neither company has complete oversight, in the supplier sense, of the other company. Each business partner wants recognition of their respective DO in a certification project. Such is the situation, in any business structure, where one company does not have full supplier oversight responsibilities over another company.

Industry members of the WG currently have such business structures with US-based companies and foreign-based companies. As the aviation business continues to grow globally, Industry members foresee continued growth of such business structures to mitigate associated business risks and leverage various strengths of different companies.

The WG discussed the notion that a US-based DO, when the consortium company is foreign-based, should be able to manage the integration responsibilities for the FAA portion of a certification project, including the project statement of compliance to the applicable airworthiness standards knowing that the US-based DO may not possess or have access to all project descriptive and/or substantiating data. This is a result of the need to protect proprietary methods and information within each respective company. Completion of all the required determinations of compliance must be documented to support the project statement of compliance.

The WG also discussed the notion that a consortium may consist of two US-based DO's. Industry recognizes there is one applicant and certificate holder for each TC; that has historically been the consortium company. The FAA has worked with consortium companies to execute shared-responsibility agreements between the members, which are acceptable to the FAA. That has allowed each consortium member to autonomously execute its role independent of the other consortium

members, including findings of compliance under FAA delegators. This protecting its proprietary data. A similar FAA-accepted working agreement could describe how those autonomatisly performed design and compliance determination processes under DO are integrated into a single type design, a single TC and a single continued airworthings process.

The WG believes further discussion is needed between the FAA and Industry to develop how this would be accomplished. Therefore, the WG recommends further development of how such husiness soluctores will be accommodated under the proposed DO financivotic linput from industry, especially those who presently linve consortium programs, should be considered further.

IV.E. Scope, Limitations, and Privileges of DO Certificates

There are many variations in design and production organizations and their products throughout the aviation system. They range from organizations dealing with a full line of products, like transport airplanes, high-tech general aviation aircraft, helicopters, and high-bypass engines, to PMA holders with a more narrow focus. In some cases, the FAA compliance approvals for the activities of these organizations are made either directly by FAA resources, or by using individual or organizational delegation approvals from the FAA. Repair stations may have ODA authorizations or their own company designees, or may contract with consultant designees to perform DA functions. This describes but a few of the organizations that make a business of engaging in design and production certification activities.

The challenges and benefits of the DO concept are outlined in Section II. In addition to the ability for FAA to implement SMS, the safety benefits of a more complete corporate focus on compliance and safety can further permeate the Industry if these organizations are required to obtain DO certificates. For this reason, the criteria for obtaining and holding a DO certificate must be such that they can be tailored to the size and functions of the specific DO certificate holder.

As has always been the case, Industry is responsible for compliance with the regulations; this will not change under the DO concept of operation. The rigor associated with that compliance is contained within the FAA-approved DO Procedures Manual, and it must be tailored to the size of the organization and the complexity of the items that the DO certificate holder designs and / or produces.

A total "culture of compliance" must exist within each DO company, but how that culture is established will likely differ for each DO certificate holder. The key is to define criteria against which all potential DO certificate holders will be measured, but recognize that there will be variables in how the criteria are met based on different types and sizes of companies and associated regulatory obligations.

IV.E.(1) FAA Limitations on the DO Certificate

A DO certificate may cover type certification activities, supplemental type certification activities, and activities leading to the issuance of PMAs, as well as activities associated with a production approval. For a particular DO, the FAA may limit the scope of activities that might be accomplished by that DO.

Type Certificates. For type certification activities, it would be rare that a certificate holder would be able to perform <u>all</u> the responsibilities necessary for demonstrating compliance for <u>all</u> products that are eligible to receive a TC. For this reason, the FAA may restrict a DO certificate to only products covered by a specific

part of the airworthiness/design requirements, such as 14 CFR part 23 (small orphanes), 25 (large airplanes), 27 (small rotorcraft), 29 (large rotorcraft), 11 (balloons), 33 (engines), or 35 (propellers).

The FAA may further limit the scope of DO certificate activities within a proof regulatory part. For instance, a manufacturer might only have the experience necessary to properly comply with DO requirements for small transport airplanes under 14 CFR Part 25 airplanes, but not large transport airplanes, or for reciprocating engines under 14 CFR Part 33, but not large nirbofan engines. The FAA may use other parameters it determines to be necessary to further limit the scope of a DO certificate. The intent is to allow the widest scope of certificate for which the upplicant has been able to demonstrate its capability to comply with the relevant design and airworthiness requirements.

Supplemental Type Certificates. In the case of SFC L the scope would also likely be defined in more narrow terms. For example, the scope might be limited by the products that a particular atrine operates, or by technical discipline and subpart (Pari 23 structures, for instance) or by the complexity of the product (large turbolan engines, for instance), or by other generic parameters the FAA determines to be appropriate.

PMA. In the case of PMA the scope would likely be tailored to each certificate holder

Determining the Appropriate Scope of a DO. When determining an appropriate scope for a DO cartificate, the FAA must ensure that the certificate holder has, and will continue to maintain, the capability to meet all the requirements of the subpart within the scope of us certificate.

As part of this determination, the FAA may consider providing multiple DO certificates in unusual situations for applicants with substantially decentralized organizations, or who have a wide range of products or capabilities. When evaluating whether a single certificate or multiple certificates is most appropriate, the FAA would consider the organizational structure of the applicant, interactions of remote or co-located design and production facilities, and the use of continon processes and procedures.

The scope of any DO certificate will be clearly defined so that all persons, including other civil aviation authorities (CAA), will understand the scope of authority for FAA-approved data granted under that certificate.

Transfer of a DO Certificate. A DO certificate holder cannot transfer the DO certificate

Design Organization Working, Group Pan 21/SMS Available Rolenaking Committee

IV.E.(2) Scope of DO Certificate Privileges Must Be "Functionally Complete"

The DO certificate privileges must be functionally complete, which means they must cover all activities that would have to be undertaken in order to fully complete a DA project within its scope of a DO's authority. The DO holder's competence and capabilities must also be functionally complete in order to certify compliance with the applicable airworthiness safety standards within the scope of its authority. Those include:

- All certification activities leading to the issuance of an original or amended design approval, including design, airworthiness, manufacturing, and maintenance and operations activities as they relate to a DA. This includes; engineering inspection, analysis and tests; flight tests; ICA, aircraft flight manuals, etc.
- All determinations of compliance, including those that involve a subjective evaluation.
- **Continued airworthiness activities,** including changes to those approved designs for product improvements or safety enhancements, such as those contained in service bulletins, or repair data.
- Manufacturing and airworthiness activities, such as the pre-production manufacturing of parts, components, and subassemblies; and conformity of test articles and products, and their airworthiness certification for flight test.
- **The development and testing of designs and processes** for possible inclusion in future approved designs (i.e., "eligible data").
- Any other activities leading to the development of data necessary for the FAA to determine compliance with the requirements issued by those countries from which validation is sought and with which the FAA has a bilateral airworthiness agreement covering that compliance activity.

[Note: While not a privilege of the DO certificate, the FAA may wish to grant the DO holder the additional privilege of making compliance determinations to CAA requirements.]

IV.E.(3) Form of a DO Certificate

Examples of EASA DOA Certificates and Terms of Approval were reviewed and discussed in assessing application to DOs. The WG recommends a structure similar in nature for a FAA DO Certificate. This would provide consistency between the FAA Certificate and CAA Certificates.

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

IV.E.(4) Transfer of Design Approvals under DO

A TC/STC issued after the effective date of maulation (e.g. under the DO framework) is transferable provided the following requirements are mut-

(a) The transferor of the TC2STC makes a statement, in writing, to the FAA that they are providing all descriptive data and providing, or making available all substantiating data to the transferce. This statement must describe the conditions under which the substantiating data is made available to the transferee if the data will not be provided (i.e., contractual agreement).

A TC/STC issued prior to the effective date of regulation is transferable provided the descriptive data and substantiating data the transferor has, or has access to, is provided to the transferee. The WG recognized that existing TC/STU holders may or may not have the descriptive data and/or substantiating data and imposing a (egulatory restriction on the transfer would negotively impact the asset value

The WG recommends non-DO DA transfer requirements he provided by a separate WG to the ARC

IV.E.(4).(a) PMA Specific Design Approvals under DO

One to the unique combined design and production nature of a PMA, PMA is not eligible for transfer

IV.E.(5) The Generation of "Eligible Data"

Under a DO certificate, compliance is an intended by-product of an FAA-approved DO system properly functioning order us Procedures Manual, which includes a formal internal aidit and oversight process. For the compliance determination for a particular part or component to be complete, it is essential that a certification basis be established for the product on which it is to be instabled or for the arnels itself. Another essential element is that a type or article design be fully defined so that the interaction of products, parts, and components may be assessed, since that interaction may establish additional certification peeds.

It is common for a company within its normal engineering and production system to develop products, parts, components, and processes for future use in certification programs. In the case of a DO certificate holder, if that development is accomplished under the approved DO system, then that development could be eligible for inclusion in subsequent designs, except for the establishment of a product final certification basis and complete product or article definition. It would be inappropriate to consider such development activity as meeting the standards for

> Design Originization Working Group Part 21/SMS Availor Rolenaking Committee

-14

complete compliance determination because thuse two elements would be missing. It is appropriate, however, to give credit for any compliance autivities accomplished under a DO. The WG refers to this as "eligible data."

"Eligible" data are data developed under the processes of an approved DO system, given a specified, but not necessarily final, certification basis and product type dusign or article design.

To use "eligible" data, the DO holder must assess the data's compliance against the final type certification basis of the product or article and final type design or article design, respectively, in which it is to be used. It winitd not be necessary to repeat the compliance activities, provided those activities were appropriate for the final product or article and its certification basis.

The creation of "eligible" data is a concept that is intended for use internal to the DO. No approval or compliance determination can be conferred upon the data if the data are provided for use outside the DO.

IV.E.(6) Voluntary Disclosure Privileges

The FAA has several active volontary disolosure programs for air varitor, production approval holders and organizations that have an ODA, among others. These programs are designed to encourage the reporting of product and process deficiencies so they can be corrected before unsafe conditions occur. The programs also apply to discovered deviations from FAA-approved procedures manuals and inadvertent regulatory violations. If the deficiencies or non-compliance activities reported were not intentional or criminal in nature, the FAA will refrain from using the disclosures as the basis for any civil penalty, as long as the certificate holder takes swift action to correct the deficiencies discovered.

As stated on the FAA's Volontary Safety Programs Branch website

" the FAA believes that avtation safety is well xerved by providing incentives for certificate budgers to correct their own instances of non-compliance and to invest more resources in efforts to preclude their recurrence. The FAA's policy of forgoing civil penalty actions when a certificate holder meets the requirements of this program, is designed to encourage compliance with the FAA's regulations, foster safe operating practices, and promote the development of internal evaluation programs."

Although the DO is a new type of certificate, the WG concludes that the information presented above remains equally applicable for a DO, and the FAA voluntary disclosure policy should be extended to DO certificate holders. Activities under a production approval are already covered by FAA voluntary disclosure policy.

Design Organization Working Group Phd 21/SMS Aviation Rulemaking Committee

The FAA voluntary disclosure reporting program is presented in AC 00-58A. Under the section entitled "Purpose" there is an important exception that must be recognized. The AC states, in part:

"The procedures and practices outlined in this AC <u>cannot</u> be applied to those persons who are required to report failures, malfunctions, and defects under 14 CFR Part 21, section 21.3, and do not make those reports in the timeframe required by the regulation."

This exception to the program is in recognition of a determination made by the FAA that, since there is a regulatory requirement to report under §21.3, the voluntary disclosure of a failure to report cannot relieve the certificate holder from any enforcement that might be based on that failure to report. This exception still appears to be appropriate for a DO certificate holder under the same defined

IV.E.(7) Manufacturing and Production Functions Under DO

There are two types of manufacturing and production functions that need to be addressed with respect to a DO and a production organization:

- Those pre-production manufacturing functions associated with obtaining a DA, and
- Those associated with a production approval (i.e., post-design approval production).

The intention is to allow a DO to utilize its existing FAA-approved production system during pre-production manufacturing functions associated with obtaining a DA. If the organization, within its scope of its design authority, chooses to use their approved production quality system, then it must use the DO procedures manual processes for any of the following:

- Conformity inspection
- Determining conformity of parts and test articles
- Determining conformity of test setup
- Determining conformity of installations

For DOs performing pre-production manufacturing as part of its scope of activities, in addition to the above, the DO Procedures Manual must also contain procedures for:

- Controlling documents and data associated with pre-production manufacturing
- Ensuring that each supplier furnished product, part or appliance conforms to its design.
- Controlling manufacturing processes to ensure conformity to its design.
- Conducting inspections and tests.
- Ensuring calibration and control of all inspection, measuring, and test equipment.
- Documenting the inspection and test status of products, parts, and appliances supplied or manufactured to the design.
- · Ensuring that discarded articles are rendered unusable.
- Implementing corrective and preventive actions to eliminate the causes of an actual or potential nonconformity to the design or non-compliance with the approved DO Procedures Manual.
- Preventing damage and deterioration of each product, part, and appliance during handling, storage, preservation, packaging, and delivery.
- Identifying, storing, protecting, retrieving, and retaining quality records.
- Planning, conducting, and documenting internal audits to ensure compliance with the approved DO Procedures Manual.

With respect to post-design approval production, the production approval requirements remain the same.

The DO processes would also support the FAA's issuance of special airworthiness certificates in the experimental category for the purpose of research and development or show compliance.

While the WG intended to further discuss the manufacturing and production functions to include the combined design and production organizations (i.e., one certificate comprising design and production), the WG was limited to the preproduction concepts noted above due to schedule constraints. Further discussion would be required to propose a recommendation on additional privileges that may be available for a combined Design Production Organization (DPO).

IV.E.(8) Flight Standards Functions

Section 21.17(a)(1) requires an applicant for a TC to show that its product meets "the applicable requirements of this subchapter that are in effect on the date of application for that certificate." Part 21 resides in 14 CFR chapter 1, subchapter C,

titled "Aircraft." This subchapter covers Parts 21 through 59, which includes the type certification airworthiness standards found in Parts 23-35. The operating rules applicable to these same type certificated aircraft are found in subchapters F and G, which include Parts 91 through 139.

While a TC may legally be awarded without the product complying with appropriate operating requirements, the practice has been to provide an initial operational evaluation of aircraft during the type certification program. That operational evaluation is carried out by the Flight Standards Aircraft Evaluation Group (AEG) that has the responsibility for the particular product being type certificated. The AEG performs or coordinates the following activities associated with the type certification of products, which are discussed in FAA Order 8900.1:

- ICA Review and find acceptable the maintenance aspects of the ICA which are required under 14 CFR §21.50, and §XX.1529 in the respective aircraft certification standards.
- Flight Operations Evaluation Board (FOEB) The primary tasks are the development and revision of the master minimum equipment list (MMEL).
- Flight Standardization Board (FSB) The primary responsibilities are to determine the requirements for pilot type ratings, to develop minimum training recommendations, and to ensure initial flight crewmember competency.
- Maintenance Review Board (MRB) Establish the minimum maintenance and inspection requirements for transport category aircraft, engines, propellers, and auxiliary power units. Participate in industry steering committee meetings to review the Maintenance Steering Group (MSG)-3 analyses.
- Participate in type certification board and flight manual review board activities.

During type certification, all determinations of compliance to the airworthiness requirements in Parts 23-35 are made by the DO certificate holder, with appropriate FAA oversight. Since compliance with the ICA requirement in §XX.1529, §31.82, §33.4, and §35.4 are to be determined by the DO certificate holder, the WG recommends that the maintenance aspects of those requirements also be determined by the DO. The DO procedures manual would have to contain appropriate procedures that ensure the maintenance aspects of the ICA are properly addressed, consistent with §21.50 and FAA Flight Standards' regulatory guidance.

The WG believes that the formulation and execution of the FOEB, FSB, and MRB should continue as Flight Standards AEG functions, with support from the DO certificate holder. All determinations of compliance to airworthiness standards associated with those boards would be made by the DO certificate holder consistent with its Procedures Manual. Some additional responsibilities associated with the operation of those boards might be assigned to a DO certificate holder, under Flight

Standards policy, after experience is gained. This would necessitate a revision to the DO Procedures Mamual

AEG participation in type certification board and flight manual review board netwines would continue to the degree that the FAA Aircraft Certification Service participates in those functions. For new TCs and amended TCs requiring a model change there would be a review by the type certification boards, but it is expected that most major changes would be conducted under DO procedures and would not require board review. This is because the type boards are identified in an FAA order and the DO certificate holder is free to propose its own procedures in lieu of those identified in existing FAA orders.

For a DO, there would not be a flight manual review board as the sole responsibility. for determining compliance for the flight manual would reside with the DO certificate holder. Any operational regulations and associated Flight Standards guidance with respect to flight manuals would be complied with through processes and procedures defined in the DO Procedures Manual

IV.E.(9) Noise, Fuel Venting, and Exhaust Emissions

While Congress has granted the FAA full statutory authority over the airworthiness certification of civil aviation products in the US, the Environmental Protection Agency (EPA) actually guides FAA requirements with respon to noise. Itel venting, and exhaust emissions (14 CFR Parts 34 and 36)

Under the current system, FAA Order 1050 TE sets policies and procedures and assigns responsibilities for ensuring that the FAA complies with environmental procedures as required by the National Environmental Policy Act under the direction of the Council on Environmental Quality. The Order contains examples of actions that normally require an environmental assessment, including noise and emission requirements.

In addition, The Noise Control Act of 1972 requires the FAA to make findings, not withstanding any delegation to companies, other private persons, CAAs, or any procedures for type certificating foreign-manufactured aircraft. The FAA's Office of Environment and Energy (AEE) delegates the authority to make these types of findings to the appropriate FAA Certification Directorate, depending on the type of micraft involved. That Directorate may not re-delegate the authority and the FAA outst base its finding on actual examination of each type design. Individual delegations have been granted by the FAA but they are only for recommending approval, and not finding compliance.

While the WG recognizes the distinction between the arrorithmess requirements of D/CFR and the noise, fuel venting, and emissions requirements, the WG believes that a DO could be found to have the necessary capabilities and expertise to make

Design Organization Working Group Bin 21/SMS Aviation Rulemaking Committee

compliance determinations with respect to the environmental requirements contained in 11 CFR Parts 14 and 36. Specific noise, fael venting, and emissions processes would be developed within the DO compliance, safety, and quality systems to ensure proper compliance determinations. This is in keeping with the WG's principle of a DO making 100% of the compliance determinations.

The WG recommends that the FAA propose to the EPA that the process-based approach to compliance, as established by DO program principles, is far more robust than the normal delegation process and is sufficient to ensure compliance with the environmental aspects of the 14 CFR Parts 34 and 36. The WG believes this to be consistent with the recommendations set forth by the FAA in response to section 312 of the FAA Modernization and Reform Act of 2012, which recommenda "expanding delegation capability to include support for all certification anyworthmess standards when appropriate, particularly low-nisk or romme activities such [as] those related to noise and emissions tests and iCA."

IV.E.(10) Establishing a Certification Basis

An applicant for a DA may propose a certification basis to the FAA and the FAA establishes the certification basis. With regard to design changes under DO, the WG discussed that the FAA could establish a vertification basis by utilizingwithin limited boundaries contained within the DO approved procedures manual. This is considered a FAA pre-decisional approval process provided to the DO. Projects that fall within the epproved boundaries would be performed order the domain of the DO without additional FAA input. Industry has identified such pre-decisional approval as being critical to the success of DOs.

IV.E.(11) Changed Product Rule

The derivative type certification requirements which apply to TCs and STCs, specify the need for FAA to make a determination of the appropriateness of the original type certification basis. This is sometimes referred to as the "changed product rule requirements". As a general rule, those projects would be subject to FAA LOPT and it is expected that a DO would notify the FAA when it undertakes such projects.

As the FAA gains more confidence in specific DO certificate holders, it may be willing to rely on specific DO approved processes to assist the FAA in making its determinations under the changed product ride

> Design Organization Working, Group Prin 21/SMS Available Rolenaking Committee

IV. F. DO Approval of Data

The DO privileges associated with approval of data discussed below are made in consideration of a 14 CFR part 21 accountability framework policy initiative which the FAA has indicated it is currently pursuing for FAA managed certification projects. The FAA has advised that the policy will be formalized in Order 8110.4D and in future changes to Orders 8110.37 and 8100.15. The new policy will clarify that discrete substantiating or descriptive data generated in support of a part 21 certification project do not need to be discretely 'approved'. While the WG accepts the FAA's intent to return to the basic foundations of part 21, the WG believes that this specific change alters an accepted paradigm and practice that has been established over several decades within FAA policy. However, if the FAA continues to pursue this new policy, the WG strongly believes that whether a project is managed by the FAA with designees or by a DO, the status of any data, irrespective of the part 21 process that produced it, should be the same. Should the FAA not pursue this policy change, the FAA must ensure that the DO regulation includes the privilege of making discrete 'compliance determinations' that result in data that 'are' or 'are equivalent' to "FAA-approved" and that the data is internationally recognized within the scope of bilateral agreements.

Data Supporting a Design Approval under Part 21

Current practice has facilitated "FAA approval" of discrete 'substantiating data' as well as discrete 'descriptive data' that ultimately makes up the type design. However, part 21 contains no requirement for "approval" of these discrete data separate from issuing the TC (ref: 14 CFR 21.41 where the TC is defined to include the type design, any operating limitations, the certificate data sheet, the certification basis, and any other conditions or limitations). The 'substantiating data' is the documentation related to the applicant's 'showing', while the 'descriptive data' defines the type design that needs to be determined 'compliant' to the CFRs and ultimately approved when the FAA issues a DA. The WG concurs with the FAA that making this clear would eliminate the perceived value and pedigree many currently attach to data and the assumption that the reason data has value is because the FAA has labeled it "FAA approved".

The significance of getting back to the basics of part 21 is relevant to all discrete data 'showings'. However, it will become even more relevant as the FAA begins to implement risk-based decision making and when it chooses not to be involved in certain aspects of a certification project. In such cases, it would not be logical to require the FAA to status discrete data "showings" when it has chosen not to be involved in making a discrete "finding" of compliance. Doing so wrongly implies that all data need to be statused by the FAA, which implies some level of FAA involvement, even when the FAA has already determined not to be involved in a given aspect of a certification program.

The WG acknowledges that independent of how any data was used in the past, it remains the applicant's responsibility to 'show' compliance even when such data is used to support subsequent part 21 projects. The WG understands that the FAA intends to clarify this in future type certification policy to further reinforce the accountability framework concepts related to design certification. This should include revisions to its 8110-3 and 8110-9 forms to distinguish between when the form is being used for a part 21 project and when it is supporting a part 43 maintenance or other operational requirement associated with an existing DA.

While not the case for part 43 maintenance actions, which will be discussed later, and unlike today's practice, the data produced for the purposes of showing compliance under part 21 needs no label or pedigree designating it as "approved" by the FAA. Instead, the 'descriptive' and 'substantiating' data will only be designated as "found to have shown compliance to the airworthiness standards by the FAA" or "determined compliant to the airworthiness standards by a DO." In either case, only when the FAA issues its DA will it make the single "finding" required by part 21 and in so doing "approve" the 'descriptive data' that defines the complete type design.

Data Supporting Major Design Changes under Part 21

Part 21 currently contains provisions for FAA approval of both major and minor design changes. In promulgating the DO regulation, once a type design is held by a DO, the DO must have the privilege to make all necessary "determinations of compliance" associated with any change and also "approve" the 'descriptive data' by incorporating the change into its type design. While existing statutory law (Title 49, Section 44702) allows only the FAA to issue a certificate, there is no restriction preventing the FAA from allowing a DO to make changes to an existing TC as a privilege of its DO certificate (i.e., DO is not a delegation). In exercising this privilege, the DO will "approve" the 'descriptive data' associated with any design change once it incorporates the change in accordance with its approved procedures. This privilege should be given with respect to any type of DA held by a DO. including PMAs (which are not addressed in Title 49). This privilege should also be given to a DO that is contracted to act as an agent of the applicant to manage its type design. In a supplier role, a DO may provide 'determinations of compliance' that support a design change, however, as a supplier a DO cannot 'approve' design changes since it is not the custodian of the type design.

In promulgating this privilege, the existing part 21 regulations regarding major changes will need to be changed. For major changes, the current regulations *require* that 1) an application be made, 2) the regulations under 21.101 be considered, and 3) the "person" obtain either a STC or an amendment to the TC. There is no alternative process that allows this to happen without the FAA. Today, only the FAA or its designee can address the certification basis or issue an amended TC or STC. Therefore, today the approval of the major change is an FAA activity. In the case of 21.101 the function is inherently governmental and will require the

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

FAA to be involved. However, the WG foresees that the FAA could facilitate its involvement through a "pre-decisional" process that is part of the DO procedures manual. The DO could follow such a process to ensure that the FAA is properly engaged on certification basis decisions that are outside what is allowed by the pre-decisional process. In this way the DO is not held up and forced to wait on the FAA for the majority of its project activity.

Data Supporting Part 43 Maintenance or any Operational Requirement

While DO "determinations of compliance" associated with seeking a DA or supporting a minor or major change do not need to result in discretely "approved data", a DO must have the authority to make "determinations of compliance" that result in the creation of "approved data" when required to support part 43 maintenance or any operational requirement associated with an existing type design. In such cases, the data may or may not be part of the approved type design. For example, when a DAH issues a service bulletin, the data is conveyed as a change to the type design. However, a holder (or even the FAA) can provide data to support a unique repair or alteration for an individual owner/operator, an action which does not result in a change to the type design.

The WG contends that the "approved data" required by part 65 for use in part 43 is the approved 'descriptive data' and any other "technical data" required to perform the maintenance. (This includes any drawings, material specifications, process specifications, procedures, and other data describing an approved repair or alteration). The WG contends that part 43 does not require any 'substantiating data' used to 'show' compliance to be approved directly by the FAA. Thus, the DO's authority to approve data in support of part 43 maintenance applies only to the 'descriptive data' (i.e., not the 'substantiating data') associated with the DAs it holds. This means that, at its discretion and without any action by the FAA, the DO may create "approved data" to support repairs and alterations by third party owner/operators for DAs held by the DO (or with respect to any other DA for which the DO is authorized to provide data for maintenance under its scope of authority).

As a result of its discussions, the WG sees three possible regulatory options for the FAA to consider in addressing DO "approved data" to support major changes and to support maintenance:

<u>Option 1 – DO has authority to create "FAA approved data"</u>: This option assumes the DO will be authorized to create "FAA-Approved Data" in a manner that does not include delegation. In the FAA's current system, all type design data "approved" by the FAA for use in the global aviation system have been referred to as "FAA-approved." This option continues with that approach.

For decades 14 CFR §21.95 ("Approval of minor changes in type design") has allowed minor changes to a type design to be "approved under a method acceptable to the Administrator before submitting to the Administrator any substantiating or

descriptive data." Additionally, in 14 CFR part 1, the term "approved" is defined as *approved by the Administrator, unless used with reference to another person*. Since section 21.95 makes no reference to another person, the regulation allows for the creation of FAA-approved data without the data being submitted to the FAA or reviewed by the FAA. The FAA-approved data are created when the TC holder executes the "method acceptable to the Administrator."

While section 21.95 applies only to minor changes to a TC, the WG believes that this existing approach can be applied to major type design changes determined to be compliant by a DO (i.e., DO creation of FAA-approved data before any substantiating or descriptive data are submitted to the FAA). The DO regulatory requirements, along with the processes and procedures contained in an FAA-approved DO Procedures Manual, must be sufficiently thorough for the FAA to approve the data resulting from them prior to it being submitted to the FAA. As with minor type design changes, the FAA may review any 'compliance determinations' and supporting data after it has been determined to be compliant by the DO.

Under this concept a DO is not approving data on behalf of the FAA, because a DO is not a delegation. Once the FAA-approved DO process for making a compliance determination has been properly executed, the 'descriptive data' are FAA-approved.

Option 2 – DO has authority to create "DO-approved data" equivalent to "FAAapproved data": This option would facilitate recognition of "DO-approved data". It would require a change to 14 CFR Part 1, where the term "approved" is defined as *approved by the Administrator, unless used with reference to another person.* The regulation would need to be changed to include other entities entrusted by the administrator to "approve". Under this option, the WG would recommend the definition be revised to: *approved by the Administrator or under the authority of a certificate granted by the administrator, unless used with reference to another person.* In granting this privilege to each DO, it is essential that the FAA affirm to its international airworthiness partners that such DO data is equivalent to being "FAA-approved."

<u>Option 3 – Create a different term for DO "approved data"</u>: This would require a change to parts 65, 121, 135 and 145 instead of part 43 to make it clear that it can be used.

While it may seem easy to simply permit certificate holders to issue "approved" data, this seemingly easy function would actually be far more difficult to reconcile with current regulatory practice than it appears at first glance. The FAA Chief Counsel's office has already met with the WG to explain that the word "approved" is currently defined to encompass inherently governmental tasks, and that office has expressed reservations at permitting a certificate holder to issue approved data. The word "approved" is also a difficult word to redefine in this situation because it used in a variety of different contexts in the FAA regulations to reflect things that are
approved by the Administrator, and this use imposes certain constraints on the ability to make changes in the use of the term.

Options 1 and 2 are trying to affect a very specific use of the term "approved" (a subset of the ways that it is used) – the use of the term in the context of data upon which maintenance providers may rely in the case of major repairs and major alterations. Rather than trying to craft language thath does not adversely affect the other uses of the term "approved," it might be preferable to adopt a different adjective to describe the data that is appropriate for use to support major repairs and major alterations. In order to effect this change, one could define a new term ("purple data" and "eligible data" have been used as placeholders by the WG, but any adjective not already in use in the regulations could be acceptable). The data included under the definition of this adjective would include both FAA approved data and DO-approved data (it would need to be defined in section 1.1 of the regulations to have global impact on all FAA Parts of the regulations). Then, the requirements for approved data currently found in Parts 65, 121, 135, and 145 could simply be updated to reflect the new adjective.

Some positive aspects of this proposed change include (1) it would accomplish the goal of permitting DOs to issues data upon which the maintenance community could rely in a manner that is nearly identical to current practice, except with limited FAA involvement (limited to FAA-chosen LOPI), (2) it would be consistent with existing statutory authority, (3) it would limit the possibility of unintended consequences (impact on other uses in the regulations of the term "approved"), and (4) It would also avoid potential delegation to the public of inherently governmental functions. The negative aspect of this proposal would be the potential need to update international executive agreements (like the maintenance-acceptance provisions of the bilateral agreement with Canada) in order to reflect the new terminology; however this impact would likely apply also to expanded use of the term "approved," due to the fact that we would be changing the inherent definition of a term upon which the bilaterals rely, which may cause our trading partners to want to revisit the affected bilateral agreements, anyway.

While any of the above three options would be acceptable to industry, the FAA members of the WG indicated that option 3 is likely the most viable option for supporting part 43 maintenance without the use of delegation (an action which was achieved in the past through recognition of SFAR-36 organizations). The FAA expressed concern regarding whether it was statutorily possible to grant DOs the privilege of approving major changes to the DAs under part 21, as Industry prefers. Current FAA thinking is that some form of delegation would be required. The FAA members of the WG acknowledge that, in order to work within the DO construct, any delegation would essentially be performing an administrative function in making its statutorily-required finding. In this role, the delegation would perform the following administrative actions prior to certificate issuance or approving a type design change:

• Verify the FAA's planned project LOPI is complete without open issues,

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

- Verify that the statement of compliance was issued by a DO representative authorized to make the statement, and
- There is no knowledge of any non-compliance or unsafe condition conveyed by FAA or DO personnel involved in the project. (Note: This is intended to be a yes or no answer without conducting research.)

In summary, the WG identified three key privileges that a DO must be granted with regard to how it documents its compliance decisions and dispositions data. In all cases the data would require no further 'showing' in regard to the applicability and acceptance of its intended use:

- 1. <u>New Part 21 Design Approvals</u> -- Make "determinations of compliance" related to the DAs it is seeking to hold under part 21 and similarly to make them as either a supplier to another DO or as an agent contracted to manage a certification project for an applicant seeking a DA, when authorized through a formal interface agreement.
- 2. <u>Part 21 Type Design Changes</u> --- Make 'determinations of compliance' related to both major and minor changes to the DAs it holds under part 21, or for those it has been contracted as an agent for another holder evidenced by a formal interface agreement, and to also "approve" those type design changes. (Note: in a supplier role, a DO does not 'approve' design changes since it is not the custodian of the type design.)
- <u>Maintenance and Operations</u> -- Create and distribute "approved data" to support part 43 maintenance or any operational requirement associated with a DA it holds, and similarly with regard to third party DAs (i.e., not held by the DO), but where the data approval is executed within the DO's FAA-authorized scope of authority.

Dissenting Opinion – Approved Data

A member of the WG does not agree with the opinion expressed above, related to data considered to be approved by the FAA. A WG member agrees that the type design, as defined in §21.31, is not only found to comply to applicable airworthiness requirements but is considered to be approved by the FAA, with the issuance of a DA. Thus, the issue is with the FAA desire to not call the substantiating data "approved data."

Section 44704 of Title 49 U.S.C. requires the Administrator to issue a TC (including STC) when a finding is made that the product "meets the regulations and minimum standards prescribed under section 44701(a) of this title." FAA policy has extended this basic premise to all DAs. Those minimum standards include the airworthiness requirements found in 14 CFR. The only way the FAA can make that finding is by reviewing, to whatever degree it deems appropriate, the substantiating data submitted by the applicant to show compliance with applicable airworthiness standards. This statutory requirement is implemented in 14 CFR Part 21. Thus, all

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

substantiating data must be found to comply by the FAA. It would be proper to refer to this data as "data found to comply."

For many decades the FAA has called that "found-to-comply" data "approved data." The words "approved data" have been common place within the U.S. aviation system and have been recognized as having significance by international authorities. Any change to this entrenched concept in U.S. aviation, and the enormous task of reeducating FAA employees, U.S. industry, ICAO, and other airworthiness authorities, should only be undertaken after the FAA has presented a good cause argument for that change. In all of the discussions of this issue within the WG, FAA has never defined the problem they are trying to correct or the new message they are trying to convey by no longer allowing substantiating data to be called "approved data."

A WG member has been unable to identify any process or other action the FAA undertakes to create "approved data" from "data found to comply." A WG member company designees cannot recall any training from the FAA that defines an action they must undertake once they have found data to be compliant, in order for them to check the box on the DER approval form saying the data is approved. A WG member has not been able to find any instructions in DER guidance material that describes any action a designee needs to take to create approved data, other than simply checking the box on the form. Based on that absence of direction and years of history working with the FAA on type certification programs, a WG member believes that "approved" is merely a shorthand way of referring to "data found to comply."

While it could be argued from the above discussions that there is nothing changed about the data itself by not calling it "approved," a WG member fails to see the need for the FAA to change several orders and reeducate the entire international aviation community on why it is not calling substantiating data approved, when in fact the character of the data has not change. A WG member would rather see valuable FAA resources used for more productive purposes, such as continued airworthiness oversight and eliminating design project sequencing. Also, with the FAA calling the type design "approved data" yet not using the same term with respect to substantiating data, it is only natural for some people to conclude that substantiating data has a somewhat lesser safety pedigree in the eyes of the FAA, no matter what the FAA does to combat such a position. To remove the words "FAA approved" will likely be seen as a change in safety or airworthiness status of the data.

IV.F.(1) Service Bulletins and a Standardized form for DO Transmittal of Approved Data

DO-issued service bulletins should be a means for DOs to provide "approved data" (reference approved data option discussions in IV.F) for general use. When issued, service bulletins constitute a change in type design by the holder and convey the

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

necessary "approved data" to implement the change by owner/operators under part 43.

A new or revised form is also needed for domestic and international recognition of "approved data" created under the DO concept. The WG recommends that the form be similar to the FAA Forms 8110-3 and 8100-9 that are currently used to approve data in the FAA's delegation system. The form should be titled to recognize the source of the data approval and should be traceable to the originator. In addition, there should be no provisions for recommending approval of data. Thus, "approved" is the only statement that can be made about the data.

The new or revised form should contain the same basic information as the Forms 8110-3 and 8100-9 with respect to the compliance data and purpose of the approval. The form should also address both the date the determination of compliance was made as well as the date the form was signed.

The WG believes that allowing FAA designees and DOs to use the same form to be the preferred option. This would help reinforce the equivalency of the data when executed by a DO. The FAA should also consider whether electronic formats would be acceptable for transmitting this type of information to owner/operators.

IV.F.(2) DO Use of Previously Approved Data

The concept of previously approved 'substantiating' or 'descriptive' data implies that the data has an established pedigree with an FAA approval that makes the data more valuable than data without such a pedigree. However, no inference can be made with respect to the applicability of such data to another DA project unless a "determination of compliance" has been made by the DO. This does not mean that the data has to be regenerated, recalculated or retested by the DO, but no relief regarding the part 21 showing is implied with respect to the new project based on how it was used in the past. The DO will still have to assess the data to determine its applicability to the new project and to make a "determination of compliance" to the cert basis of the new certification project.

In addition, when a DO incorporates a TC'd or Technical Standard Order (TSO) component into its type design, it is only required to show that its type design is compliant with the installation of that component. There is no requirement for the installer to 'look behind' the design or compliance status of the component itself. A similar approach can be taken with STCs and PMA parts when they are installed by owner/operators. However, if a DO desires to make an STC or PMA part directly a part of the type design on which it is being installed, then the DO will need to obtain, or have access to, both the 'substantiating' and 'descriptive' data associated with the STC in order to make its 'determination of compliance'. Drawing this distinction in no way prevents a TC holder from installing an STC in its production line, but in such cases the STC is not a part of the DO's type design, but rather a

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

change to that design, installed at the time it is manufactured, but held by another entity.

IV.F.(3) Use of FAA's Delegation System

One of the basic principles developed for DO is that the FAA makes no discrete 'findings' of compliance. Design organizations are issued a DO certificate because they have a demonstrated engineering capability and commitment to compliance. This enables the FAA, using its discretionary authority, not to direct its resources to making numerous discrete compliance 'findings'. Instead, it can rely upon the DO's statement of compliance in making its overall compliance 'finding' when issuing a TC or other DA.

Since the FAA is making no discrete compliance 'findings', there is no basis for allowing the use of engineering designees, either within the DO itself or at its partners/suppliers. Designees are authorized only to perform tasks the FAA itself would otherwise perform. Since the FAA is not making any discrete 'findings' of compliance under the DO concept, there is nothing to delegate. Thus, the advantage to Industry of being able to make all 'determinations of compliance' is that the DO is not dependent upon the existing delegation system.

This does not mean that DOs cannot use individuals and companies that also hold FAA delegations, but those designees would be acting solely as a design supplier resource to the DO and any compliance determinations made by such suppliers must be conducted under a system determined acceptable by the DO. They are not acting as representatives of the FAA Administrator.

IV.F.(4) DO Recognition and Use of Design Suppliers with FAA Credentials (including other DOs)

A DO may take the status of an FAA designee or another DO into consideration when determining the appropriate method and level of supplier oversight it needs to perform. That oversight must be defined within the DO supplier procedures and must include both the qualification of that supplier and periodic oversight. In conducting its oversight of the supplier, the 'project DO' may include as one of its considerations the fact that the supplier is a designee of the FAA, but it must recognize that the FAA will not be conducting oversight on any non-delegation activity. If the supplier is another DO, the FAA will perform oversight of the 'supplier DO', but this does not relieve the 'project DO' of performing oversight.

Irrespective of the FAA credentials of a particular supplier, the 'project DO' must, under its CAS and SMS processes, assess and find acceptable the compliance and safety risk associated with its degree of reliance on this type of supplier. The

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

'project DO' must also be satisfied that these organizations or individuals are performing as expected, and must be aware of any FAA corrective action related to their performance. The 'project DO' could achieve this awareness by contractually requiring the designee or 'supplier DO' to provide records of any FAA corrective action, such as designee counseling letters or audit records.

A formal supplier interface agreement must exist between a 'project DO' and every 'supplier DO' providing it with 'determinations of compliance'. The interface document should address the scope of what the 'supplier DO' may accomplish for the 'project DO.' The document may authorize the 'supplier DO' to follow its existing DO procedures when making "determinations of compliance" associated with the 'project DO's' certification plan. The presence of a supplier interface document in no way relieves the 'project DO' from its showing compliance responsibility as an applicant.

V. CHANGE MANAGEMENT PLAN

This section is intended to contain a change management plan which may include industry and regulatory awareness initialives industry and regulatory training, coordination with other regulatory authorities (including potential changes to bilateral agreements), and a transition period. As of the original release of this report, a change management plan was not prepared in fieu of completing the development of the DO concepts contained in the prior sections. Further action is recommended to develop a change transgement plan as intended.

Denga Organization Working Group Biri 21/8MS Aviation Rulemaking Committee

- 64

Appendix A. DO Working Group Members

NAME

ORGANIZATION

Industry Representatives: Beck, Tony Chick, Mike Desroster, Walter Dramant Boustead, Judi Dickstein Jasen Dionne, Paul Fernandez, Félipe Eudes Pontes Grove, Robert Haerr, Mike Hill, Paul Hubbs, Chris Linder, Ben (WO Chais) Markhant, Pal Piotrowski, Denna Rogoranski, Trau Trusis, Rick

FAA Representatives:

Clourier, Paul Collins, Michael Cook, Bob Deutschman, Jason DiPaulo, Tomaso Geddie, Scon Jambor, Jacquellice Janco, Tony Law, Deug Linegang, Michael Quiles, Carlos

Foreign Civil Aviation Authority Representatives: Forguson, Bob Hall, Julian

Labor Representatives: Collins, Michael DiPaolo, Tomaso

15&S Dungan Aviation GAMA Pratt & Whitney, Canada MARPA Sikorsky Embraer Garman Rolls Royce GE Aireraft Enginem Rochwell Cellins Boeing THEREON BELAC Honeywell Gulfstream

AFS-540 Labor - NATCA A4R-200 A8W-150 Labor - NATCA A1R-140 ACE-112 AFS-304 A4R-120 A4R-120 A4R-120 A4FS-340

Transport Canada EASA

NATCA NATCA

Design Organization Working Group Pan 21/SMS Availon Bulenoking Committee

Appendix B. DO WG Charter

Part 21:5MS ARC. Charter for the Organizational Workley Group

1 Scope and Boundaries

 The Working Groun (WG) shall define a Transporting environment (e.g. ground by the First Multimedi segmentation and system measurements and priority is write the The Second Organization angular of making a angular disemption with which the TAX page felty.

- To accompled, the 405 will produce a Report Recommende door gluppers land) and question 32 (Delwarders)
- 2. The logort will more in official to EdiA organization of systematics (4) (10) (not /)
- The Keren will monoscillance will be boundaries of costing C.S. subject for
- b. The Will preference as used that the User of reprintment theorem and will include used in patient relations of the F/A that could entryfere it indices and by the F/A to do a part of the first of the data and by the F/A to do a part of the first of the data and by the F/A to do a part of the first of the data and by the F/A to do a part of the first of the data and by the F/A to do a part of the first of the data and by the da
- Discussion of windows it is appropriate to investigate the equilation of the first the world by a data or organization and by our of so you dow Alice may denote the program of the program control meets.

2 Assumptions and Guidance

2.1 Assumptions

- Einike tim mereri FAA vegulanee-
- 1 That # Auf-issues desigh approvalia-
- The EAA may defegue the ensurements, testing and may compare any processing without providence matching. EAA may delegative output of a question contingent.
- Chain an or second do leave a company representation of appropriate or to Adappend approval.

b. The EAA should have been acount organized interview to that the EAA may import constrained have years to a start of a second seco

- [South Transport Sygnetizes at shall be prepared to have some solved senses to merger using states (some solves).
- Cash Design Organization sour have note on the Entroy Operational Solicy responsibility, and the inferences of the address functional and any start of the WC process.
- 17 The manufaction of a state of the spectra of the second second system of a state of the test of test of
- 1. The preparad registration shall an even with Dimans sty promition over hall rewriting a world multilifer more than a Caronichia ce Tor 10 Profil the develop recessory to must a perifician or approval.
 - The proposed repeation shall analogize the Design Linguisticities were find privilegies would have the privilegies of source a performance approach.
- The FAA will develop an overagilar will administration with covered and desage opportunities will combine an evenight pendot to developed by the Overagits (Vc)).
 - The EAA shall respond to plotting to these information to the of second model due to AAA finds, appropriate based on the AAAA model wild be addressed to standing program and togot (10° office processes, 6 and 10° of standard to 10° o (2.4.4)
- Implementation is table in Section devices plan in Acaty SEAA, gating to relation or and record system and any temport? In terms. Reference for some compared with the rest of the rest on only an att 7m of any her work? The rest of some at 5m of gates longer compared to attempts and attempts of the rest of the rest of gates longer compared to attempts at the rest of the rest of the rest of gates longer compared to attempts at the rest of the rest of the rest of gates longer compared to attempts at the rest of the rest of the rest of gates longer compared to attempt of the rest of the r

Commentation of the Art of Art of Arts

ι.

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

Part 21 SMS ARC

Charity for the Organitational Working Group

2.2 Guidance

- The KAA is contributed with the siles of a system more in third party or thirm party authors conduct, emplanue, audits of Design Organizations, and in which the FAA may version the results of such audits in application (in FAA) = consult to not all delegating the proportion of the regarding correct().
- 1. The WC entry wave to relax in the h/s/s revolution setuplines regulations as a medicilly for how or by the design regulations and transformation medical. The scale (3.4 CE-10.22.13 CE) are applied by the requirements, and entry and regulations of the CE-10. CE and CE and CE are applied by the transformation regulation medical by the transformation of the CE-10.22.13 CE are applied by the transformation of the CE-10.22.13 CE are applied by the transformation of the CE-10.22.13 CE are applied by the transformation of the CE-10.22.13 CE are applied by the transformation of the CE-10.22.13 CE are applied by the transformation of the CE-10.22.13 CE are applied by the transformation of the CE-10.22.13 CE are applied by the transformation of the CE-10.22.13 CE are applied by the transformation of the CE-10.22.13 CE are applied by the transformation of the CE-10.22.13 CE are applied by the transformation of the CE-10.22.13 CE are applied by the transformation of the CE-10.22.13 CE are applied by the transformation of the CE-10.22.13 CE are applied by the transformation of the CE-10.22.13 CE are applied by the transformation of the CE-10.22.13 CE are applied by the transformation of the CE-10.22.13 CE are applied by the transformation of the transformation of
- () The WA may want to othe to the TECA Design Dignation on regulations for guiden-
- 4 The BVT star man to order to the EATA Origin Organization Astronomics organization for goaline (
- The ARC hads dow despinencement are impropriate for time which wild fine following downs approval: TV, 407, 4545 and 51025

3 Tasking

3.1 Relevant Questions

- Conv Classiciani. Any exist is these spectrum are maint to owner hanglos do as the international data sources another them alreadying the proposition game with the Var's Department's of the latter resources and all there assessments that is should be based operating assessment the base of the Data converse statements or the data (bay easy be included; the Depart). The based questions are data.
- 1. The ABC/managered One are generated as a spectrum to be detered to a determine the second seco
- 2. Which the purpose of a design organization one species social benefits? Soliday Exercises with the automation performance of the automation performance of the automation of the automatio
- 3. What provide end only the afforded to design representation? If there are defined in over of design organizations, then have be the providence of and Afford.
- A first at the second or dependence what e Q and Q monoton of each have to ensure a debth as responsible as not effectively ensures at providence.
- 3. Altria non all all'aren dange organizzano banda ena "Altrial Seni tradime anche a sumantand bar fan une applaante a narmal avai da regiong comprises og altrigen two hat genomen of hom hum gib net processis sugged traditionale are process and a significant measure tool (a) - processo hum gib net processis sugged traditionale are process and a significant measure tool (a) - processo hum gib net processis sugged traditionale are processis on a significant measure tool (a) - processo hum gib net processis sugged traditionale are processis (b).

In Assument Querrary. These de side periods development and dealing since one has not needed a group memory accordingly accorded to these spectrams. These gas some should be peak is utilitated a someting the peak logent. Where the state of a side site fact and so the fact and so the state of a side site fact and so the fact and so the state of a side site.

- 1. Or address a spectral backlet or similar to save a similar program case is some at what singly a small different favore for using organizations. All other point profession devices approval approach devices favore at the device favore favore and the general sector favore for a profession of a sector favore for a profession of the device favore f
- 3) Which are group with balance which there is the group argument of "Do some dialogs opposed helders" (1) made before arguments or "Lo example, as applicants is field a spectral disease. repeated on on."
- 3. True will patient stating encoder or stems to when some interface straight organizations .
- 4. (This papers in additioned has a free as a first the post topes and beserve approaching).

Terraneous Collans, pharty of Solidan

64

Design Organization Working Group Part 21/SMS Aviation Ralemaking Committee

Part 21/SMS ABC Charter for the Organizational Working Group

- How will us must be the providence of the state of the set of the set of the state of the function.
- 9 What increases all assesses would be appreciate to adjust the privilege and componential quart

3.2 Deliverables

- 8 The Wool multi-assume to perpending altitude (comparing posed) could investigate of our registration and a resting 14 (17 %, Part 31) observe of the United Science. The regulations and I multicle.
 - 1. The or more drough organization models
 - (a) DVInne crows in regenations
 - (b) Organity experimentaries to obtaine each level of design organization to registeriors
 - (c) containing the second of the second seco
 - (a) Den lagar associated with hud-law int darges again association gainer
 - l = l containing or planning programs for these stands with these design regards are recognized.
 - 1. A contractor provincion
 - 4 Metrica (a) devices many masses of the charge on-seminary types are constant of a the travelant person on
- Fight WG, must delive a proposed change transportant plan. The course monogramming plan WG have been broad transplate that results when compressed by the FAA and weddillower deback. The change monogramming plan may reduck.
 - 3 Juliany approximate primating reads, and trade organize and reading
 -) Pepulators retain constitutes, sceparing moder and lates gamming an observation
 - Industry inserting
 - A. Regulatory presents
 - 5. Conditions with one automass including possible charges to beaved assumption
 - A Strattory proof

3.3 Milestones and Dates

Date	Aerom	Representat Distiversities
8mi (3,4,2003)	Leve WiS Marton: When any De-wypertor insertion and proposed of a Cleaner. Organization Weiner using a Cognitization of a web to a decision the experimit beneficies and complete.	Let a permit a Double of a set approximate approximate the permitted of a set approximate and a comparison of the set of the permitted of the permitted of the set of the set of a set of the set of the will set of the set of the set of the set of the set of the million of the set of the
Appin (2, 293)	Wis Members in Austing how of privalegal	Luna di mandana mineritati da the Chair na 1719 agai 12 (Chair aith dane bahar App) Pirmen ng
A(0.02, 001 fame).	Tringhamed Wei Mooting: What provide an along a Design Opportunities extended in a play to officiency by searching or provide latter.) lat of proof go a concreted with dying) algorization that works <u>pro-topy</u> of the responsible

Commence Wardsmark address of a statistic

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee 65

ч.

114tr	Action	Expected Differences		
vies (30, 29(3	For the second statement when statements and the second statements (approximation) area is under to detection, and that the second statements and pressure that participant	Ching 6, and 6		
wareld, 2003	Transienes With Monteng With a Addings in news sylphoid? Working 10.5 science with the TC 1754 one W? Working 10.5 science perge- replanation of the signation			
kine (6,200+	Tagge Daves Taive Keismen, Association of first diath ordered of the registance factories of the registance factories of the second	Evolution on compare or where the Decision in granitation or analy for writen the regulation of our regimization for our equip- tion work of the SMU to us? I our could prove proper coust and provide or of the Techny Compared and Share with With AWT and AIX 100		
(ups 5-3)	Will, having the Design Organization.	Henders among comment on Design. Department finitive by June 20		
Afference of Sec. 22	 Charles Dalamarca, Deal 76A.2, ADR-010 Albard Drame ASS - 400 and Security Equipment of endings, "every security of the term Ch.A. approximation of the term 	(Benn F.), A Albolov, AB-20, and AFS 300 Connects of Design (Agreeming) (Agree		
lune	WO Live Manting: Toxican Discon- Dynamics Office	Research appropriate and too on an Learning Transmission of the		
July 2003	Construction of the second sec	(Dominion with 2019 WO		
ກໍພື້ມມີຮະສານ	AliCatesting	AbCommune of Opportune Station		
1007876201	Tiger Level Take Manthey Court Design Communities (April) and Of Callon and Web Compilers	Lines of Datasang		
/mgan 205	Wei the Charlenge Discon Dealt Things (Approximation Depart	Estand Dynifi (d' Uppenni)		
Servenber 2001	(Table - Odepart And Indonesia A A Adds-104 (Alight Gaar A Palline in Table Theorem (BRACE) (Series Construction Report (Series (Series Odepart)) - FA Supressment of the factor	(Innus 5 AA fanta) (Innum on DEAFT Theory Doponosition (Innut		
Remaining 2011	Tese Deligner Missower Mills Wo	state language of doubling out		

Part 21/5335 Alif. Charter for the Oryanizational Working Group

which is the state of the state

31

00

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

Dar	Asten	Expects 0276 (creater
	Charles and the second se	
September 3167	A Year Discussion March and Changes Web Characterisation station makes work to an an Characterisation with Characterist Web Characterist Statistics (MCT 100 m)	Combination with Even physics Web Condition of Standard pharelog physics Conditions with Conditional Web 5 provide Analis Start
Separating 17, 3011	ARC MEMORY	ARTS www.anunus.co.kuh.layor.
legnnolis; UE-10 2054	W9 Live Minering Trocare allowing and provident of the Control Organization allows	TerminantBuyOwla
lary familier 301	Tigge Town I are Maring Develop a second or planetary the Design or presidences Report from the Will colling	(Promotion Plan
foundestiles 2015	Web Lose Munitipg to edit and thealf or Disagnet/Opportunity in Report	Talogs/ operation (cent)
Anishipped 107 2013	Timu (Operat Dou ARC counting:	County Report From ARC provide County Instant

Part 21/SMS ABC: Charter for the Organizational Working Group

3.4 WG Integration

- customers of synethet behalts and a south deate of the proposed constance to some other others for the transfer Parally dealers (W).
- Gaugestion a Talley Foregroup Rynamic and surger than some Organization of the 0-D decision of the second s
-) Upward WWP well-dates to critical for Low Collaboration of well the Decourse of Manuality of Cognition on

4 References

- 1 Filo pdp? logent
- In EALADOA/ARDUA Reservation
- / 0.23 have at inversion reportation
- 5 WG Lead(s) and Members

contraction of the state of the

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

Appendix C. Glossary of Terms

ACCOUNTABILITY FRAMEWORK: An established set of responsibilities and commitments of the FAA and Industry

AGENT DO: A DO that has been contracted by a DA applicam to manage a certification project

APPROVED DATA: Descriptive data approved by FAA employees, its designees, or in accordance with Part 1

AUDIT: Formal scheduled review by the FAA or the DO of the DO's processes, projects, and compliance with DO regulatory requirements, as determined by the FAA or the DO internal audit function. It is expected to include some review of compliance findings on closed projects. For the FAA, audit is a part of FAA's Certificate Management function.

DO EXECUTIVE: The company individual directly responsible for ensuring that the DO meets all of its regulatory responsibilities

DO POINT(s) OF CONTACT: The individual(s) within the DO responsible for all communications with the FAA

CERTIFICATE MANAGEMENT FAA actions to monitor the DO certificate holder and to determine the holder's compliance with the provisions of its certificate(s)

COMPLIANCE ASSURANCE SYSTEM (CAS): DO holder a system for maintaining design configuration control and ensuring that it complies with the applicable regulations.

COMPLIANCE FINDING: FAA decision (either directly or through a designee) that compliance has been shown with the applicable airworthiness standards

CULTURE OF COMPLIANCE: Knowledge beliefs anitudes, and behaviors of an organization that are focused on ensuring regulatory compliance within all its activities.

DESCRIPTIVE DATA; Data that defines the type design that needs to be determined compliant to the applicable airworthiness standards. The descriptive data is what is approved by the FAA when a DA certificate is issued.

DETERMINATION OF COMPLIANCE: A decision made by the DO that compliance has been shown and verified with the applicable airworthiness standards. It may also be a decision made by the DO that data previously used as part of another

> Design Organization Working Group Furt 21/SMS Avronon Rolemaking Committee

project are applicable to the design of the new product, part, or appliance for which it is to be used, including the applicable certification or approval basis

ELIGIBLE DATA: Data developed under an assumed, but not FAA established, certification basis, and product type design if appropriate.

FINDING OF COMPLIANCE. FAA decision that the applicable regulatory requirements/airworthiness standards have been mer. Statutorily required action to issue a certificate.

PROJECT DO The DO responsible for overall integration of a specific certification project. The project DO ensures all determinations of compliance are complete and may be the DA applicant or may act on behalf of a DA applicant (i.e. contracted as an agent DO).

SAFETY CULTURE: The product of individual and group values, attitudes, competencies, and patterns of behavior that determine the commutment to an organization's safety programs.

SAFETY MANAGEMENT) The act of understanding and making decisions and taking actions to lower risk, inherent in all human activity, to acceptable levels

SAFETY MANAGEMENT SYSTEM: An integrated collection of processes procedures, and programs that ensures a formalized and proactive approach to system safety through risk management. Risk analysis and assessment are required for all changes to identify safety impacts. The SMS is a closed-loop system, ansumg all changes are documented and all problems or issues are tracked to conclusion. When properly implemented, an SMS establishes a safety philosophy or culture that permeates the entire organization in the monitoring and continuous improvement of safety.

SCOPE OF AUTHORITY: A combination of explicit statements of the DO Cartificate and Terms of Approval that identify the capability of the DO and to which category of aircraft (e.g., large airplanes, commuter airplanes, etc.) and products that the DO is approved to conduct its operations

SENIOR COMPANY MANAGEMENT: Those in the company management chain above the DO Executive who are accountable for the actions of the DO.

SHOWING OF COMPLIANCE: An applicant's demonstration of compliance to an airworthiness standard (i.e., 14 CFR 23, 25, 27, 29, 33, 35) and 18 comprised of substantiating data, statements, and/or other acceptable methods of demonstration (e.g., acceptable MOC).

STATEMENT OF COMPLIANCE: A statement from the DQ to the Administrator certifying that compliance with the applicable airworthiness standards for the project

Design Organization Working Group Bin 21/SMS Aviation Rulemaking Committee

has been determined in accordance with the procedures listed in its FAA-approved DO Procedures Manual. With this statement, the DO is indicating that the project is complete and ready for the FAA to issue the certificate or DA.

SUBSTANTIATING DATA: Documentation related to a DA applicant's showing of compliance to the applicable airworthiness standards.

SUPPLIER DO: A separate DO entity in its own right that provides engineering data to a project DO. The supplied data supports the project DO's certification project.

VERIFICATION OF COMPLIANCE: An independent check, or equivalent, of the showing of compliance leading to a DO determination of compliance.

Appendix D. List of Acronyms

AC	Advisory Circular
AD	Airworthmess Directive
AEE	FAA Office of Environment and Energy
ARG	Aircraft Evaluation Group
ARC	Avminon Rulemaking Committee
CAA	Civil Aviation Anthority of mather country
CAS	Compliance Assurance System
CCS	Compliance Certification System
CDO	Certified Design Organization
CFR	Code of Federal Regulation=
COS	Continued Operational Safety
CPN	Certification Project Notification
DO	Design Organization
DA	Design Approval
DAH	Design Approval Holder
DMS	Design Management System
DOA	Design Organization Approval
DPO	Design Production Organization
EASA	European Ayrabon Safety Agency
ELOS	Equivalent Level of Safety
EPA	Environmental Protection Agency
FAA	Federal Aviation Administration
PME	Failure Modes Effects
FMEA	Failure Modes and Effects Analysis
FOEB	Flight Operations Evaluation Board
FSB	Flight Standardization Board
ICA	Instructions for Continued Anworthiness
ICAO	International Civil Aviation Organization
112	Issue Paper
LOPI	Level of Project Involvement

Denga Organization Working Group Part 21/SMS Aviation Rulemaking Committee

MOC	Means of Compliance
MRB	Maintenance Review Board
NHTSA	National Highway Transportation Safety Administration
NPRM	Notice of Proposed Rulemaking
OCS	Organization Control System
ODA	Organization Designation Authorization
PMA	Parts Manufacturer Approval
POC	Point(s) of Contact
PSE	Principal Structural Element
SC	Special Condition
SFAR	Special Federal Aviation Regulation
SMS	Safety Management System
SSP	Specialty Service Provider
STC	Supplemental Type Certificate
ТС	Type Certificate
TSO	Technical Standard Order
TSOA	Technical Standard Order Approval
US	United States
USC	United States Code
WG	Working Group

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

Appendix E. DO Challenges, Benefits and Concerns

The WG began discussions by identifying various challenges, benefits, and concerns with the existing system and the move to a DO model. The following challenges: benefits and concerns were identified as a reference for the issues that might need to be addressed. It merely represents a starting point for the WG that led to the WG report. The WG report contains the final product, without a revision to the initial discussion topics captured in this Appendix. This Appendix is intended to serve only as a reference to the initial thoughts. Note that these initial thoughts may have been captured prior to full partnepution of the WG members listed in Appendix A.

General

- Challenge F Meeting SMS requirements established by International Civil Aviation Organization (ICAO) A regulatory SMS for design activity cannot be implemented without regulatory recognition of a design organization. SMS cannot be applied to a FAA-delegated organization.
 - Benefit: A DO model would enable implementation of SMS by FAA on design organizations as required by ICAO and FAA requirements, such that no differences would be filed with ICAO.
 - Concern. The benefits of the DO model will not be commensurate with or greater than the potential regulatory SMS burden imposed on the design organization.

FAA

- Challenge 2: Any applicant, regardless of qualifications, may apply for a DA under existing requirements. The FAA is experiencing a significant increase in the number of STC applications by persons who do not currently hold a DA and are unfamiliar with applicant and holder responsibilities.
 - Benefit: DO approval/certification/oversight would give the FAA a mechanism to require minimum applicant and holder qualifications
 - Concern. The DO model may not be flexible enough to support small business
 operations, the establishment of new companies, and a range of capability levels
 appropriate to the range of aviation products being designed, especially new or
 novel designs.
- Challenge 3: Existing practices rely heavily on the "show/find" actions occurring at the end of the design process, regardless of any compliance design processes that may exist.
 - Benefit: DO approval/certification/oversight would give the FAA a mechanism to require compliance assurance within the design processes.

Design Organization Working Group Pin 21/SMS Aviation Rulemaking Committee

.0

- **Concern:** FAA approval of the DO manual may be used to restrict a DO from choosing where to place an emphasis in compliance processes (e.g., some companies may wish to emphasize early processes for assuring compliance, while others may rely more heavily on late-term processes for verifying compliance).
- **Challenge 4:** FAA currently expends significant resources making discrete compliance findings related to specific products and supporting DAs. Future federal budget expectations do not allow for growth in the federal workforce, so growth of the aviation industry requires a different approach.
 - **Benefit:** DO approval/certification/oversight would give the FAA a mechanism to focus on the system used for compliance assurance and aviation system safety. Additionally, the DO model provides the opportunity for greater capacity in the system.
 - **Concern**: Because the FAA does not fully utilize its ability to rely on ODA to remove itself from discrete compliance findings, FAA and ODA holders do not receive the perceived benefits associated with the scope of the ODA. Similarly, DO implementation may have similar issues and a reduced benefit relative to the notion of a fully-delegated ODA.
- Challenge 5: Once a DA has been issued, the only mandatory actions FAA can take to address aircraft design compliance or safety issues is to issue ADs against products or continued airworthiness actions as defined in 14 CFR part 26. ADs do not address root cause (i.e., DAH practices that may have created an unsafe condition). Other than under ODA, the FAA cannot take certificate action against non-compliant DAH practices without affecting the eligibility of products manufactured by that holder for operation. In other words, FAA certificate action has significant downstream effects that may be considered inappropriate.
 - **Benefit 1:** DO approval/certification/oversight creates an accountable organization which may be held responsible for mandatory action to address safety issues.
 - **Concern 1**: It is unclear how a DO addresses this challenge differently or better than an ODA.
 - **Benefit 2:** DO approval/certification/oversight opens up new opportunities for implementing safety improvements that do not rise to the level of an unsafe condition.
 - **Concern 2:** FAA may use the opportunity of DO approval and oversight to mandate perceived improvements that are above and beyond actual regulatory requirements.
 - **Benefit 3:** DOs reinforce both a safe culture within a company and the accountability framework addressed in Part 21 and encourage a reactive compliance culture to become a proactive compliance culture.

Industry:

- Challenge 6: Industry control of development and certification schedules is difficult. Delays in certification plan approval and Issue Paper (IP) closure generate additional uncertainty on program schedule and costs. DA applicants cannot currently rely on internal capabilities for determining compliance (i.e., applicant compliance capabilities on any given project are always subject to question by the FAA and FAA inquiries result in product delivery schedule impacts). Currently, for a new airplane TC, Special Conditions (SC), Equivalent Level of Safety (ELOS), and specific agreed MOC are generated. In some cases IPs may take as much as two to three years to close. Delays in IP closures (Stage 4) generate uncertainty on program schedule and cost.
 - **Benefit:** DO would give an applicant greater control over product delivery schedule. Moving the FAA from specific compliance findings to actual governmental actions is in the direction of best use of FAA resources to achieve safer products.
 - **Concern**: The FAA may be reluctant to rely on applicant capabilities (see Challenge 9). Additionally, agreement on the MOC may still delay projects.
- **Challenge 7:** Accountable applicants are currently unable to rely on the acceptability of supplier or consortium-member contributions to certification.
 - **Benefit:** DO would give the accountable applicant an independent mechanism to assure the validity of supplier or consortium-member's contributions to aircraft certification.
 - **Concern**: Supplier DOs and consortiums with member company DOs may confuse the accountability picture. A single point of accountability is still needed. Accountability still needs to be clearly attributable to the appropriate applicant organization.
 - **Concern**: The FAA will likely not have adequate resources to oversee multiple suppliers or complex consortiums. As such, product applicants must be responsible for suppliers.
- Challenge 8: ODA places limitations on the flexibility of a company's processes.
 - **Benefit:** DO should afford greater flexibility in company processes for assuring compliance.

Other tangential challenges:

- **Challenge 9:** FAA culture is hesitant to fully rely on applicant accountability for certificate issuance.
 - **Benefit:** DO certification and oversight would provide the FAA both a predesign-approval and post-design-approval opportunity to assess the knowledge, capabilities and practices of aircraft certification design organizations.

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

- **Concern**: The DO model may not be fully utilized by the FAA. The limit of this WG's ability to affect this is to provide a recommendation to FAA management regarding the benefits and need for the DO model. Significant cultural changes within the FAA and industry will be required to fully realize DO benefits.
- **Challenge 10:** No requirements currently exist for STC applicants to assess the compatibility of their designs with other STCs.
 - **Benefit:** DOs could be qualified for the privilege of supporting compatibility assessments for integration of multiple STCs.
 - **Concern:** Requirements for this privilege have not yet been identified. This may be out of scope for this WG.
- **Challenge 11:** Industry has significant concern over the lack of consistency in FAA LOPI decisions. There are three significant areas where LOPI decision are inconsistent:
 - 1) FAA involvement in compliance activities not related to safety critical areas;
 - 2) FAA involvement in activities where the applicant has full competence to make compliance determination; and
 - 3) FAA involvement at the level of approving documents such as test plans or results.

Industry members believe these are neither the proper nor efficient means of FAA involvement.

- **Benefit:** DO would provide a better model for the FAA to recognize capabilities and establish a consistent risk-based LOPI framework.
- Concern: This may be out of scope for this WG.





Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee



Figure F-2 - Secondary Threshold Determination

78

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee



Figure F-3 - Result Assessment

Design Organization Working Group Part 21/SMS Aviation Rulemaking Committee

APPENDIX G—SAFETY MANAGEMENT SYSTEMS WORKING GROUP REPORT

21ARC Working Document - Not for Distribution

FAA Part 21 / Safety Management Systems Aviation Rulemaking Committee (Part 21 / SMS-ARC)

Safety Management Systems Working Group Report

January 17, 2014

PAR	11 / SMS-Aviation Rulemaking Communee SMS Working Group Report	Page Revision Date:	1 of ; Original Jan 17, 2014
		: C	
TABLE OF	CONTENTS		_
SECTION 1.0	Executive Summary		
1.1 Safety N	fanagement System Working Group (SMS WG) Overv	iew	
1.2 Summar	y of Recommendations		
1.3 Addition	al Recommendations		
SECTION 2.0	SMS WG Charter / Membership / Taskings		
2.1 SMS W	G Charter		
2,2 SMS W	G Membership		3
2,3 SMS W	G Taskings		
2.4 SMS W	5 Meetings	(1+++()1)(+++)	
SECTION 3.0	Key Assumptions		
SECTION 4.0	Responses and Recommendations to Taskings and	Charter	
4.1 Task I:	Regulatory Gap Assessment		
4.11 78	sk 1 SMS WG Response		
412 Ta	sk 1 SMS WG Conclusion		
4.2 Task 21	Incorporation of SMS into 14 CFR		E
42.1 Ta	sk 2 SMS WG Response		
4.2.2 Ta	sk 2 §5 27 Coordination of Emergency Response Plan	ning	
4.2.3 Ta	sk 2 SMS WG Recommendation		
4.3 Tasks 3	and 5 SMS Applicability, Scalability, and Flexibility		
4.3.1 Ta	sks 3 and 5: SMS WG Response		
4.3.2 Ta	sks 3 and 5. SMS WG Recommendation.		
4.4 Task 4	Recommended Preamble, Policy, and Guidance Materia	il	
4.4) Ta	sk 4 SMS WG Response		
4.4.2 Ta	sk 4 SMS WG Recommendation		
4.5 Task 6 a	nd 7 CONOPS for SMS		
4.5.1 CO	ONOPS Summary		
4.5.2 Ta	sk 6 and 7 SMS WG Response		
4.5.3 Ta	sks 6 and 7: SMS WG Recommendation		
4.6 Task 8 and CB/	Coordinate SMS WG Efforts with Organizational WG, A WG	Oversight W	G. 33
4.7 Task 9:	14 CFR 21 3 for Organizations with an SMS		

Part 11 / SMS-Aviation Rulemaking Communee SMS Working Group Report		2 of Original Jan 17, 2014
4 7.1 SMS WG Response		
4.7.2 Task 9 SMS WG Recommendation		
SECTION 5.0 Additional Recommendations		
5.1 Acceptable Safety Risk		40
5.1.1 SMS WG Recommendation		40
52 Availability of Data for SRM		40
5.2.1 SMS WG Recommendation		
SECTION 6.0 APPENDICES		

15:10

SECTION 1.0 Executive Summary

1.1 Safety Management System Working Group (SMS WG) Overview

The Safety Management System Working Group (SMS WG) was chartered by the Part 2)/SMS Aviation Rutemaking Committee (ARC) to provide recommendations and guidance to the Federal Aviation Administration (FAA) for development and implementation of SMS rulemaking applicable to organizations involved in the Design and Manufacturing (D&M) of aviation products and articles. The charter included several taskings each of which have been addressed by the SMS WG SMS WG membership included individuals with expertise in SMS representing organizations for the design and manufacture of aviation products, articles, and replacement parts, association representatives on behalf of general aviation, and contributors from the FAA along with observers from European Aviation Safety Agency (EASA). Transport Canada Civil Aviation (TCCA), and National Civil Aviation Agency - Brazil (ANAC) Nine meetings were held in 2013 and 2014 during which the SMS WG addressed the taskings and developed recommendations.

1.2 Summary of Recommendations

The following is a summary listing of SMS WG recommendations in response to the charter and associated taskings:

- 14 CFR Incorporation: The SMS WG has determined that the SMS requirements in the proposed 14 Code of Federal Regulations (CFR) Part 5 (Docket Number: FAA-2009-0671) (hereafter referred to as "Part 5"), with consideration given to D&M sector comments for recommended changes contained in the Docket, are appropriate for D&M organizations. In addition, the 8M8 WG evaluated 65.27 (Coordination of Emergency Response Planning). and determined it is not necessary for D&M organizations. Therefore, we recommend that FAA modify 14 CER Part 21 (hereafter referred to as "Part 21") to make Part 5, excluding \$5.27, the SMS requirements for organizations meeting the applicability threshold. (Refer to section 4.2)
- Applicability: The SMS WG recommends that the FAA develop an applicability threshold that requires SMS for organizations that
 - Design or manufacture products (), e., aircraft, engines, propellers) or,
 - Design or manufacture articles whose failure could directly prevent continued safe flight and landing; or
 - Make design changes to a product, through a Supplemental Type Certificate (STC). failure of which could directly prevent continued safe flight and landing.

This recommendation is not intended to discourage volumary implementation of SMS for organizations producing articles with criticality falling below the applicability threshold. (Refer to section 4.4)

Scalability: Scalability of SMS has been addressed through the applicability threshold Therefore, the SMS WG recommends that all Part 5 SMS components / elements (with the exception of \$5.27) be applied for those design and/or manufacturing organizations

Part 21 / SMS-Axiation Bulenaking Communica	Page Revision	1 of 5 Ungmil
SMS Working Group Report	15:00	Ian 17, 20114

that meet the applicability threshold. Ukefor to section 4-31

- Policy and Guidance Material: The SMS WG developed SMS regulatory material and a basis for preamble, policy, and guidance material, as provided in this report, however, we determined that more work was necessary to produce supporting guidance material. We recommend that the SMS WG continue to develop supporting guidance material with an objective to complete the task by October 2014 by providing an addendum to this report. (Refer to section 4.4)
- SMS Concept of Operations: The SMS WG has developed a Concept of Operations (CONOPS) which describes the intent of the Part 5 SMS framework (Safety Policy Safety Risk Management, Safety Assurance, and Safety Promotion) for D&M organizations as it applies to each life cycle phase (design and certification, production and airworthiness certification, and continued airworthiness) of a product or article. We recommend that the CONOPS form the basis for the development of preamble, policy, and guidance material. The SMS WG also recommends that, as described in the CONOPS, existing processes and procedures be considered as meeting the intent of Part 5.

(Refer to section 4.5)

 14 CFR 21.3 for SMS: The SMS WG has determined that 5213 (Reporting of Failures, Maffunctions, or Defects) should be updated. We recommend the FAA task an ARC subteant to develop recommendations to minimize or eliminate redundant and inconsistent reporting, record keeping, and risk assessment requirements for organizations with an SMS. Additionally, the sub-team should address appropriate reporting requirements and systems to enable the FAA to exercise its oversight and system level risk management responsibilities (Reference action 4.7).

1.3 Additional Recommendations

The following is a summary list of additional SMS WG recommendations which were not directly ned to the charter or taskings:

- Acceptable Safety Risk Determination: The SMS WG recommends FAA establish WGs to develop risk acceptance criteria for products and articles. Advisory Circular (AC) 39-8 provides one example of a risk acceptance criteria that has been developed [for Transport category engines and Auxiliary Power Units (APUs)] to conjunction with the D&M industry and is mutually accepted by FAA and industry (§5-55) (Refer to section 5.1)
- Availability of Data for Safety Risk Management (SRM): The SMS WG recommends that the FAA develop an approach to make fleet data already provided to the FAA (hours flights, reported failures, multianctions, and defects and service difficulty reports) readily available to D&M organizations, in support of executing SRM (§5.71). (Refer to section 5.2)

Page

13:00

SECTION 2.0 SMS WG Charter / Membership / Taskings

The Part 21 / SMS-ARC developed a draft charter, membership, and tasking list for the SMS WG. These were further developed by the SMS WG and submitted to the full ARC for approval. Summaries of the charter, membership, taskings, and meeting schedules are provided in the following sections

2.1 SMS WG Charter

The SMS WG was chartered (refer to Appendix A) by the Part 21 / SMS-ARC to develop a report which provides comments and recommendations for developing and implementing SMS regulatory requirements in a D&M environment. Key charter deliverables include the following:

- Recommendations for rulemaking, suggested processes, policies, and auidance to align Parr 21 with the SMS requirements documented in Part 5.
- Identification and assessment of any differences between the recently adopted SMS. requirements in International Civil Aviation Organization (ICAO) Annex 19 and Part 5
- Evaluation of options and provide recommendations to the FAA on the optimum solution for the application of SMS to D&M organizations.
- Recommendation as to which Design Approval Holder (DAH) and Production Approval Holder (PAH) organizations should have SMS requirements applied to them. This recommendation should include the scalability of SMS to D&M organizations based on their size, complexity, and safety risk introduced by their products-

2.2 SMS WG Membership

The SMS WG membership (refer to Appendix, B) is comprised of a diverse group of individuals with expertise in SMS and related subject matter areas representing organizations regulated under Part 21 for the D&M of type certificated aircraft and engines, approved avionics articles and systems, association representatives on behalf of general aviation and modification and replacement part manufacturers, contributors from the FAA Aircraft Certification Service, and observers from EASA, TCCA, and ANAC

2.3 SMS WG Taskings

The SMS WG was tasked by the Part 21 / SMS-ARC to complete Task= 1 through 8 as shown in sections 4.1 Imough 4.6. The SMS WG added Task 9 during the course of our meetings. The nine tasks we summarized as follows

- Perform a regulatory gap analysis between the SMS requirements defined in Part 5, ICAO Annes 19, and Part 21.
- Develop, evaluate options, and make a recommendation(s). For incorporation of SMS inte-14 CFR for D&M organizations
- Provide a recommendation as to which D&M approval holder organizations should have ж. SMS requirements applied to them

Part 21/5405-Aciation Rulemaking Communi-	Page	h of =
SMS Working Group Report	(Jato)	Jan 17, 2014

- 4 Produce preamble, regulatory, and gardance material for SMS application to D&M organizations, include an "operational" definition of a hazard throughout the life cycle of a product in SRM.
- Provide guidance that accounts for scalability of SMS to D&M organizations based on their size, complexity, and safety risk introduced by their products.
- 6 Define the end state (roles, relationships, and responsibilities) for SMS in a D&M organization considering design, certification, production, and in-service support
- Perform a review of "strategic" safety decisions that are required to be made during the design, certification, and production phases throughout the life cycle of the product
- 8 Coordinate SMS WG activities with Part 21 ARC and supporting WGs [Oversight, Cost-Benefit-Analysis (CBA), and Organizational]
- Provide a recommendation as to what \$21.3 should look like for a D&M organization with a SMS.

2.4 SMS WG Meetings

The SMS WG conducted nine meetings in 2013 and 2014 to address the charter and associated taskings. These meetings were held on the dates and locations identified in the following paragraph and represented over 100 hours of detailed discussion. WG members attended either in person or by phone with web link.

- 1. Kickoff Meeting: April 4-5, 2013. Washington, DC
- 2 WG Meeting: April 30-May 1 2013. Phoenix, AZ
- 3 WG Meeting: June 11-12, 2013; Seattle, WA
- 4 WG Meeting, July 17-18, 2013; Washington, DC
- 3 WG Meeting, August 13-14, 2013; Wichita, KS
- 6 WG Meeting, September 18-19, 2013; Washington, DC
- 7 WG Meeting: October 30-31, 2013; Hartford, CT
- 8. WG Meeting, December 10-11, 2013; Washington, DC
- 99 Final Meeting: January 13-15, 2014; Phoenix, AZ

Page

13:00

SECTION 3.0 Key Assumptions

The SMS WG has documented the following assumptions which were key to our discussions and recommendations

- 1 The SMS requirements considered by the SMS WG are as written in Part 5 (including vousiduration of the D&M sector comments in response to the Notice of Proposed Rulemaking (NPRM) [Docket No. FAA-2009-0671: Notice No. 10-15]).
 - NOTE If this assumption is shown to be invalid, the recommendations of the SMS WG should be reviewed by industry against the new Part 5.
- 2 The Organizational WG will set the applicability requirements for D&M organizations.

SMS is a management system that is described by assigned responsibilities and the functional relationship of those responsible for safety to management and other organizational components.

- The Organizational WG will recommend that approved D&M organizations be upven a certificate. This will allow Part 5 to remain applicable to a certificate holder
- The definition of a "hazard" is a condition that could foreseeably cause or contribute to an aircraft accident. Reference Order 8040.4A for the definition of an aircraft accident.
- The LAA inspector workforce will need to possess a high level of knowledge and б. understanding of SMS fundamentals to evaluate an applicant's SMS in a context of performance-based requirements. For example, evaluating initial SMS application and conducting SMS audits requires that inspectors be able to assess whether an organization's processes simultaneously satisfy multiple regulatory requirements, e.g., Part 5, Part 21 and Part 183.
- The SMS requirements as written in Part 5 represent "performance based" requirements, FAA oversight of an approval or certificate holder will be primarily focused on the existence and function of SMS processes, rather than the discrete outputs of those mocesses
- From an 5MS perspective, there are three phases to the life cycle of a product or article. that need to be addressed: design and certification, production and airworthingacertification, and continued airworthiness.

Page

15:10

SECTION 4.0 Responses and Recommendations to Taskings and Charter

1.1 Task 1: Regulatory Gap Assessment

Task 1 required the SMS WG to perform a regulatory gap analysis between the SMS requirements in Part 5, ICAO Annes 19, and Part 21

4.1.1 Task 1: SMS WG Response

The SMS WG completed a detailed gap assessment between Part 5, ICAO Anney 19, and Part 2) (see Appendix C). The assessment was conducted for three D&M life cycle phases, design and certification, production and airworthiness certification, and continued airworthiness (in-service) The SMS WG determined that Part 5 contains all of the SMS components and elements addressed by ICAO Annex 19. The objective of the gap assessment was to determine which elements of Part 5 "are covered"/"are not covered" by Part 21.

The ICAO SMS framework has four main components. Safety Policy, Safety Risk Management (SRM), Safety Assurance (SA), and Safety Promotion; which is consistent with the FAA's (proposed) Part 5 SMS regulations that have been structured around these four components as

- Subpart B Safety Policy
- Subpart C Safety Risk Management (SRM)
- Subpart D Safety Assurance (SA).
- Subpart E Safety Promotion

Part 5 also meludes Subpart A (General) which includes Applicability. General requirements and Definitions and Subpart F (SMS Documentation and recordkeeping), which, in the ICAO Framework is meluded as part of Safery Policy

A summary of the results from the gap assessment is provided or Table 1. The colors in Table 1. indicate the following:

- Green: Part 21 fully addresses the Part 5 element.
- Yellow: Part 21 partially addresses the Part 5 eloment ٠
- · Red: Part 21 does not address the Part 5 element

There are no green entries in Table 1, which indicates that Part 21 does not fully address any of the SMS elements in Part 5. This result was expected since Part 21 addresses certification procedures for products and articles and does not mention SMS, SMS Framework (i.e., Safety Policy, SRM, SA, Safety Promotion], or associated documentation and record keeping requirements

The Part 21 coverage for each Part 5 subpart is discussed in the following sections

2 PARC Working (Decaloant - Net for Distribution)

			-	7an 2011 17) 2014	
Does Part	Table 1: 21 Currently A	Gap Assessment S ddress SMS Requi	Summary rements in Defit	red Part 5?	
SUBPART A- GENERAL	DESIGN AND FRATING STION	PRODUCTION AND ADRWORTHINESS CENTOR CENTS	LONDIG TO AGINORTIONESE	COMMENTS	
s i applicability				The OL days not mention of	
5.3 Oktional Requirements	-			define any SMS requirement).	
SUBPART B - SAFRTA POLICY	DESIGN AND LERFIELATION	PRODUCTION AND ABOVORTHINESS CONTRACTION	LONTINUED MRWOICHIMASS	COSEDUATS.	
5.21 Safary Policy		a solution desited a			
5.23 Safety Accountability and Aution dy				Part 11 days and property of the	
23 Desergention and Responsibilities of Dequised Sutery Minutgement Reconstant				policy, descends cafety accountabilities or reagonale littles, or respon-	
5.27 Coordination(a) Emergency Response Planning				rementioner's acid onnie frommente	
SUBPARTIC -SAFETY RISK MASAGEMENT	DESIGN AND CERTIFIC STRON	PRODUCTION AND ENITIAL ABUWORTHINESS	IN SERVICE	COMMENTS	
f f) Applicability	Prolice	-	-	Part 21 certification process addresses the inform of SRM	
2.21 System Analysis and Historia Usenification	J'welent		-	for products Orabley system partially addresses SRM:	
5.55 Saluts Rule Associateri und Control	Weiline -			Inwarsef, Dawe so in martion of SRADiremscrywe iscars Organization Designation Authors/mins (ODA) ever (JRAO) does require inserving from the ring of products, but was SRM.	
AUBPART D - SAFRTY ASSURANCE	DEMIGN AND CERTIFICATION	PRODUCTION AND INITIAL AURWORTHINKSS	IN-SLRVICE	COMMENTS	
5.71 Salaty Performance Monitoring and Actuation				Part 21 coefficiency process and quality evolute partially addresses 8.4. Several 8.4	
5.71 Salidy Performance Assessment			1	requirements related to SMS are not addressed. ODA (see \$153.53) down constructs	
5.75 Continuous Improvement		1		wrvite motioning of product	
STRPART F - SALETY PROMOTION	DESIGN AND CENTERCATION	PRODUCTION AND INITIAL ADDWORTDINESS	15-56289364	COMMENTS.	
59) Componencies and Unioning				Part 21 does not require SMS or required terming, and commitmenties	
SUBPART F SNIS DOCUMENTATION AND READED KOLENARG	DISIGN AND CERTIFICATION	FRODUCTION AND INITIAL AIRWORDHIAFSS	IS SERVICE	COMMENTS	
5.95 SMS Documentation				Pare 21 docs and require SMS in	
5.07 SMS Records				regined accumuntations and	

Part 71 /5MS: As other Halenaking Promotion	Poge	10.01.11
SMS-Warking Group Report	Bevicino 15ato	Ungmil Ian 17, 2014

4.1.1.1 Subpart A: General

Subpart A contains three elements §5.1 (Applicability), §5.3 (General requiriments) and §5.5 (Definitions). None of these elements are addressed in Part 2) for any of the D&M lifecycle phases.

Element §5.1 specifically discusses 14 CFR Part 119 certificate holders and 14 CFR Part 121 operators. Modification of this element will be required to include Part 21 design and production organization certificate holders.

4.1.1.2 Subpart B: Safety Policy

Subpart B contains four elements (§5.2) (Safety policy), §5.23 (Safety accountability and authority), §5.25 (Designation and responsibilities of required safety management personnel), and §5.27 (Coordination of emergency response planning). None of these elements are addressed in Part 21 for any of the D&M life cycle phases.

4.1.1.3 Subpart C: SRM

Subpart ("contains three elements \$5.51 (Applicability), \$5.53 (System analysis and hazard identification), and \$5.55 (Safety risk assessment and control). These elements are discussed in the following sections for the three life cycle phases.

4.1.1.3.1 Design and Certification Life Cycle Phase

The SMS WG decided to evaluate the design and certification life cycle phase for both products and organizations. This allowed a more detailed assessment increase Part 5 and Part 21

- For products, Part 21 partially addresses SRM through compliance with the airworthinese standards (for design and certification). In general, the airworthinese standards are based on a known hazard and prescribe required design features and characteristics to mitigate the risk posed by the hazard. In addition, some aspects of SRM are addressed through §183-63 for organizations with an ODA.
- For organizations, Part 21 does not require any of the SRM elements during the design and certification life cycle phase.

4.1.1.3.2 Production and Airworthiness Certification Life Cycle Phase

Part 21 partially addresses the SRM requirements for the production and arrworthiness certification life cycle phase through the quality system. The key SMS activities during production are to build conforming products which are in a condition for safe operation. A quality system currently addresses many elements to ensure that safe and conforming parts are produced.

Some examples of how the quality system partially addresses SRM during this life cycle phase include §21.137(m) which requires that the PAH have procedures to receive and process feedback on failures, malfunctions, and detects. In addition, §21.137(n) requires procedures for identifying, analyzing, and initiating appropriate corrective action for quality escapes. These two regulations partially address the SRM requirements to identify, assess, and control the risk of potential hazards which could be introduced during this life cycle phase.
Part 21/ SMS-Aviation Balemaking Communi-	Page	11 10/ 12
SMS Working Group Report	Date	Jan 17, 2014

4.1.1.3.3 Continued Airworthiness Life Cycle Phase

Part 21 does not address SRM for the continued airworthiness life cycle phase with the potential exception of 21 137(m) which does mention in-service feedback, but does not address risk management explicitly. Given this limited coverage, the SMS WG determined that there is no coverage for this life cycle phase.

4.1.1.4 Subpart D: Safety Assurance

Subpart D contains three elements: §5.71 (Safety performance manuaring and measurement), §5.73 (Safety performance assessment), and §5.75 (Continuous improvement).

During the design and certification life cycle phase, §5 71 is partially addressed through compliance with the airworthiness standards. The remaining SA elements (i.e., §5 73, §5 75) are not addressed.

The quality system partially addresses the SA requirements for the production and antworthiness certification life cycle phase through procedures focused on producing conforming and sale products and articles

During the in-service life cycle phase, §5 71 is partially addressed through §21.3 reporting and the quality management system. The remaining SA elements (§5 73 and §5.75) are not addressed.

4.1.1.5 Subpart E: Safety Promotion

Subpart E contains two elements §5.91 (Competencies and treating) and §5.93 (Subty communication) None of these elements are addressed in Part 21 for any of the D&M life cycle phases.

4.1.1.6 Subpart F: SMS Documentation and Recordkeeping

Subpart F contains two elements §5.95 (SMS documentation) and §5.97 (SMS records). None of these elements are addressed in Part 21 for any of the D&M life cycle phases.

There are significant document and recordkeeping requirements in a quality system which address several elements of a SMS. The SMS WG rated this section red (i.e., not covered), since a quality system does not require SMS recordkeeping.

4.1.2 Task 1: SMS WG Conclusion

The SMS WG determined that Part 21 partially addresses the SRM and SA elements of Part 5 through compliance with airworthiness standards (for products) and the quality system. Elements of SRM and SA relating to the organization and the organization's processes are not specifically addressed in Part 21.

The SMS WG further identified that Part 5 is written to be applicable to a 14 CFR Part 119 certificate holder's organization and requires an update to incorporate Part 21 D&M organizations.

4.2 Task 2: Incorporation of SMS into 14 CFR

Task 2 required the SMS WG to develop, evaluate options, and make a recommendation for incorporation of SMS into 14 CFR for D&M organizations

4.2.1 Task 2: SMS WG Response

The SMS WG used a decision-making approach to identify and evaluate options for incorporation of SMS into 14 CFR. The approach involved the following basic elements.

- I Identify criteria to be used for evaluation of options;
- 2. Assign a weighting or level of importance for each criteria.
- 3 Identify available options.
- 4 Assess advantages and disadvantages of each option, and
- 5 Score each option against the weighted criteria.

4.2.1.1 Identification and Weighting of Evaluation Criteria

Criteria were weighted using a scale of 1 to 10, with 10 being the most "important" relative to the other criteria. The following criteria were identified as being appropriate for consideration.

- The option selected should be consistent with international equivalency of regulations and State obligations with regard to ICAO Annex 19, with emphasis on equivalency between States and efficiency within D&M organizations. (Weighting = 10)
- The option selected should be readily acceptable to the FAA, with emphasis on facilitation of FAA oversight of D&M organizations (Weighting = 9)
- T The option selected should allow for simple and practical application of SMS requirements across appropriate Part 21 provisions, with emphasis on ease of understanding "what has to be done" within a regulated SMS (Weighting = N)
- I The option selected should accommodate applicability to organizations holding multiple approvals, with emphasis on efficiency of regulated SMS within D&M organizationa, especially in avoiding any requirement for multiple regulated SMS within an organization. (Weighting = 7)
- 5 The option selected should facilitate scalability and flexibility of SMS, with emphasis on ability to apply regulated SMS to different size organizations (Watganing = 5)
- The option selected should facilitate consistency of application to sub-tier suppliers of D&M organizations, with emphasis on minimizing negative impact of flow-down requirements on the efficiency of a regulated SMS. (Weighting = 1)

4.2.1.2 Options

The SMS WG considered four basic options

- Option 1: Extract portions of Part 5 applicable to D&M organizations and insert where appropriate in Part 21 to establish SMS requirements for D&M without reference to Part 5.
- 2 Option 2: Establish recognition of SMS elements already existing in FAA regulations, with modification of Part 5 to allow an option to implement the "missing" elements via certification or acceptance through a voluntary SMS implementation program acceptable to the FAA
- Option 3: Modify Part 21 by addition of a new Subpart for SMS requirements, tailored for D&M organizations by copying applicable Part 5 provisions to the new subpart.
- Option 4: Adopt Part 5 m its entirety as applicable to D&M organizations, with modification to Part 21 only as necessary to incorporate reference to Part 5.

4.2.1.3 Evaluation of Options.

Option 1. To the extent that Part 5 provisions are applicable to D&M, the SMS WG considered it would be redundant to duplicate those provisions in Part 21. Duplicative requirements might create difficulty for the FAA in maintaining consistency between the two parts. A set of extracted requirements likely would result in a regulated SMS for D&M being different from SMS for other kinds of organizations. Such differences might make it difficult to achieve equivalency of regulation between the US and other States, and efficient oversight by the FAA. With separated and different SMS provisions, this option would not avoid potential for an organization holding multiple certificates to be required to have multiple SMS, and would not contribute to effective scalability and flexibility of SMS. This option was evaluated as not achieving any of the criteria, and was scored as "zero."

Therefore the SMS WO removed Option 1 from consideration

- Option 2: Creation of a certification process to cover SMS clements "missing" from existing regulations, including any that exist outside of Part 21: would present considerable burden for FAA in its oversight responsibilities. Use of a voluntary standard would require that the FAA evaluate what likely would be multiple competing commercial SMS programs to determine acceptability, and would require FAA oversight of such multiple programs. It is unclear whether or how voluntary SMS implementation programs might contribute to equivalency of international regulations. This option was evaluated as not achieving any of the criteria, and was scored as "zero." Therefore, the SMS WG removed Option 2 from consideration
- 3 Option 3: In this approach, requirements for applicability to Part 21 approval holders would be established in the new subpart. A new subpart would provide a vehicle for ready tailoring of SMS requirements for D&M organization, and thereby also readily accomplish appropriately scaled and flexible application of requirements. Such alignment of SMS requirements to Part 21 would facilitate straightforward implementation and compliance by D&M organizations, and efficient oversight by the FAA. With tailored SMS requirements concentrated in a new subpart, there would be low risk of any Part 21 SMS changes imadvertently inducing SMS conflict in another industry segment or

Farr 21 / 540s- Avention Rubmalcing Commune	Page	11 of 15
SM5-Working Group Report	Revision Date:	Original Jan 17, 2014

requiring change to Part 5. This approach would provide for a clear definition of "what has to be dono" within Part 21 for the various certificate types.

A new Subpart for Part 21 would be advantageous only if Part 5 were determined nor to be appropriate for D&M organizations. As a result of the gap analysis discussed in section 4.1, the WG determined that Part 5 is appropriate for D&M organization, with modification of §5.1 (*Applicability*), and §5.5 (*Definitions*). Consequently, addition of a new subpart to Part 21 would create redundant SMS requirements having potential for the difficulties noted for Option 1.

Therefore, the SMS WG removed Option 3 from consideration

4 Option 4: The WG determined that Part 5 is appropriate for D&M organizations, with modification of §5.1 (*Applicability*) and §5.5 (*Definitions*). When accompanied by policy and guidance material (recommended by the SMS WG elsewhere in this report to be developed) describing SMS application for D&M organizations, use of Part 5 would minimize need for revision. Part 21 would need revision only to the extent necessary to refer to Part 5 for SMS requirements. The SMS WG considers that adoption of Part 5 is likely to best promote straightforward implementation by affected organizations and oversight by the FAA. Part 5 contains all of the SMS components and elements addressed by ICAO Annex 19, thereby maximizing US equivalency of regulations. Option 4 would require montain and compliance by D&M organizations, and efficient and standardized oversight by the FAA.

Theraforal the SMS WG selected Option 4 as the preferred option

4.2.1.4 Discussion of Selected Option

The SMS WG response for Tasks 3 and 5 provides a recommendation for applicability and scalability. Further, the SMS WG anticipates as noted in section 3.0 that the Organizational WG will define applicability requirements for D&M organizations, and will recommend that such approved organization be given a certificate. If those recommendations are adopted, Part 5 references to "the certificate holder" would remain valid for affected D&M organizations.

Notwithstanding the incorporation considerations discussed above, the SMS WG considers that §5.1 should be clarified with regard to the specified deadlines. Per §5.1(a), a certificate holder "must have" a SMS "that meets the requirements of this part and is acceptable to the Administrator by [3 years after the effective date of final rule]," and §5.1(b) specifies a certificate holder "must submit" an implementation plan "for approval" no later than six months after the effective date of the final rule.

A certificate holder must be able to understand the process of determining acceptability, track the progress of the submission, and respond to comments and questions in order to "have" an acceptable SMS by the deadline. The SMS WG recommends that the FAA clarify the activities and timing expected between submission of an implementation plan by a certificate holder and a determination that the certificate holder's SMS is "acceptable." Examples of needed clarification include a definition of "have an acceptable SMS." explanation of the relationship between "approval" of the submitted implementation plan and "acceptance" of the holder's SMS. Additionally, the specified dates should account for an application made after the effective date of the final rule.

Port 71 /SMS Availan Balanakara Commune	Page	15 of 10
TATLAL STISSAN AND BUILDING COMMANY	Revision	Ongmil
2 MS ACOLETON PLOUD REDULT	Date	- Jan 17, 2014

To facilitate appropriate management of the approval process, the SMS WG recommends that the specified deadlines be referenced to approvals rather than to the effective date of the final rule. For example, once the implementation plan is submitted for approval, the next logical step would be approval of the plan. At that point, the certificate holder would have a basis for proceeding with implementation, eventually arriving at a point of "having" an acceptable SMS Under such a concept, only the deadline for plan submitted would be appropriate to be field to the effective date of the final rule.

A certificate holder must be able to understand the process of determining acceptability, track the progress of the submission, and respond to comments and questions in order to "have" an acceptable SMS by the deadline. Clarification of the respective deadlines may require development of specific policy and guidance material.

Section 3.0, Key Assumptions, includes an assumption that the definition of "hazard" is a condition that could foreseeably cause or contribute to an aircraft accident. The SMS WG recommends that Part 5, §5.5 Definitions, be changed to incorporate that definition. The change is necessary in order to appropriately focus attention during SMS implementation by an organization and during oversight by the FAA. Conditions that, for example, affect occupational health and safety, environmental protection, and security are not appropriate to be included in hazard identification requirements for a regulated aviation SMS for D&M.

The following changes would be required for Part 5

- §5.1 (Applicability): Change §5.1(a) to include design and/or manufacturing organization approvals under Part 21, or make reference to "certificate holders."
- §5.1 (Applicability): Clarify the activities and timing expected between submission of an implementation plan by a certificate holder and a determination that the certificate holder's SMS is "acceptable."
- §5.1 (Applicability): Change 5.1(a) to refer to a period of time following approval of an implementation plan, rather than the effective date of the final rule, as the deadline to "have" an SMS.
- §5.5 (Definitions): Change the definition of "hazard" to refer to a condition that could foreseeably cause or contribute to an aircraft accident.

4.2.2 Task 2: §5.27 Coordination of Emergency Response Planning

4.2.2.1 Definition and Context of "Emergency"

According to The American Heritage Dictionary of the English Language, Fourte Edition, the definition of "emergency" is, "A serious situation or occurrence that happens unexpectedly and demands immediate action."

In the Supplementary Information of Notice of Proposed Rulemaking, No. 10-15 (Docket No FAA-2009-0671), the FAA describes emergency response planning as, "... provides the basis for a systematic approach to managing the organization's operations in the aftermath of a significant unplanned event or during an ongoing emergency situation. The overall objective is the safe continuation of operations and the return to normal operations as soon as possible."

As noted in section 3.0 of this report, the SMS WG anticipates that a regulated SMS will address mazards that could foreseeably cause or contribute to an arcraft accident. Therefore,

Part 21/5405-Accation Rulemaking Communi-	Page Bevision Date	Lo of 17
SMS Working Group Report	(Jato)	Jan 17, 2014

"coordination of emergency response planning" refers to planning for activities that take place within a limited period of time during which an aircraft operational amergency situation exists. It includes the period of time required to re-establish "normal" operations following the emergency

4.2.2.2 Discussion of Emergency Response Planning

The SMS WG considers that situations or events (e.g., natural disaster tostile attack of facilities five) that result to interruption of the business activities of a design and/or manufacturing organization are distantly removed from contribution to an aircraft accident. This is true as well for situations or events that might affect an organization's ability to provide continuing airworthiness support for a product or article. Therefore, such business interruptions do not constitute an "emergency" for which "emergency response plaining" is appropriate to the context of a regulated SMS.

Generally, aircraft operational emergency situations may be relevant to design and/ormanufacturing organizations in three ways.

- The product(s) or article(s) of the organization are installed, maintained, and/or operated by another entity.
- The organization itself conducts aviation operations, or
- 3 There is a need to interface with another organization during that organization a emergency response activities

4.2.2.2.1 Products or Articles Installed, Maintained, and/or Operated by Other Entities

With regard to the operations of other entities, involvement of D&M organizations is primarily limited to airworthiness activities following an emergency situation (e.g., a design organization may need to consider whether accident investigation findings indicate a need for airworthiness action). Those activities are part of existing regulatory responsibilities of [e.g., a Type Certifican (TC) holder complying with §21.99, and are normal activities for such an organization]. Therefore, those activities, even if they occur after an aircraft accident, do not constitute "emergency response" on the part of D&M organizations.

4.2.2.2.2 Flight Operations by a D&M Organization

A variety of aviation operations may be conducted by a design and/or manufacturing organization. A design and/or manufacturing organization may conduct any, all, or none of the following kinds of operations.

- Corporate transportation.
- Sales and demonstration.
- + Ferry flights
- · Air shows and displays
- Arial photography.
- Engineering flight test (including ground tests).
- · Production flight test

Part 21/SMS-Aviation Bulemalong Communes	Page	17 of 18
SMS Working Group Report	15ata	Im 17 2011

Typically, such operations are conducted in accordance with 14 CFR Part 91. The SMS WG understands that Part 91 operations are not to be subject to SMS requirements. Corporate transportation flights, sales demonstration flights, ferry flights, air shows, and aerial photography flights are typical of flying operations that might be undertaken by virtually any Part 91 operator. There is therefore no basis for any requirement that a design or manufacturing organization prepare emergency response planning for such operations. The WG recognizes that angineering and production flight use operations conducted for the purpose of Part 21 activities encompass a broad range of hazards that are unique to such operations. Nevertheless, such operations are not appropriate to be subjected to §5.27 for D&M organizations for the following reasons.

- Production flight test operations normally occur within the conlines of a quality
 management system. The products and articles under test normally are subjected to
 mumerous inspections during assembly, and before and after flight. Thus, these operation
 already are conducted in a manner that significantly exceeds safety requirements for
 Part 91 operations. Additional requirements for coordination of emergency planning are
 not warranted.
- Engineering flight test operations conducted with the participation of FAA personnel or designees generally are required to be subjected to risk management as specified in FAA Order 4040.26B. Aircraft Certification Service Flight Test Risk Management Program Anong the specified requirements are project phonong, descriptions of emergency procedures to be accomplished, reporting of significant events, and an accident response plan. Many, if not most, organizations that conduct engineering flight test operations apply the risk management process(es) equally to their operations that do not involve FAA personnel. Thus, these operations already typically are conducted in a manner that significantly exceeds safety requirements for Part 91 operations. Additional requirements for coordination of emergency planing are not warranted.
- Any arrworthiness issues that might arise during production or engineering flight test, even an issue that might affect in-service aircraft, are dealt with using the same process(es) used to deal with airworthiness issues otherwise discovered during the normal course of business. Therefore, the exercise of those processes does not constitute "emergency response" for which coordination of planning would be required.

4.2.2.2.3 Coordination with Other Organizations

The coordination of emergency response planning, as noted in §5.27(c) applies to coordination of a "certificate holder's emergency response plans" with the plans of "other organizations it must interface with during the provision of its services." For the reasons described above, there should be no requirement for a D&M organization to develop an emergency response plan as part of its regulated SMS. Thus, any requirement to coordinate such a plan with another organization is most. There may concervably be situations where an organization that does engage in emergency response activities might benefit from information provided by a D&M organization. Examples might include an Airport Rescue Fite Fighting unit for accident first response netivities, or an accident investigation authority for investigation activities. The SMS WC considers that provision of such information is not within the meaning of an "emergency response plan" that a D&M organization must coordinate. Further, provision of such information does not constitute "interface with" another organization that "must" be accomplished during the provision of a design and/or manufacturing organization first services.

Page

15:00

4.2.3 Task 2: SMS WG Recommendation

The WG concludes that §5 27 requirements should not apply to design and/or manufacturing organizations. Consequently, §5.2)(a)(6) also should not apply to design and/or manufacturing. organizations.

The SMS WG meanmends:

- 1 That Part 5 be made applicable to certain D&M organizations.
 - §51 (Applicability) Change §51(a) to include design and/or manufacturing. organization approvals under Part 21, or make reference to "certificate holders"
 - §5.1 (Applicability). Clarify the activities and timing expected between submission of in implementation plan by a certificate holder and a determination that the certificate limbler's SMS is "acceptable."
 - 45.1 (Applicability). Change 5.1(a) to refer to a period of time following approval of an implementation plan, rather than the effective date of the final rule, as the deadline to "liave" an SMS
 - 55.5 (Definitions) Change the definition of "hazard" to refer to a condition that could foreseeably cause or contribute to an aircraft accident.
- 2 That Part 21 be modified to refer to Part 5 as SMS requirements for D&M organizations meeting the applicability threshold. The specific changes to Part 21 will depend upon the manner in which the Organizational WG recommendations are implemented.
- 18 That §5.27 and §5.21(a)(6) should not apply to D&M organizations:

4.3 Tasks 3 and 5: NMN Applicability, Scalability, and Flexibility

Task 3 required the SMS WG to provide a recommendation as to which D&M approval holder organizations should have SMS requirements applied to them. Task 5 is related to Task 3 and required the SMS WG to provide guidance that accounts for scalability and flexibility of SMS to D&M organizations based on their size, complexity, and safety risk introduced by their products. The SMS WG decided to combine Tasks 3 and 5 into a single response.

4.3.1 Tasks 3 and 5: SMS WG Response

4.3.1.1 Applicability

ICAO Annex 19 currently requires SMS for aircraft TC and Production Certificate (PC) holders and is planning to expand the requirement to engine and propeller TC and PC holders in the future. The SMS WG agrees with the future ICAO requirement and concluded that SMS should be applied to all TC and PC holders for aircraft, engines, and propellers

The next step in the SMS applicability decision-making process addressed Parts Manufacture Approval (PMA), Technical Standard Order Authorization (TSOA) and STC holders. The SMS WG evaluated several criteria, but determined that safety risk introduced by the holder's products/articles was the key criteria that should be used to determine whether SMS should be applied to a PMA, TSOA, or STC holder

The WG noted past FAA precedence with the current requirement under Order 8110.42 Chapter 5 paragraph (d) that requires a PMA applicant to conduct a safety assessment to

Burn 21 / SMS, Astudian Balamakian Communian	Page	19.07.20
PATE 217 SAIS SORTING FOR THE AND THE	Revision	Original
2012 ACCREDING OLDER REDUCT	12:010	Jan 17, 2014

establish if a part is critical or non-critical. Consideration was also given to the Aircraft Conffication Service Category Parts List (Revision G dated 8/18/09)¹ which categorizes parts from Category 1 to III, with 1 being most critical. This was overlaid with the FAA's Certificate Management Information System (CM18)² which defines criticality from Level 1 to 5, with 5 being most critical. Using these criterion, the WG rated Category 17 Level 5 parts as having the largest importance for the SMS applicability evaluation. A description of these levels is provided in Table 2 which has been excerpted from the FAA 2012 Manufacturers Safety Management System Pilot Project Report for Design and Manufacturing Organizations

Therefore, from an aviation safety perspective the SMS WG determined that SMS should be applied to PMA, TSOA, and STC parts, products and articles that are considered critical. Critical is defined as *parts, products and articles thereof whose failure could directly prevent continued safe flight and landing, which corresponds to Category 1 or Level 5*

E DECAT	LEVEL 1	LEVEL J	LEVET 4	LEVELS
A predict of south) thereof where, failing would have LIFTLE TO (40 choice on ensummed cuterfloor and have/mesol the atoend)	A product or part(s) (hereof whose failure would not prevent continued safe flight and fanding, resulting consequences COULD reduce the capability of the nineartt or the ability of the crew to cope with adverse spending conditions or subsequent failures.	A product or part(a) thereof whose future would not prevent continued safe flight and landing, resulting consequences WOULD reduce the capability of the aircraft or the ability of the crew to cope with adverse operating conditions or subsequent failures	A product or part(s) thereof whose Galace COULD. IF OTHER CONDITIONS EXISTED prevent continued safe flight and harding: resulting consequence) could reduce sifety marging, degenda performance, for exists loss of capability to conduct certain flight operations	A printing (or particle) thereof whose fishing COULD DIRECTLY provem continued safe (high) and functing: resulting consequences could reduce safety mayours, deputid performance, or consectors of capacities to sending consectors of capacities to sending consectors.
CRETATION		CPL CAT III		CPL CAT1

Table 2: Criticality of Products Produced by Production Approval Holders

4.3.1.2 Scalability and Flexibility

Scalability: The SMS WG determined that SMS needs to be applied in full (with the exception of §5.27) to all D&M organizations meeting the applicability threshold. This applicability threshold also addresses scalability in that the SMS WG does not see benefit in applying scaled. (i.e., partial) SMS requirements. Considering the level of activities already associated with certifying a critical PMA part or an equivalent Category 1 or Level 5 part or article under a TSOA or STC, the SMS WG does not believe scalability of SMS will be a concern for affected D&M organizations. The SMS WG feels that an organization that is capable of designing.

AIRCRAFT DERTIFICATION SERVICE GATEGORY PARTS LIST can be found on the FAA website and was previously part of the now canceled Order 8120.2 and is still referred to in AC 43-18 "Fabrication of Aircraft Parts by Miwritenance Personne" and AC 21-43 "Production Under 14 CFR Part 21 Subparts F. G. K. and O" CMIS is the system used by FAA manufacturing inspection offices to prioritize oversight activities.

Part 21 / SMS-Aviation Bulenaking Lominum-	Page	20 n/ 21
SM5 Working Group Report	Data	Jan 17, 2014

certifying and performing continued airworthiness of a critical part or article should be capable of unplementing an SMS similar to a TC or PC holder.

I lexibility: The SMS WG recognizes that SMS application will need to be flexible based on the size and complexity of the D&M organization. For example, in a small organization, the accountable executive and SMS management representative may be the same person. The SMS WG plans to take this into account when developing SMS guidance material as part of our Task 4 recommendation based on the ARC's recommendation for applicability of SMS.

4.3.1.3 Additional Discussion Points

1.4A Oversight Resource Demand: A consideration in the application of SMS to STC, TSOA and PMA companies will be the resource demand to provide oversight by the FAA. The Part 21 SMS ARC Oversight and CBA WGs will need to evaluate if applying SMS to STC, TSOA or PMA is the most efficient application of FAA resources to address safety, meaning would the FAA resources have a greater impact to safety if the FAA resources were applied to other areas than SMS oversight of the STC, TSOA or PMA companies.

STC Holders: STC holders do not corrently have an organizational requirement or a requirement for a quality management system. Therefore, a greater impact could be oremred by some STC companies to implement SMS. Similar to companies that own TCs and have continued airworthiness responsibilities, but may not produce the type certificated product, some STC companies are only design and engineering and do not manufacture or sell parts. The impact to these companies and the connection with the manufacturer and installer of the STC along with the FAA oversight will need to be evaluated by the Oversight and CBA WGs.

Foluntary Implementation of SMS: The SMS WG identified potential value through voluntary SMS implementation for parts or articles below the critical threshold. Additional privileges would need to be realized for companies that voluntarily implemented SMS. A potential option would be to have an industry SMS program that would accredit the PMA, STC, and TSOA company's SMS. Future consideration will be required to develop an industry accreditation program. Additional privileges could be avoidance of sequencing and more expeditious project reviews. An example of an industry accreditation program would be the quality management system Aerospace Standard (AS) AS9100 developed by International Aerospace Quality Oroup (IAQG) released by the SAE International and the Aerospace and Defense Industries Association of Europe – Standardization (ASD-STAN).

4.3.2 Tasks 3 and 5: SMS WG Recommendation

Applicability; The SMS WG recommends that FAA develop an applicability threshold that requires SMS for organizations that.

- · Design or manufacture products () e., aircraft, engines, propellers).
- Design or manufacture articles whose failure could directly prevent continued safe flight and landing; or
- Make design changes to a product, through a Supplemental Type Certificate (STC), failure of which could directly prevent continued safe flight and landing.

This recommendation is not intended to discourage voluntary implementation of SMS for organizations producing articles with criticality falling below the applicability threshold

Part 21 / 5415- Actation Rulemaking Commune-	Page Beyiston	21 n/ 22
SMS-Working Group Report	15ate	Jan 17, 2014

Scalability Scalability of SMS has been addressed through the applicability threshold. Therefore the SMS WG recommends that all components / elements (with the exception a)' §5.27) of SMS be applied for those design and /or manufacturing organizations that meet the applicability threshold.

4.4 Task 4: Recommended Preamble, Policy, and Guidance Material

Task 4 required the SMS WG to produce recommended preamble, policy, and guidance material to support SMS application to D&M organizations. This effort also included development of an "operational" definition of a hazard throughout the life cycle of a product in SRM.

4.4.1 Task 4: 5MS WG Response

The SMS WG developed SMS regulatory material and a basis for preamble, policy, and guidance material as provided in this report. However, we determined that more work was necessary to produce detailed guidance material. The SMS WG believes it is important that this guidance material be developed before FAA tulemaking begins, and that the surrent SMS WG be involved in the development. It also is noted that the guidance material will be dependent on the outputs from, and be used by, the Organizational and Oversight WGs

Specific guidance that requires further development includes:

- How safety objectives are established
- · Evaluating the performance of an organization's SMS
- Development of an "operational" definition of a bazard throughout the life cycle of a product
- Acceptable criteria for the effectiveness of the safety risk controls at a system level.
- Process that the D&M organization implements to meet §5.55(b) and the extent to which the LAA is engaged
- · Flexibility in applying SMS to the D&M organization

4.4.1.1 Safety Performance Assessment and Effectiveness

Part 5 (§5 73) requires that the D&M organization evaluate the performance of its SMS and the effectiveness of its SRM safety risk controls. With regards to safety risk controls, there will typically be hundreds of risk controls for a product and it is unnecessary and an ineffective use of safety resources to evaluate the effectiveness of each individual safety risk control. Articles and STCs can also have a significant number of safety risk controls. A more effective approach is to evaluate the product or article at a system level, for example, engine inflight shutdown rate provides a system level assessment of an engine's reliability and the risk of loss of critical thrust on an ETOPS flight. Appropriate, acceptable system level evaluation criteria are needed to effectively and efficiently evaluate the safety risk controls of a product or article.

Additionally, guidance is needed regarding an organization's safety objective and how its SMS performance is evaluated against those safety objectives. Section 5.1 defines a safety objective a a measurable goal or desirable outcome related to aviation safety. While zero accidents is a desirable safety objective it does not provide a useful timely enterna for evaluating the performance of an organization's SMS. Additionally, the goal of SMS is to improve safety by

Part 21 /5415-Actation Rulemaking Commune-	Page	22 n/ 25
SMS Working Group Report	13-ma	Jan 17, 2014

managing the with the understanding that risk is always present and cannot be completely climinated. Consumos on criteria and guidance for determining safety objectives and SMS performance against those safety objectives will also facilitate efficient process oversight of the D&M SMS.

4.4.2 Task 4: SMS WG Recommendation

The SMS WG recommends that we continue our efforts in 2014 to develop SMS policy and guidance material and provide as an addendium to this report.

4.5 Task 6 and 7: CONOPS for SMS

Task to required the SMS WG to define the end state (roles, relationships, and responsibilities) for SMS in a D& M organization considering design and certification_production and airworthness certification, and continued airworthness. Task 7 is related to Task 6 and requires the SMS WG to perform a review of "strategic" safety decisions that are required to be made utring the design and certification, production and airworthness certification, and continued airworthness that are required to be made utring the design and certification, production and airworthness certification, and continued airworthness that are required to be made utring the design and certification, production and airworthness certification, and continued airworthness life cycles of the product. The SMS WG decided to combine Tasks 6 and 7 into a single response.

4.5.1 CONOPS Summary

The SMS WG has developed a CONOPS which describes the intent of the SMS framework (i.e. Safety Policy, Safety Risk Management, Safety Assurance, Safety Promotion) for D&M organizations as it applies to each life cycle phase (design and certification, production and airworthiness certification, and continued airworthiness) of a product or article. The CONOPS locuses primarily on SMS framework components SRM and SA. Part 5 Subparts C and D respectively. Safety Policy and Safety Promotion, Part 5 Subparts B and E together are intended to foster and promote a healthy "Safety Culture," which is a necessary enabler of SMS. Subparts B and E, do not exist in current regulations for D&M organizations and will need to be implemented to provide a regulatory basis for enabling enhancement to a D&M's safety culture.

The CONOPS defines how the existing applicable certification procedures or airworthiness regulations (e.g., 14 CFR §21 17, §21 101, §21 137, §33 75, §25 (309) and their associated policy and guidance (FAA Orders and ACs) satisfy the intent of Part 5 Subparts C and D. The CONOPS also identifies where the existing regulations do not fully meet the intent of Part 5, and what the additional intent of the Part 5 regulation is. Further, the CONOPS discusses how the organizational regulations required for Organizational Delegated Authority (ODA) under Part 183 Subpart D satisfy the intent of SMS framework components SRM and SA, Part 5 Subparts C and D, except that Part 183 Subpart D defines delegated responsibilities and Part 5 defines responsibilities of a certificated organization.

The procedural and airworthiness regulations that are complied with during the design and certification phase satisfy the intent of Part 5 Subparts C and D SRM and SA except that consideration of safety lessons learned from existing certified products need to be considered. During the continued airworthiness phase of a product/article life cycle, the existing Part 21 regulations do not explicitly satisfy the intent of SRM and SA. However, Continued Operational

See ICAO Safety Management Manual Doc. 9859 – 1 Entrion (section 1.5) for more discussion and domition of Safety Culture

Part 21/5MS-Astation Balenaking Communis	Page	27 n/ 21
SMS-Working Group Report	Date	Jan 17, 2014

Safety (COS) programs and §183-63 satisfies the intent, although §183-63 includes requirements to notify the FAA whereas Part 5 defines responsibilities of the certificated organization

During the production and airworthiness phase, the quality management system requirements suisfy much of the intent of Part 5 Subparts C and D SRM and SA. The primary function of a certificated production organization SRM and SA is to support COS programs by providing information regarding in-service failures, malfunctions, and defects.

We recommend that the CONOPS form the basis for the development of preamble, policy, and guidance material

4.5.2 Tash 6 and 7: SMS WG Response

SMS for D&M is an organizational requirement intended to enhance safety of products or articles with the ultimate goal of enhancing aviation safety. SMS is not intended to corcumvent compliance to the applicable certification procedures or an worthiness regulations (14 CFR). Nor is SMS intended to revisit obligations or compliance with the existing applicable certification procedures or airworthiness regulations (e.g., 14 CFR §21.17, §21.101, §25.1309), or their associated FAA Orders and ACs

The life cycle of a product or part can be logically divided into three phases.

- 1 Design and certification, and
- 2 Production and meworthmess certification; and
- 3 Commied airworthingss

Current certification procedures and airworthiness regulations are divided up in accordance with these phases, except that continued airworthiness, in the current regulatory structure, is predominately an FAA function. However, with the introduction of Part 5 which contains the proposed SMS regulations applied to certificate holders, continued airworthiness also becomes an obligation of a DAH. Currently, most, if not all, DAHs have implemented voluntary COS programs for their products or parts or, if they hold an ODA, are conducting COS as part of their obligation in complying with §183 o3 (° ODA – Continuing requirements: Products, parts or appliances?").

4.5.2.1 Part 5: Four Main Elements of SMS

The ICAO SMS framework has four main components. Safety Policy, Safety Risk Management (SRM), Safety Assurance (SA), and Safety Promotion, which is consistent with the PAA's Part 7 SMS regulations that have been structured around these four components as

- + Subpart B Safety Policy
- Subpart C Safety Risk Management (SRM)
- Subpart D Safety Assurance (SA)
- Subpart E Safety Promotion

Part 5 also includes Subpart A (General) which includes Applicability, General requirements and Definitions and Subpart F (SMS Documentation and record keeping), which in the ICAO Framework is included as part of Safety Policy.

3	Jorr 21/5405-Auntion Rulemaking Communics	Page	24 pf 25
	SMS Working Group Report	13:00	Jan 17, 2014

NOTE: In the remainder of this document, the references to Subpart B (L) D, and E are considered as described in the NPRM for 14 CFR Parts 5 and 119 titled "Safety Management Systems for Cartificate Holders Operating" under 14 CFR Part [2] (Docket No. FAA-2000-0671) excepted as discussed in section 3.0.

Subpart B establishes the foundation for safety management through senior management's commitment to continually improve safety and defining the methods, processes, and organizational structure needed to meet safety goals. Subparts B and E together are intended to foster and promote a healthy Safety Culture, ⁴ which is a necessary enabler of SMS. Again, while most, if not all, DAHs have an inherent positive safety culture, the proposed requirements defined in Subparts B and E. do not exist in current regulations for DAHs and will need to be implemented to provide a regulatory basis for enhancing DAHs safety culture.

The major objective of this CONOPS is to address the intent of Subparts C (SRM) and D (SA) for the life cycle of a product or article, from an aviation safety perspective. Figure 1 through Figure 6 (contained herein), provide a graphical flowohau representation of Subparts C and D, and the intent of these Subparts as they relate to the three phases of a product or article. As in discussed in the following sections, much of the intent of these Subparts is already addressed under existing regulations or by current practices to comply with the existing regulations. The discussion also identifies existing regulations as well as the additional avivities that folly meet the intent of these Subparts for a DAH and/or PAH.

4.5.2.2 Subpart C: Safety Risk Management

Subpart C is comprised of three sections:

- + §5.51 Applicability.
- · §5.53: System analysis and hazard identification, and
- · §5.55 Safety risk assessment and control

Specifically these sections require.

- Identification of hazards or ineffective tisk controls.
- · Assessment of safety lisk, and
- · Development of safety risk controls.

4.5.2.2.1 Design and Type Certification

One aspect of Subpart C (§5.51) requires a safety tisk determination. If a new system (a implemented or a revision is made to an existing system (§5.51(a) and (b)). For DAHs designing and certifying a new or changed product or article, the intent of Subpart C [i.e., §5.51(a) - *Implementation of new systems*, and §5.51(b) - *Revision of existing systems*) is satisfied by the current Part 21 certification procedures (i.e., §21.17, §21.19, or §21.101), which require that the appropriate airworthiness regulations are complied with 16 appropriate airworthiness regulations do not exist because the proposed design is novel or unusual, then the FAA will issue special

¹See ICAO Safety Management Manual Doc. 9859 – 3^m Entries (section 1.6) for more discussion and domition of Safety Culture

Port 21 / SMS-As ention Rulemaking Longmonos	Page	25 nf 20
SMS-Working Group Report	Beyedine Date:	Drigmal Jan 17, 2014

conditions to ensure that a level of safety equivalent to that defined in the regulations of combinisted (reference §21-16).

Airworthiness regulations are intended to define the conditions for risk controls to be developed based on known hazards. These hazards are systemic in nature and have often been defined as the result of accident/incident investigations. The intent of SRM has therefore already been performed and its ourcome established in airworthiness requirements and fulfilled in certification processes. Hazards unique to a certain design are typically the topic of special conditions. As in airworthiness regulations, the process for developing special conditions meets the intent of SRM and establishes the means to manage risk to acceptable levels.

Current §21.33, §21.37, §21.39, and §21.35 require that all the necessary inspections and tests are conducted, and §21.50 ensures that the Instructions for Continued Airworthiness (ICA) are available. Additionally, even if the design meets all the airworthiness requirements and special conditions, it will not be certified if it has any feature or characteristic that makes it unsafe (reference §21.21(b)(2)).

The issuance of these regulations is evidence of how the certification procedures and nitworthiness regulations have evolved over the years, and have been promulgated based on identified hazards and subsequent risk initigations, are in essence already addressing SRM (CFR Part 5 Subpart C). Some of these rules have addressed a very specific hazard (e.g., §25.607) for fastener retention), while others are more general and allow innovation of design, provided that the hazards associated with the design are identified and appropriately initigated (e.g., §23.1109 for system design). Additional examples are included in Appendix D.

Under Subpart C. §5.53 and §5.55 require a SRM process to perform the activities to comply with §5.51, that is, if a new system is implemented or a revision is made to an existing system. The D&M processes used to comply with Part 21 design certification procedural regulations and the airworthiness regulations (e.g., Parts 23, 25, 27, 26, 29, and 33) can meet the intent of Subpart C. §5.53 and §5.55 SRM for products or articles.

For an ODA holder, the processes required to satisfy Part 183 Subpart D (*Urganization Designation Authorization*) (e.g., §183-53), considering compliance with Part 24 design certification procedural regulations and the airworthiness regulations as described in the preceding paragraph. further supports meeting the intent of §5-53 and §5-55

For the ARC's proposed Design Organization (DO), it is expected that one of the obligations urder Part 21 will be to comply with the applicable sections of Part 5, which will include compliance with §5.51, §5.53, and §5.55.

Sections §5.51, §5.53, and §5.55 require SRM for the "system," which includes not only the product or article, but the organization's operating environment, supporting processes and procedures, and personnel, equipment, and facilities (organizational/operational) required for the design or production of the product or article. In the design and certification life cycle phase, the organizational /operational hazard is the condition(s) that could foreseeably lead to a noncompliant design. For organizational / operational changes: planned or unplanned, business practices need to ensure that the changes will not lead to a non-compliant design. The intent of §5.51(c) for a DAH is addressed by 14 CFR Part 183. Subpart D (e.g., §183.53 for current ODA holders), and by the expected organizational regulations for the proposed certificated Part 21 DOs. The intent of §5.51, §5.53, and §5.55 for these organizations is depicted in Figure 6 and is meant to ensure that the

Part 21/5915-Actation Bulgmaking Communi-	Page	20.6(27
SMS Working Group Report	(5ate)	Jan 17, 2014

organization not only produces a compliant and/or conforming product or article, but manages organizational processes and changes to these processes ensure it continues to produce a compliant and/or conforming product or article.

The intent of §5.51(d) for the design and type certification phase of the life cycle of a product or article is discussed in conjunction with the intent of Subpart D. Safety Assurance (SA)

4.5.2.2.2 Continued Airworthiness

Once a product or article has been designed, certified or approved and is in-aervice the continued airworthiness phase of the aviation safety life cycle begins. Most, if not all DAHs have some form of COS program which monitors reported in-service events related to the product or article certificates or approvals they hold, assesses the safety risk of those events and if necessary defines safety risk controls such that an acceptable level of risk is maintained in the fleet (reference Figure 3 and Figure 5). Monitoring of products or articles is addressed in the Subpart D. Safety Assurance, section 4.5.2.3.

In most, if not all cases, the DAH COS program satisfies the certificate or approval holder's obligations under §21.3 or the ODA holder's responsibilities under §183.63. However, under the current §21.3 regulation the DAH is only required to report, and only for thirteen specific occurrences. From a regulatory perspective §21.3 does not fully address §5.51(d), §5.53 or §5.55 identification of hazards or ineffective risk controls, safety risk assessment, and control; however, in most, if not all cases, the DAH COS program does contain processes and procedures to perform identification of hazards or ineffective risk controls, safety risk assessments, and control as depicted in Figure 3 and Figure 5. Once the COS program identifies a need for a safety risk control the same process for §5.51(a) through (c) described above is used to implement any design or organizational safety risk controls necessary, and if the FAA determines an unsafe condition exists and will likely exist, or develop in the fleet they will issue an Airworthiness Directive (AD).

For an ODA holder, the intent of §5.51(d), §5.53, and §5.55 to have SRM processes to identify bazards or ineffective risk controls, make safety risk assessments and identify safety risk controls is satisfied by the obligations of 14 CFR Part 183, Subpart D (*Commung requirements*, *Products, parts or appliances*) §183.63(c). Part 183.63(d) meets the intent of §5.51(a) through (c) If a COS program is used to satisfy §183.63(c) and (d) it also satisfies the intent of §5.51(b), (c), and (d), §5.53 and §5.55.

For the ARC's proposed DO, it is expected that one of the obligations inder Part 21 will be to comply with the applicable sections of Part 5 which will include compliance with 55.51 §5.53, and §5.55 A COS program can meet the intent of §5.51, §5.53, and §5.55 provided it identifies hagards or ineffective risk controls, makes safety risk assessments and identifies safety risk controls when unacceptable risk is identified by the safety risk assessment.

As discussed in section 4.5.2.2.2, under Design and Type Certification, vections §5.51, §5.53, and §5.55 require SRM for the system which includes the organizational/operational bazards as well as the product or article bazards. In the continued airworthiness life cycle phase, the organizational / operational bazards is the condition(s) that could foreseeably lead to an uncontrolled unacceptable risk. For organizational / operational changes, planned or unplanned, business practices need to assure that the changes will not lead to an uncontrolled unacceptable risk. For a DAH, (either ODA or DO) the intent of §5.51, §5.53, and §5.55 for organizational/operational bazards can be met by assuring that the COS program continues to

Part 21.75315-Acation Bulenaking Communi-	Page	27 0/ 28
SMS Working Group Report	Date	Jan 17, 2014

meet its intended function and that changes to the COS program are assessed to assure the system will continue to function properly. The intent of §5.51(c) for a DAH can be addressed by Part 183, Subpart D, §183.53(c)(14), for current ODA folders, and by the DO regulations for the proposed certificated Part 21 DOs. The intent of §5.51, §5.53, and §5.55 for COS programs is depicted in Figure 6.

4.5.2.2.3 Production and Airworthiness Certification

PAH organizations under Part 21 Subparts G, K, and O are authorized to produce products or articles. STCs are not addressed here because they are only a design approval and production of an STC needs to be performed under a separate authorization (e.g., a repart station authorized under Part (45). Because PAHs operating under Parts 21 Subparts G, K, or O and/or Part 183 Subpart (D) don't necessarily have design engineering expertise to conduct aviation safety risk assessments, their function for Part 5 Subpart (C) would be to support the organization's COS program by providing information on §21.3(b) and §21 (37(m) and (n) quality excapes, inaervice failures, multiunctions, and defects to the appropriate design organization which has the expertise to properly conduct an aviation safety risk assessment. The intent of §5.51, §5.53 and §5.55 can be met by §21 (37(m) and (n), and §183.63(a) is satisfied, when this information is provided to the DAH for use in the COS program

§5.51, §5.53, and §5.55 for PAHs are not intended to require these organizations to duplicate the design engineering expertise to conduct aviation safety risk assessments. With regards to implementation of new systems, revised systems, or development of operational procedures the intent of §5.51 is to ensuring a conformed product or article is being produced, as depicted in Figure 6.

In the production and airworthiness certification life cycle phase, the organizational / operational hazard is the condition(s) that could foreseeably lead to a nonconforming product or article with an unacceptable risk. For organizational / operational changes, planned or unplanned, business practices in accordance with §21.137, §21.140, and AC 21-43 assures that the changes will not lead to a nonconforming product or article with an unacceptable risk, as depicted in Figure 6.

4.5.2.3 Subpart D: Safety Assurance

Subjort D provides the process for developing procedures within the SMS that function systematically to ensure performance and effectiveness of the safety risk controls. It is comprised of three sections.

- 65.71 Safety performance monitoring and measurement;
- · 65.73 Safety performance assessment, and
- §5.75 Continuous improvement.

Specifically, these sections require:

- · Data to be acquired and analyzed.
- Assessments to be made of the system's performance and effectiveness of §5.55(c) (safety risk assessment and control).
- A change management process be in place,

COT. 11/5MS-Auation Rulgmaking Lonmun-	Page	28 6/ 20
SMS-Warking Group Report	15:00	Jan 17, 2011

- A means to provide feedback on new identified huzards in ineffective sities risk control to the SRM process required by §5.51(ii), and
- A process to be in place to correct substandard safety performance.

4.5.2.3.1 Design and Type Certification

The intent of these sections, during the initial design and type certification phase of the product or article, is to consider the safety lessons learned from the data gathered and analyzed on existing designs for compliance with §5.71. During the initial design and type certification phase of the product or article, safety performance is indeterminate (i.e., unti) the product or article is put into operation its actual safety performance cannot be determined) and compliance to §5.71. Jand meeting the intent of Subpart D (SA)) is by consideration of safety lessons learned into the initial design and type certification of the product or article. In addition, an evaluation of design processes based on safety tessons learned should be conducted and decisions made whether changes to the design and certification process should be enacted to avoid future unacceptable safety risk being introduced into service.

For an ODA holder, the intent of §5.71(a)(1-6) process and procedural requirements (e.g. auditing of processes and systems) can be addressed by current 14 CFR Part 183. Subpart D, assuming the SMS processes and systems are controlled under the same command utedia system as the ODA. For example, if the organization's processes and procedures are documented and the organization uses internal or self-audit to ensure it is following its documented processes and procedures, and this system is used to satisfy 14 CFR Part 183, Subpart D §183.53(c)(5) *A process and a procedure for periodic andit by the ODA Holder of the ODA Unit and us procedures*, then it meets the intent of §5.71(a)(3).

For the ARC's proposed DO, the intent of §5.71(a)(1-b) process and procedural requirements would be part of the expected Part 21 DO certificated organization's regulations equivalent to 14 CFR Part 183, Subpart D (N5.53

Section 5.71(a)(7), which requires a confidential employee reporting system, is a new requirement not addressed by the current DAH or ODA regulations, and is intended to be in place during the initial design and type certification phase. Many large DAHs have already implemented employee-reporting systems with various structures.

Section 5.73(a)(1) through (3), require evaluating the performance of the SMS and evaluating the effectiveness of the safety risk controls. Section §5.73(a)(1) through (3) are new requirements not addressed by the current DAH or ODA regulations, and are intended to be in place during all life cycle phases of the product or article. It is recognized that processes and data to evaluate those processes may in some cases be unavailable or not field developed. Therefore, evaluation of the safety performance against the safety objectives may be done in an incremental findmon. Additional policy and guidance material is required.

Section §5.73(a)(4), which requires identifying changes in the DAH operational environment that may introduce new hazards [i.e., change management – reference Figure 6, is a regulatory input to implement §5.51, the organization must manage changes to the organization to ensure it continues to produce a compliant and/or conformed product or article. For example, if an organization has used an Airplane Level System Safety/Certification process like that depicted in Figure 7 to produce a compliant type design and is proposing to introduce a new process to produce a compliant type design then they would need to conduct a change management review.

Part 71 / SMS & ention Rubrooknes Examples	Page	29.61.51
SMS-Working Group Report	Beytsimi (Jato)	Original Jan 17, 2014

to ensure the new process would still produce a compliant type design to meet the intent of §5.73(a)(4).

For an ODA holder, the intent of §5 73(a8/4) can be satisfied by 14 CTR Part 183, Subpart D §183 33 (Provedures manual).

For the ARC's proposed DO, the intent of §5 73(a)(4) would be part of the Part 21 DO certificated organization's regulations equivalent to 14 CFR Part 183, Subpart D §187 53 (Procedures manual)

Section 5.75 is a new requirement not addressed by the current DAH or ODA regulations, and is lotended to be in place during all life cycle phases of the product or article. It is recognized that processes and data to evaluate those processes may in some cases be unavailable or not fully developed. Therefore, the continuous improvement based on the safety performance against the safety objectives may be done in an incremental fashion. Additional policy and guidance material is required.

4.12.3.2 Continued Airworthiness

One of the primary intents of Subpart D – Safety Assurance during the continued arrworthiness phase is to monitor reported in-service and other problems and to read the information into the organization's SRM process as depicted in Figure 3 and Figure 5. In most, if not all COS programs, monitoring of reported problems is the first step in the process. The reporting process may include in-service feedback received via the quality system program used to comply with \$21.137(m) (*Invervice feedback*). The reported problems are then assessed by the organization's SRM as discussed in section 4.5.2.2 in the Subpart C. Safety Risk Management section to identify hazards or ineffective risk controls, makes safety risk assessments, and identify safety risk controls when an unsafe condition is identified. A COS program that monitors reported inservice and other (reference Figure 5) problems, failures, malfunctions, and defects per §21.147(m) or potential non-compliances meets the intent of §5.71(a)(1), (2), and (6).

For an ODA holder, the intent of \$5.71(a)(1), (2) and (6) can be satisfied by 14 CFR Part 183, Subjurt D §183.63(a) and (b)

For the ARC's proposed DO, one of the expected obligations under Part 21 will be to comply with the applicable sections of Part 5, which will include compliance with §5.71(a)(1), (2), and (6). A COS program that monitors reported in-service and other (reference Figure 5) problems. Failures, mulfunctions, and defects per §21.137(m) or potential non-compliances meets the intenof §5.71(a)(1), (2), and (6).

With regards to §5 71(a)(5), because incident and accident investigation is the purview of the NTSB (or other national agency) and controlled by their protocols, the intent is to have a process or system in place to support an incident and accident investigation, if requested by the NTSB (or other national agency)

With regard to §5 71(a)(5) for an ODA holder, as previously stated the intent of §5 71(a)(5) can be met by the same method used to satisfy 14 CFR Part 183, Subpart D §183 53(c)(5) 'A process and a procedure for periodic and it by the ODA Holder of the (10)A Unit and its procedures

Part 21/SMS-Aviation Rulemaking Communica	Page	50 n/ 31
SMS Working Group Report	(Jato)	Jan 17, 20114

For the ARC's proposed DO, one of the expected obligations under Part 21 will be to comply with the applicable sections of Part 5 which will include compliance with §5.71(a)(5) which will require an audit process equivalent to that for §183.53(c)(5) or §21.137(1) (Internet andits)

As previously stated section 5.71(a)(7), which requires a confidential employee reporting system, is a new requirement not addressed by the current DAH or ODA regulations, and is intended to be in place during the continued airworthiness phase as well as the other phases of the lifecycle of the product or article. However, many DAHs have implemented voluntary confidential employee reporting systems on their own.

Section 5.73(a)(1) through (3), require evaluation of the performance of the SMS and the effectiveness of the safety risk controls. The intent of §5 73(a)(1) is to ensure compliance with the risk controls established by the D&M organization. The intent of §5 73(a)(2) is to evaluate how well the SMS processes are working. The SMS process performance evaluation indicators may change over time as better measures of the health of the system are understood in as the system matures. Additionally, the SMS process performance indicators may not have an absolutivalue nor are they intended to indicate compliance or non-compliance to §5 73(a)(2). Normally, the SMS will be looking for trends in the performance indicators (e.g., if the indicators changes one standard deviation negatively from the norm). In this case an assessment of the SMS may be warranted to hetter understand the negative trend and an action plan developed to improve the functioning of the SMS.

The intent of \$5 70(a)(3) is to evaluate how the product or article is performing in-service against the organization's safety objectives defined in accordance with §5.21(a). The organization is expected to develop and maintain appropriate safety related performance indicators. These indicators should be precursor indicators of incidents or accidents (e.g., high-speed refused takeoffs). It is not expected that there would be a safety performance indicator for every safety risk control, given that compliance to each airworthiness regulation is considered a risk control-However, depending on the product or article there may need to be more than a single indicator and for products it is expected that there be more than one indicator. As previously discussed with regards to \$5.73(a)(2) the safety risk control performance indicators may change over time to better monitor the safety risk of the current fleet. Additionally, the safety risk control performance indicators will not have an absolute value nor are they intended to indicate compliance or non-compliance to §5 73(a)(3). Normally, the SMS will be looking for trends in the safety risk control performance indicators (e.g., if the indicators change one standard deviation negatively from the norm). In which case, an assessment of the existing safety risk control may be warranted to better understand the negative trend and an action plan developed to reduce the safety risk in the fleet

Section 5.73(a)(1) through (3) are new requirements not addressed by the current DAH or ODA regulations, and are intended to be in place during the continued involutionesic phase as well as the other phases of the breezele of the product or article.

As previously stated §5.73(a)(4) requires identifying changes in the operational environment that may introduce new hazards (i.e., change management – reference Figure 6), has basically the same intent as §5.51. The organization must manage changes to the organization to ensure it continues to meet its safety objectives.

For an ODA holder, the intent of §5 73(a)(4) can be satisfied by 14 CFR Part 183. Subpart D §183.53 (Procedures manual)

Part 21 / 5MS-Acottion Rulemaking Lommittue SMS-Working Group Report	Page	71 m/ 72
	15:00	Jan 17, 2014

For the ARC's proposed DO, the intent of §5.73(a)(4) is expected to be part of the Pair 21 DO, certificated organization's regulations, which can be considered equivalent to Part 183. Subpart D 183.53 Procedures manual.

Section §5.73(a)(3) is an outcome of the evaluation of the performance indicators required by §5.73(a)(2) and (a)(3). If new hazards or safety issues or concerns are identified by the evaluations performed for §5.73(a)(2) and (a)(3) they are feed into the SRM process per §5.51(d).

Section §5.75, requires processes to provide angoing improvement in the performance identified in the assessments conducted under §5.71. The intent of this requirement is to use the data from §5.73 to focus improvement where indicators identify a need for action.

4.1.2.2.3 Production and Airworthiness Certification

PAH organizations under Pari 21 Subparis G. K. and O are authorized to produce products or unclas. STCs are only a design approval and production of an STC needs to be performed undea separate authorization. (e.g., a repair station authorized under 14 CVR Part 145). With regards to PMA and TSOA (per Part 21 Subparts K and O), this section only describes the production aspects of the organization. The design aspects of PMA and TSOA are discussed in the design and type conflication and continued airworthingss sections. Each production approval organization under Part 21 Subparts G, K and O is required to have a quality system in accordance with §21137 FAA AC 21-43 establishes acceptable means of compliance for PAH organizations as well as quality system for §21 137 14 CFR Part 183 Subpart D defines organizational requirements for ODA including Production Certification (PC), PMA, and TSOA functions, FAA Order \$100.15B establishes the procedures, guidance, and limitations of sutherity for organizations under ODA. Under §23-137(m) PAH organizations and/or §183 Subpart D PC, PMA or TSOA ODAs must have procedures for receiving and processing feedback on m-service failures, malfunctions, and defects. These procedures must include a process for assisting the DAH to address any in-service problem involving design changes, and determine if any changes to the ICA are occessary. Because the production organizations may not have to design engineering expertise to conduct SRM their primary function for SMS is to support the COS program by providing information regarding in-service failures, malfunctions, and defects. Table 3 provides a correlation showing how the existing 14 CI/R Part 21 and/or the Part 183 Subpart D regulations, and guidance can meet the intent of Part 5 Subpart D regulations

2 FARC Working (Recurant - Not for Distribution

for 1	/5415-Auration Halemaking Lommittee
	SMS-Working George Report

Page 32 n/ 3 Bevision Original Data Ian (7, 2014

NPRM PART 3	PART 21.137/ AC 21-43	PART 183 SUBPART D ORDER SD00,15B	COMMENTS/NOTES
571(a)			The Revenue of the second seco
57laikin	21 117(αδ. (b), (c), (d) (c) (D, (g) (b), (b) and (c), AF 21(4) % 2-1 (b), 2-11, and 2-36	18141,183,47 183,53 183,6140)(34,75) (75,016)(35,15) (75,016)(65,11)(21) and 183,63	For a PC. PATA, or TSOA holder (in mean of \$5.71(ii) process and procedural requirements are used by compliance to 21.137(a). (b). (c). (d). (c). (f). (g). (h) (f) and (f). For an PC. PMA, or TSOA ODA holder. (he mean of \$5.71(a) process and procedural requirements are addressed by 14.07R. Part 18.1. Subpart D: assuming frue SAAS processes and system as the ODA.
27(682)	21 147(a), (b), (c), (d), (c), (f), (g), (h), (l), and (n) AF-21-43 *s 2-6 = 10 2-14, and 2-16	1113:53(b) (c)(5), (0), and 183.63	See comments to §5.71(a)(1) With regards to detecting operational environment changes, the mean of §5.71(a)(2) is satisfied by §187.67 and §187.53(b). (c)(5), and (0) as they relate to ensuring a conformed product or article is still being produced, as depicted in Figure 6.
578(083)	21.17700 AC-21-13 4.14	(83,557(c)(5),16) Mater 8100 (AH * 7-41	See comments to §5.71(a)(1) If the organization is processes and procedure sum documented and the organization poes internal or self- andii to ensure it is following us documented processes and procedures: and this system is med to smish 14 CFR Pan 183, Subpart D, (183.53(cu.5) – A process and a procedure for periodic audit by the ODA Rolder of the ODA Unit and its procedures," then in- meets the ment of §5.71(a)(3)
5.78(à)(4)	21:137(a), (b), (c), (d), (c), (f), (g), (l), (l), and (n) AE 21-13.%5 2-1 - 10 2-14, and 2-36	187.53(6), (59.5), (6), and 181.83	PAH evaluation of its SMS is within the scope of the PAH's SMS, which is primarily reporting of major in- service safety feedback or infervisionificant quality escopes to the DO. Major in-warvice safety feedback or safety regulierant quality escopes are those items that are potential COS items, as defined by the DO.
TTINK51	ΝA	NA	Because incident and accident investigation to the purview of the National Transportation Safety Hoard (NTSED (or other national agency) and counteiled by their protocols, the ment is to have a process or system in place to support the DO in their support of an incident and modern investigation if requested by the NTSB (or other national agency).
\$71(a)(6)	2) 137(a) AL 21-43 5 2-10	183 ((ite) Order-8100,15B % 3-18	The intent of $$5.71(a)(6)$ is met by contribute 0 \$21 + 37(a) and $$185, 63(a)$ for PE. PMA and TSOA ODA halders.
\$31(m(7)	12.4	NA.	Confidential employee reporting is not required by (21)117 or 14 UPR 188 Subpart D A PAH confidential reporting system can be confidential with of the organization s cristing confidential reporting provided the system can distinguish three reported iteris that refute to the safe operation of the product or arrele to the NAS.

Table 3: Correlation Between Part 5 and Part 21 / 183

	SM5-Wa	ising Group Real	Bevision Organi Date Jan 17, 2014	
NPRM PARTS	PART 21, 137/ AC 21-43	PART 193 SUBPART D ORDEB 8106,150	COMMENTS/NOTES	
≦'TI(n)	21 137(6), (no), and (n) AC J1-41 % 2-40, 2-15, mint 2-16	183 (346) Order 8100 (548 * 3-18	In accordance with (21) (3700, 000, and 00) AC 21-47 *0 2-10, 2-13, mit 2-10 mad 518167(a) Order 8100, 156 * 1-18 me intent of 85 7100 is addressed by a PAH baying a QMS with appropriately trained initiarized individuals who can make date minations of major in-service safety fordinack or safety significant quality excupes such that its information is provided to the DO. Major in-service safety feedback or wifety significant quality escapes are those across them to potential COS items, as defined by the DO.	
3 79(a)				
# 21(a)(2)	21 1370) AC 21-43 22-15	187,53(e)(5) (6) Order & Ion 158 * 3-14	The intent of §5.75(0)(2) is met by compliance with either §21 (17(i) or §181 55(c)(5), and (6) provided the ordif includes the procedures associated with the PAH a QMS for has no appropriately tenned authorized individual, who can make determination of import in-service nately feedback or safety significant quality escapes with that this information is provided in the DO.	
5 #Max31	21 19700 (m) and (m) AC 21-1935 2-10 2-15 and 2-10	181.00(a) (c) Order 8 (e) 158 ¶ 3-18	The intent of §5.73(a)(7) is met by compliance with cillier §21 +17(h) and (a) or, §183 63(a) and (c), limited to the PAT+5 QMS determination and reportin of major in-service solety feedback or salety sterifican and the vacanes to the DO.	
3710041	$\begin{array}{c} 21 + 17(a), (b), (c), \\ (d), (c), (f), (g), (b), \\ (l), and (n) \\ \Delta (-2) + 3 \cdot 5 \cdot 2 \cdot 1 - 10, \\ 2 \cdot (4, and 2 \cdot)6 \end{array}$	183 53(b), (c)(44), 55(a), and 57(a) Order 8400, 15B %, 3-9 and 3-12	As depicted Figure 6, the mem of §5.73(a)(4) for a PAH is ensure any changes do not would to not producing a conformed product or article in a condita for safe operation, and to perform their OMS function to described herein.	
7760051	21 (37(a), (b) (c) (d), (c), (f), (g) (b) (b) (nat (n) AU 21-13 % 2-1 - 10 2-14, and 2-10	183-55(b), (c)(14), 55(6), (c)(14), 55(6), (c)(157(c)) Circlet 8100, 15B % 3-9 and 3-12	As depicted in Figure 6: the intent of §5.75(a)(5) for a PAH is ensure any changes do not result fit not producing a conformed product or attacke in a condition for safe operation, and to perform their (2MS functions as described herein	
571061	21.137(b)/(m), and (n) AC 21-43 %5 2-(0), 2-15, and 2-16	183,63(c) Minder 8300 15H * 3-18	In accordance with §21–137(h), (m), and (n) and AC 21-43 §92-10, 2-15, and 2-16 and §183.63(c)/Order 8100-15B § 3-18 the intent of §5 75(b) is addressed by a PAH having a QMS with appropriately minod authorized individuals, who can make determinations of major mesoryice safety feedback or safety significant quality escapes such than this information is provided 6 the DO. Major mesoryice safety feedback or safety significant quality escapes into those norms that are potential COS items is defined by the DO.	
016	2) (3700 AC 2143 (246	183, 63(c) and (d) Order 8100, (5B * 3-18	The instan of \$5 71(4)(6) is met by contributed to \$21,137(a) and \$185,63(a) and (d) for FC, PMA, and TSOA ODA bolders.	



Figure 1: Three Phases of the Life Cycle of a Product or Article





Figure 2: Intent of SMS Subpart C (SRM) and D (SA) During Design and Type Certification

Part 21 / SMS-Aviation Rulemaking Committee SMS Working Group Report	Page: Revision: Date	35 of 36 Orignui) Jan 17, 2014
---	----------------------------	--------------------------------------

Subparts C (Safety Risk Management) & D (Safety Assurance) for Continued Airworthiness



Figure 3: Intent of SMS Subpart C (SRM) and D (SA) for Continued Airworthiness



Subparts C (Safety Risk Management) & D (Safety Assurance) for Production and Conformity / Initial Airworthiness

Figure 4: Intent of SMS Subpart C (SRM) and D (SA) during Production, Conformity, and Initial Airworthiness



Figure 5: SMS Subparts C (SRM) and D (SA) Process Description

Subparts C (Safety Risk Management) & D (Safety Assurance) Operational Procedures & Change Management



Figure 6: Intent of SMS Subparts C (SRM) and D (SA) for Operational Procedures and Change Management

21ARC Working Document - Not for Distribution

A Report from the Part 21/SMS ARC to the FAA





4.5.1 Tasks 6 and 7: SMS WG Recommendation

Notwithstanding the gap assessment (section 4.1) which identifies regulatory gaps between Part 21 and Part 5, the D&M community has established business practices that accomplish to varying degrees the intent of SRM and SA as part of their design, certification, production and continued airworthiness efforts although not using the SMS terminology. However, much of what has been written about SRM and SA has been from an operational sense (e.g., airline flight operations, airport air traffic operations), and limited documentation/guidance exists for how these SMS elements relate or are equivalent to what D&M organizations do as part of their compliance to the current design and certification procedures, and airworthiness regulations.

To ensure that the intent of SRM and SA, when applied to D&M, are applied in the most effective and efficient manner the SMS WG recommends that interpretations and explanations provided herein be included in any preamble, policy, or guidance material. As well, depending on the future state of delegation, approved or certificated organizations, as described herein, regulation, policy and guidance should allow an organization to use the same processes and procedures to satisfy the intent of the equivalent regulations. This approach would minimize the economic burden to industry, while maximizing the enhancement to aviation safety. The SMS WG recommends that, as described in this CONOPS, existing processes and procedures be considered as meeting the intent of Part 5.

Part 21/5405-Aviation Rulemaking Communi-	Page	38 67 39
SLPs D orking Crosses Danard	Revision	()rigmi)
Safet in number (number) without	12:010	Jan 17, 2014

4.6 Task 8: Coordinate SMS WG Efforts with Organizational WG, Oversight WG, and CBA WG

Task 8 required the SMS WG to coordinate our efforts between the Organizational WG. Oversight WG, and CBA WG. This effort has occurred throughout the WG activities with key coordination occurring at weekly telecome for WG leads and also at full Part 21 / SMS ARC meetings.

Key assumptions from this coordination are provided in section 3.0 of this report

4.7 Task 9: 14 CFR 21.3 for Organizations with an SMS

Task 9 required the SMS WG to provide a recommendation regarding what §213 should tooklike for a D&M organization with an SMS.

4.7.1 SMS II G Response

Advisory Circular (AC) 21-9A (now superseded to AC 21-9B), "Manufacturers Reporting Failures, Malfunctions, or Defects," included an explanation of the rationale for §21-9. According to the AC, "FAR 21.3 is directed to those deficiencies in the product that only the manufacturers would normally be expected to know (i.e., a design or a quality control deficiency). It is expected that compliance with FAR 21.3 will provide the earliest possible notification to the FAA of a hazardous condition, and that appropriate corrective action will be torinated by the manufacturer."

The SMS WG notes that SMS, and Part 5 in particular, introduces processes that far exceed the objectives of §21.3. Following are some points of comparison between §21.3 and Part 5.

- Absent a specific request by the FAA for more information, §21.1 requires the responsible approval holder to report only certain occurrences.
 - §5 \$1(c) would require processes intended to identify hazards brouder in scope than \$21.3, and \$5.55 would require processes to usiess safety risk associated with the identified hazards and to establish safety risk controls.
- There is not an explicit requirement that the approval holder establish a process or system to seek our and collact information, nor an explicit requirement to act upon a report unless requested by the FAA.
 - §5 71 would require SA processes that actively monitor and acquire data with respect to the holder's products, §5 53 would require that the holder identify hazards based on the collected data, §5 51 would require that SRM be applied to the identified hazards, §5.55 would require processes to assess safety risk associated with the identified hazards and to establish safety risk controls.

The SMS WG considers that exercise of an SMS inherently will accomplish the intent of §21.3 to a far greater extent than the regulation (self, §5,93(c) would require the certificate holder to have a means to convey safety critical information, which could include communicating information to the FAA that it needs in order to carry out its safety responsibilities. The WG recognizes that in order to accomplish effective oversight (including management of system level tisk, 14 CFR 39 airworthiness directive requirements, etc.), the FAA may require that certain safety information be formally reported. Such reports and the requirements) for them should be enablished based upon Part 5. Subparts E and 1

Part 21 / SMS-Aviation Bulemaking Communica	Page	59 of 10
SMS Working Group Report	13-ana	Jan 17, 2014

Some approval or certificate holders are subject to the requirements of 14 CFR Part 181 §183.61 prescribes requirements for maintenance of records and reports. §183.63 requires an ODA holder to notify the FAA of a condition in a product, part, or appliance that could result in a finding of unsafe condition by the Administrator. For the same reasons as noted above for §21.3, the requirements of Part 5 meet the intent of sections §183.61 and §183.63. Thus, an ODA holder having an acceptable SMS would be subject to redundant process and notification requirements.

Many organizations that would be required to have an SMS also hold ODA and have \$21.3 reporting requirements. For such an organization, there now exists high probability they would be required to have and use at least three separate processes, making separate respective reports to achieve a single common objective.

Similarly, such an organization may also be subject to other reporting redundant requirements or processes to some extent during exercise of SMS processes. Those other reporting requirements or processes might include, but not be limited to

- \$13.1. "Reports of violations of, inter alia, rules, regulations, and orders issued under the Federal Aviation Act of 1958."
- §21 139, "Location of or change to manufacturing facilities."
- &21 150, "Changes in quality system."
- §1915, "Submission of safety information."
- 5193.11, "What is the notice procedure?"
- §193 13, "What is the no-notice procedure?"

The SMS WG concludes that there is significant potential for design and/or manufacturing, organizations that have an acceptable SMS to remain subject to redundant reporting requirements. Redundant requirements create the potential for inefficient and inconsistent reporting by an organization, and inefficient and ineffective oversight by the FAA. The SMS WG further concludes that the FAA should task an ARC sub-team to develop recommendations to minimize or eliminate redundant and inconsistent reporting and record keeping requirements for an SMS organization. The WG also should recommend appropriate reporting requirements and systems to enable the FAA to exercise its oversight and system level risk management responsibilities. The sub-team tasking should include consideration of any potential for redundant risk assessment requirements, such as those within ODA processes or agreements that a COS program may be used to satisfy §21.3

4.7.2 Task 9: SMS WG Recommendation

The SMS WG recommends that the FAA task an ARC sub-team to develop recommendations to minimize or eliminate redundant and inconsistent reporting, record keeping, and tisk assessment requirements for an SMS organization. Additionally, the sub-team should address appropriate reporting requirements and systems to enable the FAA to exercise its oversight and system level with management responsibilitie.

Fort 21./5405-Acotton Rulemaking Lommone-	Page
SMS-Warking Group Report	10alo

SECTION 5.0 Additional Recommendations

The SMS WG has identified the following additional recommendations based on our deliberations

5.1 Acceptable Safety Risk

Part 5 requires that the D&M organization define a process for conducting risk assessment that allows for the determination of acceptable safety risk. The FAA would then accept the risk criteria proposed by the D&M as part of FAA acceptance of the SMS. However, there are no documented processes for determining acceptable safety risk for in-service issues on many products and articles. For example, the FAA has developed several risk assessment criteria. These include.

- AC 39-8: Transport Category Engines and APUs.
- FARAM: Transport Category Aircraft Risk Assessment Methodology
- + SARA Small Aircraft Risk Assessment.
- RRA, Rotorcraft Risk Assessment.
- E&PD RA: Non-Transport Category Engines Risk Assessment.

Note that only AC39-8 has reached the level of manufity that it is mutually accepted by both D&M and FAA as an acceptable risk assessment criteria.

5.1.1 SMS WG Recommendation

The SMS WG recommends FAA establish WGs to develop and refine risk acceptance criteria for all types of products and articles AC 39-8 provides one example of a risk acceptance criteria that has been developed (for transport category engines and APUs) in conjunction with D&M and is mutually accepted by FAA and industry. Agreement between the D&M and the FAA on acceptable safety risk criteria will reduce the need for the FAA to duplicate safety risk assessments conducted by the D&M. This will also facilitate efficient process oversight of the D&M SMS.

5.2 Availability of Data for SRM

The SMS WG noted that D&M organizations' ability to effectively implement SRM to address in-service risks could be constrained by the availability of in-service fleet data. The quality of SRM for in-service risks depends upon having data such as the number of homs/flights over a given time period, the number of component failures, component rentovals/reptacements or inspections, and similar information. This is used to develop statistical models of fleet behavior, estimate failure rules, and projected numbers of future failures, recognize a failure does not necessarily denote an aircraft accident/incident in this discussion.

There is no regulatory requirement for operators to report any of this information to the TC / STE / TSOA / PMA holder or PC holder. Large D&M organizations, especially with a small number of high-volume customers, have been able to invest or a reporting infrastructure, which enables them to support a COS program. Smaller D&M organizations or those with a very large.

21ARC Working Document - Not for Distribution

10 of 11

Finguni Ian 17, 2014

Port 71 / SAPS As ration Robertal inter Communes	Page	41 of 43
SMS Working Group Report	Bevision	Original Inv IT 2011
	12/00/	

number of low volume customers may not have the same infrastructure or availability of data to conduct SRM similar to previously mentioned D&M organizations

Since quantitative SRM can provide benefits beyond those found in qualitative SRM, enabling data flow from operators to D&M organizations may be beneficial. Possible approaches include

- Providing data already reported to regulators back to D&M organizations in a timely and efficient way. Currently, any organization may request this data, but there is a significant time lapse between the data request and provision of the data, a subscription approach might be possible.
- Making the data already reported to regulators readily available [e.g., establishing a web interface accessing FAA Service Difficulty Reports (SDRs) database that D&M organizations could query).

5.2.1 SMS WG Recommendation

We recommend that the FAA develop an approach to make fleet data atready provided to the FAA thours, llights, reported failures, unifilaterizes, and defects and service difficulty reports) madily available to D&M organizations, in support of executing SRM

Part 21 /5415-Accation Rulemaking Lonnonnee	Page
SMS-Working Group Report	15alo

12 of 13 Organi) Jan 17, 2014

.

SECTION 6.0 APPENDICES

Table of Contents

Appendix A.	SMS WG Charter
Appendix B.	SMS WG Membership
Appendix C	Detailed Gap Assessment between Part 21; Part 5, and ICAO Annex 19
Appendix D	Safety Risk Management (SRM) and Safety Assurance (SA) in Certification Description of How Compliance with Airworthiness Standards Meets SRM and SA Requirements During Design

21ARC Working Desauaant - Not for Distribution

Page

13210:

Appendix A. SMS WG Charter

A-IScope and Boundaries

As recommended by the ICAO FAA safety program identifies the hundamental safety management concepts and practices applicable to the implementation and oversight of SMSs for product providers. The FAA has proposed to incorporate these fundamental safety management concepts and practices into a new 14 CFR Part 5. Safety Management Systems

The SMS WG will review the applicability of the proposed Part 5 SMS requirements within Part 21 for the D&M environment. The goal of the WG is to provide the FAA recommendations for rulemaking, suggested processes, policies, and guidance to align Part 21 with the SMS requirements documented in proposed Part 5. The WG should also identify and assess any differences between the recently adopted SMS requirements in ICAO Annex 19 and the proposed Part 5.

One option for the WG to evaluate is the integration, to the maximum extent practical, of the fundamental SMS concepts and practices into Part 21. The SMS WG will identify where and when each element of SMS can be used as a tool in the design, organization, and production processes so that once integrated, Part 21 can be considered SMS compliant. Therefore, each design and/or manufacturing organization producing a type design or product under Part 21 can be also considered to satisfy the intent of SMS in the international regulatory marketplace.

Another option is to point from Part 21 to Part 5. In consideration of this option, the WG should provide the FAA recommendations of any preamble material, guidance material, and proposed changes to Part 5 that are necessary from a D&M perspective.

The WG should evaluate these options (plug any others fell appropriate) and provide. recommendations to the FAA on the optimum solution for the application of SMS to D&M organizations.

Finally, the WG should provide a recommendation to the FAA as to which D&M approval holder organizations should have SMS requirements applied to them. This recommendation should include scalability of SMS to D&M organizations based on their size, complexity, and safety risk introduced by their products.

4-2 Assumptions

- 1. The WG will use the SMS requirements as defined in the FAA proposed Part 5 and ICAO Annex 19 for comparison and evaluation. SMS implementation into 14 CFR is part of the broader ARC work to improve the regulatory processes and privileges for D&M.
- 2. Hazard is defined in accordance with FAA Order 3040.4a (Safety Risk Management Policy)

Tasking 1-3

- Perform a regulatory gap analysis between the SMS regultements in Part 5. ICAO Aones 19 and Part 21.
- 2 Develop, evaluate options, and make a recommendation, for incorporation of SMS into 14 CTR for D&M organizations

Part 21./ SMS-Aviation Relemaking Community	Page	AJ of A.2
Appendix A. 8348 Working Group Charter	Data	Oct to. 201

- Frovide a recommendation as to which design and production approval holder organizations should have SMS requirements applied to them
- 4 Produce recommended preamble, regulatory, and guidance material for SMS application to D&M organizations. Include an "operational" definition of a hazard throughout the life cycle of a product in SRM.
- 5 Provide guidance that accounts for scalability of SMS to D&M organizations based on their size, complexity, and safety risk introduced by their products
- 6 Define the end state (roles, relationships, and responsibilities) for SMS in a D&M organization considering design, certification, production, and in-service support.
- 7 Perform a review of "strategic" safety decisions that are required to be made during the design, certification, and production phases throughout the life cycle of the product.
- Coordinate SM5 WG activities with 21 ARC and Oversight, CBA, and Organizational WGs.

A-4 Deliverables

- Options, pros, cons, and recommended option for incorporation of SMS into 14 CFR for D&M organizations
- Recommended CONOPS for SMS (end state and potential strategic safety decisions per Tasks 6 and 7)
- 3 Recommendations on how to implement SMS including preamble, regulatory, and guidance material for incorporation of SMS into 14 CFR which includes applicability and scalability for D&M organizations.

Part ZL/3	AIS.	Anta	tion.	Lulem	along l	ommone
Append	In A	. 53.00	-14:01	riute	Group	Chirter

4-5 Milestones and Dates

DATE	ACTION	EXPECTED DELIVERABLE
Apal Sh	Review and finalize Channe with SMS WO. Haseline Wisson (CAF) domes 19 and Part 5 SMS readorements	Finalized SMS WC Courses tennolouselined
/April 30 / ይ እርድ 1	Flenence WEI Meaning	Parview sub-tenar progress? molecu 1. That 5 to Part 23 gap review mont 2. RCAO Atomy 19 to Part 1 year resourcement Degit form 18 or part to 3. Incorporation april no 4. COBO 55.
7ime 11 42 (2	Nenifi3 WLT Nizeting	Employe Englishing a set of the second sec
ABCTC	Present states to Part 21 / SMS-ARC	1 Gap Assessment / Kananax 2 Jucinparalina anti-ar 1 CONOTS
tali 17 k≤48	Threather WG Meading	Employ Encommended to a model on option CONOUS Reveny solution program on Design approval budget on SMS scalateday one flee onlow SMS scalateday one flee onlow Sentegie safety doct m Hourt four discussion Recommended prombine enclatery and grademental
()1 <u>111</u> 1111 (736,14	* Wieinen Wiff Meeting	Timilize 1 Danger approval babbles with supposed bits 2 SMS scalables and the article 3. Strategic safety decision Review safetiam program in 1. 4. Recommended pseudologic regulations and point action 5. SMS WG Trail Report 5. SMS WG Trail Report
5q (7	Present recommendations to Fars 71 (SMS- ABC)	Gap assessment formulars Incorporation government CONTRES Dealgn approval transfer SND coplastiking SNS scalarifly, used the during SnS scalarifly, used the during
640 18 & 19	Rosden Will Alzeing	Initize Kernemendeal preamitie, regulatory and guidance spaterul Withite essentiated / rejubri totaline Similar essentiated
(0c) 30 (£ 31	Throwing WG Memory	Drail SMS WG Benni mysey and and un
December 11/ # 11	Rosslyri WG Messing	Final report
Der D & 13	71 ASC Marine	Report on to TRU
Summy 13.25 2054	Phinems WG Meeting	Finalize assist
January 17 2014	Final Report Due	Final Report delivery in ARC
August August	T THE THE OTT AND	a room for hard set the over all filles

A-6 References

- ICAO Annes 19 (C-WP/13935)
- Proposed Part 5 [Docket No. FAA-2009-0671; Notice No.10-15]
- Part 21 / SMS-ARC Report
- Aircraft Certification Service (AIR) Manufacturers Safety Management System Pilot Project-Report – Design and Manufacturing Organizations
- Certificated Design Organization (CDO) ARC Report

4-7 WG Lead(s) and Members

Chris Eick (SMS WG Chair, Honeywell) Bicardo Aoyama (Embraer) Michael Bartron (P&W) Rafael Ximenes Borges (ANAC) Fabro Catani (Embraer) Peter Corbeel (EASA) Marco Cuberos (HEICO) Mike Deer (Bell) Fred Etheridge (Gulfstream) Cluick Huber (FAA) Doug Kihm (Boeing) Sarah Knife (GE) Jason Lewis (BE Aerospace) John Loehr (Cessiii) Stacy Mason (TCCA) Claire Pelegrin (Arthus) Ric Peri (AEA) Rogério Possi Jr (ANAC) Enc Sivel (EASA) Dean Thompson (Beechcraft) Chris Willingham (Wencor) Jell Wood (Boeing) Paul Vocera (Cessna)
Appendix B. SMS WG Membership

SEGMENT	ORGANIZATION	NAME	
Association - Alternit Electronics Association (AEA)	Aircraff Electronics Association (AEA)	Rac Peri	
Association AeroSpace and Defense Industries Association of Europe (ASD)	Airbus	Cinice Felegana	
Association Modification and Replacement Phys Association (MARPA)	BE Aerosphee	Jason Lewis	
Association Acrospic: Industries Association of Canada (AIAC)	Bell	Mike Deer	
Association Accessance Industries Association of Brazil (AIAB)	Eminner	Barandri Auxanna	
Association Acospace Inducates Association of Beazil (#IAB)	Embrace	Entries Catalor	
Association Modification and Replacement Parts Association (MARPA)	Helco	Marco Cubenn	
Association Modification and Replacement Parts Association (MARPA)	Wenzor Comp	Chris Willingham	
Transport	Boring	Doug Kihos	
Transport	fioann;	Juil Wood	
Transport General Aviation	Gullistram	Fred Etheridgo	
General Aviation Part 23	Beecherati	Dean Thompson	
General Aviation Part 23	Cessia	Pnul Vnoem	
Isonarol Avtaison Part 23	Cessar	dala Loehr	
Engines	General Eleptric	Sarah Knif:	
Systems	Honeywell	Chris Elek (WG Chair)	
Romanni	Profil & Wonney	Michael Bartron	
Federal Avoino Administration	ANM-ITER	Clinick Hoher	
Observer	European Astronan Salers Avenes (EASA)	lani Novidi	
Objectives	Transport Cantala Civil Astanion (TCCA)	Succy Mason	
Observer	Sentonal Civil Aviation Anamos- Brazil (ANAC)	Balaci Norman Borges	
Observer	Summer Civil Availon Amarcy - Benzitt ANACI	Reiserio Possi Innior	

Patrill	SARS-	Seimma J	lalemuking	L'amminue
				and the second se

Conte Recruitme Date:

4. J. of P. J. (Anglea) 134, 10, 2015

Appendix C. Detailed Gap Assessment between Part 2J, Part 5, and ICAO Annex 19

ICAUANNA II IICAMEWERE COMPONENT	FRAMEWORK REALEWORK ROOT REALESTS	HETH PARTS ADDIST LANDON STO HELDENT	PART & APRM . REALIZED ARCATE	DUTERIACEA INFORMEDA ICANANSEA (PARI) EFETERARIA	GAPSIAL ACTAILSY HETWITY GATE PARTS AND PART 20
Philips (F Solity Manuellarty Rydiae (SMb)	11.5 The SMR (Fair regrestation resembled for Q-space for gen- ment of the Communication) means (Add) for mana- europeable (One Sing) of Heaps 41,0; fb; MMR of an engineering resemble development man- resemble development resemble development (Communication) and a second second development (Communication) and a second second development (Communication) and a second with the Sing Second Wateriment	Xt application	Lei JA, et et musici le traine sustante 19 ETU. Since 100 et d'un admigne autoinne de la constituit de generation en canadicate. Ferdi- tion avair internet na 1976 (10, 1981 E2) et 1000 dégres a specificação a facilita- tiva esta aparte a servicio a facilita- tiva esta aparte autointe a canada de internet esta esta esta a facilitada de la servição de la canada de la des parte esta esta esta esta esta esta esta Directional.	si menti sun sun angkonistra Con Van Li angeon in	Contains 167 / in orth application on (ACDA [16] [2] oppose
Augenalis 7 Francesor, Ro a Sadoj) Manegenerar System (Shing	1.1.1.1. The supervise previous block devices an ALE symplectic presidence press invested to an atomic the engine entropy of a statement to engine entropy of a statement or trading entropy of a statements. Instance efficie source the engineering of a statement of the engineering of the statement of the engineering of the enginee		(b) A serificate behavior to a total re- instituciation of the noise (1) A software the analysis of the the share the strength of the the strength share the strength of the the strength of the first twee.	Part 2 states for payour ing discount on payour provide any order on of payour	1
Appendit, 2 Promonante for a Autory Management System (MIS)	1.1.1 The on-type provide shall devide as 54.0 stypersection (they brindly and type as the organization of the device due organization of the device due organization of the device due organization of the device due to the device of the device due to the device of the device due of the device of the device of the device of the device of the device of the device of the device of the device of the device of the device of the device of the device of the device of the device of the device of the device of the device of the device of the device of		(a) The application of a plantary addition are of the conflictive settles excerning the settle plantary in a data the obtained of the neuron of the many settlement of the plantary interpretation of the plantary building interpretation of the settlement building.	The Transmission of the power of the power of the second s	
Aggentity 2 Processors for a Kongy Klaningsaulus System (RAS) (RAS) (RAS) (RAS)	 Jenno and a and dipatonic Santo syst annequient Anno pastonic Anno pastonic b = 0 mmm too 	 County Hypothesises 	 (c) disc auffiliary (other required to be explored). Management Academic analysis (blacks), was a defined for field (blacks) and the second of the Management of the samelysis (blacks) (blacks) (blacks) (blacks), and (blacks) (black) (blacks), and (blacks) (blacks), and (blacks) (black) (blacks), and (blacks), and (blacks), and (blacks), and (blacks), (black	12.2.10 panelogy no ne addressed to the strates P' Boles constants, a comparison Descentings, (proposed) to maximum terms to develop the maximum terms to develop model to request to the organization of the to-sell the sense to develop model to request to the proposed to develop model to request to the proposed to develop model to request to the the sense to develop model to request to the proposed to develop to the proposed to develop to the terms to develop to the terms to develop model to the terms to develop to the ter	

"TARE Working Documents - Time for Completions

IF ALS ADARES 19 FRAMEWYSRE CUSIONICAT ED EMPORT	SPECTIC LASTEN AN ENANGWORK BEAR BRENEATS	HATTR PART 5 WHOLCOMDINEST FLOADEST	FAILTS SPRM BEIGHBREADINTS	ULTUERIASCOS ULTWEEN ULTUERIASTA ULTUERIASTA	GAPS ALIGNAUNT INTIVITY IA CTR FAUTS AND PAIR 20
			 a.d. The countravistic of Science and Science 10 of 10 of 10 percent and 10 of 10 percent and 10 p	annullementative, and any retifies the an parameterization of an 2000	
hagur 1 Defiaillea		5.4 Deficition	Marady means to million that yet would be receive. Here, or head to parall themps in or true of a dataset, conservation on property with the antibility of a dataset. The means the energy with of provided datasets and systematic with the minimal what of a head of the second second second and a second second second second second and the second second second second second with a second second second second second with a second second second second second datasets and second second second second second the second se	Part 2 Antis many and Participane in KN-200 In 2 (Holized 1984) Part 2 Init (Known 1984) Part 2 Init (Known 2004) Part 2	

21 ARC Working Document - Not for Daurdwitten

Append	Part 21-54	S-Aviation Ridemak ont Hobseen Part 21.	ne Coordillo Propased Part 5, and ICAO	Vanes 19	Page J_i of C Revision (Megina Date Oct 10, 201
IF ALS ASSUED IN PRANE POWER CREATERN AT STRATEGY	SPECTOR SOUTH AND SUB- BROADWOOK BROADWOOK	HACTR PART 7 WHOLCONID (NY M) FLOOREST	FART # SPRM BEDY MIEMENTS	INFIGRANCES INFIGURES INFIGURES INFORMATS	GAPS (J.3075AD-87) (R.TWAT, 974 CTR 7.407-5 (550 P50 P50 P2)
			Incomposite the set of a consequence of forward to set with the set of an inclusion of the income processing of a set of a set of a set of the set of a set of a set of a set of a set of a set of processing of a set of a set of a set of processing of a set of a set of a set of processing of a set of the set of a set of processing of a set of the set of a set of processing of a set of the set of a set of processing of a set of the set of a set of a set of the set of the set of a set of a set of the set of the set of a set of the set of the set of the set of a set of the set of the set of the set of a set of the set of the set of the set of a set of the set of the set of the set of the set of the other set of the set of the set of the set of the set of a set of the set of the other set of the set of		
 salaty Policy and Diskerition 	and the second second	Subseri B - Sides Priky			
Objective This as was a monoley doll doma to Community doll doma to set to ack to a conference with Responsibility 31 This as was a monoley for a conference Responsibility This as was a monoley for a conference interview of the conference response of t	The server receive delt dense as such and a to a contrary with summeries and a solitor many received as solitor many received as a solitor	573 soldy balan	(iii) The captinear better much time a softwarmity provides a being the hybriding. (b) The offer provides on the on- transformer of the applicable conditions expressions in Chapter 1 of 1916. In or the Code of London 1 Manathers and heart priviles of construction and heart priviles of construction and heart priviles of the active and heart priviles of the active and heart priviles of the construction of the priviles of the software of the priviles of the priviles of the priviles of the software of the priviles of the priviles of the priviles of the software of the priviles of the priviles of the priviles of the software of the priviles of the priviles of the priviles of the software of the priviles of the priviles of the priviles of the software of the priviles of the priviles of the priviles of the software of the priviles of the priviles of the priviles of the software of the priviles of the priviles of the priviles of the software of the priviles of the priviles of the priviles of the priviles of the software of the priviles of the priviles of the priviles of the priviles of the software of the priviles of th		When is Connecting Part 21 Only was contrasted induced manufacture of the 23 (10% and 10%). When it Mining in Part 21 Safety price in conversa of Marin produces of the decision context of produces of the decision context of produces.
		 Be cartificant for the result time a summitty on the "the intertainent that are tune the definition of tune and the definition of the second se	Part A manual according at the left (description description of and explore a solidy commitment)	When to Convert by Port 21 Part 21 in brown have up 10 in 4 in internet product when will be up to produce the Westmann in Sect 21: Ports in a Constant in Sect 21: Ports in Constant in Sect 21: Ports in Constant in the Sect 21: Ports in Constant in the Sect 21: Ports in Constant in the Optimization of Constant intervent with Optimization of Constant intervent with Optimization	
	To send a follow second ward to protect of the one-way recovery on the applications of constant on the		(a) Character When P and a constrained three 4 microscopy positive that the transmission of the United sector of the processory of the processors of the processory measurement for the hypothematical of the SMIS	Plan 5 in town organic register growt meeting preservation of 5,46 and part labor or labor and part labor or labor	When the Concerned an Hure Tax by a compared on Prior (TaY) (b) of twitten in the origination are prior of the required traceous and the twitten of the Name of the Name of the Tax Tax Harmon, Concerned on Hure the Annual of the Annual Annual Prior and the origination are constrained on any prior constrained on transformations of the Name Prior and the Name of the Name State of the Name of the Name

2) ARE Working Document - Not for Complexition

Append	Part 21 - 5M	S-Aviation Ridemal out Between Part 21.	ant Convolution Proposed Part 5, and ICAO	Annes 19	Fuge J. 4 of the Resistant (Anglow Date: Oct 10, 2017
II ALS AANES IT FRAMEWORK CONFIDENT AT STRENT	SPECTOC SUSTS IN PRATOCORK BENTHERMENTS	HACTR PART 9 WHOLCOMPARTST	FARTS STRAT	INTERNALDA DETWEEN ICAS ATACKTRAND LICENTRAND	Сарынданулган. Эк түүлээ таста глог 5 озо руйг 2)
	stitetee officements (integras		(c) The compliance body read have a study on they free to failer a most time. Otherway, and the complexity of the endpoint complexity of the complexity operating of the body failude of proteins.		What is Concern to Durit 1)- (2): UV is expression to consistence (pression of the 1 eV During private index yours) to collection to expression index of the collection of the <u>What is Officient</u> of the <u>During Concernsion</u> (2): There is every plus happened of transferse transferse (constrained of transferse transferse)
	(i) charle independent op and behavior op case condition model is the entities provide an observe with operand instability for entraneous order of add the intervention of add in the intervention of add in the intervention.		(iii) Discontinuo Astilia eren turo t serin, rotes nel le tuto e te ordin Tettare e 1911 à polegi dali delle er concessitata bei nel a subgradure re- ciscipticari arter.	(201) Program (Astronomication officer Galery/Dears - extend 11 monthal strands, 101 registre- riftser manifolisien activity 	When the Conversion (2011) (1) (2) - 2 - Address of the Historica (1) approximate prepare and converse When the Minister in Mart 12) There is not a planet management demonstration of the Second Second demonstration of the Second Second demonstration of the Second Second Second demonstration of the Second Second Second Second demonstration of the Second Second Second Second Second demonstration of the Second S
	and a regard to the accordance annuality of the channel by		(c) He takes pairs much organ b the second file means the stability of \$22	"Hit 5 provides antana lon sommitting fra provident March 2000	When h Committle Port 21 (2) 103 regimes the radio, the menufication, and pathoget out- obs (3)(1) memory of memory When (5)(Minimitin Port 11) The memory of the out-constraints of standards and approximated.
	 As commenced with particle activesment, the solution due optimization and 		eli The miley pefer unit to demonstration and demonstration distingues (Demonstration) organismo.	1997 9 and 10 may 10 respondly apprairie	Witness Conversion Cart 21- Warm in Minimalin Cart 25- the address of the Cart 25- the address of the Cart 25- the address of the Cart 25- the conversion of the Cart 25- the
	(1) Repartments of the order and the counter for measure calls and valid appropriate to the service principal		In The addition prior counts and any many play is a council of maximum of many formation depend and any operation of a confilmente fordate	We for a second the processory production of the formation account of the constraints.	What is Governing in Part 11 What is Moding in the 121 The offer party of constraints characterized constraints on one of the area constraints on the second constraints (proc.)
12 sinn	Wassering provide (kall) with only frequencies and the constants of the Despension of Africa Development for Optimization	5.21 maters (commissible) and (sufficients)	(c) The constant books and de ne- an considering for analy system to organization is unrelated within the organization is unrelated.		
	Deviction (and Oberra) responsibility, and associated by an instant of the opport over the Device function and considerations of the 331%.		(1) Associated Advancements on the effective of PLDS of the effective of the process of the effective of the process of the effective of the process of the other effective of the process of the process of the other effective of the process of the process of the other effective of the process of the process of the other effective of the process of the process of the other effective of the process of the process of the process of the other effective of the process of the process of the process of the other effective of the process of the process of the process of the other effective of the process of the process of the process of the other effective of the process of the process of the process of the other effective of the process of the process of the process of the other effective of the process of the process of the process of the other effective of the process of the process of the process of the other effective of the process of the process of the process of the other effective of the process of the process of the process of the other effective of the process of the process of the process of the other effective of the process of the process of the process of the other effective of the process of the p	Part 2 opposite in committee and an representation and a com- singutamination and a com- with scattle forms. For and Pi- with scattle forms. To and Pi-	When it Converting Part 21 Print 1: Mining in Part 21 The later print of the second Part 21 T

2) ARC Working Document - Not for Daugh-unit

Syguniti	Part 21- 54 S. Dytailed Gap Assessio	5-Avration Ridemal out Between Part 21.	and Coordians Propaged Part 5, and ICAO	Annes 19	Page F3 of P Recision (Argin Date Oct 10, 201
IF ALS ASSAULT PF PRANEWORK CUMPERINE AT EXAMPLE	SPECIFIC STOLES AN ENAMEWORK DESCRIPTION OF APA	HACTR PAGE 1 MIRAL COMPLETENT	FARTS STRAT	INFIGURACES NETWERN ICAR PASES IN AND INCOMPASES	GAPSINGARNARINT INTERNAL PACENTARY 5 0500 PMPC20
1	By dead of See Jens of server, as an used in the operation of a second order of the server of server in the server of the server of server in the server of the second or interaction of dif- second or an use server. It is possible to use interactions of other of the server interaction of order of applications of the server of the server potential second of a MME.		 (12) All manufactor d'intergenciet in operation autobieux d'intergenciet relegioportente, interfacientes en la procession establica llaco anno el trapportente establica llaco anno el trapportente establica llaco anno el trapportente establica llaco anno el trapportente establica en llaco en llaco establica en llaco establica en llaco establica establica establica della trapporte establica della establica de llaco establica della establica de llaco establica della establica de la della establica della establica de la della establica della establica de la della establica della establica della establica della establica establica establica della establica establica establica establica establica establica establica establica establica establica establica establica establica establica establica establica est	We for an end of the respective compared (respective compared (respective) is a second of the of MdE	When is Conversion Must The (2) TPS for the second control of economic for the PAN for the metry instance When (A Schemmun 2) instance When (A Schemmun 2) Schemmun 2) instance Schemmun 2) instance S
			 Thirdphyse comparations due years and data to any years performent 	Apple 5 marked and the	When it Concerns in Part 11 Prime is Alsoning in Part 11 327-145 data darks of the relation sequentiation for part 341 memory sector and and sector part and any beam sector transitioned in the relation of the relative memory sector is a sector transition.
	(f) (b) specific and source content sports approximation of a specific and a specific and and a single content of a specific and a specifi	25 Designation and Herquindlikes of Required formy Almograms Presented	 Despitations 1.00 per sense of systematic periods of the sense of the systematic periods of the sense of the	bei 2 manungen ander minn frederinge bereinfilde southing	When is Convert to Part 15 <u>Mark 1, Martin Tart 15</u> Tong 1 day and 16 and 16 and 16 and 16 and 16 and 16 and 16 and 16 and 16 a

21 ARC Working Document - Nor for Daugh-tem-

Appointi	Part 11 - 14 C. Detailed Gap Assessme	S. Avration Ridemal out Between Part 21.	ang Committee Proposed Part 5, and ICAO	Annes 19	Page 1 to of Co Reservations (Sergina (Selfe) (Sel 10, 2017
IF ALS ADDRES 19 FRAMEWORK COMPONENTS EXAMPLE	SPECIFIC SPECIES IN ENCOMPAGES DEPENDENT SPECIES	HICTR PAGE 2 (HIOLCOHDERS) FLEMENT	FARTS SPRM REPORTED ATS	INTERPOLET BLIWERS ICAS ATALESTICAND LICENTRAST 7	GAPS CLARTANEAT IN TWEFY FACTOR FAULTS 5500 PART 25
			 (4) Renew Altitude an powerfully, the record programming of the operations and/or inferior to the article of the sector of the sector (10) means and/or inferior to the sector (11) house the net file sector result in sector of the sector of the sector of the operation of the sector of the sector (11) house the net MO is proper in represented and program bolds of the sector of the sector of the sector (11) house the net MO is proper in represented and provide the operation. (11) house the sector provide the operation. (11) house the sector provide the operation. (11) house the sector provide the holds of the sector of the continues holds of the sector of the sector operating of the sector of the sector operating of the sector of the sector holds of the		When is Connected to Direct Di- When is Vision on Kert Di- Fox Di Domo od General Di Wite en and Mill Dave For a Wite destruction production
	 skiller fra første skreatigenen og in dattrifte for som det forme fysterførste er tv ock referetillet 	S21 Sales, Arrandokliky and Antherby	(b) The confident's locale must shorely the Avenue. Training and the bit has motioned to aven a neuron regarding additional and an end the second states.	Port 5 consecution mails and polaris. Apple 10 con- adatas call consecution, parameters and any phone parameters and any phone parameters and	When the Control to Part 21: (2): 17 response the equivalence of descript the left from general and edge one other 17 or 10: (2): 21 (2): 20 (2): 20 (2): 20 (2): (2): 20 (2): 20 (2): 20 (2): 20 (2): (3): 20 (2): 20 (2): 20 (2): 20 (2): (4): 20 (2): 20 (2): 20 (2): 20 (2): (4): 20 (2): 20 (2): 20 (2): 20 (2): (5): 20 (2): 20 (2): 20 (2): 20 (2): 20 (2): (5): 20 (2): 20 (2): 20 (2): 20 (2): 20 (2): 20 (2): (5): 20 (2): 20 (
Ad Apprintment of Key- Suber Dynamid	The certain products and deposition and the operation of the product of the line opply contrasts and the transferrer (of an efficiency ball)	7625 Designation and Research Uttors or Required Statisf Shaning result Performed	(i) Distinct on (if a residuariant opproaching the accounts reasons and desired at instruction and desired at highlight of the second by consec- tions for a more shared to be formed (if the ith result have of the before (if the ith result have of the before (if the ith result have of the before of the before the second second second second of the second second second (if the ith result have of the second (Part 2 hote prophenesistem and mergeneous equivariant data science to environmental executivy. Uncourse of the total science excitate the science of the environment of the science of the prophenesistem excitation (the prophenesistem) and data promotioned and and the science and and	What is forward in Part 21: What is forward in Part 21: For 21 forward in Part 21: For 21 forward in an Ut with subscription and in a Dark assessed for an Ut with a Part assessed forward 10.000

21 GRU Working Disableste - Not for Daughstern

Appointin	Pari 11 - 64 C. Detailed Corp Accession	S.Aviation Ridemal on Holsson Part 21.	any Counciliant Propaged Part'S, and ICAO	Vanes 19	Data 4, 5 of 0 Resident (Angin (7662) Okt 10, 201
IF ALS ANARES 19 TRANSPORT CONDICINE AT A DATE OF	SPECTOC SOUTH AN ENAMPWOOK DESCRIPTION AND ANY S	HIGTR PART 1 MIDI COMDATATI FLAMENT	FART 5 SPIRA BEDY MILLAR ATS	INFIGURALES: INTWEEN ICAG OPPERTIAND INCOMPARENT	GAPSINGUENT INTIVATIVE DATE FAULTS USU PAIRE 21
			(E) Evidence and Comparations, introduced the synthesis burgers of paraticles or comparad on Sumpart to Athenian (10) Regulation and the paratic Regulation and the synthesis of the paraticipants of the State action on paraticipants of the State action on paraticipants.	organities des consentituites et entrancement representantes analitates en module de consentitate de con- mentante de consentitates de consentitates entrancement de consentitates entranc	
J Conductive Conceptury Respond Theorem	Activation Was served a multiple from that and dary new served in the time that the dary is not complete the Co- ponents of the result of the served served is a served served of the served served is a served served of the served served is a served served served served is a served served served served served served served served is a served served served served served served served served is a served served is a served serv	5.21 Coerdination of Franzysky Disestant Ybusenu	We are ensured to common the second s	Part Simily reserves response prioriting the stars investories (Net? The more explicit ensempting compares and the phase of compares and the Vision 19	What is Connect to Part 21- What is Maying (orkert 1)- There is an energy of y force materials the foreign (WM 2)- durations of the other work of the duration of the other sector of the stratefore the record particle
		 The verifie of https://www.intensity.org/ wide/compares reports/provide provide on the orbit consistence of provide on the orbit consistence of reported to assess astry projection of assessing orbit by paragramme of a 22. 	(Mr.) Programme an analysis of registration factors (19 angletics for the effective of research) constitution of research respectic place	When a Common by Dark 21 - What is All-sharing in Part 21 - There are a monthly proper in order to place an energy in property of the Statement's inspect of the stat	
5 515 (Actonomic)	1.9.1 The service provide staff dramp in SVM exploremental plan, foreidly nature; and dram ensuration that ASTech the ensuration that ASTech the ensuration of activities are borner and south the companies of south the ensurements.	TT Application	(b) In contribute to the energy setting of regularization of the attribute 10.4. Advancement for generation of the theory of the setting of the theory of the setting of the regularization of the setting of the quadramenter principle of the setting of the quadramenter principle of the setting of the setting of the setting of the quadramenter principle of the origination of the setting of the origination of the setting of the origination of the setting which constructed of the setting which is constructed of the setting which is constructed of the setting which is	Party notation provide statement plate administration in the statement for 2 modules advected for 2 modules advected for 2 modules advected for 2 modules advected provide statement part of the state	When is Converted in Part 11- Water is Massime in Cert 21- Theories and AUX requestment further in the attent working with the argument of the f

21 GRU Working Doministry - Not for Distribution

Appointi	Part 21-14 S. C. Detailed Gap Account	Sodyration Ridemal out Bobacen Part 21	ant Courdants, and ICAO	Annes 19	Contractions (2010) Researchers (2010) Role: Oct 10, 201			
IF ALL ANNUL PT FRAME PARE COMPANY AT BERMENT	REALTING STOLES IN ENCORPAGING INFORTUNATIONS	HACTR PART 2 WHOLCOMINGEST THEATEST	FARTS STRM BEDFBREMENTS	INFIGURACES INTRODUCES ICAG OTNES IN AND INCOMPASES	GAPSING AUXADIAT AUXALIA GACTA FAULTS AND PART 21			
	 The corp or provider shall device productions (MMP invariants) with a baseline up at a corporation with a baseline up of the corporation of the providence of the first matterarged. SEM (more as a static providence). SEM (more as a static providence). A constitution of a constitution and a constitution and a constitution and a constitution and a constitution and a constitution and a constitution and 	5155015Dopmentation	The completion bookler used database to the continuer SNDs fractionarian architec- derschement architecture fractionarian databased products (0122005) protocours and proceedings	Play 5 dia menderia dan a maliky saga pina famara 10	What is Convert to Dig Th- Sour Acceptories or reaction 26 (17) in or the PAU. When is Minimum for Fault 31: There are SAD responsible directory and other sectors and the discourse of the sectors price. TeXP (19) in our sectors of the sectors of the sec- tors are set of proceeding.			
		2.07 Henry	(a) the contribution and a near magnitude measure on any price of indepty and matter in mark processor in the optimal in Subject C of the part back mountly using the column of the address of the comparison processor processor of comparison.	Third of the specific mand- mation approximate the limit departs for some th	When to Conneck to Part 21: Scenar control are singled for \$29 (19)(er) to the PAD and so when places that dracket controls as \$105			
					_	the the synthesis independent products overlap of oppole of which another produces an electrical in Separat 11 of 01 years Such works and a set		When is Missing in Part 25 There is no SES responses, department SES responses, such the SES to reach a drive event of communications.
		T AT Bounds	 (c) The contribution between environmentary is recented and individual of produce and and individual recenter provide performance of pro- traction provide a plannat from mentional (22) and and the pro- tractional of the complement of the individual of the complement of the mention of plannate and sound recent of plannate and sound and recent of plannate and recent of plannate provide a landor (27) to an economous of 26 recent pro-subject and sound and of 26 recent plannate and and and and and and and and and and and and and and and and and and	Hard Analogo, Hardweid Annellin wahn wordt Har In an Appentin yn wy H	Bank in Community Part 21 Simil records and explored in Simil records and explored in Similar that the Antonia Similar observation of the Antonia Similar When in Statements Part 21: Party ones Statements United Sec. 11 All International Society of SMM community and records of SMM community of SMM sec. 11 Statements			
	1.5.1 The service provide shaft devides and tasks are 1999 register particular to 400 desentationes;	2 Arren Brennigerie	The orallities builds near device and nations 2505 between device the smartled backgrift no beauty of an Shutte policy (0) SMS process and non-daps.	1/et 5 allows multi- per environment of the following the environment provide: a grant by commu- nation instance	What is Converting Part 12: Super Astronom Date is associated in 24: (17) and the part of the part of What is Musica to Fart 12: There is Musica to Fart 12: There is a 10 to proportional distribution of 10 to proportional in discount case is parts. Solid parts and the superference			

"LARE Working Document - Not for Despiration

Appointin	Part 11 - 5A C. Detailed Gap Assessme	Ps-Avration Ridemako mit Bobacon Part 21.	nu Coordialte Propased Part 5, and 10 Mi	Vinney 19	Page L 9 of Co Recision (Anglos Date Ukt 10, 201
IF ALS ADARD 19 FILADIE WARE CONDENCAT SUBMICIT	SPECTOC STOLS IN PRODUCTOR DEDUCTORATION	HICTR PART 9 (HOLCOMDAY 51) TLAMEST	FART 5 STRIM BEDOLUENENTS	INTERNALSA INTWEEN ICAG ATALSTICAND INCOMPARTS	GAPSING GONDAT IN TWEET A GAPTS OND PAIRE 21
7. Sales-Risk Managements		sala Pik Muniputat			
3.1 (here's fideritiked be-	21.1. The arrange provide shall develop and realizant A resum- but commendation and a sector system of the solution products on a resumption of estimate.	3.43 System, Londyst, and Honorit Benefilication	(a) When optimity setter via manipures, her uniform invariant and there is a setting in the setting output on a plant in the setting paragraph (d) in (10-incluse, and developing action) and optimizer multi- setting (d) in the setting of the setting output of the setting of the set is the setting of the setting of the setting of the setting of the set is the setting of the setting of the setting of the set is the se	Verif 12 Systematic many Net requiring any the based three many controls are monomial. Some 100 test systematic from the systematic many effective 2.1 and with momomphism 2.2.	When in Lowered in Pari 11 Protomo the American Terms regulation of the strength form (Larger American Strength) regulation of the strength form (Larger American Strength) The New York (Larger American Strength) The N

2) ARC Working Document - Not for Daugh turn

II AD ANARD IT FILAND PROBLEM CREATER AND EXTERNAL AND EXTERNAL	SPECIFIC ANDREAMS ENCODEMONOR DEPENDENT/ATS	HACTR PART 1 WHOLCOMD/9757/ FLAMENT	PARTS SPIN BUILD BUILD VIS	INTERNALES INTWEEN ICAN AMERIKANI INTRIPARTA	GADSSAUDALAUAL MUTWERY FACTS FAUY 5 050 P50220
			 (b) to recence/b) (after system surfaces the influency intermediates or only for executive). (1) I have one to improve relation executive. (2) Maccontrols Comparison Automation. (3) Maccollabor of the system is provided and the system. (3) Maccollabor of the system. (4) Concernson and provides and for the system. (5) Concernson and provides and the system. 	4. This census all first of specific all more particular for all more particular therefore a community of the specific of t	When is Concern to Mar Li- Tensor, Corp. Solar Tensor, Corp. Solar Tensor, Corp. Solar Tensor, Corp. Solar Tensor, Corp. Solar Tensor, Solar Solar Construction and a solar contension for an approximation of a solar tensor and approximation contension and approximation contension and the document contension and the document contension and the document contension and the document description, with tended and contension and the document description, with tended and contension and the document description and the document description and the document description and the approximation of the market of Marine Life 2017 (1997). When the Marine Life 2017 (1997) Market of Marine Life 2017 (1997) Market of Marine Life 2017 (1997) Tensor and contensistic and the document of the operation of the document of the compared and the document of the document of the operation of the document of the document of the operation of the document of the document of the operation of the document.
	1.1.2 Harris Open Gauton digit to Band in 2 yanotheside of matters parally and problem multiplied adaption and or half the		ter The symbol and her day word in each op- main market by the day of the day of the function of the day of the day of the ended position.	1998 2 Di rener garini di Ale Rene di Uradimilia Officio D'Antoni D'Antoni Dari di Antoni Directi a di Bri- bari di Antoni Directi antoni aggi mari Antoni Directi Antoni Presiditta di Balanda di Pre-	When informer the Part 21 Tree, we for conversions: we converse the conversions: we converse the conversions: all them only we descend only all them only we descend only all them only we descend only the only of the second only the only of the second only the encounter of the second only the conversion only and the second only the only of the second only the second only of t

2) ARC Working Document - Not for Double-term

Appointin	Part 11 - 58 C. Dytailed Gap Assessme	Bodyration Ridemal and Bobacen Part 21	any Computition Proposed Part 5, and ICAO	Annes 19	Poplar 1, 11 of 1, 1 Revealmen (Seigler, Onlar (Sei 10, 201
IF ALS ADDRES IN FRAMEWORK CONDENTATION	STRATISC STOLES IN PRODUCTION OF STRATES	HICTR PART 2 MICLEOMBAST 51 FLAMENT	FARTS STON BUILDERNENDS	BETTERNALDA- BETWEEN ICAGAMANEXTRAND INCOMPARENT	GAPSINGARTANIAT IR TWEET FARTS 350 PART21
					 Meleccia: P. 311 (Characon 4) (11) P. 321 (Characon 4) (11) P. 311 (Characon 4) (11) P. 311 (Characon 4) (11) P. 312 (Characon 4) (11) P. 313 (Characon 4) (11) P. 313 (Characon 4) (11) P. 314 (11)
12 Sink Insentation	The weaks provider shell denotes and account appears of a second with an anomalian of a discount to a subpress on a stand space omatified leaves	3.93 Application	 A contribute deservation operational deservation (e) The control operations of a control operation (e) The control operations (e) The control operation (f) deservation (f) control operation (f) deservation (f) deservation (f) control operation (f) deservation (f) deservation (f) control (f) (f) (f) (f) deservation (f) (f) (f) (f) (f) (f) deservation (f)	Juni 2-machinema postation-posto visiti and transcompression must transpirited	Where is from the Part 21) Barry Contrast of Part 21) Internet Contrast of the State of the regeneration IS proceedings with the and year of proceedings with the and year of proceedings of the and year of proceedings of the and year of the proceedings of the and year of the year of the and year of the year of the and year of the and year the proceedings of the and year of the year of the and year of the and year of the the year of the and year of the and year of the the year of the and the year of the and year of the year of the and year of the and year of the and year of the of the the year of the and year of the and year of the and year of the of the the year of the and year of the and year of the and year of the of the the year of the and year of the

Appointis B ADAAMES IT PRANEWORK COMPANY (AT)	C. Dytailed Gap Avenue sesator sectors in Enconvoloc	Incre part 2 Incre part 2 Incre part 2	Proposed Part 5, and ICAO	MINEARAEZA BITWEEN ICNO ANNEX IN AND	Gales dationst
					0, Commune with the network and the service the Body and the and the service the Body and the Additional terms of the Body and State (add) and the service of the Body and State and the service of the Body and State (add) and the service of the State (add) and the service of the State (add) and the service of the State (add) (add) and the service of the State (add) (add) and the service of the State (add) (add) and the service of the service of the (add) and the service of the service of the service of the (add) and the service of the service of the service of the (add) and the service of the s
		ESS heating flash Associated and Executed	Int The configure behavior and density and eventual processes to analyze analyze with an excitation of the sectors of the file sectors and other sectors of the file sectors are proceeding and processes to complexity and intermination of the other sectors intermination of the other sectors intermination of an excitation of the intermination of an excitation of the intermination of an excitation of the intermination of the other sectors of the intermination of the other sectors of biographic actions.	(1) M optim the channel 199 basis mapping and a particular particular particular and a transition of the particular particular methods and a support of the particular and (press), particular and official and additional and particular and differences particular and the support particular and the support particular particular is a particular of the particular particular of the particular and the particular and the support of the particular and the particular and the support of the particular and the particular and the support of the particular and the particular and the particular and the support of the particular and the particular and the particular and the support of the particular and the particular and the particular and the support of the particular and the particular and the particular and the particular and the particular and the support of the particular and the pa	What is Drowned in Part 10: What is Monitor in Part 11: There are trapping processing address of the processing Minute Drowned in Part 11: Instance in Part 11: Instance in Part 11: Instance in Part 11: What is Minute Drowned in What is Minute Drowned in Minute Drowned Information in the Part 11: Minute Drowned Information in the Part 11: Minute Drowned Information Minute Drowned In

"TARC Working Dominant - Toy for Doministion

Appointi	Part 11 - 14 6 C. Detailed Gap Assessme	So Avriation Bademid out Between Part 21	and Committee Proposed Part 5, and ICAO	Annes 19	Tonja E. F. Strof E. F. Novikimi, ? Nigina Data: Okt 10, 201
IF ALS ADDRES FY FILAMED VIAL CYSTOPOLIAT STRAND	SPECIFIC SPECIAL ENCOMPAGES DEPENDENCES	HACTR PAGE 2 SPIOLCOMINSEST	FART S SPRIM	INFOEMACEA INTERNALISA ICAG APPERTAND INCOMPARENT	GAPS INTRADAT IN TWITS DATE FAULTS ISSUE WRE 25
			(a) The preferrance between each discussion and there is presented with the control of the transmission presented with the control of the transmission presented with the control of the control of the control of the monotonic of the control of the monotonic of the control of the comparation of the theorem of the control of the control of the comparation of the theorem of the control of the control of the monotonic of the control of the monotonic of the control of t	(ii) "Tray (C) and takin an and distance in the provide sequence of the seq	When in Concern to Diret Liver When in Montrag in Mort Liver (1) provide the nodes a part of Minerson (2) processes to develop rade service in the nodes and the nodes of the node factors (2) CO255 (1) Mancreasted datance (p2)(2) (7) Nodeshift interface (
2. salett American		Interior According			
1) Surey, Development Mankering and Stransportat	11.1. The service provides shall devide product address the conset for system that address provements on the system that address provides of the system of the conset of the system of the system of the system of the system of the system of the system of the setting of the system of the setting of the system of the system of the setting of the system of the system of the setting of the system of the syst	Art a soft to Newfortunates, Also also esting and Also were reserved.	(m) The synthesis books must do of pro- sent standard possession of a scheme is accurate due to the response in the synthesis post-part, and environment operations, post-part, and environment operations and problem provides on operations and problem provides a sentence (north on-data), post-part operations and problem in the provides of the problem in the problem operation.	Dal A composigned (0) administration append Ramado (0) de variantes (0)	Man to Conserve in Part 21 Franktine 33. (107) meaning proceeding to implementing conserve and potential and the inService (2). (17) min (2). (3). (2) A 32.1 (3). (07) min (2). (3). (2) A 32.1 (3). (07) min (2). (3). (2) protocol (2). (3). (3). protocol (2). (3). (3). protocol (2). (3). (3). Man (3). Mining in Part 21.1 Man (3). (3). (3). (3). (3). (4). (5). (4). (5). (4). (5).
			111 Valuenas perping of approximation	1	Wheels Connect in Pire 20 Trackation 224-07 Susperso massifications (Winess Mittel 26-50 No. 271 (Upper approx.) providence for unserved), 271 performance for unserved), 271 performance for uniformized and a formation and formation performance of a statistication performance of a statistication performance of a stati

2) ARC Working Document - Not for Double-term

Part 11 - Mps.Aytanon Ridemaking Fourtuin Appointis C. Detailed Corp Assessment Between Part 21, Proposed Part 5, and ICAO Annex 19					Cons. Friday (1) Bornaine (2) Cons. (201) Cons. (201)	
IF ALS ADDRES IN PRAMEWORK CUMPONE AT EXTEMPONE AT	SPECTOC SPATS IN PRAMOWING DEPENDENCE SPATS	HACTR PART 1 (HICLCONTEXTS) FLOATENT	FARTS STORM UTIONNEMENTS	INFORMATICS INTERNATION ICAN PARTY AND INCOMPARTY	GAPS (GADNARN) IN TWEET (ACT) FAULTS (STUP) IN 121	
					What is Missing in Part 11. Discussed spectra 20 counts resolving a finance spectra 20 counts with the assumption of number and the second count of number and 20 finals 1 Provide all the second counts with the second count of the second of the second count of the second of the second counts of the second with the second counts of the second counts with the second counts of the second counts of the second counts of the second counts of the second counts of the second count of the second counts of	
			 III - Methodia, prevalenza per del Das entreralmente entremandati finalment plangas 		When the transmitting Rate Tri- Transistions (12) (12) addresses (2) Transmitting (12) addresses (2) constructions and Constant with complementary (2) and a substantial complementary Writes (12) (12) (12) (12) (12) (12) (12) (12) (12) (12) (12) (12) (12) (12)	
			(3) during () successful provides addressers.		When its Connecting Part 21: Transforming (3): 31(2) to apprecise online, the de- operation of the bar on one of the operation of the bar one of the de- doment of the bar of the transforming When the Assessment of the bar of the Part of the transforming of the transfor- ing the transforming of the transfor- ed the transforming of the transforming of the operation.	
		(- 10 Extension or the Millard synthetic type evolved (many)		When to Convert to Part 21 When to Avanuar in Part 20 Construction of the WW and Construction Transmission and Avanta	
			. C Dittor and transford to the		When to Connect to Phys 21 Incline and 2017 Study repression material of the Study and Physics of the Study of the Study of the Study Study of the Study of the Study of the Study Study of the Study of the Study of the Study Study of the Study of the Study of the Study Study of the Study of the Study of the Study Study of the Study of the Study of the Study Study of the Study of the Study of the Study Study of the Study of the Study of the Study Study of the Study of the Study of the Study Study of the Study of the Study of the Study Study of the Study of the Study of the Study Study of the Study of the Study of the Study Study of the Study of the Study of the Study Study of the Study of the Study of the Study Study of the Study of the Study of the Study Study of the Study of the Study of the Study of the Study Study of the Study of the Study of the Study of the Study Study of the Study of the Study of the Study of the Study Study of the Study	

2) ARC Working Discussion - Provint Daugh-tion

Appoint	Pari 21 - sAr s C. Dviailed Casp Assessme	S. Avranon Ridemid on Bobicon Part 21	any Computition Proposed Part 5, and ICAO	Annes 19	Date J. 1 Car L 1 Normins (Might Date Oct. 10, 201
II ALS ANARD PT PRANE VORE CONDENT AT SUBARD	SERVICE CONTRACTOR ENCOMPAGINE INTERNETION	HATTR PAGE 9 SPICIL COMPLETENT	FARTS STRAT	INFOEMANCES- DETWEEN ICAUS APPECTN AND LICENTRASES	GAPSINIAGNADAT IRTWOLA DATE FAULTS 350 PART21
		_			 Shendd Fac on an Externation Treast anotheral or incident ant utiligation.
			and the second secon		Whee Is Constant in Part 11: Traduction 22: (17) (Congruent prosperse II) and an of (part) prompt Which is Observed regimes affectively records the same regimes affectively records of the set of a parts are much to the set of the s
			 a contribution entropy of controls or part to the theorem property of a majority to station and provide the fact to goal interact incomes controls on the part of the property of the control of the control of the term of the control of the control of the term of the control of the control of the control of the term of the control of the control of the control of the control of the term of the control of the	The Treffson de Annue (Primarkes VIII) conducts solution yet ingeloads partial the arguments a SNR mayanet.	When is America in Part IV When is Mission in Part IV Draw rest request as completion requesting typical
			(b) (The spectral pair boddes must does not seen exercises processing) from earlier the data warraw it must be in processing and a second must be an experimental and a second must be analygoing spectra data of the must due on second second must be and a second any university and data with must due on second second must be and associated.	The (50 massing one of the second of the sec	What is Convert in Part 21- Translation (2): 170 and proceedings of the data and even (2): 170 and proceedings of the processing of the processing of the pro- ceeding of the processing of the pro- ceeding of the processing of the pro-
11 Subject Production Measurement Measurement	11.7. The set trace provides to entry provide the second set of the second set of the second set of the second set (set), a protocol set of the set of the set of protocol set of the set of the set of second set of the set of the set of the set of second set of the set of the set of the set of second set of the set of the set of the set of the set of second set of the set of the set of the set of the set of the set of t	Aria salaty Peritinaan (Anasonant	Fig.176 certificate horizon de certifica- sano a casa estado e de certa de productamen- anya esta esta esta esta esta esta esta est	(byt)) (for an - to tonk for Benzinentificher oder s	When it is more in Part SP When it is more the Part SP Deep user appear is more than of characteristic activities (f) characteristic activities (f) activities (f)

IF ALS ASSAULT PT PRANEWORK CUMPENDAY BURNENT	SPEATER: SPEATS AN ENCODEMONS INTERNATIONS	HACTR PAGE 9 (HEALCOATERSTS 3) FLEATERST	NABLE STORM	INFOCUMENTS INFOCUMENTS	GAPSINGSDAMAT IN TWOTA DATE FAULTS 35D PARTS
			(1) Approximation of a product and a probability of the product of a product of the product o	Mart Indexage 2	When is Concern to Durt Di- Teri 21 and the second
			12) The loads the performance of the SNIS	In 2016/2019 - option Variables option (16 Section 12) that listed of Section (12) - Stappy of 2 - 41000- improvements.	What is Converting Part Dy What is Converting Transfer Dy Transfer of space as TVES or constants of the space transfer (CCD) 1. May be descent yes a solution constant before the space transfer of the strandplatter used. They in the strandplatter used They in the strandplatter
			(7) Produces and dimension of the adaptive production of a product of under \$255 for an adaptive pro- method for a product of methods.	¹⁰⁷ 2014 (Sharaqonaletti L urazy (P.11.2 cm.e4 (m be opport of §1.25)c) that performed Anne 19	When is Frequent to Part 11: Weat is Monitor for Vart 13: Provide 1 (Marco for Vart 13) Provide 10 (Marco for example 2017) Dipartity and the formation of the provided
			(1) Alamita aragas a tas mantana aramanan fita any mandanana ing pada	The Trincly events areas 10 Appendix Mapley 2 1	When is Committee Part The Production (2) 117 addresses Connect to the south's restore the set character the spectrum of the set character (b) and the set of the set of the When (1) Admittee (in Part 7).

21 ARC Working Disauteaut - Nor for Distribution

Appointi	Part 11 - 54 s C. Detailed Gap Ascenne	Solvianon Ridemal on Between Part 21	Ante Loon matter , Proposed Part 5, and 30 Att	Vanex 19	Revision (Arigina Doue: 134, 10, 2017
IF ALS ADARD 19 FRAMEWORK CONDONLAT KERMINT	STRATIC STRATE IN PRODUCTION OF A	HICTR PART 7 MHOLCOMINSTAT	FARTS STRAL UNIO MERICATS	BETWEEN BETWEEN ICAG OTNEX IS AND DOTTI PARTY	GAPS CLAUPANEST IN TWEET LACTS FAULTS USD Public 21
		-	1000		Present all in the apprend <u>240125</u> 1. Consider overset Manufactor SPR structure
			(i) shall permit instrume transfer edit and order of a frame.	MCMDer (Marchenen e. Bergehausz, K. (2002). J. (2002). J. (2002). Automotive register and Amount View Investment (J.).	When is Consul in First 31 Prior 99 Addresses the order to solve 40 10.5 or a brand 10.5 or 10.0 mm states of a series of the consult of the order of the order 10.5 mm states of the order of the When is the Manage in Prior 21: These next of these first requestions 2000 at the order of the order of the series of the order of the order of the order of the order of the order of the
			(11) proc. 2018 phases of the association of the Direct exceeding wave bigmeds, an evolution between one demandial tender parameters (wall22) direct phases of the excitory disc association of the evolution of the apply which phase phases of the apply direct body at Tabipert C of the apply.	1201 yanazile da situa polytek bastur polytek bastur daparti bastu	When is Non-year in Fast 31: When is Moreover in Fast 31: They of these off-mapping distant off-scale particle despination. But also particle despination for the engineering Al (223) 1. Instead from the engineering off-spination constants (b) al. (1) When with the engineering mapping of the engineering of the engineering.
12 The Management of Phongs	He are to provide datil device our relation errors is detail in a constraint of the second and the second data with a second and the second data data with a performer relation of the second and the second data as and the the the data as	551 Spylinskilty	5 serificate locale or strong oppin oddy that traditioners is a transmission of (characteristic paralities as (characteristic paralities as (characteristic comparison) (characteristic comparison) (characteristic comparison) procession (characteristic comparison) procession (characteristic comparison) (characteristic comparison) real control form of tradition (characteristic real controls form of tradition) assumption (characteristic control of tradi- tic controls for a control of tradition) (characteristic control of tradition) (characteristic control of tradition) (characteristic control of tradition) (characteristic control of tradition))	The 3 might is specific event or table which is a specific address of the second or specific be consistent.	Mediation of the Pariation (Mediation of the States 21, 127) (a) (Prophers are actual the the IM data and yes applied states (b) Proceeding for a state that of the second decrement of the other second decrement
		5.13 holi ly Portorman's Aleximanii	Co. The orbifs all involve your coulding and the profit October - profit Dataset income is a starty of pacifier - 1-40 of MAN Annual Institution and MI 20 (2010).	Ner 573,00100 minutes	When is Conversion Part 21: When I. Mining in Part 21:

2) ARC Working Documents - New Inc Doctal Street

Part 71 5M5-Avration Ridemaking Countilline Appointin C. Detailed Gap Assessment Between Part 21, Proposed Part 5, and ICAO Annes, 19					Revision (Mighaa Date: UKI 10, 2017	
IF ALS ANALS IN PRAME PARA CUMPER (A) SE MARKET	BESTEDE SESTES IN ENCODIMUNE DEPENDENTION	HICTR PART 7 HHOLCOHDSTST FLEATEST	FARTS STRM BEDT DESERVIS	BETTERINGEN BETWEEN ICAG ANNEX DE AND DECHEPAREN	GAPS CLICKNERT IN TWEEVER FAULTS USU POPEZI	
100 m			(a) Mercili charge-milia reconcellarivementa darmas alexadar confusción			
L4 Continues Superconnel (CR) 535	The server of model and function and assess the Aldersverse of the MM (processing to and/or anomalow high serverse of the Local High serverse of the Local Institutement of the NYS.	6.55 unlineans	We contribute a backet and and tables and implement on the processing of the second model of the second processing of the second model field of the proceeding to condition with the gO 75.	Port of regions, come tank of other space and other performance and others the register control of the state register control of the state register control of the state of the state register control of the state of the state of the state register control of the state of the sta	Point in Concerning Control of Part 21 Production 22: 107501 interprets forward on and processition activativity product concerning and product systems weakers (2): 07 and/amount product (a-o) respirated to any or Manana Manharatin Part 21: Descent to any or Manana Manharatin Part 21: Descent to any or interpret forward on a defense interpret production of a state on the angle production of a state on the an	
A mility fromium	and the second second	Subpirt 1 -	and the second s	and the second se	And and a second se	
LI Industria	4.1.1 The series provide shall develop and maintent yearbay many, expression to be based only provide or Toront and protection (1) provided and protection (1) provided and once	5.11(response on and Frankey	The confliction field and product transmission and confirmation to product a 0.23 memory with only of the minimal constance for conflictions on times to perform working and with 0.152 (perform and promoving of the 57,05	"Plift 5 or aquisalate to Analas 10	Wheele Connect (a Part 20 (2) 142 periode despectives of the full scheme generated proceeding over an end of the full of an effort of the full year and the full of the full year and	
	413 The surgest the offer training programme doll to supervise in each after bland second arrived in the Solls.		1		What is Money in Part 21 These is no 50% performance, Used and an official segme enset for training of 55% performance.	
42 saları Loomentation	The service neuroble shall denote provide membrane in Service research for addi- correlation statement of the models correlation processing that wave of the SMS to add party of memory and the MMS to add party of the service work of the service statement of the memory and the method party memory work of the service statement of add party of the service statement of add party of the service statement and the service statement of the memory of the service statement of add party of the service statement and the service statement of the service statement of the service statement of the service statement of the service statement of the service statement of the service statement of the service statement of the service statement of the service statement of the service statement of the service statement of the service	Serianit	(c) The certificate intellection of endowing and anticelar many hit optimization. They for end at the optimization of the formation of the second of the second of the many of the second of the se	Part 3 (agains 40) percent times (seen 34) table adapting all of the all tables are strateging as 60) more position	When is Drowers in Part 21s United replacing and 21 (17) (a) Samilar costs of Part 21s (37) Inc. Inc. (31) (17) (37) Inc. Inc. (31) (17) (37) Inc. (31) (17) (17) (37) (17)	

2) ARC Working Document - Not for Daugh sum

ł

1/2073	21 (SMS-Avinton) Bulemaking Committee
Appendix H. I	Description of How Compliance with Aleworthings's
Signida	uls Meets SRM Requirements During Design

Paget

Daie,

Appendix D. Safety Risk Management (SRM) and Safety Assurance (SA) in Certification: Description of How Compliance with Airworthiness Standards Meets SRM and SA Requirements **During Design**

Concept Hypothesis

The certification processes and business practices used to show compliance can meet the intent of Part 5 SRM.

SRM Application (refer to Figure D-1)

SRM is applied by the regulator in developing regulations (above the horizontal dashed line in Figure D-1) SRM is applied by the design organization to comply with the regulations (belowthe dashed line)



Figure D-1: SRM Process Linkage Between Regulation Development and Design Implementation

Part 11 (SAIS-Aviation Bulemalong Committee	Page:	D.2 of H.2
Appender II, Description of How Compligner - and Aleworthiness	Resiston	Original
SPAI Provident States SPAI Provident States SPAI	Date	Gaj 10, 2011
Standards Meets SKM kequirements mutha Design		

A design transforms safety requirements into risk controls for a product or article. A safety requirement, in the form of an airworthiness regulation is a safety risk control that, when complied with, constitutes an acceptable level of safety risk.

Airworthiness regulations are developed when systemic hanards are discovered and the related uncome(s) have unacceptable risk. Acceptable level of risk is determined as part of the rulemaking process and summarized in the "Costs and Benefits" section of each new regulation. The acceptable level of safety can change over time, e.g., §25 561 per Amendment 25-64 versus §25 562 Amendment 25-64. Each airworthiness regulation provides boundary conditions for a design that, when met, are expected to manage risk to acceptable levels. These boundary conditions range from being prescriptive to being purely performance based. The regulations typically do not specify a detailed design solution. The design organization will design risk controls to comply with the regulations. Validating that the risk controls fulfill the safety requirements is accomplished during the certification process.

The preamble to each regulation contains a discussion of the bazard that has been identified. Many of the bazards that are being controlled via a regulation were discovered following an accident and the subsequent accident investigation. Most bazards, through combinations of other conditions/events of certain probability, can result in multiple outcomes. These outcomes will likely have varying degrees of tisk due to different severity and likelihood of occurrence.

The actual regulation itself describes criteria which a design must satisfy. These criteria can be considered boundary conditions for risk controls that a design organization includes in the design of their product or article. The resulting design solution is validated against the regulatory requirements. Often the validation is supported by agreed-upon methodology and performance thresholds as contained in associated ACs.

Holders of production approvals or type certificates are subject to regulatory requirements during the Continued Airworthiness life cycle phase. These include requirements for reporting a variety of events and conditions such as failures, malfunctions, and defects (14 CFR 21.3, 21 137(m)), and for sharing certain data with the FAA and operators (14 CFR 21.99). These activities necessarily include the creation and exercise of processes and systems by the approval holders to monitor both production of a product or article and its operation in service, collect information and data, analyze the information and data, and communicate the information and date to the FAA. The FAA uses the information and data supplied by the approval holders and other source to develop airworthiness regulations as described above.

Following are examples of how SRM is satisfied through the activities of certification and design approval. Example 1 describes how SRM was used in the development of regulations, in this example a prescriptive regulation, which in turn are used by the design organizations as tisk controls without necessity to re-create or repeat SRM. Example 2 explains how specific regulatory design requirements, combined with a requirement that the design organization and yze for unspecified other additional risks, constitutes a "safety analysis regulation" and accomplishes the intent of Part 5 SRM requirements. Examples 3 and 4 illustrate how SRM is accomplished when a design organization devises and incorporates dorign risk controls to comply with regulations, leading to a certified product. Example 5 shows how compliance with the regulations by a design organization accomplishes SRM for those design aspects that are not specifically covered by the regulations.

Part 11 / SMIS-A viation Bulemaking Committee	Page	13:3 of 13.3
Appendix II. Description of How Compligneer with Airson drines.	Tiole	Original Circl 10, 701
Standards Ments SRM Requirements During Design	dennes.	1001 (NO 2011

Based on these examples, the SMS WG has concluded that the certification processes and arryorthiness standards leading to certification of a safe and compliant product or article, accomplish the intent of SRM. With the introduction of Pari 5, hittine malemaking should be based on SRM transparently applied by the FAA's processes.

Example 1: 14 CFR 23.607, 25.607, 27.607, 29.607 Fasteners

This example describes (above the horizontal dashed line in Figure D-1) how SRM was used in the development of the regulations, in this example a prescriptive regulation, which in turn are used by the design organizations as risk controls without necessity to re-create or repeat SRM.

The development of 14 CFR 23:007, 25:607, 27:607, and 29:007 is a prime example of Safery Risk Management and Safety Assurance that has already been accounted for in the design certification process by way of airworthiness regulations. With regards to fastenets, in the past (prior to Amendments 23-48, 25-23, 27-4 and 29-5) It had been common practice in the manufacture of aircraft to secure fasteners (i.e., bolts, screws, and plns) with a single locking device. While this had proven adequate for most arcraft, there had been a number of instances of loss of fastener integrity involving fasteners installed on rotorcraft and secured with a single locking device. This service experience represents the §5 71 safety performance monitoring, i.e., acquiring data with respect to the safety of the product or article. This adverse service experience was attributed in large part to the fact that fasteners used on rotorcraft are subject to greater than normal vibration. In addition, the FAA was aware that the locking devices can be adversely affected by the environmental conditions existing at the particular installation. The FAA related this service experience to fasteners whose loss could jeopardize the safe operation of the aircraft This thought process is equivalent to performing the system analysis and hazard identification regulted by §5 53, i.e., identifying the function and purpose of the system, the system's operating environment, and identifying the hazards within the context of the system analysis.

In view of the adverse service experience with fasteners and the hazards identified, the FAA proposed and subsequently adopted regulations (\$§23.607, 25.607, 27.607, and 29.607) requiring two separate looking devices on all removable fasteners in any installation in which the loss of the fastener could jeopardize the safe operation of the aircraft. Additionally, the FAA adopted, to the regulations, requirements that consideration be given to the environmental conditions associated with a particular fastener in determining the appropriate looking device for that fastener. The adoption of airwortbiness regulations to address the identified hazard is equivalent to §5.55 safety risk assessment and control, i.e., an unacceptable risk was identified and a risk control put in place to maintain an acceptable safety risk.

The design certification process accounts for §5.51 applicability, i.e., if a new system (product or article) or a change to the system (product or article) is being introduced or a change in the environment conditions, e.g., higher rotor speeds, then a reassessment of the bounds would take place and as appropriate, based on system analysis and bazard identification, additional compliance to §23.607, 25.607, 27.607 or 29.607 would be conducted

Compliance with §§ 73 safety performance assessment would continue to be accomplished as part of the Continued Operational Safety (COS) programs

Part 11 (SMS-Aviation Bulemalong Committee	Page	D.4 of 19.4
Appendix II. Description of How Compliance with Meworthings.	Resision	Original
Standards Meets SRM Repotrements Uprint Design	Cintes	VICE 107/2011

Example 2: 14 CFR 33.75 Safety Analysis

This example explains how specific regulatory design requirements, combined with a requirement that the design organization analyze for unspecified other additional risks, constitutes a "safety analysis regulation" and accomplishes the intent of Part 5 SRM requirements

The engine certification process requires that the new design demonstrate compliance to a large number of regulations (14 CFR 33), each one of which represents a hazard or group of bazards identified by the FAA, and an acceptable mitigation strategy which will reduce risk to an acceptable level. In addition to these specific hazards, \$33.75 requires that the applicant analyze the proposed design for additional risks, if an engine failure should occur, and demonstrate that these risks are controlled to an acceptable level. In doing so, the intent of SRM is accomplished

Following is a comparison of Part 5, Subpart C – Safety Risk Management, to existing 14 CFR 33 75 showing the correlation of §33 75 requirements to Part 5. Comments about the correlation appear in blue, italicized font.

ELEMENT	CFH PART 5 REQUIREMENTS	14 DFR PART 33.75 REQUIREMENTS
5,53 System analysis and hazard identification	(a) When applying safety risk management, the cartificate holder must have a process to describe anti- analyze the system for use in identifying bazarde under paragraph (c) of this section, and developing and into-ementing itsk controls related to the system under \$5.55(n).	(a) (1) The applicant must areavize the engine including the control system to assess the rikely unmanumices of all tailures that can reasonably be expected to occur.
	 (b) In numburing the system analysis the following information must we can identify the formation must we can identify the system of the system. (1) Function and purpose of the system. (2) The system's operating invitoriment. (3) An outline of the system's processes and procedure. (4) The performation equipment, and facilities necessary for operation of the system. 	 (ii) If the addety analysis depends on one or more of the following items, these items must be identified in the analysis and appropriately substantiated. (1) Meintenance addens being canned out at stated intervals. (2) Verification of the satisfactory trunctioning of safety or other devices at pre-flight or other stated pends. (3) The provisions of specific in intervals. (3) The provisions of specific in intervals. (4) Flight crew actions to be specified in the operating instructions intabilities inductions in the operating instructions in the operating instructions in advised in the operating instructions must be induced and a statem. (4) Flight crew actions to be specified in the operating instructions introduced in intervals. (5) Flight on the safety analyse must also move, but not be limited to, investigation of the following. (7) Indicating equipment. (8) Compressor bleed systems. (4) Refigerant injection systems. (5) Eas temperature control systems.

11 ARC Working Decument Not for Dispitution

Part 21 (SAUS-Aviation Rulemaking Committee Appendix II. Description of How Compliance with Alexarchiness Standards Morte SRM Requirements During Design

Page: D.5 of D.5 Revision Original Date: Oct 10, 2011

ELEMENT	UFR PART & REQUIREMENTS	14 DER PART 33.75 REQUIREMENTS
		 (5) Engine speed, power, or thrust governors and fuel control systems. (7) Engine overspeed, overlamperature or topping limiters. (8) Proposer control systems; and (9) Engine or propeller thrust reversal systems. The requiration states the system boundaries (it be included in this analysis).
	(c) The certificate holder must develop and maintain processes to iterrity hazards within the context of the system analysis	 (g) Unless otherwise approved by the FAA and stated in the safety analysis. For compliance with part 33, the following failure definitions apply to the engine. (1) An engine failure in which the only consequence is partial or complete toss of brust or power (and associated engine services) from the engine will be regarded as a minor engine effect. (2) The following effects will be regarded as a minor engine effect. (2) The following effects will be regarded as a startious engine effects. (i) Non-containment of mgh-energy debris. (ii) Concentration of toxic products in the engine bled an interdet for the cobin sufficient to incospicate crew or passengers. (iii) Significant thosal in the opposite direction to this commanded by the pilot; (iv) Uncontrolled frie. (vi) Failure of the engine incount system leading to indevenent engine separation. (vii) Rerease of the propeller by the engine. (a) Rome second of the short her engine engine. A pilotable, and (vii) Complete insability to shull the engine (g)(1) and (g)(2) of this anction will be regarded as a major engine effect.
5,51 Applicability	A certificate holder must apply safety mix management to a system under any of the following conditions: (a) Implementation of new systems	Addressed and their sevenities and effects to a addressed and their sevenities. Showing completives with CER33 75 is required for each new angine design, and for designs Operational procedures.
	 (b) Revision of existing systems. (c) Development of operational procedures. 	(a) (1) The applicant must analyze the engine. Including the control system to assess the way consequences of all failures that can reasonably be

Part 21 (SMS-Aviation Rulemaking Committee Appendix II, Description of How Compliance with Airworthiness Standards Meets SRM Requirements During Design

D.6 of D.6 Original Oct 10, 2011 Page: Resiston

Dale,

ELEMENT	OFR PART & REQUIREMENTS	14 DFR PART 33.75 REQUIREMENTS
	(ii) Identification of hazards or ineffective mak controls through the safety assurance processes in subcart D of this part.	expected to occur. This analysis will take into account if applicable (i) Aircraft-level devices and procedures assumed to be associated with a typical installation. Such assumptions must be stated in the analysis."
		The rule limits the safety analysis to operation, procedules and the effectiveness of risk conful is with a manipulation activity.
		(d) If reliance is placed on a safety system to prevent a failure from progressing to nazardoos engine effects, the possibility of a safety system failure in combination with in oasic engine failure must be included in the analysis. Such a safety system may include safety devices, matrumentation, early werning devices maintenance checks, and other similar equipment or procedures.
		This paragraph requires consideration of the effectiveness of thick contrals.
		(6) If the safety analysis depends on one or more of the following items, those items must be identified in the analysis and oppropriately substantiated.
		(1) Maintenance actions being carried or at stated intervals. Additionally, if errors in maintenance of the engine including the control system could read to hazardous engine effects, the appropriate procedures must be included in the relevant engine manuals.
		(2) Verification of the satisfactory functioning of satisfy or other devices all prestight or other stated periods.
		 (3) The provisions of specific imatumentation risk otherwise required
		(4) Flight crew actions to be specified in the operating instructiona established under \$33.5.
		One of Immune fourned Rom service exponence in the course of the safety energy month the intent of indentification of flacardic finality the safety assurance process
5.55 Safety risk assessment and control	(a) The certificate holder must develop and maintain processes to analyze satisfy risk associated with the hazards identified in §5.53(c).	(a) (1) The applicant must enalyze the engine including the control system to assess the tikely consequences of all failures that dan reasonably be expected to occur.

Part 21 (SADS: A vintion Rulemaking Committee Appendix II, Description of How Compliance with Meworthiness Standards Meets SRM Requirements During Design D.7 of D.7 Original Oct 10, 201

Page:

Dale,

Resision

ELEMENT	UFR PART & REQUIREMENTS	14 DFR PART 33.75 REQUIREMENTS
	(b) The certificate holder mixit define a process for conducting risk annest ment that allows for the determination of acceptable safety risk Acceptable safety risk must, at a menimum, compty with the applicable regulatory requirements set forth in Chapter I of title 14 of the Code of Federal Regulations	 The rule specifies the acceptable level of risk. (4) (3) The applicant must show that hazardous engine effects are precided to occur at a rate not in excess of that defined as extremely remole (probability range of 10° to 10° per engine tight hour) (4) The applicant must show that major engine effects are precided to occur at a rate not in excess of that defined as romote (probability range of 10° to 10° per engine fight hour)
	 (c) The certificate holder must develop and maintain processes to develop failing (ink controls that are invises any as a result of the safety risk animation process under jurgraph (b) of this section (1) This certificate holder must evaluate whether the risk will be acceptable whether the risk will be acceptable whether the proposed safety risk control is implemented (2) The safety risk control is implemented (2) The safety risk control must, at a minimum, comply with the applicable regulatory requirements set forth in Chapter I of title 14 of the Code of Federal 	The engine will not manye a type contraste, if the risks evoked scoptible levels defined in the waltation. The applicant introduces itsk controls where mediat and repeals the satisfy analysis, will be risks can be shown to meet the ingularcy requirements (§13.75). The risk controls will not wished any bart of 11. CER, Part 32, since the waltations of all 11. CER, Part 32, since the waltations of all 11. CER, Part 32, since the waltation of an individual destinat abjective of car wing inquire incidence of waltavaller.

Example 3: Ice Detection System

This example illustrates how SRM is accomplished when a design organization devises and incorporates design risk controls to comply with regulations, leading to a certified product.

SRM typically begins with a description of the environment, operating conditions and general product design parameters necessary to identify hazards. When showing compliance to regulations, this description is generally defined within the regulatory material and associated guidance material. The design organization will translate this general description to the actual design for which certification is being sought. With respect to ice detection system this information is contained in 14 CFR 25 1419 (Amendment 25-129) and ACs 25 1419-2 and 20-73A. These actions correspond to "Safety Analysis Regulation" and "Hazard Assassment" in Figure D-1 (below the horizontal dashed line), and are illustrated in the text of the regulation and background information provided in the following associated rulemaking material.

1417R 25.1419 Ice protection

If the applicant seeks certification for flight in icong conditions, the airplane must be able to safely operate in the continuous maximum and intermittent maximum teing conditions of Appendix C. To establish this—

Pari 11 / SMS-A vintion: Rulemalong Committee	Page	D.8 of 11.8
Appendix II. Description of How Compliance - th Microorthiness	Restation	Original
Stationards Meets SRM Requirements Paring Design	(cinus)	YICP 117, 2271 1

(c) One of the following methods of icing detection and activation of the airframe ice protection system must be provided:

 A primary ice detection system that automatically activates or alerts the flighterew to activate the airframe ice protection system.

From the Final Rule

On October 31, 1994, an accident involving an Avians de Transport Regional (ATR) 72 series airplane occurred in icing conditions. This prompted the FAA to initiate a review of aircraft inflight icing safety and determine changes that could be made to increase the level of safety. In May 1996, the FAA sponsored the International Conference on Aircraft Inflight loing specialists recommended improvements to increase the level of safety of aircraft operating in icing conditions. The FAA reviewed the conference recommendations and developed a comprehensive multi-year icing plan. The FAA Inflight Aircraft loing Plan (loing Plan), dated April 1997, described various activities the FAA was contemplating to improve safety when operating in icing conditions. In accordance with the loing Plan, the FAA tasked the Aviation Rulemaking Advisory Committee (ARAC), through its lee Protection Harmonization Working Group, to consider the need for ice detectors or other acceptable means to warn flightnews of ice accretion on critical surfaces requiring crew action. This rule is based on ARAC's recommendations to the FAA.

From the NPRM

The notice of proposed rulemaking (NPRM). Notice No. 07-07, published in the Federal Register on April 26, 2007 (72 FR 20924), is the basis for this amendment. The common period closed July 25, 2007. In the NPRM, we proposed to revise the airworthiness standards for type certification of transport category airplanes to add requirements to unsure the timely activation of an airframe (ce protection system (IPS). We also proposed to add requirements to reduce the flighterew workload associated with operation of an airframe IPS that is manually cycled, and to ensure the Airplane Flight Manual includes. IPS procedures for operation.

Neither the operating regulations nor the certification regulations require a means to warn flighterews of ica accretion on critical surfaces requiring crew action.

The ARAC Ice Protection Harmonization Working Group reviewed icing events and found accidents and incidents where the flighterew was either completely unaware of ice accretion on the airframe, or was aware of ice accretion, but judged that it was not significant enough to warrant operation of the airframe ice protection system (IPS). The ARAC Ice Protection Harmonization Working Group concluded and recommended to the FAA that flighterews must be provided with a clear means to know when to activate the IPS. Flighterew's observation of ice accretions and such observations can be difficult during times of high workload, operations at night, or when clear ice has accumulated

Hazard scenarios are used to model the rock analysis ("Risk Assessment" in Figure D-1, below the horizontal dashed line)

Hozard Scenario

Model Hazard coupled with Linking Mechanism = Bad Outcome Hazard: Flight in icing conditions Linking Mechanism

Part 11 (SMS-Aviation Malemalang Committee	Page	D-8 mf 18.9
Appendix II. Description of How Compliances of Meworthiness	Bay roton	Concurat
Standards Many SRM Renautometry Junior, Design	Elate	3461-10-201

- Fullure to recognize ice build-up
- Failure to activate ice protection system ٠
- Failure to activate ice protection system in a timely manner
- Bad Outcome Loss of lift, leading to crash

Risk Control Criterio

Install an ice detection system to automatically alert crew to turn nn ice protection system. Required for transport category auplanes by 14 CFR 25 (419)

Acceptable Risk Crueria

Guidance for making a determination of acceptable risk is contatoed to ACs 25 1419-2. "Compliance with Ice Detection Requirements of 25,1419 (e), (f), (g) and (h)", and AC 20-73A; "Aircraft fee Protection"

For example, AC 25 1419-2 says in paragraph 4a(7)o.

"System Safety Considerations: The applicant should consult AC 25 1309-1A for guidance on compliance with \$25 1309 In accordance with AC 25 1309-1A, the applicant should accomplish a functional hazard assessment to determine the hazard level associated with failure of the ice detection system. The probability of uncountering the joing conditions defined in Appendix C to Part 25 should be considered to be 1. The unannunciated failure of a primary ice detection system is assumed to be a catastrophic failure condition, unless the characteristics of the airplane in Joing conditions without activation of the airframe IPS(s) are demonstrated to result in a less severe hazard category. If visual cues are the primary means, the pilots retain responsibility to monitor and detect ice accretionwhen an advisory ice detection system is installed. However, the natural tendency of flightcrews to become accustomed to using the advisory ice detection system clevates the importance of the detector and increases the need to make flighterews aware of an advisory ice detection system failure. Therefore, an undetected failure of the advisory ice detector should be considered as at least Major unless substantiated to be a lower failure condition classification

The hazard(s) and risk control criteria are defined and applied during design and certification. The design processes used to satisfy risk control criteria associated with compliance to 25 1410 and 25 [309 necessitate that the likelihood and severity of the bad ourcome(s) are used for a risk based decision on the acceptability of the type design of the ice detection system. (See also section 4.5 of this Working Group Report, Subpart C. Design and Type Certification).

"Risk Assessment" in Figure D-1, below the horizontal dashed line is often satisfied through industry best practices for performing safety assessments in accordance with 14 CFR 25 1309. The AC material provides guidance for a determination of acceptable risk, and design decisions are thusly made ("Define Design Safety Features. Figure D-1) The risk control, as defined through regulation culminates in a "Safe & Compliant Design" Figure D-1) considering the specific design solution presented for compliance.

Testing and further analysis are performed to validate that the as-designed risk controls meet the acceptible tisk criteria contained in the rule and guidance material ("Safety Objective Defined", tigare D-1 above the fiorizontal dashed line). Such validation enables, with regard to design and certification of the product or article, a conclusion of "Safery Objective Met" (Figure D-1).

214 RC Working Document - Not for Distribution

Part 11 (SMS-Avianou Bulemaking Committee	ł
Appendix II. Description of How Compligner with Meworthiness	I
Standard; Meets SRM Repotements Juring Design	I

c: D.10 of D.10 rsion Original c. Oci 10, 2011

Example 4: 14 CFR 23.783(b) Doors

This example illustrates how SRM is accomplished when a design organization devises and incorporates design risk controls to comply with regulations, leading to a certified product.

SRM typically begins with a description of the environment, operating conditions and product design necessary to identify hazards. When showing comphance to regulations this description in generally defined within the regulatory material and associated guidance material. The design organization will translate this general description to the actual design for which certification is being sought. With respect to location of passenger doors, this information is contained in 14 CFR 23. Subpart D, Design and Construction, Personnel and Cargo Accommodations, §23.783(b) (Amendment 25-49), and AC 23-17C, Systems and Equipment Guide for Certification of Part 23 Airplanes and Airships. These actions correspond to "Safety Analysis Regulation" and "Hazard Assessment" in Figure D-1 (below the horizontal dashed line), and are illustrated in the text of the regulation and background information provided in the following associated rulemaking material.

14 CI R 23 763 DOWNER

(h) Passenger doors must not be located with respect to any propeller disk or any other potential hazard so as to endanger persons using the door

From the Final Rule

This amendment completes part of an effort to harmonize the requirements of Part 23 and JAR 23. The revisions to Part 23 in this amendment pertain to systems and equipment airworthiness standards. Three other final rules are being issued in this Federal Register that pertain to airworthiness standards for flight, powerplant, and airframe. These related rulemakings are also part of the harmonization effort

In January 1991, the FAA established the Aviation Rulemaking Advisory Committee (ARAC) (56 FR 2190, January 22, 1991). At an FAA/JAA Harmonization Conference in Canada in June 1992, the FAA announced that it would consolidate the harmonization effort within the ARAC structure. The FAA assigned to ARAC the rulemakings related to JAR/Part 23 harmonization, which ARAC assigned to the JAR/FAR 23 Harmonization Working Group. The proposals for systems and equipment airworthiness standards contained in Notice 94-21 were a result of both the working group's efforts and the efforts at harmonization that occurred before the formation of the working group.

Section 23.785 Doors

Proposed paragraph (b) would add a requirement that passenger doors must not be located nuar any propeller disk or any other potential hazard that could endanger persons using the door. The propeller disk remains the prominent hazard but other items, such as lot deteer surfaces or sharp objects on the arplane structure, are also hazards.

Section 23.783 is amended by revising paragraph (b) to read as follows.

- Sec 23 783 DOM:N.
- (b) Passenger doors must not be located with respect to any propeller disk or any other potential hazard so as to endanger persons using the door.

From the NPRM

During the Part 23 harmonization effort, the FAA established an Aviation Ralemaking Advisory Committee (ARAC) (50 FR 2190) January 22, 1991), which held its first meeting on May 23, 1991. The ARAC on General Aviation and Business Airplane.

Part 11 (SMS-Avinton) Bullemaking Committee	Page:	D.11 of D.11
Appendix II. Description of How Compliance with Meworthiness	Resiston	Original Oct 10, 1011
Staudarits Meets SRM Requirements During Design	doutes.	10ct 10, 2011

(GABA) Issues was established at that meeting to provide advice and recommendations to the Director. Aircraft Certification Service, FAA, regarding the airworthiness standards in Part 23 as well as related provisions of parts 91 and 135 of the regulations.

The FAA announced, on June 2-5, 1092, at the JAA/FAA Harmonization Conference in Toronto. Ontario, Canada, that it would consolidate within the ARAC structure an ongoing objective to "harmonize" the JAR and the FAR. Coinciding with that announcement, the FAA assigned the ARAC on GABA Issues those rulemaking project related to JAR/Part 23 harmonization that were to final coordination between the JAA and the FAA. The harmonization process included the intertuon to present the results of JAA/FAA coordination to the public as NPRM's. Subsequently, the AARC on GABA issues established an ARAC-JAR 23 Study Group.

The IAR 23 Study Group made recommendations to the ARAC on GABA Issues concerning the FAA disposition of the rulemaking issues coordinated between the IAA and the FAA. The draft NPRMs previously prepared by the FAA harmonization team were made available to the harmonization working group to assist them in their effort

A notice of the formation of the JAR 23 Harmonization Working Group was published on November 30, 1992 (57 ER 56626). The group held its first meeting on February 2, 1993. These efforts resulted in the proposals for systems and equipment drivortionessstandards contained in this notice. The ARAC on GABA Issues agreed with these proposals."

Section 23,783 Doors

Furrent Sec 23 783(b) requires that passenger doors not be located with respect to any propeller disk so as to endanger persons using the door. Proposed paragraph (b) would add that passenger doors must be located in relation to any other potential hazard that yould endanger persons using the door. The propellet disk remains the prominent hazard but other items, such as hot deicer surfaces or sharp objects on the airplane structure, are also hazards.

Section 23,783 is amended by revising paragraph (b) to read as follows.

Sec. 23 783 Doors.

(b) Passenger doors must not be located with respect to any propeller disk or any other potential hazard so as to endanger persons using the door

Hazard scenarios are used to model the risk analysis ("Risk Assessment" in Figure D-1, below the horizontal dashed line)

Hazard Scenaria

Model Hazard coupled with Linking Mechanism - Bad Outcome

Hazard. Proximity of passenger door to propeller disk or other hazarda Linking Mechanism

- Failure to observe moving propeller
- Failure to observe/recognize other hazards, e.g., bot deicer surfaces or sharp objects on the airplane structure
- Inability to maintain clearance from propeller disk or other hazard while operating door
- Slip/trip/fall while passing through door

Her water and the second	
Appender II. Description of How Compliance with Mexor Drives Date	Conemal (Ja) 10, 301

Bad Dutcome: Injury due to contact with moving propeller, hot surface, or sharp, object

Risk Control Uriteria

Locine the passenger door(s) away from potential hazards so as not to endanger personsusing the door.

Acceptable Risk Criteria

Criteria and guidance for making a determination of acceptable task are contained in 14 CFR 23 783, 23 811, 23 813, 23 1401, 27 1309, and AC 23 170

Examples of risk criteria:

- 14 CFR 23 783(b) Location of passenger door with respect to a propeller disk or any other hazard most not endanger persons using the door
- 14 CFR 23.811(a)(2) Each external door must be externally marked by a permanent decal which shows the means of opening including any special instructions, if applicable
 - "Applicable" might include instructions intended to ensure clearance from hazards
- 14 CFR 23.813(b)(1). The location of the passenger door must still allow for a
 passageway leading from the aisle to the door that is unobstructed and at least 20
 inches wide
- J4 CFR 25 1301(b) Installed equipment must be labeled as to its identification, function, or operating limitations or any applicable combination of these factors
- F4 CFR 23.1309(a)(2). Any equipment and system does not adversely affect the safety of the airplane or its occupants
- AC 23-17C 23.783 Doors: the ELOS finding process for Part 21. §21.783(b), should include
 - (e) A railing or guard that would deploy to guide passengers away from the propeller plane should be provided as an integral part of the door
 - (f) If engagement of the engine starter would be an immediate hazard to a person near the propeller, an interconnection between the door and the engine starting circuit should be included in the design

The hazard(s) and risk control criteria are defined and applied during design and certification. The design processes used to satisfy risk control criteria associated with compliance to 14 CFR 23.783 and 25.1309 necessitate that the likelihood and severity of the bad outcome(s) are used for a risk-based decision on the acceptability of the type design of the ice detection system. (See also section 4.5 of this Working Group Report, Subpart C. Design and Type Certification)

"Risk Assessment" in Figure D-1, below the hortzontal dashed line is often sansfied through industry best practices for performing safety assessments in accordance with 14 CFR 25.1309 The AC material provides guidance for a determination of acceptable risk, and design decisions are thusly made ("Define Design Safety Features," Figure D-1). The risk control, as defined through regulation, culminates in a "Safe & Compliant Design" (Figure D-1) considering the specific design solution presented for compliance.

Testing and further analysis are performed to validate that the as-designed risk controls meet the acceptable risk criteria contained in the rule and guidance material ("Safety Objective Defined",

Part 11 (SMS-Aviation Rolemaking Committee	Page:	D.13 of D.11
Appendix D. Description of How Complignee with Airworthiness	Restion	Original Col 10, 7011
Standards Meets SRM Requirements During Design	- Cillica	ther myserrin

Figure D=1 above the horizontal dashed line). Such validation enables, with regard to design and certification of the product or article, a conclusion of "Safety Objective Met" (Figure D=1)

Example 5: 14 CFR 23.1309 Equipment, systems, and installations (Amendment 23-49); and 14 CFR 23.1431 Electronic equipment (Amendment 23-49)

This example shows how compliance with the regulations by a design organization accomplishes SRM for those design aspects that are not specifically covered by the regulations. The regulations are minimally prescriptive, and thus require the design organization to expansively identify hazards based on the design of the system and the environment in which it will operate Referring to Figure D-1, compliance with these regulations begins with "System Analysis," (Part 5.51 and 5.53) then to the SRM-equivalent (Part 5.53 and 5.55) path of "Safety Analysis Regulation" leading to a "Safe & Compliant Design."

14 CFR 23.1309 Equipment, systems, and installations

- (b) The design of each item of equipment, each system, and each installation must be examined separately and in relationship to other airplane systems and installations to determine if the airplane is dependent upon its function for continued safe flight and fauding and, for airplanes not lonited to VFR conditions, if failure of a system would significantly reduce the capability of the airplane or the ability of the crew to cope with adverse operating conditions. Each item of equipment, each system, and each installation identified by this examination as one upon which the airplane is dependent for proper functioning to ensure continued safe flight and landing, or whose failure would significantly reduce the capability of the airplane or the ability of the airplane or the ability or the airplane is dependent for proper functioning to ensure continued safe flight and landing, or whose failure would significantly reduce the capability of the airplane or the ability or the ability or the ability or the airplane is dependent for group with adverse operating conditions, must be designed to comply with the toflowing additional requirements.
 - It must perform its intended function under any foreseeable operating condition
 - (1) When systems and associated components are considered separately and in relation to other systems—
 - (i) The occurrence of any failure condition that would prevent the continued safe flight and landing of the airplane must be extremely improbable, and
 - (ii) The occurrence of any other failure condition that would aignificantly reduce the capability of the airplane or the ability of the crew to cope with adverse operating conditions must be improbable.

14 CLR 23.1431 Electronic Laupment

(a) In showing compliance with §23.1309(b)(1) and (2) with respect to radio and electronic equipment and their installations, critical environmental conditions must be considered.

From the Final Rule

(23) 1900

If the design of the airplane includes equipment, systems, and installations that perform functions whose failure condition would prevent combined safe flight and lauring of the airplane, the occurrence of each failure conditions roust he extremely improbable. To addition, on airplanes designed for any type of operation not limited to VFR, the systems whose failure conditions would significantly reduce the airplane's capability or the ability of the orew, to cope with the adverse operating conditions must be improbable. It was recognized that any failure would reduce the airplane's or crew's capability by some

Part 11 (SMS-Aviation Rulemaking Committee	Page:	D. 14 of D. 14
Appendix II, Description of How Compliance with Micworthiness	Resiston	Original
Standards Mean SRM Renausaness During Design	Date	Gay 10, 2011
Stationaria, stear stear herdorean pass when be being		

degree, but that reduction may not be of the degree that would make operation of the airplane potentially catastrophic. The intent of §23-1300(b) is to require that systems whose failure would be catastrophic or potentially catastrophic be evaluated using the latest available analysis techniques.

Although future airplane designs limited to VFR operations are not likely to include uquipment, systems, and installations whose failure condition would prevent continued safe flight and landing of the airplane, the applicability of this requirement, as discussed above, will provide airworthiness standards if the applicant elects to include such systemin the airplane's design. Therefore, the applicability of this requirement has not been revised as suggested by this commentar.

923.1431

This proposes to amend §23 1431 to revise the current rule that addresses radio equipment only by including other electronic equipment that is installed in a Part 23 airplane. Two comments were received. One commenter asks for a definition of the words, "critical environmental conditions" used in proposed §23 1431(a). Critical environmental conditions are those environmental conditions under which a piece of equipment will not perform its intended function. By including this requirement, conditions that may be critical to the operation of a piece of equipment must be considered. Consideration of such conditions would include, but not be limited to, temperature extremes, vibration levels, and humidity.

From the NPRM

\$23 1309

A new §23 1309(b) is proposed which will require a detailed examination of each item of equipment, system, and installation. This examination is to determine whether a failure would affect the airplane's continued safe flight and landing. Each item of equipment, each system, and each installation identified by such an examination as being critical to the safe operation of the airplane would be required to meet additional requirements. This will permit the approval of more advanced systems that were not envisioned when §23 1309 was added to Part 23, without the need for special conditions.

623.1431

This proposal would include electronic equipment that is being installed in Part 23 airplanes as well as radio equipment. When the existing regulation was adopted, radio equipment was the primary electronic equipment installed. For standardization in the application of FAA requirements, this proposal is consistent with §25.1431(a) and (c) §23.1309(b)(1) and (2) that are referenced are the proposed regulations in Notice 5. Small Airplane Airworthiness Review Program

Huzard Scenario

Model Hazard coupled with Linking Mechanism - Bad Outcome

Hazard: Failure or malfunction of highly complex safety-critical equipment which could prevent continued safe flight/familing.

Linking Mechanism. The complexity of such items and equipment precludes performance of only a bottoms-up FMEA type analysis) top-down functional decomposition is required as well to properly allocate safety requirements to items and equipment.

Bad Ourcome Loss/malfunction of item/equipment, leading to trash

Part 11 / SMS-A vintour Rulemaking Committee	Page	DISOTON	
Appendix II. Description of How Complignee with Meworthiness Statistical Meets SRM Requirements During Design	Date	Get 10, 201	

link Control Criteria: This is defined in \$23,1309(b)(1)(2). Additionally, $$23,1431(\pi)$ requires that critical environmental conditions be considered, i.e., it is not sufficient to consider only failures which could occur within the established environmental limits of the alternit.

Acceptable flisk Criteria: Acceptable quantitative risk criteria is provided by the regulation and AC 23 1309-1. Additionally, qualitative risk criteria (design rigor requirements expressed as Development Assurance Levels, or DAL) are imposed by AC 23 1309-1, as well as system specific ACs.

Design organizations consider safety aspects of each design through consideration of §23.1309. Draft Advisory Circular AC AMJ 25.1309 provides a methodical and systematic manner which ensures that this process and its findings are visible and readily assimilated. Figure D-2 illustrates a systematic approach to safety assessments used throughout industry today without the guidance of additional SMS regulation.

Per §23-1309(b), the design of each item of equipment, each system, and each installation is examined separately and in relationship to other airplane systems and installations to determine if the airplane is dependent upon its function for continued safe flight and landing.

For each system reviewed, the failure conditions are identified and classified. All relevant engineering organizations are involved in this process. This identification and classification may be done by conducting a Functional Mazard Assessment. These actions correspond to "Hazard Assessment" and "Risk Assessment" in Figure D-1 (below the horizontal dashed line, right hand column).

The analysis would be conducted to produce the data which is agreed to by the certification authority as being acceptable to show compliance. The analyses and conclusions of multiple safety assessments would be assessed to ensure compliance with the requirements for all arreraft level failure conditions.

The Design organization would then choose the means to be used to determine compliance with \$23 (309(b) whether it be through qualitative or quantitative measures or both

This data is then used to prepare compliance statements, establish certification maintenance requirements, airworthiness limitations and flight manual requirements at the product level, thus dollining type design



Figure D-2: Airplane-Level System Safety/Certification Process

Simple systems and systems with failure effects classified as no greater than Major do not require quantitative assessment to satisfactorily show compliance to the requirements of §23-1369. A qualitative analysis relying on experienced engineering judgment, service history or operational experience is sufficient. For Major failure conditions, a functional Fault Tree Analysis (FTA) or Failure Modes and Effects Analysis (FMEA) may be required for complex systems. Figure D-3 (taken from AC 23-1309), below, lists the qualitative probability requirements for each failure condition classification.

For failure conditions classified as Hazardous or Caustrophic, quantitative analysis is generally required to supplement qualitative analysis. Failure rate data is assigned to events at the lowest FTA event or FMEA level which can be substantiated. Figure D-3, below, lists the allowable quantitative probabilities for each failure condition classification per class of 14 CFR 23 airplane class.

In addition to qualitative and quantitative requirements Figure D-3 lists the Development Assurance Levels (DALs) for each failure condition classification per class of 14 CFR Part 23 airplane class. These DALs are intended to specify a level of development rigor per FAA AC 20 115 and FAA AC 20-152 "Safety Objective Defined", Figure D-1 above the borizontal dashed line. Compliance to such standards enables, with regard to design and certification of the product or article, a conclusion of "Safety Objective Met" (Figure D-1).

Part 21 / SMS-Aviation Rulemaking Committee
Appendix D. Description of How Compliance with Airworthiness
Standards Meets SRM Requirements During Design

Page:

Date:

Revision:

Failure Conditions	No Safety Effect	<minor></minor>	<major></major>	<-Hazardous>	< Catastrophic>
Aliowable Qualitative Probability	No Probability Requirement	Probable	Remote	Extremely Remote	Extremely Improbable
Effect on Ainplane	No effect on operational capabilities or safety	Slight reduction in functional capabilities or safety margins	Significant reduction in functional capabilities or safety margins	Large reduction in functional capabilities or safety margins	Normally with hull loss
Effect on Occupants	Inconvenience for passengers	Physical discomfort for passengers	Physical distress to passengers, possibly including injuries	Serious or fatal injury to an occupant	Multiple fatalities
Effect on Flight Crew	No effect on flight crew	Slight increase in workload or use of emergency procedures	Physical discomfort or a significant increase in workload	Physical distress or excessive workload impairs ability to perform tasks	Final Injury or incapileration
Classes of Airplanes:	Allowable Quantitative Probabilities and Software (SW) and Complex Hardware (HW) Development Assurance Levels (Note 2)				
Chess I (Typically SRE 6,000 pounds or less)	No Probability or SW and HW Development Assurance Levels Requirement	<10 ³ Note 1 P-D	<10 ⁴ Notes 1 and 4 P=C, S=D	<10 ⁶ Note 4 P=C S=D	<10 ⁻⁶ Note 3 P=C, S=C
Classi II (Typically MRE, STE, or MTE 6,000 pounds or less)	No Probability or SW and HW Development Assurance Levels Requirement	<10 ⁻³ Note 1 P=D	<10 ³ Notes 1 and 4 P=C ₄ S=D	~10 ⁴ Note 4 P=C, S=C	<10 ⁷ Note 3 P=C, S=C
Class III (Typically SRE, STE, MRE, and MTE greater than 6,000 pounds)	No Probability or SW and HW Development Assorance Levels Requirement	<10 ³ Note 1 P-D	<10 ¹ Notes 1 and 4 P=C, S=D	<10 ⁷ Note 4 P=C, S=C	<10 ⁴ Note 3 P-B, S-C
Class TV Typically Commuter Category)	No Probability or SW and HW Development Assurance Levels Requirement	<10 ⁻³ Note 1 P=D	<10 ⁸ Notes 1 and 4 P=C, S=D	<10 ⁷ Note 4 P=R: S=C	<10 ³⁰ Note 3 P=A, 5=B

Figure D-3: Probability Requirements for Failure Condition Classification

The basic SRM requirements of a Safety Management System are clearly integrated into the abovementioned 14 CFR Part 23 regulations that drive applicants to design safe products as part of the current process required by 14 CFR 21.21. The creation of a new Safety Management regulation in place of applicable airworthiness standards would not contribute materially to the level of safety over that realized by utilizing the existing system for the design and certification of products
APPENDIX H—OVERSIGHT WORKING GROUP REPORT

21ARC Working Document – Not for Distribution Oversight Working Group Report

Oversight Working Group

Report to Full ARC

February 5, 2014

Approval Signatures

Bill Whitton Gulfstream

John Hunter Heico-

Carlos Quiles FAA, AFS-340



Dave Chapel GE



Cesar Gomez FAA, ANE-170

Approved via Email Sarah Thatcher FAA, AIR-110

Chris Watkisa Bombardier

Jose Luiz Belderrain Embraer

Page 1-2

Eric Lesage Airbus

Rick Baggette Boeing

Kurt Krumlauf FAA, ANM-108L

Van Ruggles Garmin

John Bouma Cessna

Approved via Empil Marisa Stephenson Rockwell Collins

Dan Shapiro Sikorsky Global

Table of Contents

L.Everaltive Summary

- 1.a. introduction
- L.D. Summary of Recommondations
 - 1 b.1 Dversight Model Recommendations
 - 1.b.2 Assessment Methodology Recommendations
 - 1.b.3 Dversight Implementation Recommendations
- 2. Charter: Requirements / Approach / Taxonomy
 - 2.a. Requirements.
 - 2.8. Working Group Approach
 - 2.c Taxonomy
- 3 Existing FAA Oversight Model
 - 3.a. Existing FAA Oversight Activities / Descriptions
 - 1.p. Existing FAA Oversight Models for Complex ODAs
 - 3.b.1. FAA Oversight of the Boeing ODA
 - 3.b.2. FAA Oversight of the Gulfstream ODA
 - 3.: Opportunities with Today's Oversight
- 4. Future Oversight Model
 - 4.a. Oversight Key Areas
 - 4 in Design / Manufacturing Organizations
 - 4.c. Product / Article
 - 4.d. Post Certification (Continued Operational Safety)
 - A.e LOPI for Manufacturing
- 5 Oversight Assessment Methodologies
 - 5.a. mittal Assessment
 - 5.b. Certificated Design Diganization (DD) Surveillance
 - 5.c. Expansion Assessment
- a Future Oversight Assessment Model Implementation
 - 6.a. implementation Coundaries
 - 6.0 Implementation Assumptions
 - 5.c. Implementation Approach

Page | I

- 6.c.1 Proof of concept projects.
- 6.c.2. FAA transition plan
- 6.c.3. Industry transition plan
- 6.d Implementation issues Requiring Full ARC Foundination
- 6.e. Applicant Types in "Design Organization World"
- 6.1 Implementation Evolution
- 7 Membership

Appendices:

- A Oversight Working Group Charter
- B. Boeing Aviation Safety Oversight Office (BASOO) and Gulfstream Aviation Safety Oversight Office (GASOO) charts.
- C. DO Astessment Tool
- D. Aviation Safety (AVS) Safety Management System (SIMS) Strategy and Framework to Manage Safety Porformance in AVS
- E EASA Embodiment of Level of Involvement (LOI) and Safety Management System (SMS) Requirements into Part 21
- F. Responses to Oversight / Assessment Relevant Questions

1,086.1 4

1. Executive Summary

This section introduces the Oversight Working Group, its Part 21 / SMS ARC purpose, and a summary of recommendations in this report.

1.a. Introduction

The Oversight Working Group (WG) was created and commissioned to support ARC recommendations and to submit a report addressing the following:

Changes to the FAA oversight methodology based on recommendations for alranges to Part 21 that takes into account existing FAA processes and oversight and delegation programs for design and manufacturing related certificates and approvals and authorizations.

1.b Summary of Recommendations

1.b.1 Dversight Model Recommendations are:

- Create a single oversight presence by the FAA. The three key oversight areas are:
 - Organizational: Transition from traditional show / find compliance to organizational performance based oversight model.
 - Product and Articles: Transition from FAA's traditional role of direct project involvement to a Level
 of Project Involvement (LOPI) approach that is incused on performing governmental functions.
 - Post Certification (Continued Operational Safety) Transition from today's traditional reactionary approach to a systemic (process hased) surveillance model
- Performance Based oversight: Charter a dedicated effort with industry and FAA to develop recognizable performance indicators prior to implementation of the new oversight model.

1.b.2 Assessment Methodology Recommendations

- The Oversight working group recommends a centralized, systemic (process based) oversight approach for initial and impuing assessments.
- The assessment methodology will cover a standardized approach to quality, design, and safety
- In support of this recommendation, the team has provided a capability based assessment tool, PRO: / COI): analysis of oversight management options, and supporting rationale for the recommendation.
- FAA Oversight Teams report to a centralized FAA organization. Establishing a central FAA oversight organization will achieve standard surveillance practices, create centralized policy, be a single source/repository for the oversight data which will drive the risk based modeling controls, allow for a highly trained staff in system surveillance, skill management, and a single source for corrective actions.

10001-5

1.0.3 Oversight implementation Recommendations: The Oversight working group recommends the oversight implementation includes three Major Transition Steps:

- 1 Proof of Concept Pre-implementation:
 - Ensure through proof of concept plans that the requirements proposed by the Part21/SMS ARC are
 practical, effective, and efficient. Determine if the transition from "Mature ODA" to DD has benefits to.
 FAA and Industry.
- AA Transition Plan Transition principle:
 - FAA should not release a linal rule before FAA has demonstrated the necessary cultural shift to perform system oversight. To achieve a cultural shift, policy and organizational changes may be required
- 3 Industry Transition Plan:
 - The organization must establish the systems required of an approved organization while still working as a non-certificated applicant or a delegated organization. Applicants that are working toward becoming a design organization (DO) demonstrate compliance to those requirements on an "as ready" basis.

Page | b

2. Charter: Requirements / Approach / Taxonomy

This section will describe the foundational elements the Working Group used to build its recommendations.

- Section 2 a. captures the underlying inquirements for creating the future oversight. These requirements
 evolved from "assumptions" created initially by the full ARC and refined by the Working Group as the team
 studied, compared, and collaborated with the other Working Groups.
- Section 2.b. captures the approach, or the tasks established by the full ARC and refined by the Working Group.
- Section 2.c. captures the taxonomy adopted by the Working Group and shared / collaborated with the other Working Groups. Early development of the taxonomy was critical to baseline discussions, recommendations and interface with other Working Groups / full AIC.

2.n. Requirements

Requirements established by the Oversight Working Group are categorized into three sections:

- 1. Oversight
- 2 Assessment
- 3. Implementation
 - Oversight Requirements: Establishment of an FAA systems oversight approach to certification processes and continued all worthiness assumes the part 21 rewrite will require.
 - A single, integrated FAA oversight / surveillance / enforcement approach for organizations that inic certificate holders that includes production & design, certification project activities and continued operational safety (COS); and:
 - Will not include Repair Station oversight for these facilities that have multiple approvals.
 - Will be aligned with design organization requirements established by the Organizational Working Group
 - Will consider direct FAA oversight and FAA accepted "other party" oversight
 - Will be based on a risk-based approach
 - For organizational oversight
 - Will address FAA surveillance in terms of frequency and types of audits (design, manufacturing, projects, continued operational safety, etc.)
 - Will include oversight of COS processes considering the general surveillance process model currently used in Canada
 - Will include performance based oversight methods / measures
 - For product and article oversight:
 - Considering "Level of Project Involvement" (LOPI) process models currently used in Canada and under development in Europe
 - Recognizing the IOPI may be at the "specialist" level
 - Will not require FAA delegation

Page 7

- 2 Assessment Requirements: Assessment methodologies will be required to
 - Accommodate / Harmonize processes across FAA organizations (ACO, MIDO, F5DO = AEG, etc.)
 - Establish the initial compliance leading to FAA issuance of an organizational approval (operating certificate) based on qualification criteria (level of acceptance) established by the Organizational Working Ecoup.
 - Establish the compliance for FAA acceptance for a design organization progression based on qualification criteria (level of acceptance) established by the Organizational Working Group
 - Istablish performance requirements for this on going oversight / surveillance to establish the effectiveness of the systems in place
- Implementation Requirements: Establish the transition plan of today's FAA oversight system to the recommended revised oversight system necessary for part 21 changes. Robust change management will be critical to implementing the revised part 21. Transition plan should include criteria based gates, benefits metrics, etc.

2.b. Working Group Approach

The approach established by the Working Group was to establish task in the charter statement, establish a work plun and execute the work plan with frequent interaction with the Full ARC and other Working Groups. The tasks were broken into Oversight / Assessment Tasks and implementation Tasks. In addition, relevant questions were brainstormed at the Full ARC and Working Group levels. The relevant questions were applied to the deliverables as a means to ensure completeness.

Dversight / Assessment Tasks / Deliverables:

- A definition of the FAA surveillance, oversight requirements for the organizations that are certificate holders
 given the predicted outcome of the part 21 rewrite including the who, how, and what using a risk-based
 approach, including a definition of FAA assessment methods for issuance of new organizational approvals /
 progressions
- A definition of the FAA surveillance / oversight requirements for project activity including level of Project (involvement (LDPI) requirements
- A definition of the FAA surveillance / oversight requirements for COS activities

implementation Tasks / Deliverables:

- It definition of what the transition model would look like, including gaining the needed understanding and support of the changes from key stakeholders in order for the organizational changes to be implemented effectively.
- Determine the skills, competencies, and associated training requirements for FAA. Trind Party and/or "Organizational" individuals performing oversight activities
- Maintain FAA technical expertise while transitioning to oversight focus.
- Create or revise LAA Orders and advisory material necessary for NPRM, including 8000.367A.

Page | S

Relevant Questions:

Questions were generated by the Full ARC and by the Working Groups as a means to "test" in "exercise" the deliverables to ensure completeness. The relevant questions were categorized by Oversight / Assessment and implementation. The relevant questions can be found in Appendix F. The oversight working group reviewed on in question and made a determination if it was addressed in the report.

2.c. Taxonomy

The following is a list of definitions adopted as part of the Part 21 SMS ARC for use across all Working Groups and the Fall ARC:

- Accountability Framework: An established set of responsibilities and commitments of the FAA and industry.
- · Applicant DO: thd
- Approved Data: Data approved by FAA employees, III designees, or a DO acting under the authority of III certificate. (May need to be re-visited based on approved data discussion)
- Assessment: Process of measuring or judging the value or tevel of something. Envision this as the means of determining the level certification granted to the design organization.
- <u>Certificate Surveillance</u> FAA actions to monitor the DC certificate holder and to determine the holder's compliance with the provisions of its certificate(s).
 - Note. In the Oversight section we discuss managing these organizations through surveillance.
- <u>Compliance Assurance System (CAS)</u>: DO holder's system for ensuring that it compiles with the applicable regulations.
- Determination of Compliance: A decision made by the certificate holder that compliance has been shown with the applicable regulatory requirements. [NOTE: The ARC has referred to "regulatory requirements" rather than just "anworthiness standards" because of (t) recommendation that DO eventually include determination of compliance with other 14 CFR Parts, such as Parts 26, 34, and 36.]. It may also be a decision made by the certificate holder that data previously approved by the FAA or data determined to comply by another CAA under the provisions of a bilateral intworthiness agreement between the United States and a foreign country or jurisdiction, are valid and applicable to the design of the product, part, or appliance for which it is to imused, including the applicable certification or approval basis.
- Compliance Finding. FAA decision (either directly of through a designee) that compliance has been shown with the applicable regulatory requirements
- <u>Corrective Action</u>): An action required to be taken by the Design Organization to address non-compliances and problems with the organization's procedures or performance. The non-compliances may result from:
 - I Internal Audits conducted by the BO
 - IAA Surveillance
 - DO Employee Observations
 - Voluntary Disclosures.
- Culture of Compliance: Knowledge, beliefs, attitudes, and behaviors of an organization that are focused on ensuring regulatory compliance within all its activities.

PARE

- Descriptive Data
 - From the Organization WG: Outa that defines the type design that needs to be determined compliant to the applicable airworthiness standards. The descriptive data is what is approved by the TAA when a design approval certificate is issued.
 - The drawings and specifications necessary to define the configuration shown to comply.
- DO Executive: The company individual directly responsible for ensuring that the DO meets all of the regulatory, responsibilities.
- + DO Point(s) Of Contact: The individual(s) within the DO responsible for all communications with the FAF.
- Eligible Data: Data developed under an approved DO system, assuming a specified, but not FAA-established, certification basis, and product type design if appropriate.
- Enforcement: An action taken by the EAA most appropriate to promote safety and compliance with the statutory and regulatory requirements
 - The program provides a wide range of options for addressing noncommitance:
 - Educationial and remedial training efforts,
 - · Administrative action in the form of either a warning notice or letter of correction,
 - Certificate suspensions for a fixed printed of time.
 - Civil penalties,
 - Indefinite certificate suspensions pending compliance or demonstration of qualifications.
 - · Certificate revocations,
 - injunctions, and
 - Refertals for criminal prosecution.
- Evaluation: Differmining the adequacy and effectiveness of an organization through a review of organizational policies, procedures and systems.
- FAA Approved Documents: See Section IV, F. of the CDO ARC Report, titled 'CDO Approval of Data'.
- FAA Oversight Team: FAA personnel assigned to provide guidance and oversight of the DO in meeting (I), regulatory requirements
- Finding of Compliance: See Section IV. F. of the EDO ARC Report, tilled 'CDO Approval of Data', and Section IV. H. of the CDO ARC Report, titled 'International Considerations'.
- Governmental Function(s): Statutory and regulatory functions, actions, processes, and responsibilities of the
 government which are not eligible to be assigned to private persons, including designees. Particularly because
 they are so intimately related to the public interest as to mandate performance by Government employees.
 - Examples of Governmental Functions may include:
 - Rolemaking, including issuing an Airworthiness Directives (AD)
 - Establishing the Certification Basis
 - Issue Paper Resolution
 - Establishing Special Conditions
 - Granting Exemptions
 - Defining Equivalent Level(s) Of Safety (ELOS)
 - Determination of unsate condition(s).
 - Quality System Approvals
- Holder DOt tod

- Inspection: A formal systematic and independent review of organizational polices, procedures and systems.
- Level of Project involvement (LOPI). The interactive process that the besign Organization (DO) shares with their assigned Airctaft Certification Office (ACO) for specific engineering / design elements and with the Menufacturing inspection District Office (MIDO) for specific production elements during certification projects.

The criteria/factors influencing the decision of when to be involved will include but is not limited to Governmental Functions, such as

- Novel or unusual features which may require issuance of Special Conditions
- Significant issues which may require issue Papers.
- Defining Equivalent Level(L) Of Safety (ELOS)
- Management System: See the definition for System.
- Methods (or Means) of Compliance:
 - Need a definition from Organization WG
 - Notes;
 - · Method Process
 - Means, Capability
 - Novel or Unusual: "The phrase "hovel or unusual" as used in Sec. 21.16 is a very relative term. As used hereafter in applying Sec. 21.16 to justify the issuance of spoulal conditions, "hovel or unusual" will be taken with respect to the state of technology enviraged by the applicable airworthiness standards of this subchapter. It must be recognized that in some areas which will vary from time to time the state of the regulations may somewhat lag the state of the art in new design because of the rapidity in which the state of the rat is idvancing in civil aeronautical design and because of the time required to develop the experiment base needed by the FAA to proceed with general rule making. Applicants for type certification of a new design have the opportunity to mitigate the impact of not knowing the precise anivorthinesis standards to be applied for "novel or unusual design features" by consulting with the FAA early in their certification planning when such features are suspected or known by the applicant to exist. If should also be recognized that, because of the intentional objective nature of the sitworthinesis standards of this subchapter, many new design features which might be thought of as "novel or unusual design features" may almady be adequately covered by existing regulations, thus obvioling the need to issue special conditions. – Preamible matinia to 14CFR 21.15
- + Oversight:

A systems approach to review on organization's performance, validate the development of their defined system and verify compliance to the requirements of a certified design organization to determine sufficiency.

Oversight activities, include:

- Reviewing the work performed.
- · Evaluating performance for quality assurance
- Ensuring that required training has been completed, and
- Providing constructive feedback
- Taking corrective action, including enforcement as necessary

Procedure: A fixed, step-by-step sequence of activities or course of action (with definite start and end points) that must be followed in the same order to correctly perform a task.

Quality Management System:

- From the Organization WG A set of interneliated or interacting quality processes accomplianed by the organization through the establishment of policy and objectives, and achieving those objectives Notes:
 - . The baseline Part 21,137 for PAII how are we defining this for DO?
- Safety Assurance

Need a definition from SMS WG

Safety Critical

Need a definition from SMS WG

- Safety Culture: The product of individual and group values, attitudes, competencies, and patterns of behavior that determine the commitment to an organization's safety programs
- Safety Management: The act of understanding and making decisions and taking actions to lower mik, inherent
 in all human activity, to acceptable levels
- Safety Management System:
 - Need a definition from SM5 WG

From the Organization WG:

- Safety Management System: An integrated collection of processes, procedures, and programs
 that ensures a formalized and proactive approach to system rafety through risk management.
 Risk analysis and assessment are required for all changes to identify safety impacts. The SMS is
 a closed-loop system, ensuring all changes are documented and all problems or issues are
 tracked to conclusion. When properly implemented, an SMS establishes a safety philosophy or
 culture that permeates the entire organization in the monitoring and continuous improvement
 of safety.
- Notes:
 - is the baseline definition in 14 CFR Part 5 or are we expanding/redefining this for DO/
- Senior Company Management. These in the company management chain above the DO executive who are accountable for the actions of the DO
- Showing:
 - From Organization WG
 - Showing of Compliance: tbd
 - Notes
 - The DO is responsible for 100 a showings
- Statement of Compliance: A statement from the DO to the Administrator certifying that compliance with the applicable regulatory requirements has been determined and the procedures likted in to FAA approved b() Procedures Manual have been followed.
- Substantiating Data:

From the Organisation WG: Documentation related to a design approval applicant's showing of compliance to the applicable airworthiness standards

- Supplier DO: A separate DO entity in its own right provides an article to an applicant/holder DO.
- Surveillance: The combination of Evaluation and Inspection to accomplety a review of organizational system to
 determine the adequacy and effectiveness of an organization.

Page 12

- System: An integrated set of constituent elements that are combined in an operational or support
 environment to accomplish a defined objective. These elements include people, hurdware, software, firmware,
 information, procedures, facilities, services and other support facets.
- Validation: Validation is the process of proving that the functions, procedures, controls, and safety standards are correct and the right system is being built. I.e. the requirements are unambiguous, correct, complete, and verifiable
- Verification: The process that ensures that the system requirements have been met by the design solution and the system is ready to be used in the operational environment for which it is intentied.

J. Existing FAA Oversight Model

The Oversight Working Group established as one of its beliverables an existing definition of Loday's FAA oversight. Today's FAA oversight is captured with:

- Emsting FAA oversight delineated by FAA lines of business (ACO; MIDO; and Hight Standards)
- Existing uniquely designed Quersight Management Teams created for larger_complex industry Organization Designation Authorizations (ODAs)
- Existing FAA Oversight Opportunities

3.a Existing FAA Oversight Activities / Descriptions

Table 3-1 outlines existing FAA certification policy, oversight functions and activities for the FAA organizations at the time of this report. The heading in each row shows various certification and oversight activities that cross FAA organizations. The columns allow all the functions of each FAA organization to support each of the certification and oversight activities.

Table 3-1. Current FAA Certification and Oversight Functions

ACO Functions	MIDO Functions	FSDO / AEG Functions
 Design Certification Activity TC/STC/TSO/PMA applications + X110, 4, 81 tot 42, 8150-1 RBRT system + X110, 4 Issue papers + 8110, 4 Ondae Bunden - 8100, 11 Contineation basis + 8140, 4 Operational smitability + 8110, 4 FLOS + 8140, 4 CA - 8110, 54 Commercial Paris - 8110, 118 Confineation of deviationale 8110, 4 Disposition of deviational 8110, 4 Flucture Strep & CIP - 8110, 4 Pingbi nosis test - 8110, 4 Flucture Strep & CIP - 8110, 4 Compliance Inspection - 8110, 4 Compliance Inspection - 8110, 4 TRA/TIR - 8110, 4, 4000, 1 PMA Life lumited Ameles - 8110, 42, 8120, 22 Jonae TC/STC/TSDA - IGF except STC - 8110, 4, 8110, 42, 8150, 1 	 Besign Certification Activity RBRT system - 8110.4 Certification and contornally plans PSCP & CIP - 8110.4 Finding compliance - 8110.4 Finding compliance - 8110.4 Conformity Inspections - 8110.4 Review process specification - 8410.4 Winness toxt - 8410.4 Winness toxt - 8410.4 NDT evaluation - 8140.4 NDT evaluation - 8140.4 Configuration control of design changes - 8110.4 Configuration control of design changes - 8110.4 Airworthinest certification for flight testing (B&D, Show Compliance, SFA) - 8140.4 8130.20 Beriodic Safety checks for flight yest aircraft-8110.4 	 Design Certification Activity by AEG ICA 8110.4 Operational and Maintenance Evolutions 8110.4 TIA flight reas-8110.4 Flight Standardization Board - 8110.4 Mamierance Review Board - 8110.4 Flight Operations Evoluation Board - 8110.4 Type Runnings for jurnian -8110.4
 Besign Certification Diversight Investigations - IGF = 8020 11 2150 3 Voluntary disclosures - 2150.3 Special Tachment Audia - IGF - 8110 4 COS Service Difficulty Reports 21.3 - 8110 107 NTSE recommendations - 1220.2 Issue ADS - 8040 4 8040.5 IR-A48040.1 AMOC - 8110 107 MSAD - Pathures, millinguone, defects- 	Besign Contilication Diversight Special Technical Audas - IGF -#110/4 DOS: D Service Patheolic Reports 21/3 - \$110.007 MSAD - Palates authmetions delvets - \$110.007	Design Certification Dversight N/A

in the second second	FSDO / ALG FUNCTIONS
 Foreign TCSTC V didarious International agreements Finding compliance - 8110 52 	Foreign TC/s/Ff Validation. /9/4
 Delegation oversight Delegation of individual designees and designations (100.2, 8100.8, 8100.15) Airworthiness application package and limitions review (8100.8) Review and approve ODA procedures minuals (\$100.15) Conformity report review ordividual designees 8100.8 Delegated Organization tropection Program (DOIP) - 8100.15 Supervisory visits - 8100.15 Supervisory visits - 8100.15 Over the shoulder reviews individual designees 8100.8 Review internal audits - \$100.15 	 Delegation reversight Delegation to individual designees and designees to individual designees and designees and environments application package & forsidations review - 8100 X Review and approve ODA procedures, manuals - 8100,15 Delegated Organization Inspection Program (DOIP) - 8100,15 Supervisory vosits - 8100,15 Review internal audits - 8100,15 AFS Designees and Designations Volume 17 Chaptors 1 thm 10
 Production Certification PC/PMA applications = 8120/22 Review and approve new quality g)stem = 1GF - 8120/22, 8120/23 PC/PMA/TSOA pre-production approval and/ts =1GF - 8120/22, Conformity glans (CIP) = 8110/4 Production Conformity inspection - 8120/22, 8110/4 Writess test production - 8120/22, 8110/4 Undue Bardon - 8100/11 Commercial Parts = 8110, 118, 8120/22 Issue PC/PMA = 1GF - 8120/22 Issue PC/PMA = 1GF - 8120/22 Issue PC/PMA = 1GF - 8120/22 Instance PLR = 4GF - 8120/22 International agreements - 1GF - BAA/IPA, AC21-27, 21-1 	Production Certification N/A
	 International agreements Finding compliance - 8110142 Delogation of individual designees and designations - 1100.2, 8100,8, 8100.15 Airworthiness application package and limations review (8100.8) Review mut approve ODA procedures manuals - 8100.15 Conformity report review ordividual designers 8100.8 Delegated Organization frequency rules - 8100.15 Supervisory visits - 8100.15 Supervisory visits - 8100.15 Over the shoulder reviews individual designers 8100.8 Review matrial audits - 8100.15 Over the shoulder reviews individual designers 8100.8 Review matrial audits - 8100.15 Over the shoulder reviews individual designers 8100.8 Review matrial audits - 8120.22 Review and approve new quality is stem - 1GF - 8120.22, 8110.4 Witness test production - 8120.22, 8110.4 Witness test production - 8120.22, 8110.4 Witness test production - 8120.22, 8110.4 Witness test production - 8120.22 Issue PC/PMA - IGF - 8120.22 Issue PC/PMA - IGF - 8120.22 International agreements - 1GF - BAA/IPA, AC21-21, 21-3

ACO Functions	MIDO Functions	FSDO / AEG Functions
 Production Confidentian Oversight Review changes to quality system - 8120.22 Production Flight Testing Procedure (PFTP) - 8120.22 Design change procedures - 8120.22 MRB procedures - 8120.22 Service difficulty reporting - 8120.22 Audits QSA = 10F - 8120.23 Periodic clocks of production flight test - 8120.22 	 Production Certification Decrement Review and approve changes to qualify system = 8120.21. Attoo 15 Audio: REIRT system = 8120.21. Principal Inspector (PD) = 8120.23. Principal Inspector (PD) = 8120.23. Special Technical Andim = 8120.23. SCA - (GF = 8120.23. SCA (Supplier Control, Audity) = 8120.23. Corrective action vertification = 8120.23. International agreements = 1GF = 2150.3. Volumitry disclosures = 2150.3. COS: Service Difficulty Report 21.50.1. COS: Service Difficulty Report 21.50.25. Quality failures multimerions, defects = 8120.23.8140.307. NTSB recommendations = 1270.2. 	Production Exercification Oversight
Aircraft Certification Review openitional / envergency procedures used for experimental certificates &(10.2) Review and resolution of nonconformates for anreaft manufactured under a TC only star 7	Aircraft Certification (Original) • Issue airworthiness certificates/approvals: • Export + 8130-2: AC21-23, 11-2 • Standard A/W- 8130-2 • Unport - 8130-2, AC21- 11-31-2 • Section Add - 8170-7	Amerafi Certification (Recommu) • Issue anworthiness confificatestapprovali. a Export - 8130.2 • Standaril 4/W-8130.2 • Import - 8130.2 • Import - 8130.2 • Special A/W - 8130.2 8110.56, 8130.29 • Stand Cloub Ambridge

(SFA) = 8130/2 = Special High permits = 8130/2 = 8130-3 tags used parts = 8130/21
 Repute Station Certification Repute Station (RS) Conflication: Volume 2 Chapter 11 Fort 145 Reput Statums Certification of RS and Satelines RS w(USA not order a MIP V2 C11 S3 Certification of R5 Located outside the USA not order a MIP V2 C11 S3 Evaluating R5 Manual and Quality Mironit or Revision V2 C11 S4 Facilities and Equipmen V2 C11 S6 Certification of R5 located in the US applying for EASA 145 V2 C11 S0 Maintenance Implementation Procedures (MIP) FAA Order 8000 85A Overaight Activity - Certificate Management System (FS1MS) 8900.) Satyeitlance Volume 6 Chapter 13 Part 145 Risk Management V6 C11 S2 Importion Activity Surveillance Volume 6 Chapter 13 Part 145 Applications - Vo C9 S 2, S1 & S7 Certificate Requirements to C9 S1 In-Depili Inspection of Part 145 R5 V0 C9 S2 Hevord System V6 C9 S3 Monual System V6 C9 S4 Honsing and Facilities V6 C9 S8 Ports and Marenal Program V6 C9 S8 Ports and Marenal Program V6 C9 S1 Training Program V6 C9 S11 Manneanian Drogram V6 C9 S1 Personnel V6 C9 S10 Training Program V6 C9 S11 Manneanian Program V6 C9 S12 Work Away from Pix location V6 C9 S1

ACO Functions	MIDO Functions	FSDO / AEG Functions
		V0 C0 S20 COS • Service (Allfrenh) Reports 21:3 • St20.25 • Outling fulfares, multimetions, defects- 8120.25, 8110,107 NTSB recommendations - 1220 I Investigations - 2150.3 • Volumenty disclosures -3130.3 • Corrective action verification - 2150.3
		Air Operator Confidention Activity 2 Certification Process: 8900-1 Volume 2 C 3, 4, 5, & 7 Air Agency Certification Activity • Certification 8900-1 Volume 2 C 8, 9, 10, 114-12 Airccuft Equipment Authorization • 8900,1 Volume 4 C, Ulbri 19

3.b. Existing FAA Oversight Models for Complex ODAs

With the introduction of ODA, the LAA instituted Oversight Management Teams (OMT) for every ODA. The DMT operations and requirements are captured in FAA Order 8100-15. The fundamental purpose of an DMT is to oversee the ODA holder which includes the ODA unit. In certain ODA applications, the FAA created "customized" OMTs for companies with large and / or complex ODAs. In general, these OMTs utilize dedicated resources to oversee the volume, complexity and integration of ODA operations. Two examples will be provided.

- Boeing Aviation Safety Oversight Office (BASOD)
 - Gulfstream Aviation Safety Oversight Office (GASOO).

The DMT structure for BASOD and GASOD could serve as a preliminary step toward future oversight. The structure is designed to combine current FAA offices (ACO, MIDO, FSDD). The centralized nature of these OM1 structures enables increased commonality for ODA oversight. The dedicated resources reduce priority ODA support from competing with other FAA priorities.

3.b.1 FAA Oversight of the Boeing ODA

The Boeing Aviation Safety Oversight Office (BASOD) is the Boeing Commercial Airplanes ODA Drganization Management Team.

- The BASOD Structure comists of the BASOC Oversight and Certification Services branches, a COS representative, a FSOD representative, an AEB representative, a Manufacturing representative, and ACO specialists when performing. ODA oversight activities.
 - Program related items are reviewed with the BASOO Program Managers.
 - Technical Cross Model related items are reviewed with the BASOO Technical Project Managers.
 - Program related retained items are worked within the Airplane Program Compliance Team (APCT).
 - The structure also has dashed line support for manufacturing, continued operational safety, AEG, and Flight Standards activities.
- · Oversight
 - The BASOD Oversight branch manages all oversight: enforcement, supervision, inspection, voluntary disclosure and notification.
 - The Boeing Company has seen a growing volume of BASOO oversight since becoming an UDA.

Organizational Charts can be found in Appendix #

Page 20

3.b.2 FAA Oversight of the Gulfstream ODA

The Gulfstream Aviation Safety Oversignt Office (GASOO) (a Gulfstream ODA's Organization Managament Team.

- The GASDO Structure was driven by a cross regional, enterprise implementation of Guilstream ODA.
 - Single QDA unit /procedures manual covering multiple sites (FAA regions) with TC, STC, MRA, PC, PMA OBA authorities.
 - Traditional regional FAA ACO, MIDO, FSDO oversight processes needed adjustment to accommodate complex DDA;
- Program, and functional, and compliance roles are defined.
- Dversight role is growing
 - The GASOD organizational structure enables a single FAA Program Manager to oversee the company's STC activity organizational structure enables and business units. Dedicated FAA TC Program Managers enable priority support for Gulfstream's new and amended TC projects. A didicated Delegation Program Manager enables single oversight of ODA designee management, ODA procedures and audit / corrective action management activities. The structure also has dashed line support for manufacturing, continued operational safety and flight Standards activities. All of these TAA support activities report up to the GASOD manager.
 - The GASDO depends on dedicated part time support from FAA technical specialists, pilots, inspectors, etc. Having the formal dedication documented helps with priority ODA support, there remains competing priorities at this working level. In addition, sometimes the competing priorities come from different Guilstream DDA areas and / or Guilstream company support inquirements.

Organizational Charts can be found in Appendix 8.

3.c. Opportunities with Today's Oversight

Before describing the desired oversight model for a certificated design organization, it is important to consider opportunities for improvement in the overent DDA oversight model, and ensure that those are addressed in the future state. Based on information shared within the ARC, opportunities exist in the following areas:

- Risk based volume / frequency of oversight there is a high degree of inconsistency in the volume or frequency of oversight performed on an ODA holder. One ODA holder received over 200 supervision inquiries in the past year, while others only received an annual visit. Furthermore, there is no apparent correlation to risk or performance in today's QDA oversight model.
- Transition to Process focused oversight approach ODA oversight today is often based on a product based review. For example, "please provide the project folder to the DIMT when available for supervision review", or "please make available the following project folders for the approxima DOP impection". This approach can undermine the integrity of the process system, and lacks the benefits of a more systemic approach.
- Consideration for self surveillance Another opportunity in today's overlight system is the approach to using an
 ODA fielder's self-surveillance data. A healthy organization continually muniters itself through internal audit, as well
 as self disclosure, and non-compliance notification. An effective corrective action system within the organization
 will ensure continuous improvement of the organization and its processes. The FAA should work with an
 organization to recognize, help develop, and promote this system of self-surveillance as a measure of the

organization's performance. The FAA believes that the upen shoring of apporent violations and a cooperative as well as in advisory approach to solving problems will enhance and promate aviation safety." FAA AC 00-3

- Findings are often subjective or inconsistent A healthy compliance organization requires stability to allow
 expectise and skills in a given area or process to develop over time. In today's ODA oversight environment,
 previoutly accepted processes are often overturned based on involvement of new involvations or changes in
 interpretation. In order to maintain an effective compliance system, an applicant needs to be assured that
 established / accepted procedures will remain valid, unless compelling safety reasons noomsitate changes.
- Fully integrated oversight organization Linder ODA today, most OMTs are made up of a loose association (not directly reporting to the OMT leader) of ACO, AEG, MIDO and FSDO resources. This often results in inefficient resolution of cross organizational issues, poor communication, and lock of accountability. The future oversight organization should integrate around a common systemic (process based) approach to organizational oversight for design, manufacture and operational activities.
- More concise definition of oversight OMTr have developed oversight processes & procedures without robust
 national guidelines / standards or braining. Many ODA manual procedures, project management, audit / corrective
 action management processes, and designee appointment processes were created with a level of OMT memine limit
 judgments.
- More concise criteria for ODA autonomy ODA autonomy for delegated hindings, delegated projects, designee appointments / expansions, root cause / corrective action have somewhat loose criteria for "acceptability". In other words, an acceptable root cause analysis, an acceptable corrective action plan, an acceptable designer appointment process / package, etc. may be determined by OMT member best judgment versus a national standard. Without a national standard for "acceptability" criteria, there is and will be different standards and potential "unlevel playing fields" for many ODAs. "Recognizing the balance between prescriptive national requirements and flexibility of general guidelines, there is an opportunity to establish a level of orteria for inductry."
- Infrastructure enhancement to better assuring oversight success It is not apparent there are measures, merrics, or incentives for today's ODA oversight to produce the desired nutcome of reducing FAA resources on fourtine certification functions to enable resource reassignment to aviation safety enhancements. The cultural adjustments required to improve the success should address infrastructure issues like incentives, training, skill management and metrics designed for the desired behavior.

A. Future Oversight Model

This section will provide the Warking Group's recommended elements for the future aversight made.



The recommended future oversight model is fundamentally comprised of assessment and surveillance. Figure 1 depicts the general oversight concept. The concept is presented graphically by a company capability over time, with certain capability "thresholds". It is envisioned that in the future the FAA will require minimum organization capabilities to justify application. Those capabilities and expanded capabilities will be determined by an FAA oversight assessment methodology. Similarly, capability requirements to become a Design Organization will be defined. The same FAA assessment approach will be used to determine whether a company meets the Design Organization qualifications. At that Design Organization company develops additional capabilities that are further assessed by the FAA, they may gain additional privileges. In between the application, Design Organization certification, and expanded privileges, the future FAA oversight will be surveillance of the company's performance.

As a point of reference, in the figures depicting the actual oversight models, any time there is an assessment or surveillance in the figure, it represents an FAA oversight touch point.

The future oversight model concept Figure 2 introduces oversight for design / manufacturing organizations, product / article, and Post Certification (Continued Diperational Safety). Each of these areas will be reviewed in detail. The threeareas introduce level of capability (organization), level of project involvement (product/article), and level of surveillance (post certification).

Page 23

A.a. Oversight Key Areas

This section will present the recommended oversight models. The oversight basis will be risk based / performance based. The future oversight model addresses three key areas while striving to maintain a single oversight preserve by the TAA. The three key areas are:

- Organizational: Transition from traditional show / find compliance to organizational performance based oversight model (see section 4b).
- Product and Articles. Transition from FAA's traditional role of direct project involvement to a performance based project oversight model (see section 4d).
 - FAA participation will be limited to Level of Project Involvement (LOPI).

Post Certification (Continued Operational Safety): Transition from today's (raditional Reactionary to a systemic (process based) surveillance model (see tertion 4c).



Figure 2 depicts the three areas (1-Organizational; 2-Post Certification (continued operations); 3-Product and Article) in a "swim-lane" chart. The top half of the chart shows the organizational and post-certification (continued operations), while the bottom half presents the product and articles "swim-lane". Each of these oversight areas will be described in more detail in this chapter.

The future FAA oversight system will be based on two principles: Performance Based Oversight (PBO) and Compliance Based Oversight (CBO). Each oversight principle has advantages and disadvantages and collectively they balance the safety performance goals. PBO has greater advantages than CBO. PBO focuses FAA resources to areas of higher risk in the aviation system and moves the FAA from a total dependence on compliance findings, auditing and inspections to a more effective approach of monitoring safety and compliance performance data from the aviation industry. CBO is at

point of manufacture / integration vs storefront, while PBO can be virtual. Increases in compliance, conformity and safety performance will adjust the traditional Compliance Based Oversight activity and frequency.

A fully functioning PBO system provides risk based prioritization of FAA resources to areas of high safety and compliance risk. It would ensure that acceptable levels of safety and compliance risk are not exceeded in the system. Companies with high levels of safety and compliance performance would see reductions of FAA LOPI and oversight. The FAA can then focus its LOPI and oversight on companies with lower safety and compliance performance. In addition, the performance based oversight system would be more proactive vs reactive by emphasizing the use of data-driven decisions before an accident occurs.

PBO establishes performance objectives, measures and expectations, and focuses on results rather than prescriptive requirements. PBO assesses and monitors safety and compliance performance using various indicators. PBO relies on a certain amount of trust and accountability of data being assessed; it fosters improved and timely communications between the company and the FAA; it encourages a company to maintain and continuously improve its safety and compliance performance. Capability of a company is another determining factor in PBO (experience, training and systems in place).

Performance assessments include ongoing monitoring and analysis of performance data. Section 5 will introduce the assessment methodology approach for initial and expansion assessments. In addition, a prototype assessment tool has been developed (reference Appendix C). This tool could also be used for: Periodic audits, evaluations and inspections to validate the performance when necessary. Good and reliable performance data from the company will influence the frequency and level of FAA oversight. The FAA will scale the kinds of performance data collected from each company.

Performance data has target goals that are mutually agreed to by the company and the FAA (vitals show health of company's compliance, conformity and safety). A company maintains its system to move in the direction of acceptable / better performance. The FAA and industry need to develop recognizable performance indicators to be used (see recommendation). This would enable selection of performance indicators best suited in establishing the safety and compliance health of a company. The types of indicators used would be added, deleted or adjusted based on acceptable performance trends. The FAA would also identify minimum types of performance data the FAA should monitor that is risk based.

Recommendation: The oversight working group recommends a dedicated effort with industry and FAA to develop recognizable performance indicators prior to implementation of the new oversight model. Consideration for the effort would include review of an existing documentation such as the 'Strategy and Framework to Manage Safety Performance in AVS', reference Appendix D. In addition, the development of the performance level indicators needs to consider the EASA approach which intends to assign a performance level to a company based on performance parameters from its surveillance of the company's organization and as well as its involvement in projects activities . This is captured in EASA 'Embodiment of LOI and SMS Requirements into Part 21', reference Appendix E.

4.b Design / Manufacturing Organizations

The design / manufacturing oversight model for system / organization is identified in Figure 3. Any time there is in assessment or surveillance in the figure. It represents an FAA oversight touch point. The D&M applicant has a level of capability or is looking to expand that capability makes a declaration of capability to the FAA. The FAA makes an assessment of that capability (see section 5) and ultimately acknowledges the organization's capability through certification. The result is a D&M certificate holder with specified capabilities. Capability is shown and determined based on the product (e.g. aircraft, engine, propeller) and approval types (e.g. STC, TC, PMA). There may be instances where an existing BO would like to expand their capabilities to support their existing approvals. An example of this could be an Engine TC DO (part 33) wanting to expand an authority into 14 CFR 25.903.



Figure 3

Note: The DO concept does not address manufacturing privileges yet the single FAA oversight model will address both design and manufacturing, and MRA aspects of the business organization. The DO defines requirements only for design aspects of SMS, CAS, and QMS. Additional rule changes will be required to minimize FAA manufacturing involvement in DO activities. For example, most ODAs today have delegated authority to issue routine Standard Airworthiness Certificates. Without specific regulatory provisions, a DO would not have the ability to perform this government function. The solution could be to include specific regulatory provisions for the retention of select delegated functions under a certificated DD company. See section 4.e for the expected manufacturing level of project involvement with the current definition of design organization privileges.

Capability is shown / determined by the experience of the organization, or the demonstration of capability by existence of adequate processes and their use to conduct projects with FAA involvement prior to BO. Additional third party suppliers can be used to achieve capability not resident within the DO.

FAA oversight consists of the organization processes that include DC required processes (SMS, CAS, DMS) as well as manufacturing and MRA processes. As discussed, the FAA will focus on the process evaluation vs individual product / article compliance.

4.c. Product / Article

The design / manufacturing oversight model for product / article certification is identified in Figure 4. The key item is the Level of Project involvement (LOP). The two extremes of LOP) are "()" or manufacturing from the FAA to "1" where the FAA has to perform a governmental function.

Page 28

Figure 4



In the case where a certificated DO has previously demonstrated their engineering / design capabilities and they are not seeking to expand that capability, the LOPI should be "0" outside of any required governmental functions and the expectation is that the design organization has the autonomy to complete all certification efforts. Once these efforts are complete, the design organization makes a declaration of compliance which does not require an assessment by the FAA before it issues the certification / approval to the DO for that product / article.

In cases where special conditions, equivalent level of safety, issue papers, alternate methods of compliance, or any other governmental functions are required, there will be a LOPI by the FAA. The intent here is that a LOPI is required but should minimize impact to the DO's critical path activities.

In terms of oversight, the expectation is that regardless of the LOPI, surveillance of the DO is the primary function performed by the FAA. Note that DO activities that required LOPI for a specific product / article can be used as demonstration of capability for future product / articles efforts.

4.d Post-Certification (Continued Operational Safety)

FAA and certificate holders will transition from today's traditional reactionary to a systemic (process based) surreliance model. The tevel of Surveillance (LOS) will be dependent on processes for hazard identification, criticality of products, and a risk based safety criteria determinations. The process shall include outputs for corrective actions to the FAA (58/s, AU, SAIB, etc) (*One has to have parameters in place to be careful on the imposing of corrective actions such that they do not impose added costs to the operators, i.e. improved part *[.



Hazard Identification will include monitoring and trending safety analysis data with multilevel inputs by incorporating a data driven risk based approach for safety assurance and safety risk management. Current governmental and holder processes should be evaluated for effectiveness and readiness. (MSAD, BOEING Process, Bombardier, etc.)

Hazard Identification inputs

- Service Alert Notifications (Declared Emergencies)
- Service Difficulty Reports, Incidents, and Accidents
- o Flight Test
- · Manufacturing
- o Audits
- Non Compliance
- Anonymous reporting



The expectation is that a DO with a functional SMS has the responsibility to propose the corrective actions required to address safety issues to the FAA. For example, a DD could make a request to the FAA to release an AD based on their risk assessment.

4.e LOPI for Manufacturing

LOP(has been defined for the engineering activity in section 2.c. The following identifies and defines the potential accas for level of project (manufacturing) involvement depending on the performance of the organization as discussed m section 4.a.

Type Certification - Areas where agency manufacturing inspection may be involved during a TC program:

- Conformity inspections critical articles, parts and installations on flight test aircraft (an worthiness and production certification risk areas).
- Conformity inspections developmental (proto-type) layouts / drawings that turn into final production drawings
- Undue Burden decisions for prototype articles manufactured at new production locations. Mitigation of burden may require conformity inspections and FAA audits. Sec 21.43, Sec 21.122, 21.139
- I light test sizeraft:
 - Establishing experimental certification requirements.
 - Issue experimental certificates, R&D and Show Compliance.
 - Issue Special Flight Authorizations (SFA) for foreign registered test aircraft.
 - Coordinating and approval of flight test locations and operating limitations.

Review and approval of design change procedures (configuration control) that are part of the 21.157 quality system:

Progressive conformity inspections of flight test aircraft that will be presented for Standard Airworthiness Certificate after type certification

Production certification - Areas where agency manufacturing inspection may be involved during a PC/PLR project:

- Inspections for product manufactured under a Type Certificate Only. Inspection or test at a supplier facility, necessary to determine compliance. Sec 21 123.
- Conformity inspections of developmental layouts or drawings into final production drawings
- Production locations approvals (Undue Burden decisions may require conformity inspections and FAA audits) Sec 21,139.
- Concurrent production conformity inspections on prototype products during TC projects.
- Production Inspections of design changes during manufacturing.
- Review and approval of quality system changes during TC projects.
- Review and approval of MRB procedures, design change procedures, and production flight test procedures.
- Inspection of quality system, articles, witness test to determine compliance. Sec 21:140 and 21:147.

Airworthiness Certification - Areas where agency manufacturing inspection may be involved during a TC/PC project:

- Progressive conformity inspection of flight test aim aff presented for Mandard Airworthiness Certificate
- Establishing experimental certification requirements
- . Issuing Experimental certificates, R&D, Show compliance, crew training.
- Issuing Special Flight Authorizations (SFA) for foreign registered test aircraft
- Coordinating and approval of flight test locations and operating limitations.

5. Oversight Assessment Methodologies

5.a. Initial Assessment

The initial assessment (referred to as approisal in the 2008 CDO ARC Report, see section IV, 1,1 of a Design Organization (DO) should be an evaluation of the organization's ability to satisfy the requirements for a DO. The initial assessment establishes that the organization meets the basic requirements for the authority and has process capabilities in place in accordance with the requirements for a DO. The effectiveness of that organization in ensuring compliance, safety of its products, and managing safety risk will be assessed by performance measures as discussed in section 4. These performance measures will determine the frequency and depth of orgoing surveillance activities (see section 5.6).

The initial assessment is intended to establish whether an organization has the required process coverage and the required level of process maturity. There are several models in use across industry to assess process maturity. In April 2012, the Safety Management International Collaboration Group (SM ICG) published the Safety Management System Evaluation Tool' as an objective method "to indicate the expected standard of on organisation's SMS in terms of compilance with the SMS regulation and its performance to effectively manage safety risk". This tool provides an internationally harmonized standard for assessing process maturity, and can be expanded to include additional regulatory requirements for a DO. The report can be found on the SM ICG 'SKYbrary' page, at http://www.skybrary.aero/index.php.

The Oversight Working Group has expanded the SM ICG SMS Assessment Tool to include requirements for a Compliance Assurance System and Quality Management System. These components, SMS, CAS, and QMS form the basic components of a Certificated Design Organization. In advance of final requirements for CAS and QMS, the working group has used the draft proposed regulations from the 2008 CDO ARC Report. The working group presents this 'prototype' tool in Appendix C, as an example of what an assessment tool could look like. Before endorsing this as a tool for broader use, FAA and industry should revise the model to incorporate lessons learned from recent applications of the ICG SMS Assessment tool.

In the 2001 CDO ARC Report, the ARC recommends that "an organization applying for a CDO certificate or an expansion of its existing certificate undertake a self-assessment. This self-assessment should be a formal undertaking with records generated of the findings and observations of the evaluators." The self-assessment should apply the same evaluation tool as the FAA assessment, however, the FAA may consider the extent and thoroughness of the self-assessment in developing their plan for assessment. Upon satisfactory completion of the self-assessment, the candidate organization may submit application for Design Organization to the central FAA DO office (see recommendation below).

The application process (preferably electronic / on-line), will be defined by FAA policy and should consist of:

- Completed application form
- Written request for assessment
- Identification of requested capabilities
- Enclosure of self-assessment

It is the recommendation of this ARC that the FAA Assessment Team report to a centralized FAA DO office rather than placing that responsibility on an ACO. The ARC evaluated various options as shown in Table 5-1, below.

Table 5-1. Oversight Management Option Evaluation

Oversight Management	PROs	CONI
Local Surveillance and Assessment	 Team familiar with the products, the company, and the organization's performance to determine level of capability Some familiarity with DO's processes 	 Difficult to charge the existing culture to systemic oversight mode! Lack of standardination of the oversight across DO's Variations in DO Manual Content Additional workload lovied on strained resources.
Local Surveillance / Central Assessment	Team familiar with the products, the rampany, and the performance of the organization to determine appropriate surveillance (L5) Some familiarity with DCP's processes (L5) Consistency in the assessment of organization seeking DD privileges (CA) Unblance Assessment form an untravioled ream of specialized DD system auditors. (CA) Amily to gauge relative performance screes multiple DC applicance (CA)	 Difficult to change the everying culture to systemic oversight model (LS) Lack of standardization of the oversight acress DO's (LS) Lack of tamiliarity of DO products, the containty, and the performance of the organization prior to assessment (CA) Additional workload levied on strained renources (LS)
Local Assessment / Central Surveillance	 Team familiar with the products, the company, and the organization's performance to determine level of capability (LA) Some tamiliarity with DD's processes (LA) Consistency in the surveillance of organizations (CS) Unblased surveillance from a team of specialized DO system auditors. (C5) Ability to gauge relative performance across multiple DO (CS) 	 Lack of standardization of the assessment for applicant/expanding DO's (LA) Variation in DO Manual Content (LA) Additional workload levied on strained resources (LA) Lack of familiarity of DO products, the company, and the performance of the urganization prior to beginning, surveillance (CS) Dom not address oversight when there are targets of opportunity during project work (CS)
Central Surveillance and Assessment	 Unbrased surveillance from a least of specialized DO system available. Allows ACPs to for us in on Safety Printifie Overlaght completed in OO Folicy Events (no interpretations) Gate/experts of EVI Manual Content Data onalysis Scoup for Continuous System Improvorment work forward (EnA) and eliternal (DO) 	 Dues not address oversight when there are targets of opportunity during protect work Lack of tamtianty of DO produces, the company, and the performance of the organization prior to Sectoming adrivellance

Once the "Applicant DO" has applied and requested capabilities as defined in the DO regulations, the FAA will conduct an initial Assessment. This will include

- Inspection of the "Applicant DO" for compliance to the regulatory requirements
- Evaluation of the "Applicant DO's" processes abilities to meet the regulatory requirements with the
 respect to the canabilities requested.

An effective assessment will both *validute* that the organization has procedures and supporting processes in place to satisfy the DO requirements, and *verify* that the processes are followed by personnel who meet the qualifications defined in those processes. The validation step can be described as "*did you build the right product*?", and the verification step can be thought of as "*did you build the product right*?"

It is important to ensure that both steps are effective. For example, an applicant needs to rely on the expectation that if people in the organization follow approved processes, then the result will be compliant products. This requires effective validation of the process during their development and their initial assessment. Effective validation ensures that all applicable requirements are allocated to the procedures (no requirements are missed), that requirements captured are valid (based in regulatory requirements), and that supporting processes define clear, unambiguous steps to be performed (see definitions of verify and validate in section 2.c.).

Furthermore, the industry needs to rely on the expectation that a certificated design organization will meet their obligations, and ensure that qualified personnel follow their approved processes. This includes the individual performing a self-assessment of their own work, the organization performing self assessments of their own work, the organization performing self assessments of their own activities, and the FAA performing ongoing surveillance activity of the design organization.



Figure 6 DO Validation and Verification

The result of the initial assessment is a determination that an organization either does or does not have adequate procedures and the necessary level of process maturity to ensure compliance and manage safety risk. This requires validation of the organization's procedures and supporting processes, as well as verification that those processes are followed. This assessment can be accomplished using a process maturity approach, combined with a risk / performance based surveillance program.

5.b. Certificated Design Organization (DO) Surveillance

As part of their fundamental oversight responsibilities, the FAA will conduct surveillance on the organizations to evaluate performance through inspection. The surveillance conducted by the FAA will utilize a systematic approach that will focus on the validation of the processes/procedures by means of inspection, and a verification of the organizations capability of following their procedures through an evaluation of the products and/or approvals that are the result of the system. Surveillance could include (but is not limited to):
- Review of the DO Process / Procedures within the Operating Manual,
- Review of the DO self-assessment / self-surveillance data and corrective actions,
- inspecting the DOs for compliance with their Process / Procedures within the Operating Menual.
- Reviewing the work performed and Evaluating performance for quality assurance.
- Ensuring that required training has been completed,
- Providing constructive feedback, and
- Taking intractive action, as necessary.

Surveillance does not include program specific involvement; it is based at the system and process levels. The FAA will still have the responsibilities that are contained in governmental functions. However, the ARC has termed these actions as Level of Program Involvement (LOPI) and will be the tasks required of the FAA office in charge of the respective program and not within the job duties the Oversight Office (See Figure 4).

Oversight Team

The DO Oversight Team will be a ream of individuals that have a corresponding role with the organization. For example, if a company holds design, manufacturing, and/or repair certificates, the Dversight Team will consist of Engineers (ASE), Manufacturing and Flight Standards Inspectors (ASI), and Aircraft Evaluation Group (AEG) members in order to parallel the DO's capabilities.

It is the recommendation of this ARC that these FAA Oversight Teams report to a centralized organization. As shown providesly in Table 5-1, the ARC found that establishing a central oversight organization will

- Achieve standard surveillance practices,
- Centralize policy responsibility ensuring consistency in interpretations,
- Allow the ACO to focus on safety critical functions,
- Provide "3th-party objectivity" as the office does not work programs with the DO,
- Provide a single source/repository for the oversight data which will drive the risk based modeling controls,
- Manage skill development practices for surveillance staff.
- · Allow for a highly trained staff in system surveillance, and
- Provide a single source for oversight of corrective actions.

Surveillance by 3rd party independent organization within a DO is acceptable as a means of validating or performing a DO's own self-audir / surveillance.

Figure 7 depicts the different offices and their roles, responsibilities, and actions with respect to standard applicants and design organizations.





Surveillance Scope

As mentioned previously, the surveillance activities are not program specific, but at a process, system level. Therefore, there will be no specific criteria contained in policy by which the FAA will inspect and evaluate the organization. The oversight team will base their surveillance on the procedures contained within the organization's operating manual, the adherence to the procedures/processes, and evaluating the outcomes from the process for quality.

Risk Based Decision Making

The interval and depth of surveillance activities should be based on the safety risk of the product on article, capability of the organization, past performance of the organization, complexity of the programs, and the authonity/privileges granted to the organization. First, an organization's self-surveillance activity, including their performance in addressing voluntary disclosures and notifications of non-compliance, is an indication of a closed loop system that will ensure continual improvement of the organization and address lessons learned. In addition, as discussed in section 4, an organization's performance, as indicated by the factors below, will determine the level and frequency of surveillance.

- Capability
- Past Performance
 - ADs / Safety Findings on approved products
 - . Quality Escapes
 - Non-compliances
 - Self-Surveillance Findings
- Complexity
 - New and Novel Technology with respect to current processes
 - LOPI from ACO: If the program has a high level of involvement from the ACO, is it a good candidate for oversight?
 - The ability to manage and maintain control over large programs
 - Process Robustness
- Multiple Authorizations/Privileges/Certificates

5.c. Expansion Assessment

When a DO requests an expansion to the scope of its authority that involves substantially new processes that have not been previously demonstrated, the DO must demonstrate that it is capable of operating under its proposed processes prior to receiving an amended certificate with the expanded capabilities. As a tool in the performance of the self-assessment and the subsequent FAA evaluation, the DO applicant should have traceability between its processes and the regulatory requirements they are intended to address (see Figure 6).

Solf-assessment: The application for expanded scope would include a self-assessment to show its readinem to function with the new processes, and exercise its capabilities to make determinations under the changed scope. The self-assessment should be a formal undertaking with records generated of the findings and observations of the evaluators utilizing the same process/performance measures used for the initial assessment (reference 5.a.). This assessment may be abbreviated based on the delta between regulatory requirements with respect to the additional capabilities requested. The FAA may use this assessment in focusing its assessment activities.

*AA Assessment: Following DO application for expanded scope, the FAA will perform an assessment based on the criteria used for the initial assessment (reference 5.a.) to determine the DO has shown that it is fully capable of operating within the changed scope.

- In the case of an expansion in scope for a DO in good standing, the FAA may rely on the self-assessment in issuing the expansion of the DO certificate.
- If the expansion in scope is minor, applicant DO self assessment may be sufficient to allow the FAA to expand The certificate scope with no further demonstration
- The use of the applicant's self-assessment to adjust the scope of FAA activities is solely at the discretion of the FAA and should follow the safety management principles of targeting safety critical efforts
- The FAA is under no obligation to complete its assessment within a minimum time limit or number of projects.

6. Future Oversight - Assessment Model Implementation

6.a. Implementation Boundaries

The implementation approach covers the portion of the industry located above the DO threshold. This approach addresses the implementation of the Organizational and SMS elements as applicable to a Design Organization. The current underlying assumption is that there is no change for the portion of the industry below the threshold.

6. b. Implementation Assumptions

Implementation covers global aspects of DO Implementation:

- Implementation plan uplit into Industry specific actions and FAA specific actions.
- Implementation plan for industry and FAA will be limited to above the DO threshold.
- Implementation plan will address the transition from a "Mature ODA" to a DO
- Implementation plan will not address the production and maintenance, including the applicable ODA Junction codes

Implementation plan for FAA apply to the different FAA lines of business having oversight responsibilities, under the DO oversight model, for design organizations, design projects and fleet activity.

- For design approval holders and applicants below the DG threshold, confirmation of the requirements from the other WGs needs to be provided before determining need for extending the scope of the implementation plan. The current assumption is that this oversight system of this portion of the industry does not change.
- + DO/SMS
 - Not all design approval holders will be required to be DOs
 - DOs require an SMS, SMS will be evaluated at Initial DO assessment and during DO surveillance.
 - Some non-DOs may require SMS. Non-DOs may elect to develop a SMS without a certificate or privileges and it will reduce risk indicators and influence the level of direct oversight by the EAA.

DO will only be in effect after new application above the DO threshold-

- Delegation
 - DOs will not have a delegation for design approval activities (DMIR, ODA, DAR may be used for production / maintenance - see section 4.b.).
 - Non DOs may have a delegation or delegated persons
- The FAA Dyetsight approach covers 3 major areas.
 - System / organization
 - Products, articles certification activities
 - Post-certification (COS)
- TAA will perform direct Oversight of all DOs. TAA will not delegate direct oversight of a DO to a third party.

- DO and SMS WG defines the need for independent auditing / monitoring that may be carried out in-house or by a third party. The extent of independent auditing may influence the level of direct oversight by the FAA;
- FAA issues DO certificate_DD perform product / articles certification process with minimal FAA involvement
- Central FAA Oversight organization performs organizational assessment & surveillance in Integration / coordination with local FAA field offices.
- Engineering LOPI only at local ACO level. Manufacturing LOPI will be performed by agency manufacturing implements (e.g. MIDO, ODA, etc.).
- . The minimum qualifications for DAH will be defined. The PAH minimum requirements remain the same.

6.c. Implementation Approach

Three Major Transition Steps:

- 1. Proof of concept projects Pre-implementation
- 2 FAA Transition Plan
- Industry Transition Plan.
- Note: A phased-in approach to implementation may be effective where regulator/industry defines organizational regulation regulation and applicants show compliance to those requirements on an "as ready" basis

6.c.1. Proof of concept projects

- · Objectives.
 - Ensure through proof of concept plans, that the requirements proposed by the Part21 / SMS ARC are
 practical to implement and achieve their effectiveness.
 - Determine If the transition from "Mature ODA" to DO has benefits to TAA and Industry and therefore should be executed or not.
 - Validate the DO elements and get operational experience on LOPI and fleet monitoring.
- Who will be involved?
 - Digaminational (including SMS).
 - One large and experienced company and one medium-size company above DO threshold.
 - Products and Articles certification (I.DPI)
 - Central Office of Oversight and FAA field offices of selected companies involved in both organizations assessment /surveillance and Products and Articles certification.
 - Post-certification (COS) / Neet activities
- How to run the simulation?

Organizational (locluding 5M5 / Pleef activities) requirements

- I" phase:
 - DO test-companies are requirements applicable to the simulation.

- DO test-companies perform gap-analysis, implement missing elements, conducts selfassessment and contract the FAA (central office + field offices) when they are ready for assessment by the FAA.
- · 2" pitase:
 - FAA conducts assessment and when satisfied, grants letter of authorization to the DE test-companies to work on projects as a "DO".
- T phase:
 - FAA / DO test companies perform routine organization external/internal oversignt. See figure 9 "applicant types diagram".
- Products and Articles certification (LDPI)
 - 1 phases FAA consults Project test-companies to develop guidance material on LOP((DO testcompanies should be part of this group)
 - I phase: Simulation on selected projects
 - 31° phase: FAA review and uses feedback to develop fine-tuned guidance materials
- Benefits, Metrics

To be defined for pass/fall. Mature ODA needs to be defined in order to set the baseline (see below). Capture any cost, resource, timing or autonomy benefits.

Use metrics lessons learned to develop implementation metrics.

6,c.2. FAA transition plan

- Objectives: identifying how the FAA can transition from its traditional role of project involvement to a role of coach and mention with system oversight responsibilities.
- 2. Transition principle:

FAA needs to be ahead of the industry to be ready to support industry transition to DO (on an "as ready basis) and project work (LOPI).

- Develop performance measures prior to implementation.
- FAA should not release a limal rule before FAA has implemented the intrastructure (training, policy change, incentives, organizational changes, metrics, etc.) to better ensure the necessary cultural shift to perform system oversight.

"Mature ODA FAA Oversight" is defined as the baseline for determining the DD benefit metrics for the industry and the FAA Level of FAA oversight appropriate for a Mature ODA is considered as the minimum criteria before moving to DD oversight. "Mature ODA oversight" may be understood as FAA ungagement being limited only to inherent governmental functions, with ODA holding responsibilities intended by the ODA regulation.

- 3 Transition methodology
 - FAA Office of Rulemaking
 - · Review axisting regulatory and guidance material for consistency and revision as requirad-

Page 43

- Publish necessary policies for pushing the cultural changes.
- Apply lessons learned from proof of concept and incorporate into future rulemaking.
- Release or issue new rule(s).
- FAA Central Dversight organization
 - Formalize roles & responsibilities including coordination with local offices (Orders)
 - . Complete training and qualification of staff (Organization and Projects oversight)
 - Insure effectiveness of FAA's surveillance processes, procedures and tools.
 - Develop metrics that measure the desired outcome / results / behaviors for both regulators and industry. For FAA, measuring effective auditing, LOPI disruption. For the DO, measuring effectiveness of corrective actions.

Local Field Offices

- Roles & responsibilities well-established including coordination with Lentral Eversight organization.
- Complete training and qualification of staff.
- T Ensure effectiveness of LOPI and COS Surveillance Processes, Procedures and Tools

FAA International Branch

 Bilateral agreements and associated Technical Implementations Procedures should be updated based on new DO system.

6.c.3. Industry transition plan

- Objectives, identifying how an organization could establish the systems required of an approved organization while still working as a non-certificated applicant or a delegated organization
- Transition principle:
 - FAA publishes final requirements with a cut off date for industry to implement them. FAA published guidance materials using lessons learned from proof of concept projects.
 - Applicants above timeshold (e.g. TCH) demonstrate compliance to those requirements on an "a+ ready" basis
 - Other DAH below threshold may apply for a DO un a voluntary basis.
- Immition methodology:
 - Drganization processes.
 - Identify and implement needed organizational changes: Structure, Roles and responsibilities, Procedures & processes, Documentation, List of changes depends on current system (e.g. ODA vs. Delegates) and DO Objectives (Privileges / Functions).
 - (bentify and develop necessary skills and competencies (e.g. Training, 3" party utilization, etc.)
 - Bon the preliminary steps before application for a DO (Implementation of requirements, selfassessment).

Page 44

- Product / Articles (LOPI) and Fleet activities:
 - Hon projects under LOPI principles, as soon as DO certificate is word by the LAA; IAW Limitations and Privileges defined in the DO certificate

6.d. Implementation issues Requiring Full ARC Coordination

PBD / CBO Metrics at the DO system Level

- · Metrics remain to be developed jointly by FAA and Industry
- · What metrics should be established and how would they be applied consistentily?

- DO needs to have the ability to issue certain certificates

- Recommend the ARC put together a team to evaluate the production aspects of a DO and the ability to issue certain certificates (ie Airworthmess Certificates)
- This learn also will evaluate 5M5 benefits to production

6.e Applicant Types in "Design Organization World"

Figure 8 is presented as a concept to identify the various applicant types in the future state including Design Organizations. The checklist represents the QMS, SMS, DAS / CAS requirements to be met to qualify for the desired level of certification.



Figure 8

6.f Implementation Evolution

Figures 9 -11 and Table 6-1 capture the concepts and details of future oversight.

Oversight Today:



Oversight tomorrow



CBO/PBO Chart





FAA oversight of a CDPO and ODA includes SMS, Quality system and Design System. It shows FAA oversight is predominately performance based and full delegation. As the CDPO system matures the FAA oversight and ODA functions will decrease. When the CDPO reaches full maturity and with acceptable safety risk the FAA oversight is significantly lowered and delegation is reduced to only those functions that can only be performed by FAA or designees, i.e. issuing experimental certificates or airworthiness certificates. The FAA can concentrate more on high risk companies.

Page 49

- Future oversight table ("4" Column")

Table 6-1. Proposed FAA Certification and Oversight Functions

ACO Functions	MIDO Functions	FSDO / AEG Functions	Future State ACO/MIDO/FSDO/AEG
 Design Certification Activity TC/STC/TSO/PMA applications - 8110.4, 8110.42, 8150.1 RBRT system - 8110.4 Issne papers - 8110.4 Undue Burden 8100.11 Certification basis - 8110.4, 8110.48 Operational satiability - 8110.4 BLOS - 8110.4 DCA - 8110.54 Commercial Paren 8110.118 Certification and conformity plans PSCP & C1P - 8110.4 Deposition of deviations - 8110.4 Finding compliance - 8110.4, 8110.51 Review substitution data 8110.4 Finding twitter 8110.4 Witness lest - 8110.4 Witness lest - 8110.4 Flight twitter 8110.4 TLATTR - 8110.4 PMA Late fonoted Antelse - 8110.42, 8120.22 Issne TC/STC/TSOA- IGF except STC - 810.4, 8110.42, 8150.1 	 Design Certification Activity RERT system - \$130.4 Certification and conformity plans PSCP & CIP - 8110.4 Pinding compliance - 8110.4 Conformity Inspections - 8110.4 Review process specifications - 8110.4 Witness test - \$110.4 Witness test - \$110.4 TLA/TIR - 8110.4 NDT evaluation - 8110.4 Teardown mapeerions - 8110.4 Configuration control of design changes - 8110.4 Airworthaness certification for flight resting (R&D, Show Compliance, SFA1 - 8110.4, 8130.29 Periodic Safety checks for Bight test arcmit- 8110.4 	 Design Certification Activity by AEG ICA - \$110.4 Operational and Maintenance Eviduations - \$140.4 ThA Olgh tass-\$110.4 Flight Standardization Board - \$110.4 Maintenance Review Board - \$110.4 Flight Operations Eviduation Board - \$110.4 Type Ramps for airpain -\$110.4 	 Design Configuration Activity Contine Burgleri, ACO, MIDO Configuration basis, ACO RESEACO LOPP 4020 Commercial purity ACO, MIROO Issue TCSNC T2824 A.W. complement 52:4, R&D, Menne Complement Flight Standardment Brand ATO 4 Manutements Responstitution found Algor Operations Evolution found Algor 4

ACO Functions	MIDO Functions	FSDO / AEG Functions	Future State ACO/MIDO/FSDO/AEG
Design Corrification Oversight • Investigations - IGF - \$020,11,2150,3 • Voluntary disclosures - 2150,3 • Special Technical Andus - IGF - 8110,4 • COS: • Service Difficulty Reports 21 V 8110,107 • NTSB recommandations - 1220,2 • Issue AD5 - 8040,4, S040,5,1R- M8040,1 • AMOC - 81(0,105 • MSAD - Failures, malfunctions, defects - 8110,107	 Design Certification Dversight Special Technical Audits - RiF - 8110 4 COS: Service Difficulty Réports 21.3 - 8110.107 MSAD - Failures multimetions, delects - 81100107 	Design Cortification Overview N/A	 Design Cortification (Newsight Investigations (ICF + 8020.1) 2150.3 5 Voluntery disclosures - 2150.5 Special Technical Audits (2010 - 8110.4) COS: 5 Service Difficulty Reports 21.3 8110.10⁸ COS: 5 Service Difficulty Reports 21.3 8110.10⁸ VISB recommendations 1220.5 Issue ADS - 5040.4, 2010.8, bb M6040.1 MEGO - 8110.103 MEGO - 8110.103 MEGO - 8110.103 MEGO - 8110.103
Foreign TC/STC Validations • International agreements • Finding compliance -8140-52, 8110-51 • Witness test - 8110-52, 8110-51 • Issue VTC and VSTC - 8140-52, 8110-51	Foreign TC/STC Validations • International agreements © Finding compliance -8110.52	Porrigo TC/STC Validadous X/A	Foreign TCSTC Validations • International agreements • Finding compliance - 8110.52, 8110.53 • Witness text - 811052 8110.51 • Issue VTC and VSTC - 8110.32 %110.51
 Belegation Oversight Delegation to mdividual designees and designees and designations - 1100.2. \$100.8, \$100.15. \$110.17 Review and approve ODA procedures manuals - \$100.15 Delegated Organization Inspection Program (DOIP) - \$100.15 Supervisory Visits - 	 Delegation overslepti Delegation to individual designees and designees and designetions - 1100.2: \$100.8; \$100.15 Airworthinesy application package and timitations review - \$100.8 Review and approve ODA procedures manuals - \$100.15 Conformity report review individual 	 Delegation oversight Delegation to individual designees and designations – 1100.2, 8100.8, 8100.15 Airworthiness application package & limitations review - 8100.8 Review and approve ODA procedures manuals - 8100.15 Delegated Organization Inspection Program 	 Delegation aversight Delegation in individual designers - 1100.2 ×100.8 Over the shoulder reviews - ×100.8 Arrecardinases application review 8100.8 Conformity report review individual designees - ×100.8 Over the shoulder reviews individual designees - ×100.8 Over the shoulder reviews individual designees - ×100.8

ACO Functions	MIDO Functions	FSDO / AEG Functions	Future State ACO/MIDO/FSDO/AEG
N100 L5 • Review unternal andits - 8100 L5	designees \$100.8 Delegated Organization Inspection Program (DOIP) - \$100.15 Supervisory visits- \$100.15 Over the shoulder reviews individual designees \$100.8 Review internal andits - \$100.15	 (DOIP) - 8100.15 Supervisory visits - 8100.15 Review internal audits - 8100.15 AFS Designees and Designations - Volume 13 Chapters 1 thru 10 	
 Production Confinention Review new (path) system = 81/20,22 Production Flight Tesamg Procedure (PFTP) - 81/20,22 Design change procedures 81/20,22 MRB procedures = 81/20,22 Service difficulty reporting = 81/20,22 	8100-15 ation Production Confiltration (By) • OC/PMA applications - (2) • 8120-22 (Flight • Review and approve needure • Review and approve (GF - \$120-22, \$120-21) (GF - \$120-22, \$120-21) (GF - \$120-22, \$120-23) (GF - \$120-22, \$120-23) (GF - \$120-22, \$120-23) (addits - (GF - \$120-22, \$120-24) (conformity plans (CIP) - \$110-4 * Conformity plans (CIP) - \$110-4 * Witness (est production - \$120, 22, \$110-4 * Witness (est production - \$120, 22, \$110-4 * Witness (est production - \$120, 22, \$110-4 * Undits - Burden - \$100, 11 * Commercial Parts: \$110, 118, \$120-22 * Issue PC/PMA -IGF- \$120, 12 * Issue PC/PMA -IGF- \$120, 22 * Issue PC/PMA -IGF - \$120, 22 <td>Production Contification N/A</td> <td> Pendinetion Contribution AC (Ph1) applications = \$120.22 Receive and approve new quiday wears (AFE - \$120.22, \$130.25) PC (Ph1)/13814 pre-presidential approval audits (Pr1 - \$120.22) Conformity plans (CFE) (\$106.4 Production ("unformity inspection - \$120.22, \$170.4) Writness lest production - \$120.22, \$170.4) Writness lest production - \$120.22, \$170.4) Confinement (Parts - \$170.11), \$126.22 Jonae Brashou (Pr0) 11 Commercial Parts - \$170.11, \$126.22 Jonae Production Constance Review (Pr0), (GF = \$120.22) International agreements - 1(d) (P4.11PA, 90.21-24, 21-2) </td>	Production Contification N/A	 Pendinetion Contribution AC (Ph1) applications = \$120.22 Receive and approve new quiday wears (AFE - \$120.22, \$130.25) PC (Ph1)/13814 pre-presidential approval audits (Pr1 - \$120.22) Conformity plans (CFE) (\$106.4 Production ("unformity inspection - \$120.22, \$170.4) Writness lest production - \$120.22, \$170.4) Writness lest production - \$120.22, \$170.4) Confinement (Parts - \$170.11), \$126.22 Jonae Brashou (Pr0) 11 Commercial Parts - \$170.11, \$126.22 Jonae Production Constance Review (Pr0), (GF = \$120.22) International agreements - 1(d) (P4.11PA, 90.21-24, 21-2)
Production Certification Oversight Review changes to quality system— 8120.22 Production Flight Testing Procedure (PFTP) - 8120.22 Design change procedures - 8120.22	Production Certification Oversight Review and approve changes to quality system = 8120-23, 8100-15	Production Locification Overvient N/A	Pendarthum Cortification Oversight Review and approve changes to guidity potent - \$120.23

ACO Functions	MIDO Functions	FSDO / AEG Functions	Future State ACO/MIDO/FSDO/AEG
 MBB procedures	 Audite: RISR T as each RISR T as each Prostar as the second (PID - 4120 27) Product - 8120 27) Special Technical Aodus - 8120 27 SCA (Sappley Control Audits) R120 23 SCA (Sappley Control Audits) S120 23 SCA (Sappley S120 23 SCA (Sappley Control Audits) S120 ScA (Sappley Sappley S120 ScA (Sappley Sappley Sappley ScA (Sappley Sappley Sappley ScA (Sappley S		 Interview -IGF - 2000 (* 1000) Interview above verification (2000) Interview above verification (2000) Interview of a statement of the statem
	 Volumpy disclosures - 2150,3 Corrective action verification - 2150,5 COS- Service Difficulty Reports 21,1 8120,24 		 Corrective action verification - 2150.1 6/08: Service (Officulty Repress 21.1 - 8120.21 Quality fathews, and functions
	 Quality faibures, multimetures, detects = 8179(23) 8110-107 NTSB recommendations = 1220.3 MSAD = 8110.107 		delegas =8120.23 8110.107 = 51581 recommendations = 1,220.2 = 5155111 = 8110.107
Aircraft Circlification Review operational / emergency procedures used for experimental pertificates #120.2 Review and resolution	Aircraft Centification (Origonal) • issue anyonthiness centroleanes approvaic • Export - \$130.2. AC23-23, 21-2 • Standard A/W-	Aircraft Certification (Recurrent) Eaue airworthiness certificates approvals: Esport = 8170.2 Standard A/W- 8130.2	Aircraft Cartification - Insur actron channels - Expanse approvals - Expanse 8130 2, 10:21-25, 21-2 - Standard 4 W- 8130 1 - Improve 8130 2, 11:21-21

ACO Functions	MIDO Functions	FSDO / AEG Functions	Future State ACO/MIDO/FSDO/AEG
of noncomornatives for aircraft manufactured under a TC only \$130.2 • Evaluation of noncome for essens anoma weight \$130.2 • Evaluation of mod Dual use cannots #130.2 • Special purpose operations for restricted Category #130.2	 SUMLE Import - SUMLE Special A/W - SUMLE A/W - SUML	 Import - \$130.7 Special A/W - 8150.2 #140.56, \$150.29 Special Flight Autoorizations (SFA) - \$110.7 Special flight permits - \$130.2 \$150.5 mgs used parts - \$130.2) 	21-2 Special AW + 81/10.2 81/02/6 A130.29 A130.29 Special Flight Ladorectoria (SF 2) - Ref - M10/2 Special Ref - M10/2 Special Ref - M10/2 A130.3 deep severy works/1 81/20.21
Repute Station Corrification N/A	Repair Station Certification N/A	 Repair Station Certification Repair Station (RS) Certification: Volume 2 Chapter 11 Part 145 Repair Stations Certification of RS and Solellites RS w/i USA and its Territories V2 C11 S2 Certification of RS Located outside the USA not under a MIP V2 C11 S3 Evaluating RS Manual and Quality Manuelt or Revision V2 C11 S4 Facilities and Equipment V2 C11 S5 Certification of RS located in the US applying for EASA 145 V2 C11 S6 Manuscumuc bioplementation Procedures (MIP) FAA Order 8000 85A 	 Repair Station Confidential Repair Station (R8) Confidential Contribute 2 Chapter 11 Duri 145 Repair Stations Contribution of RS and Stability (R5) is a 1783 and its Preenance 12111 S2 Contification of RS Located nutrified the USA number in 501PT 2011 S3 Evaluating RS Manual and Quality Manual on Revision V2011 S4 Fourfilters and Kaptionent V2011 S5 Certification of RS Incated in the US applement of RS Incated in the US applementation (R5) Incated in the US applementation (R5) Incated in the US applementation (R119 T.11) Onless 5000,854
		Dversight Activity - Cortificate Management and Surveillance Activities per Flight Standards Information Management System (FSDMS) 2000.1	Oversight Lating Certificane Management and Surveillance Activities pat Flight Namherly Information Damagement Search (FNDIS) 8 vml 1
		Sorveillance Volume 6	Sometiume I al & Car - 13 Part

Flate | 54

ACO Functions	MIDO Functions	FSDO / AEG Functions	Future State ACO/MIDO/FSDO/AEG
		 Chapter 13 Part 145 Risk Management Vi- C13 S2 Inspection Activity Surveillance Volume // Chapter 9 Part 143 Applications – V6 C9 S 2, 53 4, 57 Certificate Requirements, V6 C9 S1 In-Depth inspection of Part 143 R5, V6 C9 S2 Record System V6 C9 S3 Manual System V6 C9 S3 Manual System V6 C9 S4 Housing and Facilines V6 C9 S5 Tools and Equipment V6 C9 S6 Technical Data V6 C9 S7 Quality Control System V6 C9 S8 Parts and Material Program V6 C9 S9 Parts and Material Program V6 C9 S10 Training Program V6 C9 S11 Maintenance Process V6 C9 S12 Work Away from Fix Jocanion V6 C9 S15 Connact Mimenance of C 	 Risk Management 2 n (13.50) Risk Management 2 n (13.50) Survey Rance Volume & Chapter 9 Part 143 Applications 10 (19.2.3) & 37 Certailizate Requirement 16 (27.3) In-Depth Inspection of Part 148 RS (16.0.9.3) Record Somm 15 (0.9.5) Manual Somm 15 (0.9.5) Housing and Facilitate 10 (19.8) Tools and Equipment 16 (19.8) Tools and Equipment 16 (19.8) Record Vola (16.1.9.8) Recond Control (10.0.8) Recond Control (10.0.8) Recond Program 16 (19.8) Doubley Control (10.0.8) Recond Program 16 (10.8) Recond Vola (10.0.8) Maintenance Process (10.0.8) Maintenance and Alteration 10.09 \$13 Maintenance and Alteration (10.09 \$13 Maintenance i.c. Suppliere(10.0.9) (19.82)
		COS Service Difficulty Reports 21:3 - 8120.23 Quality Endones. ontificaciones. defects - 8120.23 RE(0.107 WTSB recommendations - 1220.2	CON • Serves Difficulty Reports 21.3 - 87.0.25 • Quality Robusts and Dimensions, any 10 -8120.25 \$110.102 • TNB recommendations (250.3
		Investigations - 2150.3 + Voluntary disclosures -	Investigations - 2130.1 • Foluntary disclosures - 2130.1

Fige | 55 -

ACO Functions	MIDO Functions	FSDO / AEG Functions	Future State ACO/MIDO/FSDO/AEG
		2150.3 • Corrective action verification = 2150.1	 Corrective action verification - \$150,3
		Air Operator Certification Activity • Certification Process 8900 I Valume 2 C 3, 4, 5 & 7 Air Agency Certification Activity • Certification 8900 (Volume 2 C 8, 9, 10, 11& 12 Aircraft Equipment Authorization • 8900,1 Volume 4 C 1 days 16	Air Operator Certification Activity • Certification Process 8900,1 Valume 2 (*5, 2, 3, 4, 7) Air Agency Certification Activity • Confidential 8900,1 Volume 2 C 8, 9, 40, 116, 12 Aircraft Equipment Authorization 8900,1 Volume 4 C 1 thru 16

Segment	Organization	Name	
Assoc-ASD	Airbas	Entriclesage	
ASNOC ANAC	Bombardien	Chine Weathins	
Assoc ANA	Embraier	lose Luiz Beldemain	
Assoc - MARPA	Heico	John Hunter	
Assoc-MARRA	Jet Parts Engineering	Rod Sands	
ASSDC MARPA	will Enterprises	Simph Gonzales	
ANNUE MARPA	Primasure Teas	Todd Heroutuman	
rausport	Boence	Rick Baggette	
Panspurt GA	Gulistream	Bill Waitton"	
Full ARC Co-chair	104	Mike Reinert	
GA Part 23	Beecharam	Bandy Shields	
GA Part 23	Seedingall	Latry Moore	
GA Part 23	Beethetatt	Wes Klinkerman	
SA Part 23	(Cesone)	Joon Bourra	
Engines	GE	Dave Chapel	
Awarnics	Rackwell Collins	Marisa Stephenson	
Awamics	Exercite	Vac Ruggies	
Retorers*	Silversey Globel	Dan Shapiro	
EAA	-41E-110	Saran Thatcher	
FÁN	AWE-170	Cesar Gomes	
AA	ANM-1081	Kurt Krumlauf	
Claseryet	ERSA	Jan Sinuak	
Diserver	EASa	EricSivel	
abserver	EASA	Pater Corbeel	
Observet	TUCH	Marcus Tittiger	
Facar	365-340	Carlos Otnles	
TÁÁ	A15-340	Faul Clouter	
AA		Tont Janco	
Observer	AMAC	Falino Provesan	
Observer	AMOC.	Guilherme Maceool	
Gbserver	SAE	Bruce Maltone	

7. Membertlup

* Shannan

Page 177

LLARCMonling Discument - Notion Distribution-Overlight Working State Report

Appendix A

Oversight Working Group Charter

Dated April 18, 2013

Page AS

Scope and Boundaries

This Working Group (WG) will support ARC recommendations and automatal to a report addressing the following: Changes in the CLA oversight methodology beam on recommendations for changes to part 21 that takes interfocute extrange? Ut processes and oversight and delegation programs for descent and manufacturing related verificates and apprecials and autoritations.

Assumptions

- Oversight Requirements' Establishment of an FAA system: oversight approach in certification processes and continued anyonthaness assumes the part 23 rewrite will require
 - 1 A single integrated FAA oversight/surveillance/enforcement approach for organizations that are certificate holders that includes production & design certification project activates and continued operational safety (COS); and:
 - (a) Will not include Repair Station oversight for those facilities that have inalighe approvals.
 - Will be aligned with design organization requirements established by the Organizational Working Group
 - Will consider direct FAA oversight and FAA accepted "other pany" oversight
 - Will be based on a red-based approach -
 - For organizational oversight
 - Will address FAA surveillance in terms of frequency and types of wordns (design, manufacturing, projects continued operatorial safety, etc.)
 - (2) Will include oversight of COS processes considering the general surveillance process model currently us if in Canada
 - (5) Will include performance based oversight methods / measures
 - For project activity oversight:
 - (1) Considering "Level of involvement" (LOI) process models currently used in Europe and Canada
 - Recognizing the I OI may be in the "specialist" level
 - Will not require FAA delegation
 - Assessment methodologies that
 - (a) Accommudate / Harmonize processes across FAA organizations (ACO, MIDO, FSDO AEG, etc.)
 - (b) Establish the ionial compliance leading to FAA issuance of an organizational approval (operating certificate) for a funqualification enteria (level of acceptance) established by the Organizational Working Group
 - (c) Establish the compliance for FAA acceptance for a design organization progression based on qualification enterna dever of acceptance) established by the Organizational Working Group
 - (d) Establish performance requirements for the on-going oversight / surveillance to establish the offectiveness of the systems in place.

Oversight Implementation: Extablish the manufaction plan of roday's PAA oversight system to the recommended revised oversight system necessary for plan 21 changes. Robart change manufactured will be critical to implementing the revised part 21. Transition plan should tuchate criteria based gates, benefits over sets etc.

Page 155

Tasking

WG Tasking

- a. Decought Requirements
 - A definition of the FAA-surveillance / oversight requirements for the oversignmentions that are cortificate folders given the predicted outcome of the part 2) rewrite including the who, how, and what using a risk-based approach, including a definition of FAA assessment methods for issuance of any organizational approvals / progressions
 - A definition of the FAA surveillance / oversidil requirements for project activity osciluling k evel of Involvement (LOI) requirements

A definition of the FAA surveillance (aversight requirements for COS activities)

Oversight Implementation'

- A definition of what the transition model would look like, including gaining the needed understanding and support in the changes from key stakeholders in order for the organizational changes to be implemented effectively.
- Determine the skills, competencies, and associated training requirements for FAA. Third-Party and/or "Organizational" individuals performing oversight activities

Mainting FAA (coloucal expertise while transitioning to oversight focus

Create or revise FAA Drders and advisory material necessary for SPRM, including 8000.367A

Relevant Questions

- is Oversight Requirements
 - 1 What FAA organizational charges have to be implemented to support robust oversight? Will the FAA need to have Contenof Excellence?

What in the appropriate intent level of the oversight function?

- How will "one" oversight process be designed for AVS" "Single surveillance model between explaneting and production/ Should the oversight be implemented for AIR minially with AFS in follow?
- Is it necessary to differentiate the overshall approach based on organization size, certification maturity, etc.? If so what are the oversight criteria?

How will oversight be aligned to appropriately match the privilege and responsibility level of the organization?

- How is systems approach organizational oversight differentiated from level of twolvenieur (LOI) at a project level
- If origoing oversight is based on performance and not compliance how can it be normalized by company and scalable?
- How important is it that the US model is harmonized, or at least recognized as fully equivalent, with other authorities. DOA/DO systems - inday and in the future as they evolve?

How does oversight change from overseeing a delegated organization (ODA holder) to an applicant showing organization?

Should there be routine FAA oversight and audit of the design organization's processes or should this be accomplished via unifandir and reported to the FAA? Should it be a combination of both?

What eafery risk-based structure will be imposed for assessed deviations to procedure?

- How does AS9100, or equivalent, outside auditing equate to the FAA oversight in stem? (Compare actual FAA Approved Production OMS audit versus AS9100 findings.)
- Should internet PAA OMS audits be used as a model? If not, what best practices can be benefmanized in develop transivions? (Consider other industries)

Should a system be generated to allow PAA resources to perform maintenance of organizations instead of a formal audit process (i.e., review of internal audits with a focus or verification and validation)? (Audit the management system elements not the findings.)

Oversight Implementation

- 1 What metrics / memores used to be evaluabled to ensure a successful implementation of organizational showing overagin?"
- What is the impact on oversight much? approach if any part of rulay is adegined system (minysinal or organization) authorization is retained?
- CarrEAA accept Third-Party conflication in lieu of FAA audit/
- What entireal entirena will be used to identify frequency and scope of oversight activity?

What are the other "traggers" that would gauge additional FAA overeight?

Deliverables

Deliver a report that dottines an oversigan model that addresses overstallit requirements (that "what") and sversigan tapleaningation (the "who add how") as identified by 3.1

Date	Action	Expected Deliverable
April 2015	Define current FAA oversight in terms of activities and descriptions	Documented baseline of carrent $V\Delta\Delta$ oversight
May 2017	Enve-to-Face monting in Washington DC to review / refine future oversight model and establish preliminary remplate for final report	1. Inserim "Duture -overwhit model 2. Phelminiary template for final report
6dy 2011	Ence-to-Face incetting in Washington DC to baseline oversight model against current development of design organization requirements as developed by the Organizational and SMS Working Groups	Agreement with Organizational and SMS WGs on digned oversight model with design organization / SMS requirements
August 2013	Establish assessment methodologus for level of acceptance for initial and enhanced design organization. Establish assessment methodologies for performance based effectiveness of ongoing operations.	Assessment methodologies.
Septemider 2013	Face-to-Face in Scattle or Washington DC to establish preliminary musition model baseline transition model with current development of cost benefit analysis methodologies not data agreements Update alignment with SM5 WG deliverables.	Transition model Agreement with Cost Benefit Analysis WG on benefit methodologies and data agreements. Agreement with SMS WG on ongoing alignment
(leiober 20)1	Establish shifts, competencies, and training argumentants for right iduals performing	Skills, competencies, and training requirements for future oversight model

Milestones and Dates

Page | BI

Date	Action	Expansed Deliverable
	oversight	
Sevember 2003	Reway impacts to existing FAA willets, advisory material, etc.	Definition of necessary 64 Å ritramient changes
December 2011#	Complete Wersight W(7 final report and submit to ARC	Final Report

WG Integration

 Oversight WG will need give independent of the Openneutronia and SMS WG recommendations to develop specific exercisitie requirements and implementation strategies.

Coa Benchi Analysis (CBA) WG will identify data needs for CBA from other working groups: evaluated benefits are dependent inpart on what WE's recommend.

References

Working Group Lead(s) and Members

Aŭbjus	End Lesage
Bornondier	Chris Watkes
Emiltract	Jose Linz Bekärnan
Haico	Joins Hunter
let Parts Engineering	Rod Samis
NAS Enlerprises	Simon Gouzales
Measure Tech	Todd Haveutumian
Bueany.	Rusk Baggeire
Gullitream	Bill Whitton (Working Group Lend)
Beeslicati	Randy Sindids
Cessiii	heaper Holloway
GL	Dayo Chapel
Rochwell Collins	Marisa Stephenstor
Granmin	Nam Rognics
Silucity Global	Dan Shaporo
FAA AIR-110	Satali Thaicitar
FAA.ANE 170	Casar Gomer
FAA ANM-1081	Kort Kronilaal
FAA AFS-340	Cartos Quiles
FAA AFS-340	Paul Clonier
EASA	hau Novali
TECA	Manuas Finiger
ALIAC	Fahin Hinggan)
abilitie -	Gatthiarma Magado

Page | 1/2

21ARC Working Document - Notifor Distribution -OpenUppt Working Group Report

Tornee Mahone

SAT.

- 0

Page 14

LLARC Writing Document - Notion Distribution -Overland Working Discuss Apport

Appendix B

Boeing Aviation Safety Oversight Office (BA500)

and

Guifstream Aviation Safety Oversight Office (GASOO)

Charts

Paux | hd

1. FAA Oversight of the Boeing ODA



2. FAA Oversight of the Gulfstream ODA



21ARC vice ong Desument - Notifor Distribution -Oymunit Working Steam Report

Appendix C

DO Assessment Tool





Com	elencies
The	cell should be used by regulating stall with Insining and competency in Cafety Management Systems based on the KCAC SMIS Francework Understanding of Quality Management Systems, compliance and auditing Understanding of Compliance Assurance Systems, and regulatory compliance interview techniques Understanding of raik management Appreciation of the difference between compliance and parformance Report writing techniques (a allow nermitive to be used to comprise the ensurement of the answer
hasep buong	commended that as well as being framed to use the final in the glassoom, aswonment staff are for additional training using a live assessment to familiance themselves with the tool and its rai use
Meling Index Index Index Each and e olome	The loci (instructions) out avaluates the compliance and effectiveness of the CDO timuigh a deties of indicators. It is it issing the 12 elements of the ICAO SMS Frantework with the Francework definition followed by an Winewa statement for that element, plus the 15 elements of a Compliance Assurance System , and the 7 elements of a County Management System (CMS). For each element, a series of itors for compliance and performance is listed followed by a series of indicators of beat practice indicator about the reviewed to determine whether the indicator is present, subtitie on operating fractive using the definitions and guidance set out below, so that the overalk effectiveness of the in can be justified and supported.
The to it can regula	tor would normally be used by the regulator to record and document the assessment. Alternatively the partially completed by the organization to meressifiand ("How it is achieved" column) and by the for to verify and validate the organization's assessment ("Verification" column and "Summary ante" box)
Apple The s should find to	cability volunion form can be used to assues any regulated organisation. However, due consideration a be given to the size, nature and complexity of an organisation in certying out the assessment and in emailer organisations a reduced number of inocators may be used as defined by the regulator
Defin	mons used in the tool
Prese	m) is suggement that the "publicator" is clearly visibly and is discovershot within the organisations CDCF mentation
Sullai The in Ite ac	hie Idicator is suitable based on the start, ratius, concernity of the organisation and the innovant milk in they, including consideration of the industry vector
Opera	Nong is evidence lital the indicator is in use and an output is being produced
Effect	tive is evidence that the indicator is effective and echanology the mestions meaning
¹ Sinte (Cho APS Elegist Appandie J. Engosien J. FM

Evidence Evidence includes documentation reports, records of interviews and discussions and is likely to vary far different levels of indicator assessment. For example, for an indicator to be present the evidence of legy to be documented only whereas for assessing whether it is seersting a mey involve associating records. as well as face to face ossuescens with personnel within an organization. "How it is achieved' should include summary statements and any relevances is documentation and PRINTINGS. Ventication The Verification Column should be for the regulator to report any observations, conversations, records and documents sampled. Summary comments Once all indicators have been assessed by the regulator, a judgement can be made on when our the everall effectiveness of the CEO - - - - - int has been achieved; the should be need in the summary comments box Modifying the Tool A regulator may adapt the terminology and tool to meet its own national regumements out alimning to every version may enable mutual recognition acrom States Developing Procedures Each regulator will need to define procedures around the use of the tool, customised to its own -organisations! structure and approach to COC oversight activity. **Gomplementary SM IGG products** The loci may be used in conjunction with other SM /CG products

Page | V1

La C The dim opposite mocente	COMPLIANCE ASSURANCE BYSTEM proceed controls system (could provide could use of the Vielders do requirements. The or operation of public by a counter of with Wind Vielders and proceeding on the Unit CVC throughout the second of the second of	1 000	Lync Fried	NU HO	100	press a the diameter of any character of a a the diameter of any character of a	unen), - e lo como dance o punto completo e The ChS read que su
INCOM	TORS OF COMPLIANCE + PENERPHANICE	TW	TR	m	F	Lines it in meninum	1 Van die natione
14.5	Miles and able of a principle result for and republicly	f	1	1	1	THE R. LEWIS CO., LANSING MICH.	A ST HOUSE IN ST A
1	Displace more an example of the example of the countration	1			-		
1.0.2	Emulaim communicationum.	-		-		1	
104	Execute companion parts	+	+	-	-		
1.0.5	Linearly and datase cannots for the transcens between complanets planning, the contravers of committeet, and the completence contravers automotic defined to (b)(2), (b)(3), and dataset	T					
1,3,8	Document processes and procedures for the contractment management of provides, compared to parts, appliances, and contractment down.			ĺ.			
127	Countries who the countries FAA allow to establish entification requirements and proceptable settions of complexion and.			ľ			
146	Estuare the statement settiging comparative is provide by executed					16 - S	1
BESTR	RACTICE INDICATORS	P	-	0	f	How it is activesed	Verification
_		Þ			F		-
		1	-	1	-		
La SUM	MARY COMMENTS						
NDICA	TORE OF COMPLIANCE + PERFORMANCE	2	5	D	ε	How & Le achieved	Vertucation
--------	---	---	---	-----	---	--------------------	-------------
1.0.2	Take being pathoned The continuion qualification of the individual	+	-	H	-	-	
6.4	A percentamentament to a varie to performed to waitly a to concentration with the control and instances system objections, and	Ì	T				
CE 4	A recard at lost of the reducts of sconerregistment of the committees tablety				0		
DAST P	RACTICE INDICATORS	p	A	(a)	E	Now II is achieved	Ventication
	A second s				_		
h sin	WWWWYCOWSKEW73	H	5				
a sla	W4589 (C3)+545 (K73						
h sta	WALLEY CONSIGNTS						
h Sia	WARRY (CI) and KITS						
h sta	WARRY CONSIGNTS						
is sta	WALLEY CONSIGNITS						
a Sid	WALLEY CONSIGNITS						

11 12		1.77	8	D	E	Hum d is achieved	Verturation
45	The teal centerma ta required faitakers	1		1.1	٦,	the second s	
-	Included the design was being contribute server a manufactory representation of product			Ì.	1		
F3.	The Holds percenter (verified to the application of the process is not realized for which the weed, and				11		
0.4	Ay no good a logal 1 is finalization for final france is and 1 is a provide the contract of th						
ESTA	RACTICE INDICE TORS	P.	5	1	Ē	How It is achieved	VerMaatum
_		H	_	- 1	-		
_		1		1			

Page | /4

Enables If a CDD to evalue it made the quality place	1			1		
Contertant D. Office the presentation of presentations A limit.						
Conducts summaries and quelts of procession and provedures, attention in the period and ground to ensure that and takened and continue to much their infended periods.						
Using protocological and associate come time action processes, and optimes three processes were present any above for model		_				_
Indiales a ortanicided change control crowers.			1			
Violation of the performance and training and in additional to condependent					-	
heckades high development oversign (of surveillance and - out friends, and						
Encount of processes and processing proceeding access and processing of all datasets processing them.						
RACTICE INDICATORS	P	18	a	Ê	How # Is activitied	Ventication
	H	-	1	-		-
			1			
MARY COMMENTS	-	-	-	-		
	_	_	_	_		
	Charlen de CDO de endre in heade e se apartaciano remainmente de las adaptar far las examp el na mato ado ado end contra ao fa fall e servicio en la carlína a la comuna da se a manda la comunación en la carlína a processor a de las estas atendidas comunacións en la carlína a processor a de las estas atendidas de las estas entre las estas en las estas entre estas en las estas estas entre las estas entre entre las processors en la carlína entre comunación processors a una estas en las estas entre las estas entre las estas entre estas en las estas estas entre entre entre entre entre las estas entre entre entre entre entre entre entre entre las estas entre entre entre entre entre entre entre entre las estas entre entre entre entre entre entre entre entre las estas entre entre entre entre entre entre entre entre las estas entre entre entre entre entre entre entre entre las estas entre entre entre entre entre entre entre entre entre las estas entre en	Charles of a CDO consistent in the example of a number of the second of the matter of the solution of the	Charles of a CDO consistent of mount or an additional or and consistent of the additional for the open sets and prevention of mount of the additional formation of a constraint of the additional of the additional of the additional of the constraint of the additional of the addi	Charles of a CDO consistence in proceeding of the mail of an of the operation operatioperation operation operation operation operation operation operat	Charles of a CDO developer for the example of har matter and or off any large presentation of the advectory of an matter off the advectory of the matter off the advectory of the matter of the advectory of the second off and the second off the matter off the advectory of the second off the second of	Enabled of a CDO or evaluate it is available in the superconduct. Image: CDO or evaluate it is available in the superconduct. Contracts 0: OCD are then previous care (previous) and is matter and or evaluate it is available. Image: CDO or evaluate it is available. Conduct: summarize it is available in the superconduct or evaluate it is available. Image: CDO or evaluate it is available. Conduct: summarize it is available in the superconduct or evaluate it is available. Image: CDO or evaluate it is available. Conduct: summarize it is available in the superconduct or evaluate it is available. Image: CDO or evaluate it is available. Conduct: summarize it is available in the superconduct or evaluate it is available. Image: CDO or evaluate it is available. Up or evaluate it is available. Image: CDO or evaluate it is available. Image: CDO or evaluate it is available. Up or evaluate it is available. Image: CDO or evaluate it is available. Image: CDO or evaluate it is available. Up or evaluate it is available. Image: CDO or evaluate it is available. Image: CDO or evaluate it is available. Up or evaluate it is available. Image: CDO or evaluate it is available. Image: CDO or evaluate it is available. Up or evaluate it is available. Image: CDO or evaluate it is available. Image: CDO or evaluate it is available. Up or evaluate it is available. Image: CDO or evaluate it is avail

A set of a s	AT CITY MARAGEMENT SYSTEM In Totar model relinition is subty in APETY FOLICY AND OBJECTIVES MAREEMENT COMMITMENT AND RESPONSIBILITY instantion about actions actually poly which insult is in according the Executive of the object actually of the object is only only the industry of the incomenty function and important according to the object of the incomenty function and incompation according to the object of the incomenty function and incompation according to the object of the incomenty function and according provident and according to according to the object and according to the object of the object is an according to the object of the object of the object of the object which determines and an include not apply. The safety call which according of many function include the incompation of the object of the safety which according of the object of the incompation active according to the object of the safety which according of the object of the incompation active according to the object of the safety which according of the object of the object of the incompation active according to the object of the safety call which according to the object of the object of the safety call which according to the object of the object of the safety call which according to the object of the object of the object of the safety object of the object of the objec		anti anti anti anti anti anti anti anti	inter alter out in terr	ina i national a line a national a const line a line a lin	the Following: I and wild in the of presentation control over 1990 and 9 wild year over a second with 1990 and 9 wild present of balance of the second of the second second second of the second of the second second second of the second second and years of a reference of the second second second second	E, and which which he cannot by the instance, a elem maximum aloud the recorded. Theorytopart the committee metable and walk under the two instable and will under the metable state relation and automation to make relation and automation to a gradient and automation to
HIDEAT	DES DE COMPLIANCE + PEREDRAANCE	1.0	1.5	In	IE	How it is achieved	Avertilization
10.211	There is a under parker the intustry a committee broards acreaving the highest salidy scandards agreed by the Accountable & accident	Î	Ĩ	Ĩ	T	(one are anticipal	
M.T.Y.2:	The regarded text based its allefy rearrangement extrem to the salety policy			1			
11:44	The Automotive Executive and the end of paramy must been promote and decrementation their numbers of its factory Policy through active and vestice parameterian in the addity management system.						
111.5 7.4	The salink sciecy is communicated to all personnel with the initial that they are trade evalue of block manifolds configurations and daughtates with reason to Saless						
11.1.15	The safety policy includes a committee it to charge out anytophic legal requirements, clanicards are beel accord- providing appropriate measurement and defining which and providing appropriate sof all Managers.						
MICT.F	The safety policy activity innovanges safety maximal, The safety policy activity innovanges safety interface management anneales and technicate to contract a programmer to the safety from	-				i	
WL++B	The address pointy to revealed periodscole to be used to be addressed parents	Г	F		T		

1.2.2.8	The concentration of all the concentrations are seened	(01			
	if the subdy natragoromi system.						
11.2.42.40	A dampletary unley may been defined one doors dealine the preditions under which pumples adver would be constructed a grieque admity registerios of elifi- improved ().						
rinn.	There is avalence of depicer making willing and behaviours that related a process table colline.			ι,	į,		
		10	1.	10		bing if at semicinary	11/ in illing them
H + + 12	Permanential of the path are invariant in the establishment and matrix area of the safety manual establishment.	Î				They it is achieved	a manual set
0.8.8.18	There is one remains callely pointy can't instantion in construction, and it is implementation at all ferrors of the organization						
11 11 14	The solidy policy is case/or or data, as consolider, for all performance and in anti-case of data doubt performance and perpendiculation method.				(
01.77745	Sprinty servery objectment of we the experimentation is presented and reduction obtainments.		1				
11 1 1 16	The organization regularly softline that policium dishoughout the regularization are function with well have understitled the endowed the second softline with well have				ſ		
117331	The Assar Live Excession demonstrated the scontinent is exercised on any and in painty scattered on					N	
0.11.0	Concer Management time adopted on operated astery plan and produce with publicited and or an orbitality angue profit bijectory		1				
II. T. T. BLW	MMRY COMMENTS.						

Page | //

and naths to make 5	ent, unexpective of alter functions, or well as university with re- view what be downmining and yourn uncoded throughout two or induces nagrocary calls your history and the second by	three and the second	A.Q. V	Paga Int. P	in in info mili	the one country the second as by performance of the SMS 3 multi-reducte a second room of the multi-reducte a second room of the	Salely research block, accountable n books of research with adherity
EFFECT	VENESS to achieved view view are ready hims of safety account consentativity for the SMS and the Account actor Examples and in	intri 	ning print	altern be	inter	usit) the organisation inclution fully autoestayon the rules for) See See countrible portions when they and by the opportunity o
INLICATI	DRS OF COMPLIANCE + PERFORMANCE An Accountible Feedbrief has been insponies with full responsibility and utimesis accounteday for the SAS to consist if is inspoly implemented and partners allocated	9	5	0	E	How It is achieved	Minification
10.=2.2	The Accounted in Executive has economical the Prancial and terms recording department for the propular maintaining of un- uffective SMS	1					
10.123	The Account intel Execution is 6/by system of their SMS roles and recommendations or presents of the carding pairty working characteris and unless carbon of the recommendation						
11124	Safety accountationers autorately and inspeciality and official and does not field the obginal file on a new nexts	1		1	1		
m c2.6	Persential of all levels and expressed each contential likely safely accounted free solesaries and respondenties remaining of solery reactions will recommend to the actions.						
111.7.2.1	Safuty management is proted aprove the argumenter, and is not just the resourcebley of the Safety Menager and those terror.						
11127	There are decomposed in adapted or gas access degraps and ich description for all personnel						

RESTOR	ACTICE INDICATORI	10	15	1213	ET.	House of the architement	Unittientimy
0.12.0	The or endower if personant involvement and something on a two endowed out specific of the Soft	Î			1	THE D IS NOT	
1120	Terms is proceeding that watery management system price plea- parts presidential all have and the expression and solution is part of the everyday toppologie.	Ì			Ì		
NC 7.2.10	Carding comparison to adjoin the encounter are description of and new Woods down the fact stranged by the	T			T		
11.11.11	King walding with times and closing values that for work of management blacks and response bill its and incorrections with personnel tectormatical language.	t			İ		
1 1 2 12	These according that server management each years the lag. We anys of record server from of leads of the organization and how a mechanism for information of a server organization.	T			T		
1.1.2 50							
11.230							

Page | /9

The orth	minuters straff aboutly in Saintly Identican to be the compoundable and	comb	ini.	03%c	ШĄ	point hat the incommentation a	-d mentional or branche 1981
EFFECT	WENERS is achieved when the SMS is facilitated by the response of energy of the organization. Each was a new results are actively	ilde TEPR	josh god (ninal n ihe	-	d Overe is a satisfy strailare a big namagaranti system.	they present from the passes
NOCI	DAS OF COMPLIANCE - PERFORMANCE	P	15	01	E	How B is achieved	Ventication
00.6354	W amorphism person with the amphonistic knowlings, while and expension has been insumated to manage the operation of the SMS	F		Π			
11.432	The planar macaging her canadaw in his SMS futbolic required per forctions and responsibilities	I	1	Π			
M 123	Three is a chevel respecting time testymene the Salety Managari and the Amountable Disordary.						
M 1.24	The organization has obscribed addresso consumer to manage the SMS industrial management to calcul investigation, motivate, average and promotion	1					4
11115	Processes in any addity roles and tax common framph on these transmission and anomalance of contenues and set that						
		12				March 19 10 and 10 and	T to site set
11130	The organisation rate establishment a procession actiony committee at supply attim appropriate for the size and memploxy of the organisation, committing of a foll range of sector monogeneous processions.	ľ	1		E	How U IS ACHIEVED	UNITALISATION
111.4.21.7	The Salley Coordinate in the approximation memory. The salledy performance of the operations and the effectiveness of the SAS and is normally related by the Assessmente Exponence.					-	
MILE	The person (is) necessarily in the memory and maintaining the DME transferred executions dimension to the suggestion indicating the responsion of the safety retrollarity the argumention.	1					
1011.2.0	Subity common structure on control and superficient contracted organizations.	k					

U 1 T XUMMERY COMME	m.		

	DHS OF COMPLIANCE * PERFORMANCE	4	18	0	E	How It is achieved	- 1	Variation	-
10.9-4 1	An energy rep response plan (ERP pair other in factors nation and complexity of the operation has been developed and define the reconciliant, other, mayout tables and anti- other control according to the operation.			Π					
11.842	Kay paraonnal in an omorganey have only become in the ERP at all threes	T							
A 143	The equilibrium has a present to distribute the ERP procedures and 6) communicate the crosent to at personnel				1				
0.74.4	The ERP is presentially terrarility on admission of the plan and the real discovery of the preserve its allocations			11	1				
BEST PR	ACTIGE VERICATORS		15	101	E)	How it is achieved	1	//esthicition	_
11.1.45	The improvement of the American interview of Electron United by (Block) or approximation with effort organizations for restaul all and the provision of interview by Services.								
HI LAN	The arganisation has implemented Critical Incident Science Manuforment for the Principles	T		Ħ	1				
IN FAT SU	MARARY COMMENTS	-	-	-			_		-

MILLION & T	THE OF PALSE ISSUES . RENEARSHIES	1.00		-	-	Cheve Max Authority	Lingtheating
1151	Train is decimentation the describes the safety servicement system and the interviduoraters testimory of of the observed.	1	-			How A to actived	Ammicance
0.162	SMS decomentation is running inviewed and undered with accountain vanish cellular in place.	1					-
114.8	SMS decommentation is machine would be to all parson and	Г	T				
11:54	The SMS decision (full) or default and indecessors the mass of fail the decision of class, EMS excited corrects	Í	1	f	Ì		-
IST PR	MACTICE INDICATORS	F	13	0	m	How It is activened	T.Martic/hon
1155	Earliefly minangement processions = 0 viologisplexitions existing argunizations minimals.	Г	Γ		Ĩ		
134	The arguinisation has analysized and scenarios runted appropriate reactions for the deformy of discongenitatives of table the empropriate and operational layout	1			Ϊ	_	1

M.2.4 H Two cargo unidents callety car	AZARD IDENTIFICATION manifest shall carefully and standards a formal process that assume and anademic to spentry process frazients. Interact identificance is collection	a let	ul an	e isas	i ma	ing haanina are denigeri. Yo a containilos of readings	Tree attended in attoartioner an	ni) presiding new provinsion of ni) presiding new provinsion of
EFFECT	INCREASES and the state of the second state of the second se	ol/1		net a	atmin	and the opposition the opposition	on Hadaou	a non communities a and nor
INDIGAT	ORS OF COMPLIANCE . PERFORMANCE	1	13	U	E.	Him it is actioned	Ver	Michilden .
MIZ & A	The organization has a reporting update to ceptions error. but articland near remains that to simple to ease and account be- to of Personnel.			ľ				
012.12	The organisation loss presentency contillant of the major houseds and essented flor road related to be control or in the.	T	T	T				
112.0.2	This validly importing optimit brenchin Levillance for the reporter of any account levin (or hell taken) and, where occurations, to the rest of the organism on		T	T	Î			
11.2.7.4	Salaty investigations are converted to identity underwise coveres and potential forcereds for existing and issue operations	Ī	T					
11275	Salaty reports are saind on in a interry manual.	T	T					
n.r.r.n	Hazard chertification is on dropout process and involves all key personnel and appropriate (takefunder	t	t	T				
11237	Раздилий пороканах тор тиразграния герита аге нажето на тиратириот воспларная	t	T	T				
0.2.1.8	Investigations relations balancins antibuling (actors (Wey V Impounds and pair what balances).	l	T	T				
11212	Personnel systems contributed and loter in the provident term importing pavey and process.	T	T	T		-		

11.2.1 10	The (ucon) - Unrolled are documented and kents equilible (at folgae televence						
HI.2.7.19	The organization must be reacted of investigation of providents and accolumn as a vector for hazard blandfication in the system.			1			
0257 20	ACTICS MARCHTOPS	1 P	14	I m		Man II is achieved	Vertilization
11.2 9.12	These is preasants receiving system indicate try reporting layers of mean than, on a servings. I report our patron par patro					There is a construction	
M210	Safuly Reports include the exporter is over entrol and events that the executive yearship remove the events where mo- ume view events must			Ī			
162734	This reparting update anticovery personnel to prevente preventative and copyrights defeats						
MC32 m 18	Tomo in nonlinear limit the reporting realisty in articlely used lipid, and the nume organization in each lagarithesi and a nearly location.			1	1		
10.2.1.18	The control worker is available to contended anyone both and outcome to make against						
11(21.37	There is a previous in place to dealy a remark and macant legs to took for frends and your mounth management information				1		
0.2.1 Buy	nija by Comments						

in and the	directioned			3	1	must man examine of must just	a la companya ao ao amin'ny faritr'i A
line)	WENESS IS addressed which index is a control (manual that minutes	b cm				munital Ana control of the same	all upped as dimensional to the subschemen-
INDICAT	ORS OF COMPLIANCE . PERFORMANCE	v	3	17	Ē	How It is achieved	Vertilication
11.1.3 1	There is a sinchine of process for the memory must of non-Heat included the absenced of this accordance was merelined hazards, capterauld in forms of basenity and protocolday						
0.22.2	There are orders for evaluating the taxet of risk the regimeention in Willing to accept			Π			
m.2.2.3	(fer arganization has net contex strategies that make hazart) alimitation, rule context, net avaidance, net acceptance, rule milipation, and where spritciple an action plan.	ŀ					
m.5.2.4	Miligency accounting from the risk summariant, terticidag tradition and allocation of responsibilities are documented				1		
112,2.5	Web memory and a middally assued in terminal indian processor	1					
HLTT E	Ethesive and related mingalized and realized an equipmented						
11227	Triate answerman to and and onlying one appropriately justified.						
10.2.7.6	Similar in an agrowth have velicity of conducts and betternly business and linde relagation and controls.				1		
BESTRE	RACTICE INDICATORS	P	15	10	E	How die achieved	Vestfacations
M.2.2.9	There is evidence that risks are being wassaged to as key as manufacturally enabled.			ľ	1	1	10000
WIZ2 10	The organisation data in care messagement could, to devine family new two grideleses that if chance with the industry						
11.2.2.31	The rick manuan resil price sees are revealed and improved - on a principal basis.		1				

LLARC (kouring Document - Not for Distribution -Oversight Working Distury Report

CLARKER COMPANY		

SMS EFFECT	IVENESS is a downed Views the regimment is now developed in a	0.01	ol e	atory	(00	folosoane e orașteis Mut ana	ananepointos de 15a Mario di oportamento
INDICA	TORS OF COMPLIANCE + PERFORMANCE	TP	18	10	E	How it is achinged	Menilication
M211	Safety objectives have been established.		1	1			
HEAT	Safety performance indicators have been defined, promotopied, and any rear minimum concerning and analysis.	T					
#314	Alsk integrations and contrasts are assist combativations to confirm they are wanting and effective	T	T	1			1.
masa.	Subdy and its are carried foul that from an the vallesy performance of the organisation and in services and assess to mill opening	Ī	1	1	Ī		
10.3 (5	Safety depiction and performance indexities are reviewed and replaced periodically.	T	T	T	1		
01216	Solicity abjectives, and largest are specific, mulsurable, spread to indicate and largest even a	1	1	T			
M.3.9.9	Internation obtained from Safely Resources and Compliance Mentioning activities lacid, back into the safety red true commit processing	T	Ī	t	1		
0(3-10	Safety ensurements and indexist the efficiences of the controls including these applied by contracted enjamiliations			1			

BESTPR	ACTICE INDICATORS	10	16	102	1E	How it is an mercel	Vertikanites
HI 0. Y.B	The organisation is manipulated to concern, house and trifed party suffray that and is tribere without a solar to an excernation suffray miles.	ſ					
df 3 7.10	When indexedually and environing dramation and personnance entropy the dramatical consideral interacts and ratios from all operations tool temperation processing allocation processing operations.				1		
11.27 19	Sanny свростука дечі дачілочально в свідалого алекторась ай аниан о Рисстураннаяти						
113112	Photosmenico measuremente have Case defined foi sensio satory riska ecombod on the satory risk profes	1					
11(27)13	Permittent at all levals the almost of the unity performance measurements in their second of perpendicitity and the resour- of performance measurements are transmitted to them	T	t		ſ		
113 [14	Safety performance indicators, are lineed in the angeotosite softery objectives, and State SPIs are considered veters appropriate.	T			I.		
M 3 x 16	The end as and dispatent of restains are based on table from the pollours and metallicity of						
13150	IMARY COMMENTS						

MENCAT	ORS OF COMPLIANCE + PERFORMANCE	TP	10	-D	E	How II to achieved	Varitication
11331	The impartation has indefinited a present unit content. formal heperformal particular analysis and for image remains of charges, many organizational charges and charges in key						
W.2.2,2	Ballely Come Rok managements and dealth addety for and	t	1	Ħ			
1)122	Rich stracholders are repained in the cleaner memory aneral programs.			T			
11.22.4	During the charge management process towices that approximents and public features and reviewed her recently infect.						
0257-01	ACTICE INDIGATORS	P	3	U	1.0	How & is achieved	Venticution
11 225	Validation of the adverte transmission of a conservation and operational develope trace token place to account account place preservation of the conservation of the servation				1	1.	
0126	All or gampartitud and contracted contracts are subject to too change management process.						
91.2.2.F	Safety personabilities, automices and responsibilities are promotiling part of the charger.				1		
		-	-		_		

screaks mail of the so text/sty-						
in takity		-	_			
_	-					
	2	-	ri l		iteres it in actions and	Vertication
nd mos illur				-	Cont of the boomedice	(Concept)
or the						
					1	
formation from	1					
In turns						
de unit						
	in the not correct to terres the mot	ur fluk mer sammad Er Tapre alle and	ur flut. unt taattant In Tupor	ur flue une suurmed In Turne dag uned	ur flue unt saartunt ko Tapoe	un flue une saurrund Er Tapres

Therappy	manien shull devalup and marking a salah finance programma e of the safety training shall be appropriate to ason molyclusing a	UPPER N	(ma	uren i Vitiko	het tre	permission frames and and or GMS	and inspection of the local state of the second s
EFFECT In officia	IVENESS is additional where an percentise and counted and countries	white:	01 fa	ntom		an 1742 i materi disfora anti fi	a press of 00% states = 0000 molt foll
INDICAT	ORS OF COMPLIANCE . FEREDRMANCE	P	5	12	Ē.	How to is achieved	Wardination
10,4 7.5	Them to a maintenance process to issuinty Salary Management transition reconcensions in that personal one component to contrary line dates.						
11,412	There is a process in place to measure the effectiveness of leveling and to fails accordinate women to increave excessioned entering.					-	
0412	There is a produce that weeks with the tody/strains completence) and taken a produced presented of edited your recent of						
10.4.3.8	Instance includes initial and recurrent fouriery.				1		
W.475	A training record to resonance he such all possessed fronted				1		
BEST PE	AACTICE INDICATORS	IR	Is	a	E	How it is achieved	Verification
0410	Training includes / union and expansion in prezes including registeringial deal with the intert of including tubical pres-						
10.43.7	Freezety responsements are decomented for each own of activity white the oppression, inclusion areas where foreiting and any settle are so do not be resulting.			i.			
HIA+a	A training contain analysis is carried and for-oil pathoened and in resultanty remembers				1		
10.4,1.9	Therease in a conclust for pseudoned inciding for contractial organizations related to the operation.						
10 4 7 10	Paravarnal Govery mechanism to consent additional SMS fromtage to relation to think roth in SMS						
0.410	Printing and period on early management.				1		

	-	And a mark		
III II I III Taaloony Largent (or officiency - for for grant	n		
ILA. I SUMMARY C	DIVINENTS		 	1

NDICA	OHS OF COMPLIANCE + PERFORMANCE	P	2	R	Ē	How It is achieved	Vertication
0.627	 Josty (how out) chalegies are communisabled incogrand the community is all permutant. 	1		17		14 A.M. 1. 1. 1. 1. 1.	
11433	Summer diversity and two spatial reliances are with the support where commences of the spatial forces of the contracted areas to with strate agreement.						
ESTP	RACTICE INOICATORS	TR	1		E	How # lo achiever!	Vertication
142	Then a safety commune of the trade proceeding as a second line commune of the trade proceeding as a second line commune of the trade of the SAC such as setting and commune of the trade of the SAC such as setting and commune of the trade of the SAC such as setting as a set of the setting of the setting of the set	ľ	-	Ĩ	Ĩ	UNIT REPORTS	
142.0	Time to the first of the second poly is work to the first external second to the second second by a nervanical instability only and the second seco	Π					
0.4.2.5	The effective work of safety common scale in recordingly	1					
1142.0	Safety sole) in the size of a record with the party of the size of	t,					
14281	ANNAARY SCHMENTS						

21ARCMpring Document - Notion Distribution -Oversight Working Broup Report

Appendix D

Aviation Safety (AVS) Safety Management System (SMS)

Strategy and Framework to Manage

Safety Performance in AVS

Auto P



Sitalem and Fromewook to Manage Blacky Piotomerica in 1/31 Document Overview The surpcosi of this document is to establish a strategy and homowork to measure and margage safety performance in AVS. The intended audience of the deservent is the individuals involvent in the sesion, development, implementation, and operation of the Avation Safety Safety. Management Symmetry (AV\$5M5). Episcifically, this document will favorate the AV\$5M5 Coordinates Group in setting appropriations regarding the Safety Performance Design Yeam act/villes and the methodology the loarn is excluding to employee in AVS for safety performance management The Stivlegy becker starts with background information followed by a disception of the concept. The objective of these subsections is to provide the reader the overarching context for univery performance management in the surrespace system with a focus on the regulator's activities. The background and concert is based on work done by the Safety Management international Collaboration Group. Nort, the scope of safety performance management in AVS. is described to focus on the expected results in AVS. Then, the requirements and assumptions used by the Safety Performance Design Team are listed to further specify the expected risults. in AVS, as well as highlight some of the inizied, but likely necessary activities/actions that the mutable of the Safety Performance Design Team's control. In the insmerkoli section the high level model of safety performance management in AVS is described. This model is placed in the pontext of the Aviation Gallety Safety Management System (AVSEMS) Safety purformance management is part of Safety Assurance and has Important interfaces with Safety Risk Management (SRM). The model describes the high-level toos at safety performance measurement at the corporate AVS-level and within AVS Service and Office... The Safety Performance Decion Team with develop more detail inclusion moshigh-lower steps through the course of development of the safety performance menagement FUDCHES IN AVS. (the Development and Implementation section describes the high-level plan for establishment of addity performance management in AVS. The section includes the Safety Performance Design Team's high-level activities, schedule, and militationes. This document will serve as a touchistone for the design of safety performance management in : AVE. Throughout the development process the stategy and homework may be simplerif as a result of testing, interface needs and the critical thinking in any design seturity. As seed, any changes will be documented and thered encomingly

1 Strategy to Measure Safety Performance in AVS 1 Background 12 Concept. 12.1 Safety Performance Management System Summary 12.2 Safety Performance Indicator Framework 1.3 Scope 1.4 Requirements 15 Assumptione 2 Framework to Measure Safety Performance in AVS. 2.1 Overview 2.2 Framework 2.2 Tobe Acquisition 2.3 Date Acquisition 2.4 System 2.5 Action	11 11 41 41 44 44 44 44 44 44 44 44 44 4
Strategy to Measure Safety Performance in AVS Stategy to Measure Safety Performance in AVS Safety Performance Inducator Framework Safety Performance Inducator Framework Safety Performance Inducator Framework Safety Performance in AVS Framework to Measure Safety Performance in AVS Safety Performance in	14 14 44 45 66 17 80
12 Concept. 12.1 Safety Performance Management System Summary 12.2 Safety Performance Indicator Framework 1.3 Scope 1.4 Requirements 1.5 Assumptions 2 Framework to Measure Safety Performance in AVS. 2.1 Overview 2.2 Framework 2.1 Overview 2.2 Framework 2.1 Monitor System 3.2.2 Data Angustion 2.2.3 Data Angustion 2.2.4 System Analysis 2.2.5 Action	44,46,8,四,8,8,
12.1 Safety Performance Management System Summary 12.2 Safety Performance Indicator Framework 1.3 Scope 1.4 Requirements 1.5 Assumptione 2 Framework to Measure Safety Performance in AVS. 2.1 Overview 2.2 Framework 2.1 Monitor System 3.2.2 Data Angualitori 2.3 Data Angualitori 2.4 System 2.2.4 System 2.2.5 Action	1 - 2 , 5, 6, 12, 18, 19
1.2.2 Safety Performance Indicator Framework 1.3 Scope 1.4 Requirements 1.5 Assumptions 2 Framework to Measure Safety Performance in AVS 2.1 Overview 2.2 Framework 2.2 Framework 2.2 Data Acquisition 2.3 Data Acquisition 2.4 System 2.2 A System 2.2 Suba Acquisition 2.3 Data Acquisition 2.4 System 2.2 System 2.2 System 2.2 System	(5,68, 12, 18, 1
1.3 Soope 1.4 Requirements 1.5 Assumptions Framework to Measure Safety Performance in AVS 21 Overview 22 Framework 221 Monitor System 32.2 Data Angualitori 223 Data Angualitori 223 Data Angualitori 224 System Analysis 224 Assion	56 17 58
1.4 Requirements 1.5 Assumptione 2 Framework to Measure Safety Performance in AVS 2.1 Overview 2.2 Framework 2.2 Framework 2.2 I Monitor System 3.2.2 Data Anguation 2.2.3 Data Anguation 2.2.3 Data Anguate 2.4 System Analysis 2.2.4 Action	17 Si
Assumptione Framework Assumptione Assumptione	
21 Overview 22 Framework 221 Monitor System 22.2 Data Angualition 22.3 Data Angualition 22.3 System Analysis 22.4 System Analysis 22.5 Action	
2.2 Framework 2.2.1 Monitor System 3.2.2 Data Angustion 2.2.3 Data Angustion 2.2.4 System Analysis 2.2.5 Action	30
221 Monitor System 223 Data Angulation 223 Data Analysis 224 System Analysis 225 Action	1710
2.2 Data Angulation 2.23 Data Analysis 2.24 System Analysis 2.25 Action	1243
22.4 Evalue Analysia 22.5 Action	1343
2,25 Action	1343
	1444
3 Development and Implementation	
31 Overview	1414
er song	111111111
Index of Figures	
Frank a Andrew Territor of Course	
Figure 1 - Accident Henris and Caulas Figure 2 - Relationship Between Regulatory Regulatements and Risk	33
Figure 3 - Safety Performance Measurement Matrix	. 44
Figure 4 - Safety Performance Indicator Framework	
Figure 6 - SRM and Safety Performance Measurement	1144
Figure 7 - Monitoring Continuum	12:12
Index of Tables	
Table 1 - Debon Requirements	78
Table 2 - Development Activities	1545
Table 1 - Development Activities	15



Page 99











_	subjective muse on a manage station reaction of a very constraints of the
32	adjuities is saged upon the relationship betwarp the received oversight adjuites and then
33	influence in service provider behaviors and cultomer.
34	
35	Ter 1 subcome measures come in two verifiesoverall event tales to a accident rates hull
38	loss fates) and event related to significant settive issues. The main distinction between
37	these event roles is level at detail. Overall event rates are the highest level metrics used in the
30	system. Event rates related to significent safety issues are typically lower level memory.
39	specifically forstand on measuring the effectiveness of mitigations/controls established to
401	adoress identified eighticant calety issues. In any case, eveni types in Tier 1 are triase
41	respectively with permittion gauge hazards - those hazards to which all or large segments of the
10	ungregenate buarde community are extrated
43	The Technic configuration in Particle and the Induction of a configuration statement
30	ter a safety participante indicators address the deharder of whateh service systems wheat
40	performence relates to parcy durcantes. Will have been adding durantee shall be demined
47	projecteding an econolidan with common cause intended. The set of addative should also
64	include measures related to heavide that are undue to the product/autoice provide.
45	
50	Compliance with regulations (AVS's specifications for control of huttards common to the
51	productiservice provider a population) is part of the process of this management. Therefore,
52	measurement of compliance should also include measures of how well the service provider has
53	used its SRM process to incorporate relevant regulations into their processos and product
19-4	design. Productnessess providem should also develop measures for fastance that are unique to
55	their operational environments.
56	
57	For a indicatory are process and outcome measures to gauge the safety interventions and
20	initiatives of the regulator. Effective regulator activities should mativitie and facilitate
60	product a more provider denoviors that, in the appropriate, result in overse teleprovements in
41	the information of the manufacture into a final data introduce and and a state and the second
60	within word long patients any use of endland. The ability to endlance a biture protomorphic is an
63	Immotionit characteristic of both Tier 2 and Tier 3 invicators
144	Charles and the second s
65	At Tier 1, regulator activities must be based upon influencing live behaviors of prosiect and
66	service providers. Regulator action of Tier 1 considers this online aurouption system or major-
67	Nystem companients or domains. Thus, this responsibility entails management of common
68	cause huzards. Accountablesy for identifying and designing risk centrols for these common
68	cause hazards reals primarily with the regulator. Effectiveness of the regulator's
202	accomplighment of this responsibility is, therefore, a matter of evaluating these functions and
71	nyn rymdal to AVS befety performance
	A A MARINE
12	La scope
13	The annough described above provides the centert through which safety performance
7-9	management will be used at the personal AVE local to the safety performance management
76	process will be prevention of the process of the process of the process will be processed on the stranger of the process of th
77	model). Further, AVS Schoos, and Offices with respectively interpretatively (The Chinese and
70	(AFS), Averall Couldication (AFR), Auropage Medicina (AAM), and so Traffic Sando Couldidit
75	(ACV), spoty safety pattermanuo intragriment to medistine watery within the components of the
eu	annequase system for which they have oversight responsibility on Tex 3 with the expective of
	Anna Marina Marina

influencing Tier 2) and contribute to the aggregate AVS-level (at Tier 1). Account and Prevention (AVP) will have the primary responsibility for aggregation of rate measures and developing the system-view. The safety performance happagers opplicable to ony AVS organization that diracity influences (in regulates) industry manage tisk. The Safety Performance Design Team was established to develop processes an	nt investigation ty performance int process if y behaviors in		
and Prevention (AVP) will have the primary responsibility for aggregation of sala measures and developing the system-view. The sarety performance transgema applicable to any AVS organization that dwardly intrances (in regulates) industry manage risk. The Safety Performance Design Team was established to develop processes an	dy performance int pracess if y behaviors in		
measures and developing the system-wew. The sarety performance managema opplicable to any AVS organization that directly influences (in regulates) industry manage risk. The Safety Performance Design Team was established to develop processes an	nt pracess in behaviors lin		
applicable to any AVS organization that directly influences (in regulates) induction manage risk. The Safety Performance Design Team was established to develop processes an	y behaviors in		
manage tisk The Safety Performance Design Team was established to develop processes an			
The Safety Performance Design Team was established to develop processes an			
The Safety Performance Design Team was established to develop processes an			
and the second	nd an		
implementation plan to incorporate addely performance management into AV5.	Throughout the		
development, processes for measuring and monitoring safety performance at the	e evs-level and		
winnin the domains of AVa Services Ondes will be projetyped and rested. Vide a	to the solidly		
management system (2003) has a rices on contenants in provinent is too we	by on determine		
measurement through testing, we may re-scope the application of the processes	in the second second		
14 Benulasanata			
The more frequencies	Time in All a		
Orders VS 5000 397A Aviation Salaty (4VS) Salaty Management Surface Rese	Concrustite and		
VS 6000 370A. Aviation Safety (AVS) Safety Policy. These mountements are su	mmar2nd m		
Table 1			
Table 1 - Design Requirements			
Design Requirement	Reletence		
FAA Order VS 8000 367A, Aviation Safety (AVS) Safety Management System	Requirements.		
AVE and AVS annuces officers must establish and document organizational solicity	Diagree V.		
objectives that are inservicible	Sertium 2.5		
AV5 and AV5 entropy of them must have safety reportive a pro- geometry of entropy to ensure that the safety reportive to boing mat	Compare 2 Success 2 s		
W/S and 6//S avvicance/fixed must develop and maintain parts to meet induced and such and must startly economics.	Chapter 2 Berbert		
AvG appropriations must involve their systems to masture free effectives else of active mellicontrols	Lindplief 4 Section 1 b.1		
Avising requiring on the france method of the sense the set of performance of MVS and enoting common an another method.	Stager 4 Second and		
AVE and AVE services/citizes must perform system assessments to determine whether monitoritized other viter has been mail	Crupter 3		
The AVE Management Team must mention register reviews of AVE to assess the	Compton 4		
Mategement reviewe must include assegang the head for changes to the AVSBMB	Chupter #		
FAA Order VS 1000 370A, Avention Safety (AVS) Selety Power	and driving the		
Stempo and Office Directors are traid associatable by AVS 110 Imperiant, operational of an economic and an exception of a second statement of the AVS SMS	Section Fairi		
Service and Officer Directors are traid accountable by AVIS 1.00 imperment, vicewall- nic and a sector of an encrose the AVISINE and a sector activity of the AVISINE for continuative management of entropy performance interruption the everyperice performance	Sector Fishti Sector 6 c		
	Avidential produces the intervention of the second of the intervention of any period of the latter, Just a Management System (SMB) has a faces on continuous imperiod and itelact, Just a Management System (SMB) has a faces on continuous improvement so too will perform the maxement through testing, we may re-scope the addition of the processes of the produces of the produces of the produces of the processes of the produces o		

	fair and the second sec	and the second sec
	Design Requiring	Referance
	AVS Services and Offices with sarety oversight recompositility read. Each shi measurable safety performance ratigets, with an emphanic nn quarthetive data over quarterive information	Sealen 14.2
	AVX Services and Diffuse with safety even up the ownedulity must Dor unent safety observes and early performance to use of FAA intotegic plane (e.g. Decideation 2023), the AV8 Bounders Plan and the included Service of Office Service Plane for the early people AVS with included early we performance plane to the organization's safety, objectives and performance of streets.	Sector 141
	At a minimum, the AVENIT most assess its astery performance interve in its satury performance targets on a quarterity brian.	Section 15 a
	At a renormal individual Service and Lifese Directors must exercise their Service of Office schehr) certormance, milli verto to safety performence targets, each month	Second 19 m
B 0123455789012	 Results from the safety performance measurement processes should business planning activities and establishment of strategic objection support management decision making and resource a focation Safety performance measurement processes developed by the SM (C greatest extent that is practical Safety performance measurement processes will not rely on informa- tiffy performance measurement processes will not rely on informa- tiffy for initial introduction into AVS. The extent to which IT systems a safety performance measurement processes will be ditterrived after processes have coan developed, treated, and put the operations. The design requirements we be used to glide the sevelopment of safety perfor- molecurement processes and validate that they meet the intern of AVS polos. The requirements may be modified as a result of learning during process development important to role that SM ICG safety performance materials have been during. 	I Inform Lagn with the G to the librit technology can support the business manual halse design nt. Finally, If is but no!
3	Trialized. Once they are finalized, this document may need to be modified in ord align with the final product.	ter to belter
5	1.5 Assumptions	
6	The Safety Performance Design Team made the following assumptions when e	stablishing its
2	 An AVS-level SRM process will be developed as identified in the AV. 	SEMS
1	include)	No. 11 Ven
3	Standard methodologies for schipping ink across AVB	
3	Risk acceptance processes. Development of risk centrols, and	

	currently examined and the specific specific sector sector and the matter in the sector of
à i	Requirements for AVS Samides Onice: to manifer risk combals (a - rows the
ê.	affectivenese
б	 There will be AVS-level processes, as well as processes used with AV6-
21	Bervices/Offices for safety performance massurament
8.	 Cirganutational changes might be possible to support safety performance.
¥	management. For instance, establishing a dedicated group responsible for
0	managing/optimating users) performance measurement processes might be the most
1.	effoctive and efficient way to accomption the objectives
	 AVIS management is committed to salety performance management and will aspend the develop and adole patient bademance management discasses. Submit
	performance missionered will eatail performing different advises which will an be
5	require some number of AVS personnel shifting their time and focus. Further,
6	outcomes of these pioceases will likely impact how management applies responses
7	It is assumed that management will support this allocation of resolucion
Б	 Necessary skills to identify effective and actionable ways of measuring safety.
9	parformance suitst in AVS. It enhanced competenders are required to implement.
0	safety performance measurement processes, they will be defined and inclusied in the
	safety performance management implamentation plan
5	2 Commenced in Marcon Calman and
4	z Pramework to measure safety Performance III AVS
1010	its sarety countries intrough the collectedy, intelligets, and assumement of information. It is important to understand how SRM and Safety Assurance work legither to understand the context in which Safety Renformance is manifyed. The SRM process provides a system analysis, the dentification of historids, and the analysis and astrosomment of contrivial. As a result, safety risk controls are developed and, once they are notemmend to be granticable in
587880122450789	redocing safety role to an acceptable level, these controls are employed operationally. Safety Assurance is used to ensure that safety risk control strategies are in place, asserts whether they are achieving their intended safety risk reduction objectives, and recorder for unintended consequences. If the controls are hot address to taken a series whether they additional safety risk controls are developed through SRM. This is one way SRM and Safety Assurance are integrated. Abother way these functions work together is through SRM. This is one way SRM and Safety assaring are integrated. Abother way these functions work together is through the identification of potential new negarity or indiffective controls through Safety Assurance measures, which are then analyzed and assessed using SRM. Figure 5 decids the SRM and Safety Amorance processes and their relationship to one another. If shows the mest obvious and focusion interactions between the SMS components of safety Assurance and SRM. There are three brain findings in the system assessent within Safety Assurance. In conformance with requirements.
15078001224507890	redocing safety role to an acceptable level, these controls are employed operationally. Safety Assurance is used to ensure that safety risk control strategies are in place, asserts whether they are achieving their intended safety risk reduction objectives, and recoder, for unintended consequences, if the controls are not adequately reducing safety risk, here are modified analysis additional safety risk controls are developed through SPM. This is one way SRM and Safety Assurance are integrated. Abother way these functions work together is through SPM. This is one way SRM and Safety assigned using SRM. Figure 5 decids the SRM and Safety Assurance processes and their relationship to one another. If shows the mest obvious and from on interactions between the SMS components of Safety Assurance and SRM. There are three beau findings in the system assessment within Gafety Assurance and SRM. There are three beau findings in the system assessment within Gafety Assurance and SRM. There are three beau findings in the system assessment within Gafety Assurance and SRM. There are three beau findings in the system assessment within Gafety Assurance and SRM. There are three beau findings in the system assessment within Gafety Assurance. • In conformance with requirements. • Not a conformance with requirements.
150700012215070900	redocing safety role to an acceptable level, these controls are employed operationally. Safety Assurance is used to ensure that safety hall control strategies are implayed operationally. Safety are achieving their intended safety role reduction objectives, and montor for unintended consequences. If the controls are hot addicately reducing safety risk, they are modified analysis additional safety risk controls are developed through SRM. This is one way SRM and Safety Assurance are integrated. Abother way these functions work logother is through SRM. This is one way SRM and Safety Assurance are integrated. Abother way these functions work logother is through SRM. This is one way SRM and Safety Assurance are integrated. Abother way these functions work logother is through SRM and Safety Australia new negative or notificative controls through Safety Austrance measures, which are then analyzed and assessed using SRM. Figure 5 decides the SRM and Safety Austrance processes and their relationship to one another. If shows the mest obvious and frequent interactions between the SMS components of Safety Assurance and SRM. There are three beau findings in the system assessment within Safety Assurance and SRM. There are three beau findings in the system assessment within Safety Assurance and SRM. There are three beau findings in the system assessment within Safety Assurance and SRM. There are three beau findings in the system assessment within Safety Assurance and SRM. There are three beau findings in the system assessment within Safety Assurance with requirements. • The origination of a new facand m ineffective continit






_	sitale@rearch.outawarkeowendatesiae/headutation.prw/3 crites1t
141	AVS Services/Offices will also identify the areas to monitor within the component of the
42	denospace system for which they have everalght responsibility. The Safety Ferrometers Design
543	Team will develop materials to ensure that these ames are derd/led in a consistent reportation
44.	manney Again, these areas alread be based on the natpets of SRM within the applicable
45	component of the aerospace system SRM could be conducted by AVS organizations or
H6	preduct/service provider organizations. Ether way the outputs of SRM will include plans to
47	monitor the effectiveness of the resultant safety risk complet. These plans become the basis in
48	the monitoring plans identified in the unlary performance measurement process illustrated in
149	Equire 6. When its in altermined to be acceptable in SRM. the Momiloring Plans may be
50	updated to remove ar charge what is manifored
151	
152	Monitoring plans will include indicators as described in Figure 4. Safety Partermance indicati-
169	Framework. The skulpture of individual monitoring plane describe the foliationship to the Tax 1
54	Integrated Civil Aviabon System Level. The deslied behaviors of the product/service provider
55	Ter 2 and the activities that AVS will undertake to influence the beingvior of ploduct/service
60	providers. Appropriate safety or formance terget, will be established to assist in the System
10	Analysis step. Gallety performance targets are typically a predetermined risk reduction objective
00	That prings reas written acceptation toyota.
109	Contention Presidential process the million monitoring plan is developed in the Update
00	wontering risk stepping, while the number of the constant, it is important to some the
10	trate and then obtaine the monitoring plane, barried on the buccuts for Section
102	Monitroops many will be developed at the AVS level and writes the Dendess Offices. These
64	nings must take into account aviation data sources from within and outside of the FAB
100	Montration many must also ensure that the requirements in EAA (many VS MOD) 3704 remeting
06	The periodicity for missioning addets and immine largests and met. Management must appear
67	tation, performance relative to safety performance targets at least questions the AVS-level and
600	of least monthly within AVS Services/Offices
69	2.2.2 Data Acquisition
70	The report own plan identifies the data to member Al the AVS-level, the flate may be an
0.03	agg signin of data collected of a lower-invellant two search as the Aviather Safety Information
12	Analysis and Sharing (ASIAS) System will be laveraged. Similarly, within AVS Services/Officer
77	there are aways data sources and tools that will be incomposible relief the data recomment to
14	implement the monitoring plan
15	2.2.3 Data Analysis
76	In the next stap, data is analyzed and turneri into information such that decisions can be marked
117	in the following step: Date analysis is not an milliomated (IT fased) antivity. If requires an
18	individual or individuals consecting the anniver, and publing it in a formal and structure that s
79	unconstandable and manageable by decision-makers. This information must be presented in a
50	convision formal. Therefore, part of the safety performance process development effort will tak
81	the antabienment of consistent methodologies for preventing information to decision-makers.
87.	2.2.4 System Analysis
63	In this step, decisions are made regarding the effectiveness of safety management activities at
84	Inv AVS-level and within AVS Services/Offices Specifically, detorminations are made
85	reparating whether the safety performance targets are being mat. If safety performance targets
06	are met, it is presumed that allely risk controls are effective and are having the desired effect.
	1.000 AV

	Strategy and Fromework to Manage Sarety Pestonnaria to oVS (Jone 31, 2010)
87	If safety pertormance targets are not met, additional analyses are necessary to determine if a
BB	potential new interard events and/or if there are ineffective controls. If potential new iparards event
60	and/or if there are ineffective controls, SRM is applied to (dept/y additional ections increasar)
08	2.2.5 Action
er.	There are basically two paths in this step. First, # a was determined that there are obtential new
58	hazants and/or il there are ineffective controls and SRM was conducted, the monitoring plan is
163	updated to tike into account this change. The other, more positive path is that the performance
94	larges, were neit in the step above and a decision is made seguriting whether the monitoring
95	plan shered be updated as a result. For restance of the AVS-toret, the AVS Manupement Team
698	(AVSMT) may decide to change the 3 proficant Safety issues to move an issue of the list in
197	favor of a wately examined that has become a higher contern
EB.	
999	The important characteristic of this step and the whole presides, in general, in that safety
00	performance management is a dynamic activity. System changes, whether planned or
01	unplanned may result in changes to wrist is monitored and measured. The focus is on
02	continuous regrovement to continually drive risk out of the annespace system.
00	
64	Part of the second s
105	3 Development and Implementation
106	11 Overview
07	This BUSTARS investmentation Blan datastrone for burn interim the post-start of action
ine .	the domatics responsement in AVE. EAA has mean documents that lat asterio objections and
DAT -	describe statisticates for activities to examine the p. Overfination 2025 Next Committee Are
10	Transparticle System (ArytGen) goals and parformance measures Busiceses Phase of various
TT	Investing allowances, there he not fully meet the parents for AVSISIAS to determine how effective
12	AVS is a manufaction callely. Salary performance manufactured should encap upde realized and
03	programstimal safety objectives and simuld be programered in the appendance of safety risk in the
14	an musule system components that AVS Services/Offices oversee
THE.	
18	Therefore, AVS established the Safety Parkemanos Design Team to develop a process and
07	muscums for measuring safety performance in the asinspace system. Please rists that the
38	loam is called the Sality Performance Team in the AVSSMS Implementation Plan
nn.	Understanding the salety performance of the senspace system, will allow AVS to omisince is
20	management of safety. The Safety Performance Design Team will develop processes that
01	Support management decision making and facilitian change within AVS and industry. The lower
22	is starting with the development of this document, which builds on the SM ICIS's work. By value
2.1	The week done by the SM ICE, safety performance actoss AVS will align with safety
2.4	performance in other civil Aviation Autourtaes (CAAs) and their State Safety Programs (SSPI)
28	and the state of t
28	The must product the fourn will devising is a product to identify Significant Safety issues (called
11	Significant Misk lineas in the AVSSMS implementation Plan Significant Sakely issued deposite
20	the nazaros of events that are expected to have a considerable impact on the accordinate and
28	are preserve candidates hit medulinimmit. The fearly working with standardow
124	cervices und industry, will use the process it developed to contact system-level
100	table and the meany being the approvent barety to see in the setupping and the term
35	Significant Native sciences are savely because a essential savely performance indicators for the Significant Native issues. The salety performance indicators will remainer both elastione and
	L= 3 01 Par. U

process measures as described in Figure & These safety performance indicators for the FAA, as well as the affected productivervice provider i the Safety Performance Daugh Team will work with Services/Offices to the affectiveness of their safety management processes. While AVS Si is and developing Service-lavel safety objectives and measures, it is im condinated and consistant. Therefore, AVS Services/Offices will work Performance Design Team to ensure alignment of safety process and p the team will also make use that the measure of parts on the measure in the processes (e.g., the positive affect the processes thave on safe f there processes (e.g., the positive affect the processes thave on safe always objectives, including those described in Destination 2025. With it	indicators will include organizations a relability measures for envices/Chices may have portant that they are with the Safety performance massures with the Safety performance massures and of the effectivetons by performance), a prospece systems will
The Safety Performance Design Team will work with Services/Offices to the effectiveness of their safety management processes. While AVS So in an developing Service-lavel safety objectives and measures, it is im soridinated and consistent. Therefore, AVS Services/Offices will work enformance Design Team to ensure alignment of safety process and p the team will also make sue that the measures docus on the measurer of these processes (e.g., the positive affect the processes have on safe forward/files-level safety measures must also simpoin the higher-level affety objectives, including those disortated in Destination 2025. With it affety objectives.	e establish measures fin ervices/Oraces may have portant that they are with the Safety serformance mascures next of the effective-tonia by performance), aerrespice system-1 with
The Safety Performance Design Team will work with Services/Offices b the affactiveness of their safety management processes. While AVS S in any developing Service-lavel allerly objectives and impassing it is im- condinated and consistant. Therefore, AVS Services/Offices will work forformance Design Team to ensure alignment of safety process and p the team will also imake see that the measures focus on the measurer of these processes (e.g., the positive affect the processes free on safe invisio/Office-level safety measures must also singuisi the higher-level affety objectives, including those disscribed in Destination 2025. With it	a establish measures for envices/Chices may have portant that they are with the Safety performance massures next of the effective-tonia- ty performance.
The affactiveness of their safety management processes. While AVS 5 and are developing Service-lavel safety objectives and measures, it is im condinated and consistant. Therefore, AVS Services/Offices will work forformance Design Team to ensure alignment of sefety process and p the team will also inside sale (that his inside condition) and the measurer of brene processes (e.g., the positive affect the processes have on sofe investoffice-level caliby mensures much also singurit the higher-level affect volgetives, including those described in Dectination 2025. With it	enuces/concerney have portant that they are with the Safety performance manuare- nent of the effectivenes. By performance (by performance), aerospace system
In an developing service-have subly objectives and measures, it is im- condinated and consistant. Therefore, AVS Services/Thices will work forformance Decay Team to ensure alignment of selaty process and p the team will and make sure that the measures Jocks on the measurer f there processes (e.g. the positive affect the processes have on sofe forward/thise-level calicity measures much alice singusit the higher-level affect work with the described in Dectination 2025. With t	portant that they are with the Safety serformance massures read of the effective tests. by performance) fairtospace cystem-1 mil
commance and consistent internetia, AVS services unices will work Performance Design Teem to ensure alignment of sefety process and p the team will also make sure that the measures focus on the measurer of these processes (e.g. the positive affect the processes have on sofe iervice/Office-level calaty mensures must also support the higher-level aNty objectives, including those described in Destination 2025. With it	with the parety performance mancures nent of the offective task. As performance), elemespace system-the of
The team will also make sure that he measure focus on the measurer if here provide the sure that he measure focus on the measurer invited the provides have a sure the provides have on soft invited the relation of the provided in the surgest the higher-level after objectives, including those discribed in Destination 2025. With t	nent of the effectiveness. by performance). Factospace systems ====================================
If these processes (e.g. the positive affect the processes have on soft ionvice)(Office-level salidly memories must also simport the higher-level affect volgettives, including those directibled in Destination 2025. With t	la performance) la performance)
service/Office-level salety mensions must also support the higher-level aftery objectives, including those described in Destination 2025. With t	armaphen system-ment
aNity objectives, including those described in Destination 2025. With t	successive about the second
and asheed as a summer the summer of the summer of a to the	ne manues in place
VS can implement the measure processes to measure safety. These	e marserer must else
e communicated to management and used to determine the need to a	system changes.
	and the second sec
The final activity for the Safuty Performance Design Teem will be to dee	notation an implementation
day for AVS to fully incorporate addity performance memogentiant proce	onnas. Tria Eaferty
forformance Douken Foury will also tracktate implementation of this plan	n within AVS. Three
clivities are summarized in the Table below	and a set of a
Table 2 - Development Activities	
Sentery Performance Gerlage Term Actually	Due Della
Feasiert Strategy, and Framework to Measure Salety Partomones in 5/5	Juran 10, 2010
Develop Process to Identify Eignificant Ballety Impres	Cercontheri 37, 2013
Determine Significant Subtry modes and Establish Salety Performance -	4000 ato 2004
ments sty Menautries in Consistence Pfontovisions of Solida Materialist and Photosess	Desparator by 15044
Gevelop instrumentation Plan is constructing Safety Peldomanical Measurement in Avia	June 30, 2015
12 Strategy	
is contractly that a specify of the solidity performance meter upment and	term will be limiterented
n physics of states. As dements the developed, they will be tested an	d prototyped. Rayad au
asson learned from this testing, they will be refined and more broadly a	colled Implementation
I processes will be iterative and evolve while adhering to the general o	unlive and bracking
leachined above (and in the AVSSMS Implementation Plan)	
full implementation, which will lead to continual improvement (Pitase d	are cheanaideera in broi
VSSMS Implementation Plan, is expected in 2016. The implementation	ion ofan te lincorporate
afety performance measurement in AVS will entablish more upportion	rivities, meastones, and
Ue dates for full implementation.	
	The final activity for the Safaty Performance Design Techn will be to doublen for AVS to fully incorporate addity porformance menugament price Serior AVS to fully incorporate addition to the technic Serior AVS to fully incorporate addition to the price Serior AVS to fully incorporate addition to the price Serior AVS to fully incorporate addition of the price Serior AVS to full incorporate Serior Serior Serior Serior Serior Serior Serior Serior Serior Serior Serior Serior br>Serior Serior

Appendix E

EASA

Embodiment of Level of Involvement (LOI) and

Safety Management System (SMS) Requirements

into Part 21

¥	for a ru	Iemaking task	_
Emb Mana R/IT/0262.8	odiment of Level of Inv agement System (SMS . RMT 0611 and RMT 0550 & BMT DM	volvement (LOI) and Safe) requirements into Part- 512 (MOH 060 project) - lawe 1 - 27/	21 /00/201
	Aminability	Produced trans	
Atomical	Conternations (bigrowanning / 2011	Enternation mark	-
Inapliation	No 749/2012 and de America (Part	Concord Minum.	Vela:
Duo Mecsonity	ED Derman 201 Journal (AMC 6	Highermaking gravity.	Not
	Cirris Terr 23.)	reason during .	PARTY .
Alternal	Applicants for SASA (Server) design.	MMT.0282 - LOCNPA & (IN)	
"instruction"	ignorranie, Part-21 Design and	out type.	Light.
	Production organization, NAAS of	Factorial consultation through NW F	
	methodal	dialgood;	115
		President date of 17% 1 (11);	1013/0
furse//neger	y withing applications of	Durwhin of NPA 1 consultations	2 marga
Daisbargant		Polycething dates of the Optiologic	20114/10
		BMT, UGIT - LDE MPA 2 (AMC/GM)	
		nilé type	Digto-
		Technical consultation storing HDA 2	
		dealthing.	You.
		Postocation anto of MPW 2	2014/0
		Foregoing to see the 2 contraction	210100
		Publication dament the Decretary	makin
		HMT.0550 & RHT.0613 - SHS WPA	sult and
		(IN # AMC/SM);	
		HLA type:	Full
		diation	Von.
		Publication cale of NPA 7	7014/0
		Duration of HPA 1 consentition	3 Munh
		Fortissed classification:	Yes
		Phillipsinn date at the Option	2015/0
		Philineattion-state of the Deceman	501010
		MODE While the woman method a formally "Agency. The Agency will be segrand by two prope (for LOI are SMS) with installing/MAA neuroscilative in alow and methods a set of odd.	

EM	ropean Aviation Safety Agency	HMT.0382 & RMT.0511 and HMT.0550 & HMT.0542 (MOM.080) 700 June 1
1.	Issue and reasoning for re	gulatory change
	In older to further improve the a fCAQ neveloared and adopted betw concept of "Safety Management" (5 State Safety Programme (SSP) ary (SMS) by a visition industry organis shall require that service provid organisation	Insetty good safety records of the civil avieton industri neer 2001 and 2009 in a number of ICAO Anneses the M). These new SM provisions require implementation of the Contracting States and a safety Management Syntem attons (Lervice privaters). An part of the SSP, each state less under its suffernity implement an SMS in the
	On 25 February 2013, ICAO edop Edition 11: Annes 19 consolitates actoss Annexes 1, 6, 8, 11: 13 em SMS which are generally applicable 5: Minorthikess of auctant. Anne Investigation 2013 Ed. MS will be rea 14 invjetnice 2013 Ed. MS will be rea Standards in Annes 19, Such mult thematter, the States shall notify to	Red a new Annox for Safety Hanadomitor (Annex 19 all the general Sbill and TMS processions connectly spread 1.14. Annex 19 transidicated the transmoster for SSP and to all available consister, no finiting the deman of Annex (15 will become applicate the of network downers on 1/ uested to restly ICAO of any differences that will exist an restored requireme or practices and the provision of the facilities are to be thanks before 14 Demote 2017, and CAD of any further difference that area.
	To reflect these IEAO revelopment of Total System Approach, the Are the IEAO provisions for Setery fra- thin Resplacian (IEC) 216/2008 ³ (In obligation that starts from the pro- trial the provisions at the rEAO to account (ref. Article 2(2)(6)). In the for answerthness, with Sersing, anonese to the Basic Regulation) in a Management System relating (where, The Agency has, becogni- Safety Management concept and) Balateri actions wate introduced in support the Member States to inte- houstry.	Is and in line with the generally accepted EASA principle net is in the pracess of implementing, in a modulual lense representer of the aviation introduction within the scope of encounter of the data is the Basic Regulation?. It is an examine of the data Regulation which requires to ensure histopy the basic Regulation which requires to ensure histopy the basic Regulation in the comptain requirement of generalizes, ATM/ANS, and annothed requirement is admity, and an is contraction of the solution, the basic Regulation to traditionate the problem with the contractions to tradements of the solution, the basic Regulation to tradements and module of admity, and an is contractions to tradement and neuronal angulation or address improvement, and domain to the European Aviation Safety data (EASt) is outled to informed them SSPs and tasks implementation of AHs in the model of the SSPs and tasks implementation of AHs in
	The domain of minut anwardium implementation of the Salley Mana implementing rules based to be one RMT 0611 and RMT 0550 % RMT 0 the TASA 2014 2017 minimulang 575 2.29.	es as the lass demann for which passiplate (or the gement providings has to be blanched), and EASA fail 11 and di accordingly licitated paleovalong bargs (int 1262) (7) (homeny all under the MDM ONT order () are and a Programmer to address the CKUp actions (VCI) to and
d.	Objectives	
	The principal objective of the truth with the framework of safety the check the ownoing Part-21 miles a provinsive too hat two Design & Me Part-21 are ICAO SMS pomptionit, and performing toes continuing o model from TCAO SSP obligations.	making project to to ensure a full compliance of Part JJ negatives providents of ICAQ Amous 19. This reasons to individual an encoursery, the missing Sofety Narragements neutriciting 10 % M encourses approved order UASJ and the competent acthorities granting them approval versight one supported with the resultatory provisions to
A and	spatiation (SCI 76-21.07/2000 of the E-super- ter Table of some obtained one catalitations a period table. The statement (SCI 16-11/07/2001 on execution by Communic Regionality of the trans-	Personners and of the Course of all 20 reactions 2000 to another scale of buildening another builders Agains, and subarray "stand buildening of Demonstrative Software (13) 5 50, 150 1,200% to 13,0 demonstrative buildening and the standard scale of the Software Course (13) 110/11 4 5 1 2517 = 143
-	and all results in the line	











RMT.0262 & RMT.0611 and RMT.0550 & RMT.0612 (MDM.060) Stropand Actorion Safety Againey European Aviation Safety Agency Annex to ToR RMT.0262 & RMT.0611 and RMT.0550 & RMT.0612 (MDM.060 PROJECT) CONCEPT PAPER **Embodiment of Level of** Involvement (LOI) and Safety Management System (SMS) requirements into Part-21

_		(MOM.060) Concept Face
		ECUTIVE SUMMARY
This Cor adaptato Salety M topics Managen Commission Official International International International International International	repil Paper presents a top o or of the unplementary rules anagement principles introdu Usyal of Involvement (ru- heri System (rulemaning tais ani Regulation (EU) No 742/2 design approvals as well as a we been classified as comple- sh which arms at -	down, systemic approach in the makes related to the site limital advertisities in order to carely with the care by ICAO Annex 19, to assist with two closely relates inmaking tasks (RMT 05/2). The tasks aim at attention 912 and its Annex 1 (2017-21), as well as related AMC a compliant Salinex (Cont-21), as well as related AMC a control and the saline set of th
141	ediacing a risk-based approac product contribution, and	It to determine text) of Involvement (LDI) of the Agency
(h) bay ICA est cor	rewing Part-21 in check in O Autors 19 and add, an meo M) organisetions related to ultrements for competent wit throung oversight to support (compliance with the Safety Manupérnent provisions of estany, the requirements for Design & Manufacturing (D The Safety Management System (ShS), as well as fronties granting them approvali with performing them here State Safety Programme (SSP).
the Ney orgalation	elements of the (CAD Sat	fety Nanagement concept which are new to fleet at
- sal	nty rick managiwhern (Kazarri ely performance massuremen forminge-based oversight; er en unnerster	identification, risk assessment and milioation); f and community improvements d
Uno/ame organisal organisal organisal organisal organisal organisal organisal organisal organisal	Interiori of Salety Managen tonii Nai il potential to brin tone ani competent authoriti immerse application of a pri missi mues 0 will also fell fety Management principlet. Novezes, Sintety Management fers.	Next principles. In product certification and E. & M is significant safety and curt benefit to hoth D & M lies due to better allocation and use of resources and castiverpredictive approach in safety allowing from m manutasis a even playing finit with competitions since the are going to be implemented verticities in all adjustor in implementation will certainly require investment from
the side / 11/200 assisting needs to	un of initial analothiness in A Descivitier referred to a rapionenting rules with the 5 the assessed and full compiler	The test domain within the scope of Regulation (CC, is the "Besic Regulation") where the conrealizability of allery Management proceptes faild down in Annax 19 shi is enceds to be established, as resulted.
This Con Herd The J	oept Paper is intended to faci best way forward,	utate discussion, among all parties involved, in order to
fugion in loss in loss ind and in loss in loss	ian (eC) (a. 7) (acc) al me base balan (ec) (ac 7) (ac 7) (ac 7) (ec) (ac 7) (ac 7) (ac 7) (ec) (ac 7) (ac	a ten service are of the survey of 10 paintage and tenerate when a first service definition fields, foundly, on framework, Control D and service definition fields, foundly, on framework, Control D (1) 40(4)(2013) 40 (2014) (2014) (2014), 41, 50(2, p. 34) (3) 40(4)(2013) 40 (2014) (2014) (2014), 41, 50(2, p. 34) (3) 40(4)(2014) (2014), 2014 (2014), 41, 50(2, p. 34) (4) 40(4) (2014) (2014), 2014 (2014), 2014), 41, 50(2, p. 34)

-	100000	concept
and in the second	Table of Contents	
EXECUTIVE	SUMMARY .	
Table of Co		
1 Denin	Centors of the INALE	
Z RATA L		
LASA L	ACKAPOLITA	14
12 40		1.6
13 44	Operations	15
TH AT		16
3.5 40	not comis	17
16 00	atinuing Airworthiness	.17
37 10	tial Ainvortiliness	18
37.1	History of (OL	18
3.7.2	Stert of HEM, 060.	.20
4 (01 m	cept for Part-21	
3.1 1.0	Loncept parameters.	
1.2 Ru	lemaking process for 10]	
4.7.1	Warking method	
\$ 7.2	Past ordinets .	
4.7.7	Delverables	
3.2.4	Time schadule for LOL	
13 10	Econcept open issues	
4.3.1	Apple apility issue	
432	Content of AMC/GM material	. 25
4.3.3	Output from FAA 21/SMS ANC project	
an m	etfaces	
4.4.1	Interface with Jusies Rett, 0550 & Rett, 0612 (5MS)	
3.4.2	Interface with the EAA 21/SMS ARC project	
5M5.co	except for Part-21	
LL SM	S consept parameters	
511	Part-21 SMS and compliance with ICAD	
5.1.2	Eart 21 SNS framework and its consistency with other dom	<u>ains</u>
511	Stucture	
CONTRACTOR	With Somesse Austan Libert Agancy, Alleght meaned	April Papert

	8.53	and a second	
	214	No stand-stone SMS organisation amountal	
	515	EASH SMS scope ve ICAD SMS scope for D & M	471127
	51.6	Consistency of SMS frameworks between DOA and POA-	
	517	Requiation of State Salety Programme (55P)	
	518	Alternative Means of Contrillings	
	\$1.9	Human Instans	
2	2 10	ernalising process for SHS	
	521	Working method	
	522	Ellal projects for SMS	PS
	5.7.7	(bell/rerebits	29
	\$ 2.4	Time schedule log SNS	
3	3 54	5 concent open issues	30
	531	Form of SMS Implementation In Part-21	
1	A Off	ter open issues	
	241	Construction of Perc-21 provenisation approvals	
	5.4.2	Fart-11 review to introduce performance based rulemaking _/v/ma/ves	
5	i in	Witness	
	3.5.1	Interface with pases RMT 0262 & RMT 0611 (201)	
	4 5.2 No 204	Interface with task HNT.0251 (NUM.055) - SMS for Communications)	n Regulation (EC
	5.5.3	Interface with the FAA 21/SMS ARC project	
8	Conclus	nion .	
1	Attach	limits	
A	Le	vei of involvement /LOTI project - Descriptive note	
	Co	mparison of the Annex L9 SMS framework with EASA SMS	
1	fra	mework	32

Concept Page	
	1. Identification of the issu
icharg the ICAG provisions for 'Sofaty Kanagement the scope of the Basic Regulation. This series from the I to the Chicago Convention (and its Annexes) which a needuly fallers into account in this Regulation and in rt (Anticle 2 (2) (d)).	The Agency is to the process of imp (SM) trial the aviation domains with intovision in the Basic Regulation relativity impress, to vinsure that 1. Its privity the roles drawn up for its implementation
concept, which was introduced in the past years o abov, orients, of two mean components. The State magement requirecontribution of contracting States, are implemented by specified eviation service or produc- rel ancient.	The IGNU Salvay Monogenine" (SM Inneral Americes to the Chicago Con Salvay Programme (SSP) with salvay Salvay Managament System (SMS) to provides in support of the sale opera-
a) is, the Agency has domaged its own EASA 55 the ICAG harmoworks to 61 Net specific needs of the ICAG harmoworks to 61 Net specific needs of the action of the Constant of ATHYANS, but adminish completed for the domain of ATHYANS, but adminish the idamain into full completings with the ICAG of process has been survived and is in various with a process has been survived and is in various with a ad Continuum Alawarchings.	Powel ou the RCAO SSP/SMS Instal community which is the adaptation IU/EASA twice and EU legal envolveme men completed in the domains of AI P AG crimptiant ² . The process has be calendaring affort will be needed to b provisions in Asings 19. The pulsemakt immerses in the domains of Acroditories
ing donuelt in which SM implementation lies to be ex 19. Certain mierrutking attempts were made in the aris kelow). The main make and ubjective of tim to ament Commission Regulation (Co) the 7nd 2012 is introduce therein the ICAO compliant EASA SM	Initial environtimens is the left remo- evaluated for compliance with ICAO A paid but they were unsuccessful () remaining relemating task is, therefore and its: Annex 1 - Part-71 in unle transmosts.
results achieved and respecterum gained from the 58 in particular the domains of Continuung Anworktinges 3M framework developed and edupted there will be abon in too fide of Ditual Anworktimess.	This submissions activity will build on implementations in the other domains. An Crew and Wr Operations. The EA considered as a model for 5M implement.
ain will recourt adaptations to fillaria specific reada o id a Destor Disputsition Annoval (000A) and/or a () issued under Part-21, These organizations have principles, as equired by Part-11, and run approach in these organisations. It is, therefore, insection that does not deflact from the current effective framework fely benefit. Previous attenuits to change the current confersional and resuming appointion on the skin gency steff involved. Therefore, we considered undu- tive internating status, all the million states and the cubernating activity needs to be emillioned into a w- a cupport a smooth 5th full implementation. The more	However, it's implementation in the all Nerv21. Many organisations already Winduction Organisation. Approval () almost empedded in number of the 3 her resulted in i-good atfirty means in introduction of the tail 50 pendip or yes. If in unneschang costs far no nerv23 system prived difficult rule in out-23 system prived difficult rule in of hoth statietholders and some of the is describe and explain rule, before private southers. Then the subsorption munique change management prices.
Second Strategy and an end of the second strategy and the second strategy a	These CU is rOA or TOAD South which Makes that which Programme (S.C.) and can estimate the CAD for any south of the estimate the CAD for any south of the theory (DAD) of the south of the comparison of the CAD of the south of the souther in the comparison of the souther the method programmed. Comparison of the souther



RM1.0262 & RM1.0011 and RM1.0550 & RM1.0612 (MDA.050) European Aviation Safety Agency Concept Paper representatives of a wimper of Connacting States as well as representatives of come international industry organisations (e.g. ICCAIA) and the RU. The Agency, more-sensing the EU, participate) (and continues to participate) in the SNII work. The new Around 19 (Editor 4) rear was adopted by the ICAD Counce on 12 February 2012. becomes effective in Aily 2013 and shart he applicable in all swatten domains on 14 Rowamber 2013. Amies 19 commidates all the general SSP and SHS provinions currently sproud arrow. Americas 1, 5, 8, 11, 13 and 18 and adds elements for the collection and last of safety sime and State satery oversight dottes. Anexx 19 demons two frameworks, one for FSP (see Attachment A of Annex 15) and one for \$15 (see Appendix 2 of Annex 19). These frameworks, we generally applicable to all aviation dominine. Including the dominin of Annesi a 'Airworthiness of an raft/ The general SM provisions of Annex 19 will be complemented, as necessary in each Annex, by domain specific 5M provisions. The Annexies will contexn appropriate cross relatences to the general provisions of Annas E9: An affected annexes will shortly undergo revisions to remove the general provisions new contained in Armas 19 in order to avail depincations. By creating Annes 14 with Salety Hahapement provisions applicable to all avietian domains. ICAG Timmented a new literarchical structure of SSP/SM5-provisions immediately after the new Amer becomes effective (July 2013), the Contracting States, metuding this EU MS, will be inquested by the Secondary Seneral to write UCAO of any Ultransies that with estst on 14 November 2013 netween their instantal regulations or practices and the annualism of the Stimulanda in Anneal 19. Such multilambian as to be made luglage us Deraber 2013, and Theroarter the Status shall notify the Organization of any further differences this anse furthes the Contracting States will also be invited to polity ICAD of any differences between their own gracking and these established by the ICAD Recommences Practices, when the multication of such differences is imparting for the fatety of an taxingstan. The Competing States will disc to requested to instally ICAO, before 14 Octoper 2013, of the date or date by which they will have complied with the provisions of the Standards in Arrea 19. market and its er f taxes

Entranse wheneve result whench	(MDALOGO (MDALOGO (Anocht Face)
3. EASA background	
3.1. General	
As explained in section 1 above, th PLA is unlight to origine half of Regulation take due arrived of Management specified in Ameri 19	in Agency, being responsible for availation poletrawing in the I the implementing rouns within the ways of the Bala 7 19AQ provisions, including the anavision for Spiely and the other affected Ameson
bi edition, the Besic Regulation livership, wit operations, ATMANS regularis aviation organisations in in subry, and aim fac combinities into	teelt, in its essential mutanmonts for environments, you and aerutiones (see live annova, is) the Basic Republication aplement and materials a Management System relation to revenees of the system.
To imprement the above provisions concept and model transvorii for adopted to the specific LU legal very	s of the Basic Regulation, the Agency has developed in own in Safety Management which is noty TCAO compliant or visonment.
This made, adapted to its form to experimenting raise to all the dam damene, the work is in different of mis-matters on the independence	to fit space domain, to gradpaily balog introduced not ma- more within the scope of the Basic Requisition. In different faths of progress (see Figure 1), Rekwe you one final ma- senate in each demain.
manual on the manufacture	sums in expression.
I Provide the second design of	anny of Mada to many
	"Man as the Well's and the state of the stat





Employer where a survey where the	(MDM.06 Ganogat Vac
and GMs are contained in ED Decision 20 GEB. Sections $\ell_{\rm e}$ 0, and 01	12/006/R ⁸ (AMC and GM to Part-ARA), Subp
The organisation requirements are conta Requirements for Air Crew. The SHS (General) and II (Management). Support 2012/007/RF (AMC and GM to Part-ORA).	aned in Annes VII 'Part DRA', Organisati milated provisions are contained in Section ing AMCs and GMs are contained in ED Detail Subpart GEN, SectionA Lant II
The organisando requirements, exception	ng me 5MS elements apply to
 approved training organisations (At commercial and/or private licences) borning, as well as reganisations ang 	De), i.e. all droatnisectory providing training maccinumice with the EASA coles on flight on ages in flight contraining;
- all holders of a Flight Simulatino Tinin	ing Unitice (F510) qualification bertilicato; and
- all nerv-metrical centres (AeMCs).	and the second se
4 fant, comparison between the ICAO Ann system provisions is available in the Ensla (Rf to Part/ORA) ⁶	ex 6 SHS numework and the EASA management natury Note: to the Decision 2012/007/8 (AHC a
3.3. Air Operations	
The rolemaking process for undienceding in adoption of Commission Singulation (EU) N Organisation and Technical Requirements october 2012:	Desition All Operations has been complianed with the paints/2012_of 5 October 2012 deriving Authors for air operations. It entered into force bin
Requirements for Competent Authenties a Requirements for Air Operations. The SEN, Section I (Control), Section II (Nau and entercement), Supporting AMC and Support GTN, Sections 1, II, and III	ve contained in Annex II "Part-ARO" Apthor ISSP printed acovisions are contained in Supp agriment) and Soction III (oversight, certificati GPs are contained in ED Decision 2012/016/
The SHS related providing for organisati "Part-ORO" Organisation Requirement (General) and B (Management). The supp 2012/017/99). Subject GEN, Sectors 1 and	ons of air cuestoos are contained in Arinex I 5 for Air Operations , Subpirit GEN, Section ording AMC and GHs are contained in ED Decla II
The wpanisation requirements, enumprassi	ing trio SMS clements, apply to:
 all operators who are required to not rules for air operations; and 	an ADC/organisation certificate under the new
 all operators who will be required to non-commetcal operations of Completions 	destare their adjivity under the new EU rules a Motor Rowered Altaraft
3.4. ATM/ANS	
5M related requirements have been adopted	t and ane contained in:
The second section of provide the second sec	- 2014 Annual Charles at Sent at 2020 Annual Sent 2015 Annual Charles at Sent Sent Sent Sent 2017 Annual Charles at Sent Sent Sent Sent 2017 Sent Sent Sent Sent Sent Sent Sent 2017 Sent Sent Sent Sent Sent Sent Sent





Ē

RMT.0262 & KMT.0011 and MMT.0550 & KMT.001 (MOM.068) Concept / Upe	mapadani divlasticos Santatly Againey	Ente
The opplicant for design approval is (under 21 A.257) a allow the Agency to review any report, make an in and ground test necessary to check the validity of the original to resolve this conflict and provide uniform, in the programma, a grosser legal containty on the level o process.	then ventication At the same time 1.71A 33 (2) and (d)) oblight t vector, perform or writess any fill A compliance statements. The task ph or approved or the cattleador have involvement to the certification	turti and liqqe bolk stelp Agel
(with different views being regressed rule) internality a visition commentity. Following the publication and 006, some fundamental changes in the concept were all declate. The ESC participated to this debate on man 2007 - D6 May 2008. The external Review Group set in cedure did not support the proposal.	bask was contentious from the un- lier the Agoncy and in the was immose moment on VEA by Dr2 roduced, resulting significant mas- serve between the period 04 the second mose with the milemaking and coordinace with the milemaking and	The write cone infro
Inv CRD No. 16/2006, which interrupted to establish a g product approval processes by adopting a mak base of responsibilities between the applicant and the Agancy effectives.	In the accorded proposal, covered impa between the organisation ar mach with a clearer democration a not equest, Among (he main colu	(vien beitin optio diptio
respective roles and responsibilities of desarr approva t certification many and DDA monsplit tears may lead certification role of the Agency;	The sads of clerity regenting the applicants/huiden, DOA, problem to unerlags or gaps in the owner.	ŧ,
ten will reduce safety (product castification considered her autiling organisations). Experience tas shown the net always be of the standard expected by the Agency confidence in DOA by certification seams who view both a decign organisation should be in addition to and your oct itself.	The perception that the approximate efficient in safety terms in compliance documentation may this tasked to a general lack of as too remote and that escrean sobstitute for absenced the prod	
DiOAs (due to recordenisations, metablicity and lack o	The himited confidence in sume resources);	
ocy involvement will lead to a loss of Agency teamina knowledge, standaritisation between projects and	The concern that elimited Age compatence, protect specific credibility,	1
iquires grader coordination between the DOA town are perits. Princedures would name to be developed to ensuin 1	The concern that the proposal r specific PCNs and certification at this countination is effective; an	1
al agreements which could result in the next to an on the validation withouty	The probability implact on bilater increased invest of involvement for	-
elivity was contained in Opinion No 03-2010 which only of the proposal. This part, later adapted time Commission and immed changes to Part-21 related to the obligation at to interent to the Agency or certification, imagination temportarillon, and to keep this document updated as easy Additionally, DDA iniders were given the involu- cat Flight Nanual and as supplements. The rest of the is sequence of events implements the rest of the is sequence of events implements.	That outcome of the retensions torsed ton non-controvertail part putation (EU) the 748/2012; introdu- the applicant for a design appro- ailing the means of compliance in essary during the certification pro- ipprove introduce outcome or the Au- poration LCG was put on held. The retension control to width a color	The local Reduction of the detail in period chain
in our second	nity and the presence index other pro-	1.1-





-	urberd wetenen struit Allenet.	IMON.060 Concept/Sev	
	(2) the level of performance of th	e design organisation	
(d)	Is introduces the possibility of extending, under certain conditions, the Dr0A arryinges to approve mater changes to the Type (beign (including changes to associated require manuals/dam), major reports and \$TC.		
to)	II estillations product pertification and design argumention approvals as two processe sharing the same common tisk-based elements;		
(07	It improves the clarity of row Part- product certification and those appl	It improves the clarity of the Part 91 feet, by expecting the inquirements application to product cartification and those application to design organisation approvals.	
(9)	It clarifies the responsibilities of the process. The applicant remain (all certification basis;	e applicant and the Agency in the product cartification y responsible for the compliance of its design with th	
100	It better defines the responsibilities of the Agency in product certification. In peritodal the controllation of the present test in 21.6.35 (d), 31.6.257 (b) and 21.6.253 (b) resolved. The current privilege formpliance documents should be accepted by the Agency without further verification' is replaced by a risk-based determination of ins level o undetermined.		
10	It is consident with and estimate the correct DOA concept with all the new element introduced, as appropriate, including the descention of the privileges:		
00	It supports innovations by examine the industry to safely discrored their responsabilities in a manual appropriate with the extended supply chain models used by industry.		
(11	It introduces a consident and proportionave demonstration of capability for all design impanishment (including these applying for EFSDA)		
d)	B free the potential to make the DOA concept effective and projectivester for 5mail an Neution Enterprises (SMEs), in terms of complexity and provides		
R 14	assumed that the UO1 concept will be	e applicable to	
E .	appression for TCa and APU ETSCH	ann	
8	applicents for major/minur changes E) ann ETSDAs (Subpart O)	a and mejor/minor repairs, 57Ca (babpanta D, M, an	
la z nepti Pres	he first phase, the locals will be an icents/bolders. The applicants for/ho edures to DDA (APDDA) and under a	(21) implementation in certification projects of UC oders of design approvals designing under Alternativ etification throughtme only will be addressed inter.	
Han com dive Ceri sant unici	corregol does not change the we privace with the type-certification is and of work in the cartification p likation staff and DGA staff) or wol 6 5 DGA staff) but should improve to mailton and encourage threfs integra	countability of the applicant for deviationary for asis. It ains does not comparitie respective roles on process within the Agency team (i.e. between the in the applicant team (i.e. between their Certification he callaboration between this teams, the exchange of their	
Fue His min	hurther details about the LOI concept Concept Paper. The concept is accordinges.	I by Use Calification Directorale, and Atlactament A L enganised with the definited traffic proposal for Part J	
-	No 2019 To 12 Barriera Indian Taking Against . 2019 Supervised, Copper you sign categories	an a state research and the construction of the state of	



a.2.4 Time sch	edule for LOI	
4.2.4.1 Time The time scredul	schedule for implement to the IRs is proposed as	ing rules i tollows'
April 2/313	Start of the LDI SG for plint projects	
May 2013	Adoption of tins Concept Paper (hoth on LOT and SMS) by DSC/LETC	
June 2013	RAG/TAG/SSCC are consulted on the Concept Paper + arefe TuR	
iune/Joly 2013	TOR is published, NPA 1 (IRs only) internal consultation	
Aligiett 2013	NPA 1 (IRs only) is published (short-enet contaillation period wat for contailmaid)	
02/2014	Optinion (+ CRD + mant)	Commission Regulation) is published
Q4/2015	Commission Requisition is adopted Vestimate)	
4.2.4.2 Time The lime sciential	-21 iulu changes, civitar be appliet. schedule for AMC/GM m e for ANC/GM is proposed	o cernicanun procedures, devore ine i r domis, auch es lhe enviaged ext alerial Wikkes
4.2.4.2 Time 4.2.4.2 Time The line scientul up/2011	-21 into changes, civitar be appliet. actiedule for AMC/GM m e for ANC/GM is proposed	o conviction procedures, devote the r r domits, such as the envisaged extr aterial Weiwes
4.2.4.2 Time 4.2.4.2 Time The lime scientul (Upren) 1 1) 1/2013	-21 rule changes, civitae be appliet. schedule for AMC/GM m tor AMC/GM is proposed i t t t t t t t t t t t t t t t t t t t	o centrication procedures, denote the , i domits, and as the environged extr aterial whoms fait of picol projects, the "Longood aper and Toik are approved (rait domiting MPA 2 (for AME)(3) miterial) based on the output thim pico
4.2.4.2 Fine car 4.2.4.2 Fine The time schedul ((2720) i 5 1/2013	-21 Julu changes, civitau be appiller. schedule for AMC/GM m e for AMC/GM is proposed for AMC/GM is proposed f f	o centrication procedures, denote the r docaids, auch as the enviraged exis aterial lookwas fant of pical projects, the Loncoor aper and Toik are approved fant dombing MPA 2 (for AMC (cen refertal) based on the outpot them pico rojects
(2)/2013	-21 Julu changes, civitau be applier. schedule for AMC/GM m tor AMC/GN is proposed i tor AMC/GN is proposed i f f f f f f f f f f f f f f f f f f f	o centricentor procedures, denote the , i domits, and as the environged extr aterial tert of picol projects, the "concept aper and Toik are approved tert aperting MPA 2 (for AMC)(3) miterial) based on the output thim picol rojects IPA 2 published
(2),2015 (2),2015 (2),2015 (2),2015 (2),2015 (2),2015	-21 Julu changes, civitau be appillen schedule for AMC/GM m e for ANIC/GM is proposed to 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	o centrication procedures, denote the l intentis, and as the environged extr aterial working tart of pilot projects, the English are and Toik are approved tart anothing NPA 2 (for AME/Cat reformation means) based on the output them also rejects IPA 2 published rath Decision ready
1002 01 100 000 00 01.024.2 Time 10.02011 0.02011 0.02013 0.02013 0.02015	-21 rule changes, civitae be applied. schedule for AMC/GM m e for AMC/GM is proposed for AMC/GM is proposed f f f f f f f f f f f f f f f f f f f	o centrication procedures, denote the i intential south as the intenaged extra aterial withins faits of picor projects, the Except aper and ToR are approved trait duriting MPA 2 (for AME/(an reternal) bread on the outpot them picor rolects IPA 2 publisher rolects IPA 3 publisher rolects

ŕ

Turopean Aviation Setery Agency RMI.0262 & 4MI.0011 and MMI.0550 & MMI. (NOM Loncept
(ANDIGA) or, for ELN 1 arriv, by presenting the certification programme. The essue also inc applications for L150As.
4.3.2 Content of AMC/GM material
The Certification Directorate envisages in their concept for LOT their a ingritionity amon AMC and GNS will need to be produced, in particular in the following iteration.
 Control and level of detail of the sertilication programme;
- Criteria for calamining the level of performance of design organisation;
- Critecia for determining the criticality of an item of the certification programme, and
- Oritma for viewermining The mivelay of an item of the certification programme.
There are other candidates for AMC/GHs such as :
 Definition of a mechanism which will assume a smooth respection of and a interactions between the different purcles involved in the product commandan ection to established (applicant, LASA 200A team, and EASA Product contilication term);
 Definition of a mechanism for the minaigement of certification programmer three an established, on what basis, in which form, how they are sepreved, how they a amended, etc.):
Enters for the determination of the depth of involvement. For a specific item o certification programme, the determination of the LOI may not be (list "Yes/ko" but deep".
 Criteria and catelitions for granting the provide to a DOA trailer to approve t thempet, minor repairs and/or STC under their DOA;
NDTE) before tarmal addotion of the Commission Regulation with the ruleted LOI Part 21 and synchronised adoption of related ED Decision with supporting AMC/GM and Destituation Netros may be used to allow the LOI process to be applied to indi- certification projects nowever, DOA balance will unit be entitled to the LOI related prior after the Continuanter Regulation is formally adapted.
 Access of the Agency certification learn to all commitmic ancuments whether or ne agency is monivert in the task.
 Reporting adligations of the Applicant for significant events (led index dama certification process, and
 Differin and procedure for reconscience the LOI for a specific dem or items a certification programme.
4.3.3 Dutput from FAA 21/SHS ARE project
The debails of the LGF concept may be influenced by the orthogene of the FAA 21/SHS Av intercenting "Commenter (ARG) project which has been set up to provide FAA recommendations on how to amend FAR Part-21 and Part-5 to implement performance- oversight (Level of Involvement) in modulit certification and SHS in 0 ft 41 organisal there is a need to align the results of the work on each cide to support harmonication mitual remignition of performance-based oversight (Level of Involvement) based on equivalency.

RM1.0202 & RM1.0811 and MM1.0550 & RM1 (MDA Turopean Aviation Substy Agency er Heer 4.4. Interfaces A & 1 Interface with tacks RMT 0550 & RMT.0612 (5MS) The vertexity under Loke BNT 0262 5 0017 0211 on LOT struke is the else surfacement with the a fieldy under tasks 0x1 0550 5 0m1 0x02 for the imprementation within an particular s-address the design argument of existing automatic else factor considered struc-detormining a LOI in contribution projects. 1 4 2 Interface with the FAA 21/SMS ARC project The LDL SC should establish contacts with and manifer the aut & 19 for 8.04 (1)/945 480 (proof.) on the implementation of the performance based isorrangle due to a fair-

RM1.0262 & RM1.0011 and HM1.0550 & RM1.0612 (MDM.050) European Adation Substy Agency Environt Fater 5. SMS concept for Part-21 5.1. SMS concept parameters S.1.1 Part-21 SM5 and compliance with ICAO the transwork of the SMS implementation in Part-21, including its oversight, must be ICAO compliant. This will be assured by conducting a thorough gap analysis between the criteria tal-Annex 19 and Parr 25 to identify SHS/SSP elaments that will used to be anneal. 5.1.2 Part 21 5H5 framework and its consistency with other domains The framework of the SMS implementation in Part-21 should be consumed with the UMA model transvork as implemented for going to be implemented) in the other domains within the scape of the Basic Resultation. This is required by the priorsple, generative anomatic to tasia, of a Total System Approach. This system requires that aviation system components-(organisations, evidems, minition, products, inguistions, atc.) are addressed in a uniform or, at best, consistent way as they are all part of a single aviation betweek buch system to states, interroperability of the system components and innovan its overall performance. If eliminants or minimum over for safety gaps, conflicting or overlapping explorements, conflicted responsibilities monotequetations etc. Applied to a single organization moliferg multiple FASA organisation approvals, a supports smooth integration of ristlar opprovals unsertheir company integrated management system. Therefore, Internally (within Part-11) and extendily (against the other EASA domains) the SMS/SSP hameworks alongo an construct. However, a consistent approach does not mean that all implementation asserts railable the Land. Differences, where supported by an appropriate citionale, can be activenesizable. D is ing invented to represently enforce a specific framework aspect where it is inopimumine for (IBP) (add also 5.3.1 below) **II** Structure When introducing the SMS/SSP regularments into Part-21. Its emisting structure (its division for Sectors and Subparts) will be respected as far as possible; unions unlifted otherways. 5.1.4 No stand-alone SH5 organisation approval The existing organisation and authority requirements of DOA and PDA approachs will be amended to embady those 5M5/SSP elevisority which are currently missing toese. The continuopprovals will be amended but the Agenry does on foresee a separate films approval. 5.1.5 EASA SM5 scope vs ICAO 5M5 scope for D & M The ICAO SARPs, in in Annex 19, apply to organizations designing of minutacturing inscattly unly. ICAO has already communicated its plan for teation 2 of Annex 19 in ordered its allow to insule engines, properties, and some other where establish introducion on poneous. The Agence's approach is based on the essential requirements of the basic Regulation and require implementation of a Management System with safety aspects from all raddees of or applicants. to EASA organisation approvals while applying it to the full scene of these approvals. In the unial seworthiness domain the organization of SHS will be required from design organisations designing under OOA and production organisations producing under PGA and we cover their half approved acope which may pover products and/or their changes and regains, and parts and appliances (actualing £750, and and to avoid a model a prominent dependence rachely dependent, main plane researches, states and second and the second s

_	Incluein division Serety Agency RMT.0202 6 RMT.0011 and RMT.0550 6 RMT.021. (MDM.060) Concept Face
State on the	principles will also need to be implemented. In zonie propriotienete comi to be explored, or ne designing under Alternative Processner to DDA (APDGA) or under presentation or tification programme unity. In the manufacturing web, the SN principles will need to be servented by holders of or appleans <i>h</i> = a fetter of agreement under Sobpart F of Part 3 produce products, parts, or appleans without POA. Bit how assistly SNS will be immented in these fields is an open usar.
٩.	6 Consistency of 5MS frameworks between DOA and POA
	Interconnect that concentrations wholey or partly involved in aviation D&M are likely, as will agency, to have a boat system approach to safety management and as such will have on insgement system transeverk for their complete organisation, which may extend beyond withtom to be a principle of the EASA SM implementation runemation at every afternative with a making no complete and at their 21.5M processes into a adder transevery, while making no completes on attention of Part-21.5M processes into a adder transevery, while making no completes on attention to the ICAO Armes if numericants (see also 5.1.2 above).
O M D M D	resonantly, as too as then contents is concerned, the resulting DOA and POA SMS/SSI memoriks need to be consistent. The eacht way how these tramewings will be acceptible the relevant Part-21 Subparts (), (), (1) in an open issue. Another open issue is it and how to wove, apart from SMS, the overall consistency of the DOA on POA requirements (see also is below).
'n.	X Regulation of State Safety Programme (SSP)
NAL DE MAN	(EASA aviation marks, and them is no plan to change the that demonstrate being calculated aviation domains and it is due to legal reasons as well as due to a well not to interfere with current SSPs recently implemented in the MS. Consistent the SSPs of the EV MS need to led the adjusting EVEASA aviation legalative transmoverk as well as the Evopean Aviation ledy Programme (EASP) and alian (EASP) designed to support the SSPs of the SV effect researcedly. The requirements for complement authorities (for Section 8 of Part-21) will no addy refer to the SSP for well support their implementation by the States.
5.	.5 Alternative Means of Compliance
Thun b the mon man	5.5KS/SSP concept for EL6.4 M improviduations will incorporate the EA5A concept of voluments of ulternative means of compliance (withoc) with the implementing Side), as adopted a minimum volument to the conjunctions in the track involution (e.g. Air Crew and Air Operations). The is offer Die M expanded are seen in molitiky an provider to the other service movaler tails extern Apoptoch), its events a monor of the alternative for proposed by the information of the alternative for C proposed by the information of the alternative for C proposed by the information of the alternative for C proposed by the information of the alternative for C proposed by the information of the alternative for C proposed by the information of the alternative for C proposed by the information of the alternative for C proposed by the information of the alternative for C proposed by the information of the alternative for C proposed by the information of the alternative for C proposed by the information of the alternative for C proposed by the completent information of the alternative for C proposed by the completent formation of the alternative for C proposed by the completent formation of the alternative for C proposed by the completent formation of the approval, it may be alternative for completent formation of the alternative for C will reactive the provale an equilibriated level of safety. The requirements in the trace of altifices.
5.	1.9 Human factors
114	i scope of MOM 000 hands will include foe iembodimount links frame 25 of Homen Fectors is agreed from the other annulate dominance, p. Fart 145) to fit Fart-21 and the 0 \pm 5

Emphany wetanon Preside allered.	RM1.0262 & 4041.0011 and 4041.0550 & 4047.001. (MDM.060, Concept.Paper	
organisations. Homan Factors are c response to commends on the initial (Group.	currently masting in Part-21 and their incursion is in fail and a recommendation of the Turopean HF Advisor	
5.2. Rulemaking process for SM	45	
5.2.1 Working method		
The working metricel will formally be and NAA stakeholders will be assure State by Broup with reduring and has and monitor a number of the 5165 memorie diate s(45 miles for part 21 an investing diate s(45 miles for part 21 an inter TAG at 0.4. M Sub-SBCC, world modesary. This will be stated in the p	The 'Agency' but adequate Involvement of the Industr- et by various means. The Agency will set up on SM (representatives which will be presently tasked to launce plot projects (see below) and getter invierventation of projects (see below) and getter invierventation of related AMC/GM material. In addition, consultations of Simps, focused, consultations do: will be consultationed emicet FoR.	
5.2.3 Plint projects for SMS		
In a similar way to 1.01, the Agency implementation of SMS in D & M orgo of line rules and AM2/GM material, an the Agency in their drafting. The em Standardisation (Triclia) Artworthiness within the remti of EU fMAs, their shift	c plane to lowerth a set of what empirits for testing the presences. The pills empirits are means for development is the SNS SS in change of the pile projects will approximate you require anythermost of the Agency's PLA on the start because contours POAs on the PLANS on it was testing anythermost and POAs on the PLANS on it was also be invited to participate in the pilot project.	
The profile of these plicit projects an Resty as follows	id participating ID IK IA organisations will ideally re-furb	
 CS-25 aeropiane (TC/significant) 	CS-25 acropiane (TC/significant/mojor change protect);	
CS-23/VLA/LSA aeropiane (TC/skpulicant/major change propert);		
 Engenn (Co segnificant/major change project); Rotorcraft (TC/ significant/major change project) 		
		 Hon-TC holder marter change (ST
- Part of Appliance Involution		
5.1.3 Deliverables		
Une Opinion for Part-21 employmentil generate The draft rule energies and on contained in a single NPA leading in	ag rules and one becalan for AMC2014 material will be a research AMC2/CMS will be developed semillaneously to b the Opinion and Decision	
5.2.4 Time schedule for SHS		
The time schedule processil for the SM	15 implomentarion is as fullows:	
Muy 2013	Adoption of the concept Philer by ISC/TETC	
June 2013	TAG/SSCC consulted on the Consegl	
programmer and the previous doubted subset depen-	and the state of the	

CALIFIC COMPLEX	La La
	Pamer = draft Yakt
lone/July 2013	TOR published
Q3/2013	Start or piket projects for SMS
01/2015	HPA (usid) TIL & AHC/GH) published
01 (2016	Osimien + CBD + draft Commission Regulation 1 draft FD Decision publicities
03/2017	Communion Regulation adopted (estimate) + ED Decision adopted
j017	Full limplementation of 945 (synchronisati with 170) unplementation).

5.3. SMS concept open issues

S.1.1 Form of SMS implementation in Part-21

The form of implementation of SMS into Part-21 may differ from other domains because of the questic structure of Part-21. Since DDA and POA already turbain a minuter of elements upon which to build SMS up competent and mained presimes (addition, a beef lack to accountable minuters) compliance monitoring system, etc.) the unbodiment must be such that their will be no superfluores or contradicting requirements. To instant this, a fracturing gap adalyses was to performed with the Annex 10 framework. D will need to be needed whether to only complement the costing requirements of the relevant Part-21 Subparts (a, G, and F) will missing SMS/SSI provisions place by prese or to introduce (e.g. into Subpart A – Termari Invision Comments for all the other Part-21 Subparts) a full set of basic high level impurpresents on the Hampement System and adapt Subjarts (a, G, and F to avoid obvious uppriminations.

5.4. Other open issues

5.4.1 Consistency of Part-21 organisation approvals

The meet by SHS implementation in ODA and 20A engineerance offers agon the opportunity for implementation of the old JAY initiative of restoring consistency of various orbital inguisation approvals, in the formework of Pert 21 field, it would apply to the equinementaof ODA & PGA approvals and production without POA (contrained currently in Subparts 1, o, and 1). The aim would no to make these three approvals more consistent of home without situations, matteries and terminarings. These approvals were originally developed for DAR 20 hy the TAA progressively and mare at less independently (as were the other regardiation approvals for other JARs) and that the regulatory material varies in mony approxitation approvals for other JARs) and that the regulatory material varies in mony approxitacionation the induction approvals became apparent while implementing from The TAA recognised this induction approvals became apparent while implementing rises in the other "Consistences of Degain abon approvals became apparent while induction in front report "Consistences of Degain abon approvals (CDIA report) and the figure() later, based on the COVA report, progressing the same with an Advance NIVA (#-fD/A No 15-JD00) bollowed by it

Tanal, april 2015 Status Indian Colling States, al faith memory)

100.00111.02
RM1.0262 & RM1.0011 and RM1.0550 & RMT.0013 (MDM.050) European Aviation Safety Agency Concept Fages CRD. However, the implementation of the idea was deferred. For the polyose of this rolemaking project these are two ophoses Option 1. Use the opportunity of 9MT 0550 task with the area to-lowed-one the 9MS related repowersients in a consummy way and amand also the axisting remaining requirements is hitiparts 1, G, and F to inclusive the overall connectoncy of the ODA and POA approval. Wivantages of Option 1 Consistency, well implemented, should improve transpirency and clarity of regularments and reduce main for misinterpretations. Bolli file D & H industry and the competent authorities. may profit if the potential for salety and cost benefits is realised, particularly where organizations hold more than one approval. It is tikely to support integration of born design and production management systems as consistent initials under a einole 'integrated mmagement system of the organisation. Disadvantuges of Option 1. If extends the scope of the tasks and scene of the necessary changes to Part-21. When and implemented sensitively, it could have an administrative impact on DDA and PDA communities init both staketoidars and the competent authorities. Clution 2: Limit the basic to strictly necessary changes to Part 21 for introduction of SMS (Imoples Advantages of Option 2 It might simplify conduct of MDH 060 taxs and limit changes to Part-21 to what 6 strictly impresent by ICAO Ariona, 15, it might avoid industric stakeholders having to make changes to Unit established organisations based on odministrative rather than safety related reduces. Ulsadivintages of Oplion 2 Lowing the issue of consistency of organisation approvals for (in) follow rulemaking host(ii) might result in more work for DMM organizations to align their internal single SMS with inclassistent requirements. It might result in a situation not uptimised for safety, given that Obtion 1 was seen as having poperical to improve saliety. 5.8.2 Part-21 review to introduce performance based rulemaking principles The medito introduce rule changes and potentially transfer rules within the structure of Fari-If to implement SMS raises the question of whether to use the opportunity to make a full review of Part-21, and veloces AMC/GM in order to check and, if needed, improve its quality by introducing the principles of performance-based rulemaking (Option 1). The objective would tar to dearry distinguish between the essential safety elements to be kept in Part 21 and the numessential implementation aspects that simuld be transferred to the AMC/BM to Part-21. This electronitive option (Option 2) woold the multiple the energias to (a) future rulemolog testion) Preterred potion: Clation 1, including Part-21 review, to inflictuse performance-based rolemaking plinciples. The extra work for arganisations with guisting DOA and POA approvals reads to be millinated by ensuring the revised, consistent rule and guidance waternal is no more prescriptive than it needs to be. The basic style and agenoach of Part 25 should not be changed n professionen (s. 19. prominise mediter talen) ärjenen et sin steller remainer Annen av den et start i start i et steller av de steller et sin steller steller av de stelle present/juszymi Auto 21 of All

RM1.0262 & RM1.0011 and HM1.0550 & RM1.0012 (MOM.058) European Aviation Safety Agency Concept Factor 5.5. Interfaces 5.5.1 Interface with tasks RMT 0262 & RMT.0611 (LOI) The activity under tasks RMT-0558 & BHT-0512 on SMS should be clearly coordinated with the activity under tasks RHT-0262 & RHT-0612 on LOL. For example the level of havdversent of the Agency is certification projects of Design Organisations that have artificed lives compliance is relevant to both tables. 5 22 Interface with task RMT.0251 (MDM.055) - SMS for Commission Regulation (EC) No 2042/2003 organisations) The activity under tasks RMT.0550 & RMT.0612 on SMS implementation should be ardinated with the activity under task MON.055 in implementation of SMS by organisation within one scope of Communication Regulation (EC) for 2612/2003. The aim is to implement connected SMS systems, in publicular, in organisations having publicle approvals and to allow computent authorities to streamline and patientiaus their procedures. 5.5.3 Interface with the FAA 21/SMS ARC project The future SMS SG should establish contacts and munitor the work of the FAA 21/SMS ARC ymud on the implementation of SNS and performance-based oversight in FAR Raw 11. Puropean D & M Industry will profit if burn the US and EU SRS systems (and ultimates) the one systems of other countries) are as close as potable, providing that the combinant syllenis are repropriate 6. Conclusions (It) be completed with conclusions for the open course and, it mested, changes to the asiamptions.) 1. Attachments A Level of availanment (LCII) movest - Description univer-B. Comparison of the Associate SMS (barrenoris with LABA SME Assessor)

Canosat
ATTACHMENT A
LEVEL OF INVOLVEMENT (LOI) PROJECT DESCRIPTIVE NOTE
ionale for the proposed changes
I publication of the protoci Part 21 sucluded as the Armos to Commenced Regu- 46/2012 feet clarified the duties of application in protect certification activities is instituted type certification, changes to 7C, supplemental type certification protects, Comparent technical standard options).
and reveaue of Part-21 acts at improving the content regulation in the Isi
tak netroduction of a risk-based approach both in product certification and o manhon approvals:
aling the Agency to determine its lever of involvement in product certification is all connectency tangible criterin, based on sately rely.
afuction of the passibility in approve trajer changes, major repairs, and STE in privilego, under center controls;
bisinners of product as Disation and design impaniation opprovide as law as esses shelts) the same continion rela lased elements; and
reventent of the clearty of Part 21 (text, by separating the requirements opport det certification and ilexign organization approvalu
pe of the proposed changes
ciple of the proposed changes
and changes be not change the rules and repartition of the looks below and the Agency for product certification articities when an TALE variation from
iple is that the Agency will review and verify the demonstrations of come by the applicant according to a predefined over of involvement (101)
In this Agency does not verify, or, alternatively, when it is saturied an location remich may need completion of several iterations), it does not sever ement, row any estimatement. The explorance process's with its finel declarat planae, in response to which the Agency bases the approval contribute. This ent Pan-21 process and it is not modified; and
n, on the other road, the Agency is not satisfied after its verify action it node it is indication of demonstration of compliance, which automatically results in a ng and blocks the issues of of the certificate until the demonstration is modified be accepted by the Agency. This element is new.
IV, to awind immedies any administrative twinder without any added recrimical validate is introduced to allow design imperisation approval (SOA) holders to is mator design change or introd approval or STE, when the Agency has decide lead in the verification of the demonstrations of controllance.
SHOLY Exception Analog Salvas, Advantacional Analogo International Advantacional Advanta Advantacional Advantacional Advantacional Advantacional Advantacional Advantacional Advantacional Advantacional A

	Ganoph Fax
2.2. Impact on the chronology	of certification projects
Cardilipation projects lyncally comprise	e the following romanilogical physics
(r) Farmilarisation,	
(3) Tstablishment of the type certified	cation basas
(8) Establishment of the pertilication	a pointrammer,
(4) Compliance demonstration,	
(1) Decertation of compliance, and	
(ii) Final report and essance of the	cerificate.
The LOJ project builds on the intro legislation (EU) No 748/2012. It dos establishment of the certification and the LOJ by the Agency.	duction of the certification programme by Commis- es not, therefore, madity the chronology except for to gamme phase which allor includes the determination
1.3. Scope of the changes in Pa	wt 21
This project tuny takes advantage of main symmetrical paragraphs.	the structure of Part-2), and essentially comprises t
 Elements related to requirement essentially contained in paregraph auded, one of which coming trents 	Is set on the applicant for unidust contration in 1.21.4.70 For this purpose, two new clauses have be an existing peraprephy and
 Elements relided to requirements resentative contained in a new part 	set on the Agency for product certification, estivibles / agraph (1):8,206
An additional cause is added to para elimitithe Agency is hat satisfied with	graph (1.4.21 to prevent the istance of the centric the demonstrations of compliance
middler editional blanc nas neen a miling), between type pertitionition an compliance b formally rejected by the	odel (o 71,4,75)) (o introduce an automatic (int (o 1) 8 design arganisation appravol when a demonstration Agency.
ricolity, perceptopic 21.A.263 has be aliantations are misposed in perceptop	en modified to introduce the new DOA primleye, in In-21 A 103 21 A 113 and 21 A 037
2.4. Acceptable means of comp	llance and guidance material
A significant amount of AMC and CM damains	will need to be produced, in particular in the follows
- Content and level of desail of the	e certification programme;
- Criteria for determining the level	of performance of design organisation;
- Criteria for determining the citte	shiry of an item of the certification programme, and
- Onuria for betermining the new	ity of an item of the conflication programme.
The soft own material is loady to b discipline (or group of disciplines) cooperature with industry, B will be results of picture projects". These may concepts are retrospectively tested, o which the LOI concept will be applied. Interprotection takes in the applied.	C dependant on the type of product and the enjoy 31 is produced to develop the soft any material based on the current acts produces augmented by 6 consist of elements of angeling projects on which i or consistery new projects (in elements of projects) in Allights hearing in the statements in ASA temporation. See 1970.

(MOA) OI (Concept Fo
mpatibility with OSD
resol changes have been tested and lated compatible with the 0517 requirement i on the process to be incorporated in Part 21 according to the most month version minim.
oposed changes to Part-21
.1 contains the typographical conventions land to describe the proposed thinnou the following accounts respectively address.
riges to section A implementing the LOI concept for type certificates and resilva
nges to section B initiatienting the LOI concept for the Agency.
right in section A estimating the LOI monoph to Subharts D. E. H. and D. and
nges to section A implementing the additional primingle described in the proper allest paragraph 21 A $2h3(r)$ (8).
pographical conventions
and convertions are used at this part of the decomment.
nanged last is in this fait.
inged or additional text is in this fant, and
11 1 1
posed changes to section A implementing the LOI concept for TC an C
Compliance with the type-certification basis and environmental protoct requirements
applicant for a type-cartificate or a restricted, type-cartificate shall demonstre plicative with the applicable type-cartification best, and environmental protect alemonts and shall provide the Agency with the means by which such complia- been demonstratester.
-opplicant shall provide the Agency with a contribution programme detailing an for compliance demonstration. This document shall be updated as necess- ing the certification process.
applicant shall record (astification of compliance within compliance demonstrations) according to the certification programme established under point (b)
applicant shall declare that it has demonstrated compliance with the applicable ty iffeation basis and environmental infelection requirements, according to iffeation programme establishest under point (b).
one the applicant holds as appropriate besign organisation approval, the distant solid (d) shell be reade according to the provisions of Subpart 3.
55-55, Subject rotation when Agencia as lighty reserved.

page any according a general sector of the s
The Applicant stall tomely allow the spency to review any centri and make a inspection and perform or witness any hight and ground test necessary to check r would of its compliance perministrations and to determine that in badure characteristic makes the install for the uses for which pertitication is requeste as defined by the level of involviment specified under point 21.8,200 (b).
The Applicant shall requising inform the Agency of the progress and or any signifier anticuity annumbered whilet demonstrating compliance.
21 Issue of a type certificate
pperant shall be entitled to trive a product type certificate exactly the Agency after.
demonstrating its capability in econdence with print 21.6,14.
submitting the declaration reterred to in point 21 A 20/d); and
demonstration that
 the product to be certificated month the applicable type-certification beam a givenumental protection requirements designation in acceptance in points 21.4.17 and 21.4.19;
(2) any airworthiness provinions out completed with are complemented for by factors to provise an equivalent level at safety.
(3) no instante or classectentile makes it unsafe for the uses to which certification requested; and
(4) the type-certificant opplicant loss expressly states that it is prepared to certifie w soles 71.4.44
In the case of an accretit type-certificate, the engine or propeller, or both, if instabilit the alrecaft, have a type-certificate usual or reseminat in accordance with th Regulation.
All demonstrations of compliance subject to indency intrification exponencing, to the 21.B.209 (b) have been verified to the instraction or the Agency.
23 Issue of a restricted type certificate
da - I
and the second se
All demonstrations of compliance subject to indency varification according to 20 21,8,208 (p) have been verified to the satisfaction of the Agency.
33 Inspectors and tests
The second
(s.d)
StackS-Story Sunder Infactor Server Agenci, All Institu Internation



	(MOM.OG Canosof Vije
	1 - 1 - 0"
100 - 12	$(a_1, a_2, a_3) = (a_1, a_2, a_3) + (a_2, a_3) + (a_3, $
hi) R	in feats performed or willnessed by the Agency to verify the dimonstration immittance of the applicant according to point 21.8.208 (a):
	(i) Smaller
	and the second se
21.A.Z	57 Investigations
Une de investig coropita	sign organisation stall make immediations. that allow the Apeney to tracke an atlans, including investigations of partners and subcantractors, necessary to determin nor and communical compliance with the applicable requirements of this Subpart.
100	
00.00	for the Eliza
21.A.2	58 Findinys
05	and an
	a second
-	
ini ti	le rejention of a demossivation of complumice mode by file Agency or assordance wi LS 200 (c) when maint in a DOA finding massified according to point (a) and (b)
21.A.2	63. Privileges
1	D. (Change)
	and the second se
10.00	(in (i)) (in any)
-	state in the second sec
30) When, pursuant la 20.8.268(b), the Agency has concluded that it is not involve the daugh organisation approval haddan shall be ontitled to move the design androve).
3.3. P	roposed changes to section B implementing the LOT concept
21.8.1	Level of performance of the design organisation
When e	Cling in accordance with Subdevia B, D. E. M and D, the Agency show establish the low misings of the design organization
TEANICS Prostation	Residual Elevante Al Islam Samuri, Ala Islam Institution. Residual Elevante International Islam Communication Institution Research International Advances International



_	(MDALOS Cancept Fac
21.1	206 Level of involvement
11978	n acting of approximity with Subparts B, D, E, M, and O
100	The Againcy anali monty the applicant onto it has accepted the contribution projection provided accepting to 31.4.2016).
ini	For each item in the certification programme established under point 21.4.20 (b), it ligency shall determine its level of invaluement in the verification of the demanstration compleance and in determining that no feature of characteristic makes the modul unside for the uses for which certification /= requested, based on a combination of
	(1) the novelry and critically of the certification liter, including means or compliant and
	(2) the level of performance of the design organisation, as established und 21.6.34, complemented is necessary by nervices resulting from the organisation listed under 21.8,208 (c).
ħđ.	The following incumstantives may be used by the Agency to complement or revise level of involvement
	 accident, impdoint or unsafe condition related to the concerned item of the certification programme, or its technology, on any product, or
	(2) change in the pattormence of the dusge organisation, or
	(2) demonstration of compliance rejectual by the Agency whilst werdying h distantial affinite of compliance previded by the Applicant, for any argum continuation project with this applicants or
	(4) mutuan of the carbinesticn programminestablectual under point Q1-ALIO (b), or
	(5) technical or operational evolution in the field of the concurred item of h certification programme, is lectinology, or in the means of compliance identified the pertification programme, or
	(6) information provided by the applicant under 21 ± 20 (g).
(0)	According to point (b), the Agency shall verify the demonstrations of complian scientified by the spaniant and astermine that no require or characteristic makes to spolust unsafe for the uses for which settification is requireded. For this purpose, if m require further classifications from the approach.
092	When the Agency is not subsfield with the demonstration of compliance of the applian annial attempts to ocsure the situation have failed, it shall notify the rejection of th demonstration of compliance in the applicant, and the reasons thereof
3,4	Proposed changes to section A extending the LOI concept to Subparts L E, M, and O
2.4	1 Extension to major changes and STC
21.1	1.97 Major changes
1	(m. 1961))
12.80	(0.0000-001) European All societ Games Agenci, As hans tweetige.

senter processing and	(MOALOGO Lanoph Park
Al a sec	
111	
[3] comply-with points 21.A.20	(0), (3), (0), (1) and (y), and
()) in manufacture	
())	
0 0 1 N	
Note:	
3.4.2 Extension in repair design a	ppravals
11.4.433 Repair design	
10 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	
1111.10.00.000	
(2) comply will points 21.A.20	(b), P.3, (0), III and Ig1, and
11	
and the second s	
14.3 Extension to ETSO	
21.4.605 Data requirement	
and the second s	
0 0 0	
a company	
 for ell atter articles, the proc curtomation programme referial 	etures referred to an minist 21.43602m(b)(2) and (b m in 21.4.20(b)
21.A.615 Verification by the Agene	EA
The applicant for an ETSU approval shi	an exempty with 22 A.20(Tr and (g)-
3.5. Proposed changes to sectio	on A resulting from the new privilege
Changes are introduced to 21 4,103, Indias when the Agency has concluded	PLA 115 and 21 A.437 to Introduce approval by ISO (Nat It is not Involven).
15.3 mill 389(5-m) V. Sumocar Milaton Salam April Trivantery example. Color and recommended. Com	n da natus hastanda. In statular at at an innustri das 5555 Trianya Adristanti da su at an innustri da



RM1.0262 & 4041.0011 and KM1.0550 & KMT.00 (MDM.05 Concept Fac	muobeen yenemos zelety yilensä
	5.1 Extension to major changes
	A.103 Tasue of approval
	a set parents
	the second second
major chieron lo the topo decorr may be approved. • regularisticker of the applicant under a procedu logen i her concrutes that it is nut involved accord block (a) and (b) of the paragraph are red.	 By dentrystion to this paragraph, the appropriately, approval data appage with the spearce, when the lar21.8(206(b) and given that con- ter21.8(206(b) and given that con-
	3.2 Extension to STC
on certificate	LA.115 Tasue of a supplemental t
	O - CAR COMPANY
	Colline College
e supplemental lype certificate may be insued by h investion of the applicant upder a considure autom is the concluded that it is not involved eccanting its (a) to (d) of this periodical) are mail.	 B) designation to this propagaph, spacentater approximat design or with the signace, when the Ager 21.8.308(b) and given that condi-
rovals	5.2 Extension to repair design ap
proval	1.4.437 Issue of a repair design a
noropoates approved rieson organismus order ar 12 - 12 - 1 - 0 - 1 - 0	i for micol repairs only, by an procedure approval by the Agence
by the appropriatery approved steagn crustilisation agraed with the Agency, when the Agency A anding to 33.8.208(6)	 Fac major rupor design addresse the Applicant under a provide conclused that is is cost invested a
a tigeta taunnya. Inana tautu taunga tau (XIX) jeonyal/2019991	Find State State Street Printer and Article

Confidence (Fritmane Security Talment)	INVERTIGATION AND THE LONG PARTICIPAL PRATICITY IN THE CALLS INCOME	e,cesa))
	ATTACHMENT B	
COMPARISON OF ANNEX 19 SM	IS FRAMEWORK WITH THE EASA SMS FRAMEWOR	Ŕ.



wropera Aviation Safety Agency	RMT.0262 & BHT.0611 and RHT.0550 & RHT.0613 (MDH.060) Chronic Pasar
	 Conversional and a provide state of the second state
	a Hallymont of Hamps
	+ Company accounted
	Persenne) - Salety prevaition Salety variant
	 Staty community
ICAD Annex 19 SMS Standards	EASA Management System Implementing Roles
Safety Policy and Olijectives	ORS.GEN.200 Hanagement System
1 Management carmitment and staponishills	(incomposition until manager, incomposite and reaction is reacting with a state of the case.
terper could and participal control to market. The soluty training from the control of the solution of the sol	111 datasy primer insult spatializity and according to transprint the application decading a direct which experiments of the constraint remains.
 Indiana is one download state the process in it the Accusary released at the Weilington state in it fid even y policy. 	[77] a description of the juncted process and process of the argument of an integral to staticly, otherwall for an two orbits.
 Control and so that the control of the second data material to an other second data material to an other second data and the second d	(1) (1) (1) (1) (1) (1) (1) (1) (1) (1)
supported to the state decourses permitted and state,	the second of second se
 Description of the analytic executive of the examination. 	the para Oni res. and with their effectiveness

Coold, year	and these	allenn inninial anna b allenich	CHA.
 (4) Transmission process framed line: completer in second View Labor, 	- 90	In comparisonal, with order and the second second to a second s	0
(5) data when a plant administration on process, including a process for maline because when of the	191	The periodic state comparison for the set of the set o	9
reservations, and the providers for amandery pro-		Gately accountablement	a.
The market of the		second provide the later	in -
(c) a function to months controllerup of the ingeneration with the mediant legislation of functions to the exception of use a function of another of the exception manages to ensure effective explorementation of mone pre- actions are receivery and	10)	where $r_{\rm g}$ the μ -contraction data and the state μ -contraction data in the first data in the properties of the energy radius $r_{\rm g}$ and $r_{\rm g}$ and $r_{\rm g}$ are presented at the energy radius $r_{\rm g}$ and $r_{\rm g}$ are presented at the energy radius $r_{\rm g}$ and $r_{\rm g}$ and $r_{\rm g}$ are presented at the energy radius $r_{\rm g}$ and $r_{\rm g}$ and $r_{\rm g}$ are presented at the energy radius $r_{\rm g}$ and $r_{\rm g}$ and $r_{\rm g}$ are the energy radius $r_{\rm g}$ and $r_{\rm g}$ and $r_{\rm g}$ are presented at the energy radius $r_{\rm g}$ and $r_{\rm g}$ are presented at the energy radius $r_{\rm g}$ and $r_{\rm g}$ and $r_{\rm g}$ are presented at the energy radius $r_{\rm g}$ and $r_{\rm g}$ and $r_{\rm g}$ are presented at the energy radius $r_{\rm g}$ and $r_{\rm g}$ and $r_{\rm g}$ are presented at the energy radius $r_{\rm g}$ and $r_{\rm g}$ are presented at the energy radius $r_{\rm g}$ and $r_{\rm g}$ are presented at the energy radius $r_{\rm g}$ and $r_{\rm g}$ are presented at the energy radius a	
(1) any applying approximate that are presented in the relevant Support of the areas or draw applying	(77	channy chines insure in senity assessmentiating for unknown thingsoft are providently including a chine proceedability for unknown fragment of server minimum (AF).	8
This framework water and accepted to the task to fra- organistic and to many imposing of the system, taking	-	Harrish Sana and Andreas and all transformed di compartenet, transportenet di aller dana di un part an el anteriorena anti- resport di Oris antere part esterna el dira 2015.	1
which because their bounds and reactived road indefend in the	aller She	accument and on the set of yough the difference in to both the and a 35 million throughout the promitizer in moti-	1
GEN.255 Contracted activities	LIGEN	indicative basis of surragement with taking the to main decrease many high stars, on the surrage,	6
access of logarity of fault and performed by teacher community orders must see the date of the servery cut safety and set of the set	100	Appointment of key satury personnel	.a.
spring some the unput domains approved. This is good in the second secon		which provide the product is leaving the only only and a region to the exploration of an effective CMS	tine t
adally, (in improved a particular areas particular to furne-	lo de	Control subtract guestsy newspaces platenting	4
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	With March	any a provinci sharkar un that any ana anny ying ana shar niy yan di Mada atin'na anna yang yang yang yang ang tina tinan nganana hinggi bing ana atin yang yang bing tina tinan nganana hinggi bing ang ating tina tina tinang tina tinang tinang tinang tinang tinang tinang tinang tinang tina	he s
man of a dr.	Millionel	and the second state of th	-

fundingen fragmen farmer Ditteriet.		Congot Ager
nd my me. LS SMS documentation		with under the account of the preparation. The processing approaches and more than the annual authors, a good source to the annual approaches in community accounting
LTT These are provide into devolution SHE important of them.	Country of high put total programmer.	
 (They included by Device The Database The Database (the organization of the Constraints) of the Database (the States) of a States of the Operation (c) and the observation. (The Section provides and Devices are provided for the encoded black they then then the 	(x)	GDI-230 Personant requirements The opposition of opport in synamic methods, we can the ochoicy for end or gitter of payment for the Therpest and cannot use a second second for any store to measure on the methods and the second second second second second second second second second second second second second second second second second
and a service and a service of the s		comparison of supercipal interpretation of Amount
Idea incomments, Destinations and investments account and investment and index rule for SPES presence on account and in	107	is present a drive of present real by controlling the tra- inguishing on the independent of controls the real to a control of the real of the real to the opportunity of the real of the real to the opportunity of the real of the real of the opportunity of the real of the real of the the real to the real of the real of the real of the the real to the real of the real of the real of the the real to the real of the real of the real of the real of the the real to the real of the real of the real of the real of the the real to the real of the real of the real of the real of the the real of the real of the real of the real of the real of the the real of the real of the real of the real of the real of the the real of the br>the real of the br>the real of the br>the real of the br>the real of the br>the real of the real of
 SKS repres The second participant contraction and second participants and second participants. 	(re)	The incorporation share have a diment specified percentage on the content party and adhybrid to be particular to the system with the approximate the system with the approximate the system of the sys
t, Sallety Hills managament .1 House's alternity way	101	The supervise is an explore appropriate sources, sources and the set of the
1.2 The same a provider start asymptotic memory is an one that makes that have been provided with the available methods in services the periods."	(v) (011	Training evolution of antiki encours from oil, percentions and second of the roles of a processory constant by the second of their relation of ENLISE Facility requirements.
1.3 Paraget plantification point for same, by a constitution of resonance membra and practition methods of safety table coherence.	True Joint	ergenerator intel hand facilitati aliantes that performance bet reprinted of an present name and services of anticellance with the OBIs the anticellance
A Rule assessment and Arthqueser		
The sector is presented at the common sector sector at a presence book on the	(III)	The source from the section is going of possible and the
A designed a set of the second	www.utv	ment man stat at at

East	DM.060) NDF Planes
a, and a second of you takey not a maximal or to all the product at a strong and , which is meaning of an all harms. Second of you takey not a mean and on the second of the low strong of of the low stro	ant series
ety assurtance OV The herrys of the reports that he possible multiple point in the reports that he possible multiple in the reports that her possible multiple in the report of the reports that her possible multiple in the report of the reports that her possible multiple in the report of the reports that her possible multiple in the report of the reports that her possible multiple in the report of the reports that her possible multiple in the report of the reports that her possible multiple in the report of the reports that her possible multiple in the report of the reports that her possible multiple in the report of the reports that her possible multiple in the report of	(1911)
The served provide and develop and memory of the served of	1000,0000
If a second process is alway port-money and be period in the CD Secondary contension is believed and select participants means of the SMC	-
te neregenert of derive	
ns de president a fuit conservation de presentant à proposal la (domain au contra en palle en par la colleption et de conservation de dense la president de conservation de la conservation aviante en terrete de desençais.	
Internal Personals, of the SPE	
ne is province presidence of and presso the effective sport in 2MS and presside province representation of the pressed performance Mag	
disty Promotion	
are not and a data service of the se	
The same of a constant second constant is adding the state of the constant is a state of the constant	
The entry of the sales investig and entry the beat sources a	

RMT.0267 & RMT.0613 and RMT.0550 & RMT.0613 (MDM.060) fimpers Aviation Safety Agency Canonal Parent Berg and a stranger of the SMS 1.2 Salary polymorication The service is provide all the country and traction all the insure it is a service and the country of the service of the servi with the same property of the state of the SMS to grad an experimental sector the spectrum $_{\rm c}$ of towner have setting the second to restart any late plan a transmission 32' sugarus yey aday providence or orreduced constraint. An of some second states in the second s -----The state of the state of the Invite Company Ave

LLARCMININg Forument - Notion Justibusion -Operturnt Working Stourt Report

Appendix F

Responses to Oversight / Assessment Relevant Questions

Paux | 10.8

Oversight / Assessment Relevant Questions:

- What FAA organizational changes have to be implemented to support robust oversight? Will the FAA need to
 have Centers of Excellence? Yes. Addressed in report.
- 2. What is the appropriate intent level of the oversight function? The intent of oversight is to validate the development of the defined system and verify compliance to the requirements. The level of oversight is determined by the risk level of the organization and activity (see risk section).
- How will "one" oversight process be designed for AVS? Single surveillance model between engineering and production? Should the oversight be implemented for AIR initially with AFS to follow? Single centralized FAA. Oversight organization.
- 4. Is it necessary to differentiate the oversight approach based on organization size, certificates, maturity, etc? If so what are the oversight criteria? The oversight approach is the same, however, the privileges / functions will vary, as will the depth and interval of surveillance based on organization size, maturity, certificates, etc. (see surveillance section).
- How will oversight be aligned to appropriately match the privilege and responsibility level of the organization? The level of oversight is determined by the risk level of the organization and activity (see risk section).
- How is systems approach organizational oversight differentiated from level of involvement (LOI) at a project level? Organizational surveillance will be conducted by the Oversight organization, while LOPI will be conducted at local ACO level.
- 7. If ongoing oversight is based on performance and not compliance how can it be normalized by company and scalable? Oversight will be based on privileges / functions (defined by the Organization WG), and scale will be determined by the risk level of the organization and activity.
- How important is it that the US model is harmonized, or at least recognized as fully equivalent, with other authorities' DOA/DO systems – today and in the future as they evolve? It is important and is addressed in report.
- How does oversight change from overseeing a delegated organization (ODA holder) to an applicant showing organization? DO oversight is not over a delegated entity. See Section 5.
- 10. Should there be routine FAA oversight and audit of the design organization's processes or should this be accomplished via self-audit and reported to the FAA? Should it be a combination of both? It would be a combination of both. A DO's routine internal surveillance results will be fed into the risk model to determine interval and scope of FAA oversight. The FAA oversight will sample DO processes and products to validate strengths and potential gaps or 'blind spots' in the organization.
- 11. What safety risk-based structure will be imposed for assessed deviations to procedure? Assessed deviations to procedure will follow corrective action procedures defined in FAA Order 2150.3. The risk based evaluation of these will follow the principles established in the SMS
- How does AS9100, or equivalent, outside auditing equate to the FAA oversight system? (Compare actual FAA Approved Production QMS audit versus AS9100 findings.) Not addressed in report.
- 13. Should internal FAA QMS audits be used as a model? If not, what best practices can be benchmarked to develop framework? (Consider other Industries.) Not addressed in report.
- 14. Should a system be generated to allow FAA resources to perform maintenance of organizations instead of a formal audit process (i.e., review of internal audits with a focus on verification and validation)? (Audit the management system elements not the findings.) Addressed with performance based oversight approach

Implementation Relevant Questions:

 What metrics / measures need to be established to ensure a successful implementation of organizational showing oversight? Recommending a separate activity to develop performance measures (Included In section 6).

- What is the impact on oversight model / approach if any part of today's delegation system (individual or
 organization) authorization is retained? No impact to oversight approach. Retained items (e.g. LOPI) are
 handled separately and independently from oversight. If areas of a delegation are still necessary, the single
 oversight model must be addressed.
- 3. Can FAA accept Third-Party certification in lieu of FAA audit? Yes in some cases. Addressed in sections 4 and 6.
- What critical criteria will be used to identify frequency and scope of oversight activity? Addressed in sections 4 and 5.
- 5. What are the other "triggers" that would cause additional FAA oversight? Addressed in section 5.

APPENDIX I—TSO SUBTEAM REPORT

21ARC Working Document - Not for Distribution

GAMA TSO Sub-team Report to the Part 21/SMS Aviation Rulemaking Committee

TSO Design Approval Clarification and Program Enhancement

January 31, 2014

I. INTRODUCTION

This report summarizes the activities and recommendations developed by the General Aviation Manufacturers Association (GAMA) Technical Standard Order (TSO) Subteam chartered to support the deliberations of the FAA Part 21/SMS Aviation Rulemaking Committee (ARC) and its Design Organization (DO) Working Group (WG)

The report is intended to clarify the FAA TSO Design Approval processes, identify practical TSO implementation issues from both the FAA and industry perspectives, and recommend specific actions to resolve them. It is not intended to address the question of DO applicability to TSOA applicants or holders, which remains the responsibility of the Design Organization WG. Similarly, questions related to Safety Management Systems and Oversight requirements have been assigned to other ARC Working Groups and are not considered here.

The objective of this report is to present the background material and specific rulemaking, proposals to be considered by the ARC in making its recommendations to the EAA regarding the clarification and enhancement of the current TSO program, as well as the establishment of an effective, efficient and globally recognized TSO program within the haure DO accountability framework envisaged and to be proposed by the ARC

Faken in its entirety, this report constitutes the TSO Sub-team recommendations detailing the gaiding principles and attributes necessary to prepare regulatory language for the drafting of any NPRM, Policy and/or Guidance materials. While not a complete listing, recommended rule and preamble language is provided in the respective sections.

Now

This report unitzes the content structure and formating created by the Part 21 SMS ARC Using Cychiazation Working Group in the Design Organization Concept Report, for consistency, case of reference and incorporation in the final ARC report if as applicable.

-

GAMA TSU Sub-team Part 21/SMS Availor Rolemaking (prominec

II. BACKGROUND

II. A. FAA Part 21/SMS ARC and Design Organization Working Group Charter

On October 5, 2012, the UAA Administrator chartered an Aviation Rulemaking Committee for Part 21/Safety Management Systems (SMS). The Part 21/SMS ARC was tasked to recommend improvements to the effectiveness and efficiency of existing tertification procedures for products and parts, along with incorporating SMS in the design and manufacturing environments. The ARC was tasked with making recommendations, including proposals for rulemaking, suggested processes, policies, and guidance as well as other actions the agency should take in support of its goal.

In addressing its charter, the ARC further chartered the Design Organization Working Group on March 17, 2013 to assist it in developing details associated with FAA recognition of Design Organizations in 14 CFR Part 21. The Working Group was tasked to define and address the details necessary for the FAA to recognize a Design Organization that needs organizational and system requirements inflicient to ensure that a Design Organization is capable of making compliance determinations upon which the FAA may rely.

The Design Organization WG adopted, as part of its charter, the task to recommond the types of organizations and product/part thresholds above which the application of Design Organization, and SMS, requirements should be considered mandatory.

II. B. TSO Sub-team Charter

The FAA Part 21/SMS ARC charter requests "consideration of proposed revisions to vlarify and update design-oriented regulatory requirements in Part 21", including "clarifying TSO design approval processes"

In early planning discussions, the ARC directed that Design Organization applicability to TSO parts suppliers should be considered by the ARC's Design Organization WG. Other, practical TSO implementation issues were to be drafted as problem statements for consideration by a separate TSO Sub-team, within the ARC's overarching objectives for continuing improvement in safety, incorporating SMS/Continued Operations Safety (COS) processes and increased efficiency in the certification processes, as well as transitioning to a systems oversight framework that does not require FAA delegation

GAMA's Technical Policy Committee approved the formation of the TSO Sub-team, which met initially on Judy 25, 2013, to consider the TSO problem statements brought

> GAMA TSO Sub-team Part 21/SMS Avoition Rolennaking (committee

forward by the FAA and industry and to recommend solutions in the form of Rulemaking, Policy and/or Guidance that, to the extent possible, support harmonization with international airworthiness authorities. The TSO Sub-team comprises FAA and industry representatives for multiple categories of TSO articles, including Avionics, Aircraft Instruments, APUs, Oxygen Systems, Seats, Cabin Safety and Cargo Handling Systems.

A copy of the GAMA Project Charter and a list of TSO Sub-team members are contained in Appendices A and B to this report, respectively.

While this document is the TSO Sub-team's final report to the ARC, the Sub-team has also identified topics for further discussion beyond the scope of the ARC. As such, the group will continue to operate as a GAMA Sub-team to serve as an industry forum for recommending and supporting ongoing improvements to the FAA TSO program, via stand-alone policy and guidance.

GAMA TSO Sub-team Part 21/SMS Aviation Rulemaking Committee

4

III. SUMMARY OF ACTIVITIES

III. A. Sub-team Work Plan

Following its initial meeting, in the course of three additional meetings— August 27, October 2-3 (Industry only) and November 13-14, together with an October 10 online meeting and several subsequent FAA coordination and editing meetings, the TSO Subteam reconfirmed both the FAA's and industry's views of the continuing value and relevance of a sub-aircraft level approval process that installation developers can take credit for and can be reased across multiple installations.

A brief summary of the Sub-team's discussion of the TSO Program value and challenges is contained in Appendix C to this report. It is noteworthy that there was very close alignment in the views of the FAA and industry members representing the multiple categories of TSO articles.

The Sub-team reviewed all open items in the FAA's Part 21 Clean-up Matrix carried forward from the Part 21 (Production) re-write, including clarification of TSO Holder Part 21 3 responsibilities, clarification of TSO Deviation and acceptable Equivalent Level Of Safety (FLOS) procedures, as well as clarification of TSO design changes part marking and integrated non-TSO functions. In addition, the problem statements identified by the Part 21/SMS ARC members and additional issues raised by industry throug the 2013 FAA Parts Approval Workshop were considered for potential resolution by rulemaking, policy and/or gaidance.

A list of the TSO issues and problem statements from the Part 21/SMS ARC members and from industry during the 2013 FAA Parts Approval Workshop is contained in Appendix D to this report

It was reported at the October 2-3 meeting that the Design Organization WG had determined TSO will fall below the threshold for mandatory application of Design Organization, and SMS, requirements. The Sub-team proceeded to develop its recommendations on this basis but without precluding the possibility of a TSO organization voluntarily adopting DO, and SMS, requirements in a scalable fashion, consistent with its technical capabilities and the scope of TSO privileges envisaged.

Assumptions:

The TSO Sub-team assumes the following regarding broader ARC recommendations and related FAA policy activities.

 The ARC will recommend that TSO organizations fall "below the threshold" of any mandatory requirement to become a Design Organization

> GAMA TSU Sub-team Part 21/SMS Availon Rolennking Committee

Q.

2. The ARC will recommend that an SMS will only be required for those organizations that fall "above the threshold" of any mandatory requirement to become a Design Organization.

3. The ARC will provide recommendations for DO organization requirements, whether applicability is mandatory or voluntary. The FAA is expected to adopt a subset of these DO organization requirements as TSO organization requirements under 14 CFR 21.605.

4. The ARC is expected to clarify that "substantiation data" does not need to be FAA approved. As such, the TSO Sub-team has tabled recommendations that would grant privileges to appropriately qualified TSO organizations for the approval of substantiation data (e.g., identification of "approved" DO-178 data, DO-160 data, flammability data, or other forms of data).

III. B. Product

With regard to the TSO Program value and challenges identified in Appendix C, the TSO Sub-team reviewed the list of issues and problem statements from the three sources (FAA, ARC Members and FAA-Industry Workshop), to distill the key issues that, if appropriately addressed, would lead to material improvements in the effectiveness and efficiency of the TSO program. These key issues are:

- o Acceptance/approval of integrated Non-TSO functions
- Management of post-TSOA design discrepancies
- o Definition of substantiation data and data submittal items
- o Minimizing re-review of TSOA substantiation data at installation approval
- Elimination of unnecessary TSO deviation requests

The TSO Sub-team continued to develop six Rulemaking recommendations and proposals for related Policy & Guidance, for consideration by the Part 21/SMS ARC in its final report to the FAA. While Recommendation 1 below is written with reference to a "Certified TSO Organization", these recommendations are generally intended to be implementable by either a Certified TSO Organization or an expanded FAA (ODA) delegation system incorporating TSO Design functions.

Each of the following recommendations is supported with a Discussion Paper/draft narrative for a Background/Preamble section describing the issue being addressed; and a Recommendation/Proposed Rule section outlining the proposed rule or rule revision, and any related policy & guidance. In some instances a Benefits/Metrics section is provided, summarizing the expected benefits to the FAA and industry, together with associated metrics.

GAMA TSO Sub-team Part 21/SMS Aviation Rulemaking Committee

6

Summary of Recommendations:

1) Allowance for TSO Organizations to issue their own TSO authorizations, relative to scalable privileges for particular types of TSO standards

Alternate approaches via a Certified TSO Organization or expansion of TSO ODA functions

2) Clarifying the types of data that can be approved under TSOA (Type design of the article, and declared performance of the article including Non-TSO functions & Incomplete TSO), and expectations for acceptance of approved TSO data for installation

Require and approve Declaration of Design and Performance (DDP) via revision to 21.601(b)(2) and proposed new 21.603(a)(3)

3) Requirement to declare Non-TSO functions

Proposed new 21.603(a)(3) and 21.619(d) for subsequent design changes. Additional guidance to include a "decision table" to assist in differentiating between TSO Supporting Features and Integrated Non-TSO Functions

4) Rule revision to remove the term "model number" from TSO rules and replace it with a requirement for a "unique identifier"

Revision to 21.603(b) and 21.619 for subsequent design changes

5) Change to Part 21 to establish the effective TSO revision level at the beginning of the project, not at the end

Revision to 21.603(a)

6) Process for TSO Holder to continue marking TSO articles following a determination of "a design discrepancy that does not result in an unsafe condition"

Revision of 45.10(b) and proposed new 21.616(i)

Summary of Parking Lot Topics: Proposals for Future Policy and/or Guidance

1) Maintain privilege for TSO Holders to make minor and insignificant (sub-minor) changes to articles without further approval

2) Clarify TSO Application Data, Manufacturer Data and Furnished Data requirements (Ref. Sub-team Discussion Paper – "TSO Documents")

7

GAMA TSO Sub-team Part 21/SMS Aviation Rulemaking Committee

3) Develop expanded guidance to promote the uniform definition and treatment of Integrated Non-TSO Functions by applicants, installation developers and the FAA (Ref. Sub-team Discussion Paper – "Non-TSO Functions")

> GAMA TSO Sub-team Part 21/SMS Aviation Rulemaking Committee

8

21ARC Wanting Document - Not to: Distribution

IV. RECOMMENDATIONS

IV. 1) Self-issuance of TSOA

Part 21 ARC GAMA TSO Subgroup

Subject: Self issuance of TSO Authorizations by Certified TSO Organizations

Tasking: Determine if the FAA's statutory authority permits the FAA to change Part 21, Subpart O to allow TSO organizations to issue their own TSO authorizations with no FAA involvement.

Background: The TSO system has been described as a "self-certification process but in reality, if currently requires FAA involvement to verify the manufacturer's compliance statement and issue an authorization to apply TSO marking for each article. The FAA could better manage its resources and streamline the TSO process by allowing the traditional FAA review portion to be performed by Certified TSO Organizations or appropriately delegated organizations, up to and including issuance of the TSO authorization letter. The FAA already has a rule-basis to qualify TSO organizations under 21.605. But Part 21 Subpart O currently requires an application to be submitted to the ACO (under 21.603) and the FAA to issue the authorization (under 21.611).

Recommendation: Rule changes to § 21 603 and 21.611 should be considered to allow for TSO organizations to apply for the privilege of issuing their own TSO authorizations. The expectation is that this privilege would require specific capabilities that may not be required of all TSO organizations under 21 605. As such, 21 605 would also be amended to specify expectations for organizations seeking the privilege of self-issuance. The FAA via policy, would be expected to use these new regulatory capabilities to describe a scalable system of organizational privileges. For example, the FAA may specify organizational capabilities necessary for an organization to receive the privilege for each TSO standard, or for groups/classes of TSO standards. The FAA would specify oversight requirements to monitor use of this privilege, and to take action against improper use of the privilege.

Proposal: Based on the above the following changes are proposed for the committee's consideration to address self-issuance of TSO authorizations

§ 21.601 Applicability and definitions. (Moprand)

(a) This subpart prescribes-

GAMA TSO Sub-lean Part 21/SMS Avoition Ridemaking Committee

ā.

(b) For the purposes of this subpart-

 A TSO issued by the FAA is a minimum performance standard for specified articles used on civil aircraft;

(2) A TSO authorization is an FAA design and production approval issued to the manufacturer of an article that has been found to meet a specific TSO;
(3) A letter of TSO design approval is an FAA design approval for an article that has been found to meet a specific TSO in accordance with the procedures of §21.621.

(4) An article manufactured under a TSO authorization, an FAA letter of acceptance as described in §21.613(b), or an article manufactured under a letter of TSO design approval described in §21.621 is an approved article for the purpose of meeting the regulations of this chapter that require the article to be approved; and

(5) An article manufacturer is the person who controls the design and quality of the article produced (or to be produced, in the case of an application), including any related parts, processes, or services procured from an outside source.

(6) A Certified TSO Organization is an FAA operating certificate for the issuance of TSO authorizations.

§ 21.603 Application (Proposed)

(a) Other than as described under 21.603(d), an applicant for a TSO authorization must apply to the appropriate aircraft cartification office in the form and manner prescribed by the FAA. The applicant must include the following documents in the application:

(d) An applicant for a Certified TSO Organization certificate must apply by submitting an application to the appropriate arcrait certification office in the form and manner prescribed by the FAA.

§ 21,605 Organization. [Proposed]

(a) Each applicant for or holder of a TSO authorization must provide the FAA with a document describing how the applicant's organization will ensure compliance with the provisions of this subpart. At a minimum, the document must describe assigned responsibilities and delegated authority, and the functional relationship of those responsible for quality to management and other organizational components.

> GAMA TSU Sub-team Part 21/SMS Aviation Rulemaking (countried

- (b) Each applicant for or holder of a Certified TSO Organization certificate must provide and maintain documentation describing the applicant's organization and system for managing safety and compliance of TSO authorizations issued under its approval.
- § 21.611 Issuance. [Proposed]
- (a) If the FAA finds that the applicant for TSO authorization complies with the requirements of this subchapter, the FAA issues a TSO authorization to the applicant (including all TSO deviations granted to the applicant).
- (b) A Certified TSO Organization certificate holder may issue a TSO authorization when it determines that an article complies with the requirements of this subchaptor and may be approved in accordance with the Certified TSO Organization operating certificate issued under 21.6xx, (See Note 2 below).

NOTES:

- 1 This recommendation suggests the FAA will maintain 2 systems for TSOA, i.e. The Certified TSO Organization and the traditional, FAA managed TSOA project. Certified TSO Organizations will not be authorized to act as TSOA Agents for third party TSOA projects.
- TBD. Will need an explicit rule describing the FAA's additional responsibility for issuing Certified TSO Organization operating certificates and retained responsibility for the review and approval of TSO Deviation requests.
- 3 TBD Will need an explicit rule describing the FAA's authority to oversee Certified TSO Organizations and to revoke a Certified TSO Organization's Certificate.

Parking Lot Question: Would the FAA directly oversee Certified TSO Organizations or would it be possible for the FAA to assign oversight responsibilities to a third-party industry organization? The latter approach may diminish the reusability of the 'FAA TSOA' credentials, especially with regard to acceptance by foreign authorities.

u

GAMA TSO Sub-team Part 21/SMS Aviation Ralemaking Committee

IV. 2) Require and Approve Declaration of Design and Performance (DDP)

Part 21 ARC GAMA TSO Subgroup

Subject: Incorporate a Declaration of Design and Performance as a Required Element for FAA TSO Approval

Background: Under current rules, the holder of a TSO authorization is expected to maintain the performance of the TSO article relative to the TSO standard. However, in order to support all worthiness or contractual requirements. TSO manufacturers are typically required by their customers to maintain performance of the article relative to requirements that are not part of the TSO standard. The TSO system could be modified to allow manufacturers to better align their requirements under the TSO system with those requirements they are electing to meet as part of sirworthiness or contractual obligations. EASA's system supports the declaration of this type of manufacturer defined performance by requiring the submission of a declaration of design and performance, and expects TSO manufacturers to control the article's performance relative to that DDP.

Recommendation: The FAA should require submission of a DDP as part of a TSO application. TSO manufacturers should be required to maintain the performance of the article relative to the DDP. Except where required by other FAA policy (e.g. see Section IV.3 discussion of required declaration of non-TSO functions). The manufacturer should not be required to declare any performance that goes beyond the minimum requirements of the TSO. Instead, the manufacturer may elect to align their DDP with the TSO requirements, or they may elect to include additional requirements or performance levels that go beyond the minimum requirements or performance levels that go beyond the minimum requirements or performance levels that go beyond the minimum requirements of the TSO. This approach affords flexibility to the TSO manufacturer so they can align their regulatory responsibilities under the TSO system with any external contractual or ainvorthiness requirements that also apply to the TSO article. But no additional builden is added in requiring the TSO, unless the performance declared in the DDP exceeds the minimum TSO requirements.

Proposal: Based on the above, the following changes are proposed for the subcommittee's consideration to adoress the incorporation of a DDP into the TSO system.

§ 21.601 Applicability and definitions. ///reposed/

17.4344 TSO Substant Part 21/SMS Available Enternating Committee

11

(b) For the purposes of this subpart-

 A TSO issued by the FAA is a minimum performance standard for specified articles used on civil aircraft;

(2) A TSO authorization is an FAA design and production approval issued to the manufacturer of an article that has been found to meet applicable TSO standards. Any data specified in the TSO authorization's declaration of design performance is considered to be approved.

§ 21.603 Application. [Proposed]

(a) An applicant for a TSO authorization must apply to the appropriate aircraft certification office in the form and manner prescribed by the FAA. The applicant must include the following documents in the application.
(1) A statement of conformance certifying that the applicant has met the requirements of this subpart and that the article concerned meets the applicable TSO that is effective on the date of application for that article (2) One copy of the technical data required in the applicable TSO.
(3) One copy of a declaration of design and performance for the arucle and supplementary data. If needed to support any declaration of performance exceeding the minimum TSO requirements.

NOTES:

- Minimum required DDP content should be defined in a revision to AC21-46 (Reference EASA AMC 21.A 608 for minimum DDP content).
- 2 If this proposal to require and approve a DDP is not accepted, another solution will be needed to address the question of what TSO approval actually approves. This may require further definition that any data specified in the TSO authorization is approved data.

GAMA TSU Sub-team Pan 21/SMS Aviation Rulemaking Committee

IV. 3) Requirement to Declare Non-TSO Function

Part 21 ARC GAMA TSO Subgroup

Subject: Non-TSO functions included in a TSO article

Tasking: Determine if Part 21, Subpart O, changes are necessary to address added non-TSO functionality in a TSO article, or can it be handled by TSO policy only (i.e. Order 8150.1C).

Background: Non-TSO function(s) have become more and more prevalent in avionics systems mainly due to added processing capabilities in newer integrated circuits as well as larger and cheaper memory capabilities. However, mechanical systems TSOs, such as seats, can also contain provisions to host non-TSO functions (e.g., embedded passenger entertainment devices). These added non-TSO functions are mainly market or OEM driven to optimize installation capabilities while minimizing certification costs. Added non-TSO functions have historically been handled through FAA policy, since Part 21 Subpart O does not specifically address or codify their embedded existence. This has resulted in a problem with some ACOs who feel that current Part 21 Subpart O does not allow for a more detailed evaluation of the non-TSO function(s) at the time of TSO approval, even though in reality, the added non-TSO function(s) is/are inseparable from the hosting TSO article design at the time of manufacture.

Recommendation: A rule change to § 21.603 should be considered to address added non-TSO function(s) incorporated in a TSO article approval. Building under a TSO design and manufacturing approval is a privilege. Since the TSO label is affixed at the time of final inspection it represents the final approved configuration, which may include embedded non-TSO functionality. For this reason the FAA should have the authority to require the manufacturer to declare, at the time of TSO application, all non-TSO functionality contained in the final approved configuration that is intended to be acknowledged in the TSO authorization. It should be noted that the non-TSO function(s), once acknowledged by the FAA, are considered inseparable in the approved design. Design changes (minor/major) to the TSO function(s) still requires an analysis of its impact on the required performance of the hosting TSO.

Proposal: Based on the above, the following changes to § 21.603 are proposed for the committee's consideration to address added non-TSO functionality.

14

§ 21,603 Application (Proposed addition of new subparagraph (a)(3) and minor revision to subparagraph (b) in accordance with Recommendation 4)]

(a) An applicant for a TSO authorization must apply to the appropriate aircraft certification office in the form and manner prescribed by the FAA. The applicant must include the following documents in the application. (1) A statement of conformance certifying that the applicant has met the requirements of this subpart and that the article concerned meets the applicable TSO that is effective on the date of application for that article (2) One copy of the technical data required in the applicable TSO, and (3) The applicant must also submit one copy of the technical data for any additional declared function(s) not required for the performance of the TSO function. This data must include the manufacturer's declared performance and design assurance level of the added function(s). (b) If the applicant anticipates a senes of minor changes in accordance with Sec. 21,619, the applicant may set forth in its application the basic unique identifier of the article and the part number of the components with open brackets after it to denote that suffix charge letters or numbers (or combinations of them) will be added from time to time (c) If the application is deficient, the applicant must, when requested by the

FAA, provide any additional information necessary to show compliance with this part. If the applicant fails to provide the additional information within 30 days after the FAA's request, the FAA denies the application and notifies the applicant.

§ 21.619 Design changes (Processed addition of new subparagraph (d))

(a) Minor changes by the manufacturer holding a TSO authorization. The manufacturer of an article under an authorization issued under this part may make minor design changes (any change other than a major change) without further approval by the FAA. In this case, the changed article keeps the original model number (part numbers may be used to identify minor changes) and the manufacturer must forward to the appropriate aircraft certification office, any revised data that are necessary for compliance with Sec. 21 603(b)

(b) Major changes by the manufacturer holding a TSO authorization. Any design change by the manufacturer extensive enough to require a substantially complete investigation to determine compliance with a TSO is a major change. Before making a major change, the manufacturer must assign a new type or model designation to the article and apply for an authorization under Sec. 21,603.

(c) Changes by persons other than the manufacturer. No design change by any person (other than the manufacturer who provided the statement of conformance for the article) is eligible for approval under this part unless the

15

person seeking the approval is a manufacturer and applies under Sec. 21 603(a) for a separate TSO authorization. Persons other than a manufacturer may obtain approval for design changes under part 43 or under the applicable airworthiness regulations of this chapter.

(d) If additional functions are declared in accordance with 21.803(a)(3), then the manufacturer must ensure that any design changes to those functions do not.

t) impact the performance of approved TSO function(s); or

2) Clumps the declared performance of the additional function(s).

Benefit: For the TSO applicant – an initial review of the manufacturer-declared performance can be made, and credit for software, hardware, and environmental testing, in support of installation, can be acknowledged in the TSOA letter. For the FAA – more efficient use of engineering resources since the review of the added function(s) can be more thoroughly accomplished with the TSO manufacturer as opposed to attempting to accomplish this review during each subsequent installation approval. It should be noted that both the TSO article as well as the integrated non-TSO function(s) must be appropriate to support the intended installation.

NOTES:

- The development of additional policy and/or guidance is essential to assist in determining a Supporting TSO Feature vs. an Integrated Non-TSO Function and to promote standardized and unambiguous interpretation Table 1 below may be used as basis for developing additional common policy and/or guidance in Order 8110.4, Order 8150.1. AC-21-46 and/or AC 21-50.
- 2. The minimum DDP content definition noted in Recommendation 2) above should include. For any additional declared function(s) not required for the performance of the TSO function, appropriate reference to i). An analysis showing that the added function(s) will not interfere with required TSO performance in the article, and ii) Any limitations or continued maintenance and preventive maintenance actions necessary to ensure that the added function(s) continues to meet the manufacturer's declared performance
- 3 Proposed new subparagraph § 21 603(a)(3) would become new subparagraph § 21 603(a)(4) following adoption of Recommendation 2) above which also adds a new subparagraph § 21 603(a)(3).

GAMA TSU Sub-team Part 21/SMS Avoiton Rolennation (committee
Table 1.

Determination of a Supporting TSO Feature vs. an Integrated Non-TSO Function

Identifying any integrated Non-TSO Function within a TSO article is necessary to assure the function is properly evaluated as part of the TSOA and for installation approval. The following table will help you distinguish between a supporting TSO feature and an integrated non-TSO function hosted within the TSO article¹. Keep in mind that whether a "supporting TSO feature" or an "integrated non-TSO function", both require an installation approval to evaluate compatibility and intended function.

Jop	den.	Constitutions	91=-	d==_
	It directly support the unability, performance, or design assurance of the honting 1507	In many cases the TSO manufacturer may add features to improve the dislify of the article, to differentiate their product from a competitor s, or to support a design assurance level. Examples of such teatures include the Rep-Day of the active and standby frequency in a VHP radie, providing topplemental artport information in an FMS database; or adding an electrically operated fectime to a paroinger snat. If you are traving difficulty making this optermination, then ask the reverse question: "Would it have any aseful purpose if the hinting TSO did not exist?" In the example of a dualt in-test-equipment (B/TE) check in a transponder, the onsider would in no, since the B/TE check has no purpose beyond supporting the horting TSO and it should be considered a feature	It should be treated as a "supporting heature of the hasting TSO	Go tá Štep "t
ž	Rmother TSO cover this fonction?	If the function is covered by another 750, the applicant should comply with that 150% performance requirements as, by themision, it can no longer be considered a "Non-150 Function". If should be noted that a TSO sufficient to any the only means to achieve an article approval—an explicant can also use the 1C or STC orocess and then, using the approved talax- sets menufacture authority for the article under a 14 CTR 121.303 PMA authorization. Therefore, the applicant may choose to entition comply with the applicable 150 for inter one-time, or approve the entire article under the TC/STC process as appropriate	Camply with the performance requirements of the applicable TSO, or use the TSO, or use the TSO/STC species to achieve article approval.	TSO uple and should declare the as an "Integrated Non-TSO function"

GAMA TSU Sub-team Part 21/SMS Aviolitici Rolemaking (promitiec

T:

 $^{
m 1}$ Whether a supporting TSO feature or an integrated Non-TSO function, the manufacturer must

demonstrate neither will adversely impact the required performance of the TSO(s) included in the article. ² Depending on the complexity of the integrated Non-TSO function, a concurrent TC or STC evaluation may

be necessary to properly evaluate the integrated Non-TSO functionality.

18

IV. 4) Replace Model Number with Unique Identifier

Part 21 ARC GAMA TSO Subgroup

Subject: Replace "Model Number" requirements with "Unique Identifier" requirement

Background: Current regulations require the use of model numbers to maintain configuration control of TSO authorizations. But from a practical perspective, model numbers are frequently used as marketing identifiers; they do not consistently provide a meaningful reference for configuration control of TSO article designs.

Recommendation: The Group recommends eliminating the requirement to use a model number, but should not preclude the use of a model number. Group recommends making the rule require a "unique identifier". Policy and guidance can specify that base part number is the recommended means of compliance. Policy or guidance is needed to clarify that "sub-approval level" changes do not need to be documented as a "minor change" under the rule. The current rule language structure is appropriate (keep the current definition for major changes but better define what constitutes a minor change... require new application for major changes and allow flexibility for minor changes).

The intent of the rule is to maintain configuration control of the approved descriptive data and approved performance of the article. Any change to the approved descriptive data for the article or approved performance of the article must be documented as either a major or minor change. Changes that do not affect approved descriptive data (e.g., sub-assembly, MRB, lower-level design drawing changes) are addressed under production configuration control requirements in 21.137, are not considered a change to the TSOA, and do not require a change to the unique identifier for the article (e.g., re-substantiation to verify that sub-minor changes yield the same approved performance level) are not considered a change to the approved performance level of all articles covered by the TSOA (e.g., new substantiation to qualify existing and future articles to a higher performance level) are considered to be a change to the TSOA but do not require a change to the unique identifier for the articles to a higher performance level) are considered to be a change to the TSOA but do not require a change to the unique identifier for the articles to a higher performance level) are considered to be a change to the TSOA but do not require a change to the unique identifier for the articles.

Proposal: Based on the above, the following changes are proposed for the committee's consideration to address replacement of model number configuration control requirements with unique identifier requirements:

GAMA TSO Sub-team Part 21/SMS Aviation Rulemaking Committee

19

§ 21.603 Application [Property]

(b) If the applicant anticipates a series of minor changes in accordance with §21 619, the applicant may set forth in its application the basic unique identifier of the article and the part number of the components with open brackets after it to denote that suffix change letters or numbers (or combinations of them) will be added from time to time

§ 21.619 Design changes (Proposed)

(a) Minor changes by the manufacturer holding a TSO authonization. The manufacturer of an article under an authonization issued under this part may make minor design changes (any change other than a major change) without further approval by the FAA. In this case, the changed article keeps the original unique identifier (part numbers may be used to identify minor changes) and the manufacturer must farward to the appropriate aircraft carrier provide the second data that are assessed for compliance with \$21,03(b)

(b) Major changes by the manufacturer holding a TSO authorization. Any design change by the manufacturer extensive enough to require a substantially complete investigation to determine compliance with a TSO is a major change. Before making a major change, the manufacturer must assign a new *unique identifier* to the article and apply for an authorization under §21,603.

> GAMA TSU Sub-team Pan 21/SMS Availon Rolemaking (promitice

IV. 5) Establishing the Effective TSO Revision Level

Part 21 ARC GAMA TSO Subgroup

Subject: Change to TSO Application Process in Part 21 to establish the effective TSO revision level at the beginning of the project, not at the end.

Tasking: Determine language changes to Part 21, Subpart O, to establish the Effective TSO Revision Level at the beginning of a TSO development project.

Background: Current regulation (14 CFR 21.603(a)(1)) implies that application for TSO is made after all design and development is completed. Designing and developing a TSO article is a lengthy process especially for complex articles. During this time, TSOs could be revised or new TSOs could be introduced. Current policy, as defined in FAA Order 8150.1C, section 6-1.b., allows an applicant six (6) months from release of a newer revision to apply with the previous revision. Although six (6) months seems to be a reasonable timeframe, many complex system developments take significantly longer and the six (6) month grace period is not sufficient. The two (2) current options to address this situation are:

- 1. Request a petition for exemption (per 14 CFR 11) which an applicant is required to submit at least 120 days before the exemption is needed. After submission, approval can take many months due to the requirement of publication in the Federal Register and a public comment period.
- 2. Comply with the newer revision TSO or add a newly released TSO.

If a new revision of a TSO or newly introduced TSO provides no benefit or does not impact flight safety, this additional work to submit a petition for exemption or complying with the latest TSO during an in-process development project could be a large burden on an applicant, potentially driving re-redesign and/or re-test and preventing expeditious introduction of safety enhancing products to market.

Since TSOA is a self certification based upon a statement of conformance, the responsibility of reviewing and ensuring any new or revised TSO(s) does not affect the certification basis or design is the burden of the TSO applicant regardless of when application is made.

Recommendation: A rule change to 14 CFR 21.603(a) should be considered to address the allowance of declaring the effective TSO revision at the beginning of a design and development project, in the same manner of Type Certificate applicants (reference 14 CFR 21.17(c)). When a TSO project is started, the applicant submits a Project Specific Certification Plan (PSCP) or equivalent. The PSCP or equivalent would be used to define the effective TSO revision level and the agreed upon time period for which it will remain effective.

GAMA TSO Sub-team Part 21/SMS Aviation Rulemaking Committee

§ 21.603 Application [Proposed]

(a) An applicant for a TSO authorization must apply to the appropriate aircraft certification office in the form and manner prescribed by the FAA. The application must declare the effective TSO revision (evel(a) that describe the certification basis and a unique identifier for the article. In addition, the applicant must propose and the FAA must accept the deadline by which the development project will be completed and the following documents submitted to the aircraft certification office.

Benefit: If the applicant is allowed to declare the effective TSO revision levels al the beginning of a project, FAA/Applicant communication on complex certification issues could be improved. Presently, when application is submitted, there could be several iterations due to certification basis disagreements causing potential applicant design rework and weeks to months of delay in issuance of a TSOA letter. Based upon the regulation change, adoption of the new or recently revised TSO(s) would be voluntary unless the change is due to a safety of flight issue or required for other reasons, such as interoperability, and deemed mandatory by the FAA.

The benefits of this change can be tracked through improved turnaround time of TSOA Letter issuance from the FAA, upon final submittal of a statement of conformance certifying that the applicant has met the requirements of the subpart and that the article meets the applicable TSO(s)

NOTES:

- Develop specific guidance for declaring the effective TSO revision level basis and agreed upon time period for the application for inclusion in Order 8150.1 and/or AC21-46. The Sub-team anticipates developing a more specific set of expected time frames for the development of different types of TSO articles for inclusion in future Policy/Guidance. Both of these initiatives would also benefit from review and coordination with foreign authorities.
- If adopted as drafted, this Recommendation does not conflict with other changes to §21 603(a) resulting from Recommendations 1, 2, 3 and 4

GAMA TSU Sub-team Pon 21/SMS Aviation Rulemaking (committee

IV. 6) Continued Marking of TSO Articles

Part 21 ARC GAMA TSO Subgroup

Subject: Marking of an article upon discovery of a post-TSO approval design discrepancy

Tasking: Determine if recent Part 21, Subpart O, changes, implemented with Amendment 21-92, dated 4/16/2011, now allows TSO holders the flexibility to continue to mark TSO articles that are found, "post-TSO approval" to contain a design deficiency that does not result in a unsafe condition.

Finding: The requirement that the article must meet all applicable TSO performance standards in order to be marked is still in effect, but was moved to § 45.10(b)

Background: Prior to the Part 21 and 45 changes implemented by Amendment 21-92, dated 4/16/2011 and Amendment 45-26, dated 4/16/2011, respectively, the requirement that an article meet applicable TSO performance standards in order to be so marked was defined in § 21 603 as shown below.

§ 21.603 TSO marking and privileges

(a) Except as provided in paragraph (b) of this section and Sec. 21 617(c). In parson may identify an arricle with a TSO marking unless that person holds a TSO authorization and the article month opplicable TSO participance standards.

With the Part 21 and 45 Amendment changes implemented in 4/16/2011, the previous § 21.603(a) requirements were recodified into § 45.10(b) as shown below, which still requires the article to meet applicable performance standards before marking can be applied:

§ 45.10 Markings

No person may mark a product or sincle in accordance with mis subpart unless-

(a) That person produced the product or article -

(1) Umlet part 21, nubpart F, G, K, or O of this chapter, or

(2) For export to the United States under the provisions of an agreement between the United States and another country or jurisdiction for the acceptance of products and anticles and

(b) that product or article confirms to its approved designs and s in a

GAMA TSO Sub-team Part 21/SMS Aviation Rulemaking Committee

condition for safe operation and, for a TSO article that TSO article meets the applicable performance standants.

Currently, when a TSO holder identifies a design deficiency, the holder must stop shipment (i.e., stop marking per § 45 10(b)) and report the deficiency to the ACO. If the ACO determines that the deficiency does not result in an unsafe condition (i.e., does not require a 14 CFR § 39.1 action), then to resume shipping of articles the TSO holder must either correct the deficiency immediately (which may not be practical), or, more typically, request a deviation under § 21.618. However, the intent of the § 21.618 provision was to provide a means for the TSO holder. Therefore, in an effort to apply a "managed risk" approach, it is recommended that a revision to § 45.10(b) and the addition of a new paragraph to § 21.616, "Responsibility of holder", be proposed by the Part 21 ARC to allow a TSO holder to resume shipping with the TSO marking when an ACO-accepted corrective action plan has been proposed and implemented.

Proposal: Based on the above, the following change to § 45.10(b) and the addition of a paragraph (i) to § 21.616 are proposed for the committee's consideration.

§ 45.10 Markings (Proposed deletion in paragraph (b))

No person may mark a product or anticle in accordance with this subpart unless--

(a) That person produced the product or article --

(1) Under part 21, subpart F, G, K, or O of this chapter, or

(2) For export to the United States under the provisions of an agreement between the United States and another country or jurisdiction for the acceptance of products and articles, and

(b) That product or article conforms to its approved design, and is in a condition for safe operation; and, for a 7SO article; that TSO article musts the applicable performance standards.

§ 21.616 Responsibility of holder (Proposed new paragraph (1))

(c) Ensure that each manufactured article conforms to its approved design ts in a condition for safe operation, and meets the applicable TSO;

(i) Notwithstanding paragraph (c) of this section and § 21.3, a holder of a TSO authorization may, upon discovery of a discrepancy in the approved design, continue to manufacture a TSO article if:

> GAMA TSO Sub-team Part 21/SMS Aviation Rulemaking Committee

14

(1) The discrepancy is reported to the FAA;
(2) The FAA determines that the discrepancy does not result in an unsafe condition; and
(3) The TSO holder implements a corrective action plan that is accepted by the FAA.

Benefit: Provides for a risk-based approach to handle TSO design deficiencies that do not rise to the level of an "unsafe" condition. May also be in the public's interest in the case where the stop shipment could result in a major economic burden to the end-user of the article.

75.

GAMA TSO Sub-team Part 21/SMS Aviation Rulemaking Committee

	Appendix A
(GAMA TSO Sub-team Charter
1) GAMA	Eveneral Availan Musuriverureen Association Motor Samerika Taren III Musingen DC Bace-Set and 351 1507 (g.w. (2021 MD-Rod
No. of Lot.	April 24, 2013
inject Chatter	Teconical Standard Order (150) Program improvement
ponson GAMA	Avionics & Electronic Systems Subcounitien (AVI) & FAA Part 21/SMS ARC
ackground: The FAA: Fr	ed group wit compare TSO Problem Balements betaget forward by the FAA and a durity, unit and work are in the form of Rulemaking Policy and/or Godbarce at 21, 2009 ARIs charter requests' consideration in propagation excerning to carry and catally average
NEW PRIMARY ISSUED	eute en Law 31, manage crewhell recharge statewai backers
i baity planning discussion uppliers alreads be raminde a Fraham Statements for the AFG's oversething object interaining processes, as w pulse. The provided group	In the ARC determined that Danger Organization, IS/MC, and Diverselm applications to ISD and by the ARC and propagation WS. Other, product TSC implementation assure alread the another due to make due to the due tothe due to the
Early planning decasions upplers should be conside a Problem Schemers for the Ari Sciencesching ober- mitisching processes, as w mitis Disproved group communications Aritimeter glanlog requirements, pot entremt mitis what retaining	In the ARC determined that Design Organization: (5)MC, and Diversell application, which is obtained by a MRC's Organization WS. Other, product 130 implementation actuates alread the ordered vibring moderation by a supplementary WS to be formed taset. Such consideration shall be written transferred tasets are advected and shall be written to see Such consideration shall be written to see the source to see a supervised action to any by the Part 2416M 5 ARC, and the FAA directly for any ing independing, user the rest 5-4 months. The project alread was defined and the swetch or density of a supplementary of the Part 2416M 5 ARC, and the FAA directly for any ing independing, user the rest 5-4 months. The project alread was defined was date away to be for a project alread or and any other average of the text to be average of the text of
Lany planning de casuent uppliers alroads be carrielle a Problem Statements for eAS & case tarbing other, intrication processes, de la auto: Disconsidigation (communications of requiri- gatistics y requirements, pol energia ratio allo an estering a 14 CFR Part 21 (Product apprivations) TRIO Depresent TRIO Depresent apprivations and responses.	The ARC determined that Darigh Organization (S/MC) and Diversity applications to 1600 and by 8 ARC's Organization WS. Other, product 130 implementation assues alread the ordered table to which there is nonlineing improvement on safety, reacryproxing S/MSIGON and invest readered determinations for an observation of the Part 21/SMC ARC, and the FAA directly for any no presentation for own by the Part 21/SMC ARC, and the FAA directly for any no presentation, will device incommendation for any by Report by Signature and the control of the read of the re
 Lany planning decasions uppliers simulation consider a Problem Schlaments for e Artics constanting others intercommunications of regular spatial processes, as a spatial processes, as a spatial processes of regular spatial processes of regular spatial processes of regular approval process. Develop unit anthers it with the FA3's Close publics. 	The ARC determined that Danger Organization (S/MC) and Diversity applications of the Table et by the ARC's Organization WIS. Other, product T3C implementation assues alreaded the another due to move address by a supplementary WIG to be twented table. Such considerables in that be writen tries for continuing to an ournal systemic means philling SMS/ICO) and must read efficiency in the of the testing once to an ournal systemic means philling SMS/ICO) and must read efficiency in the of the testing once to an ournal systemic means philling SMS/ICO) and the EAA directly for any organization, used the read SH months. The protect already will address not expand the award the organization, used the read SH months. The protect already will address we would be the following three obscurves. Include will permit a state of the address of the available already and already and any patients of the triangle the EAA's Part 21, Subpart 11 (21 0XX) Chemical Multim control forward limit on prevention should be addression in a conditione with the following the available address of the the following three obscurves. Include with permittion with EAA's Part 21, Subpart 11 (21 0XX) Chemical Multim control forward limit acceptable 12,011) processions and configuration of TSO Houses part that the TBC supports thes, donical address acceptable 12,011) processions and configuration of TSO theory the read of the read includes and part of parts and marking addression and the field and the address and write and parts and parts addression and configuration of TSO theory the address and write and acceptable 12,011) processions and configuration of TSO theory and address and write and write the marking addression and configuration of TSO theory and address and write and write the marking addression and configuration of TSO theory the address and write and write the marking addression and configuration of TSO theory to a state marking and writemark addression theory the addression and configuration of TSO theory the address o
 Lany planning discussion upplices straid be consider a Frabuen Salatements for a ARC's constanting object ministram processes, or we make. The proceed group score and anone not require splantary requirements, pot entrementations not require splantary requirements, pot entrement with which retaining in 14 OFR Part 21 (Fraduct approval process). TSO Depresent pro- TSO Depresent pro- taining the FA3's Close proteines. Develop the Constition; 	The AFC determined that Dange Organization (E)MC and Diversity applications of the Ep- ed by the AFC dispersional of the Other, product TSC implementation and a photometer that have determined by a suppresenting. We have functioned that Such conductation shall be written that have determined to an experiment operaginities awards that the investment english and the experiment of the terminance to an exercise system of the Part 2/16MC ARC, and the EAA directly for any ing terminance to an exercise system of the Part 2/16MC ARC. and the EAA directly for any ing terminance, when the result system of the Part 2/16MC ARC. and the EAA directly for any ing terminance, when the result shall months. The protect around will denotify and review exciting on guidance, industry process specific T ince an receivery, and any other shall be worther as well and to the following three observes. Incular will report the shall be Part 2/1, Subpart ID (21.6XX) Diversing Matrix cannot be worther and the following three observes. Incular will report the best set of each of the organization of TSO that error and the diversity for the three observes. These includes without terminents submitted by the F&A and industry, set of the TBO engineers are proved to the recommendations, and our tradement TSO that error of the organization and information acceptable ED.000 processores, and our tradement TSO that error of the organization and information acceptable ED.000 processores and our tradement TSO that error of the recent provide the defension of the tradement of the recent provides and information of TSO. Physics of the recent provides the additional and information acceptable ED.000 processores and our tradement TSO that and industry, sent that and an error of the recent provides and our tradements and industry and the recent provides and information approximation of the recommendations and that could no additional tradements provide policy above above and timescales for recommendations and additional provides and the prodements of the r
 Lany planning discussion upplices alread be conside a Frideen Salatements for a ARC a constraining object minimum processes, or want autor. The process of regun communications (of regun guided y requirements, pol- energy reformation reading apprivation of the guided interfer Salat 21 (Friduct apprivations). 1340 Devices in the salat region of process. Devices the consistent in minimum FA3 to Gore proteines. Devices the consistent process of process. Devices the consistent process. 	The APC determined that Design Organization (S/MC and Diversity application) to TSD ed by Brit Si Organization Wits. Other, product TSD implementation assues alread the analysis that have been by a supplementary Wis to be from Such condensation shall be written treas for continuing improvement to safety, incorporating SMS/GOL and numerical efficiency in the clicit teaching to an aureat systematic operagin the owned that their net region. FAA delogation and develop incorrected atom for own by the Part 21(606) ARC, and the FAA directly for any ing internating, over the next S-B months. The project along AB actifity and review existing or guidance, inducting an aureat systematic operaging the owned that develop and review existing or guidance, inducting an access specific TLOS on receivery, and my other around the aurore of to the following three observes. Inductions giftee observes. Induction giftee observes. Induction of the structure shall be addressed on a contained and the ARC clicking the around the aurore of to the following three observes. Induction when shall be addressed in a contained and the ARC clicking that they done and the aurore of acceptable 2000 processes, and contribution of TSO theory Part 21 S response interve of weir acceptable 2000 processes, and contribution of TSO theory transmission and through parts of weir about an intervention action, as well as threat that could be addressed through policy about about and timescales for recommendations, including applicability to existing lenger probability provides that account will be a transmission of the Part 21(200) ARC in its frequency index reading and threadoure in recommendations for parts and working advects to about a project group shall be addressed in runner advects to the part of the ord 21(200) ARC in its frequency reading the structure that acoustic actions in runner advects on the ord 21(200) ARC in its reading project group shall be advected as any recommendations for parts and advected in the ord and a trun p
 Lawy planning decasions uppliers alread be consider a Problem Stitutioneds for the ARCS constantion processes, or we make the constant of the par- communications of require communications of require splatticy requirements, pol- enterest of the union relating a 14 OFR Part 21. In par- me Fart 21. (Product approval process). I TSO Devices in Process. Develop unit atches it with the FAA's Close publics. The process prop- tive FAA conv. 2014. The restly with FAA's process for products the process pro- cess process of FBO of other categories of FBO 	The ARC determined that Dange Organization (SME) and Diversity applications to 160 ed by the ARCs organization Wis- Other, product T30 implementation anise should be analysis that have detailed by a supplementation Wish be twenter take. Such consideration should be invested that have detailed by a supplementation with the most take Such consideration should be written tries for continuing to excerning systems in perceptil the owerk that their net engages FAA delogation will deviate recommendations for own by the Part 21(6Mc) ARC, and the FAA directly for any ing excerning, one the exercise specific T300 an receivery will deviate even exercise or guidance, industry produce specific T300 an receivery, and phy other available associate available. Industry produce specific T300 an receivery, and phy other available associate to the knowing three observes. Industry of the Weat SH months. The protect around will deviate the available associate when the work of the triangle of the PAA's Part 21. Subpart 17 (21.000) Clean- up Matter centred framework from entry to work of weat the AA's Part 21. Subpart 17 (21.000) Clean- up Matter centred interval frame on protection when all immediately of and cate or 150. House is T31. It respects they, don't and interval acceptable 12.010 processores and configuration of T50 thates T47.11 is respectively policy attriangle and microards when interval and configuration of T50 thates T47.11 is respectively policy attriangle and an exceptable 12.010 processores and configuration of T50 design changes part marking and intervals acceptable 12.010 processores and configurations T50 thates the Part 21.31 respectively policy attriangle and an exceptible 12.011 processores and configurations of the additions at the taken acceptable 12.011 and additions in travelations at the additionation additionation acceptable 12.011 and deviations in travelations in processory and/or guidance in additionation and an exception of the approve is in the extent policy and FAA instrument o
 Lawy planning discussions upplices strand be consider a Frabuent Statements for the ARCS constantion processes, or we make the processes, or we make the processes, or we make the processes of the communications Art require planting requirements, pol- entiment with which reducing a 14 CFR Plant 21 (Insper me Flant 21 (Freduct) approved process." TSD Depresent pro- tradicts The process pro- product the process pro- product. The process pro- product cher process pro- restly with FIAA (On reserv- tion) and statements to another calegories of TSD another calegories of tSD ano	The APC determined that Dange Organization (EVMC and Diversity applications of the Top of by the ARC organization Wiss, Other, product TSC implementation assues alreaded the another data family activation by a suppresent or Wisk to be formed that. Such consideration into the when there for continuing improvement to sately, incorporating DMSIGOL and increased of decay in the of an transforming to an aureal splittmin operagin tracework that their net regions FAA delogation will device incorporate back to all the PMT 21/BMC ARC, and the FAA directly for any ing internating, over the next 5-0 months. The project along will dentify and review exciting on guidance, industry process specific T toP an receivery, and any other available aurorate and on guidance, industry process specific T toP an receivery, and any other available aurorate and to the following three observes. Incute will reper times in the FAA (Part 21, Subpart 11 (21 BAX) Diversing Mahrin canned Immund Im- remine with, shield the defension in a profession of TRO teach Part 21 is respectively top, they following these industry of the receivers. Incute will reper times allowing defension in a profession TRO teacer there are the Clarky file TRO camp- tion re-write, shield teaching, during and relation of TRO teacer Part 21 is respectively top, and information acceptable ED.001 processores and curritization of TRO teacer thereary through policy above above and therearies the recommendations including application to be exching accept probability and internation and exclusion in repert transfer and devices any recommendations including application to be exching accept to allow allows are readed and teaching actions in incurring the size by the Part 21/BMC ARC in its treat repor- tion and teaching and devices in incurring the size by the Part 21/BMC ARC in its treat repor- ation and teaching and devices in incurring the size by the Part 21/BMC ARC in its treat project functionaries and devices any recommendations in proving the file and the part provided th
 Lawy planning decasions uppliers straid be conside a Frabaen Salatements for a ARCA constantion gates miticianon processes, or we make. The proceed group communications not requir gateling view ensures, poly ensurements to make resulting in 14 OFR Part 21 (Fraduct apprivation process). TSO Devices for 20 (Fraduct apprivations). Develop view atthes to apprivations. Develop view atthes to apprivations. Develop the constituent in the FA3's (Con- continues). Develop the constituent in the FA3's (Con- continues). Develop the constituent in the FA3's (Con- continues). Develop the constituent indices the process group the FA4 every 2014. The process of TSO contains the process of TSO constructions without take to one compones of TSO conversitions within the conversition in solution. 	The APC determined find Dange Organization (SMC) and Diversity applications to 160 ed by Br ARC si Organization With Orbits, product TSC implementation assues already the tomber due to involve due to the properties of With the twinned true. Such conductation shall be written (real for confineing improvement) is safely, incorporating SMS/GOL and numerical efficiency in the clicit teaching or an atematisylation. One and by the Part 2/16MC ARC, and the FAA directly for any ingle-manufacture to an atematisylation. The properties of an atematical efficiency in the clicit teaching, over the result of months. The properties of any atematical and review existing on guidence, industry process specific TLOS an receivery, and any other analytic access of to the following three observes. Insule will report there are the FAA siles of Part 21, Subpart 15 (21, 5XK) Chinesson Matter second line and the analytic access of to the following three observes. Insule will report the second by an accession and the ARC Scharter for Clarky file TRO receivers insule will report there are the FAA siles of Part 21, Subpart 15 (21, 5XK) Chinesson Matter and the approach they done and receivers when a transfer of a solution of TSO Hodger Part 21 3 responses to writing and receivers when a transfer of a solution of TSO Hodger Part 21 3 responses to writing on processores and coordination of TSO Hodger Carl 21 3 responses of writing acceptable ELO(1) processores and coordination of TSO Hodger Part 21 3 responses to writing outprints and lineary and the approach as writing applicability to exchange teacky finder report project proves that and devices any recommendations in additional and additional transfer devices are about an develop any mounterchildows in transmentations in applicability be Part 21 (20, XAEC ARC and a finder applicability and develop any recommendations in transmentations in applicability be Part 21 (20, XAEC ARC and a finder applicability approach and anode and approach to the extend parent and and at t

Appendix B GAMA TSO Sub-team Members

NAME

ORGANIZATION

Industry Representatives:

Anderson Tenn Barber, Fred Boullay, Edmond Chevrel, Cedno Gamer, Robert Istam, Raki Mahone, Bruce Moynihan, Pat Neeld, Jonathon Parelon, Eric Rogozanski Tom Ruggles, Van Sambiase Jue Spross, Limit Southgate, Roger Tew Paul

FAA Representatives:

Brackmann, Matthew Greer, Paul Jennings, Richard Law, Douglas Linegang, Michael Powell, Victor

EVACUTAS, UTC & SAES/4 Avidyne US Crest Group Thales Odyssey If Solutions & SAE A-10 Zodias Aerospace & SAE SEAT SAE International UTAS, UTC & SAE A-4 CSafe Global & SAE AGE-2A Thules Honeywell Carmon GAMA 1.4 Rockwell Collins Universal Avionità -

AIR-130 AGC-200 AIR-130 AIR-120 AIR-120 AIR-110

GAMA TSU Sub-team Part 21/SMS Avoiton Rolemaking Committee

Appendix C TSO Program Values and Challenges

FAA Team Members' Perspective

Value:

- TSO reduces duplicate use of LAA resources to evaluate the same article multiple times for different (installations (e.g., (ewer designees))
- TSO supports simplification of after-market installation approvals (e.g., facilitates field approval process for simple modifications & replacement) and charges to type design (e.g., major/mmor).
 - Note does not apply well to things like slide rafts, fly-hy-oviru_EICAS, etc. because the TSO unicle is linghly integrated with the aircraft installation
 - Note does not apply well to cargo equipment or portable articles because there is no installation approval for the equipment.
- TSO supports interoperability of avionies with the an traffic system
- Note: primarily applies to Comm. Nav. and Surveillance equipment
- TSO supports approval of designs for component that is not installed in the aucuati
- TSO supports establishment of equipment-based operational requirements.
- TSO provides production approval for equipment independent of aircrafi-lavel approval
- TSO provides an avenue to identify common-cause COS issues that are field to a specific arbide (vir without TSO, COS issues would be field to each aircraft individually, but difficult to identify across multiple aircraft)
- FAA recognition of industry consensus standards encourages use of (celinical standards (and can require the use of the newer standard in the case of TSO)
- FAA recognition of now and updated standards encourages advancement of technology
- TSO encourages the addressing of requirements at the lower lovel supplier's level (vs. relying only in auranti-level requirements to mekle down to suppliers)

Thallengest

- Standards become outdated or misaligned with an worthiness requirements
- TSO approval can be misunderstood when TSO standards are outdated.
- TSO approval can be misunderstood when the installation requires specific performance from TSO article (beyond the minimum TSO requirements) or the TSO article's performance is highly dependent on the installation
- TSO approval can be misunderstood when the TSO article includes meanplete TSO approval, non-TSO functions, or other performance not specific to the TSO standard
- EAA does not have clear policy / emena for oversight of TSO anticle designers
- TSO adds complexity in continued airworthiness/maintenance due to the used to maintain two approval bases for the article and arcraft
- Configuration management TSO minor changes and arcraft-level changes both need to be assessed

GAMA TSU Sub-team Part 21/SMS Avialian Rulemaking Committee

795

Industry Team Members' Perspective

Value:

- Marketability
- o FAA TSO approval provides international recognition
- FAA TSO approval process is relatively harmonized with other international article approval processes
- TSO allows the manufacturer to make minor changes to the design
- o TSO supports lower cost / economies of scale for equipment manufacturers
- TSO allows the TSO holder to design, manufacture, and deliver products under a single approved quality system (vs. needing to operate under multiple higher-level quality systems)
- o TSO supports the supply of spare parts directly from the TSO holder
- \circ $\,$ TSO facilitates sharing of design data by the TSO holder with affiliated repair
- stations, supporting the provision of maintenance/return-to-service actions
 TSO provides reliable "partial credit" for compliance with installation requirements
- (TSO reduces the burden on the installer)
- TSO approval supports reduction of international authority resources for validation of articles
- TSO provides ability to generate approved data
- Use of industry standard provides accessible understanding of the article's design
- TSO supports after-market customers

Challenges:

- o Outdated standards reduce the value of TSO credit towards installation
- Integrated/complex systems are bound by multiple TSO standards which may not be consistent/compatible
- TSO and aircraft-level approval can create ambiguity regarding responsibilities for COS issues and reporting of events under Part 21.3
- Combined design and production approval for APUs limits the TSO holders' ability to take advantage of Part 21.101 changed product rule and other privileges afforded to engine type certificate holders



29

Appendix D Issues and Problem Statements Considered

Sources 1 and 2. FAA Part 21 Matrix and ARC Member Discussion

llar Xo	Story: A. Problem Statement	Bute at	Berannajadimai	Harma also a more service of	Brile
14	31 AP(C: Consult - riscon litence- a (AA), Piet 31 Else rectation			· · · · · · · · · · · · · · · · · · ·	P
15	21 ARC (Chaptorn Palaco Inactors) to cologoned department partial analysis toil and 190 Directorae			alli 120 deci minederi mazzal - si cella i a qui	2.7
10	 А.С. 1. полнот сероптонить стать different 150 системово 		Separate processing of 73Eb Electronics v. M. changed Permanently unreliantly anniv- receivable?	1	
2	21 MRC 9 compared Design and Brachistrate approval	-	Counter reportions.	AB 11/1 and rainmaning propagation for ProceED3 Non-maplicage	
-	21 MRC Design charge management performing and interests	121.019 V 121.01	True Dominio Artes de Governo -		
TP.	21 ABC Mana Shore class- Demandant to Lauffordel Interior in commun. Tallourum	1			P
54	31 APE Value of Vinces and 150 or New-TSO 1 and 150			11 second according to TWO	
P.,	3) ABC Ligamoun of APC 700. (CVIB) y For ELD	ABCMEY	-	(Decign Association) protection by Others in (1)	-
_					
-					-

GAMA TSO Sub-team Part 21/SMS Aviation Relenating Committee

°1)

Source 3, 2013 FAA Parts Approval Workshop

568 56	Saus - & Prolden Statement	itale A Statebalists	Beaumond incom	Bannak- Comount	Sub: Pair:
70	Industry specific total responsibilities and productor of solutions of total approxid		10 C 10 C	Madagar in control Tal	
Pr-	Industry There of TSQ supplications - and adaption of new TeO processes			 Strong Science of Long Compares and Alexandro Decard Contraction on American Science 	R. P
.9	Lindnery Public tetrack patting of commenders				
Pr-	Industry, Optimital replacement for all includes USO features a manufator discharmon of 01 formania (TS14 & MarcTS0)			10-10-10-10-10-10-10-10-10-10-10-10-10-1	
30	Tentowery Claraf- Data enformation idease to the specular of Tribit manyland				-Ð-
217	Industry, Automatica completions of require the TL for the CARDON Software			Control 1 1974 Annual Annual Annual Agustration annual Annual Maanamaana	
				1	
	2				

71-

GAMA TSO Sub-team Part 21/SMS Aviation Rulemaking Committee

Appendix E List of Acronyms

AC	Advisory Circular
AEG	Aircraft Evaluation Group
ARC	Aviation Rulemaking Committee
AVS	FAA Office of Aviation Safety
CAA	Civil Aviation Authority of another country
CCA	Common Cause Analysis
COS	Continued Operational Safety
DO	Design Organization
CFR	Code of Federal Regulations
DAR	Designated Airworthiness Representative
DER	Designated Engineering Representative
EASA	European Aviation Safety Agency
FHA	Functional Hazard Assessment
FAA	Federal Aviation Administration
ICA	Instructions for Continued Airworthiness
ICAO	International Civil Aviation Organization
JPDO	Joint Planning and Development Office
NAS	National Airspace System
ODA	Organization Designation Authorization
PAH	Production Approval Holder
PMA	Parts Manufacturer Approval
PSSA	Preliminary System Safety Assessment
QMS	Quality Management System
SMS	Safety Management System
SSA	System Safety Assessment
STC	Supplemental Type Certificate
тс	Type Certificate
TSO	Technical Standard Order
USC	United States Code
WG	Working Group

32

GAMA TSO Sub-team Part 21/SMS Aviation Rulemaking Committee

APPENDIX J—COST BENEFIT ANALYSIS WORKING GROUP REPORT

Part 21/SMS Aviation Rulemaking Committee Cost-Benefit Analysis Working Group

•

Final Report

June 2014

Cost-Benefit Analysis WG Final Report

TABLE OF CONTENTS

TABLE OF CONTENTS	F
II INTRODUCTION	
III. BENEFITS	
III.a. BACKGROUND: SYSTEM CONCEPT EVOLUTION	
III:6. INDUSTRY SYSTEM CONCEPTS AND BENEFITS	
III.b.I. Mandatory DO with Safety Management Systems (SMS)	
III.b.ii. Certification Concepts for "Below the Risk-Based Threshold"	
III.b.iji. Industry Benefits	
III.c. FAA NEED FOR CHANGE AND BENEFITS	
III.c.I. Industry Growth and the Need for Change	
III.c. II. FAA Benefits	
IV. INTERNATIONAL HARMONIZATION	
V. CBA METHODOLOGY	
V.a. CBA METHODOLOGY DVERVIEW	
V.6. PROCESS IDENTIFICATION	
V.c. RESEARCH	
V.d. COLLECTION AND ANALYZING THE DATA.	
V.d.i. COST BENEFIT SURVEY OF SMALL ORGANIZATIONS	
V.d.i. COST BENEFIT SURVEY OF LARGE ORGANIZATIONS.	41
V.d.III COST BENEFIT ANALYSIS OF 5MS	
VI. LESSONS LEARNED	
VI.a. ROLE OF A CBA WG IN ARC ACTIVITIES	
VII. CONCLUSION	
REFERENCES.	
APPENDIX A	
Part 21/SMS Cost-Benufit Working Group Cost Survey of Large Companies	
APPENDIX B.	
Part 21/SMS Cost-Benefit Working Group Cost Survey of Small Companies	
APPENDIX C	
	-

Cost-Benefit Analysis WG Final Report

68A Working Group Members

Cust-Benefit Analysis W& Final Report

Vage I

100

II. INTRODUCTION

Historically, aviation safety rulemaking cost-benefit analysis (CBA) has focused on the prevention of accidents and lives lost as the benefit justification to offset costs of implementation of new rules. In evaluating the proposed transition to a systems-based approach to new rules, the purpose of this Part 21/SMSARC activity was to create a rule that ralies not upon accident prevention as a justification, but rather a rule that can be justified and supported by enhanced improvements to the affectiveness and afficiency for both industry and the FAA. As a result, the Cost-Benefit Analysis Working Group (CBA WG) was tasked to find new ways to capture the benefits and costs associated with such improvements and identify a methodology that supports thic.

The CBA WG charter identified five deliverables that have been addressed and are contained inthis report. Those deliverables are:

- · Benefits associated with moving to a systems-based approach;
- Benefits associated with harmonized regulatory frameworks for organizations with bilateral partners;
- Overall industry benefits related to safety, efficiency, and effectiveness;
- Benefits associated with FAA process changes;
- New and/or alternative cost-benefit methodologies to assist the FAA's required aconomic analysis of rulemating.

Data gathered by the CBA WG was intended to be supporting data only and does not represent a formal economic analysis. All data gathered by the CBA WG has been shared with the Economic Analysis Division (APO-300) in the Office of Aviation Policy and Plans at the FAA and will be referenced during the formal economic analysis in the event a rulemaking project takes place.

Lost-Benefil Analysis WG Final Report

lage I

III. BENEFITS

III.a. BACKGROUND: SYSTEM CONCEPT EVOLUTION

As the aviation industry steadily continues to grow, FAA resources continue to remain level. The increase in design and manufacturing is a clear benefit for the U.S. according; however, if our system does not adjust with the growth in aviation, the FAA will not be able to provide the same level of oversight and support as is currently provided today. A "systems approach" must be adopted to accommodate the increase in aviation design and manufacturing. Shifting toward a systems approach will enable the industry to continue on its current path while FAA resources can focus on higher risk projects and areas where resources are better utilized.

Figure 1 below represents an illustration of the transition from the current Organization Designation Authoritation (ODA) system of today to the future state of a "Design Organization (DO)" (as envisioned by the Part 21/SMS ARC).

The left side of the illustration depicts the DDA model as it is intended to function today. Today's system for certification is based on a "showing" (i.e., of compliance) by the applicant and a "finding" by the FAA. For every action by the applicant, there has to be an equal action by the authority on a project-by-project basis. Each star in the image in <u>Figure 1</u> represents a "showing" by the applicant and a "finding" by the FAA.

Transitioning to a DO would reduce the amount of FAA involvement and allow the industry to work with more independence and responsibility. Following the transition arrow in Figure 1 shows how the current ODA model would transition from one using discrete findings of compliance to one that applies systems oversight. A DO is a regulatorily-recognized organization that meets organizational and system requirements sufficient to ensure that it is capable of making compliance determinations upon which the FAA may rely in support of obtaining design approvals under 14 Code of Federal Regulations (CFR) part 21. As such, the DO maintains and follows processes to manage its certification projects as well as the continued airworthiness of its products. This method allows for FAA oversight of the process, less direct project involvement by the FAA, and reduced one-for-one show/find, all while maintaining and, ultimately, enhancing safety for U S, design and manufacturing organizations.

The CBA WG explored this transition by researching the costs and benefits of moving to a systems approach. The results of our preliminary review have been articulated in subsequent sections

Lost-Benefil Analysis WG Hinal Report

Jage I



Cost-Benefit Analysis WG Final Report

III.b. INDUSTRY SYSTEM CONCEPTS AND BENEFITS

III.b.i. Mandatory DO with Safety Management Systems (SMS)

The CBA WG approached its methodology under the ARC's original assumption that the optimal organization of the future would be comprised of a mandatory DO with the inclusion of SMS. The specific attributes of a "DO plus SMS" can be referenced in the Organization and the SMS Working Groups' final reports.

A definition of the methodology assumption has been documented below. This definition comes from the Organization Working Group's final report.

"A DO is a regulatory-recognized [sic] organization that meets organizational and system requirements sufficient to ensure that it is cupuble of making compliance statements upon which the FAA may rely in support of obtaining Design Approvals (DA) under 14 Code of Federal Regulations (CFR) part 21. As such, the DO maintains and follows processes to manage its certification projects, as well as the continued airworthiness of its products.

A DO will include organizations applying for, supporting the opplication for, or holding Type Certificates (TC), Supplemental Type Certificates (STC), or Parts Manufacture: Approvals (PMA)."

In addition to this definition, it was assumed that implementation of SMS would be required of all parties in order to become a DO. Therefore, the CBA WG used this assumption and definition in developing the methodology, conducting surveys, and all other data gathering techniques.

[It should be noted here, however, that, at this point in time, the final conclusion reached by the ARC was that 5MS should be mondatory, but that DO should be voluntary. This conclusion was not reached until late into the time frame for which the ARC was chartered. The CBA WG activity and this report are based on the anginal assumption that both SMS and DO would be mandated based on the definition given above]

Lost-Benefil Analysis WG Hnal Report

lage (

III.b.ii. Certification Concepts for "Below the Risk-Based Threshold"

The ARC has identified three potential options for any applicant falling below the risk-based threshold identified as the requirement for SMS implementation and a DO. The risk-based threshold is based on the potential risks posed by the applicant's products to aircraft safety. If a DO threshold is defined at a later date, the three options will be applicable to that threshold. The risk-based threshold is suggested in this case due to the uncertainty of the DO threshold. The applicability threshold requires SMS for organizations that:

- Design or manufacture products (i.e., aircraft, engines, propellers) or;
- Design or manufacture articles whose failure muld directly prevent continued safe flight and landing; or
- Make design changes to a product, through a Supplemental Type Certificate (STC), failure of which could directly prevent continued safe flight and landing.

The objective of having these options is to ensure that small companies that do not meet the above criteria for implementing SMS and becoming a DO have the opportunity to gain the same benefits as the large established companies without incurring the same level of cost and complexity to certificate products. The impact of requiring substantial ramp up cost for those companies falling below the risk based threshold could not only hurt small business, but could negatively immact the economy as a whole. The goal of recommending change to part 21 is to streamline certification so that industry and the FAA are able to continually improve the current high level of safety, while keeping up with industry growth. Three options for organizations "below the threshold" have been identified to encourage small businesses and innovation in the sector. Those three options are:

- Accredited Diganization (AO);
- Agent DO; and
- Modified Current Model.

Each option has been articulated below and is designed to reduce the daily involvement of the FAA on low-risk activities while still maintaining or improving aviation safety. Because of time and resource constraints, the ARC was unable to focus on companies failing below the threshold. Therefore, these options have been explored by the CBA WG as a preliminary effort. Further research and definition are required for each option following the conclusion of this ARC.

Accredited Organization:

The Accredited Organization (AO) approach would seek to achieve many of the same objectives as the Mandatory DO/5M5 approach, but with an aim toward making such achievements cost effective for smaller businesses while maintaining or improving safety. This would be an optional program, with a goal of voluntary adoption by industry. In order to parallel or

Loct-senefit Analysis WG Hinal Report

Page 1 N

harmonize with the international mandate for SMS principles, the AO would implement Continuing Operational Safety (CDS) principles or SMS principles. CDS incorporates many, but not all, principles of SMS and is a system which many "below the threshold" companies are already familiar with, or have already implemented.

Those companies seeking to export products to SMS-requiring ICAD countries may elect to implement appropriate additional SMS requirements; however, adoption of SMS requirements would not be mandatory for those implementing an AD.

Adoption of certain established COS/SMS principles would be required for accreditation to a voluntary industry standard. The AO approach would be a voluntary industry accreditation program similar to that described in the FAA's <u>Advisory Circular 00-56A</u>"Voluntary Industry <u>Distributor Accreditation Program"</u> for distributors. Under such an approach, one or more industry organizations (e.g., SAE) would develop industry accreditation standards articulating the necessary requirements for an AO system, including implementation of COS/SMS principles. Those "balow the threshold" companies successfully accredited to such standards would he identified in an FAA document indicating compliance. The objective of such an accreditation program is to encourage voluntary participation by industry, and for customers to make inclusion in such a program a preequisite to doing business. As accreditation to the standard becomes an accepted norm, more companies would seek to implement COS/SMS principles under the AC model.

The AO approach may also provide "below the threshold" companies that meet the required accreditation standard with the opportunity to utilize an approved "compliance library." The compliance library could be one that the AO develops on its own and has FAA approval, or one that is developed as part of a consensus standard that is approved by the FAA. If a consensus standard was to be utilized, the AO would use only those standards that are on a list approved for the AO by the FAA. The compliance library would enable below the threshold companies to take advantage of the reliability indicated by accreditation to self-start projects fitting in its compliance library. Self-starting projects from its compliance library permits the AO to avoid the FAA sequencing/project prioritization queue and to more quickly initiate projects, thus bringing products to market more quickly. Projects not within an AO's "compliance library" would still be subject to FAA sequencing/project prioritization. The compliance library could be expanded to demonstrate more competencies that would permit the AO to avoid the FAA queue for additional projects.

The AO approach will also provide for a reduced Level of Project Involvement (LOPI) by the FAA. The extent to which LOP) is reduced or increased will depend upon the complexity of the proposed project. Complex projects will involve more significant LOP(from the FAA in terms of systems oversight and findings, though not using to the level of the one-for-one show-find process of the current model. Those projects of diminishing complexity will allow for correspondingly reduced LOPI, in some cases almost zero LOPI, reflecting their lower level of complexity and potential effect on safety, thus preserving FAA resources.

Cost-Benefit Analysis WG Hinal Report

Page I

Agent Design Organization:

Agent DO is a concept that needs further discussion and understanding by both the industry and the FAA. Persons or companies qualifying as an Agent DO would be similar in nature to current Consultant DERs, though responsibilities would shift in a similar manner as the Accredited Organization approach.

Such Agent DOs would provide cost-effective options for small companies, STC applicants, and small PMA applicants who do not have the resources to implement an Accredited Diganization approach, or who have limited needs for FAA resources due to limited applications.

An Agent DC would have authority to provide services to comply with the 14 CFR §21.3 (Reporting of failures, mailunctions, and defects) requirements and COS monitoring on behalf of holder of a type certificate (including amended or supplemental type certificates), a PMA, or a TSO authorization, or the licensee of a type certificate. This would enable small companies to comply with the §21.3 and COS requirements without maintaining the required resources on their own.

As mentioned above, this concept needs further exploration and the Agent DO approach would not be applicable until DOs become voluntary.

Modified Current Model:

The Modified Current Model approach is a simple reflection of the use of designees by small businesses, including PMA companies and STC applicants. Although the process has not been changed, the model is "modified" in the sense that the FAA anticipates a reduction in the number of DERs who would be available to provide services to these small applicants. The number of DERs would be reduced by approximately two-thirds of current numbers by attrition, non-renewal of privileges, and limitations of new DER privileges. Sequencing, Riskbased Resource Targeting (RBRT), and/or Project Prioritization would still be applicable, causing certain applicants to be substantially delayed in the FAA queue based on the perceived value and safety considerations of their application, while giving preference to those applicants whose projects are deemed having a greater Impact on safety.

Criteria for applicant-only-showing would be developed by means of a standard (e.g., ISO, SAE) for low risk projects. Specific criteria for applicant-only-showing would include such things as:

- the article being considered low-risk;
- the project meeting criteria of the compliance library, and
- the applicant's ability to issue a §21.20 statement.

If the standard is adopted, there will be privileges gained by doing so. A compliance library would be developed and accessible in a repeatable manner, allowing the applicant to initiate low-risk projects independent of the FAA. However, in order for the applicant to take

Lost-Benefil Analysis WG Hinal Report

Page | 11)

advantage of the privileges, all criteria must be met. If there are deviations from the defined criteria, then there will be stipulations that would potentially involve the FAA or a DER. Additionally, the company would have a system in place to meet the specific criteria and have a process with proper oversight and/or checks in the system. The system would have reporting requirements back to the authority regarding self-disclosures and a corrective action program regarding non-compliances. This system would then provide for additional privileges under applicant only showing program.

This Modified Current Model will encourage those companies that can afford to implement a DO or an Accredited Organization to do so, in order to avoid being subject to increased queue time with the FAA. Smaller businesses unable to establish an Accredited Organization or DO inay be subject to delays in compliance findings by the agency in areas of FAA involvement, such as flight testing, software, human factors, noise, test witnessing, and inherently governmental functions, such as exemptions. The FAA would balance the certification needs of applicants producing future products for the National Air Space (NAS) versus maintaining an ever-growing COS responsibility within the same NAS. The majority of FAA resources are currently focused on lower-risk projects due to the sheer number of projects initiated each year. Transitioning smaller business and lower-risk companies to this type or system could decrease product time to market for the individual companies, along with allowing the FAA to focus resources in areas of greater need.

Lost-Benefil Analysis WG Hnal Report

III.b.iii. Industry Benefits

With both concepts, either above or below the threshold, there are a number of overall industry benefits related to safety, efficiency, and effectiveness

- Decreased product time to market;
- Less time spent in sequencing;
- Increased project authority;
- Reduced time spent guiding the FAA during audits.
- International harmonization;
- Scheduling control,
- Elimination of redundant conformity;
- Increased work efficiency (due to schedule control, less interruption, less wait time); and
- Centralized Oversight Office.

The benefits gained are further described in each of the ARC Working Group reports Each benefit was identified through the data gathering process during the CBA surveys, presentations and workshops. The benefits were not assumed by the CBA WG, but described by ARC members and survey participants. The CBA WG prefers to see further research be performed on each benefit listed to realize the full worth in cost and efficiencies. Due to the understandable reluctance of many survey participants to provide specific cost date and the late realization of other working group recommendations, the CBA WG did not have sufficient time to fully explore the potential of each benefit. A benefit survey is strongly encouraged as a change in methodology to the commonly applied economic analysis process. Fully understanding the benefits will assist applicants in realizing the true reduction in cost due to benefits, compared to the common survey that only requests information on costs incurred. During the data gathering process, the CBA WG found that it was difficult for survey participants assign a monetary value to benefits. They understood the benefits from an intangible and efficiency perspective, but were unsure how to document them. This was a valuable lesson learned for the CBA WG, emphasizing the importance of a benefits survey.

Lost-Benefil Analysis WG Hnal Report

Page | 1_

III.c. FAA NEED FOR CHANGE AND BENEFITS

III.c.i. Industry Growth and the Need for Change

The U.S. civil aviation industry has a crucial role in fostering trade and making any place on the global easily and quickly accessible. The U.S. industry and travelers depend on the viral services of air transportation to support the U.S. economy. Even with severe fluctuations in the scontomy and the recent government sequestration, the sylation industry has been able to continually readjust and regain stability.

The following points indicate the importance of the industry:

- In 2009, an carriers operating in U.S. airspace transported 793 million passengers over 1,039.3 billion revenue passenger miles (RPM).⁷
- More than 53 billion revenue ton-miles (RTM) of scheduled freight passed through U.S. airports in 2009,¹¹
- The U.S. civil aviation manufacturing industry continues to be the leading U.S. net exporter. According to 2009 data from the U.S. International Trade Commission (USITC), the U.S. civil aviation manufacturing industry supported a positive trade balance of over \$75 billion.⁷
- The 2011 FAA Aerospace Forecast projected an average annual growth rates of 3.8
 percent per year through 2031 for U.5. airlines.⁷

The U.S. aerospace industry may be facing some of its greatest challenges in decades. While weathering several trials in 2013, the industry produced relatively flat results compared with 2012. An overall slight decrease in sales was forecast, reaching \$220.1 billion for 2013 – down from \$222 billion in 2012 – with only civil aircraft sales showing growth.⁴ According to data from the Aerospace Industries Association (Figure 2), however, aerospace industry annual sales have increased more or less continuously from 1998 through 2014.

Lost-Benefil Analysis WG Hnal Report



FAA and industry data show that even through difficult transitions, U.S. aviation continues to innovate and grow. Although this is good news for the industry and the economy, there is increasing concern about the FAA's ability to oversee and support this continuous growth at its current and projected capacity. The Bureau of Transportation Statistics (BTS) projects employment in the broader transportation industry to increase by 0.7% annually through 2022, but conversely projects an annual decline of 1.5% in government employment over the same period. Additionally, the FAA, AVS specific, has maintained a relatively flat increase in hiring. This is due to the slow and steady hiring rate combined with the attrition rate. <u>Figure 3</u> illustrates this gap.

Cost-Benefil Analysis WG Final Report



If, as forecast by the FAA, airline passenger services grow at a rate 3.8% per year through 2031, but FAA employment steadily declines by 1.6% each year, then there will be a significant gap between the aviation industry's demands for FAA services, compared with the capacity of the FAA to support. This gap could translate into the stifling of innovation and entrepreneurship and, possibly, even a decline in air travel.

For certification activity specifically, the FAA has devised strategies to mitigate the increasing resource gap. However, strategies like designee programs and sequencing have reached their limits in their ability to mitigate the gap. The result has been an increase in sequencing wait times and the areas of focus of FAA employees. The United States has reached a point where significant change is needed in its aviation certification system. The Design Organization and SMS concepts, if implemented, could provide the solution to address this challenge.

Lost-Benefil Analysis WG Hinal Report

III.c.ii. FAA Benefits

Shifting to a systems approach could provide huge benefits to the FAA. The FAA is rapidly moving to a situation where current resources are unable to support the amount of work being requested at local offices. If the FAA were able to support certification activity through application of a systems approach, then resources would be able to better focus on areas of greater need. In its 2011 The Economic Impact of Civil Aviation on the U.S. Economy report, the FAA found that its rate of hire is equal to or less than its rate of attrition; therefore, even with current levels of hiring, the number of agency employees is decreasing. As long as this trend continues, the growth in industry will always outpace the growth of agency resources.

The CBA WG is aware that the benefits accruing to the FAA are not the types of benefits typically considered when conducting an economic analysis under OMB guidelines for new rules. However, the level of FAA resources is a key contributing factor to the industry growth rate, evolution of technology, and product time to market. Therefore, the CBA WG took FAA benefits into consideration when understanding the impacts of moving to a systems approach for both the industry and the FAA.

If the industry moves loward a DO with the inclusion of SMS, some FAA nenefits would include

- Increased standardization of oversight;
- Accountability Framework;
- Less time spent with new applicants;
- Less time spent on low-risk projects;
- Less time spent on product audits;
- More time spent on new technology, research and development, areas of higher risk (Reallocation of resources); and
- More efficient and streamlined certification processes.

Thus, according to research and the data gathered, multiple benefits would be malized by the FAA, as well as the industry.

Lost-Benefil Analysis WG Hinal Report

Page | 1)

IV. INTERNATIONAL HARMONIZATION

All parties recognize that civil aviation authorities and the aviation industry have access to finite and often very limited resources. These governmental and industry resources must be focused to produce the greatest safety benefit possible. Harmonization of aviation safety and regulatory requirements has been a long-standing goal of the United States and, in fact, of all ICAO Member States. However, it is recognized that such harmonization of aviation safety standards must be based on reliable safety data and, ultimately, on the potential for decreased risk. The initial recommendations of this ARC, if accepted, would not only move the FAA toward a certification oversight system that is more harmonized with many of the aviation trading partners of the United States, but could also help to move other ICAO Member States toward similar systems.

More than twenty years ago, the (then) U.5 General Accounting Office (GAO) investigated progress being made in the harmonization of aircraft design standards and found, *Inter alia*, that

- Varying (non-harmonized) certification processes are inefficient and raise manufacturers' and users' costs because of different interpretations of regulations, which impose additional requirements and duplicate certification activities,
- Regulatory resources spent on duplicative activities can be better used to address other safety issues, and
- 5. Common standards and practices lower costs and increase safety.¹¹

The introduction of risk-based safety management principles to certification systems is a significant change to the existing traditional certification system that relies primarily upon a large corps of government inspectors overseeing each aspect of certification. Unilateral implementation of such a risk-based system, or adoption by only a handful of authorities, could create serious problems due to the diversion of the new and old systems and the resulting problems with lack of mutual recognition and acceptance of product certifications outcomes.

In Europe, EASA has moved ahead in implementing and integrating safety management principles. With the recommendations of this ANC, the FAA is moving in the same direction to maintain harmonization and compatibility. This harmonized effort also aids the international aviation industry, with its increasingly global supply chains – ones that require "interoperability" in their SMS applications

Harmonized rulemaking on all sides is of litmost importance, not just for national civil aviation authorities, but also for their aviation and aerospace industries. In Europe, for example, certain certification approvals of major type design changes are going to be issued in the EU by the European Aviation Safety Agency (EASA)-approved organizations themselves. This entails

Cost-Benefit Analysis WG Final Report

Page | 1/

minimal (or no) certification involvement of the Agency, except the ongoing checks performed as a part of continuing oversight of the design organization in observance of the Level of involvement (LOI) principles. If such approvals were not to be accepted by the FAA because of "lack of EASA involvement," it would mean serious problems both for manufacturers and regulators, as it would require two almost entirely separate certification processes (one satisfying FAA requirements and one implementing LOI).

In addition to the expected safety and economic benefits of implemented safety management principles, harmonized rulemaking action by civil aviation authorities worldwide is necessary to maintain a level playing field for their aviation and aerospace industries. Harmonization is pritical to the economic health of the worldwide aerospace industry. For example, the FAA and EASA oversee aerospace manufacturers that represent over 60% of the global aviation market. A harmonized EASA/FAA approach to certification would greatly assist ICAO in the further development of this concept as a global standard. The adoption of development of these systems in a timely manner may be critical, as other countries may advocate different and, in many cases less vigorous, systems as the basis for ICAO Standards, which may not be consistent with our approach. As all ICAO Member States need to implement SMS in order to meet ICAO requirements, the pioneers in this area may ultimately dictate the standards by which we will need to abide.

The evolution from the reliance on individuals to the reliance on approved organizations (under continuing oversight) will align the FAA closer to the EASA system, and that of some other ICAO states, with a clear benefit to both U.S. and European industries (among others due to a potentially increased scope of mutually acceptable design changes and repairs). The FAA and EASA systems would certainly be more aligned – adopting the concept of approved organizations granted certain privileges, and their corresponding responsibilities – instead of the current FAA delegation principle, relying on the individual designees (DER, DAR) and/or the DDA concept.

Implementation of a new system of approved design organizations with privileges will support the implementation of compatible DOA/SMS models from which all civil aviation authorities would benefit both mutually and internationally. The FAA, as a certification oversight authority, would benefit in the development of an organization/management system that would enable an environment for more focused attention on areas of concern, and, where necessary and appropriate, enforcement.

If the SMS concept is implemented in a harmonized manner with other civil aviation authorities the industry will benefit from easier mutual recognition of such approved organizations and their resulting SMS systems. Again focusing on the FAA-EASA example, the U.S.-E.U. Bilateral Aviation Safety Agreement (BASA) would not need to deal with substantial system differences on both sides and this would, in turn, have a positive effect on oversight resources required within the scope of BASA activities. Similarly, if the LOI concept in certification projects is implemented on both sides in a compatible way, it would ensure that the traditional "one for one" compliance showing-finding process would be replaced by a lisk-based compliance.

Lost-Benefil Analysis WG Hinal Report

verification process. It will be accepted that the level of authority involvement may vary, depending on the level of performance of approved design organizations and an assessment by the authority of the other safety risks in product certification.

The benefits of the harmonization of certification processes and procedures may best be summarized by a paper that the FAA prepared for an ICAO regional meeting in 2011:

"The United States supports a harmonized regulatory system in which oirworthiness standards and the certification processes to ensure compliance to these standards are based on universally accepted data and feasible for implementation. Harmonization of rules and processes amongst the increasing number of international aviation regulators should be a goal for all Civil Aviation Authorities (CAA), as common requirements facilitate global acceptonce."⁴⁰

Movement loward adoption of risk-based safety management principles into certification systems, as outlined by this ARC, would not only move the FAA toward a system that would strengthen aviation safety, but one that would be harmonized with that of other major aerospace trianufacturing States.

Lost-Benefit Analysis WG Hinal Report

V. CBA METHODOLOGY

V.a. CBA METHODOLOGY OVERVIEW

The methodology identified in this document is only one alternate means of identifying the costs and benefits of rulemaking. This approach includes a number of different data collection tools and options for sampling. Historically, aviation safery rulemaking cost-benefit analysis (CBA) has focused on the prevention of accidents and lives lost as the benefit justification to offset costs of implementation of new rules. The methodology applied in this report focuses of the collection of real data and active interaction with the parties affected (small and large organizations, as well as the FAA). As the industry continues to change, the FAA must maintain adequate skills in promoting change to enhance safety. This methodology approach was chosen as a proactive method to identifying cost and benefits at a preliminary stage of the ARC and rulemaking process. The costs and benefits described have been identified through industry participation and are not the assumptions of one individual.

The CBA WG worked in synch with the full ARC during the development of ARC recommendations. Due to the uncertainty of final recommendations to be made, and the concurrent preparation of this report, the CBA WG was required to make certain assumptions based on the ARC direction at an early stage. The CBA methodology general guidelines that were actively followed throughout the ARC duration included:

- Process Identification This is the area of work where the CBA WG identified the process that would be followed in order to identify costs and benefits for the Part 21/SMS ARC This process included mapping out subsequent steps
- Research Performed research of academic, FAA and other government agencies, aviation and other industry cost benefit approaches. Used this research to determine whether approaching a methodology other than the historical approach to CBA by the FAA was feasible, attainable, and something this ARC should consider.
- Bata Collection Once research had been performed, the ARC was at a more mature stage in understanding the direction the recommendations would take. Using the recommendations, the CBA WG began the data collection phase. This phase included developing templates for surveys, presentations and workshops. The CBA WG actively worked with ARC participants, external workshops and the other working groups to populate templates, hold discussions, and gather data.
- 4. Analyze the Data Once the data was gathered, the CBA WG met multiple times either through face-to-face meetings or telecons to discuss and analyze the data as a team. This involved reviewing and understanding the responses provided in surveys and discussion, following up with participants when needed, and scheduling future

Lost-Benefil Analysis WG Final Report

Page | 21)
discussions to gather follow on data as the ARC progressed.

 Conclusions – The conclusions of this effort were documented based on the findings of the data gathered. The overall conclusions consisted of identified gaps and findings that required follow on effort or research in the event a rulemaking activity takes place.

Although this methodology is quite high level, this is the process the CBA WG used as guidance throughout the ARC timeline. This methodology served as a tool for organizing the working group as it worked in synch with the ARC and was used as a guide to know what and how data should be gathered. Each guideline required significant effort and, in some cases, took a substantial amount of time. Given the ARC's chartered timeline, each phase of this methodology needed to be shortened to accommodate the full ARC. Working in alignment with the ARC proved to be less streamlined than anticipated; however, the CBA WG adopted these concepts in order to achieve the results of the concepts being explored by the ARC. This report summarizes each phase of the methodology and contains all noteworthy data gathered.

Additionally, calculating efficiencies was a primary goal in this methodology. The CBA WG concluded that in order to calculate efficiencies effectively they must first be defined. In most cases the CBA WG found that efficiencies can be calculated if they are defined in a detailed manner. Requirements that are process oriented tend to be more difficult to calculate, because a process is intertwined with multiple facets of an organization's function. The process must be broken out by physical aspect in order to calculate the cost incurred or reduced. Examples include administrative functions and physical time. Efficiencies often translate into intangible activities that an organization is unable to define from a physical perspective. In this event, qualitative assessments must be made. An educated calculation should be determined through organizational history, research, forecasting, atc.

Calculating efficiencies is an activity that the CBA WG did not have the opportunity to spend adequate time an. This will involve working with the industry on perceived benefits and translating them into cost values.

Lost-Benefil Analysis WG Final Report

V.b. PROCESS IDENTIFICATION

The Process Identification step is the area of work that Identified the process to be followed in order to identify costs and benefits for the Part 21/SMS ARC goals. The development of a Cost Benefit Analysis Warking Group was the first step this in process identification. During the CBA WG's first face to face meeting, the team worked to develop its plan of action, identifying how the working group wanted to proceed with accomplishing its charter's goals.

The process identification took into account that the ARC was still in a premature phase and had not agreed upon final recommendations. Given that the CBA WG was assigned to work in synch with the ARC, the working group chose to perform research and to develop mithods for gathering cost and benefit data in parallel with the ARC as it explored recommendation concepts.

Some of the methods included performing research, developing bemplates, and surveys to gather data supportive of the ARC goals. Multiple templates were drafted and presented to the full ARC on how to best gather certification baseline cost. The certification baseline was defined as the cost to industry for conducting certification in today's system. The baseline cost would then be compared with the cost and benefits that would actrue if specific ARC recommendations were to be implemented.

Templates and surveys were drafted for small and large companies to populate. A decision was made very early on by the CBA WG to gather data from all companies, small and large. Once the ARC decided it would be unable to fully address organizations falling "below the threshold, the CBA WG chose to continue including small business in the event the ARC recommendations changed.

Following the development of the templates and surveys for gathering of cost and benefit data, the CBA WG began to fully research other agencies, industries and private organizations for useful methods of gathering cost and benefit data.

Lost-Benefil Analysis WG Hnal Report

V.c. RESEARCH

The benefits of aviation regulations have traditionally been calculated using a "reactive" process – that is, the analysis of past accidents and the calculation of how many lives would be potentially saved and hull tosses/accidents potentially prevented by the introduction of corrective actions to prevent the recurrence of an accident event. Because the past decade has been the safest statistically in modern airline history – having the fewest fatalities and accidents – it is becoming increasingly harder to justify the cost of new requirements in traditional safety terms (i.e., the number of accidents prevented or lives saved). New regulations can no longer claim credit for avoiding a past accident that was already addressed in a previous rule.

From the outset, the objective of the Part 21/SMS ARC was not aimed towards new design or operational requirements that would primarily save lives; rather, it was aimed towards requiring enhanced improvements in processes, procedures, and oversight that would promote comprehensive efficiencies to both Industry and the FAA, as well as enhancing safety <u>generally</u>. The CBA WG was tasked to find new ways to capture the benefits and costs associated with such changes.

With this objective, the CBA WG started out by conducting a "literature search" for cost-benefic studies that may already have been performed by government agencies, academia, or others, that focused on capturing the costs/benefits of regulations that bring "efficiencies" rather than improved safety specifically. The following discussion describes seven relevant items that were identified and reviewed.

1. FAA's Final Rule on Organization Designation Authorization (ODA)

In the Regulatory Evaluation for this rule, the FAA determined that the cost of compliance with the new rule would entail only the additional costs to apply for and to operate an ODA, over and above the costs for an existing designation authorization. The total costs to operate an ODA are <u>not</u> the costs to comply with the rule. Thus, FAA calculations for used only on those costs involved in transitioning to an ODA from a current organization model.

FAA did break down these compliance costs to two types! (1) initial compliance costs, and (2) annual compliance costs. Initial compliance costs were the one-time costs to develop the procedures and apply to the FAA for an ODA. Annual compliance costs were the yearly incremental costs resulting from any changes in the company's practices, procedures, personnel training and retraining, self-audits, recordkeeping, etc. to fulfill the ODA requirements.

Throughout their final regulatory evaluation, the FAA mentions the "officiencies" that the QDA pile will bring, both to industry and to FAA. For industry efficiencies, FAA did provide a

Lost-Benefil Analysis WG Final Report

quantitind illustration of the potential benefits by measuring the value of <u>time-savings</u> as an opportunity cost of capital approach. In this example FAA stated

"[I]die capital equipment ties up investment capital with no return. The anaartunity cost of this idle equipment equals the value of the equipment multiplied by the interest rate, multiplied by the time the equipment is not used. For example, the annual value of a new transport category production oirplane is about 550 million. Using a 5 percent interest rate, the cost of a one day increase in manufacturing time would be about \$6,850. If the rule were to save an average of one-half a day of manufacturing time, the manufacturer would save \$3,425 per airplane.

The FAA does no these officiencies back to safety where it states:

"By shifting our inspection focus from reviewing test results towards overseeing the designation program, we will more efficiently use our resources while extending our oversight coverage, thereby increasing safety."

2. EASA's Notice of Proposed Amendment (NPA) 2013-01 (A), Embodiment of Safety Management System (SMS) requirements into Commission Regulation (EC) No 2042/2003 RMT.0251 (MDM.055)

EASA describes its proposed rule as supporting a "holistic approach" towards management systems by incorporating safety management principles into the management systems of organizations as well as the regulating authorities. The rule formally requires the implementation of SMS in a number of types of organizations, including Approved Training. Drganizations (ATOs), holders of Flight Simulation Training Device (FSTD) qualification tertificates, aero medical centers, and others.

In its formal Regulatory Impact Assessment (RIA) for this proposed rule, EASA described the technical requirements of the rule as mainly focused on:

"... the creation of streamlined, consolidated management system requirements that, while built upon existing quality systems, improve consistency in organisation approvals, and introduce additional requirements related to hazard identification, risk evaluation, and effective risk mitigation."

The working method adopted for the RIA was to perform a "<u>aualitative</u> assessment of possible imputs."

Lost-Benefil Analysis WG Hnal Report

EASA's assessment contains no quantitative data, but does provide an in-denth analysis of the impact of the rule, as to:

- + safety impacts,
- social Impacts:
- + economic impacts for Industry
- economic impacts for authorities;
- · proportionality issues (i.e., size, complexity of organizations); and
- Impact on regulatory coordination and harmonization (including ICAO standards).

In its conclusion, EASA found that the need for the regulation was supported on the grounds of safety, increased efficiency, and international harmonization.

3. FAA Report, "Business Case for the Next Generation Air Transportation System (August 2012)"

In this report, the FAA addresses the potential air traffic management requirements of the Nexi Generation Air Transportation System (NextGen). It considers the shortfalls in the current system that new technologies can help to alleviate, and the "costs and benefits" of doing so.

This business case focuses on the direct benefits to aircraft operators, passengers, and texpayers from the rollout of NextGen improvements. These benefits include:

- Improvements in system <u>capacity</u> (i.e., reductions in flight and taxi times and corresponding fuel use resulting from less delay; reductions in cancelled flights; additional scheduled flights that are enabled by new capacity), and
- Improvements in system <u>efficiency</u> (i.e., reductions in flight times and fuel use due to more direct routings, and reduced fuel use due to more efficient descent profiles).

FAA was able to monetize these benefits and efficiencies by relying on various inputs to estimate a cost vs benefit conclusion. As part of the estimating process, they employed an elaborate modeling system – System Wide Analysis Capability (SWAC) – whose methodology involves a fast-time simulation model used to estimate the potential benefits of NextGen Improvements in the National Air Space (NAS).

Based on these inputs, FAA was able to show in its analysis that NextGen mid-term improvements will generate \$106 billion in benefits for the nation as a whole through 2030.

Lost-Benefil Analysis WG Hnal Report

4. "Aviation Safety Management Systems Return on Investment Study," Center for Aviation Safety Research (CASR), Parks College of Engineering, Aviation and Technology, Saint Louis University; February 2011

This report is a result of FAA's tasking of CASR to develop a model to show the investment benefit of SMS implementation and sustainment vs. the costs of developing the program, as well as costs associated with incidents and accidents. The model is built around the concept that the large costs associated with accidents could be reduced or avoided with the implementation of a safety management system. These reduced or avoided costs could then be seen as a net gam and placed into a <u>return-on-investment</u> model for safety management system investment calculations.

The CASR team reviewed how accidents and incidents affect the financial results of an aviation organization at three levels

- Macro-level, where shareholder and market value is lost due to "bad publicity" occurring post-accident/incident. (Examples of costs: drop in stock price, or change in public perception of air travel safety.)
- Mid-level, where losses are absorbed as part of the regular costs of doing business, most particularly if SMS programs do not exist that would prevent such moldents from occurring. (Examples of costs; raised workers companisation premiums, logistical costs of a product recall, or the time that vital equipment is out of service.)
- Micro-level, where individual accidents/incidents in a company that inour custs that could have been avoided or mitigated if an appropriate SMS program were in effect. (Examples of costs: damage to products or facilities, schedule delay penalties, and parts failures)

Gathering these costs, the CASR team then used the following formula to determine the return on investment (ROI) of implementing an SMS program:

Return on Investment = (Payback - Investment) + Investment

CASR concluded that:

" _ a properly executed ROI analysis of safety interventions allows organizations to compare sajety investments with other competing business investments and assist in decisions regarding financial resources."

Lost-Benefil Analysis WG Final Report

5. U.S. Environmental Protection Agency (EPA)

The CBA WG also looked at cost-benefit and cost-efficiency analyses undertaken by other U.S. government agencies. The U.S. EPA produces an immense volume of regulations annually, and has used cost-benefit analysis for many years as one of several sources of information on the impacts of alternative policy choices.

Typically, the EPA provides three types of analysis:

- estimates of the net benefits (benefits minus costs) for each regulatory alternative, along with a discussion of the non-monetized effects;
- · a schedule showing when the benefits and costs would occur; and
- a cost effectiveness analysis of each major alternative for situations when many benefits are not easily monetized or when the governing statute specifies the regulatory objectives

The cost portion of EPA's analyses normally includes:

- estimating the expenditures needed to comply with new regulations (e.g., to install contaminant removal technologies); and
- determining the market effects of these expenditures (e.g., the cost increase to the household water bill).

The <u>benefits</u> portion of the analyses generally focuses on the effects of reducing exposure to conteminants, including effects on human health and the environment.

The EPA is authorized to issue regulations under several different acts of Gorgress, among them:

- · Clean Air Act;
- Clean Water Act;
- Safe Drinking Water Acti
- Toxic Substances Control Act;
- Resource Conservation and Recovery Act;
- + Comprehensive Environmental Response, Compensation and Liability Act;
- Superfund Amendments and Reauthonization Act;
- Federal Insecticide, Fungicide, and Rodenticide Act:
- Food Quality and Protection Act;
- Pollution Prevention Act; and
- Atomic Energy Act and its amendments, including the Uranium Mill Tailings Radiation Control Act.

Lost-Benefil Analysis WG Hinal Report

Each of these laws includes statutory obligations to conduct economic analyses. Some of them give EPA relatively broad flexibility to use economic criteria in the decision-making process; whereas, others circumscribe EPA's use of economic analysis.

In order to devalog <u>quantifiable</u> estimates for benefits and costs in compliance with these different statutory obligations, EPA has needed to develop many "non-economic" analyses, such as methods for evaluating dose-response relationships for critical and hononitical effects. The benefits of regulatory action are in some cases required to be reflected in improvements in human welfare or, equivalently, the avoided damages or losses in welfare that humans experience in the absence of regulatory action. These and similar important categories of benefits are often considered in the analyses <u>qualitatively</u>, due to the difficulty of quantification and data availability.

It should be noted here that EPA has been criticized by the <u>Government Accountability Office</u>: (GAC) in the past for its failure to monetize benefits. The GAC indicated that, without monetization, a study probably can provide no more than a "cost-effectiveness" scenario – although that may be adequate in some cases (especially where there is little statutory flexibility).

(In light of the many Acts, Executive Orders, and other policies with which EPA must comply in developing its regulations, EPA has posted on-line extensive guidance for developing and performing regulatory cost-benefit impact analyses.)

6. The U.S. Food Industry: Hazard Analysis and Critical Control Points (HACCP)

The food industry in the U.S. is regulated for the most part by the Food and Drug. Administration (FDA) and the Department of Agriculture (USDA), and portions of the Industry are required by these agencies to implement "Hazard Analysis and Critical Control Points," an HACCP. HACCP is a systematic preventive approach to food safety and biological, chemical, and physical hazards in production processes that can cause the finished product to be unsafe, and the program includes the development of measurements to reduce these risks to a safe level

Similar to SMS, the HACCP is referred to as "the prevention of hazards" rather than the inspection of finished products. The HACCP system can be used at all stages of the food chain, from food production and preparation processes to packaging, distribution, etc. HACCP emphasizes control of the process as far upstream in the processing system as possible by using operator control and monitoring at critical control points. As such, HACCP "enhances the responsibility of producers and processors in quality and safety assurance."

Meat HACCP systems are regulated by the USDA, while seafood and juice are regulated by the FDA. (The use of HACCP is currently voluntary in other food industries.)

Lost-Benefil Analysis WG Final Report

Recently, HACCP has been increasingly applied to industries other than food, such as cosmetics and pharmaceuticals. This method, which in effect seeks to "plan out" unsafe practices based on science and risk, differs from traditional "produce and sort" quality control methods that do nothing to prevent hazards from occurring and must identify them at the end of the process.

Based on risk-assessment, HACCP allows both industry and government to allocate their resources efficiently in establishing and auditing sale fond production practice. While the application of HACCP systems has been recognized to aid inspection by regulatory authorities and promote international (rade by increasing confidence in food safety, those benefits have not been quantified.

7. Office of Management and Budget (OMB) Guidance on Regulatory Analysis

The Office of Management and Budget's *Circular* A-4 provides guidance to the Federal executive agencies on the development of regulatory analysis as required under Section 6(a)(3)(c) of Executive Drifer12866, "Regulatory Planning and Review," the Regulatory Right-to-Know Act, and a variety of related authorities.

Circular A-4 emphasizes that cost-benefit analysis is the primary lool used for regulatory analyses:

"Where all benefits and costs can be quantified and expressed in monetary units, benefit-cost analysis provides decision makers with a clear indication of the most efficient alternative, that is, the alternative that generates the largest net benefits to society."

However, it also acknowledges that it will not always be possible to express in monetary units all of the important benefits and costs of a rule. In this case, OMB emphasizes

"When it is not [possible to quantify costs/benefits], the most efficient alternative will not necessarily be the ane with the largest quantified and monetized net-henefit estimate. In such cases, you should exercise professional judgment in determining how important the non-quantified benefits or costs may be in the context of the overall analysis. If the nonquantified benefits and costs are likely to be important, you should carry out a threshold analysis to evaluate their significance. Threshold or breakeven analysis answers the question, 'How small could the value of the nonquantified benefits be (or how large would the value of the nonquantified benefits be (or how large would the value of the nonquantified benefits be (or how large would the value of the nonquantified benefits be (or how large would the value of the nonquantified benefits, you should indicate, where possible, which non-quantified effects are most important and why.

Cost-Benefit Analysis WG Final Report

Thus, OMB recognizes that some important benefits and costs may be inherently too difficult to quantity or monetize, given current data and methods. In those cases, Circular A-4 provides lengthy instructions to the agencies on analyzing them qualitatively – presenting a description of the unquantified effects, discussing the strengths and limitations of the qualitative information, and providing a clear explanation of the rationale behind the choice.

8. "Understanding the Nature of Past Corrective Actions and How Safety Management Systems Could Change the Scope of Future Corrective Actions in Aviation," Center for Aviation Safety Research (CASR), Parks College of Engineering, Aviation and Technology, Saint Louis University: March 2013

As stated by the CASR team, "This study was initiated to better understand how implementation and maturation of Safety Management Systems (SMS), particularly at design and manufacturing (D&M) organizations, might change the scope and nature of future AD corrective actions. Considering the already robust safety record of the aviation industry over the last ten years, some D&M industry representatives have questioned the value in pursuing SMS beyond existing levels. Since each AD involves unplanned costs, complexities, and other variables that must be managed while maintaining the already, this study was considered a first step in exploring how implementation and maturation of SMS could benefit the industry in broader terms by reducing the necessity for corrective actions."

The CASR team recommends, "...the FAA consider how industry can gain further insight into improving system safety through review of AD data. A pilot activity may be appropriate to see how FAA and industry could support, and benefit from, regular review of this data at a system level. Consideration should be given to looking at AD follow-up actions to ensure that appropriate updates are made to various controls (processes, practices, standards) to capture system-level learning that could preclude future events."

Additionally, "a large percentage of the ADs (48%) were considered revision to existing risk controls and were associated with one of six different descriptors related to the AD corrective action. This suggests that existing industry practices are recognizing areas that must be addressed for safe operations. These cases also illustrate that the existing risk controls were not sufficient and the AD action was necessary to achieve an acceptable level of safety. The industry may want to review the effectiveness of existing risk controls and ensure effective safety performance measures are in place. Improved risk controls and safety performance monitoring may significantly reduce the number of AD actions in the future. Thus, these improvements may induce costs and the potential for unintended consequences associated with unplaimed and complex AD compliance actions. The D&M industry and the FAA need to further explore SMS."

After reviewing this study, the CBA WG concluded that if SMS were to require a cost benefit analysis independent of a design organization, this study would be an adequate point of reference. The study recommends additional research, however it is a good indicator of how SMS may prevent unsafe events in the future.

Loct-senefil Analysis WG Hinal Report

Page | 31)

RESEARCH CONCLUSIONS:

Cost-benefit analyses addressing other than quantified itoms are notther unusual nor untested. While examples are not abundant, these that do exist provide ampla informative intight as to how such analyses can be used appropriately to assess the impact of key intangible:

Lost-seven Masyris WG Hval Report

Page (3)

V.d. COLLECTION AND ANALYZING THE DATA

During this phase of the process, the CBA WG worked actively with ARC participants as well as outside parties to understand the potential impacts to small and large organizations. Data was collected through a series of discussions, meetings, surveys, presentations and workshops. Data collection was a combination of formal and informal methods. The formal methods of collecting data were through three primary avenues.

- 1. Part 21/5MS ARC face-to-face meetings,
- 2. 2013 Modification and Replacement Parts Association (MARPA) conference.
- Distribution of the Cost Benefit Survey for Large Organizations to seven design and manufacturing companies.

Lost-Benefit Analysis WG Hnal Report

Page 1 3

V.d.i. COST BENEFIT SURVEY OF SMALL ORGANIZATIONS

The small company survey was developed by the CBA WG and delivered by representatives of the CBA WG to small business members and other interested names at the 2013 Modification and Replacement Parts Association (MARPA) conference (October 73-25, 2013, Las Vegas, NV). Seven companies responded to the survey request: six manufacturers and one FAA designee company. The purpose of the survey was to obtain industry cost-benefit estimates of the effect of three possible options being considered by the CBA WG for those small companies deemed "below the threshoid" at which larger companies would be required to implement a Design Organization system as well as a mandatory implementation of SMS. (See section <u>III.b.II</u> for more information on "below the threshold companies.") Adoption of one of the three options is intended to address the ARC tasking that any proposal for Design Organization and SMS be scalable to small businesses.

Survey Presentation

Members of the CBA WG, representing both industry and the FAA, offered a presentation to industry at the MARPA 2013 conference. The presentation was directed at small manufacturers, which male up a large segment of the Parts Manufacture Approval (PMA) industry. The presentation addressed the progress of the Part 21/SMS ARC and what recommendations could be expected from the ARC. The CBA WG further presented three options that could be applicable to those businesses deemed to be "below the threshold" at which mandatory implementation of a DO/SMS program recommended by the ARC would not be required or yould became economically infeasible based on business size.

The presentation began with an introduction to the ARC and working groups. It human described the CBA WG and explained that the purpose of the presentation was to obtain data from industry to assist in the development of new methodologies for performing orst-benefit analyses in lieu of an aircraft accident-based approach.

The presentation discussed the current method by which projects are approved. This explanation, known and understood by the attendees, involves direct FAA involvement, uses designees, relies on a compliance-based process of one-for-one findings of compliance, and allows for voluntary implementation of SMS principles.

The presentation then offered a discussion of three possible options being considered by the ARC at the time of the presentation. The <u>first option</u> presented contemplated a mandatory DQ/SMS program being necessary for all businesses regardless of size.

The second option, presented under the working title "Scaled DO," contemplated a systems based oversight system. This option is referred to as the "Accredited Organization" approach under section III.12.11 on "below the threshold companies." After presenting this option at the Annual MARPA Conference, the name "Scaled DO" was changed to "Accredited Organization."

Cost-Benefit Analysis WG Hinal Report

For the purpose of this section we will refer to this option as the Scaled DO, because this is the language used in the survey questions.

Under this approach, a business adopting the Scaled DO would be subject to reduced FAA oversight of individual showing of compliance. Instead, the business would be subject to oversight of systems to ensure compliance. The adopting business would exercise increased authority to initiate projects without FAA approval and operate under decreased design and production oversight. The adopting business would no longer be subject to the one-for-one show-find compliance model, but instead would be subject to systems oversight and reduced a periodic show-find by the FAA, depending on the complexity of a given project.

The <u>third option</u> described a modified version of the current model of oversight and approval. This model contemplated a reduction in Designated Engineering Representatives (DERs) as the FAA scales back the number of DER certifications it would renew or issue in an effort to conserve resources. The model also expects that project prioritization (known as sequencing) would be a factor in the processing of applications.

Survey Results

The survey was comprised of thirteen questions designed to gauge the effects of the three proposed options contemplated by the ARC. The respondents provided business names and points of contact for the purposes of follow up surveys if necessary. In order to maintain confidentiality, the respondent companies and contacts are not identified in this report.

The first three questions requested background information about each respondent:

1. Select your company size range:

1-10 employees	11-25 employees	25-50 employees	51-100 employees
101-500 employees	501-1000 employees	1001-1499 employees	1500+ employues

2. In what State is your compony primarily located?

3. Select all certificates that apply to your company:

TC.	PC	570
PMA	TSOA	MRA

Five of the seven respondents reported a company size of 1-10 employees, one company reported 26-50 employees, and one company reported 51-100 employees.

Five companies reported PMA certificates as applicable to their company, two companies reported PC certificates as applicable to their company; two companies reported STC as applicable to their company, one company reported TC, TSOA, and MRA were also applicable.

Lost-Benefil Analysis WG Final Report

A majority of the respondents were very small companies (1-10 employees) in the PMA business. This result was not surprising, as the MARPA conference is a conference for PMA companies and customers.

```
3. What percentage of total average annual casts is attributable to FAA cert/fication activities?
a. 0% d. 30%
b. 10% e. 50%
c. 20%
**If able, please provide the total average unnual casts attributable to FAA certification activities in
claifars. $
```

In response to this question:

- two companies reported that 10% of costs were attributable to certification;
- two companies reported that 20% of costs were attributable to certification,
- two companies reported that 30% of costs were attributable to certification, and
- one company reported that 100% of costs were attributable to certification activities.

Two companies were able to provide cost estimates in dollars: one company reporting 30% of costs attributable to certification estimated the real cost to be approximately \$1 million; the company reporting 100% of costs attributable to certification estimated the real cost to range between \$400,000 and \$1.5 million.

5 Effect of each method on company costs:

1. What percentage change in cost do you estimate would result from adopting a DO?

n. 10% or higher	e. 10%
b 20%	F2011
z 20%	g30% or lower
n. 0%	10 - 10 - 10 - 10 - 10 - 10 - 10 - 10 -

II. What percentage change in cast do you estimate would result from adopting a Scaled DO?

n.	30% or Higher .	10%
D.	20%	J20%
ik.	10%	g. 30% or lower
16	0%	

III. What percentage change in cast do you estimate would result from continuing in the Modified Current Model with Jower DERs?

1. 30% or higher	e 10%
20%	F. 20%
10%	0. 20% or lowe
1. 17	

Lost-Benefil Analysis WG Hnal Report

These three questions asked respondents to estimate to what degree they estimated their costs would change based upon the three proposals being considered by the ARC, as explained in the presentation.

When considering a change to the proposed mandatory DO method,

- four companies estimated costs would increase at least 30%;
- one company estimated an increase of 20%;
- one company estimated an increase of 10%; and
- · one company provided no response

The resulting average estimated cost increase of a mandatory DO method was approximately. 25%

When considering a change to the proposed "Scaled DO" method:

- three companies estimated costs would increase at least 30%;
- · three companies estimated costs would increase 10%; and
- one company provided no response.

The resulting average estimated cost increase of adoption of the Scaled DO method was approximately 20%.

When considering a change to the current model, designated a Modified Eurrent Model.

- are company estimated costs would increase at least 30%;
- one company estimated costs would increase 20%;
- four companies estimated no change in costs, and.
- one company provided no response.

The resulting average estimated cost increase of adoption of the Modified Current Model was approximately 8.3%.

5. If your company was able to initiate projects immediately (no sequencing queue) what do you estimate would be the effect on annual revenue?

ю.	30% or higher	x-10%
ħ,	20%	1. 20%
×.	10%	g30% or lower
d.	DW	and the second second

This question asked respondents to estimate the effect on revenues if their company was not required to wait in the sequencing or project prioritization queue to mitiate a project. The purpose of the question was to gauge the perceived benefit of privileges associated with both the Mandatory DC and Scaled DO models.

Lost-Benefil Analysis WG Final Report

Page | Sil

When considering the benefit of immediate project initiation :

- one company estimated at least a 30% increase in revenue;
- one company estimated a 20% increase in revenue;
- four companies estimated a 10% increase in revenue; and
- one company provided no response.

The resulting average estimated increase in revenue as a result of immediate project initiation (no sequencing queue delay) was approximately 15%

- 7. If your company was able to initiate projects immediately (no sequencing gueue) what do you enimate would be the effect on product time to market?
 - g. No Change
 - U. 3 months sooner
 - < 6 months sooner
- d. 9 months sooner e. 12 months or more

This question asked respondents to estimate the effect on product time to market if their company was not required to wait in the sequencing or project prioritization queue to initiate a project. The purpose of the question was to gauge the perceived benefit of privileges. associated with both the Mandatory DO and Scaled DO models.

When considering the benefit of immediate project initiation

- one company estimated it could bring products to market 9 months sconer.
- five companies estimated they could bring products to market 3 months sooner, and
- one company provided no response.

The resulting average estimated time savings as a result of immediate project initiation (no sequencing queue) was approximately 4 months.

What do you estimate would be the effect on annual costs given a 50% reduction in DERs1

a. 30% or higher	e10%
b. 20%	1-20%
< 10%	g30% or lower
d. 0%	

This question asked respondents to estimate the effect on annual costs if the number of certificated DERs was reduced by 50%. The purpose of this question was to gauge the perceived costs associated with a significant reduction in DERs associated with the Modified Current Meani.

Loct-seneril Analysis WG Hnal Report

When considering the costs of a 50% reduction in DERs.

- four respondents estimated their costs would increase at least 30%
- · one company estimated its cost would increase approximately 10%
- · one company estimated 0% change in costs; and
- · one company provided no response.

The resulting average estimated increase in costs as a result of a 50% reduction in DERs was approximately 21.6%,

9. What do you estimate would be the affect on product time to market given a 50% reduction in DERs?

- a. Wo Change
- d. 9 manths somer e. 12 months of more
- b. 3 months sooner c. 6 months sooner

Current Model

This guestion asked respondents to estimate the effect on product time to market if the number of certificated DERs was reduced by 50%. The purpose of this question was to gauge the perceived costs associated with a significant reduction in DERs associated with the Modified

When considering the effect on product time to market of a 50% reduction in DERs

- three respondents estimated that such a reduction would result in an increase time to market of at least 12 months,
- one respondent estimated an increase of approximately six months.
- one respondent estimated an increase of approximately 3 months.
- one respondent estimated no change, and
- one company provided no response

The resulting average estimated increase in time to market as a result in a 50% reduction of DERs was approximately 7.5 months.

10. Please answer the following questions related to designees and ODA

- a. Is your company an ODA? Y N
- b. Daes your company have DERs? _____Y___N
- c Do you utilize comultant DERs? _____ N

The purpose of this question was to solicit information from respondents with respect to the current source of their design data approval.

Loct-Benefit Analysis WG Hinal Report

In response:

- one respondent reported notaing an UUA.
- one company reported having its own DERs, and
- seven companies (all respondents) reporting utilizing consultant DERs.

Widespread use of consultant DERs may lend support to respondents' estimates of increased costs and times to market based on a 50% reduction in DERs, anticipated in guestions 8 and 9.

- For certification process requirements only, which of the following most closely approximates your average annual certification costs?
 - a. \$10,000 d. \$500,000 b. \$50,000 e. \$1,000,000
 - € \$100,000

and the second sec

This question asked respondents to astimate annual costs solely attributable to pertification. The responses varied broadly:

- two respondents estimated annual certification costs of approximately \$10,000;
- one respondent estimated annual costs of approximately \$50,000;
- one respondent estimated costs of approximately \$100,000;
- one respondent estimated costs of approximately \$500,000;
- pne respondent estimated costs of at least \$1 million; and
- one respondent provided no response.

The resulting average annual certification costs under the current model are approximately \$278,333.

12. Which option is most likely applicable to your business?

- a. Design Drganization
- b. Scaled Design Organization (autonomy w/ 3rd party autits)
- c Current system (reduced DERs plus sequencing process)

This question asked respondents to anticipate which of the three proposed models would best fit their business in response:

- one respondent stated that the DO model would be most applicable;
- two companies responded that the scaled DO or modified current model would be most applicable;
- three companies stated that the modified current model would be most applicable; and
- one company provided no response.

Lost-Benefit Analysis WG Hinal Report

 For the purpose of herter understanding the impact of any proposed changes, please provide FAA approved product/services annual sales (Optional)

Willingness to respond to this question was limited, with only two responses. One company, reported annual sales of approximately \$1.2 million, and one company reported annual sales of approximately \$5 million.

Lost-senefil Whatysis WG Heal Report

Page (40)

V.d.ii. COST BENEFIT SURVEY OF LARGE ORGANIZATIONS

The large company survey was designed by the CBA WG and sent by the Part 21/SMS ARC to ceven U.S. aerospace companies—including airframe manufacturers, engine manufacturers, and an appliance manufacturer. All seven companies were represented on the ARC. Useful responses were promptly received from all seven companies.

Survey Preamble

In a preamble to the survey questions, the document discussed the potential benefits of DO with SMS as a systems approach. The DC (with SMS) would be an FAA-certificated organization that would have the privileges and responsibilities for finding compliance rather than just showing compliance as is corrently required under the current designee system (QDA and/or individual designees). The current system requires, for each project, a highly redundant series of compliance. The FAA's level of project involvement (LOPI) is high

Under the DO system, the FAA would concentrate its limited resources on performance-based auditing and oversight of the manufacturer's systems, rather than on individual projects. The FAA's LOPI would be greatly reduced or even eliminated by eliminating redundant findings by the FAA in areas such as flight testing, conformity, etc. The FAA's LOPI would depend on the degree of project risk or novelty. For projects of low risk or nigh standardization, the FAA's LOPI would be low and could even be reduced to zero for standardized products of sufficiently low risk, in which case the manufacturer could initiate and undertake the project on its own. The only direct involvement of the FAA in the project would be to issue the certificate or approval upon project completion. For projects of greater risk or novelty, the FAA would have a nigher LOPI since the FAA would have to establish a certification basis, taking into account special conditions, exemptions, equivalent level of safety (ELOS) findings, strr. But with the FAA involved at just the beginning and at the end of the project level, the FAA's LOPI would still be greatly reduced compared to the current system.

Survey Results

The survey was comprised of fourteen questions designed to obtain industry cost-benefit estimates of a requirement being considered by the ABC that large firms be required to change from an QDA/individual Designee System to a DO along with implementation of SMS.

The first three questions requested background information about each respondent;

1.Parent company name:	
7. Aviation company name (e.g. subsidiary, division, business unit)	
3. Aviation comp any contact person, telephone number, and small address:	

Loct-senefil Analysis WG Hnal Report

It is important to distinguish between the <u>parent</u> company and the <u>aviation</u> company, which may be a subsidiary, division, business unit, etc. of the parent company. This was the case for all seven of the large aviation companies that were survived.

The respondents provided business names and points of contact for the purposes of follow up surveys differessary. In order to maintain confidentiality, the restrondent companies and contacts are not identified in this report.

4. Aviation company number of emuloyees (recent years):

5. Aviation company total revenues: 5

6. Aviation company revenues from sales of FAA-approved products or articles (aptional, but encouroged): V

7. Percentage of aviation company revenues derived from sales of FAA-approved products or articles:

Questions 4 through 7 were intended to measure the aviation company size. Employment and total revenues are standard size metrics. Even though all seven companies are considered large, there is an order of magnitude difference between the smallest company in the sample compared to the largest company, with employment size ranging from 7,500 to 80,000 and total revenues ranging from 53 billion to \$53 billion.

Cluestions 6 and 7 asked for aviation company revenues from sales of FAA-approved products or articles and for the same estimate as a percentage of aviation company revenues. Six of seven respondents answered one or both of these questions, so we either were given complete information for these questions or we could calculate it. Since the proposed rule applies only to aviation products that must be approved by the FAA, for our purpose, this is a better value measure of size than total aviation company revenues. The percentage varied widely from 25% to 85%, with a median value of 68%, attesting to the importance of distinguishing between the two measures.

 Your company uses which of the following? (Please check all that apply.) [ODA, Company Designees. Consultant Designees)

Six of seven companies reported using ODAs. Half of these also used company and consultant designees. Dire company reported using company designees only.

Lost-Benefil Analysis WG Final Report

3 Check all certificates/approvals that apply to your company-	
Type Certificate (TC) Amended Type Certificate (ATC)	
Supplemental Type Certificate (STC) Amenued Supplemental Type Certificate (ASTC)	
Parts Manufacturer Approval (PMA)	

Not surprisingly, large aviation companies generally apply for a wide range of certificate/approval types. The table below summarizes the responses received:

CERTIFICATES/APPROVALS	NO. OF COMPANIES
All	3
TC, ATC, STC, PMA	2
TC, ATC	2

10a What percentage of total costs of FAA-approved products, in a typical year, is attributable to commitance with FAA regulations? (Put differently, what percentage of your costs of FAA approved products would not occur if there were no FAA regulation of your products?) Please choose the answer closes to your estimate (0%, 10%, 30%, 50%, Other (explain)).

10b. "If possible, please provide a dollar amouni for the percentage indicated above: \$

We asked this question primarily as a percentage because we anticipated that companies would have difficulty in estimating a dollar amount. All companies answered the percentage question and three companies answered the dollar amount question. The purpose of this question was to get an objective, consistent measure of the burden of FAA regulation and, therefore, what scope there might be for regulatory changes, such as the proposed DO system that could reduce that burden.

Removing one outlier of 70% from the sample, the median response was 20%. Retaining the public, the median resignate was 30%.

Loct-benefit Analysis WG Hnal Report

11. The FAA anticipates that there may be <u>incremental non-recurring transition</u> casts in converting from an ODA (or individual designee system) to a DQ with SMS. The FAA anticipates that large firms will have two years to convert from their current system to a DQ with SMS.

Please provide your estimate of these transition costs, if any. (Unless you state otherwise, the FAA will usuame these costs would be sprend equally over the two year conversion period.) 5.____

12. The FAA unlicipates that there may be <u>incremental returning</u> costs and banefits for a DO system with SMS, and there could be offertting savings in mourning costs. As noted above, the FAA anticipates that cost savings could result from the DO privileges that eliminate redundant findings (conformity, testing, etc.).

In answering the following question please consider the current elements of SMS that your company has already implemented.

- a. Please provide an estimate of your changes in recurring costs as a result of changing from on ODA (or individual designee system) to a DO with SMS. 5 ______
- b. Please provide an estimate of your changes in recurring benefits (cost savings) as a result of changing from an ODA (or individual designee system) to a DO with SMS. \$______

These are the key questions of the survey, as they address the issue of whether or not the proposed DO system is cost-beneficial. Question 11 asked for estimates of the incremental non-recurring transition costs in moving from an ODA (or individual designee system to the DO system. Question 12 asked for estimates of incremental recurring costs and benefits of the DO system.

The WG was somewhat surprised to learn that the large companies considered the proposed DD system to be far from cost-beneficial. Two companies provided no cost-benefit estimates as they stated that estimates were "difficult to estimate at this time" or that estimates were "too premature to determine." One company's responses were large outliers, so just four of the responses were incorporated in the analysis.

For these companies the median response for question 13—non-recurring transition costs was \$2 million, but with one estimate at a very high \$30 million.

For Question 12, the four responding companies found recurring costs to be much higher than recurring benefits. The ratio of costs/benef is for the four companies were.

- Company 2: 1-2/3
- Company 3: 1-3/4
- Company 4: 2
- Company 5: 5

This gave a median response of approximately 2 to 1, costs to benefit.

Lost-Benefil Analysis WG Hinal Report

Discussion with ARC company representatives suggest that possible reasons in: this result are that the large company representatives believe that

- The DD system provides no additional benefits beyond what is provided by the ODA, or that will be provided when the ODA is fully matured; or
- the DO proposal is premature since the DDA has not fully matured; or
- the DO system is far from cost-beneficial now, but could be cost-beneficial after other companies have adopted it and worked out the implementation.

Additionally, informal discussions with company representatives after the survey was completed indicate that the companies did not sufficiently focus on their potential benefits in responding to the survey. This finding auggests that greater attention must be given to potential benefits in future surveys.

L3. What are the most important privileges that you anticipate from becoming a DO? If possible please provide estimates of your anticipated cost savings from these privileges.

With the exception of the outlier company, all companies expected some benefits from the DO system. The EBA WG expected companies to enswer this guestion with reference to the many possible redundancy eliminations discussed in the survey preamble. However, just two companies did soll both cited elimination of redundant conformity and one also cited elimination of redundant aircraft flight testing. The other four companies responded in generalities, citing "Systemic/centralized oversight,""Work more efficiently," etc. One company stated that "Over the years, we have adopted various delegation structures without prohibitive burdens to our business. In fact, many were accomplished with recognition of a benefit to husiness operators "

These results goint again to the necessity, already noted, of focusing greater attention on potential benefits in future surveys.

14. Under the several month	current regulatory framework, FAA sequencing may force a delay in applicant projects from to to over a year. The FAA anticipates that sequencing would be eliminated for DOs.
Please chuose	the astimute closest to the overage direct delay to your projects caused by current FAA
a farmer	0 months
	a months
	6 months
	9 manths
-	12 months
	More than 12 months

Loct-senefil Whatysis WG Hinal Report

Four of the seven companies reported delays caused by FAA sequencing of 3 to 6 months. One company reported a recurring annual benefit of reduced delay under the DC system of \$3 million. The cost of delay and the corresponding potential benefit of reduced delay under the DC system are important subjects for further analysis:

Cost-Benefit whatysis WG Final Report

Page

V.d.iii. COST BENEFIT ANALYSIS OF SMS

A cost benefit analysis of SMS was not specifically performed. The CBA WG chose not to perform an independent analysis of SMS for two primary reasons:

- The ARC came to an agreement (har SMS would be an integrated piece of a Design Organization: Therefore, breaking out SMS as an independent entity was not preferred by ARC members submitting cast and benefit data.
- SMS is intended to be an integrated part of a company business process and not a standalone "layer" or tasking. Given the difficulties in costing out process-oriented inquirements, SMS was not looked at independently.

However, even though follow on activity needs to be performed in this area, the CBA WG does perceive there to be a benefit in implementing SMS, with or without a DO. If SMS is not, regulated by the FAA when required by other countries or ICAO, companies interested in doing business outside of the U.S. will need to meet SMS requirements for each country in which may choose to do business. This could create multiple and duplicative Safety Management Systems for companies doing international business. Please refer to section IV for additional details on International Harmonization.

Lost-Benefil Analysis WG Hnal Report

VI. LESSONS LEARNED

VI.a. ROLE OF A CBA WG IN ARC ACTIVITIES

1. Establishing a Cost-Benefit Analysis Working Group was a beneficial step.

Cost benefit analyses (CBA) have traditionally been an afterthought in the course of previous FAA Aviation Rulemaking Committee (ARC) activities. The FAA normally takes the action to develop a regulatory analysis only during a formal rulemaking process: usually long after the ARC's final recommendations are submitted to the FAA.

The Part 21/SMS ARC took steps to break with tradition by making CBA a part of the ARC from the beginning and throughout its activity. For the first time under any FAA advisory committee a CBA WG was chartered and established to play a fundamental role in the development of the ARC's recommendations. As a result, the need for a CBA was recognized by the ARC from the time the ARC was iniliated.

The CBA WG recommends that there should, in most cases, be a role for such a group in future ARCs.

2. CBA WG membership should be drawn from the full ARC.

In course of its work, the CBA WG was directed to stay connected to the other three Working Groups to facilitate inclusion of their thinking and recommended actions into a methodology for assessing costs and benefits.

Future ARCs would benefit if CBA is required not only to be considered early on, but to be required as part of the charter of each Working Group that is established under an ARC.

On the other hand, if a separate CBA WG must be established, its membership should be drawn from the full ARC. It is important that a CBA WG be significantly informed of the ARC's direction throughout the process in order to be fully informed in its final recommendations; an integrated and assimilated connection with the full ARC will ensure this.

3. CBA WG activity should be phased in-

The introduction of a separate CBA WG should be phased in over the course of the ARC. While CBA itself should be considered by the ARC from the beginning, a working group focused on CBA should be initiated when the full ARC recommendations reach a high level of maturity At the onset, the CBA WG did not have a clear idea of the specific directional goals toward which the ARC was aiming. The CBA WG's first activities for several months were based on

Loct-Benefit Analysis WG Final Report

Page | A

assumptions of what the ARC would eventually recommend, some of which turned out to be erroneous.

Even though the CBA WG stayed connected with the full ARC, two decisions made by the other WGs late in the ARC process served to preclude much of the initial work of the CBA WG, requiring a significant amount of rework:

- The decision not to address smaller ("below-the-threshold") entities under the ARC's work. (This decision was made approximately half way through the course of the ARC.)
- The decisions to require SMS, but not to mandate DO, for the larger entities (This
 decision was made as part of the (Inal recommendations of the Organizational WG, very
 late in the ARC process)

The cost surveys that the CBA WG originally designed and deployed to obtain necessary data based on our original assumptions, were intended to capture these very costs. While the data gathered from the two surveys are still valuable, the surveys may have been quite different (and more appropriate) had the two decisions been made early on in the course of the ABC and if the CBA WG had been more embedded in the decision-making,

In light of this experience, we recommend that future CBA WGs be initiated about a third or a half of the way into the course of the ARC, when the recommendations reach a high level of maturity. This is to ensure that that the CBA WG has a fairly certain idea of the directional goals of the ARC before attempting to target and capture costs. This approach simply recognizes the deliberative and developmental nature of the ARC process, where goals and directions evolve as the ARC members consider various approaches. Because CBA is so dependent upon analyzing the impact of specifically proposed requirements, waiting for some form of consensus on an ARC approach could save some wasted effort in the future.

4. An FAA economist must be a member of the WG.

It is essential to have an FAA economist participate as a regular member of the CBA WG. While the CBA WG was encouraged to think "out of the box" for new ways to perform quantitative and qualitative analyses, having an FAA economist on the team brought a level of discipline and reality to our analyses. In addition, an FAA economist also brings detailed knowledge regarding analytical methodologies that will be acceptable to the FAA and OMB. This is of immense value in developing CBA data that can actually be used by the FAA in the rulemaking process, hopefully shortening the time required to prepare a proposed rule for publication.

Lost-Benefil Analysis WG Final Report

VII. CONCLUSION

In conclusion, the CBA WG found that overall change is needed for both the industry and for the FAA. Based on discussion and statistics, there is a uniform concern that the current certification system is not sustainable given industry growth and the FAA change in resources and budgetary constraints of recent years. The CBA WG cannot, within its limited scope, collectively conclude that a DO with the inclusion of SMS is the appropriate change; however, it is becoming more and more apparent that a major change to the current certification system is needed. This conclusion is a combination of statistical data, quantitative and qualitative analysis, and assumptions all gathered by the CBA WG.

The constraints and advantages of the CBA WG led to the following findings:

- A Design Organization that is recommended as voluntary does not require a cost benefit analysis. If the efforts for implementation are voluntary then the costs incurred are also voluntary.
- The cost vs. benefit of implementing SMS without a Design Organization must be further
 researched. Given the late decision by the Organization Working Group to recommend DO
 as optional, the CBA WG did not have sufficient time to gather data on SMS independent of
 a DO.

Of the data gathered from <u>small organizations</u> pertaining to implementing mandatory DO with SMS, a Scaled DO, or Modified Current Model, the following key findings were made:

- The 2013 MARPA Conference was extremely useful in the fact that small business does not always have the resources to participate full time on Aviation Rulemaking Committees.
 Holding a workshop and distributing a survey during the MARPA conference proved that small business does have a strong interest in proposed changes to certification.
 Additionally, there is a common feeling that change to the current system is needed.
- The resulting average estimated cost increase of adoption of:
 - DO with the inclusion of SMS was approximately 25%; Scaled DO (Arcredited Organization) was approximately 20%; and Modified Current Model was approximately \$.3%.
- The data also revealed that the resulting average estimated increase in revenue as a result
 of immediate project initiation (no sequencing queue delay) was approximately 15%. This is
 a significant benefit that may compensate for cost of a DO with SMS or the Scaled DO
 (Accredited Organization).

Cost-Benefit Analysis WG Hinal Report

Page | SI)

Respondents estimated the effect on revenues if their company was not required to wait in
the sequencing or project prioritization queue to initiate a project. The purpose of the
question was to gauge the perceived benefit of privileges associated with both the
Mandatory DD and Scaled DD models. The resulting average estimated time savings as a
result of immediate project initiation (no sequencing queue) was approximately 4 months.

Of the data gathered from <u>large organizations</u> pertaining to implementing mandatory DO with SMS, the following key findings were made:

- The large company survey responses considered a mandatory DO system to be far from
 cost-beneficial. Two companies provided no cost-benefit estimates, as they stated that
 estimates were "difficult to estimate at this time" or that estimates were "too premature to
 determine." One company's responses were large outliers, so just four useful responses
 were received. This gave a median response of approximately 2 to 1, costs to benefits.
- Discussion with ARC company representatives suggests that possible reasons for this result are that the large company representatives believe that:
 - the DO system provides no additional benefits beyond what is provided by the DDA, or that will be provided when ODA is "fully matured," or

the DO proposal is premature since ODA has not fully matured, or-

- The DO system is far from cost-beneficial now, but could be cost-beneficial after other companies have adopted it and worked out issues with implementation.
- Additionally, informal discussions with company representatives after the survey was
 completed indicated that the companies did not sufficiently focus on their potential
 benefits in responding to the survey. This finding suggests that greater attention must be
 given to potential benefits in future surveys.

Admitional key findings of the CBA WG include:

- Manufacturers had difficulty articulating baseline certification costs. Part 21 is a process
 oriented rule, therefore, this made for a very broad and difficult rule to cost out. The cost
 was calculated in a more general formal and described mostly in percentages by the
 majority of participants.
- Large and small manufacturers had difficulty articularing benefits. Although the benefits
 were understood by participants from a conceptual level, most participants were unable to
 express benefits in monetary values. Therefore, a benefits survey is strongly encouraged as
 a follow-on activity to this ARC and to future fulemaking effort:

Lost-Benefil Analysis WG Final Report

Page | SI

Lastly, in the event a formal rulemaking project takes place the formal cost benefit analysis should consider the following recommendations to the traditional process.

- Consider a separate survey to gather benefit data. A key finding was that respondents
 had significant difficulty articulating benefits. Gathering data from multiple benefit
 questions can provide adequate data to calculate efficiencies. Relying on the industry te
 calculate this information is not a viable solution. In the case of SMS and changes to
 part Z1, process oriented requirements are far too broad to calculate in one question.
- When benefits were described in a form of a question respondents were able to better estimate cost savings.
- Efficiencies can be calculated through a number of equations commonly used in industry and by other agencies. The most simple of these includes the following:

Efficiency = Expense/Revenue

This is one simple example of a possible solution to calculating efficiencies in typical business operations.

- Efficiencies must be broken out into tangible items whenever possible. In areas where
 this is not possible a qualitative assessment is acceptable to DMB according to DMB
 Circular A4.
- The CBA WG did not look at SMS independent of a DQ, however, a common response by the industry was to consider the following:

 Identify industry cost of maintaining and satisfying multiple SMSs compared with

A single SMS accepted internationally.

Lost-Benefit Analysis WG Hinal Report

Page | S

REFERENCES

- 1 Aerospace Industries Association (AIA), "2013 Year End Review and Forecast," 2013.
- "Aircraft Certification: Limited Progress on Developing International Design Standards," U.S. Government Accountability Office; RCED-92-179: Published: Aug 20, 1992. Publicly Released: Sep 21, 1992. Excerpted from highlights.
- Deloitte, "The Aerospace and Defense Industry in the U.S.) A Financial and Economic Impact Study," March 2012.
- European Aviation Safety Agency, Notice of Proposed Amendment (NPA) 2013-01 (A), "Embodiment of Safety Management System (SMS) requirements into Commission Regulation (EC) No 2042/2003 RMT 0251 (MDM.055)," January 2013. Available at, http://easa.europa.eu/document-library/notices-of-proposed-amendments/npa-2013-01
- Federal Aviation Administration, "Establishment of Organization Designation Authorization Program," Final Rule; Amendment Nos. 21–86, 121–311, 135–97, 145– 23, and 183–12; Federal Register publication (70 FR 59932, October 13, 2005).
- E. Federal Aviation Administration, "The Business Case for the Next Generation Air Transportation System – FY 2013;" Report; August 2012. Available at: http://www.faa.gov/nextgen/media/Business_Case_for_NextGen_2013.pdf
- Federal Aviation Administration, "The economic impact of civil aviation on the U.S. economy," August 2011.
- "Improving International Validation Programs: Reliance on Data-Driven Requirements," ICAO RASG-APAC/1 - WP/22, Noumea, New Caledonia, 10 - 11 October 2011, Agenda Item 4, Member State Presentation by the United States.
- Lercel, Damon; Steckel, Richard; Mondello, Suzanne; Carr, Eddie; Patanka, Mano). "Aviation Safety Management Systems Return on Investment Study," Center for Aviation Safety Research (CASR), Parks College of Engineering, Aviation and Technology, Saint Louis University; February 2011. Available at: <u>http://parks.slu.edu/research/centers-labs-facilities/CASR/research-papers/aviationsafety-management-systems-rol-study</u>
- 10. U.S. Census, "Annual Survey of Manufactures: Geographic Area Statistics," 2007-2011.

Cost-Benefil Analysis WG Hinal Report

- U.S. Department of Transportation, Bureau of Transportation Statistics (BTS), T-100 Segment, System revenue ton-miles.
- U.S. Department of Transportation, Bureau of Transportation Statistics (BTS), "New Quarterly Statistics Detail Industries' Economic Performance," April 2014. Available at: http://bea.gov/newsreleases/industries/
- U.S. Food and Drug Administration (FDA), "Hazard Analysis and Critical Control Points (HACCP) Guidelines," August 1997. Available at: http://www.fda.gov/Food/GuidanceRegulation/HACCP/ucm2006801.htm
- U.S. Office of Management and Budget, "Circular A-4, Regulatory Analysis," September 17, 2003. Available at: http://www.whitehouse.gov/omb/circulars =004 a-4

Cost-Benefit Analysis WG Final Report

Page | S

APPENDIX A

Part 21/SMS Cost-Benefit Working Group Cost Survey of Large Companies

Cost of Potential Rule Changes to Aircraft Certification Process

You are receiving this survey because your company is a large aviation design and manufacturing company (D&M) that may be affected by the FAA's planned rulemaking to streamline part 21 certification requirements. The intent of this rulemaking activity is to facilitate shifting DAHs to a systems approach similar to that used for production approval holder requirements. The FAA naticipates that this effort will most likely replace the current Organization Designation Authorization (DDA) program (or individual designee system) with a new Design Organization (DO) program, which includes implementation of a Safety Management System (SMS).

We are reaching out to you to obtain information that will help us conduct a required analysis to determine the costs and benefits associated with potential regulatory changes. Please be aware that all information provided will be held as strictly <u>confidential</u> and will be <u>de-identified</u> when used in our analysis.

Your cooperation in this activity is very much appreciated

Background

In late 2012, the FAA chartered the Part 21/Safety Management System (SMS) Aviation Rulemaking Committee (ARC). This ARC was tasked with recommanding improvements, to the effectiveness and efficiency of existing certification procedures, along with incorporating SMS into the design and manufacturing environments. The recommendations could likely be for rulemaking to integrate these improved certification procedures into Part 21, "Certification Procedures for Products and Parts." As with any rulemaking, the law requires the FAA to perform a cost-benefit analysis to determine the economic impact of the rule.

This cost survey has been developed by the Cost Benefit Analysis Working Group to solicit information from you to help us identify the changes in your certification costs. The proposed revision of Part 21 certification procedures would require your design and manufacturing (D&M) company to replace its current designee system (ODA and/or individual designees) with a new Design Dirganization (DO) and require the incorporation of a Safety Management System.

Lost-Benefil Analysis WG Final Report

Page | S

The DO Concept in Brief

Under this new concept, a DO (which includes 5M5) would be an FAA-certificated organization that would have the privileges and responsibilities for finding compliance rather than just showing compliance as is currently required for an ODA. This authority would be based on the DO having adequate engineering, design and testing capabilities, standards, and safeguards to ensure that the product being certificated is properly designed and manufactured, performs properly, and meets the regulations.

Greatly reduced or even eliminated would be the redundant testing and evaluations in live current model under which a series of compliance "showings" by the manufacturer are followed by FAA "findings" verifying compliance. These redundancies may include

1 Ground testing

- 2. Coupon/material testing
- 3. Component part resting

4. Flight testing

5. Noise Lesting

6. Conformity

7 Supplier qualification, evaluation, and use of test plans

8: Data change control

As a result, the FAA's level of project involvement (LOPI) would be greatly reduced. Projects with greater risk or novelty would have a higher--although still greatly reduced-LOPI since in establishing a certification basis the FAA would have to take into account special conditions, exemptions, and equivalent level of safety (ELOS) findings, etc. For less risky/less novel products the FAA's LOPI would be much lower and could even be reduced to zero for standardized products of sufficiently low risk. Reduced LOPI would allow the FAA to concentrate its limited resources on risk management and performance-based auditing and oversight.(Continued on next page.)

Lost-Benefil Analysis WG Hnal Report

Page | S)
Part 21/SMS ARC Large Business – Cost Survey

This survey requests your cost estimates of potential changes to the existing Part 21 regulation that would require the conversion of your company to a DO (as defined above) with full implementation of SMS.

For the following questions, please use current dollars for all dollar estimates. You may type your answers, creating space as necessary.

COMPANY DATA

1. Parent company name:

1 Aviation company name (e.g. subsidiary, division, business unit):

3 Aviation company contact person, telephone number, and email address:

Aviation company total revenues \$

 Aviation company revenues from sales of FAA-approved products or articles (optional, but encouraged);

\$______

 Percentage of aviation company revenues derived from sales of FAA-approved products or articles:

5_____

8. Your company uses which of the following? (Please check all that apply.)

Cost-Benefit Analysis WG Hinal Report

Page | S

Check	Designee System
-	ODA
_	Company Designees
_	Consultant Designees

9. Check all certificates/approvals that apply to your company:

Check	Type of Certificate/Approval	
	Type Certificate (TC)	
Amended Type Certificate (ATC)		
	Supplemental Type Certificate (STC)	
Amended Supplemental Type Certificate (ASI		
	Farts Manufacturer Approval (PMA)	

COST ESTIMATES

10. What percentage of total costs of FAA-approved products, in a Typical year, is attributable to compliance with FAA regulations? (Put differently, what percentage of your costs of FAA-approved products would not occur if there were no FAA legulation of your products?)

Please choose the answer closest to your estimate

Check	Percentage Cast	
_	0%	
	10%	
	20%	
	30%n	
	50%	
	Other (explain):	

"If possible, please provide a dollar amount for the percentage invicated above.

s.,

 The FAA anticipates that there may be incremental non-recurring transition costs in converting from an ODA (or ind/vidual designee system) to a DO with SMS. The FAA

Lost-Benefil Analysis WG Hnal Report

Page | S

anticipates that large firms will have two years to convert from their surrent system to a DO with SMS.

Please provide your estimate of these transition costs, if any. (Unless you state otherwise, the FAA will assume these costs would be spread equally over the two-year conversion period.)

12 The FAA anticipates that there may be incremental recurring costs and benefits for a DO system with SMS, and there could be offsetting savings in recurring costs. As noted above, the FAA anticipates that cost savings could result from the DO privileges that eliminate redundant findings (conformity, testing, etc.).

In answering the following question please consider the current elements of SMS that your company has already implemented.

a. Please provide an estimate of your changes in recurring costs as a result of changing from an ODA (or individual designce system) to a DO with SMS.

\$_____

b Please provide an estimate of your changes in recurring benefits (cost savings) as a result of changing from an ODA (or individual designee system) to a DO with 5M5.

- š.
- What are the most important privileges that you anticipate from becoming a DO? If possible please provide estimates of your anticipated cost savings from these privileges.

Proliege	East Saylogs	
	s	
	1	
Total Cost Savings	5	

Note: Your figure for Total Cost Savings should be consistent with your calculation of recurring cost vavings in gunstion 17.

Lost-Benefil Analysis WG Hinal Report

Page | SV

14 Under the current regulatory framework, FAA sequencing may force a delay in applicant projects from several months to over a year. The FAA anticipates that sequencing would be eliminated for DOs.

Please choose the estimate closest to the average direct delay to your projects caused by current FAA sequencing.

Check	Sequencing Delay
	0 months delay
	3 months delay
	6 months delay.
	9 months delay
	12 months delay
	More than 12 months delay (Please specily
	length of delay.)

15 Please identify any <u>additional</u> significant cost-benefit impacts, other than already, discussed in the preamble above or responded to in the survey.

For clarification of the survey questions, please contact Daniel Leach, Economist, FAA Office of Aviation Policy and Plans, Economic Analysis Division (APO-300), tel: 202-267-3335, e-maildaniel.leach@fma.gov. For technical questions please contact Amy Garzaro, Aerospace Enginee FAA Aircraft Certification Service, Safety Management Design and Analysis Branch (AIR-150), tel 727-559-1387, e-mail: amy.garzaro@faa.gov.

Lost-Benefit Analysis WG Hinal Report

Page (6))

APPENDIX B

Part 21/SMS Cost-Benefit Working Group Cost Survey of Small Companies

Part 21/SMS ARC Impact on Small Business -Questionnaire

Company Name (Optional)

October 23, 2013

Point of Contact Name/Phone Number if Interested in participating further.

1. Select your company size range:

1-10 employees	11-25 employees	26-50 employees	sesvolqma 001-17
101-500 employees	501-1000 employees	1001-1499 employees	1500+ employees

2. In what State is your company primarily located?

3. Select all certificates that apply to your company:

TC	PC	STC
PMA	TSQA	MRA

4. What percentage of total average annual costs is attributable to FAA certification activities?

0% b. 10% c. 30% cl 30% c. 50%

***If able, please provide the total overant annual costs attributante to FAS catterians

- activities in dollars. 6_____
- 5. Effect of each method on company costs:

What percentage change in cost do you estimate would result from adopting a DD?

Lost-senefit Analysis WG Hnal Report

Page | 61

 30% or higher 	e- 10%
B (089)	F 20%
10. 2000	g. 30% or lower
zi. 0%	

H.

What percentage change in cost do you estimate would result from adopting a scaled 1007

a. 10% of higher	₽10%€
b 20%	Fr - 20%6
£. 10%	g -10% or lower
d. 0%	

Ш. What percentage change in cost do you estimate would result from continuing in the modified current model with fewer DERs?

a. 30% ar higher	e -10%
b. 20%	V205
r. 10%	g. 30 ⁸ or lowe
d, 0%	

8. If your company was able to initiate projects immediately (no sequencing queue) what do you estimate would be the effect on annual revenue?

з.	30% or higher	e10%
h.	20%	8. 20%
c.	10%	g30% or lower
d.	0%	

7. If your company was able to initiate projects immediately (no sequencing queue) what do you estimate would be the effect on product time to market?

a. No Change		d. 9 months sooner	
b.	Lounths money	e. 12 months or mare	

c. 6 months sooner

Lost-Benefil Analysis WG Hinal Report

Page | G

8. What do you estimate would be the effect on annual costs given a 50% reduction in DERs?

a. Hoteor higher	e. 10%
b. 20%	V. 20tm
ç. 10%	830% a) howe

d, 0%

 What do you estimate would be the effect on product time to market given a 50% reduction in DERs?

- a. No Change d. 9 monitive or same
- 6 3 months sooner é: 13 months or sonner
- z. Bruonths aborrer

10. Please answer the following questions related to designees and OBA:

- a. Is your company an ODA7 Y N
- b. Does your company have DERs? y N
- 6. Do you utilize consultant DERs 7 V N

11. For certification process requirements only, which of the following most closely approximateyour average annual certification costs?

a, 510,000	d. 5500,000
b 550,000	e \$1,000,000
c: \$100.000	

12. Which option is most likely applicable to your business?

- a. Design Drganization
- b. Scaled Design Organization (autonomy w/ 1" party audits)
- c. Durrent system (reduced DERs plus sequencing process)

13. For the purpose of better understanding the impact of any proposed changes, please provide

FAA approved product/services annual sales (Optional)

Lost-Benefil Analysis WG Hinal Report

Page | 6

APPENDIX C

CBA Working Group Members

Segment	Organization	Name
Association - AIA	AIA	George Novak
Association - GAMA	GAMA	Jens Hennig
Association - MARPA	MARPA	Ryan Aggergaard
Transport	Boeing	Jill DeMarco
Systems	Honeywell	Joe Caldwell
FAA	AIR-100	Dave Hempe
FAA	AIR-150	Amy Garzaro*
FAA	APO-300	Daniel Leach
EASA	EASA	Miklos Kedves
Observer	EASA	Jan Novak
Observer	EASA	Eric Sivel
Observer	ANAC	Maria Clara

*Working Group Lead

Cost-Benefit Analysis WG Final Report

Page | 64

APPENDIX K—SM ICG FINDINGS ON SMS EQUIVALENCE



SM ICG Findings on SMS Equivalence





SM ICG Findings on Equivalence

SAPETY MANAGEMENT

This paper was prepared by the Safety Management International Group (SM ICG). The purpose of the SM ICG is to promote a common understanding of Safety Management System (SMS)/State Safety Program (SSP) principles and requirements: facilitating their application across the international aviation community.

The current core membership of the SM ICG includes the National Civil Aviation Agency (ANAC) of Brasil, the Civil Aviation Authomy of the Netherlands, the Civil Aviation Authomy of New Zeafand, the Civil Aviation Safety Authomy (CASA) of Australia, the Direction Generale de l'Aviation Civile (DGAC) in France, the European Aviation Safety Agency (EASA), the Federal Office of Civil Aviation (FOCA) of Switzerland, Japan Civil Aviation Bureau (JCAB), the United States Federal Aviation Administration (FAA) Aviation Safety Organization, Transport Canada Civil Aviation (TCCA) and the Civil Aviation Authority of United Kingdom (UK CAA). Additionally, the International Civil Aviation Organization (ICAO) is an observer to this group.

Members of the SM ICG

- Collaborate on common SMS/SSP topies of interest.
- Share lessons learned
- Encourage the progression of a harmonized SMS
- Share products with the aviation community
- Collaborate with international organizations such as ICAO and civil aviation authorities that have implemented or are implementing SMS and SSP

For hirther information regarding the SM ICG please contact:

Regine Hamelijnek, SM ICG Chair EASA 140 221 800 00 1000 regine humelijnek weasa curuna cu Jacqueline Broth TCCA (613) 952-7974 memoline brother to ecse Amer Younsess FAA, Avintion Solety (202):267-5164 amer to violation of lan pay

Carlos Eduardo Pellograno ANAC +55 21 3501 5147 carlos pelleuron ganac, zov br Wayne Jones-CASA 1617 5144 7494 congstrease area and

INTERNATIONAL COLI International Civil Aviation Organization (ICAO) standards require each state to implement a

TTY MANAG

State Safety Program (SSP) SSP requirements include mandates for States to require certain aviation product and service providers to implement Safety Management Systems (SMS) A number of Product and Service Providers (P/SPs) worldwide have expressed concern regarding implementation and acceptance processes for Safety Management Systems (SMSs). This is especially acute for P/SPs that are subject to certification, approval, or other authorizations of then products and services by multiple authorities

In many cases, SMSs will be a prerequisite for issuance of an organizational certificate. This is the case for those organizations whose activities are controlled by certification of their operational processes, such as Air Operator Certificate (AOC) holders and Approved Maintenance Organizations (AMOs). State-P/SP relationships for type certificate holders and production certificate holders, however, are more varied helween authorities. For example, some authorities certificate design organizations and some do not. In either case, the product itself is also subject to type design approval or certification.

For regulators, the problem becomes one of ensuring an equivalent level of safety when granung authorizations for other States" P/SPs without imposing excessive technical legal and administrative burdens on the companies. For P/SPs, the problem becomes one of not only being required to no through multiple authorities' acceptance and auditing processes, it also presents them with the possibility of conflicting requirements, a significant management problem.

While ICAO standards outline requirements for SMS, including a detailed SMS framework, the performance of any organizational system or process in practice depends not only on the requirements, but the way in which those requirements are implemented. For States to be confident in the equivalence of safety performance of product and service providers whose SMShave been accepted by another State, the two states will need to agree on standards, expectations for final implementation, acceptance processes, performance measurement strategies, and processes for communed oversight.

The Safety Management International Collaboration Group (SMICG) a chartered group of technical representatives of several aviation safety authorities, was asked to analyze the issue of equivalence of SMS and the implications of SMS acceptance among authorities. The group found two major components to this issue-

The first component concerns the nature of mutual or reciprocal recognition of SMS by different authorities. It was recognized that there are existing precedents and current processes for dealing, with this part of the issue including bi-lateral agreements that cover a broad range of aviation organizations, products, services, and activities. These involve a mixture of diplomatic, legal, and technical arrangements between States. Given the croup's charter and core expertise, it was the consensus of the SMICG that the latter element should be the sole focus of the group's analysis. Given that, the SMICG members also felt that existing frameworks for interaction between states were adequate. However, in all cases of bi-lateral actions, an assessment of technical equivalence is necessary



The second component further breaks down the subject of technical equivalence into five areas. SMICG members offered that there is more to technical equivalence than simply having a common set of standards. If the primary interest is in the "bottom line" performance of the products, services, or processes of the organizations whose SMS is being evaluated, additional elements must be evaluated. Having a commonly accepted framework for evaluating SMSs should make these evaluations more structured and efficient Those elements include:

- Basic Process Requirements. While not sufficient to establish equivalence, use of a common set of basic core requirements is necessary. These have been established in the various ICAO Annexes (with the current exception Annex 8).
- Implementation Expectations Each State will prepare specific expectations for processes, programs, methods, and tools related to implementing and demonstrating performance on the part of P/SPs. This is where the basic requirements are interpreted into operational definitions, documentation and record-keeping requirements, and procedures.
- 3. Acceptance Methodology. The methods that the State uses to evaluate the process design and management capability of the P/SP may vary between states. This is usually a function of the State's oversight system (Critical Element of Oversight number 6 – Licensing, Certification and Approval). The combination of specific requirements for implementation and the methodology for acceptance by the State is a large determinant of performance capability. The SMICG Documentation WG is developing an SMS evaluation tool that could be used to support this task.
- 4. Performance Measurement. States must measure performance of safety management practices in the SMS. Therefore, the methodology used by each State to measure safety performance of P/SPs is important in understanding the performance potential and status of P/SP/s certificated or otherwise approved/accepted by the State.
- 5. Continuing Oversight Policies and Methods. In order to assure the performance status of P/SPs and their SMSs, continuing oversight is essential. This is also part of a State's oversight responsibility (Critical Element of Oversight number 7 – Surveillance Obligations). Thus each State must have a good understanding and mutual confidence in the methods used to oversee P/SP's SMSs in order to establish equivalence. This will provide confidence that the P/SP is maintaining the same performance capability that was established at initial acceptance.

The SMICG's existing areas of study encompass most of these five critical elements. Therefore, the group offers to continue current projects and, where necessary, draw linkages to the above areas in order to contribute to mutual understanding of the elements of technical equivalence. Implementation of the products to be developed by the SMICG workgroups will provide a baseline for establishing technical equivalence of SMS performance.

The SMICG also recognizes that the issue of equivalence is most acute for design organizations, where differences in certification processes are most significant. The next most potentially problematic area would be in the case of AMOs, where organizational certification is nearly universal but where certification by multiple States' authorities is common. The special problems of these groups will be highlighted in SMICG products.

APPENDIX L—EASA DO APPROVAL CERTIFICATE



European Aviation Safety Agency

APPROVAL CERTIFICATE

EASA.21J.031

Pursuant to Regulations (EC) 1592/2002 and (EC) 1702/2003 and subject to the conditions specified below, the Agency hereby certifies

AIRBUS

1, Rond-Point Maurice Bellonte 31700 Blagnac Cedex France

as a DESIGN ORGANISATION

approved according to Part 21, Section A, Subpart J

CONDITIONS :

- The approval is limited to that specified in the enclosed Terms of Approval, and
- This approval requires compliance with the procedures specified in the Design Organisation Handbook ref. DOM in the latest revision, and
- This approval is valid whilst the approved Design Organisation remains in compliance with Part 21, Section A. Subpart J.
- Subject to compliance with the foregoing conditions, this approval shall remain valid until surrendered or revoked.

For the European Aviation Safety Agency,

Date of issue: 28 September 2004

Norbert LOHI Certification Director

APPENDIX M—RESPONSIBILITY OF EXPORTERS



Recommendation for the Removal of 14 C.F.R. § 21.335(b): Responsibilities of Exporters-Duration of Effectiveness of Packaging

> Submitted by the Aviation Suppliers Association 2233 Wisconsin Ave, NW, Suite 503 Washington, DC 20007

> > For more information, please contact: Jason Dickstein General Counsel (202) 347-6899

The only packaging metric related to packaging duration is found in ATA Spec 300. Spec 300 lists Entegory I. II. and III packagings to describe the effectiveness of reusable, multi-trip packaging containers. Category I packaging is usable for 100 mund-trips, Category II for 10 round-trips, and Category III for 1 round-trip. Spec 300 does not describe or contemplate a duration requirement for a single-use packaging. More importantly, most packaging used for export of aircraft parts is not designed or certified to ATA Spec 300, so it is not identified by a Spec 300 round-trip limit.

In addition to an absence of single-use packing efficacy duration metrics from packing manufacturers, no guidance material has been produced by the FAA to explain either to industry or to FAA personnel the appropriate methods for determining and stating the duration and effectiveness of packaging. The enforcement of this provision has never been considered an enforcement priority for the FAA

The provision appears to offer no safety benefits. It does not appear to serve any government interest. It is also a record coping requirement that fails to meet the requirements of the Paperwork Reduction Act (and is therefore unenforceable according to 44 U.S.C. 3512).

Conclusion

Given that Part 21 is currently in the process of a significant re-write, now is an appropriate time to remove § 21,355(b). The paragraph adds needless confusion to the regulations, is impossible to comply with from a practical perspective, and has never been an FAA enforcement priority, making it an unnecessary waste of scare FAA resources. For these reasons ASA recommends that 14 C F R § 21,335(b) be deleted from the Federal Aviation Regulations.

Aviation Suppliers Association

Page 2

Issue

14 C.F.R. § 21.335(b) states that each exporter of aircraft parts must "[p]reserve and package products and articles as necessary to protect them against corrosion and damage during transit or storage and state the duration of effectiveness of such preservation and packaging." There is no available guidance for compliance with this requirement, nor is data generally available regarding duration of effectiveness of packing. It is not reasonably possible for exporters to comply with this requirement, nor does this requirement serve any government interest. We therefore recommend that Paragraph 21.335(b) of the Federal Aviation Regulations be removed.

Discussion

Part 21 of the Federal Aviation Regulations is currently undergoing significant revisions to incorporate SMS principles into the regulations as well as to make other changes. 14 C.F.R. § 21.335(b) is an unworkable provision based on industry practices, does not reflect the actual nature of packaging for shipments, and has never been an enforcement priority of the FAA.

14 C.F.R. § 21.335(b) reads as follows:

Unless otherwise agreed to by the importing country or jurisdiction, each exporter must . . . [p]reserve and package products and articles as necessary to protect them against corrosion and damage during transit or storage and state the duration of effectiveness of such preservation and packaging

The duration and effectiveness of single-use packaging is not a metric that is typically available from packaging manufacturers. This is likely because the shipping industry has generally not faced an issue with packaging of such ephemeral effectiveness that its duration has been an issue during the course of a single-use shipment. The practical result of the reliable durability of single-use packaging is that no reliable metrics are produced by or available from packaging manufacturers, making the requirements of § 21.335(b) nearly impossible to satisfy.

Aviation Suppliers Association

Page 1

APPENDIX N—APPLICANT DECLARATION AND HOLDER OBLIGATION

[DATE]

Gilles Morin Chief Projects (AARDE) National Aircraft Certification Branch Transport Canada Civil Aviation 330 Sparks Street Tower C, 3rd Floor Ottawa, Ontario K1A 0N8

SUBJECT: ENGINE TYPE CERTIFICATION / APPLICANT DECLARATION AND UNDERTAKING TO CARRY OUT RESPONSIBILITIES PER CAR 521.33 AND 521.57

REFERENCES:

1/ Application For Transport Canada Type Certification Of The **ENGINE**

2/ Transport Canada ENGINE Issue Paper G-01, Edition 6 [CLOSED]

Dear Sir:

Type Certificate Applicant / Holder Corp. initially made Application for Transport Canada Type Certification of the **ENGINE** on [date] (Reference 1).

On behalf of **Type Certificate Applicant / Holder Corp.** and as required by CAR 521.33 and 521.57, I the undersigned, **[Program - Vice President or Director or Manager]**, declare that:

- **Type Certificate Applicant / Holder Corp** has demonstrated that the **ENGINE**, as defined by Top Engine Assembly Drawing XXXX, complies with the Type Certification Basis defined by Transport Canada **ENGINE** Issue Paper G-01 (Reference 4) and that the **ENGINE** incorporates no unsafe features;
- Findings of compliance have been made by the cognizant [organizational Delegates] and are recorded in the **ENGINE** Compliance Program, XXX, attached.
- As recorded in TCCA "ENGINE TCCA LOI Completion Statement" [RDIMS #XXXX], Transport Canada specialists' Level of Involvement has been completed for the ENGINE program.

Further, **Type Certificate Applicant / Holder Corp.** will undertake the responsibilities of a design approval document holder as specified in Canadian Aviation Regulations Part V, Subpart 21, Division VIII.

Yours truly **Type Certificate Applicant / Holder Corp.**

Program - Vice President or Director or Manager

TCCA CAR 521.57 Applicant Declaration / Holder Obligations

CAR 521.57 - Type Certificate Issued:

The design approval applicant has demonstrated compliance with the requirements in the cert. basis;

All compliance tests completed and reports reviewed / submitted / accepted or approved

The design approval applicant has made required Declarations

TCCA LOI is complete and no unresolved issues remain

The Minister (company delegates + TCCA specialists) has made findings of compliance with requirements of the Certification Basis;

The compliance program is completed for all requirements and completed / signed off (Finding of Compliance) by delegates

All required documents (e.g. Installation Manual, ICDs, AWL Limitations, ICAs) are approved

2

1

CAR 521 Required APPLICANT Declarations

(PRIOR TO ISSUE OF TYPE CERTIFICATE)

CAR 521.33(b)

APPLICANT Declaration attesting to the demonstration of compliance of the type design of the aeronautical product with its certification basis

CAR 521.57(b)

Signed undertaking by APPLICANT to carry out the responsibilities of the <u>HOLDER</u> specified in Division VIII

Conformity with Certification Basis	
521.33 An applicant for a type certificate in respect of an aeronautical product shall	
$\langle a\rangle$ demonstrate to the Minister that the aeronautical product conforms to the certification established by the Minister under section 521.30;	on basis
(b) submit to the Minister a declaration attesting to the demonstration of conformity of t aeronautical product with its certification basis;	the
$\langle c \rangle$ make available to the Minister the means by which conformity is established;	
(d) in the case of an aircraft, record the noise levels in its flight manual or in a suppleme manual using the Guidelines for the Administration of Noise Certification Documentation Attachment G of Annex 16, Volume 1 to the Convention; and	nt to that set out in
$\langle e\rangle$ submit to the Minister for approval any manuals, instructions and limitations that are by the certification basis established in respect of the aeronautical product.	required
Issuance of a Type Certificate	
521.57 (1) Subject to section 6.71 of the Act, the Minister shall issue a type certifica respect of an aeronautical product if the applicant	te in
(a) submits the declaration required under paragraph 521.33(b);	
(b) submits a signed undertaking to carry out the responsibilities specified in Division VIII	i; and
(c) meets the requirements set out in subsection (2) or (3) in respect of the category o aeronautical product.	f the

4

3

CAR 521.57(b)

Declaration – Type Certificate Holder Obligations

APPLICANT undertakes to carry out the responsibilities of a design approval document HOLDER, as set out in Division VIII of Part V Subpart 21 of the CARs, regarding:

- ✓ technical capability,
- ✓ service difficulty reporting,
- establishing a service difficulty reporting system,
- ✓ investigation of service difficulty reports,
- mandatory changes,
- ✓ transfers,
- record keeping and loss or disposal of records,
- ✓ manuals,
- ✓ instructions for continued airworthiness
- ✓ supplemental integrity instructions [Part 25 Only]

CAR 521 – Division VIII Design Approval Holder Responsibilities

- 521.351 Division VIII Responsibilities of a Design Approval Document Holder
 - 521.351 Application
 - 521.352 Technical Capability
 - 521.353 Service Difficulty Reporting
 - 521.354 Establishing a Service Difficulty Reporting System
 - 521.355 Investigation of Service Difficulty Reports
 - 521.356 Mandatory Changes
 - 521.357 Transfer
 - 521.365 Record Keeping
 - 521.366 Loss or Disposal of Records
 - S21.367 Manuals
 - 521.368 Instructions for Continued Airworthiness
 - 521.369 Supplemental Integrity Instructions

5

APPENDIX O—ACCOUNTABILITY FRAMEWORK

Accountability Framework --Applicant Statement of Compliance and FAA Discretionary Function

2013 ODA Seminar Seattle, WA September 11, 2013



What do we mean by Accountability Framework?

- Any process involving more than one entity requires dividing up responsibility for portions of the entire process
- Accountability Framework simply means holding all stakeholders accountable for their portion
- It's a philosophy substantiated in Title 49 of US Code and reflected in regulations and policy

Accountability Framework – Applicant SofC and Discretionary Function 2013 ODA Seminar – Seetle, VA, September 11 Function

The Underlying Premise Is ...



Governmental Discretionary Function is Not New



Accountability Framework

Applicants for a design approval have a regulatory obligation to:

- . Use means of compliance acceptable to the FAA
- Submit data that "Snows" that their designs are compliant

Design Approval Holders have an ongoing regulatory obligation to:

- Maintain compliant designs with no unsafe features
- · Report product failures, malfunctions, and defects

FAA promotes (not ensures) Aviation Safety by ...

- Issuing regulations
- Specifying the certification basis
- Providing guidance regarding acceptable means of compliance
- Overseeing compliance
- · Taking enforcement actions as necessary
- · Issuing certificates and approvals; and
- · Mandating corrective action as necessary

Wight should be an international and an international and a final should be a series of the series o

Accountability Framework in Part 21

An Applicant is required to...

- Conduct all inspections and tests to show compliance of design and product (14 CFR 21.33(b))
- Make all flight tests that the Administrator linds necessary (14 GFR 21.35(b)).
- Submit all data showing compliance (14 CFR 21.21(b)).
- Provide a statement certifying compliance (14 CFR 21 20(b))

The Administrator...

- Finds upon examination of the type design that applicable requirements have been met (14 CFR 21.21(b)1).
- Makes any inspections and tests necessary to determine compliance of applicant's design and product (14 CFR 21.33(a)).

Federal Aviation Asimmetration



- + New 14 CFR Part 21 Effective April 16, 2011
- "an applicant ... must provide a statement certifying that the applicant has complied with the [applicable] requirements"
 - § 21.20(b) for TC/STC and amended TC/STC applicants
 - § 21.97(a)(3) for major change to type design applicants
 - § 21.303(a)(5) PMA applicants
- "Compliance Listing" required
 - § 21.20(a) and § 21.97(a)(2)



Applicant's Certifying Statement of Compliance

- Applicant submits statement to ODA Administrator at end of project
 - Signed by individual "having authority over the certification, and legally representing the applicant..."
- Guidance
 - Advisory Circular, AC 21-51, Published 9/27/2011
 - Will be addressed in Order 8110.4D



FAA Discretionary Function is...

... the ability to make decisions by applying judgment and making reasonable choices as we perform our work within the bounds of the statutes, regulations, and directives (i.e.; Orders & Notices) that prescribe how the FAA will perform its work

FAA Level of Involvement Decisions

Federal Aviation Administration

Accountability Framework – Applicant SofC and Discretionary Function 2013 ODA Seminar – Sastle, WA, September 11



APPENDIX P—USE OF APPLICANT SHOWINGS ON ODA PROJECTS

Use of Applicant Showings on ODA Projects

2013 ODA Seminar Seattle, WA September 11, 2013



Overview

- Evolution/Background
- Today/Tomorrow
- 8110.4D Approach
- ODA Approach



"Applicant Showings"

- A means to implement FAA discretion in ODA projects
- Align ODA processes with
 - Accountability framework
 - 8110.4D Approach
- Recognition of applicant substantiation activities rather than "ODA unit" involvement
 - E.g. test, inspection, analysis, etc.







ODA Today without RBRT/SMS Principles



Attributes Today Without RBRT/SMS Principles

- Mainly product focused
- Typically every action by applicant requires action by ODA -- No FAA discretionary function decision making
- No Risk Management or reliance on applicant showings
- Many applicants still "throw compliance over the fence"
 - 21.20(a) Statement of Compliance attempts to change that



Possible Future with RBRT/SMS Principles



- Not required by Part 21, but supports FAA recognition of 'show only' processes

Attributes with RBRT/SMS Principles

- · Greater system focus
- Every action by applicant does not require action by ODA
- Incorporates risk management and reliance on applicant showings
- Compliance built in throughout design phase

S	ummary of Key Principles
·	Compliance should be the goal from the beginning of the process and not something inspected in at the end
÷	Compliance by process can mitigate higher risk areas
4	Discretionary Function is only exercised by FAA (not designees)
ł	ODA holder can propose and OMT can recognize "applicant showing" processes
0 55	Feddral Aviation

8110.4D Approach

- Goes back to these basic part 21 principles (Accountability Framework) and incorporates FAA discretion.
 - The applicant is responsible to make sure the substantiation data is correct (show compliance).
 - FAA makes a decision what inspections and tests it will witness or conduct and what data it will review.
- FAA involvement is not required for the applicant to show compliance and fulfill their regulatory obligations.



8110.4D Approach

- The FAA "finds" by "spot checking" (Discretion) using FAA authority to:
 - Witness applicant tests or inspections.
 - Conduct FAA tests and inspections.
 - Review substantiating data submitted by the applicant.
 - Review the type design data submitted by the applicant.
- FAA may choose to rely on applicant showing and statement of compliance.

Fectore: Avialian

Rev D - FAA "Level of Involvement"

Implementation via:

- direct FAA involvement
- use of designees allocated based on risk
- FAA may choose to rely on applicant showing and statement of compliance.



ODA Approach

- Available under TC, STC, and PMA
- Reliance on applicant showing activity may be established for individual activities:
 - Compliance data review
 - Test witnessing
 - Conformity inspections
 - Flight Test



- Procedures manual may address applicant showings activity agreed to by the OMT
- Recognition of applicant showings based on risk
 - Airworthiness Standard risk
 - Complexity of substantiation activity
 - Established/robust procedures for higher risk activity documented in ODA manual

Fectors & Hallon

ODA Approach

- Risk tools not yet established for ODA projects
 - Use of RBRT being looked at in 2013
- + OMT judgment/discretion to accept
- SMS/QMS not required to implement applicant showings

ODA Approach	
Use of applicant showi "applicant" responsibil Inspection, tests Establish compliance	ngs does not change ities
- Data review	a second and and
 Can result in compliant UM involvement 	ce findings with less
 Engineering UM approval conformity or test-witness 	without requiring UM

ODA Approach-Data Review

- OMT requires UMs to review and determine compliance unless safety risk is acceptable based on
 - Criticality of risk to the product
 - Applicant experience and history
 - Maturity of compliance assurance processes



Oversight of Applicant Showings

- · Oversight of applicant showings
 - Not addressed in ODA or any other order
 - Will be addressed in future FAA policy

